IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:	Chapter 11
HCR MANORCARE, INC.,1	Case No. 18-10467 (KG)
Debtor.	

DISCLOSURE STATEMENT FOR THE FIRST AMENDED CHAPTER 11 PLAN OF REORGANIZATION FOR HCR MANORCARE, INC.

SIDLEY AUSTIN LLP YOUNG CONAWAY STARGATT & TAYLOR, LLP

Larry J. Nyhan

Robert S. Brady (No. 2847)

Dennis M. Twomey

Edmon L. Morton (No. 3856)

William A. Evanoff

Allison Ross Stromberg

Tara C. Pakrouh (No. 6192)

Matthew E. Linder Rodney Square

One South Dearborn Street
Chicago, Illinois 60603
Telephone: (312) 853-7000

1000 North King Street
Wilmington, Delaware 19801
Telephone: (302) 571-6600

Facsimile: (312) 853-7000 Facsimile: (312) 853-7036 Facsimile: (302) 571-1253

Attorneys for the Debtor and Debtor in Possession

Dated: May 18, 2018

01:23221416.1

¹ The last four digits of the Debtor's federal tax identification number are 9231. The Debtor's mailing address is 333 N. Summit St., Toledo, OH 43604.

IMPORTANT INFORMATION ABOUT THIS DISCLOSURE STATEMENT

On March 4, 2018 (the "Petition Date"), HCR ManorCare, Inc. (the "Debtor") filed a voluntary petition for relief (the "Chapter 11 Case") under chapter 11 of title 11 of the United States Code (the "Bankruptcy Code") in the United States Bankruptcy Court for the District of Delaware (the "Bankruptcy Court"). The Debtor is providing you this document and the accompanying materials (this "Disclosure Statement") because you may be a creditor or interest holder entitled to vote to accept or reject the First Amended Chapter 11 Plan of Reorganization for HCR ManorCare, Inc. (including all exhibits and schedules attached thereto, and as may be amended, altered, modified or supplemented from time to time, the "Plan"), which is attached hereto as Exhibit A.

On May 18, 2018, the Bankruptcy Court approved this Disclosure Statement in accordance with section 1125 of the Bankruptcy Code as containing "adequate information" to enable a hypothetical, reasonable investor typical of Holders of Claims against, or Interests in, the Debtor to make an informed judgment as to whether to accept or reject the Plan, and authorized its use in connection with the solicitation of votes with respect to the Plan.

SPECIAL NOTICE REGARDING FEDERAL AND STATE SECURITIES LAWS

The securities to be issued upon the Effective Date will not have been the subject of a registration statement filed with the United States Securities and Exchange Commission (the "SEC") under the United States Securities Act of 1933 (as amended, the "Securities Act") or any securities regulatory authority of any state under any state securities law ("Blue Sky Laws"). The Plan has not been approved or disapproved by the SEC or any state regulatory authority, and neither the SEC nor any state regulatory authority has passed upon the accuracy or adequacy of the information contained in this Disclosure Statement or the Plan. Any representation to the contrary is a criminal offense.

The Debtor is relying on section 4 of the Securities Act and section 1145(a) of the Bankruptcy Code and any applicable Blue Sky Laws provisions to exempt from registration under the Securities Act and Blue Sky Laws the offer and sale of New Common Stock, including, without limitation, in connection with the Solicitation and the Plan.

Neither the solicitation of votes in connection with the Plan nor this Disclosure Statement constitutes an offer to sell or the solicitation of an offer to buy securities in any state or jurisdiction in which such offer or solicitation is not authorized.

DISCLAIMER

This Disclosure Statement contains, among other things, summaries of the Plan, certain statutory provisions, and certain events in the Chapter 11 Case. Although the Debtor believes that these summaries are fair and accurate, these summaries are qualified in their entirety to the extent that they do not set forth the entire text of such documents or statutory provisions or every

01:23221416.1

² Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to such terms in the Plan.

detail of such anticipated events. In the event of any inconsistency or discrepancy between a description in this Disclosure Statement and the terms and provisions of the Plan or any other documents incorporated herein by reference, the Plan or such other documents will govern for all purposes. Factual information contained in this Disclosure Statement has been provided by the Debtor's management except where otherwise specifically noted.

The statements contained in this Disclosure Statement are made only as of the date of this Disclosure Statement, and there can be no assurance that the statements contained herein will be correct at any time after this date. Although the Debtor may subsequently update the information in this Disclosure Statement, the Debtor has no affirmative duty to do so, except as otherwise provided in the Plan or in accordance with applicable law. The information contained in this Disclosure Statement, including the information regarding the history, businesses and operations of the Debtor and its non-debtor affiliates, the financial information regarding the Debtor and the liquidation analyses relating to the Debtor, is included for purposes of soliciting acceptances of the Plan, but, as to contested matters and adversary proceedings, is not to be construed as an admission or stipulation, but rather as a statement made in settlement negotiations as part of the Debtor's attempt to settle and resolve claims and controversies pursuant to the Plan. This Disclosure Statement will not be admissible in any non-bankruptcy proceeding, nor will it be construed to be conclusive advice on the tax, securities, or other legal effects of the Plan as to Holders of Claims against, or Interests in, either the Debtor or the Reorganized Debtor. The Debtor does not represent or warrant that the information contained herein or attached hereto is without any material inaccuracy or omission. Except where specifically noted, the financial information contained in this Disclosure Statement and the exhibits hereto has not been audited by a certified public accountant and has not been prepared in accordance with generally accepted accounting principles in the United States.

The Debtor has not authorized any Person or Entity in connection with this Disclosure Statement, the Plan or the Solicitation to give any information or make any representation or statement regarding this Disclosure Statement, the Plan or the Solicitation, in each case, other than that which is contained in this Disclosure Statement and the exhibits hereto or as otherwise incorporated by reference or referred to herein. If any such information, representation or statement is given or made, it may not be relied upon as having been authorized by the Debtor.

The Debtor's management, in consultation with the Debtor's financial advisors, prepared the financial projections provided in this Disclosure Statement. While the Debtor has presented these projections with numerical specificity, it has necessarily based the projections on a variety of estimates and assumptions that, although considered reasonable by management, may not be realized, and are inherently subject to significant business, economic, competitive, industry, regulatory, market and financial uncertainties and contingencies, many of which will be beyond the Debtor's or the Reorganized Debtor's control. The Debtor cautions that it cannot make any representations as to the accuracy of these projections or to the Debtor's or Reorganized Debtor's ability to achieve the projected results. Some assumptions inevitably will not materialize. Furthermore, events and circumstances occurring subsequent to the date on which these projections were prepared may differ from any assumed facts and circumstances. Alternatively, any events and circumstances that come to pass may well have been unanticipated, and thus may affect financial results in a materially adverse or materially beneficial manner. The projections,

therefore, may not be relied upon as a guaranty or other assurance of the actual results that will occur.

Certain of the information contained in this Disclosure Statement is by its nature forward looking and contains estimates, assumptions and projections that may be materially different from actual future results. The words "believe," "may," "will," "estimate," "continue," "anticipate," "intend," "expect" and similar expressions identify these forward-looking statements. These forward-looking statements are subject to a number of risks, uncertainties and assumptions, including those described in <u>ARTICLE VIII</u>, "Risk Factors." In light of these risks and uncertainties, the forward-looking events and circumstances discussed in this Disclosure Statement may not occur, and actual results could differ materially from those anticipated in the forward-looking statements. Except as required by law, the Debtor disclaims any obligation to update or revise any forward-looking statements, whether as a result of new information, future events or otherwise.

TABLE OF CONTENTS

INTRODUCT	TION	1
ARTICLE I.	ΓHE PLAN	4
A.	New Capital Structure	4
B.	Sources of Cash Consideration for Plan Distributions	
C.	Master Lease	
D.	Release of Guaranty	
E.	Occurrence of the Effective Date	5
F.	Overview – Satisfaction of Claims and Interests	5
G.	Unclassified Claims	7
H.	Classified Claims and Interests	8
I.	Transfer of New Common Stock	15
ARTICLE II.	VOTING PROCEDURES AND REQUIREMENTS	16
A.	Classes Entitled to Vote on the Plan	16
B.	Votes Required for Acceptance by a Class	16
C.	Certain Factors to Be Considered Prior to Voting	16
D.	Classes Not Entitled to Vote on the Plan	17
E.	Solicitation Procedures	18
F.	Voting Procedures	18
ARTICLE III	BUSINESS OVERVIEW, CORPORATE HISTORY AND THE MASTER LEASE AND SECURITY AGREEMENT	19
A.	Business Overview	19
B.	The MLSA and the Guaranty	
C.	Employees	
D.	Prepetition Capital Structure	
E.	Equity Interests in the Debtor	
ARTICLE IV	. EVENTS LEADING TO THE CHAPTER 11 FILING	23
A.	Industry-Wide Trends	23
B.	Transfers of Cash to the Long-Term Care Business from the 4H Business and Outpatient Rehab and Other Businesses	the
C.	Prior Agreements and Amendments	
D.	Default and Acceleration of the Prior Credit Facility	
Б. Е.	The Company's Segregation of Cash and HCR III's Subsequent Rent Payment	
F.	Filing of the Receivership Complaint and Entry Into the December 2017	
	Forbearance Agreement	26
ARTICLE V.	THE CHAPTER 11 CASE	27
Α.	Commencement of the Chapter 11 Case	27

В.	Confirmation of Original Plan	28
C.	Alternative Plan Sponsor Agreement and the Plan	28
ARTICLE V	I. OTHER KEY ASPECTS OF THE PLAN AND THE PLAN SPONSOR	
	AGREEMENTS	29
A.	Distributions	29
B.	Restructuring Transactions	
C.	Regulatory Approvals	
D.	Treatment of Executory Contracts and Insurance Policies	
E.	Discharges, Releases, Exculpations and Injunctions; Survival of Indemnifi	
	and Exculpation Obligations	
F.	Excess Severance Payments; Former CEO Settlement Agreement	
ARTICLE V	II. CONFIRMATION PROCEDURES	39
A.	Hearing on Confirmation	39
B.	Requirements for Confirmation of the Plan	40
C.	Best Interests of Creditors / Liquidation Analysis	41
D.	Feasibility	
E.	Acceptance by Impaired Class	
F.	Confirmation Without Acceptance by Impaired Class	44
G.	Alternatives to Confirmation and Consummation of the Plan	
ARTICLE V	III. RISK FACTORS	45
A.	Bankruptcy-Specific Considerations	46
B.	Risks Related to the Debtor's and the Reorganized Debtor's Business	51
ARTICLE I	X. IMPORTANT SECURITIES LAW DISCLOSURE	54
A.	Plan Securities	54
B.	Issuance and Resale of Plan Securities under the Plan	55
ARTICLE X	C. CERTAIN UNITED STATES FEDERAL INCOME TAX CONSEQUEN	CES
	OF THE PLAN	56
A.	Certain Federal Income Tax Consequences to the Debtor	
B.	Certain Federal Income Tax Consequences to Holders of a QCP Claim	59
C.	Certain Federal Income Tax Consequences to Holders of Interests	
D.	Reservation of Rights	60
ARTICLE X	II. CONCLUSION AND RECOMMENDATION	60

TABLE OF EXHIBITS

EXHIBIT A First Amended Chapter 11 Plan of Reorganization

EXHIBIT B Alternative Restructuring Support Agreement

EXHIBIT C Original Restructuring Support Agreement

EXHIBIT D HCR Financial Projections

EXHIBIT E 4H Global Financial Projections

INTRODUCTION

On the Petition Date, the Debtor commenced the Chapter 11 Case and filed a Prepackaged Plan of Reorganization for HCR ManorCare, Inc. dated March 3, 2018 (including all exhibits and schedules attached thereto, and as subsequently modified, the "Original Plan") and a disclosure statement with respect to the Original Plan (the "Original DS"). On April 13, 2018, the Bankruptcy Court entered the Findings of Fact, Conclusions of Law and Order Approving (I) the Adequacy of the Disclosure Statement, (II) Prepetition Solicitation Procedures and (III) Confirmation of the Original Plan of Reorganization for HCR ManorCare, Inc. [Docket No. 127] (the "Original Confirmation Order").

After entry of the Original Confirmation Order, but prior to the Effective Date and substantial consummation of the Original Plan, Welltower Inc. ("Welltower") and ProMedica Health System, Inc. ("ProMedica Parent") entered into definitive agreements to acquire, respectively, QCP and the equity of the Debtor. On April 25, 2018, Welltower entered into an Agreement and Plan of Merger (the "Merger Agreement") with a subsidiary of Welltower, Potomac Acquisition LLC ("Potomac"), QCP, and certain subsidiaries of QCP, pursuant to which the parties agreed that, subject to the terms and conditions set forth in the Merger Agreement, Welltower would acquire all of the outstanding capital stock of QCP in an all-cash merger. Pursuant to the Merger Agreement, Welltower will acquire QCP through a series of mergers, with Potomac surviving as a wholly owned subsidiary of Welltower (the "Mergers"). Each of Welltower's and QCP's obligations to consummate the Mergers is subject to a number of customary closing conditions, as well as a condition requiring the successful completion of ProMedica Parent's acquisition of the Debtor (the "HCR Acquisition"). The proposed HCR Acquisition, in turn, contemplates that a newly created subsidiary of ProMedica Parent, Suburban Healthco, Inc. ("ProMedica Sub" and, together with ProMedica Parent, "ProMedica") (or its designee(s)), will acquire 100% of the equity of the Reorganized Debtor in exchange for the ProMedica Plan Contribution (as defined below). In connection with the Mergers and the HCR Acquisition, Welltower formed a joint venture, Meerkat I LLC, with ProMedica Parent that will own substantially all of the real property assets relating to the Company's business and will lease those assets to a non-debtor subsidiary of the Debtor, HCR III Healthcare, LLC ("HCR III"). The closing of the Mergers and the HCR Acquisition - collectively, the proposed "Alternative Transaction" – are anticipated to occur in the third quarter of 2018, assuming all of the various approvals and conditions to closing have been met.

To facilitate and implement this Alternative Transaction, the Debtor has proposed the *First Amended Chapter 11 Plan of Reorganization for HCR ManorCare, Inc.*, which provides for alternative treatment for certain Classes of Claims against and Interests in the Debtor if the Alternative Transaction closes. If the Alternative Transaction does not close, the Plan provides that the Classes of Claims against and Interests in the Debtor will receive the same treatment as provided in the Original Plan.

This Disclosure Statement provides information, pursuant to section 1125 of the Bankruptcy Code, regarding the *First Amended Chapter 11 Plan of Reorganization for HCR ManorCare, Inc.* (including all exhibits and schedules attached thereto, and as may be amended,

altered, modified or supplemented from time to time, the "<u>Plan</u>"),³ dated May 18, 2018. A copy of the Plan is attached hereto as <u>Exhibit A</u>. The rules of interpretation set forth in Article I of the Plan shall govern the interpretation of this Disclosure Statement.

Until the Plan is confirmed by the Bankruptcy Court, the Original Confirmation Order and the Original Plan remain in full force and effect and are binding upon the Debtor, the Reorganized Debtor, all Holders of Claims against and Interests in the Debtor and each such Holder's respective successors and assigns in accordance with the terms of the Original Confirmation Order and the Original Plan. If the Bankruptcy Court confirms the Plan, the Original Plan shall be deemed to be modified, amended and superseded in its entirety by the Plan.

The Debtor's board of directors has approved the Plan and believes the Plan is in the best interests of the Debtor's Estate. The primary features of the Plan are as follows:

- If the Alternative Transaction closes:
 - O ProMedica Parent, as the Plan Sponsor, will make a plan contribution to the Debtor in the amount required to (i) pay the Allowed Credit Facility Claims in full, (ii) pay the Agreed Deferred Rent Obligation in the amount of \$440,190,183 to the Holders of the QCP Claims, and (iii) pay \$50 million to Holders of Interests in the Debtor (the "ProMedica Plan Contribution"). The ProMedica Plan Contribution may be in the form, at ProMedica's election, of either (x) a capital contribution to the Reorganized Debtor or (y) a combination of (A) a capital contribution to the Reorganized Debtor and (B) an unsecured, subordinated loan to the Reorganized Debtor in principal amount not to exceed \$550 million, which unsecured loan shall mature no earlier than the third anniversary of the date of the Closing. In exchange for ProMedica Parent making the ProMedica Plan Contribution, ProMedica Sub (or its designee(s)) will receive 100% of the stock of the Reorganized Debtor.
 - o No creditors other than QCP will be impaired.
 - o Equity holders will receive a share of \$50 million, to be allocated to the Holders of Preferred and Common Interests in accordance with the Plan.
- If the Alternative Transaction does not close:
 - The Debtor will consummate the transaction (the "Original Transaction") described by the Original Plan Sponsor Agreement (as defined below), subject to the terms and conditions thereof, which transaction is the same as the transaction embodied in the Original Plan and approved in the Original Confirmation Order.
 - Under the Original Transaction, QCP (or its designee(s)) will receive 100% of the stock of the Reorganized Debtor in full and final satisfaction of its Claims against

³ Capitalized terms used but not otherwise defined in this Disclosure Statement shall have the meanings ascribed to such terms in the Plan.

- the Debtor (arising under the Debtor's guarantee of the lease obligations owed to QCP by non-debtor HCR III, as described in detail below).
- No creditors other than QCP will be impaired. The Plan provides that Section 510(b) Claims are impaired; however the Debtor is not aware of any existing Section 510(b) Claims.
- o Equity holders will receive no distributions.
- The Holders of more than 80% of the equity of the Debtor (the "Majority Stockholders") have agreed to support the Plan.
- The Effective Date of the Plan is contingent upon, among other things, the satisfaction or waiver of the conditions precedent to the consummation of (i) the Alternative Transaction pursuant to the Alternative Plan Sponsor Agreement, executed by the Debtor, QCP, ProMedica, and Meerkat I LLC, dated as of April 25, 2018 (as amended, modified or supplemented from time to time in accordance with its terms and including all schedules and exhibits thereto, the "Alternative Plan Sponsor Agreement"), which sets forth detailed terms and conditions for the transfer of the Company's equity to ProMedica Sub (or its designee(s)), or in the event the closing of the Alternative Transaction fails to occur, (ii) the Original Transaction pursuant to the Original Plan Sponsor Agreement, executed by the Debtor, QCP, HCP Mezzanine Lender, LP and the lessors identified therein, dated as of March 2, 2018 (as amended, modified or supplemented from time to time in accordance with its terms (including, without limitation, by the Amendment to the Original Plan Sponsor Agreement (as defined below)) and including all schedules and exhibits thereto, the "Original Plan Sponsor Agreement"), which sets forth detailed terms and conditions for the transfer of the Company's equity to QCP. The Alternative Plan Sponsor Agreement, the Original Plan Sponsor Agreement, and the April 25, 2018 Amendment to the Original Plan Sponsor Agreement (the "Amendment to the Original Plan Sponsor Agreement") are attached as exhibits to the Plan.
- On the Effective Date, HCR III will either enter into the Alternative Master Lease with Meerkat I LLC (or its designee(s)) (if the Alternative Transaction closes) or the Original Master Lease Amendment (if the Alternative Transaction does not close and the Original Transaction closes). The Alternative Master Lease Term Sheet and the Original Master Lease Amendment are attached as exhibits to the Plan; the Alternative Master Lease shall be filed with the Plan Supplement. In either case, prior to the seventh day after the Effective Date, HCR III and the Lessors under the existing MLSA shall terminate, release and discharge the Guaranty pursuant to the applicable Plan Sponsor Agreement.

The Debtor, ProMedica Parent, and QCP anticipate that the Effective Date will occur in the third quarter of 2018, as it may take three to five months from the date of this Disclosure Statement for the closing of the Merger Agreement and to obtain the governmental approvals necessary for consummation of the transaction.

As described below, only Holders of Class 4 Claims (QCP Claims), Holders of Class 8A Interests (Preferred Interests) and Holders of Class 8B Interests (Common Interests) are entitled to vote to accept or reject the Plan. Following more than fifteen (15) months of extensive negotiations, the Debtor and QCP entered into the Original Plan Sponsor Agreement and the Original Restructuring Support Agreement. A copy of the Original Restructuring Support Agreement is attached hereto as **Exhibit C**. During such time, QCP was entitled to, and was provided, significant access to financial and operational information regarding the Debtor and the Company under the terms of the MLSA. This information included, among other things, audited annual and unaudited quarterly consolidated balance sheets of the Debtor, HCR III and their respective consolidated subsidiaries and related consolidated income statements and monthly facility-level statements of income and detailed operational statistics. Additionally, the Debtor and QCP engaged in extensive negotiations regarding the Alternative Transaction prior to the execution of the Alternative Plan Sponsor Agreement and the Alternative Restructuring Support Agreement. A copy of the Alternative Restructuring Support Agreement is attached hereto as **Exhibit B**.

ARTICLE I. THE PLAN

A. New Capital Structure

On the Effective Date, the Debtor will effectuate the transactions contemplated by the Alternative Plan Sponsor Agreement if the Closing of the Alternative Transaction occurs. Under the Alternative Transaction, ProMedica Sub (or its designee(s)) will receive 100% of the New Common Stock in consideration for the ProMedica Plan Contribution. The ProMedica Plan Contribution will be in an amount sufficient to pay the Allowed Credit Facility Claims in full, the Agreed Rent Deferral Obligation owed to the Holders of the QCP Claims, and \$50 million to current equity holders of the Debtor. The balance of the QCP Claims, including such Claims that QCP asserts are Administrative Expense Claims, shall be waived and released. The remainder of the Debtor's obligations will be paid in full in Cash, Reinstated, or otherwise left Unimpaired.

If the Closing of the Alternative Transaction does not occur, the Debtor will effectuate the transactions contemplated by the Original Plan Sponsor Agreement. If this scenario occurs, QCP (or its designee(s)) will receive 100% of the New Common Stock in full satisfaction of all of the QCP Claims, *i.e.*, all Claims of QCP and its subsidiaries against the Debtor arising under the Guaranty and due and unpaid as of the Effective Date. The current equity holders of the Debtor will not receive any distribution, and all current equity interests will be cancelled. Otherwise, the capital structure will remain largely unchanged. The secured Credit Facility and Intercompany Note will be Reinstated on existing terms, and the remainder of the Debtor's obligations will be paid in full in Cash, Reinstated, or otherwise left Unimpaired.

B. Sources of Cash Consideration for Plan Distributions

The Reorganized Debtor shall fund distributions and satisfy applicable Allowed Claims under the Plan with Cash on hand and/or, if the Closing of the Alternative Transaction occurs, from such Cash on hand together with the ProMedica Plan Contribution.

C. Master Lease

If the Closing of the Alternative Transaction occurs, on the Effective Date, the Reorganized Debtor shall cause HCR III to, and Meerkat I LLC (or its designee(s)) and the Lessors shall, enter into the Alternative Master Lease, substantially in accordance with the Alternative Master Lease Term Sheet attached as Exhibit A to the Plan. Meerkat I LLC is the joint venture owned by Welltower and ProMedica Parent and, after the Closing of the Alternative Transaction, will own the real estate previously owned by QCP.

If the Closing of the Alternative Transaction does not occur, on the Effective Date, the Reorganized Debtor shall cause HCR III to, and QCP and the Lessors shall, enter into the Original Master Lease Amendment, substantially in the form of Exhibit C to the Plan.

D. Release of Guaranty

On or prior to the third day after the Effective Date if the Alternative Transaction closes, or, on or prior to the seventh day after the Effective Date if the Original Transaction closes, HCR III and the Lessors shall terminate, release and discharge the Guaranty pursuant either to the Original Plan Sponsor Agreement or the Alternative Plan Sponsor Agreement, as applicable.

E. Occurrence of the Effective Date

The Effective Date of the Plan is contingent upon, among other things, the satisfaction or waiver of the conditions precedent to the consummation of (i) the Alternative Transaction pursuant to the Alternative Plan Sponsor Agreement, which sets forth detailed terms and conditions for the transfer of the Company's equity to ProMedica Sub (or its designee(s)), or (ii) the Original Transaction pursuant to the Original Plan Sponsor Agreement, which sets forth detailed terms and conditions for the transfer of the Company's equity to QCP. The Alternative Plan Sponsor Agreement, the Original Plan Sponsor Agreement, and Amendment to the Original Plan Sponsor Agreement are attached as Exhibits B, D, and E respectively, to the Plan.

F. Overview – Satisfaction of Claims and Interests

The Plan provides for the satisfaction of Claims and Interests through: (a) a payment of Cash equal to the Agreed Deferred Rent Obligation to the Holders of QCP Claims if the Closing of the Alternative Transaction occurs, or the issuance of New Common Stock to QCP (or its designee(s)) if the Closing of the Original Transaction occurs; (b) the Reinstatement of certain Claims; and (c) payment of Cash. As more fully described herein:

- Holders of Allowed Other Priority Claims, if any, will be paid in full in Cash, Reinstated, or otherwise rendered Unimpaired;
- Holders of Allowed Secured Claims, if any, will be paid in full in Cash, Reinstated, receive delivery of the collateral securing their claims, or otherwise rendered Unimpaired;
- Holders of Allowed Credit Facility Claims will have their claims either (i) paid in full in Cash (if the Closing of the Alternative Transaction occurs) or

- (ii) Reinstated and their liens retained with the same priority as of the Petition Date (if the Closing of the Original Transaction occurs);
- Holders of Allowed QCP Claims either will have (i) the Agreed Deferred Rent Obligation paid, in full in Cash, with the balance of their claims waived and released (if the Closing of the Alternative Transaction occurs) or (ii) their claims converted into 100% of the New Common Stock of the Reorganized Debtor (if the Closing of the Original Transaction occurs);
- Holders of Allowed General Unsecured Claims will be paid in full in Cash, Reinstated, or otherwise rendered Unimpaired;
- Holders of Allowed Severance Claims will be paid in Cash up to the maximum amount permitted under the applicable provisions of the Bankruptcy Code;
- Holders of Allowed Section 510(b) Claims, if any, will have their Claims either (i) rendered Unimpaired (if the Closing of the Alternative Transaction occurs) or (ii) discharged and extinguished without receiving or retaining any property (if the Closing of the Original Transaction occurs);
- Holders of Preferred Interests in the Debtor either (i) will receive their pro rata share of \$2 million plus any accrued unpaid dividends due and payable to Holders of Preferred Interests as of the Effective Date (the "Accrued Unpaid Dividends")⁴ (if the Closing of the Alternative Transaction occurs) or (ii) will have their Preferred Interests cancelled, annulled and extinguished and will not receive or retain any property nor receive any distributions (if the Closing of the Original Transaction occurs);
- Holders of Common Interests in the Debtor either (i) will receive their pro rata share of \$48 million less the amount of the Accrued Unpaid Dividends (if the Closing of the Alternative Transaction occurs) or (ii) will have their Common Interests cancelled, annulled and extinguished and will not receive or retain any property nor receive any distributions (if the Closing of the Original Transaction occurs).
- All Intercompany Claims will either be (a) Reinstated, (b) deemed satisfied, or
 (c) discharged and extinguished and eliminated; <u>provided</u>, <u>however</u>, that any
 Intercompany Claim held by an obligor under the Credit Facility shall not be
 discharged or extinguished without the express written consent of the Credit
 Facility Agent; and

⁴ Additional information regarding the Accrued Unpaid Dividends is set forth in <u>ARTICLE III.E</u> of this Disclosure Statement.

• Holders of Allowed Administrative Expense Claims, Allowed Priority Tax Claims, and Allowed Professional Fee Claims will be: (a) paid in full in Cash; (b) Reinstated; or (c) otherwise rendered Unimpaired, as applicable.

G. Unclassified Claims

1. Unclassified Claims Summary

Claim	Plan Treatment	Projected Plan Recovery
Administrative Claims	Paid in full in Cash	100%
Professional Fee Claims	Paid in full in Cash	100%
Priority Tax Claims	Paid in full in Cash	100%

2. Unclassified Claims Details

a. Administrative Expense Claims

Subject to the terms of Confirmation Order and Section 2.1(b) of the Plan, each Holder of an Allowed Administrative Expense Claim shall receive, on account of and in full and complete settlement, release and discharge of, and in exchange for, such Allowed Administrative Expense Claim, payment of Cash equal to the full unpaid amount of such Allowed Administrative Expense Claim (i) on the Effective Date or as soon as reasonably practicable thereafter, (ii) such other date as the Reorganized Debtor and such Holder shall have agreed, or (iii) as otherwise ordered by the Bankruptcy Court; provided, however, that Allowed Administrative Expense Claims not yet due or that represent obligations incurred by the Debtor in the ordinary course of its business during the Chapter 11 Case, or assumed by the Debtor during the Chapter 11 Case, shall be paid or performed when due in the ordinary course of business and in accordance with the terms and conditions of the particular agreements governing such obligations.

b. Professional Fee Claims

All Professionals or other entities requesting compensation or reimbursement of expenses pursuant to sections 327, 328, 330, 331, 503(b) and/or section 1103 of the Bankruptcy Code for services rendered before the Effective Date (including, without limitation, any compensation requested by any Professional or any other entity for making a substantial contribution in the Chapter 11 Case) shall file and serve final requests for payment of Professional Fee Claims no later than the first Business Day that is forty-five (45) days after the Effective Date. Objections to any Professional Fee Claim must be filed and served on the Reorganized Debtor and the applicable Professional within thirty (30) days after the filing of the final fee application with respect to the Professional Fee Claim. Any such objections that are not consensually resolved may be set for hearing on twenty-one (21) days' notice by the Professional asserting such Professional Fee Claim.

c. Priority Tax Claims

Each Holder of an Allowed Priority Tax Claim against the Debtor shall receive, on account of and in full and complete settlement, release and discharge of, and in exchange for, such Allowed Priority Tax Claim, payment in full in Cash, on the latest of: (a) the Effective

Date; (b) the date such Priority Tax Claim becomes an Allowed Claim; and (c) the date such Allowed Priority Tax Claim becomes due and payable under applicable non-bankruptcy law.

d. Post-Effective Date Fees and Expenses

Except as otherwise specifically provided in the Plan, from and after the Effective Date, the Reorganized Debtor shall, in the ordinary course of business and without any further notice to or action, order or approval of the Bankruptcy Court, pay in Cash the legal, professional or other fees and expenses related to the implementation and consummation of the Plan incurred by the Reorganized Debtor following the Effective Date. Upon the Effective Date, any requirement that Professionals comply with sections 327 through 331 and 1103 of the Bankruptcy Code in seeking retention or compensation for services rendered after such date shall terminate, and each Reorganized Debtor may employ and pay any professional for services rendered or expenses incurred after the Effective Date in the ordinary course of business without any further notice to or action, order or approval of the Bankruptcy Court.

H. Classified Claims and Interests

1. Classified Claims and Interests Summary

The Plan establishes a comprehensive classification of Claims and Interests. The table below summarizes the classification, treatment, voting rights, and estimated recoveries, estimated as of the Petition Date, of the Claims and Interests, by Class, under the Plan. Amounts assumed in the liquidation recovery analysis are estimates only. The projected liquidation recoveries are based on certain assumptions described herein. Accordingly, recoveries received by Holders of Claims and Interests in a liquidation scenario may differ materially from the projected liquidation recoveries listed in the table below.

Class	Claim or Interest	Voting Rights	Treatment	Plan Recovery	Liquidation Recovery
1	Other Priority Claims	Not Entitled to Vote / Presumed to Accept	Paid in full in Cash/Reinstated/ Unimpaired	100%	100%
2	Secured Claims	Not Entitled to Vote / Presumed to Accept	Paid in full in Cash/Reinstated/ Collateral Returned/ Unimpaired	100%	100%
3	Credit Facility Claims	Not Entitled to Vote / Presumed to Accept	Paid in full in Cash/Reinstated/ Unimpaired	100%	100%
4	QCP Claims	Entitled to Vote	Impaired	< 100%	> 0%
5	General Unsecured Claims	Not Entitled to Vote/ Presumed to Accept	Paid in full in Cash/Reinstated/ Unimpaired	100%	100%
6	Severance Claims	Not Entitled to Vote / Presumed to Accept	Paid in full in Cash	100%	100%

Class	Claim or Interest	Voting Rights	Treatment	Plan Recovery	Liquidation Recovery
7	Section 510(b) Claims	Not Entitled to Vote / Deemed to Reject	Impaired	100%* or 0%**	0%
8A	Preferred Interests	Entitled to Vote	Impaired	100%* or 0%**	0%
8B	Common Interests	Entitled to Vote	Impaired	>0%* or 0%**	0%

^{*} Plan Recovery if the Alternative Transaction closes

2. Classified Claims and Interests Details

Except to the extent that the Debtor and a Holder of an Allowed Claim or Interest, as applicable, agree to a less favorable treatment, such Holder shall receive under the Plan the treatment described below in full and final satisfaction, settlement, release, and discharge of and in exchange for such Holder's Allowed Claim or Interest. Unless otherwise indicated, the Holder of an Allowed Claim or Interest, as applicable, shall receive such treatment on the Effective Date or as soon as reasonably practicable thereafter.

a. Class 1 – Other Priority Claims

- (i) <u>Classification</u>: Class 1 consists of any Other Priority Claims.
- (ii) Treatment: Each Holder of an Allowed Other Priority Claim will receive, in the sole discretion of the Reorganized Debtor, (i) payment in full in Cash as promptly as reasonably practicable on the later of (a) the Effective Date and (b) the date on which such Other Priority Claim becomes an Allowed Claim payable under applicable law or any agreement relating thereto, or (ii) treatment of such Allowed Other Priority Claim in any other manner that renders the Claim Unimpaired, including Reinstatement. All Allowed Other Priority Claims not due and payable on or before the Effective Date shall be paid in the ordinary course of business in accordance with the terms thereof.
- (iii) <u>Voting</u>: Allowed Claims in Class 1 are Unimpaired. Each Holder of an Allowed Claim in Class 1 shall be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject the Plan.

^{**} Plan Recovery if the Original Transaction closes

b. Class 2 – Secured Claims

- (i) Classification: Class 2 consists of Secured Claims.
- (ii) Treatment: Except to the extent a Holder of a Secured Claim agrees to different treatment of that Secured Claim, each Holder of an Allowed Secured Claim shall, in the sole discretion of the Reorganized Debtor, receive on the Effective Date (or as promptly thereafter as reasonably practicable) or in the ordinary course of the Reorganized Debtor's business (a) payment in full by Reorganized Debtor in Cash, including the payment of any interest Allowed and payable under section 506(b) of the Bankruptcy Code, (b) delivery of the collateral securing such Allowed Secured Claim, or (c) treatment of such Allowed Secured Claim in any other manner that renders the Claim Unimpaired, including Reinstatement.
- (iii) <u>Voting</u>: Allowed Claims in Class 2 are Unimpaired. Each Holder of an Allowed Claim in Class 2 shall be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject the Plan.

c. Class 3 – Credit Facility Claims

- (i) <u>Classification</u>: Class 3 consists of the Credit Facility Claims.
- Allowance: If the Closing of the Alternative Transaction (ii) occurs, the Credit Facility Claims shall be Allowed in the principal amount of \$550,000,000 plus all unpaid fees, costs, expenses, interest, charges, premiums and other obligations due under the Credit Facility, including, without limitation, the Make-Whole Amount to the extent payable in accordance with the terms of the Credit Facility; provided, however, that if the Closing of the Original Transaction occurs, the Credit Facility Claims shall be Allowed to the extent required to Reinstate the Credit Facility Claims under the applicable provisions of the Bankruptcy Code. For the avoidance of doubt, if the Closing of the Original Transaction occurs and the Credit Facility Claims are Reinstated, the Make-Whole Amount shall not be due and payable at the time of Reinstatement.
- (iii) <u>Treatment</u>:

- (a) If the Closing of the Alternative Transaction occurs, the Credit Facility Agent shall receive, on the Effective Date, in full and final satisfaction, release, and discharge of (including any Liens related thereto), and in exchange for, the Allowed Credit Facility Claims, payment in full, in Cash, of all such Allowed Credit Facility Claims.
- (b) If the Closing of the Alternative Transaction does not occur, on the Effective Date, the Credit Facility Claims shall be Reinstated and any liens held by the Credit Facility Agent, on behalf of itself and the Credit Facility Lenders, against the assets of the Debtor securing the Credit Facility Claims shall survive the Effective Date in the same priority as they existed as of the Petition Date.
- (iv) <u>Voting</u>: Claims in Class 3 are Unimpaired. Each Holder of an Allowed Claim in Class 3 shall be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject the Plan.

d. Class 4 – QCP Claims

- (i) Classification: Class 4 consists of all QCP Claims.
- (ii) Allowance: The QCP Claims shall be deemed Allowed on the Effective Date in the aggregate amount of \$445,796,590.

(iii) <u>Treatment</u>:

- (a) If the Closing of the Alternative Transaction occurs, on the Effective Date, (x) the Agreed Deferred Rent Obligation shall be paid, in full, in Cash, and (y) the balance of the QCP Claims, including any such QCP Claims that are Administrative Expense Claims, shall be waived and released.
- (b) If the Closing of the Alternative Transaction does not occur, the Holders of the QCP Claims (or the designee(s) of QCP) shall receive, on the Effective Date, in full and final satisfaction, release, and discharge of, and in exchange for, their QCP Claims, 100% of the New Common Stock of Reorganized Debtor.

(iv) <u>Voting</u>: Allowed Claims in Class 4 are Impaired. The Holder of an Allowed QCP Claim in Class 4 shall be entitled to vote to accept or reject the Plan.

e. Class 5 – General Unsecured Claims

- (i) <u>Classification</u>: Class 5 consists of all General Unsecured Claims.
- General Unsecured Claim agrees to different treatment of that General Unsecured Claim, each Holder of an Allowed General Unsecured Claim, each Holder of an Allowed General Unsecured Claim shall receive, in the sole discretion of the Reorganized Debtor, (1) payment in full in Cash of such Claim (a) on the Effective Date or as promptly thereafter as reasonably practicable (b) in the ordinary course of the Reorganized Debtor's business, or (c) on the date such Allowed General Unsecured Claim becomes due and payable according to its terms; or (2) satisfaction of such Allowed General Unsecured Claim in any other manner that renders the Allowed General Unsecured Claim Unimpaired, including Reinstatement.
- (iii) <u>Voting</u>: Allowed Claims in Class 5 are Unimpaired. Each Holder of an Allowed Claim in Class 5 shall be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject the Plan.

f. Class 6 – Severance Claims

- (i) Classification: Class 6 consists of all Severance Claims.
- (ii) <u>Allowance</u>: On the Effective Date, the Severance Claims shall be Allowed in the maximum amount permitted under the applicable provisions of the Bankruptcy Code.
- (iii) Treatment: Each Holder of an Allowed Severance Claim shall receive, in the sole discretion of the Plan Sponsor, payment in full in Cash on account of such Allowed Severance Claim (a) on the Effective Date or as soon thereafter as reasonably practicable or (b) as otherwise agreed by the Reorganized Debtor and such Holder of a Severance Claim.
- (iv) <u>Voting</u>: Allowed Claims in Class 6 are Unimpaired, and the Holders of such Claims are conclusively deemed to

have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code.

g. Class 7 – Section 510(b) Claims

(i) <u>Classification</u>: Class 7 consists of all Section 510(b) Claims.

(ii) Treatment:

- (a) If the Closing of the Alternative Transaction occurs, except to the extent a Holder of an Allowed Section 510(b) Claim, if any, agrees to different treatment of that Allowed Section 510(b) Claim, each such Holder of an Allowed Section 510(b) Claim shall receive satisfaction of such Allowed Section 510(b) Claim in any manner determined by the Debtor, with the consent of the Plan Sponsor, that renders such Allowed Section 510(b) Claims Unimpaired, including Reinstatement.
- (b) If the Closing of the Alternative Transaction does not occur, on the Effective Date, all Section 510(b) Claims shall be discharged and extinguished and the Holders thereof shall not receive or retain any property under the Plan on account of such Section 510(b) Claims.
- (iii) Voting: Claims in Class 7 are Impaired because each Holder of an Allowed Section 510(b) Claim, if any, will not be entitled to receive or retain any property under the Plan if the Closing of the Alternative Transaction does not occur. Accordingly, each Holder of an Allowed Claim in Class 7 shall be conclusively deemed to have rejected the Plan pursuant to section 1126(g) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject the Plan.

h. Class 8A – Preferred Interests

(i) <u>Classification</u>: Class 8A consists of all Preferred Interests.

(ii) Treatment:

(a) If the Closing of the Alternative Transaction occurs, on the Effective Date or as soon thereafter as reasonably practicable, each Holder of an Allowed

- Preferred Interest shall receive its pro rata share of the Preferred Equity Distribution.
- (b) If the Closing of the Alternative Transaction does not occur, on the Effective Date, all Preferred Interests in the Debtor shall be cancelled, annulled, and extinguished and the Holders of such Preferred Interests shall not receive or retain any property under the Plan on account of such Preferred Interests nor receive any distributions or property under the Plan on account of such Preferred Interests.
- (iii) <u>Voting</u>: Preferred Interests in Class 8A are Impaired because each Holder of Preferred Interests will not be entitled to receive or retain any property under the Plan if the Closing of the Alternative Transaction does not occur. Each Holder of a Preferred Interest in Class 8A shall be entitled to vote to accept or reject the Plan.

i. Class 8B – Common Interests

- (i) Classification: Class 8B consists of all Common Interests.
- (ii) <u>Treatment</u>:
 - (a) If the Closing of the Alternative Transaction occurs, on the Effective Date or as soon thereafter as reasonably practicable, each Holder of an Allowed Common Interest in the Debtor shall receive its pro rata share of the Common Equity Distribution.
 - (b) If the Closing of the Alternative Transaction does not occur, on the Effective Date, all Common Interests in the Debtor shall be cancelled, annulled, and extinguished and the Holders of such Common Interests shall not receive or retain any property under the Plan on account of such Common Interests nor receive any distributions or property under the Plan on account of such Common Interests.
- (iii) <u>Voting</u>: Common Interests in Class 8B are Impaired. Each Holder of a Common Interest in Class 8B shall be entitled to vote to accept or reject the Plan.

3. Intercompany Claims

On the Effective Date, in the sole discretion of the Reorganized Debtor, all Intercompany Claims shall either be (i) Reinstated, in whole or in part, (ii) deemed satisfied, or (iii) discharged and extinguished, in full or in part, and shall be eliminated as of the Effective Date, in whole or in part, in which case the Holders thereof shall not be entitled to and shall not receive or retain any property or interest on account of such discharged and extinguished portion under the Plan; provided, however, that prior to such discharge and extinguishment such Intercompany Claims may be contributed to capital, transferred, setoff or subject to any other arrangement in the sole discretion of the Reorganized Debtor; provided further, that any Intercompany Claim held by an obligor under the Credit Facility shall not be discharged or extinguished without the express written consent of the Credit Facility Agent.

For the avoidance of doubt, the Intercompany Note Claim evidenced by the Intercompany Note shall: (i) if the Closing of the Alternative Transaction occurs, be Allowed in the amount of \$25,000,000 plus accrued and unpaid interest thereon and, in the sole discretion of the Plan Sponsor under the Alternative Plan Sponsor Agreement, either (A) HCR Home Health Care and Hospice, LLC shall receive, on the Effective Date, in full and final satisfaction, release, and discharge of (including any liens related thereto), and in exchange for, the Intercompany Note Claim, payment in full, in Cash, of the Intercompany Note Claim or (B) such Intercompany Note Claim shall be Reinstated under the Plan, and any liens held by HCR Home Health Care and Hospice, LLC against the assets of the Debtor securing the Intercompany Note shall survive the Effective Date with the same priority that such liens held as of the Petition Date; and (ii) if the Closing of the Alternative Transaction does not occur, such Intercompany Note Claim shall be Allowed in the amount of \$25,000,000 plus accrued and unpaid interest thereon and be Reinstated under the Plan and any liens held by HCR Home Health Care and Hospice, LLC against the assets of the Debtor securing the Intercompany Note shall survive the Effective Date with the same priority that such liens held as of the Petition Date.

I. Transfer of New Common Stock

If the Closing of the Alternative Transaction occurs, on the Effective Date, ProMedica Sub (or its designee(s)) shall receive, and the Debtor shall, in accordance with the Implementation Memorandum, issue, transfer, and deliver, either directly or indirectly, to ProMedica Sub (or its designee(s)), 1,000,000 newly issued shares of New Common Stock of the Reorganized Debtor, par value \$0.01 per share, which shall constitute all of the issued and outstanding capital stock and rights to purchase or otherwise acquire capital stock of the Debtor, free and clear of any lien, charge, pledge, security interest, claim, or other encumbrance in consideration for the ProMedica Plan Contribution.

If the Closing of the Alternative Transaction does not occur, on the Effective Date, QCP or its designee(s) shall receive, and the Debtor shall issue, transfer, and deliver to QCP or its designee(s), 1,000,000 newly issued shares of New Common Stock of the Reorganized Debtor, par value \$0.01 per share, which shall constitute all of the issued and outstanding capital stock and rights to purchase or otherwise acquire capital stock of the Debtor, free and clear of any lien, charge, pledge, security interest, claim, or other encumbrance, in full satisfaction of all of the OCP Claims.

ARTICLE II. VOTING PROCEDURES AND REQUIREMENTS

A. Classes Entitled to Vote on the Plan

The following Classes are the only Classes entitled to vote to accept or reject the Plan (the "Voting Classes"):

Class	Claim or Interest	Status
4	QCP Claims	Impaired
8A	Preferred Interests	Impaired ⁵
8B	Common Interests	Impaired

If your Claim is not included in the Voting Classes, you are not entitled to vote and you will not receive a Solicitation Package, including a Ballot setting forth detailed voting instructions. If your Claim or Interest is included in the Voting Classes, you should read your Ballot and carefully follow the instructions included in the Ballot. Please use only the Ballot that accompanies this Disclosure Statement or the Ballot that the Debtor, or the Solicitation Agent (as defined below) on behalf of the Debtor, otherwise provided to you.

B. Votes Required for Acceptance by a Class

Under the Bankruptcy Code, acceptance of a plan of reorganization by a class of claims or interests is determined by calculating the amount and, if a class of claims, the number, of claims and interests voting to accept, as a percentage of the allowed claims or interests, as applicable, that have voted. Acceptance by a class of claims requires an affirmative vote of more than one-half in number of total allowed claims that have voted and an affirmative vote of at least two-thirds in dollar amount of the total allowed claims that have voted. Acceptance by a class of interests requires an affirmative vote of at least two-thirds in amount of the allowed interests that have voted.

C. Certain Factors to Be Considered Prior to Voting

There are a variety of factors that all Holders of Claims and Interests entitled to vote on the Plan should consider prior to voting to accept or reject the Plan. These factors, which may impact recoveries under the Plan, include the following:

• unless otherwise specifically indicated, the financial information contained in this Disclosure Statement has not been audited and is based on an analysis of

⁵ If the Alternative Transaction closes, Preferred Interests in Class 8A will be Unimpaired. However, because each Holder of Preferred Interests will not be entitled to receive or retain any property under the Plan if the Closing of the Alternative Transaction does not occur, Class 8A is being treated as Impaired and is entitled to vote on the Plan.

data available at the time of the preparation of the Plan and this Disclosure Statement;

- although the Debtor believes that the Plan complies with all applicable provisions of the Bankruptcy Code, the Debtor can neither assure such compliance nor that the Bankruptcy Court will confirm the Plan;
- the Debtor may request Confirmation without the acceptance of all Impaired Classes entitled to vote in accordance with section 1129(b) of the Bankruptcy Code; and
- any delays of either Confirmation or the occurrence of Effective Date could result in, among other things, increased Administrative Expense Claims and Professional Fee Claims.

While these factors could affect distributions available to Holders of Allowed Claims and Interests under the Plan, the occurrence or impact of such factors will not necessarily affect the validity of the votes of the Voting Classes or necessarily require a re-solicitation of the votes of Holders of Claims or Interests in the Voting Classes.

For a further discussion of risk factors, please refer to <u>ARTICLE VIII</u>, entitled "Risk Factors" of this Disclosure Statement.

D. Classes Not Entitled to Vote on the Plan

Under the Bankruptcy Code, Holders of claims and interests are not entitled to vote if their contractual rights are unimpaired by the proposed plan or if they will receive no property under the plan. Accordingly, the following Classes of Claims are not entitled to vote to accept or reject the Plan:

Class	Claim or Interest	Status	Voting Rights
1	Other Priority Claims	Unimpaired	Presumed to Accept
2	Secured Claims	Unimpaired	Presumed to Accept
3	Credit Facility Claims	Unimpaired	Presumed to Accept
5	General Unsecured Claims	Unimpaired	Presumed to Accept
6	Severance Claims	Unimpaired	Presumed to Accept
7	Section 510(b) Claims	Impaired ⁶	Deemed to Reject

⁶ If the Closing of the Alternative Transaction occurs, Section 510(b) Claims will be Unimpaired. However, because Holders of Section 510(b) Claims will not be entitled to receive or retain any property under the Plan if the Closing of the Alternative Transaction does not occur, and because the Debtor is unaware of any Section 510(b) Claims, the Class of Section 510(b) Claims is being treated as Impaired and is deemed to Reject the Plan based on the treatment of the Class in the event the Original Transaction closes.

E. Solicitation Procedures

1. Solicitation Agent

The Debtor has retained Epiq Bankruptcy Solutions, LLC to act, among other things, as the solicitation agent (the "Solicitation Agent") in connection with the solicitation of votes to accept or reject the Plan.

2. Solicitation Package

The following materials constitute the solicitation package (the "Solicitation Package") distributed to Holders of Claims and Interests in the Voting Classes:

- this Disclosure Statement and all exhibits hereto, including the Plan and all exhibits thereto;
- the Amended Solicitation Procedures Order (as defined below) and all exhibits thereto;
- the notice providing information related to the hearing to consider confirmation of the Plan and the deadline and procedures for objecting to the Plan;
- a Ballot to accept or reject the Plan along with a pre-addressed, postage prepaid return envelope; and
- such other materials as the Bankruptcy Court may direct or approve.

3. Distribution of the Solicitation Package

The Debtor will cause the Solicitation Agent to distribute the Solicitation Packages to Holders of Claims and Interests in the Voting Classes via first class mail on or before May 22, 2018. Holders of Claims in the Voting Classes who wish to vote on the Plan are required to complete and submit their Ballots to the Solicitation Agent so that the Ballots are actually received by the Solicitation Agent before 4:00 p.m. prevailing Eastern Time on June 15, 2018 (the "Voting Deadline").

F. Voting Procedures

May 21, 2018 (the "<u>Voting Record Date</u>") is the date that will be used for determining which Holders of Claims and Interests are entitled to vote to accept or reject the Plan and receive the Solicitation Package in accordance with the solicitation procedures. Except as otherwise set forth herein, the Voting Record Date and all of the Debtor's solicitation and voting procedures shall apply to all of the Debtor's creditors, equity holders and other parties in interest.

In order for the Holder of a Claim or Interest in the Voting Classes to have such Holder's Ballot counted as a vote to accept or reject the Plan, such Holder's Ballot must be properly completed, executed, and delivered to the Solicitation Agent so that such Holder's Ballot is

<u>actually received</u> by the Solicitation Agent <u>on or before the Voting Deadline</u>, *i.e.* <u>June 15</u>, <u>2018 at 4:00 p.m. prevailing Eastern Time</u>. The Ballots must be returned to the Solicitation Agent by first class mail, overnight courier, or hand delivery, or alternatively by electronic, online submission solely through the "E-Ballot" section of the Debtor's case website (http://dm.epiq11.com/HCR), pursuant to the instructions set forth on the Ballot.

UNLESS THE COURT ORDERS OTHERWISE, IF A BALLOT IS RECEIVED AFTER THE VOTING DEADLINE, IT WILL NOT BE COUNTED UNLESS THE DEBTOR OTHERWISE DETERMINES IN ITS DISCRETION.

UNLESS THE COURT ORDERS OTHERWISE, ANY BALLOT THAT IS INCOMPLETE, ILLEGIBLE, INDICATES UNCLEAR OR INCONSISTENT VOTES WITH RESPECT TO ACCEPTANCE OR REJECTION OF THE PLAN, OR ANY BALLOT THAT IS IMPROPERLY SIGNED AND RETURNED, WILL NOT BE COUNTED UNLESS THE DEBTOR OTHERWISE DETERMINES IN ITS DISCRETION.

EACH HOLDER OF A CLAIM OR INTEREST IN THE VOTING CLASSES MUST VOTE ALL OF ITS CLAIMS OR INTERESTS WITHIN THE APPLICABLE CLASS EITHER TO ACCEPT OR REJECT THE PLAN AND MAY NOT SPLIT SUCH VOTES. BY SIGNING AND RETURNING A BALLOT, EACH HOLDER OF A CLAIM OR INTEREST IN THE VOTING CLASSES WILL CERTIFY TO THE BANKRUPTCY COURT AND THE DEBTOR THAT NO OTHER BALLOTS WITH RESPECT TO SUCH CLAIM OR INTEREST HAVE BEEN CAST OR, IF ANY OTHER BALLOTS HAVE BEEN CAST WITH RESPECT TO A CLASS OF CLAIMS OR INTERESTS, SUCH OTHER BALLOTS INDICATED THE SAME VOTE TO ACCEPT OR REJECT THE PLAN. IF A HOLDER CASTS MULTIPLE BALLOTS WITH RESPECT TO A VOTING CLASS AND THOSE BALLOTS ARE IN CONFLICT WITH EACH OTHER, ONLY THE LAST VALID BALLOT RECEIVED PRIOR TO THE VOTING DEADLINE SHALL BE DEEMED TO REFLECT THE VOTER'S INTENT AND WILL SUPERSEDE AND REVOKE ANY PRIOR BALLOTS.

IT IS IMPORTANT THAT THE HOLDER OF A CLAIM OR INTEREST IN THE VOTING CLASSES FOLLOW THE SPECIFIC INSTRUCTIONS PROVIDED ON SUCH HOLDER'S BALLOT AND THE ACCOMPANYING INSTRUCTIONS.

ARTICLE III. BUSINESS OVERVIEW, CORPORATE HISTORY AND THE MASTER LEASE AND SECURITY AGREEMENT

A. Business Overview

Headquartered in Toledo, Ohio, the Debtor is a holding company that conducts business operations through more than 300 wholly owned or controlled non-debtor subsidiaries. Such subsidiaries, together with the Debtor, are collectively referred to herein as the "Company." The Company is a leading national healthcare provider that, through certain non-debtor providers, operates an extensive network of more than 450 locations across three business segments—the Long-Term Care Business, the 4H Business, and the Outpatient Rehab and Other Businesses.

1. Long-Term Care Business

As of the Petition Date, the Long-Term Care Business operated 295 skilled nursing facilities (each, a "<u>SNF</u>") and memory care or assisted living facilities (each, an "<u>ALF</u>") in twenty-five (25) states, with the majority of such facilities located in Florida, Illinois, Michigan, Ohio, and Pennsylvania. The Long-Term Care Business is conducted through 289 facilities leased from QCP and six facilities leased from third parties other than QCP.

The SNFs operate primarily under the Heartland Health Care Center[®] and ManorCare Health Services[®] brands, and the ALFs operate primarily under the Arden Courts[®] brand. The SNFs treat a wide range of patients who are transitioning from hospital to home or to a lower level of care. The SNFs also include rehabilitation units, which are separate and distinct units within Heartland and ManorCare SNFs designed for patients' rehabilitation after shorter hospital stays. The ALFs primarily provide specialized memory care in a residential environment designed exclusively for patients living with dementia such as Alzheimer's disease. The Long-Term Care Business treated more than 143,000 patients in 2017.

The Company's Long-Term Care Business is primarily conducted through non-debtor subsidiaries of non-debtor HCR III, an intermediate holding company and indirect subsidiary of the Debtor.

2. 4H Business

As of the Petition Date, the 4H Business operated more than 100 offices and agencies located in twenty-three (23) states. None of the entities operating within the 4H Business are subject to leases with QCP, and none are debtors in the Chapter 11 Case.

The 4H Business is comprised of two primary lines of business. First, "Home Health Care" provides nursing and rehabilitation services to patients at home and in independent or assisted living facilities. Home Health Care provides services to patients who have been certified by a physician as home-bound and requiring the services of a registered nurse, occupational therapist or physical therapist. Home Health Care also provides non-skilled nursing services such as companion care and/or assistance with daily living activities that are not covered by Medicare. Second, "Hospice" with a life expectancy of six months or less. Caregivers including physicians, registered nurses, social workers, licensed practical nurses, and spiritual advisers provide services to help manage patients' pain and symptoms.

3. Outpatient Rehab and Other Businesses

As of the Petition Date, the Outpatient Rehab and Other Businesses operated fifty-two (52) outpatient rehabilitation clinics under the Heartland Rehabilitation Services[®] brand. The rehabilitation clinics are primarily located in Florida, Kentucky, New Jersey, Ohio and Virginia. None of the entities operating within the Outpatient Rehab and Other Businesses are subject to leases with QCP, and none are debtors in the Chapter 11 Case.

The licensed therapists in the Company's outpatient rehabilitation clinics, which are operated through non-debtor Heartland Rehabilitation Services, LLC and certain of its subsidiaries, provide comprehensive programs in sports medicine and industrial, pediatric,

physical, occupational, speech and orthopedic therapy for patients recovering from major surgery, strokes, heart attacks, neurological and orthopedic conditions, and other illnesses, injuries and disabilities. In addition to the outpatient rehab clinics, the Company also provides comprehensive outpatient rehabilitation services through certain SNFs and in homes, schools, hospitals and third-party independent and assisted living facilities.

In addition, non-debtor Heartland Services, LLC holds a 50% ownership and voting interest in Heartland Healthcare Services, LLC, a joint venture with Omnicare, Inc. (a subsidiary of CVS Health Corporation), which provides pharmaceutical services to the Company's patients, including patients within SNFs and ALFs.

B. The MLSA and the Guaranty

On April 7, 2011, the Company entered into a sale-lease-back transaction with HCP, Inc. and certain of its affiliates ("<u>HCP</u>") pursuant to which (i) the Company sold the real property used in the operation of a significant portion of its skilled nursing and assisted living facilities to HCP (the "<u>Sale-Lease-Back Properties</u>") and (ii) non-debtor HCR III leased the Sale-Lease-Back Properties from HCP pursuant to a Master Lease and Security Agreement, dated as of April 7, 2011 (as amended from time to time, the "<u>MLSA</u>") between HCR III and HCP. The MLSA is a long-term triple-net master lease, and the Debtor guaranteed the obligations of HCR III under the MLSA pursuant to that certain Guaranty of Obligations, effective as of February 11, 2013 (as amended or modified from time to time, the "<u>Guaranty</u>"). As noted above, the leased property under the MLSA currently comprise 289 SNFs and ALFs nationwide (collectively, the "<u>Leased Facilities</u>"). The MLSA groups the Leased Facilities into four pools with initial terms that expire in 2029, 2030, 2032, and 2033.

In 2016, HCP created QCP as a new, wholly owned subsidiary and transferred the Leased Facilities to QCP. QCP and its affiliates assumed the role of lessor under the MLSA and, on October 31, 2016, HCP completed a spin-off of QCP as an independent, publicly traded company.

The MLSA requires HCR III to pay monthly rent to QCP. The monthly rent amount is determined pursuant to certain formulas set forth in the MLSA. In addition to monthly rent payments, the MLSA requires HCR III to pay certain additional charges and other amounts specified in the MLSA (including taxes, utility charges, maintaining certain insurance and paying insurance premiums) and to spend a minimum annual amount for capital projects.

HCR III subleases the Leased Facilities to certain of its direct and indirect subsidiaries and affiliates (each a "Sublease OpCo" and, collectively, the "Sublease OpCos") pursuant to certain subleases. QCP, HCR III and the Sublease OpCos have also entered into certain Agreements Regarding Subleases made and entered as of April 7, 2011, October 23, 2015, February 16, 2016 and March 14, 2016 (as amended or modified from time to time, collectively, the "Agreements Regarding Subleases"). The MLSA and the Agreements Regarding Subleases contemplate this arrangement and prescribe the form of sublease between HCR III and the Sublease OpCos. The subleases require the Sublease OpCos to pay monthly rent to HCR III in accordance with the Agreements Regarding Subleases.

C. Employees

Five current or former members of the Company's senior management are parties to employment agreements with the Debtor. The Debtor's non-debtor affiliates employ approximately 50,000 individuals in full- and part-time positions, including nurses, therapists, medical directors and other health professionals, maintenance technicians, administrative support and other staff and corporate personnel.

D. Prepetition Capital Structure

The Debtor is a party to that certain Credit Agreement, dated as of July 17, 2017 (as amended on August 25, 2017 and December 22, 2017), among HCR Home Health Care and Hospice, LLC, an indirect non-debtor subsidiary of the Debtor, as borrower; the Debtor and certain of its direct and indirect non-debtor subsidiaries, as holdcos; the several lenders from time to time party thereto; and RD Credit, LLC, as administrative agent and collateral agent (the "Credit Facility Agent"). The Credit Agreement provides for a \$400 million term loan (the "Term Loan") and a \$150 million revolving credit facility (the "Revolving Facility" and, together with the Term Loan, the "Credit Facility").

The Credit Facility is scheduled to mature on January 17, 2019. Interest on the Credit Facility is payable quarterly at a rate per annum of 9.5%. Proceeds of the Credit Facility were used for general corporate purposes and to repay the previous \$575 million credit facility, dated as of April 6, 2011, by and between HCR Healthcare, LLC, as borrower, and JPMorgan Chase Bank, N.A, as administrative agent and collateral agent (the "Prior Credit Facility").

The Debtor is also a party to that certain Guarantee and Collateral Agreement dated as of July 17, 2017 (the "Guarantee and Collateral Agreement"). Under the Guarantee and Collateral Agreement, the Debtor, the other holdcos and certain other indirect non-debtor subsidiaries of the Debtor have guaranteed the borrower's obligations under the Credit Facility. The parties' obligations under the Credit Facility and the Guarantee and Collateral Agreement, including the Debtor's guarantee obligations, are secured by a first lien on substantially all of the parties' assets, including the Debtor's assets.

The Credit Agreement provides that the commencement of the Chapter 11 Case shall not constitute an event of default thereunder.

E. Equity Interests in the Debtor

The equity interests in the Debtor are comprised of preferred stock (the "<u>Preferred Interests</u>") and common stock (the "<u>Common Interests</u>").

There are 2,000 shares of Preferred Interests issued and outstanding. The aggregate redemption price for all issued and outstanding shares of Preferred Interests is \$2,000,000 (\$1,000 per share) plus any Accrued Unpaid Dividends. The Holders of Preferred Interests are entitled to receive dividends in respect of each share of preferred stock equal to 10% of the preference amount of that share per annum compounded semi-annually on April 1 and October 1 of each year. The amount of the Accrued Unpaid Dividends will likely be approximately \$200,000 if the Effective Date occurs on or about October 1, 2018.

There are a total of 44,925,848 shares of Common Interests in the Debtor issued and outstanding. Of the total Common Interests, 43,223,932 are fully vested, unrestricted shares, and 1,701,916 are unvested, restricted shares (the "Restricted Shares"). The Restricted Shares are held by current and former members of Debtor's management and board of directors. The vast majority (1,264,376) of the Restricted Shares will vest automatically upon the consummation of the Alternative Transaction pursuant to the terms of the applicable Restricted Stock Agreements. The remainder of the Restricted Shares do not vest automatically upon the consummation of the Alternative Transaction under the terms of the applicable Rollover Stock Agreements. At the Debtor's option, if the Alternative Transaction closes, subject to the approval of ProMedica Parent (which approval shall not be unreasonably withheld, conditioned or delayed), (a) the Debtor shall accelerate the vesting of the Restricted Shares that do not vest automatically on the Effective Date, which Restricted Shares shall be entitled to receive their pro rata share of the Common Equity Distribution applicable to the unvested Restricted Shares shall be reserved and paid if and when such Restricted Shares vest.

ARTICLE IV. EVENTS LEADING TO THE CHAPTER 11 FILING

A. Industry-Wide Trends

At the time the Company entered into the MLSA in 2011, the business environment in the post-acute/skilled nursing sector was favorable due to a number of factors, including an aging population, expected increases in aggregate skilled nursing expenditures, and supply constraints in the skilled nursing sector due to substantial barriers to entry. The parties negotiated the amount of rent payable under the MLSA against this background.

Since that time, however, the operating environment for post-acute/skilled nursing facility operators has become significantly more challenging. Unfavorable trends for operators of skilled nursing facilities include (a) a shift away from a traditional fee-for-service model toward new managed-care models, which base reimbursement on patient outcome measures; (b) increased penetration of Medicare Advantage plans, which has reduced reimbursement rates, average length of stay and average daily census; (c) increased competition from alternative healthcare services such as home health agencies, life care at home, community-based service programs, senior housing, retirement communities and convalescent centers; and (d) reduction of reimbursement rates from government payors.

B. Transfers of Cash to the Long-Term Care Business from the 4H Business and the Outpatient Rehab and Other Businesses

This challenging business environment has resulted in decreasing revenues from the Long-Term Care Business and impeded HCR III's ability to pay rent under the MLSA. Beginning as early as June 2012, the "Rent Coverage Ratio" under the MLSA fell below the permitted threshold, which resulted in a "Rent Coverage Trigger Event." Also beginning in 2012, the Facilities' net cash flows in certain months were insufficient to allow the Sublease OpCos to make the required rent payments to HCR III under the subleases and, in turn, would have rendered HCR III unable to make its rent payments to QCP under the MLSA absent support

from the Company's other business segments. Since 2012, despite the parties' agreement upon certain deferrals of rent obligations from time to time, the Company's other business segments—the 4H Business and the Outpatient Rehab and Other Businesses—frequently advanced the additional funds necessary to permit HCR III to make its rent payments to QCP under the MLSA. Since 2011, HCR III and its subsidiaries have received more than \$500 million in intercompany funds from the Company's other business segments to assist HCR III and its subsidiaries in meeting their obligations. These transfers continued until April 2017, when a default occurred under the Prior Credit Facility and the lenders thereunder prohibited any further transfers of their cash collateral to the Long-Term Care Business, including, without limitation, for the payment of rent under the MLSA.

Unlike the Long-Term Care Business, the 4H Business and Outpatient Rehab and Other Businesses have continued to grow and, in the case of the 4H Business, has been increasingly profitable in recent years.

C. Prior Agreements and Amendments

In the years prior to the Debtor's commencement of the Chapter 11 Case, HCR III, HCP and its successor-in-interest under the MLSA, QCP, executed a series of agreements and amendments relating to the MLSA, which afforded HCR III certain deferrals and temporary reductions of their rent obligations under the MLSA. As detailed below, however, none of these agreements and amendments provided a long-term solution addressing HCR III's inability to meet its rent obligations under the MLSA (absent asset sales) and the financial difficulties facing the Long-Term Care Business.

1. Sale of Non-Strategic Facilities

On February 6, 2015, HCR III and HCP entered into a letter agreement whereby HCR III and HCP agreed to cooperate in selling at least forty-eight (48) non-strategic Facilities covered by the MLSA, resulting in a negotiated reduction of the amount of rent payable under the MLSA. Overall, fifty-one (51) Facilities were sold under the letter agreement, resulting in a \$27.3 million reduction in the annual rent under the MLSA.

2. Tenth Amendment to the MLSA

Effective April 1, 2015, HCR III and HCP entered into the Tenth Amendment to Master Lease and Security Agreement (the "<u>Tenth Amendment</u>"), which provided for, among other things, (a) certain deferred rent payments, (b) the resetting of rent escalations and (c) the extension of the initial term of the MLSA by five years. To temporarily reduce HCR III's monthly payments under the MLSA, the Tenth Amendment deferred \$525 million of HCR III's rent obligations (the "<u>Deferred Rent Obligation</u>"). The Deferred Rent Obligation was divided into two tranches of \$275 million (the "<u>Tranche A DRO</u>") and \$250 million (the "<u>Tranche B DRO</u>"), respectively. The Tenth Amendment provides that, together with each of its monthly rent payments, HCR III shall also make monthly payments of a portion of the Tranche A DRO and the Tranche B DRO.

Also as part of the Tenth Amendment, the Company agreed to sell to HCP nine of its long-term care facilities not subject to the MLSA for a purchase price of \$275 million. In

accordance with the terms of the Tenth Amendment, the Company used the proceeds of the sale to pre-pay the Tranche A DRO upon the closing of the sale transaction. On July 15, 2017, QCP provided notice to HCR III that, as a result of the occurrence of an event of default under the MLSA, the aggregate outstanding amount of the Tranche B DRO of approximately \$265.225 million, was immediately due and payable. The Tranche B DRO remained outstanding as of the Petition Date.

3. 2016–17 Rent Deferrals

HCR III and QCP agreed to certain additional rent deferrals in November 2016 (the "<u>Rent Deferral Agreement</u>"). Specifically, HCR III and QCP agreed to one-time deferral of rent, in the aggregate amount of \$30 million, which consisted of (a) a \$5 million deferral for November 2016, (b) a \$15 million deferral for December 2016 and (c) a \$10 million deferral for January 2017, subject to certain conditions.

4. April 2017 Forbearance Agreement

Following the execution of the Tenth Amendment and the Rent Deferral Agreement, the Sublease OpCos continued to have difficulty generating revenues sufficient to meet their rent obligations under the subleases and pay their ordinary-course operating expenses. Beginning in November 2016, various representatives of the Company and QCP, and their respective advisors, engaged in extensive negotiations regarding a potential long-term solution to the financial difficulties facing the Long-Term Care Business. To facilitate a continuing dialogue in furtherance of a consensual restructuring and to temporarily defer certain rent payments, QCP, HCR III and the Debtor entered into the Forbearance Agreement on April 5, 2017 (the "April 2017 Forbearance Agreement, QCP, HCR III and the Debtor agreed, among other things, that HCR III's rent obligations in excess of \$32 million per month would be deferred for the months of April, May and June 2017, with payment of the deferred portion thereof (approximately \$7.5 million per month) due on the earlier of (a) July 5, 2017 and (b) termination the April 2017 Forbearance Agreement.

D. Default and Acceleration of the Prior Credit Facility

As a result of the Company's ongoing financial difficulties and in the absence of a long-term restructuring of the MLSA to reduce rents, the Company's financial statements for the fiscal year ending December 31, 2016 were issued with a "going concern" qualification. This qualification was a default under the Prior Credit Facility, which was secured by substantially all of the assets of the 4H Business, the Outpatient Rehab and Other Businesses and the Company's "services" entities, including HCR Services. On May 4, 2017, the Company entered into a short-term forbearance agreement with an ad hoc group of lenders comprising a substantial majority of the commitments under the Prior Credit Facility in the interest of facilitating continued discussions regarding a longer-term forbearance. Ultimately, the parties failed to reach agreement on a longer-term forbearance, and, on May 16, 2017, the administrative agent under the Prior Credit Facility transmitted a notice of acceleration to the Company. The acceleration notice terminated all commitments under the Prior Credit Facility and declared all outstanding amounts under the Prior Credit Facility immediately due and payable.

E. The Company's Segregation of Cash and HCR III's Subsequent Rent Payments

Beginning in May 2017, HCR III began making monthly rent payments to QCP based on the cash generated by the Sublease OpCos after payments of the Sublease OpCos' operating expenses while retaining sufficient cash to continue operations of the Long-Term Care Business. The amount of available SNF Cash was not sufficient, however, to enable HCR III to meet its full rent obligations in any month since institution of the cash segregation protocol.

F. Filing of the Receivership Complaint and Entry Into the December 2017 Forbearance Agreement

1. July 2017 Default Notice

On July 7, 2017, QCP sent a notice to HCR III and the Debtor stating, among other things, that HCR III had failed to pay in full the rent due under the MLSA for April, May, June and July 2017, as well as late charges and interest with respect to such installments, totaling \$79,628,590, plus interest (the "Default Amount"). On July 15, 2017, QCP sent another notice to HCR III and the Debtor stating that their failure to pay the foregoing amount on or prior to July 14, 2017 had resulted in the occurrence of an event of default under the MLSA affecting all Facilities. As a result of the alleged event of default, QCP demanded, among other things, immediate payment of (a) the Default Amount and (b) the total amount of the Deferred Rent Obligation in the aggregate amount of \$265.225 million. QCP also demanded that the Debtor, as guarantor under the MLSA, immediately pay the Default Amount and all other amounts owed under the MLSA. QCP also noted, however, that the MLSA, the Agreements Regarding Subleases and the Guaranty remained in full force and effect and had not been terminated.

2. Receivership Complaint

On August 17, 2017, after months of continued negotiations without an agreement on a long-term restructuring, QCP filed a complaint against the Debtor, HCR III and "John Does 1–50" in Los Angeles County Superior Court, styled as *Quality Care Properties, Inc. and HCP Properties, LP* v. *HCR III Healthcare, LLC et al.*, Los Angeles County Super. Ct. No. BC672837, seeking, among other relief, the appointment of a receiver for HCR III (the "Receivership Proceeding").

On January 26, 2018, the Debtor filed a demurrer objecting to and requesting the dismissal of the Receivership Proceeding. On February 9, 2018, QCP filed an amended complaint. On February 15, 2018, the court issued a tentative case management order setting a schedule for the Receivership Proceeding, subject to comment by the parties. The Plan provides that the Receivership Proceeding shall be deemed dismissed with prejudice as of the Effective Date and that QCP shall cause the complaint filed in the Receivership Proceeding to be withdrawn from the relevant court docket as soon as practicable after the effective date Plan.

3. December 2017 Forbearance Agreement

Following the filing of the Receivership Complaint, the parties continued negotiations for a comprehensive restructuring of the Company. In furtherance of these discussions, QCP and the

Company entered into a series of stipulations extending the deadline by which the Company was required to file a response in the Receivership Proceeding. In addition, on December 22, 2017, QCP, HCR III, and the Debtor entered into that certain Forbearance Agreement and Amendment to Master Lease and Security Agreement, dated as of December 22, 2017 (the "December 2017 Forbearance Agreement"). Pursuant to the December 2017 Forbearance Agreement, QCP, HCR III, and the Debtor agreed to a reduced cash rent under the MLSA from November 30, 2017 until November 30, 2018 in the amount of \$23.5 million per month, with the remaining amounts of rent due during the period to be deferred.

The December 2017 Forbearance Agreement acknowledged HCR III's expectation that it would be unable to pay a material portion of Reduced Cash Rent during part or all of the period ending November 30, 2018.

4. Appointment of CRO and Independent Directors

In connection with its ongoing restructuring negotiations, the Company took additional steps to address issues raised by QCP with respect to the Company's corporate governance. Effective August 31, 2017, John R. Castellano was hired as Chief Restructuring Officer of the Debtor. Mr. Castellano is charged with leading and supporting the Company's restructuring efforts with the Debtor's board of directors, senior management of the Company, and other employees of the Company, among other duties. In addition, effective as of September 14, 2017, the Debtor appointed two new independent directors to its board, Sherman Edmiston III and Kevin P. Collins. The Debtor also established a special restructuring committee of the Debtor's board of directors to develop plans, proposals and recommendations for a financial restructuring of the Company, with the assistance of the Chief Restructuring Officer and the Company's other advisors, and provide direction to Company management in connection therewith. Mr. Edmiston and Mr. Collins, as the independent directors, are the two members of the special restructuring committee.

5. The Original Plan Sponsor Agreement and the Original Plan

After more than fifteen (15) months of extensive negotiations, on March 2, 2018, the Debtor and QCP agreed to a comprehensive restructuring of the Debtor in accordance with the terms of the Original Plan Sponsor Agreement, the Original Restructuring Support Agreement and the Original Plan. On March 3, 2018, the Debtor solicited votes to accept or reject the Original Plan from Holders of QCP Claims, the only Class of Claims entitled to vote under the Original Plan. Holders of QCP Claims unanimously voted to accept the Original Plan.

ARTICLE V. THE CHAPTER 11 CASE

A. Commencement of the Chapter 11 Case

Since the commencement of the Chapter 11 Case on the Petition Date, the Debtor has continued to operate the Company through its non-debtor direct and indirect subsidiaries in the ordinary course. On the Petition Date, the Debtor filed the Original Plan, the Original DS, and a motion requesting entry of an order (a) scheduling a combined hearing (the "Combined Hearing") to consider (i) approval of the Original DS, (ii) approval of the pre-petition solicitation

procedures, and (iii) confirmation of the Original Plan and (b) establishing confirmation-related deadlines (the "Original Scheduling Order"). The Court entered the Original Scheduling Order on March 6, 2018 and established April 13, 2018 as the date for the Combined Hearing.

On the Petition Date, the Debtor also filed motions (i) seeking authority to use cash collateral during the Chapter 11 Case, (ii) seeking authorization to continue using the Debtor's pre-petition bank account, (iii) requesting a waiver of the appointment of a patient care ombudsman, and (iv) seeking authority for non-debtors to continue making certain payments on behalf of the Debtor consistent with ordinary course, pre-petition business arrangements. The Court approved the relief requested in each of these motions.

The Debtor also obtained court authority to retain Sidley Austin LLP as counsel, Young Conaway Stargatt & Taylor, LLP as co-counsel, and Moelis & Company LLC, as investment banker, and obtained an order approving the Debtor's proposed interim compensation procedures.

B. Confirmation of Original Plan

After the Petition Date, the Debtor provided notice of the Combined Hearing and the deadline for objecting to the Original Plan to all known creditors, interest holders, and other parties in interest. The Debtor filed a plan supplement on March 29, 2018 containing (i) the proposed Certificate of Incorporation for the Reorganized Debtor, (ii) the proposed By-Laws for the Reorganized Debtor, and (iii) the identities of the individuals proposed to serve as directors and officers of the Reorganized Debtor. On April 10, 2018, the Debtor filed a revised exhibit to the plan supplement with the affiliations of the individuals proposed to serve as directors and officers as well as the nature of the compensation of any insiders of the Debtor proposed to serve as directors or officers of the Reorganized Debtor.

The Debtor received one objection, one reservation of rights, and several informal comments on the Original Plan. With the exception of the objection, the comments were all resolved consensually through language added to the Original Plan and/or the Original Confirmation Order. The one objection was filed by a plaintiff in a lawsuit against a non-debtor affiliate of the Debtor, and was overruled by the Bankruptcy Court at the Combined Hearing.

Prior to the Combined Hearing, the Debtor filed (i) a memorandum of law in support of confirmation of the Original Plan, (ii) a declaration of John R. Castellano, Chief Restructuring Officer of the Debtor, in support of the Original Plan, (iii) a supplemental declaration of John R. Castellano in support of the Original Plan, and (iv) a proposed form of the Original Confirmation Order.

On April 13, 2018, the Bankruptcy Court entered the Original Confirmation Order (i) approving the Original DS, (ii) approving the pre-petition solicitation procedures, and (iii) confirming the Original Plan. In the Original Confirmation Order, the Bankruptcy Court also waived the requirement that the Debtor file statements of financial affairs, schedules of assets and liabilities, and reports required by Bankruptcy Rule 2015.3, and waived the meeting of creditors under Section 341 of the Bankruptcy Code.

C. Alternative Plan Sponsor Agreement and the Plan.

Shortly prior to the entry of the Original Confirmation Order, Welltower and ProMedica Parent submitted an offer to acquire QCP. Welltower, ProMedica Parent, QCP and the Debtor then entered into negotiations regarding the Alternative Transaction. On April 25, 2018, the applicable parties entered into the Merger Agreement, the Alternative Plan Sponsor Agreement and the Alternative Restructuring Support Agreement. Additionally, on April 25, 2018, the Debtor, QCP and HCP Mezzanine Lender, LP entered into the Amendment to the Plan Sponsor Agreement, which amended the Original Plan Sponsor Agreement to allow the parties to pursue the Alternative Transaction and extended the termination date for the Original Plan Sponsor Agreement from September 30, 2018 to January 15, 2019.

On May 2, 2018, the Debtor filed the Plan, which revises the Original Plan to facilitate the Alternative Transaction. If the Alternative Transaction does not close, the treatment of Claims and Interests and other terms of the Plan will revert to the treatment provided in the Original Plan.

On May 18, 2018, the Bankruptcy Court entered an order (the "Amended Solicitation Procedures Order") approving the Disclosure Statement and establishing solicitation and confirmation-related deadlines and procedures. Additionally, the Amended Solicitation Procedures Order authorized the Debtor to enter into the Alternative Plan Sponsor Agreement, the Alternative Restructuring Support Agreement, and the Amendment to the Original Plan Sponsor Agreement pursuant to sections 105(a) and 363(b) of the Bankruptcy Code.

ARTICLE VI. OTHER KEY ASPECTS OF THE PLAN AND THE PLAN SPONSOR AGREEMENTS

A. Distributions

One of the key concepts under the Bankruptcy Code is that only claims and interests that are "allowed" may receive distributions under a chapter 11 plan. This term is used throughout the Plan and the descriptions below. In general, an Allowed Claim or Interest means that the Debtor agrees, or if there is a dispute, the Bankruptcy Court determines, that the Claim or Interest, and the amount thereof, is in fact a valid obligation of the Debtor. Except as otherwise provided in the Plan, a Final Order, or as otherwise agreed to by the relevant parties, on the Effective Date, the Debtor shall make initial distributions under the Plan on account of Allowed Claims, including those that become Allowed as of the Effective Date, subject to the Reorganized Debtor's right to object to Claims. A more detailed discussion of the treatment and anticipated means of satisfaction for each Class of Allowed Claims or Interests is set forth in ARTICLE I.G and ARTICLE I.H above.

During the period from the Confirmation Date through and until the Effective Date, the Debtor may continue to operate its business as debtor in possession in the ordinary course in a manner consistent with its obligations under the applicable Plan Sponsor Agreement and the transactions contemplated by the Plan and the Plan Sponsor Agreement, subject to all applicable orders of the Bankruptcy Court and the provisions of the Bankruptcy Code.

B. Restructuring Transactions

1. Alternative Transaction

Under the Alternative Transaction, ProMedica Sub (or its designee(s)), on the Effective Date, shall provide the ProMedica Plan Contribution in exchange for 1,000,000 newly issued shares of the New Common Stock of the Reorganized Debtor, par value \$0.01 per share, which shall constitute all of the issued and outstanding capital stock and rights to purchase or otherwise acquire capital stock of the Debtor, free and clear of any lien, charge, pledge, security interest, claim, or other encumbrance. Under the Alternative Transaction, ProMedica Sub (or its designee(s)) will receive the New Common Stock in accordance with the Implementation Memorandum.

2. Original Transaction

If the Closing of the Alternative Transaction does not occur and the Alternative Plan Sponsor Agreement is terminated in accordance with its terms, the Debtor shall consummate the Original Transaction, subject to the terms and conditions of the Original Plan Sponsor Agreement, pursuant to which QCP or its designee(s) shall receive, and the Debtor shall issue, transfer, and deliver to QCP or its designee(s), 1,000,000 newly issued shares of New Common Stock of the Debtor, par value \$0.01 per share, which shall constitute all of the issued and outstanding capital stock and rights to purchase or otherwise acquire capital stock of the Debtor, free and clear of any lien, charge, pledge, security interest, claim, or other Encumbrance, in full satisfaction of all of the QCP Claims.

3. Master Lease

If the Closing of the Alternative Transaction occurs, on the Effective Date, the Reorganized Debtor shall cause HCR III to, and Meerkat I LLC (or its designee(s)) and the Lessors shall, enter into the Alternative Master Lease, substantially in accordance with the terms set forth in Exhibit A to the Plan. The Alternative Master Lease will replace and supersede the existing MLSA in its entirety. On or prior to the third day after the Effective Date, QCP shall cause the Lessors and HCR III to terminate, release and discharge the Guaranty pursuant to the Alternative Plan Sponsor Agreement.

Under the Alternative Master Lease, HCR III, as lessee, will be obligated to pay monthly base rent as set forth in Schedule 1 to the Alternative Master Lease. HCR III's obligations under the Alternative Master Lease will not be guaranteed by the Reorganized Debtor, but will be guaranteed by ProMedica Parent.

If the Closing of the Alternative Transaction does not occur, on the Effective Date, the Reorganized Debtor shall cause HCR III to, and QCP and the Lessors shall, enter into the Original Master Lease Amendment, substantially in the form of Exhibit C to the Plan. HCR III's obligations under the Original Master Lease shall not be guaranteed, and, on or prior to the seventh day after the Effective Date, QCP shall cause the Lessors and HCR III to terminate, release and discharge the Guaranty pursuant to the Original Plan Sponsor Agreement.

Under the Original Master Lease, HCR III, as lessee, will be obligated to pay, as partial payment of minimum rent for each month, all monthly free cash flow for such month, with the remaining minimum rent for each calendar month to be added to the amount of a new deferred rent obligation. The new deferred rent obligation will be due and payable on the earliest to occur of the anniversary of the Effective Date in 2025 or the occurrence of certain other specified events. The existing Tranche B DRO will be eliminated upon the Effective Date.

4. Corporate Governance, Directors and Officers

a. Certificate of Incorporation; By-Laws

On the Effective Date, the Certificate of Incorporation and the By-Laws shall go into effect. Consistent with, but only to the extent required by, section 1123(a)(6) of the Bankruptcy Code, on the Effective Date, the Certificate of Incorporation will be amended to prohibit the issuance of non-voting equity securities. After the Effective Date, the Reorganized Debtor may amend and restate its certificates or articles of incorporation, by-laws, or similar governing documents, as applicable, as permitted by applicable law.

b. Directors and Officers of Reorganized Debtor

Subject to any requirement of Bankruptcy Court approval pursuant to section 1129(a)(5) of the Bankruptcy Code, the initial directors and officers of the Reorganized Debtor on the Effective Date shall be as set forth on (i) **Exhibit E** to the Plan, if the Closing of the Alternative Transaction occurs, or (ii) **Exhibit F** to the Plan if the Closing of the Alternative Transaction does not occur. After the Effective Date, the Certificate of Incorporation and the By-Laws, as each may be amended thereafter from time to time, shall govern the designation and election of directors.

5. Corporate Action

On the Effective Date, (a) the selection of directors and officers for Reorganized Debtor, (b) the issuance and distribution of the New Common Stock, and (c) all other actions and transactions contemplated by the Plan and the Plan Sponsor Agreement shall be deemed authorized and approved in all respects (subject to the provisions of the Plan and the Plan Sponsor Agreement). All matters provided for in the Plan and the Plan Sponsor Agreement involving the corporate structure of the Debtor or the Reorganized Debtor, and any corporate action required by the Debtor or the Reorganized Debtor in connection with the Plan and the Plan Sponsor Agreement, shall be deemed to have timely occurred in accordance with applicable law and shall be in effect, without any requirement of further action by the security holders or directors of the Debtor or the Reorganized Debtor. On and after the Effective Date, the appropriate officers of the Reorganized Debtor and members of the boards of directors or managers of the Reorganized Debtor shall be authorized and directed to issue, execute and deliver the agreements, documents, securities and instruments contemplated by the Plan and the Plan Sponsor Agreement in the name of and on behalf of the Reorganized Debtor.

C. Regulatory Approvals

The Debtor and the business conducted by certain of its affiliates are subject to regulatory oversight by a number of Governmental Units. Among other conditions precedent to the Effective Date, the effectiveness of the Plan is subject to obtaining certain governmental approvals, except to the extent such governmental approvals would not, in the aggregate, reasonably be expected to result in losses, costs, liabilities or expenses to certain parties to the applicable Plan Sponsor Agreement and their respective subsidiaries (as set forth in the applicable Plan Sponsor Agreement) in excess of \$5,000,000. The Debtor and the Plan Sponsors anticipate seeking each of the necessary governmental approvals from the relevant Governmental Units as soon as practicable.

D. Treatment of Executory Contracts and Insurance Policies

1. Assumption or Rejection of Executory Contracts

On the Effective Date, all Executory Contracts of the Debtor will be deemed assumed in accordance with, and subject to, the provisions and requirements of sections 365 and 1123 of the Bankruptcy Code, unless such Executory Contract (a) was previously assumed or rejected by the Debtor, (b) previously expired or terminated pursuant to its own terms, or (c) upon the applicable Plan Sponsor's prior written consent or request, is subject to a motion to reject such Executory Contract filed prior to the Effective Date. Entry of the Confirmation Order by the Bankruptcy Court shall constitute approval of such assumptions and the rejection of any Executory Contract for which a motion to reject has been filed, pursuant to sections 365(a) and 1123 of the Bankruptcy Code. Each Executory Contract assumed pursuant to Article VI of the Plan shall revest in and be fully enforceable by the Reorganized Debtor in accordance with its terms, except as modified by the provisions of the Plan, or any order of the Bankruptcy Court authorizing and providing for its assumption. Any motions to assume or reject Executory Contracts pending on the Effective Date shall be subject to approval by the Bankruptcy Court on or after the Effective Date. To the maximum extent permitted by law, to the extent any provision in any Executory Contract assumed pursuant to the Plan restricts or prevents, or purports to restrict or prevent, or is breached or deemed breached by, the assumption of such Executory Contract, including any "change of control" provision, then such provision shall be deemed modified such that the transactions contemplated by the Plan shall not entitle the non-Debtor party thereto to terminate such Executory Contract or to exercise any other default-related rights with respect thereto.

2. No Cure Obligations

There are no anticipated cure obligations with respect to any Executory Contract to which the Debtor is a party. Unless otherwise determined by the Bankruptcy Court pursuant to a Final Order or agreed to by the parties thereto prior to the Effective Date, no payments will be required to cure any defaults of the Debtor existing as of the Confirmation Date with respect to any Executory Contract assumed pursuant to the Plan. Subject to the occurrence of the Effective Date, the entry of the Confirmation Order will be deemed to constitute a finding by the Bankruptcy Court that (i) each such assumption is in the best interest of the Debtor and its Estate, and (ii) the requirements of section 365(b)(l) of the Bankruptcy Code are deemed satisfied.

3. Insurance Policies and Agreements

Insurance policies issued to, or insurance agreements entered into by, the Debtor prior to the Petition Date (including, without limitation, any policies covering directors' or officers' conduct) will continue in effect after the Effective Date. To the extent that such insurance policies or agreements are considered to be Executory Contracts, the Plan will constitute a motion to assume or ratify such insurance policies and agreements, and, subject to the occurrence of the Effective Date, the entry of the Confirmation Order will constitute approval of such assumption pursuant to section 365(a) of the Bankruptcy Code and a finding by the Bankruptcy Court that each such assumption is in the best interest of the Debtor and its Estate. Unless otherwise determined by the Bankruptcy Court pursuant to a Final Order or agreed to by the parties thereto prior to the Effective Date, no payments will be required to cure any defaults of the Debtor existing as of the Confirmation Date with respect to each such insurance policy.

4. Post-Petition Contracts and Leases

All contracts, agreements and leases that were entered into by the Debtor or assumed by the Debtor after the Petition Date will be deemed assigned by the Debtor to the Reorganized Debtor on the Effective Date.

5. Modifications, Amendments, Supplements, Restatements or Other Agreements

Unless otherwise provided in the Plan, each Executory Contract that is assumed shall include all modifications, amendments, supplements, restatements or other agreements that in any manner affect such Executory Contract including easements, licenses, permits, rights, privileges, immunities, options, rights of first refusal and any other interests, unless any of the foregoing agreements has been previously rejected or repudiated or is rejected or repudiated under the Plan. Modifications, amendments, supplements and restatements to prepetition Executory Contracts that have been executed by the Debtor during the Chapter 11 Case shall not be deemed to alter the prepetition nature of the Executory Contract, or the validity, priority or amount of any Claims that may arise in connection therewith.

E. Discharges, Releases, Exculpations and Injunctions; Survival of Indemnification and Exculpation Obligations

1. Discharges

a. Discharge of Claims and Termination of Interests

Except as otherwise provided in the Plan or in the Confirmation Order including with respect to any Claims that are Reinstated under the Plan, all consideration distributed under the Plan shall be in exchange for all Claims and Interests of any nature whatsoever, whether known or unknown, against the Debtor or its Estate, assets, properties or interest in property, and shall constitute a complete satisfaction and settlement of all Claims and Interests other than Reinstated Claims, in each case regardless of whether any property shall have been distributed or retained pursuant to the Plan on account of such Claims and Interests. On the Effective Date, except for Reinstated Claims, the Debtor shall be deemed

discharged and released under section 1141(d)(l)(A) of the Bankruptcy Code from any and all Claims and Interests, including, but not limited to, demands and liabilities that arose before the Effective Date, and all debts of the kind specified in sections 502(g), 502(h) or 502(i) of the Bankruptcy Code, Section 510(b) Claims, General Unsecured Claims, and Interests in the Debtor.

b. **Discharge Injunction**

As of the Effective Date, except as otherwise expressly provided in the Plan or the Confirmation Order, all Entities (other than holders of Reinstated Claims solely in their capacities as such) shall be precluded from asserting against the Debtor or the Reorganized Debtor and their respective assets and property or the Estate, any other or further Claims (other than those Reinstated under the Plan) or Interests, or any other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities of any nature whatsoever, relating to the Debtor or Reorganized Debtor or any of their respective assets and property or the Estate, based upon any act, omission, transaction or other activity of any nature that occurred prior to the Effective Date. In accordance with the foregoing, except as expressly provided in the Plan or the Confirmation Order, the Confirmation Order shall constitute a judicial determination, as of the Effective Date, of the discharge of all non-Reinstated Claims or other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities, pursuant to sections 524 and 1141 of the Bankruptcy Code, and such discharge shall void and extinguish any judgment obtained against the Debtor, the Reorganized Debtor, or its respective assets, property and Estate at any time, to the extent such judgment is related to a discharged Claim, debt, liability or Interest. Except as otherwise specifically provided in the Plan or the Confirmation Order, all Persons or Entities who have held, hold or may hold Claims or Interests that arose prior to the Effective Date and all other parties-in-interest, along with their respective present or former employees, agents, officers, directors, principals, representatives and Affiliates, are permanently enjoined, from and after the Effective Date, from (i) commencing or continuing in any manner any action or other proceeding of any kind with respect to any such Claim (including a Section 510(b) Claim) against or Interest in the Reorganized Debtor or property of the Reorganized Debtor, other than to enforce any right to a distribution pursuant to the Plan, (ii) the enforcement, attachment, collection or recovery by any manner or means of any judgment, award, decree or order against the Reorganized Debtor or property of the Reorganized Debtor, other than to enforce any right to a distribution pursuant to the Plan, (iii) creating, perfecting or enforcing any Lien or encumbrance of any kind against the Reorganized Debtor or against the property or interests in property of the Reorganized Debtor, other than to enforce any right to a distribution pursuant to the Plan or (iv) asserting any right of setoff or subrogation of any kind against any obligation due from the Reorganized Debtor or against the property or interests in property of the Reorganized Debtor, with respect to any such Claim or Interest. Such injunction shall extend to any successors or assignees of the Reorganized Debtor and their respective properties and interest in properties. For the avoidance of doubt, the provisions of Section 9.2.2 of the Plan shall not apply with respect to Claims that are Reinstated under the Plan including, without limitation, the Credit Facility Claims (if the Closing of the Alternative Transaction does not occur) and the Intercompany Note Claim, as applicable.

2. Releases

a. Releases by the Debtor

Except as otherwise expressly provided in the Plan or the Confirmation Order, on the Effective Date, for good and valuable consideration, to the fullest extent permissible under applicable law, each of the Debtor and Reorganized Debtor on its own behalf and as a representative of its respective Estate, shall, and shall be deemed to, completely and forever release, waive, void, extinguish and discharge unconditionally, each and all of the Released Parties of and from any and all Claims and Causes of Action (including, without limitation, Avoidance Actions), any and all other obligations, suits, judgments, damages, debts, rights, remedies, causes of action and liabilities of any nature whatsoever, whether liquidated or unliquidated, fixed or contingent, direct or derivative, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are or may be based in whole or part on any act, omission, transaction, event or other circumstance taking place or existing on or prior to the Effective Date (including prior to the Petition Date) in connection with or related to the Debtor, the Reorganized Debtor, their respective assets and property, and the Estate, the Chapter 11 Case, the Plan, the Plan Supplement or this Disclosure Statement that may be asserted by or on behalf of the Debtor, the Reorganized Debtor or the Estate against any of the Released Parties; provided, however, that nothing in Section 9.3 of the Plan shall be construed to release any party from fraud, willful misconduct or gross negligence as determined by a Final Order.

b. Releases by Certain Holders of Claims

Except as otherwise expressly provided in the Plan or the Confirmation Order, on the Effective Date, for good and valuable consideration, to the fullest extent permissible under applicable law, each Holder of a Claim or an Interest that votes to accept the Plan shall be deemed to have completely and forever released, waived, and discharged unconditionally each of the Released Parties of and from any and all Claims, any and all other obligations, suits, judgments, damages, debts, rights, remedies, causes of action and liabilities of any nature whatsoever (including, without limitation, those arising under the Bankruptcy Code), whether liquidated or unliquidated, fixed or contingent, direct or derivative, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are or may be based in whole or part on any act, omission, transaction, event or other circumstance taking place or existing on or prior to the Effective Date (including prior to the Petition Date) in connection with or related to the Debtor, the Reorganized Debtor or their respective assets and property, and the Estate, the Chapter 11 Case, the Plan, the Plan Supplement, and/or this Disclosure Statement; provided, however, that nothing in Section 9.4 of the Plan shall be construed to release (x) any party from fraud, willful misconduct or gross negligence as determined by a Final Order, (y) any Reinstated Claims, including, without limitation, the Credit Facility Claims (if the Closing of the Alternative Transaction does not occur) and the Intercompany Note Claim, as applicable or (z) any obligations under the Plan Sponsor Agreement or the Merger Agreement.

3. Exculpations

a. **Exculpation**

From and after the Effective Date, the Exculpated Fiduciaries and, solely to the extent provided by section 1125(e) of the Bankruptcy Code, the Section 1125(e) Parties, shall neither have nor incur any liability to, or be subject to any right of action by, any Holder of a Claim or an Interest, or any other party in interest, or any of their respective employees, representatives, financial advisors, attorneys, or agents acting in such capacity, or Affiliates, or any of their successors or assigns, for any act or omission in connection with, relating to, or arising out of, the Chapter 11 Case, formulating, negotiating or implementing the Plan and/or previous iterations hereof, the Plan Supplement, this Disclosure Statement and/or previous iterations thereof, the Alternative Plan Sponsor Agreement, the Amendment to the Original Plan Sponsor Agreement, the Original Plan Sponsor Agreement, the Restructuring Support Agreement, the Alternative Master Lease or the Original Master Lease (as applicable), the solicitation of acceptances of the Plan and/or previous iterations hereof, the pursuit of Confirmation of the Plan, the Confirmation of the Plan, the consummation of the Plan, the administration of the Plan, the property to be distributed under the Plan, the consummation of the transactions contemplated by the Alternative Plan Sponsor Agreement, the Original Plan Sponsor Agreement, or any other act taken or omitted to be taken in connection with or in contemplation of the Chapter 11 Case or implementation of the Plan; provided, however, that Section 9.5 of the Plan shall not apply to release (x) obligations under the Plan, and obligations under the Alternative Plan Sponsor Agreement, the Amendment to the Original Plan Sponsor Agreement, the Original Plan Sponsor Agreement, the Restructuring Support Agreement, the Alternative Master Lease or the Original Master Lease (as applicable), the Merger Agreement and the contracts, instruments, releases, agreements, and documents delivered, Reinstated or assumed under the Plan (including, without limitation, the Credit Facility (if the Closing of the Alternative Transaction does not occur) and the Intercompany Note, as applicable), and (y) any Claims or Causes of Action arising out of fraud, willful misconduct or gross negligence as determined by a Final Order.

Any of the Exculpated Parties shall be entitled to rely, in all respects, upon the reasonable and informed advice of counsel with respect to their duties and responsibilities under the Plan.

b. **Injunctions**

(i) **Injunction Related to Exculpation**

Except as expressly provided in the Plan or the Confirmation Order, as of the Effective Date, all Persons and Entities that hold, have held, or may hold a Claim or any other obligation, suit, judgment, damages, debt, right, remedy, Cause of Action or liability of any nature whatsoever, of the types described in Section 9.5 of the Plan and relating to the Debtor, the Reorganized Debtor or any of their respective assets and property and/or the Estate, are, and shall be, permanently, forever and completely stayed, restrained, prohibited, barred and enjoined from taking any of the following actions against any

Exculpated Party or its property on account of such released liabilities, whether directly or indirectly, derivatively or otherwise, on account of or based on the subject matter of such discharged Claims or other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities: (i) commencing, conducting or continuing in any manner, directly or indirectly, any suit, action or other proceeding (including, without limitation, any judicial, arbitral, administrative or other proceeding) in any forum; (ii) enforcing, attaching (including, without limitation, any prejudgment attachment), collecting, or in any way seeking to recover any judgment, award, decree, or other order; (iii) creating, perfecting or in any way enforcing in any matter, directly or indirectly, any Lien; (iv) setting off, seeking reimbursement or contributions from, or subrogation against, or otherwise recouping in any manner, directly or indirectly, any amount against any liability or obligation that is discharged under Section 9.2 of the Plan; and/or (v) commencing or continuing in any manner any judicial, arbitration or administrative proceeding in any forum, that does not comply with or is inconsistent with the provisions of the Plan or the Confirmation Order.

(ii) Injunction Related to Releases

Except as expressly provided in the Plan or the Confirmation Order, as of the Effective Date, all Persons and Entities that hold, have held, or may hold a Claim or any other obligation, suit, judgment, damages, debt, right, remedy, Cause of Action or liability of any nature whatsoever, of the types described in Section 9.4 of the Plan and relating to the Debtor, the Reorganized Debtor or any of their respective assets and property and/or the Estate, the Chapter 11 Case, the Plan, the Plan Supplement and/or this Disclosure Statement are, and shall be, permanently, forever and completely stayed, restrained, prohibited, barred and enjoined from taking any of the following actions against any Released Party or its property on account of such released liabilities, whether directly or indirectly, derivatively or otherwise, on account of or based on the subject matter of such discharged Claims or other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities: (i) commencing, conducting or continuing in any manner, directly or indirectly, any suit, action or other proceeding (including, without limitation, any judicial, arbitral, administrative or other proceeding) in any forum; (ii) enforcing, attaching (including, without limitation, any prejudgment attachment), collecting, or in any way seeking to recover any judgment, award, decree, or other order; (iii) creating, perfecting or in any way enforcing in any matter, directly or indirectly, any Lien; (iv) setting off, seeking reimbursement or contributions from, or subrogation against, or otherwise recouping in any manner, directly or indirectly, any amount against any liability or obligation that is discharged under Section 9.2 of the Plan; and/or (v) commencing or continuing in any manner any judicial, arbitration or administrative proceeding in any forum, that does not comply with or is inconsistent with the provisions of the Plan or the **Confirmation Order.**

4. Survival of Indemnification and Exculpation Obligations

The obligations of the Debtor to indemnify and exculpate any past and present directors, officers, agents, employees and representatives who provided services to the Debtor prior to or after the Petition Date, pursuant to certificates or articles of incorporation, by-laws, contracts

and/or applicable statutes, in respect of all actions, suits and proceedings against any of such officers, directors, agents, employees and representatives, based upon any act or omission related to service with, for or on behalf of the Debtor, shall not be discharged or Impaired by Confirmation or consummation of the Plan and shall be assumed by the Reorganized Debtor. For the avoidance of doubt, Section 9.7 of the Plan affects only the obligations of the Debtor and Reorganized Debtor with respect to any indemnity or exculpation owed to or for the benefit of past and present directors, officers, agents, employees and representatives of the Debtor, and shall have no effect on nor in any way discharge or reduce, in whole or in part, any obligation of any other Person, including any provider of director and officer insurance, owed to or for the benefit of such past and present directors, officers, agents, employees and representatives of the Debtor. For further avoidance of doubt, the provisions regarding indemnification, exculpation and directors' and officers' insurance contained in section 4.7 of the Plan Sponsor Agreement shall be deemed incorporated into the Plan as if set forth fully therein.

5. Non-Debtor Subsidiary Releases

Pursuant to Section 3.5 of the Alternative Plan Sponsor Agreement and Section 3.4 of the Original Plan Sponsor Agreement, QCP, the Debtor and, in the event the Alternative Transaction closes, ProMedica and each Lessor (as defined in the Alternative Plan Sponsor Agreement) have agreed that, on the Effective Date of the Alternative Transaction or the Original Transaction, as applicable, each direct and indirect subsidiary of the Debtor shall be deemed to have completely and forever released, waived, and discharged unconditionally each of the shareholders or other equity holders of the Debtor that is not an employee of the Debtor and each director of the Debtor that is not an employee of the Debtor, and any Person claiming by or through any of the foregoing Persons (including their respective officers, directors, managers, shareholders, partners, employees, equity holders members, and professionals (the "Non-Debtor Subsidiary Released Parties"), and, if the Alternative Transaction closes, QCP, ProMedica and each Lessor under the Alternative Plan Sponsor Agreement, or, if the Original Transaction closes, QCP, HCP Mezzanine Lender, LP and the lessors under the Original Plan Sponsor Agreement, shall be deemed to have completely and forever released, waived, and discharged unconditionally each Non-Debtor Subsidiary Released Party, of and from any and all Claims (as such term is defined in the Plan) and all other obligations, suits, judgments, damages, debts, rights, remedies, causes of action and liabilities of any nature whatsoever (including, without limitation, those arising under the Bankruptcy Code), whether liquidated or unliquidated, fixed or contingent, direct or derivative, matures or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are or may be based in whole or part on any act, omission, transaction, event or other circumstance taking place or existing on or prior to the Effective Date (including prior to the Petition Date) in connection with or related to the Debtor, the Reorganized Debtor, their respective Subsidiaries, assets, property and estates, the Bankruptcy Case or the Restructuring Documents (as defined in the Alternative Plan Sponsor Agreement or the Original Plan Sponsor Agreement, as applicable).

F. Excess Severance Payments; Former CEO Settlement Agreement

Section 1.5 of both the Alternative Plan Sponsor Agreement and the Original Plan Sponsor Agreement provides, in pertinent part, that the applicable Plan Sponsor will establish an account to fund Excess Severance Payments (as defined in the applicable Plan Sponsor

Agreement) and, following the Effective Date, shall cause such account to be funded with all amounts necessary to make the Excess Severance Payments and, subject to certain conditions and requirements set forth in the applicable Plan Sponsor Agreement, cause the Excess Severance Payments to be paid to each of the current Chief Executive Officer, Chief Financial Officer, General Counsel and Chief Operating Officer of the Debtor.

The Debtor also intends to, with the consent of the Plan Sponsor, offer to the Former CEO (as defined in the Plan Sponsor Agreement) a settlement agreement in the form attached to the Alternative Plan Sponsor Agreement as Exhibit 4 and to the Original Plan Sponsor Agreement as Exhibit 3 (the "Former CEO Settlement Agreement"), which, if entered into among the Debtor, Plan Sponsor and the Former CEO, would entitle the Former CEO to certain amounts and payments in respect of the Former CEO's claimed interests in and entitlements to certain deferred compensation, severance and other benefits. Under the terms of the applicable Plan Sponsor Agreement and the Former CEO Settlement Agreement, such amounts and payments would be made to the Former CEO no earlier than the Effective Date, but, if the Closing of the Alternative Transaction occurs, prior to the distribution and transfer of the New Common Stock to ProMedica Sub or its designee(s).

ARTICLE VII. CONFIRMATION PROCEDURES

A. Hearing on Confirmation

Under section 1128(a) of the Bankruptcy Code, the Bankruptcy Court, after notice, may hold a hearing to confirm a plan of reorganization. Pursuant to the Amended Solicitation Procedures Order the Bankruptcy Court set a hearing for June 21, 2018 at 10:00 a.m. (prevailing Eastern Time) to consider confirmation of the Plan (the "Confirmation Hearing"). The Confirmation Hearing, once set, may be continued from time to time without further notice other than an adjournment announced in open court or a notice of adjournment filed with the Bankruptcy Court and served on those parties who have requested notice under Bankruptcy Rule 2002 and the entities who have filed an objection to the Plan, if any, without further notice to parties in interest. The Bankruptcy Court, in its discretion and prior to the Confirmation Hearing, may put in place additional procedures governing the Confirmation Hearing. Subject to section 1127 of the Bankruptcy Code and the applicable Plan Sponsor Agreement, the Plan may be modified, if necessary, prior to, during, or as a result of the Confirmation Hearing, without further notice to parties in interest.

Additionally, section 1128(b) of the Bankruptcy Code provides that any party in interest may object to Confirmation. Pursuant to the Amended Solicitation Procedures Order, the Bankruptcy Court set June 15, 2018 at 4:00 p.m. (prevailing Eastern Time) as the deadline for parties in interest to file objections to confirmation of the Plan. All objections to the Plan must be filed with the Bankruptcy Court and served on the Debtor and certain other parties in interest in accordance with the applicable order of the Bankruptcy Court so that they are received on or before the deadline to file such objections as set forth therein.

B. Requirements for Confirmation of the Plan

Among the requirements for Confirmation are that the Plan is accepted by all Impaired Classes of Claims and Interests or, if rejected by an Impaired Class, that the Plan "does not discriminate unfairly" and is "fair and equitable" as to such Class, is feasible, and is in the "best interests" of Holders of Claims and Interests that are Impaired under the Plan. The following requirements must be satisfied pursuant to section 1129(a) of the Bankruptcy Code before the Bankruptcy Court may confirm a plan of reorganization. The Plan fully complies with the statutory requirements for Confirmation listed below.

- The proponents of the Plan have complied with the applicable provisions of the Bankruptcy Code.
- The Plan has been proposed in good faith and not by any means forbidden by law.
- Any payment made or to be made by the Debtor (or any other proponent of the Plan) or by a Person issuing Securities or acquiring property under the Plan, for services or for costs and expenses in or in connection with the Chapter 11 Case, in connection with the Plan and incident to the Chapter 11 Case is subject to the approval of the Bankruptcy Court as reasonable.
- The Debtor has disclosed the identity and affiliations of any individual proposed to serve, after Confirmation, as a director, or officer, the Reorganized Debtor, any Affiliate of the Debtor reorganized under the Plan, or any successor to the Debtor under the Plan, and the appointment to, or continuance in, such office of such individual is consistent with the interests of creditors and Holders of Interests and with public policies.
- The Debtor has disclosed the identity of any Insider that will be employed or retained the Reorganized Debtor and the nature of any compensation for such Insider.
- With respect to each Holder within an Impaired Class of Claims or Interests, each such Holder has accepted the Plan or (b) will receive or retain under the Plan on account of such Claim or Interest property of a value, as of the Effective Date, that is not less than the amount that such Holder would so receive or retain if the Debtor was liquidated under chapter 7 of the Bankruptcy Code on such date.
- With respect to each Class of Claims or Interests, such Class (a) has accepted the Plan or is Unimpaired under the Plan (subject to the "cramdown" provisions discussed below).
- The Plan provides for treatment of Claims, as applicable, in accordance with the provisions of section 507(a) of the Bankruptcy Code.

- If a Class of Claims or Interests is Impaired under the Plan, at least one Class of Claims or Interests that is Impaired under the Plan has accepted the Plan, determined without including any acceptance of the Plan by any Insider.
- Confirmation is not likely to be followed by the liquidation, or the need for further financial reorganization, of the Reorganized Debtor, or any successor to the Debtor under the Plan, unless such liquidation or reorganization is proposed in the Plan.
- All fees payable under 28 U.S.C. § 1930 have been paid or the Plan provides for the payment of all such fees on the Effective Date.

C. Best Interests of Creditors / Liquidation Analysis

Section 1129(a)(7) of the Bankruptcy Code requires that each Holder of an Impaired Claim or Interest either (a) accept the Plan or (b) receive or retain under the Plan property of a value, as of the Effective Date, that is not less than the value such Holder would receive if the Debtor was liquidated under chapter 7 of the Bankruptcy Code. Four Classes of Claims or Interests (other than Intercompany Claims) are Impaired under the Plan: QCP Claims (Class 4), Section 510(b) Claims (Class 7), Preferred Interests (Class 8A), and Common Interests (Class 8B). The Holders of QCP Claims must accept the Plan in order for the Plan to be confirmed, thereby satisfying clause (a) above. The Holders of Section 510(b) Claims, Preferred Interests and Common Interests of the Debtor would receive no distribution if the Debtor were liquidated under chapter 7 of the Bankruptcy Code for the reasons explained below. However, the Holders of Preferred Interests and Common Interests are entitled to vote on the Plan. Holders of more than 80% of the Common Interests in the Debtor affirmatively support the Plan, and the Debtor is unaware of any Section 510(b) Claims.

If the Debtor were liquidated under chapter 7, QCP's Claim against the Debtor under the MLSA Guarantee is estimated to be no less than the amount of the QCP Claim plus, in accordance with section 502(b)(6) of the Bankruptcy Code, damages equal to the rent reserved by the MLSA, without acceleration, for the greater of one year, or fifteen percent (15%), not to exceed three years, of the remaining post-petition term of the MLSA (the "QCP Chapter 7 Claim"). The QCP Chapter 7 Claim would need to be paid in full before the Holders of Section 510(b) Claims and Interests of the Debtor would be entitled to any distribution in the chapter 7 case.

Other than cash on hand, the Debtor's only material asset is the stock (the "Subsidiary Stock") of its direct subsidiary HCR ManorCare Operations II, LLC (the "Subsidiary"), which is an intermediate holding company. In a hypothetical case under chapter 7 of the Bankruptcy Code, the Debtor would be required to sell the Subsidiary Stock. The value of the Subsidiary Stock derives from the going-concern value of the Company's businesses: the Long-Term Care Business, the 4H Business and the Outpatient Rehab and Other Businesses. The Long-Term Care Business, which historically has generated approximately 80% of the Company's revenues, is operated through subsidiaries of HCR III (*i.e.*, the Sublease OpCos), the lessee under the MLSA. As explained above, the Long-Term Care Business is unable to generate sufficient cash flows for the Sublease OpCos to make their sublease payments to HCR III and,

in turn, for HCR III to make the rent and other payments owing to QCP under the current version of the MLSA. Absent a restructuring of the MLSA, HCR III would continue to be unable to meet its obligations to QCP under the MLSA and would continue to be insolvent. Therefore, in a chapter 7 liquidation of the Debtor, the MLSA obligations owing to QCP would not be satisfied in full and therefore all of the value of the Long-Term Care Business would be absorbed by the direct liabilities of HCR III and the Sublease OpCos. Consequently, potential purchasers of the Subsidiary Stock would ascribe no value to the Long-Term Care Business.

Because the Long-Term Care Business would remain insolvent in a hypothetical chapter 7 case for the Debtor, the cash proceeds generated by a sale of the Subsidiary Stock would depend entirely upon the going-concern value of the 4H Business and the Outpatient Rehab and Other Businesses. More specifically, the sale proceeds would depend upon the value of those businesses in excess of the direct liabilities of the legal entities that operate such businesses and the liabilities of the intermediate holding companies (the "Excess 4H Value"). Such liabilities include:

- Approximately \$550 million plus accrued interest outstanding under the Credit Facility; and
- Approximately \$145 million in deferred compensation and other retirement obligations owed by an intermediate non-debtor subsidiary holding company, Manor Care, Inc.

Although the 4H Business and Outpatient Rehab Business are profitable and continue to grow, they historically have generated approximately 20% of the Company's revenues. The 4H Global Financial Projections (as defined below) for the 4H Business and Outpatient Rehab Business are attached hereto as **Exhibit E**. Based on the foregoing, the Debtor and its investment bank, Moelis & Company LLC, believe the Excess 4H Value would be less—very likely significantly less—than the QCP Chapter 7 Claim. Accordingly, a sale of the Subsidiary Stock in a chapter 7 liquidation, coupled with cash on hand, would yield insufficient cash proceeds to provide any distributions to the Holders of Section 510(b) Claims and Preferred and Common Interests of the Debtor in a chapter 7 liquidation of the Debtor.

D. Feasibility

The Bankruptcy Code requires that a debtor demonstrate that confirmation of a plan of reorganization is not likely to be followed by liquidation or the need for further financial reorganization. For purposes of determining whether the Plan meets this requirement, the Debtor has analyzed its ability, and that of its operating subsidiaries, to meet obligations under the Plan.

If the Closing of the Alternative Transaction occurs, then on the Effective Date, non-debtor HCR III, which is currently subject to the MLSA, will enter into the Alternative Master Lease. The monthly rent obligation under the Alternative Master Lease was negotiated between

⁷ Claims against the intermediate non-debtor subsidiary holding companies of the Debtor are structurally senior to Claims against the Debtor and, therefore, would be paid in full before Claims against the Debtor receive any distribution.

Welltower and ProMedica Parent, and is set at a level that the Company expects to be able to reasonably pay. The Debtor, with the assistance of its advisors, has prepared projections for the calendar years 2018 through 2021, including management's assumptions related thereto, which are attached hereto as **Exhibit D** (the "HCR Projections"). The HCR Projections assume that the Alternative Transaction closes, HCR III enters into the Alternative Master Lease, and the Plan will be implemented in accordance with its stated terms in all other respects. The Debtor is unaware of any circumstances as of the date of this Disclosure Statement that would require the re-forecasting of the HCR Financial Projections due to a material change in the Debtor's prospects. As reflected in the HCR Financial Projections, the Debtor anticipates that the Long-Term Care Business will be able to satisfy its obligations under the Alternative Master Lease and that the 4H Business and the Outpatient Rehab and Other Businesses will timely meet all of their collective obligations and generate significant profits for the foreseeable future.

If the Closing of the Alternative Transaction does not occur, after the Effective Date, non-debtor HCR III, which is currently subject to the MLSA, will be subject to the Original Master Lease. The monthly cash pay rent obligation under the Original Master Lease is limited, until 2025, to the operating cash generated by the SNFs and ALFs that is available after paying and reserving for all other liabilities (the "Free Cash Flow"). While the precise levels of Free Cash Flow for these businesses can vary from month to month and year to year, the Debtor is confident that Free Cash Flow for the foreseeable future will be substantially positive on a monthly basis and, accordingly, HCR III will be capable of timely satisfying all obligations as they come due through at least 2025.

With respect to the 4H Business and the Outpatient Rehab and Other Businesses, the Debtor, with the assistance of its advisors, has prepared projections for the calendar years 2018 through 2021, including management's assumptions related thereto, which are attached hereto as **Exhibit E** (the "4H Global Financial Projections"). The 4H Global Financial Projections assume that the Original Transaction will close, HCR III will enter into the Original Master Lease Amendment, and the Plan will be implemented in accordance with its stated terms in all other respects. The Debtor is unaware of any circumstances as of the date of this Disclosure Statement that would require the re-forecasting of the 4H Global Financial Projections due to a material change in the Debtor's prospects. As reflected in the 4H Global Financial Projections, the Debtor anticipates that the 4H Business and the Outpatient Rehab and Other Businesses will timely meet all of their collective obligations and generate significant profits for the foreseeable future.

Upon the Effective Date, the applicable Plan Sponsor shall make available, or cause to be made available, to the Reorganized Debtor such amounts as may be necessary to enable the Reorganized Debtor to fund or pay all of its obligations required to be funded or paid under the Plan (to the extent the Debtor does not have sufficient funds to pay such obligations when due).

⁸ The Original Master Lease requires total monthly rent of \$39.5 million. The difference between the monthly Free Cash Flow and the total rent will (a) accrue, (b) bear payment-in-kind interest at the annual rate of four percent (4%), and (c) be payable in 2025. Although the Debtor can make no assurances as to whether it will have the ability to satisfy the deferred rent (and accrued interest) in 2025, the Debtor does not believe that it will be in QCP's commercial interest, as indirect owner of the Long-Term Care Business, to enforce payment at that time.

Accordingly, the Debtor believes that Confirmation is not likely to be followed by liquidation or the need for further reorganization.

E. Acceptance by Impaired Class

The Bankruptcy Code requires, as a condition to confirmation, that, except as described in <u>ARTICLE VII.F</u> below, each class of claims or interests that is impaired under a plan of reorganization, accept the plan. A class that is not "impaired" under a plan is deemed to have accepted the plan and, therefore, solicitation of acceptances with respect to such class is not required. Section 1126(c) of the Bankruptcy Code defines acceptance of a plan by a class of impaired claims as acceptance by holders of at least two-thirds in dollar amount and more than one-half in number of allowed claims in that class, counting only those claims that actually voted to accept or to reject the plan. Section 1126(d) of the Bankruptcy Code defines acceptance of a plan by a class of impaired interests as acceptance by holders of at least two-thirds in amount of allowed interests in that class, counting only those interests that actually voted to accept or to reject the plan. Thus, the Voting Classes described herein will have voted to accept the Plan only if two-thirds in amount and, with respect to the Class of QCP Claims, a majority in number of Holders that actually vote on the Plan vote to accept.

F. Confirmation Without Acceptance by Impaired Class

The Bankruptcy Court may confirm a plan of reorganization over the rejection or deemed rejection of the plan of reorganization by a class of claims or interests if the plan of reorganization "does not discriminate unfairly" and is "fair and equitable" with respect to such class.

1. No Unfair Discrimination

This test applies to Classes of Claims or Interests that are of equal priority and are receiving different treatment under the Plan. The test does not require that the treatment be the same or equivalent, but that such treatment be "fair." The Debtor does not believe the Plan discriminates unfairly against any Impaired Class of Claims or Interests. The Debtor believes the Plan and the treatment of all Classes of Claims and Interests under the Plan satisfy the foregoing requirements for nonconsensual confirmation.

2. Fair and Equitable

This test applies to Classes of different priority and status (e.g., secured versus unsecured) and includes the general requirement that no Class of Claims or Interests receive more than 100% of the amount of the allowed Claims or Interests in such Class. As to the dissenting Class, the test sets different standards depending on the type of Claims or Interests of the Debtor in such Class. In order to demonstrate that a plan is fair and equitable, the plan proponent must demonstrate:

• <u>Class of Secured Claims</u>: That each holder of a secured claim: (a) retains its liens on property to the extent of the allowed amount of its secured claim, and receives deferred cash payments having a value, as of the effective date of the plan, of at least the allowed amount of such claim; (b) has the right to credit

bid the amount of its claim if its property is sold and retains its liens on the proceeds of the sale (or if sold, on the proceeds thereof); or (c) receives the "indubitable equivalent" of its allowed secured claim.

- <u>Class of Unsecured Claims</u>: That either (a) each holder of an impaired unsecured claim receives or retains under the plan property of a value equal to the amount of its allowed claim or (b) the holders of claims and interests that are junior to the claims of the non-accepting class will not receive or retain any property under the plan.
- <u>Class of Equity Interests</u>: That either (a) each holder of an impaired interest will receive or retain under the plan property of a value equal to the greatest of the fixed liquidation preference to which such holder is entitled, the fixed redemption price to which such holder is entitled, or the value of the interest, or (b) the holders of interests that are junior to the non-accepting class will not receive or retain any property under the plan.

The Debtor believes that the Plan satisfies the "fair and equitable" requirement notwithstanding that Class 7 is deemed to reject the Plan. If the Closing of the Alternative Transaction occurs, Class 7 will be Unimpaired, and the fair and equitable test does not apply. If the Closing of the Original Transaction occurs, Class 7 will not receive or retain any property, but there will also be no Class of equal priority receiving more favorable treatment and no Class that is junior to Class 7 will receive or retain any property on account of the Interests in such Class.

G. Alternatives to Confirmation and Consummation of the Plan

The Debtor has evaluated several alternatives to the transactions set forth in the Alternative Plan Sponsor Agreement and Original Plan Sponsor Agreement. After studying these alternatives, the Debtor believes that the Plan is the preferred approach to effectuating the Company's restructuring and maximizing the prospects of a successful turnaround and value for all stakeholders and, therefore, is in the best interests of all constituencies. If the Plan is not confirmed, the Debtor's ability to continue operating as a going concern could be severely jeopardized. The realistic alternatives likely will be either (1) the preparation and presentation of an alternative plan of reorganization, (2) a sale of some or all of the assets of the Company pursuant to section 363 of the Bankruptcy Code, or (3) a liquidation under chapter 7 of the Bankruptcy Code.

ARTICLE VIII. RISK FACTORS

The following provides a summary of various important considerations and other risk factors associated with the Plan; however, it is not exhaustive. There are risks, uncertainties, and other important factors that could cause the Debtor's actual performance or achievements to be materially different from those they may project, and the Debtor undertakes no obligation to update any such statement.

A. Bankruptcy-Specific Considerations

1. General

While the Debtor believes that the Chapter 11 Case will be of short duration and will not be materially disruptive to the Company's businesses, the Debtor cannot be certain that this will be the case. Although the Plan is designed to minimize the length of the Chapter 11 Case, it is impossible to predict with certainty the amount of time that the Debtor may spend in bankruptcy or to assure parties in interest that the Plan will be confirmed. Even if confirmed on a timely basis, bankruptcy proceedings to confirm the Plan could have a material adverse effect on the Debtor and the Company's businesses. A delay in the bankruptcy proceedings will also involve additional expense and may divert some of the attention of the Debtor's management away from business operations.

2. Objections to the Plan's Classification or Amounts of Claims and Interests

Section 1122 of the Bankruptcy Code provides that a plan may place a claim or an interest in a particular class only if such claim or interest is substantially similar to the other claims or interests in such class. The Debtor believes that the classification of the Claims and Interests under the Plan complies with the requirements set forth in the Bankruptcy Code because the Debtor created Classes of Claims and Interests, each encompassing Claims or Interests, as applicable, that are substantially similar to the other Claims and Interests in each such Class. Nevertheless, there can be no assurance that parties in interest will not object to the classification or amounts of Claims and Interests under the Plan, or that the Bankruptcy Court will approve such classification or amounts.

3. The Conditions Precedent to the Effective Date of the Plan May Not Occur

As more fully set forth in Article VIII of the Plan, the Effective Date is subject to a number of conditions precedent. If such conditions precedent are not satisfied or waived, the Effective Date will not occur.

4. Failure to Obtain or Delay in Obtaining Governmental Approvals

Among other conditions precedent to the Effective Date, the effectiveness of the Plan is subject to obtaining the governmental approvals set forth in Section 6.1(b) of the Debtor Disclosure Letter (as defined in both the Alternative Plan Sponsor Agreement and the Original Plan Sponsor Agreement), except to the extent such governmental approvals would not, in the aggregate, reasonably be expected to result in losses, costs, liabilities or expenses to certain parties to the applicable Plan Sponsor Agreement and their respective subsidiaries (as set forth in the applicable Plan Sponsor Agreement) in excess of \$5,000,000. The Debtor anticipates seeking each of the necessary governmental approvals from the relevant Governmental Units as soon as practicable.

If the Closing of the Alternative Transaction does not occur, the Debtor may need to restart the process for obtaining governmental approvals to consummate the Original Plan Sponsor

Agreement. If that scenario occurs, the Debtor and QCP may not be able obtain the Governmental Approvals prior to the maturity and expiration of the Credit Facility.

Although the effectiveness of the Plan requires that all such governmental approvals be obtained, no assurance can be given that such governmental approvals will be obtained or such conditions to effectiveness of the Plan be satisfied or waived, as such governmental approvals are beyond the control of the parties to the Plan Sponsor Agreements. Even if all such governmental approvals are obtained and such conditions are satisfied or waived, no assurance can be given as to the terms, conditions and timing of such governmental approvals. For example, certain Governmental Units providing such governmental approvals may require the Debtor or the Holders of Claims or the Holders of Interests to pay certain fees and/or accept certain conditions, requirements, limitations or restrictions that may be imposed by such Governmental Units. Such fees, conditions, requirements, limitations or restrictions may jeopardize or delay consummation of the Plan or may result in the modification of the Plan, reducing the proposed treatment of the Voting Classes and/or require re-solicitation.

5. Failure to Satisfy Voting Requirements

If votes are received in number and amount sufficient to enable the Bankruptcy Court to confirm the Plan, the Debtor intends to seek, as promptly as practicable thereafter, Confirmation. In the event that sufficient votes are not received, the Debtor may seek to confirm an alternative chapter 11 plan. There can be no assurance that the terms of any such alternative chapter 11 plan would be similar or as favorable to the Holders of Allowed Claims and Interests as those proposed in the Plan.

6. Termination of Alternative Plan Sponsor Agreement

The Alternative Plan Sponsor Agreement contains certain provisions that give certain parties the ability to terminate the Alternative Plan Sponsor Agreement upon the occurrence of certain events or if various conditions are not satisfied. The Alternative Plan Sponsor Agreement is subject to automatic termination upon the occurrence of certain events, termination by mutual consent, and, depending on the termination event, the Debtor, the Alternative Plan Sponsor, or QCP may terminate the Alternative Plan Sponsor Agreement. In the event the Alternative Plan Sponsor Agreement is terminated, the Debtor will pursue consummation of the Original Transaction, subject to the terms and conditions of the Original Plan Sponsor Agreement.

7. Termination of the Merger Agreement

The Closing of the Alternative Plan Sponsor Agreement is conditioned upon all conditions to the consummation of the Merger in the Merger Agreement (other than the occurrence of the Closing and such conditions to the consummation of the Merger as, by their nature, are to be satisfied at such consummation) having been satisfied or waived in accordance with the terms of the Merger Agreement. If the Merger Agreement is terminated, including due to a failure of the shareholders of QCP to vote to accept the Merger Agreement, the Alternative Plan Sponsor Agreement will be automatically terminated.

8. Termination of Original Plan Sponsor Agreement

The Original Plan Sponsor Agreement contains certain provisions that give certain parties the ability to terminate the Original Plan Sponsor Agreement upon the occurrence of certain events or if various conditions are not satisfied. The Original Plan Sponsor Agreement is subject to automatic termination upon the occurrence of certain events, termination by mutual consent, and, depending on the termination event, the Debtor or the Original Plan Sponsor may terminate the Original Plan Sponsor Agreement. Termination of the Original Plan Sponsor Agreement could result in a protracted Chapter 11 Case, which could significantly and detrimentally impact the Company's businesses.

9. Amendment of Plan Prior to Confirmation by the Debtor

The Debtor, subject to the terms and conditions of the Plan and the Alternative Plan Sponsor Agreement (or, upon the termination of the Alternative Plan Sponsor Agreement, the Original Plan Sponsor Agreement), reserve the right to modify the terms and conditions of the Plan or waive any conditions thereto if and to the extent necessary or desirable for Confirmation. The potential impact of any such amendment or waiver on Holders of Claims and Interests cannot presently be foreseen but may include a change in the economic impact of the Plan on some or all of the proposed Classes or a change in the relative rights of such Classes. In addition, if the Debtor seeks to modify or amend the Plan after receiving sufficient acceptances, but prior to Confirmation, re-solicitation may be required.

10. Non-Confirmation or Delay of Confirmation of the Plan

In the event that votes are received in number and amount sufficient to enable the Bankruptcy Court to confirm the Plan, the Debtor intends to seek, as promptly as practicable thereafter, Confirmation of the Plan. Section 1129 of the Bankruptcy Code sets forth the requirements for confirmation of a chapter 11 plan, and requires, among other things, findings by a bankruptcy court that: (a) such plan "does not unfairly discriminate" and is "fair and equitable" with respect to any non-accepting classes; (b) confirmation of such plan is not likely to be followed by a liquidation or a need for further financial reorganization unless such liquidation or reorganization is contemplated by the plan; and (c) the value of distributions to non-accepting holders of claims or equity interests within a particular class under such plan will not be less than the value of distributions such holders would receive if the debtor was liquidated under Chapter 7 of the Bankruptcy Code.

The Bankruptcy Court may determine that this Disclosure Statement and/or the solicitation procedures did not satisfy the requirements of the Bankruptcy Code or the Bankruptcy Rules or may decline to confirm the Plan if it finds that any of the statutory requirements for Confirmation have not been met, including the requirement that the terms of the Plan do not "unfairly discriminate" and are "fair and equitable" to non-accepting Classes, or the Plan contains other terms disapproved of by the Bankruptcy Court. If the Debtor fails to achieve Confirmation, the Chapter 11 Case would likely continue for a protracted period without indication of how or when the Chapter 11 Case may be completed. Some of the risks that the Company faces in the Chapter 11 Case would become more acute in such a scenario, including certain risks that are beyond its control, such as further deterioration or other changes in

economic conditions, changes in the industries in which the Company operates, potential revaluing of its assets due to the Chapter 11 Case, changes in patient demand for, and acceptance of, its services, regulatory difficulties and increasing expenses.

Even if the requisite acceptances of a proposed plan are received, the Bankruptcy Court is not obligated to confirm the Plan as proposed. A dissenting Holder of a Claim or Interest against the Debtor could challenge the solicitation procedures as not being in compliance with the Bankruptcy Code, which could mean that the results of the solicitation may be invalid. If the Bankruptcy Court determined that the solicitation procedures were appropriate and the results were valid, the Bankruptcy Court could still decline to confirm the Plan if the Bankruptcy Court found that any of the statutory requirements for confirmation had not been met.

If the Plan is not confirmed by the Bankruptcy Court, (a) the Debtor may not be able to successfully reorganize; (b) the distributions that Holders of Claims or Interests ultimately would receive, if any, with respect to their Claims or Interests is uncertain and (c) there is no assurance that the Debtor will be able to successfully develop, prosecute, confirm, and consummate an alternative plan that will be acceptable to the Bankruptcy Court and the Holders of Claims or Interests.

11. Non-Consensual Confirmation

The Debtor intends to seek Confirmation notwithstanding the deemed rejection of the Plan by Class 7. The Bankruptcy Court may confirm the Plan pursuant to the "cram-down" provisions of the Bankruptcy Code if the Plan satisfies section 1129(b) of the Bankruptcy Code. To confirm a plan over the objection of a dissenting class, the Bankruptcy Court also must find that at least one Impaired Class (which cannot be an "insider" class) has accepted the Plan.

12. Other Parties in Interest May Be Permitted to Propose Alternative Plans that Are Less Favorable to Certain Stakeholders

Other parties in interest could seek authority from the Bankruptcy Court to propose an alternative plan of reorganization. Under the Bankruptcy Code, a debtor in possession initially has the exclusive right to propose and solicit acceptances of a plan of reorganization for a period of 120 days from filing. Such exclusivity period can be reduced or terminated, however, upon order of the Bankruptcy Court. If such an order were to be entered, other parties in interest would then have the opportunity to propose one or more alternative plans of reorganization.

13. Conversion to Chapter 7

If no plan of reorganization can be confirmed, or if the Bankruptcy Court otherwise finds that it would be in the best interest of Holders of Claims and Interests, the Chapter 11 Case may be converted to case under chapter 7 of the Bankruptcy Code, pursuant to which a trustee would be appointed or elected to sell the Debtor's stock of the single direct subsidiary through which it owns interests in operating and non-operating subsidiaries within the Company and distribute the sale proceeds in accordance with the priorities established by the Bankruptcy Code. See the description of the Liquidation Analysis in <u>ARTICLE VII.C</u> of this Disclosure Statement for a discussion of the effects that a chapter 7 liquidation would have on the recoveries of Holders of Claims and Interests.

14. Inability to Use Cash Collateral

On April 5, 2018, the Bankruptcy Court entered an order authorizing the Debtor to use cash collateral to fund the Chapter 11 Case and to provide customary adequate protection to the prepetition secured parties. This access to cash collateral provides liquidity during the pendency of the Chapter 11 Case. The Debtor's authorization to use cash collateral may terminate after September 30, 2018. As a result, if the Chapter 11 Case takes longer than expected to conclude, the Debtor may not be permitted to use cash collateral. There is no assurance that the Debtor will be able to obtain an extension of the right to use cash collateral, in which case, the liquidity necessary for the prosecution of this Chapter 11 Case may be materially impaired.

15. Failure to Obtain Approval of Releases, Injunctions and Exculpation

Article IX of the Plan provides for certain releases, injunctions, and exculpations, including a release of liens and third-party releases that may otherwise be asserted against the Debtor, the Reorganized Debtor, the other Released Parties and their respective Related Persons, as applicable. The releases, injunctions, and exculpations (including, for the avoidance of doubt, the definitions of Released Parties and Exculpated Parties) provided in the Plan are subject to objection by parties in interest and may not be approved. If the releases are not approved, certain parties may not be considered Released Parties or Exculpated Parties, and certain Released Parties may withdraw their support for the Plan.

16. Plan Based on Assumptions

The Plan affects both the Debtor's capital structure and the ownership, structure, and operation of its businesses and reflects assumptions and analyses based on the Debtor's experience and perception of historical trends, current conditions, and expected future developments, as well as other factors that the Debtor considers appropriate under the circumstances. Whether actual future results and developments will be consistent with the Debtor's expectations and assumptions depends on a number of factors, including but not limited to the Debtor's (a) ability to obtain adequate liquidity and financing sources; (b) ability to maintain the public's confidence in the Debtor's and Company's viability as a continuing entity and enterprise and to attract and retain sufficient business; and (c) ability to retain key employees, as well as the overall strength and stability of general economic conditions of industries in which the Company operates. The failure of any of these factors could materially adversely affect the successful reorganization of the Debtor's businesses.

In addition, the Plan relies upon the Financial Projections, including with respect to revenues, EBITDA, debt service, and cash flow. Financial forecasts are necessarily speculative, and it is likely that one or more of the assumptions and estimates that are the basis of these financial forecasts will not be accurate. Consequently, there can be no assurance that the results or developments contemplated by any plan of reorganization implemented will occur or, even if they do occur, that they will have the anticipated effects on the Debtor and its subsidiaries or their businesses or operations. The failure of any such results or developments to materialize as anticipated could materially adversely affect the successful implementation of the Plan.

17. Recoveries for Holders of New Common Stock in Future Cases of Bankruptcy, Liquidation, Insolvency or Reorganization

Holders of the New Common Stock will be subordinated to all liabilities of the Reorganized Debtor and its subsidiaries (including the indebtedness under the Credit Agreement and the Intercompany Note, which shall be Reinstated on the Effective Date if the Closing of the Alternative Transaction does not occur, with any corresponding liens against the assets of the Debtor securing the Credit Facility Claims and Intercompany Note, respectively, surviving the Effective Date). Therefore, the assets of the Reorganized Debtor will not be available for distribution to any holder of the New Common Stock in any bankruptcy, liquidation, insolvency or reorganization of the Debtor unless and until all indebtedness, if any, of the Reorganized Debtor, including the Credit Agreement and the Intercompany Note, to the extent any loans thereunder are outstanding, has been paid. The remaining assets of the Debtor and each of its subsidiaries may not be sufficient to satisfy the outstanding claims of its equity holders, including the holders of the New Common Stock.

B. Risks Related to the Debtor's and the Reorganized Debtor's Business

1. Reorganized Debtor's Dependence on Its Subsidiaries' Performance

Upon the occurrence of the Effective Date, the Reorganized Debtor will continue to be a holding company and, as such, it will not have any independent operations, and its most significant asset will continue to be its ownership of the capital stock of its subsidiaries. As a consequence, the value of the New Common Stock will continue to be entirely dependent upon the future performance of its subsidiaries and will continue to depend upon the financial, business and other factors affecting its subsidiaries and the markets in which they do business, as well as general economic and financial conditions.

2. Requirements of, or Changes to, Governmental Reimbursement Programs Such as Medicare or Medicaid May Adversely Affect the Reorganized Debtor's Ability to Meet Its Financial and Other Contractual Obligations

The Debtor is, and the Reorganized Debtor will be, affected, directly or indirectly, by an extremely complex set of federal, state and local laws and regulations pertaining to governmental reimbursement programs. Such laws and regulations are subject to frequent and substantial changes that are sometimes applied retroactively. For the year ended December 31, 2017, the Long-Term Care Business generated a significant portion of its revenues from governmental payors, primarily Medicare and Medicaid. The Debtor is, and the Reorganized Debtor will be, affected by, directly or indirectly, among other things, statutory and regulatory changes; retroactive rate adjustments; recovery of program overpayments or set-offs; court decisions; administrative rulings; policy interpretations; payment or other delays by fiscal intermediaries or carriers; governmental funding restrictions (at a program level or with respect to specific facilities); and interruption or delays in payments due to any ongoing governmental investigations and audits at such properties.

If the Reorganized Debtor and its affiliates fail to comply with the extensive laws, regulations and other requirements applicable to their businesses and the operation of their properties, they could become ineligible to receive reimbursement from governmental reimbursement programs, face bans on admissions of new patients or residents, suffer civil or criminal penalties or be required to make significant changes to their operations. These laws and regulations are enforced by a variety of federal, state and local agencies and can also be enforced by private litigants through, among other things, federal and state false claims acts, which allow private litigants to bring *qui tam* or "whistleblower" actions. The Reorganized Debtor would be adversely affected by the resources required to respond to an investigation or other enforcement action. In such event, the results of operations and financial condition of the Reorganized Debtor could be materially adversely affected. The Debtor is unable to predict future federal, state and local regulations and legislation, including the Medicare and Medicaid statutes and regulations, or the intensity of enforcement efforts with respect to such regulations and legislation, and any changes in the regulatory framework could have a materially adverse effect on the Reorganized Debtor.

Sometimes governmental payors freeze or reduce payments to healthcare providers, or provide annual reimbursement rate increases that are smaller than expected, due to budgetary and other pressures. Healthcare reimbursement will likely continue to be of significant importance to federal and state authorities. The Debtor cannot make any assessment as to the ultimate timing or the effect that any future legislative reforms may have on its costs of doing business and on the amount of reimbursement by government and other third-party payors. The failure of the Reorganized Debtor or its affiliates to comply with these laws and regulations, and significant limits on the scope of services reimbursed and on reimbursement rates and fees, could materially adversely affect the Reorganized Debtor's ability to meet its financial obligations.

3. Legislation to Address Federal Government Operations and Administration Decisions Affecting the Centers for Medicare and Medicaid Services Could Have a Materially Adverse Effect on the Reorganized Debtor's Liquidity, Financial Condition or Results of Operations

Congressional consideration of legislation pertaining to the federal debt ceiling, the Patient Protection and Affordable Care Act, along with the Health Care and Education Reconciliation Act of 2010 (collectively, the "Affordable Care Act"), tax reform and entitlement programs, including reimbursement rates for physicians, could have a materially adverse effect on the Reorganized Debtor's and its affiliates' liquidity, financial condition or results of operations. In particular, reduced funding for entitlement programs such as Medicare and Medicaid may result in increased costs and fees for programs such as Medicare Advantage plans and additional reductions in reimbursements to providers. Additionally, amendments to the Affordable Care Act, implementation of the Affordable Care Act and decisions by the Centers for Medicare and Medicaid Services could impact the delivery of services and benefits under Medicare, Medicaid or Medicare Advantage plans and could affect the Reorganized Debtor and its affiliates and the manner in which they are reimbursed by such programs. Furthermore, the future of the Affordable Care Act, including its repeal, replacement or modification, is uncertain. Any such changes could have a materially adverse effect on the Reorganized Debtor's and its affiliates' liquidity, financial condition or results of operations.

Furthermore, the U.S. Supreme Court's decision upholding the constitutionality of the individual healthcare mandate while striking down the provisions linking federal funding of state Medicaid programs with a federally mandated expansion of those programs has contributed to the uncertainty regarding the impact that the law will have on healthcare delivery systems over the next decade.

4. Failure to Comply with Federal, State and Local Laws and Regulations, Including Licensure, Certification and Inspection Requirements, May Cause the Reorganized Debtor or Its Affiliates to Cease to Operate or Render Them Unable to Meet Their Financial and Other Contractual Obligations

The Debtor is, and the Reorganized Debtor will be, subject to or impacted by extensive, frequently changing federal, state and local laws and regulations. These laws and regulations include, among others: laws protecting consumers against deceptive practices; laws relating to the operation of our properties and how our tenants and operators conduct their business, such as fire, health and safety and privacy laws; federal and state laws affecting hospitals, clinics and other healthcare communities that participate in both Medicare and Medicaid that mandate allowable costs, pricing, reimbursement procedures and limitations, quality of services and care, food service and physical plants; resident rights laws (including abuse and neglect laws) and fraud laws; anti-kickback and physician referral laws; the Americans with Disabilities Act of 1990, as amended, and similar state and local laws; and safety and health standards set by the Occupational Safety and Health Administration or similar state and local agencies. Certain of the Company's properties may also require a license, registration and/or certificate of need to operate.

The Reorganized Debtor's or its affiliates' failure to comply with any of these laws, regulations or requirements could result in loss of accreditation, denial of reimbursement, imposition of fines, suspension or decertification from government healthcare programs, loss of license or closure of the facility and/or the incurrence of considerable costs arising from an investigation or regulatory action.

5. Ongoing Trends in the Healthcare Industry, Including a Shift Away From a Traditional Fee-for-Service Model and Increased Penetration of Government Reimbursement Programs with Lower Reimbursement Rates and Length of Stay, and Increased Competition in the Industry May Result in Lower Revenues and May Affect the Reorganized Debtor's and Its Affiliates' Ability to Meet Their Financial and Other Contractual Obligations

Post-acute/skilled nursing operators, including the Debtor and its affiliates, have been, and the Reorganized Debtor and its affiliates will continue to be, impacted by a challenging operating environment, which has put downward pressure on revenues and operating income. Ongoing trends in the post-acute/skilled nursing sector that are causing operators to adjust their business models include, but are not limited to, (a) a shift away from a traditional fee-for-service model towards new managed care models, which base reimbursement on patient outcome measures; (b) increased penetration of Medicare Advantage plans (*i.e.*, managed Medicare),

which has reduced reimbursement rates, average length of stay and average daily census, particularly for higher acuity patients; (c) increased competition from alternative healthcare services such as home health agencies, life care at home, community-based service programs, retirement communities and convalescent centers; and (d) increased regulatory scrutiny on government reimbursements. The Debtor is unable at this time to predict how these trends will affect the revenues and profitability of the Long-Term Care Business; however, if these trends continue, they may reduce the Reorganized Debtor's profitability.

The healthcare industry is also highly competitive. The occupancy levels at, and fee income from, the Leased Facilities are dependent on the Debtor's ability to compete with other operators on a number of different levels, including the quality of care provided, reputation, the physical appearance of a property, price, the range of services offered, family preference, alternatives for healthcare delivery, the supply of competing properties, physicians, staff, referral sources, location, and the size and demographics of the population in the surrounding area. In addition, the Debtor faces, and the Reorganized Debtor will face, an increasingly competitive labor market for skilled management personnel and nurses. An inability to attract and retain skilled management personnel and nurses and other trained personnel could negatively impact the ability of the Reorganized Debtor to meet their obligations. A shortage of nurses or other trained personnel or general inflationary pressures on wages may force the Reorganized Debtor to enhance pay and benefits packages to compete effectively for skilled personnel, or to use more expensive contract personnel, but the Reorganized Debtor may be unable to offset these added costs by increasing the rates charged to residents. Any increase in labor costs and other property operating expenses or any failure by the Reorganized Debtor to attract and retain qualified personnel would likely adversely affect the Reorganized Debtor's cash flow and have a materially adverse effect on its business, results of operations and financial condition.

The Company also competes with numerous other companies providing similar healthcare services or alternatives such as home health agencies, life care at home, community-based service programs, senior housing, retirement communities and convalescent centers. We cannot be certain that the Reorganized Debtor will be able to achieve occupancy and rate levels, and to manage its expenses, in a way that will enable it to meet all of its obligations. Further, many competing companies may have resources and attributes that are superior to those of the Debtor. The Reorganized Debtor may encounter increased competition that could limit its ability to maintain or attract residents or expand their businesses or to manage their expenses, either of which could materially adversely affect its ability to meet its financial and other contractual obligations.

ARTICLE IX. IMPORTANT SECURITIES LAW DISCLOSURE

A. Plan Securities

The Plan provides for the Reorganized Debtor to distribute New Common Stock (the "<u>Plan Securities</u>") to either (i) ProMedica Sub (or its designee(s)), if the Closing of the Alternative Transaction occurs) or (ii) the Holders of QCP Claims (or their designee(s)) (if the Closing of the Alternative Transaction does not occur). The Debtor believes that the Plan

Securities constitute "securities," as defined in Section 2(a)(1) of the Securities Act, section 101 of the Bankruptcy Code, and all applicable state securities laws ("Blue Sky Laws").

B. Issuance and Resale of Plan Securities under the Plan

1. Exemptions from Registration Requirements of the Securities Act and Blue Sky Laws

The Debtor is relying on exemptions from the registration requirements of the Securities Act, including, without limitation, section 4 thereof and section 1145(a) of the Bankruptcy Code, to exempt the offer and sale of the Plan Securities.

Section 4 of the Securities Act exempts transactions not involving a public offering, and section 506 of Regulation D of the Securities Act ("Reg D") provides a safe harbor under section 4(a)(2) for transactions that meet certain requirements, including that the investors participating therein qualify as "accredited investors" as defined in section 501 of Reg. D (17 C.F.R. § 230.501). The Debtor believes that entities receiving the Plan Securities (ProMedica Sub (or its designee(s)) and the Holders of QCP Claims) are "accredited investors." Each voting Holder of QCP Claims will be required to certify in the Ballot that it is an "accredited investor." ProMedica Sub has also represented that it qualifies as an "accredited investor."

In addition, in the event the Original Transaction closes and the Plan Securities are issued to Holders of QCP Claims, the Debtor may rely on the exemption provided under Section 1145 of the Bankruptcy Code. Section 1145 of the Bankruptcy Code provides that the registration requirements of section 5 of the Securities Act (and any applicable state Blue Sky Laws) shall not apply to the offer or sale of stock, options, warrants, or other securities by a debtor if: (x) the offer or sale occurs under a plan of reorganization; (y) the recipients of the securities hold a claim against, an interest in, or claim for administrative expense against, the debtor; and (z) the securities are issued in exchange for a claim against or interest in a debtor or are issued principally in such exchange and partly for cash and property.

In reliance upon these exemptions, the offer and sale of the Plan Securities will not be registered under the Securities Act or any applicable state Blue Sky Laws.

2. Resales of Plan Securities; Restrictions on Transfer

With respect to the issuance of Plan Securities covered by section 1145 of the Bankruptcy Code (*i.e.*, the issuance of Plan Securities to Holders of QCP Claims in the event the Original Transaction closes), such Plan Securities may be resold without registration under the Securities Act or other federal securities laws, unless the holder is an "underwriter" (as discussed below) with respect to such securities, as that term is defined in section 1145 of the Bankruptcy Code. Section 1145(b)(1) of the Bankruptcy Code includes in its definition of an "underwriter," "affiliates" of the issuer of the securities, meaning all persons who, directly or indirectly, through one or more intermediaries, control, are controlled by, or are under common control with, an issuer of securities. The Debtor believes that the Holders of QCP Claims, as the recipients of the Plan Securities in a transaction exempt from registration under section 1145(a) of the Bankruptcy Code, would be affiliates of the Reorganized Debtor if the Original Transaction closes and

therefore "underwriters" under section 1145 of the Bankruptcy Code whose Plan Securities would be subject to restrictions on transfer.

If the Alternative Transaction closes, the Debtor believes that ProMedica Sub, as the recipient of the Plan Securities in a transaction exempt from registration under section 4 of the Securities Act, would be affiliates of the Reorganized Debtor whose Plan Securities would also be subject to restrictions on transfer.

Under certain circumstances, holders of Plan Securities who are affiliates of the Reorganized Debtor may be entitled to resell their Plan Securities pursuant to the limited safe harbor resale provisions of Rule 144 of the Securities Act. Generally, Rule 144 of the Securities Act would permit the public sale of securities received by such person certain conditions are met, including, if such seller is an affiliate of the issuer, public information, volume limitations and manner of sale requirements. Given the nature of the Rule 144 requirements, it may not be possible for affiliates of the Reorganized Debtor to sell Plan Securities under Rule 144 of the Securities Act unless and until a trading market in the Plan Securities develops. The Debtor recommends that potential recipients of Plan Securities consult their own counsel concerning their ability to trade such securities without registration under the federal securities laws.

In addition, the Plan Securities may be able to be resold without registration under applicable state Blue Sky Laws pursuant to various exemptions provided by the respective Blue Sky Laws of those states; however, the availability of such exemptions cannot be known unless individual state Blue Sky Laws are examined. Therefore, recipients of the Plan Securities are advised to consult with their own legal advisors as to their ability to trade such securities without registration under state Blue Sky Laws.

ARTICLE X. CERTAIN UNITED STATES FEDERAL INCOME TAX CONSEQUENCES OF THE PLAN

The following discussion is a summary of certain U.S. federal income tax consequences of the Plan, is for general information purposes only, and should not be relied upon for purposes of determining the specific tax consequences of the Plan with respect to a particular Holder of a Claim or Interest. This discussion does not purport to be a complete analysis or listing of all potential tax considerations.

This discussion is based on existing provisions of the Internal Revenue Code of 1986, as amended (the "IRC"), existing and proposed Treasury Regulations promulgated thereunder, and current administrative rulings and court decisions. Legislative, judicial, or administrative changes or interpretations enacted or promulgated after the date hereof could alter or modify the analyses set forth below with respect to the U.S. federal income tax consequences of the Plan. Any such changes or interpretations may be retroactive and could significantly affect the U.S. federal income tax consequences of the Plan.

No ruling has been requested or obtained from the Internal Revenue Service (the "<u>IRS</u>") with respect to any tax aspects of the Plan and no opinion of counsel has been sought or obtained

with respect thereto. No representations or assurances are being made to the Holders of Claims or Interests with respect to the U.S. federal income tax consequences described herein.

This summary does not address the U.S. federal income tax consequences to Holders of Claims or Interests that are not "United States persons" (within the meaning of section 7701(a)(30) of the IRC). This discussion does not purport to address all aspects of U.S. federal income taxation that may be relevant to the Debtor or to certain Holders in light of their individual circumstances. This discussion does not address tax issues with respect to Holders subject to special treatment under the U.S. federal income tax laws (including, for example, banks, governmental authorities or agencies, pass-through entities, subchapter S corporations, dealers and traders in securities, insurance companies, financial institutions, tax-exempt organizations, small business investment companies, foreign taxpayers, persons who are related to the Debtor within the meaning of the IRC, persons using a mark-to-market method of accounting, Holders of Claims who are themselves in bankruptcy, and regulated investment companies and those holding, or who will hold, Claims, debt of the Reorganized Debtor, or shares of stock in the Company, as part of a hedge, straddle, conversion, or other integrated transaction). No aspect of state, local, estate, gift, or non-U.S. taxation is addressed. Furthermore, this summary assumes that a Holder of an Interest holds its Interest as a "capital asset" (within the meaning of section 1221 of the IRC).

ACCORDINGLY, THE FOLLOWING SUMMARY OF CERTAIN U.S. FEDERAL INCOME TAX CONSEQUENCES IS FOR INFORMATIONAL PURPOSES ONLY AND IS NOT A SUBSTITUTE FOR CAREFUL TAX PLANNING AND ADVICE BASED UPON THE INDIVIDUAL CIRCUMSTANCES PERTAINING TO A HOLDER OF A CLAIM OR INTEREST. ALL HOLDERS OF CLAIMS OR INTERESTS ARE URGED TO CONSULT THEIR OWN TAX ADVISORS FOR THE FEDERAL, STATE, LOCAL AND NON-U.S. TAX CONSEQUENCES OF THE PLAN.

A. Certain Federal Income Tax Consequences to the Debtor

Under the Alternative Transaction, the Alternative Plan Sponsor and the Debtor intend to effectuate certain restructuring transactions on the Effective Date that will result in the Debtor becoming an organization that is exempt from federal income tax under Section 501 of the IRC. Accordingly, the remainder of the discussion under this heading "Certain Federal Income Tax Consequences to the Debtor" will be inapplicable.

If the Closing of the Alternative Transaction does not occur, however, then the following summary of certain federal income tax consequences to the Debtor applies.

As of December 31, 2017, for federal income tax purposes, the Company had approximately \$270 million of consolidated net operating losses ("NOLs"), \$27 million in tax credit carryforwards and approximately \$736 million in amortizable tax basis in goodwill and other intangible assets. In addition, the Company expects to generate additional consolidated NOLs and tax credit carryforwards during the 2018 taxable year. The implementation of the Plan is not expected to materially reduce or otherwise significantly affect the amount of these tax attributes, except as noted below under "382(l)(5) Bankruptcy Exception."

The amount of the Company's consolidated NOLs, tax credit carryforwards and amortizable tax basis remains subject to adjustment by the IRS. As a general rule, NOLs generated by a debtor in taxable years beginning on or before December 31, 2017 can be carried back and deducted from its taxable income generated within the two preceding taxable years and the remainder can be carried forward and deducted from the debtor's taxable income over the 20 succeeding taxable years. NOLs generated in 2018 and later taxable years cannot be carried back and may only be used to offset up to 80% of a debtor's taxable income generated during the taxable year to which it is carried. Any remainder can be carried forward indefinitely and used to offset taxable income in a future taxable year (subject to the 80% limitation described above). Amortizable tax basis in intangible assets can generally be deducted ratably from the debtor's taxable income over the 15-year period commencing with the year the intangible asset is acquired.

Section 382 of the IRC contains certain rules limiting the ability of corporations to utilize NOLs when there has been an "ownership change" (the "Annual Section 382 Limitation"). An "ownership change" generally is defined as a more than 50 percentage point change in ownership of the value of the stock of a "loss corporation" (a corporation with NOLs) that takes place during the three-year period ending on the date on which such change in ownership is tested.

As a general rule, the Annual Section 382 Limitation equals the product of the value of the stock of the loss corporation (with certain adjustments) immediately before the ownership change and the applicable "long-term tax-exempt rate," a rate published monthly by the Treasury Department (2.30% for ownership changes that occur during May 2018). Any unused portion of the Annual Section 382 Limitation generally is available for use in subsequent years. The Annual Section 382 Limitation is increased in the case of a corporation that has net unrealized built-in gains ("NUBIG"), i.e., gains economically accrued but unrecognized at the time of the ownership change, in excess of a threshold amount. Such a corporation can use NOLs in excess of its Annual Section 382 Limitation to the extent that it realizes those NUBIGs for U.S. federal income tax purposes in the five years following the ownership change. For purposes of determining the Annual Section 382 Limitation for a consolidated tax group, the determination as to whether the group has a NUBIG generally is made on a consolidated basis (i.e., by netting the built-in losses and built-in gains of all members of the group).

In addition to limiting a corporation's ability to use NOLs, the Annual Section 382 Limitation may also apply to certain losses or deductions which are "built-in" as of the date of the ownership change and that are subsequently recognized. If a loss corporation has a net-unrealized built-in loss ("NUBIL") at the time of the ownership change, then any built-in losses or deductions (which for this purpose includes a portion of the depreciation or amortization of depreciable or amortizable assets that have a built-in loss) that are recognized during the following five years (up to the amount of the original built-in loss) generally will be subject to the Annual Section 382 Limitation. Special rules apply for purposes of determining the amount of built-in losses of a consolidated tax group that are subject to an Annual Section 382 Limitation. Absent an applicable exception (discussed below), a Company NUBIL would also be subject to the Annual Section 382 Limitation.

Section 382(l)(5) of the IRC provides an exception to the application of the Annual Section 382 Limitation when a corporation is under the jurisdiction of a court in a Title 11 case

(the "382(1)(5) Bankruptcy Exception"). The 382(1)(5) Bankruptcy Exception provides that where an ownership change occurs pursuant to a bankruptcy reorganization or similar proceeding, the Annual Section 382 Limitation will not apply if the pre-change shareholders and/or "qualified creditors" (as defined by applicable Treasury Regulations) own at least 50 percent of the stock of the reorganized corporation immediately after the ownership change. However, under the 382(1)(5) Bankruptcy Exception, a corporation's pre-change losses and excess credits that may be carried over to a post-change year must be reduced to the extent attributable to any interest paid or accrued on certain debt converted to stock in the reorganization. In addition, if the 382(1)(5) Bankruptcy Exception applies, a second ownership change of the corporation within a two-year period will cause the corporation to forfeit all of its unused NOLs that were incurred prior to the date of the second ownership change. If a corporation qualifies for the 382(1)(5) Bankruptcy Exception, the use of its NOLs will be governed by that exception unless the corporation affirmatively elects for the provisions not to apply.

If a corporation in bankruptcy is not eligible for the 382(l)(5) Bankruptcy Exception or elects out of that provision, a special rule under Section 382(l)(6) of the IRC will apply in calculating the Annual Section 382 Limitation. Under this special rule, the Annual Section 382 Limitation will be calculated by reference to the lesser of the value of the corporation's stock (with certain adjustments) immediately after the ownership change (as opposed to immediately before the ownership change, as discussed above) or the value of the debtor's assets (determined without regard to liabilities) immediately before the ownership change.

Although it is expected that the Company will undergo an ownership change as a result of the implementation of the Plan, the Company intends to take the position that it qualifies for the 382(l)(5) Bankruptcy Exception described above. Accordingly, the Company does not expect to be subject to the Annual Section 382 Limitation as a result of the implementation of the Plan.

B. Certain Federal Income Tax Consequences to Holders of a QCP Claim

If the Closing of the Alternative Transaction occurs, Cash will be distributed to Holders of QCP Claims. A Holder who receives Cash in exchange for a QCP Claim will recognize income or gain in an amount up to the amount of Cash received.

If the Closing of the Alternative Transaction does not occur, New Common Stock of the Reorganized Debtor will be distributed to Holders of QCP Claims. A Holder who receives New Common Stock of the Reorganized Debtor in exchange for a QCP Claim may realize income equal to the value of the New Common Stock received. Such Holder's initial basis in the New Common Stock received should likewise equal the value of such New Common Stock at the time of receipt. Such basis may be increased in respect of a subsequent termination of the Guaranty. A Holder should consult its tax advisor concerning the tax consequences of this and other related aspects of the Plan.

C. Certain Federal Income Tax Consequences to Holders of Interests

If the Closing of the Alternative Transaction occurs, Holders of Preferred Interests will receive their pro rata share of the Preferred Equity Distribution and Holders of Common Interests will receive their pro rata share of the Common Equity Distribution, in each case in exchange for their Interests. In such case, each such Holder will recognize gain or loss in an amount equal to the difference between (i) the adjusted tax basis in its Interests surrendered in the exchange, determined immediately prior to the Effective Date, and (ii) the amount of Cash received.

If the Closing of the Alternative Transaction does not occur, Holder of Interests will have their Claims extinguished and will not receive or retain any property under the Plan on account of their Claims. In such case, each such Holder will recognize loss in an amount equal to the Holder's adjusted tax basis in its Interests, determined immediately prior to the Effective Date.

Any gain or loss recognized by a Holder of an Interest will be long-term capital gain or loss if the Holder held its Interest for more than one year, and otherwise will be short-term capital gain or loss. Long-term capital gains are subject to preferential tax rates for individual Holders. Capital losses generally may be used by a corporate Holder only to offset capital gains, and by an individual Holder only to the extent of capital gains plus \$3,000 of other income.

D. Reservation of Rights

The foregoing discussion is subject to change (possibly substantially) based on, among other things, subsequent changes to the Plan and events that may subsequently occur that may impact the timeline for the transactions contemplated by the Plan. The Debtor and their advisors reserve the right to further modify, revise or supplement this ARTICLE X and the other tax related sections of the Plan and Disclosure Statement in accordance with the terms of the Plan and the Bankruptcy Code.

ARTICLE XI. CONCLUSION AND RECOMMENDATION

In the opinion of the Debtor, the Plan is preferable to all other available alternatives and provides for a larger distribution to the Debtor's creditors than would otherwise result from any other scenario. Any alternative to confirmation of the Plan, moreover, could result in extensive delays and increased administrative expenses. Accordingly, the Debtor urges Holders of Claims and Interests in the Voting Classes to evidence their acceptance of the Plan by returning their Ballots so that they are received by the Solicitation Agent no later than 4:00 p.m., prevailing Eastern Time, on June 15, 2018.

Dated: May 18, 2018 Wilmington, Delaware Respectfully submitted,

HCR ManorCare, Inc.

/s/ John R. Castellano

John R. Castellano Chief Restructuring Officer

SIDLEY AUSTIN LLP

Larry J. Nyhan
Dennis M. Twomey
William A. Evanoff
Allison Ross Stromberg
Matthew E. Linder
One South Dearborn Street
Chicago, Illinois 60603
Telephone: (312) 853-7000
Facsimile: (312) 853-7036

-and-

YOUNG CONAWAY STARGATT & TAYLOR, LLP

Robert S. Brady (No. 2847) Edmon L. Morton (No. 3856) Justin H. Rucki (No. 5304) Tara C. Pakrouh (No. 6192) Rodney Square 1000 North King Street Wilmington, Delaware 19801 Telephone: (302) 571-6600 Facsimile: (302) 571-1253

Attorneys for the Debtor and Debtor in Possession

EXHIBIT A

PLAN OF REORGANIZATION

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

HCR MANORCARE, INC.,1

Debtor.

Case No. 18-10467 (KG)

Chapter 11

FIRST AMENDED CHAPTER 11 PLAN OF REORGANIZATION FOR HCR MANORCARE, INC.

SIDLEY AUSTIN LLP YOUNG CONAWAY STARGATT & TAYLOR, LLP

Larry J. Nyhan

Robert S. Brady (No. 2847)

Dennis M. Twomey

Edmon L. Morton (No. 3856)

William A. Evanoff

Allison Ross Stromberg

Matthew F. Linder

Podney Square

Matthew E. Linder Rodney Square

One South Dearborn Street 1000 North King Street

Chicago, Illinois 60603 Wilmington, Delaware 19801 Facsimile: (312) 853-7036 Facsimile: (302) 571-1253

Attorneys for the Debtor and Debtor in Possession

Dated: May 18, 2018

01:23221422.1

¹ The last four digits of the Debtor's federal tax identification number are 9231. The mailing address for the Debtor is 333 N. Summit St., Toledo, OH 43604.

TABLE OF CONTENTS

			PAGE
ARTI	CLE I:	DEFINED TERMS AND RULES OF INTERPRETATION	1
ARTIC	CLE II	: TREATMENT OF ADMINISTRATIVE EXPENSE CLAIMS AND	
	PRIOF	RITY TAX CLAIMS	12
	2.1.	Administrative Expense Claims	12
	2.2.	Priority Tax Claims	
	2.3.	Post-Effective Date Fees and Expenses	
ARTI	CLE II	I: CLASSIFICATION AND TREATMENT OF CLASSIFIED CLAIMS	
		INTERESTS	13
	3.1.	Summary of Classification and Treatment of Classified Claims and	
	3.1.	Interests	13
	3.2.	Classification and Treatment of Claims Against and Interests in the Debtor.	
	3.3.	Intercompany Claims	
	3.4.	Unimpaired Claims	
	3.4.	Ommpuned Claims	
ARTI	CLE IV	: ACCEPTANCE OR REJECTION OF THE PLAN	20
	4.1.	Impaired Classes of Claims or Interests Entitled to Vote on this Plan	20
	4.2.	Acceptance by an Impaired Class of Claims or Interests	
	4.3.	Presumed Acceptance by Unimpaired Classes	
	4.4.	Presumed Rejection by Certain Impaired Classes	
	4.5.	Confirmability of this Plan and Reservation of Rights	
A D/TI			
AKII	CLE V	: MEANS FOR IMPLEMENTATION OF THE PLAN	21
	5.1.	Alternative Transaction	21
	5.2.	Original Transaction	21
	5.3.	Operations Between the Confirmation Date and Effective Date	21
	5.4.	Sources of Cash Consideration for Plan Distributions	22
	5.5.	New Common Stock.	22
	5.6.	Rollover Restricted Shares	22
	5.7.	Section 1145 Exemption	22
	5.8.	Master Lease Amendment	23
	5.9.	Release of Guaranty	23
	5.10.	Corporate Governance, Directors, Officers and Corporate Action	23
	5.11.	Continued Corporate Existence and Vesting of Assets in the Reorganized	
		Debtor	
	5.12.	Cancellation of Liens	24
	5.13.	Preservation of Rights of Action and Settlement of Ordinary Litigation	
		Claims	
	5.14.	Registration of New Common Stock	24

5	.15. Additional Transactions Authorized Under this Plan	
5	.16. Comprehensive Settlement of Claims and Controversies	25
5	.17. Tax Election	25
A DTICI	LE VI: TREATMENT OF EXECUTORY CONTRACTS AND INSURANCE	
	OLICIES	25
•		20
	.1. Assumption or Rejection of Executory Contracts	
	.2. No Cure Obligations	
	.3. Insurance Policies and Agreements	26
6	.4. Postpetition Contracts and Leases	26
6	.5. Modifications, Amendments, Supplements, Restatements or Other	
	Agreements	26
ARTICI	LE VII: PROVISIONS GOVERNING DISTRIBUTIONS	26
7	.1. Distributions on Account of Claims Allowed as of the Effective Date	26
	.2. Distributions on Account of Claims that Become Allowed after the	20
,	Effective Date	27
7	.3. Distributions by Reorganized Debtor	
	.4. Means of Cash Payment	
	.5. Withholding and Reporting Requirements	
	.6. Compliance Matters	
	.7. Setoff and Recoupment	
	.8. Reinstated Claims	
	.9. Undeliverable or Non-Negotiated Distributions	
	.10. Claims Paid by Third Parties	
,		20
ARTICI	LE VIII: CONFIRMATION AND CONSUMMATION OF THE PLAN	29
8	.1. Conditions to Effective Date of Plan Implementing the Alternative	
	Transaction	29
8	.2. Conditions to Effective Date of Plan Implementing the Original	
	Transaction	29
8	.3. Waiver of Conditions	30
8	.4. Vacatur of Confirmation Order	30
8	.5. Notice of Effective Date	30
ARTICI	LE IX: EFFECT OF PLAN CONFIRMATION	30
0	1 Dinding Effect	20
	.1. Binding Effect	
	.2. Discharge	
	J	
	J	
	.5. Exculpation	
	.6. Injunctions Related to Exculpation and Releases	
	.7. Survival of Indemnification and Exculpation Obligations	
9	.8. Term of Bankruptcy Injunction or Stays	ээ

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 73 of 441

	9.9.	Liability to Governmental Units	35
	9.10.	·	
ART	ICLE X	: RETENTION OF JURISDICTION	35
	10.1.	Retention of Jurisdiction	35
ART	ICLE X	I: MISCELLANEOUS PROVISIONS	37
	11.1.	Post-Effective Date Retention of Professionals	37
	11.2.	Effectuating Documents and Further Transactions	38
	11.3.	Exemption from Transfer Taxes	38
	11.4.	Payment of Statutory Fees	38
	11.5.	Amendment or Modification of this Plan	38
	11.6.	Severability of Plan Provisions	38
	11.7.	Successors and Assigns	
	11.8.	Non-Consummation	
	11.9.	Notice	39
	11.10.	Governing Law	
		Tax Reporting and Compliance	
		Exhibits	
		Filing of Additional Documents	
		Plan Documents	
		Reservation of Rights	

TABLE OF EXHIBITS

Exhibit A	Alternative Master Lease Term Sheet
Exhibit B	Alternative Plan Sponsor Agreement
Exhibit C	Original Master Lease Amendment
Exhibit D	Original Plan Sponsor Agreement
Exhibit E	Amendment to Original Plan Sponsor Agreement
Exhibit F	Directors and Officers of Reorganized Debtor – Alternative Transaction
Exhibit G	Directors and Officers of Reorganized Debtor – Original Transaction
Exhibit H	Certificate of Incorporation of Reorganized Debtor – Original Transaction
Exhibit I	By-Laws of Reorganized Debtor – Original Transaction

INTRODUCTION

HCR ManorCare, Inc. (the "<u>Debtor</u>") hereby proposes the following first amended plan of reorganization for the Debtor's reorganization case under Chapter 11 of the Bankruptcy Code for the resolution of the outstanding Claims against and Interests in the Debtor. Capitalized terms used but not defined in this paragraph have the meanings assigned to them in <u>Article II</u>. The classification and treatment of Claims against and Interests in the Debtor are set forth in <u>Article III</u> and <u>Article III</u>. The Debtor is the proponent of this Plan within the meaning of section 1129 of the Bankruptcy Code. Reference is made to the Disclosure Statement, distributed contemporaneously herewith, for a discussion of the Debtor's history, business, properties and operations, projections for those operations, risk factors, a summary and analysis of this Plan, and related matters.

ARTICLE I: DEFINED TERMS AND RULES OF INTERPRETATION

- A. <u>Defined Terms</u>. As used in this Plan, capitalized terms shall have the meanings set forth in this <u>Article I</u>. Any term that is not otherwise defined herein, but that is used in the Bankruptcy Code or the Bankruptcy Rules, shall have the meaning given to that term in the Bankruptcy Code or the Bankruptcy Rules, as applicable.
- 1.1 <u>Accrued Unpaid Dividends</u> means any accrued unpaid dividends due and payable to Holders of Preferred Interests as of the Effective Date.
- Administrative Expense Claim means a Claim for costs and expenses of 1.2 administration of the Chapter 11 Case arising after the Petition Date and prior to the Effective Date under sections 328, 330, 363, 364(c)(1), 365, 503(b), 507(a)(2), or 507(b) of the Bankruptcy Code, including, without limitation, (a) any actual and necessary costs and expenses of preserving the Estate and operating the businesses of the Debtor after the Petition Date and Claims of governmental units for taxes (including tax audit Claims) related to tax years commencing after the Petition Date, but excluding Claims related to tax periods, or portions thereof, ending on or before the Petition Date; (b) any Professional Fee Claim, to the extent Allowed by Final Order under sections 328, 330, 331 or 503 of the Bankruptcy Code; (c) with the exception of section 507(b) Claims, any indebtedness or obligations incurred or assumed by the Debtor during the Chapter 11 Case; (d) any cash payment required to be made under this Plan and payments to cure a default under an Executory Contract that has been or will be assumed by the Debtor; or (e) any Quarterly Fees; provided, however, in connection with the Alternative Transaction, no QCP Claim (including the Agreed Deferred Rent Obligation if the Closing of the Alternative Transaction occurs) shall be treated as an Administrative Expense Claim and all QCP Claims shall be treated only as provided in Section 3.2.4(c)(i).
- 1.3 <u>Affiliate</u> has the meaning assigned to such term in section 101(2) of the Bankruptcy Code.
- 1.4 <u>Agreed Deferred Rent Obligation</u> means, for purposes of the Alternative Transaction only, an obligation of the Debtor for deferred rent in the amount of \$440,190,183.
- 1.5 <u>Allowed</u> means, with respect to a Claim or Interest, such Claim or Equity Interest or any portion thereof that the Debtor has assented to the validity of, and to which the Plan Sponsor

has not objected, or that has been (a) allowed by an order of the Bankruptcy Court, (b) allowed pursuant to the terms of the Plan, (c) following the Effective Date, allowed by agreement between the Holder of such Claim, on one hand, and the Reorganized Debtor, as applicable, on the other hand, or (d) allowed by an order of a court in which such Claim could have been determined, resolved or adjudicated if the Chapter 11 Case had not been commenced; provided, however, that an Administrative Expense Claim, other than Professional Fee Claim, incurred by the Debtor in the ordinary course of its business during the Chapter 11 Case, or assumed by the Debtor during the Chapter 11 Case, may be Allowed if the Debtor assents to the validity of such Claim; provided, further that, notwithstanding the foregoing, the Reorganized Debtor shall retain all Causes of Action and defenses with respect to Allowed Claims that are Reinstated or otherwise Unimpaired pursuant to the Plan (including, for the avoidance of doubt, Administrative Expense Claims not paid prior to the Effective Date).

- 1.6 <u>Allowed Claim or Interest</u> means a Claim or Interest in a particular Class or of a particular type that is also an Allowed Claim or Interest. For example, an Allowed Administrative Expense Claim is an Administrative Expense Claim that is also an Allowed Claim.
- 1.7 <u>Alternative Master Lease</u> has the meaning given to the term "New Master Lease" in the Alternative Plan Sponsor Agreement. The form of Alternative Master Lease shall be filed with the Plan Supplement. A term sheet describing the terms of the Alternative Master Lease is attached hereto as **Exhibit A**.
- 1.8 <u>Alternative Plan Sponsor Agreement</u> means the Plan Sponsor Agreement among the Debtor, QCP, ProMedica Health System, Inc., ProMedica, and Meerkat I LLC, dated as of April 25, 2018, as amended, modified or supplemented from time to time in accordance with its terms, and including all schedules and exhibits thereto. A copy of the Alternative Plan Sponsor Agreement is attached hereto as **Exhibit B**.
- 1.9 <u>Alternative Restructuring Support Agreement</u> means the Restructuring Support Agreement among the Debtor, Carlyle, MC Operations Investments, LLC, and ProMedica Health System, Inc., dated as of April 25, 2018, as amended, modified or supplemented from time to time in accordance with its terms.
 - 1.10 Alternative Transaction shall have the meaning ascribed to it in Article V hereof.
- 1.11 <u>Amendment to Original Plan Sponsor Agreement</u> means the Amendment to Plan Sponsor Agreement among the Debtor, QCP, HCP Mezzanine Lender, LP, a Delaware limited partnership and a wholly-owned subsidiary of QCP, and the Lessors identified therein, dated as of April 25, 2018. A copy of the Amendment to Original Plan Sponsor Agreement is attached hereto as **Exhibit E**.
- 1.12 <u>Avoidance Actions</u> means any and all actual or potential claims or causes of action to avoid a transfer of property or an obligation incurred by the Debtor arising under sections 502(d), 542, 544, 545, 547, 548, 549, 550, 551, 552 or 553(b) of the Bankruptcy Code, or under similar or related state or federal statutes or common law, including fraudulent transfer and conveyance laws, in each case whether or not litigation to prosecute such claim(s) or cause(s) of action was commenced prior to the Effective Date.

- 1.13 <u>Ballot</u> means the ballot form for accepting or rejecting this Plan, distributed with the Disclosure Statement to the Holders of Claims that are Impaired under this Plan and entitled to vote to accept or reject this Plan pursuant to <u>Article III</u> and <u>Article IV</u>.
- 1.14 <u>Bankruptcy Code</u> means Title 11 of the United States Code, 11 U.S.C. §§ 101 through 1532, as in effect on the Petition Date, together with any amendments and modifications thereto that may subsequently be made applicable to the Chapter 11 Case.
- 1.15 <u>Bankruptcy Court</u> means the United States Bankruptcy Court for the District of Delaware or any other court with jurisdiction over the Chapter 11 Case.
- 1.16 <u>Bankruptcy Rules</u> means, collectively: (a) the Federal Rules of Bankruptcy Procedure promulgated by the United States Supreme Court under section 2075 of Title 28 of the United States Code; (b) the Federal Rules of Civil Procedure, as applicable to the Chapter 11 Case or any proceedings therein; and (c) the local rules of the Bankruptcy Court, all as in effect on the Petition Date, together with any amendments and modifications thereto that may subsequently be made applicable to the Chapter 11 Case.
- 1.17 <u>Business Day</u> means any day other than a Saturday, a Sunday or "legal holiday" (as defined in Bankruptcy Rule 9006(a)).
- 1.18 <u>By-Laws</u> means the amended and restated by-laws of Reorganized Debtor which (i) if the Closing of the Alternative Transaction occurs, shall be substantially in the form to be filed with the Plan Supplement and (ii) otherwise shall be substantially in the form of **Exhibit I**.
- 1.19 <u>Carlyle</u> means Carlyle MC Partners, L.P., a Delaware limited partnership, Carlyle Partners V-A MC, L.P., a Delaware limited partnership, Carlyle Partners V MC, L.P., a Delaware limited partnership, CP V Coinvestment A, L.P., a Delaware limited partnership, and CP V Coinvestment B, L.P., a Delaware limited partnership, collectively.
 - 1.20 Cash means legal tender of the United States of America.
- 1.21 <u>Cash Collateral Order</u> means, collectively, the *Final Order Under 11 U.S.C.* §§ 105(a), 361, 362, 363, 507 and 552 and Bankruptcy Rules 2002, 4001, 6004 and 9014 (I) Authorizing Debtor to Use Cash Collateral, (II) Granting Adequate Protection to Prepetition Secured Parties, (III) Modifying the Automatic Stay and (IV) Granting Related Relief [D.I. 94] and the interim Bankruptcy Court order entered in connection therewith [D.I. 45].
- 1.22 <u>Cause of Action</u> means any action, including any Avoidance Action, cause of action, liability, obligation, account, controversy, right to legal remedy, right to equitable remedy, right to payment, suit, debt, sum of money, damage, judgment, or Claim whatsoever, whether known or unknown, now or in the future, reduced to judgment, not reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, secured, or unsecured, whether alleged, asserted or assertable directly or derivatively, in law, equity, admiralty, or otherwise, arising under any applicable law, regulation, or similar governmental pronouncement.
- 1.23 <u>Certificate of Incorporation</u> means the certificate of incorporation of Reorganized Debtor which (i) if the Closing of the Alternative Transaction occurs, shall be substantially in the

form to be filed with the Plan Supplement and (ii) otherwise shall be substantially in the form of **Exhibit H**.

- 1.24 <u>Chapter 11 Case</u> means the voluntary case under Chapter 11 of the Bankruptcy Code commenced by the Debtor in the Bankruptcy Court on the Petition Date.
 - 1.25 <u>Claim</u> means a "claim," as defined in section 101(5) of the Bankruptcy Code.
- 1.26 <u>Class</u> means each category of Holders of Claims or Interests established under Article III pursuant to sections 1122 and 1123(a)(1) of the Bankruptcy Code.
- 1.27 <u>Closing</u> has the meaning given to such term in the Alternative Plan Sponsor Agreement or the Original Plan Sponsor Agreement, as applicable.
 - 1.28 <u>Common Equity Distribution</u> means the difference between the Total Equity Distribution and the Preferred Equity Distribution.
 - 1.29 <u>Common Interest</u> means any Interest in the Debtor issued by the Debtor prior to the Effective Date (including prior to the Petition Date) other than the Preferred Interests.
 - 1.30 <u>Confirmation</u> means the entry of the Confirmation Order by the Bankruptcy Court.
- 1.31 <u>Confirmation Date</u> means the date on which the Clerk of the Bankruptcy Court enters the Confirmation Order on the Bankruptcy Court's docket.
- 1.32 <u>Confirmation Hearing</u> means the hearing held by the Bankruptcy Court on confirmation of this Plan, as such hearing may be continued from time to time.
- 1.33 <u>Confirmation Order</u> means the order of the Bankruptcy Court confirming this Plan pursuant to section 1129 of the Bankruptcy Code.
- 1.34 <u>Credit Facility</u> means that certain Credit Agreement, dated as of July 17, 2017, as subsequently amended, modified or supplemented among HCR Home Health Care and Hospice, LLC, as borrower, HCR Manorcare Operations II, LLC, HCR Manorcare Heartland, LLC, Manor Care, Inc., HCR II Healthcare, LLC and HCR Healthcare, LLC as Holdcos, the Credit Facility Lenders and the Credit Facility Agent.
- 1.35 <u>Credit Facility Agent</u> means RD Credit, LLC, in its capacity as the administrative agent and collateral agent under the Credit Facility.
- 1.36 <u>Credit Facility Claims</u> means all Claims against the Debtor of the Credit Facility Agent and Credit Facility Lenders under the Credit Facility; <u>provided</u>, <u>however</u>, for the avoidance of doubt, if the Closing of the Alternative Transaction occurs, the Credit Facility Claims shall include, without limitation, the Make-Whole Amount to the extent payable in accordance with the terms of the Credit Facility.
 - 1.37 <u>Credit Facility Lenders</u> means the lenders under the Credit Facility.

- 1.38 Debtor means HCR ManorCare, Inc., as debtor and debtor in possession.
- 1.39 <u>Disclosure Statement</u> means the written disclosure statement that relates to this Plan including, without limitation, all exhibits and schedules thereto, as the same may be amended, supplemented or otherwise modified from time to time, in a manner acceptable to the Debtor and the Plan Sponsor, as approved by the Bankruptcy Court pursuant to section 1125 of the Bankruptcy Code.
- 1.40 <u>Effective Date</u> means, and shall occur on, the Business Day on which each of the conditions precedent to the occurrence of the Effective Date set forth in <u>Article VIII</u> has been satisfied or waived in accordance with the terms thereof.
- 1.41 <u>Eligible Employee</u> means each of the current Chief Executive Officer, Chief Operating Officer, Chief Financial Officer and General Counsel of the Debtor as of the Petition Date.
 - 1.42 Entity means an entity as defined in section 101(15) of the Bankruptcy Code.
- 1.43 <u>Estate</u> means the estate of the Debtor created in the Chapter 11 Case under section 541 of the Bankruptcy Code.
- 1.44 <u>Exculpated Fiduciaries</u> means each of the following solely in their capacity as such: (a) the Debtor; (b) the Reorganized Debtor; and (c) with respect to each of the foregoing parties, such Entities' directors, officers, and professionals.
- 1.45 <u>Exculpated Parties</u> means, collectively, the Exculpated Fiduciaries and the Section 1125(e) Parties.
 - 1.46 Executory Contracts means all executory contracts to which the Debtor is a party.
 - 1.47 Exhibit(s) means, individually or collectively, the exhibits to this Plan.
- 1.48 Final Order means an order or judgment of the Bankruptcy Court (or other court of competent jurisdiction) entered by the clerk of the Bankruptcy Court on the docket in the Chapter 11 Case (or on the docket of any other court of competent jurisdiction), which has not been reversed, vacated or stayed and as to which (a) the time to appeal, petition for *certiorari* or move for a new trial, reargument or rehearing has expired and as to which no appeal, petition for *certiorari* or other proceedings for a new trial, reargument or rehearing shall then be pending, or (b) if an appeal, writ of *certiorari*, new trial, reargument or rehearing thereof has been sought, such order or judgment of the Bankruptcy Court shall have been affirmed by the highest court to which such order was appealed, or *certiorari* shall have been denied or a new trial, reargument or rehearing shall have been denied or resulted in no modification of such order, and the time to take any further appeal, petition for *certiorari* or move for a new trial, reargument or rehearing shall have expired; <u>provided</u>, <u>however</u>, that the possibility that a motion pursuant to section 502(j) or 1144 of the Bankruptcy Code or under Rule 59 or Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Bankruptcy Rules, may be filed relating to such order shall not solely cause such order not to be a Final Order.

- 1.49 <u>Former CEO Settlement Agreement</u> has the meaning given to such term in the Plan Sponsor Agreement.
- 1.50 <u>General Unsecured Claim</u> means a Claim against the Debtor that is not an Administrative Expense Claim, a Priority Tax Claims, a Secured Claim, a Credit Facility Claim, a QCP Claim, a Severance Claim, an Intercompany Claim, or a Section 510(b) Claim.
- 1.51 <u>Governmental Unit</u> has the meaning <u>provided</u> in section 101(27) of the Bankruptcy Code.
- 1.52 <u>Guaranty</u> means that certain Guaranty of Obligations made by the Debtor, effective as of February 11, 2013, as amended, supplemented or modified from time to time in accordance with its terms.
 - 1.53 HCR III means HCR III Healthcare, LLC.
 - 1.54 Holder means an Entity holding a Claim against, or Interest in, the Debtor.
- 1.55 <u>Impaired</u> means "impaired" within the meaning of section 1124 of the Bankruptcy Code.
- 1.56 <u>Implementation Memorandum</u> means that certain document describing the procedures and transactions required for implementation of the Alternative Transaction on the date of the Closing, which Implementation Memorandum shall be filed no later than thirty (30) days after execution of the Alternative Plan Sponsor Agreement and shall be subject to the consent of the Debtor (which consent shall not be unreasonably withheld or delayed); <u>provided</u> that no such procedures or transactions shall have any force or effect prior to the occurrence of the Closing of the Alternative Transaction; <u>provided further</u> that nothing in the Implementation Memorandum shall delay or impede ProMedica's obligation to obtain required regulatory approval or adversely impact the distributions to be made to any Holder of a Claim or Interest under the Plan.
- 1.57 <u>Intercompany Claim</u> means any Claim against the Debtor that is held by a non-Debtor subsidiary of the Debtor.
- 1.58 <u>Intercompany Note</u> means that certain \$25 million Subordinated Secured Demand Note, dated March 1, 2018, made by Debtor to HCR Home Health Care and Hospice, LLC.
 - 1.59 <u>Intercompany Note Claims</u> means the secured claims of HCR Home Health Care and Hospice, LLC against the Debtor in respect of the Intercompany Note.
- 1.60 <u>Interest</u> means any equity security within the meaning of section 101(16) of the Bankruptcy Code, including any issued and outstanding common stock, preferred stock, limited liability company interest, partnership interest, or any other instrument evidencing an ownership interest in the Debtor prior to the Effective Date (including prior to the Petition Date), whether or not transferable, and any restricted stock units, calls, rights, puts, awards, commitments, repurchase rights, unvested or unexercised options, rights of conversion, warrants, unvested common interests, unvested preferred interests or any other agreements of any character related to the common or preferred interests of the Debtor, obligating the Debtor to issue, transfer, purchase, redeem, or sell

any equity interests or other equity securities, and any rights under any equity incentive plans, voting agreements and registration rights agreements regarding equity securities of the Debtor.

- 1.61 <u>Lessors</u> has the meaning given to such term in the Alternative Plan Sponsor Agreement or the Original Plan Sponsor Agreement, as applicable.
- 1.62 <u>Lien</u> means, with respect to any interest in property, any mortgage, "lien" as defined in section 101(37) of the Bankruptcy Code, pledge, charge, security interest, easement or encumbrance of any kind whatsoever affecting such interest in property.
- 1.63 <u>Make-Whole Amount</u> has the meaning given to such term in section 2.4(b) of the Credit Facility.
- 1.64 Merger Agreement means the Agreement and Plan of Merger by and among Welltower Inc., Potomac Acquisition LLC, Quality Care Properties, Inc., QCP AL REIT, LLC, QCP SNF West REIT, LLC, QCP SNF Central REIT, LLC, QCP SNF East, LLC, QCP HoldCo REIT, LLC, and QCP TRS, LLC, dated as of April 25, 2018, as amended, modified, or supplemented from time to time in accordance with its terms, and including all schedules and exhibits thereto.
- 1.65 <u>MLSA</u> means that certain Master Lease and Security Agreement, among certain Affiliates of QCP who are the parties thereto, as "Lessor", and HCR III, as "Lessee", dated as of April 7, 2011, as amended, modified or supplemented from time to time.
- 1.66 New Common Stock means the new common stock to be issued by Reorganized Debtor on the Effective Date authorized by this Plan, which shall have the powers, preferences and rights and be subject to the limitations, qualifications and restrictions, in each case, as set forth in the Certificate of Incorporation.
- 1.67 Ordinary Litigation Claims means the claims, rights of action, suits or proceedings, whether in law or in equity, whether known or unknown, that the Debtor or its Estate may hold against any Person as of the Petition Date; provided, however, Ordinary Litigation Claims shall not include (a) any claim, right of action, suit or proceeding that has been settled on or prior to the Effective Date, in accordance with the Plan Sponsor Agreement, (b) any claim, right of action, suit or proceeding that the Debtor or its Estate may hold under chapter 5 of the Bankruptcy Code, including any Avoidance Action, and (c) claims, rights of action, suits or proceedings waived or released pursuant to Article IX.
- 1.68 <u>Original Master Lease</u> means the MLSA, as amended by the Original Master Lease Amendment.
- 1.69 <u>Original Master Lease Amendment</u> means the Original Master Lease Amendment attached hereto as <u>Exhibit C</u>.
- 1.70 Original Plan Sponsor Agreement means the Plan Sponsor Agreement among the Debtor, QCP, HCP Mezzanine Lender, LP, a Delaware limited partnership and a wholly-owned subsidiary of QCP, and the Lessors identified therein, dated as of March 2, 2018, as amended, modified or supplemented from time to time in accordance with its terms (including, without

limitation, by the Amendment to Original Plan Sponsor Agreement), and including all schedules and exhibits thereto. A copy of the Original Plan Sponsor Agreement is attached hereto as **Exhibit D**.

- 1.71 <u>Original Restructuring Support Agreement</u> means the Restructuring Support Agreement among the Debtor, Carlyle, and MC Operations Investments, LLC, dated as of March 2, 2018, as amended, modified or supplemented from time to time in accordance with its terms.
 - 1.72 <u>Original Transaction</u> shall have the meaning ascribed to it in <u>Article V</u> hereof.
- 1.73 Other Priority Claim means an Allowed Claim under section 507(a) of the Bankruptcy Code other than an Administrative Expense Claim or Priority Tax Claim.
- 1.74 <u>Person</u> or <u>person</u> means a person as defined in section 101(41) of the Bankruptcy Code.
- 1.75 <u>Petition Date</u> means March 4, 2018, the date on which the Debtor commenced the Chapter 11 Case.
- 1.76 <u>Plan</u> means this *First Amended Chapter 11 Plan of Reorganization for HCR ManorCare, Inc.*, including all Exhibits, supplements, appendices and schedules hereto, either in its present form or as the same may be altered, amended or modified from time to time in accordance with the provisions of the Bankruptcy Code and the terms hereof.
- 1.77 <u>Plan Sponsor</u> means (i) ProMedica Health System, Inc., the sponsor under the Alternative Plan Sponsor Agreement or (ii) QCP, the sponsor under the Original Plan Sponsor Agreement, as applicable. For the avoidance of doubt, (a) absent the consent of the Debtor and the Plan Sponsor under the Original Plan Sponsor Agreement, the Plan Sponsor under the Alternative Plan Sponsor Agreement shall not enter into any agreements or transactions on behalf of the Debtor, or take any actions that obligate or purport to obligate the Debtor, prior to the occurrence of the Closing of the Alternative Transaction, and no such agreements, transactions or actions shall become effective until the occurrence of the Closing of the Alternative Plan Sponsor Agreement, the Plan Sponsor under the Original Plan Sponsor Agreement shall not enter into any agreements or transactions on behalf of the Debtor, or take any actions that obligate or purport to obligate the Debtor, prior to the occurrence of the Closing of the Original Transaction, and no such agreements, transactions or actions shall become effective until the occurrence of the Closing of the Original Transaction, and no such agreements, transactions or actions shall become effective until the occurrence of the Closing of the Original Transaction.
- 1.78 <u>Plan Sponsor Agreement</u> means the Alternative Plan Sponsor Agreement or the Original Plan Sponsor Agreement, as applicable.
- 1.79 <u>Plan Supplement</u> means the supplement to this Plan to be filed with the Bankruptcy Court fifteen (15) days prior to the Confirmation Hearing on this Plan, together with any amendments, modifications, and/or supplements thereto, which shall be subject to the consent of the Debtor and the applicable Plan Sponsor.

- 1.80 <u>Preferred Equity Distribution</u> means Cash in the amount of \$2 million plus any Accrued Unpaid Dividends.
- 1.81 <u>Preferred Interest</u> means the preferred stock in the Debtor issued by the Debtor prior to the Effective Date (including prior to the Petition Date).
- 1.82 <u>Priority Tax Claim</u> means any Claim of a governmental unit of the kind against the Debtor entitled to priority in payment as specified in sections 502(i) and 507(a)(8) of the Bankruptcy Code.
- 1.83 <u>Professional</u> means any Person retained by the Debtor or a statutory committee, if any, pursuant to a Final Order of the Bankruptcy Court entered pursuant to sections 327, 328 or 1103 of the Bankruptcy Code.
- 1.84 <u>Professional Fee Claim</u> means any Claim of a Professional for allowance of compensation and/or reimbursement of costs and expenses incurred in the Chapter 11 Case on or before the Effective Date.
- 1.85 <u>ProMedica</u> means Suburban HealthCo, Inc., an Ohio non-profit corporation and a direct, wholly-owned subsidiary of ProMedica Health System, Inc., an Ohio non-profit corporation.
- 1.86 <u>ProMedica Plan Contribution</u> means Cash in the amount required to (i) pay the Allowed Credit Facility Claims in full, (ii) pay the Agreed Deferred Rent Obligation, and (iii) make the Total Equity Distribution, which shall take the form, at the election of ProMedica and ProMedica Health System, Inc., either of (x) a capital contribution to the Reorganized Debtor or (y) a combination of (A) the capital contribution referred to in clause (x) and (B) an unsecured, subordinated loan to the Reorganized Debtor in principal amount not to exceed \$550,000,000, which unsecured loan shall mature no earlier than the third anniversary of the date of the Closing and the other terms and conditions of which shall be reasonably acceptable to the Debtor.
 - 1.87 <u>QCP</u> means Quality Care Properties, Inc.
- 1.88 <u>QCP Claims</u> means all Claims of QCP and its subsidiaries against the Debtor, including, but not limited to, all Claims arising under the Guaranty and due and unpaid as of the Effective Date.
 - 1.89 Quarterly Fees has the meaning given to such term in Section 11.4 of the Plan.
- 1.90 <u>Receivership Complaint</u> means that certain complaint filed by QCP against the Debtor, HCR III and "John Does 1–50" in Los Angeles County Superior Court, dated on or about August 17, 2017, seeking, among other things, the appointment of a receiver for HCR III.
- 1.91 <u>Reinstate</u>, <u>Reinstated</u> or <u>Reinstatement</u> means (a) leaving unaltered the legal, equitable and contractual rights to which a Claim or Interest entitles the Holder of such Claim or Interest, or (b) notwithstanding any contractual provision or applicable law that entitles the Holder of such Claim or Interest to demand or receive accelerated payment of such Claim or Interest after the occurrence of a default, (i) curing any such default that occurred before or after the Petition Date, other than a default of a kind specified in section 365(b)(2) of the Bankruptcy Code;

- (ii) reinstating the maturity of such Claim or Interest as such maturity existed before such default; (iii) compensating the Holder of such Claim or Interest for any damages incurred as a result of any reasonable reliance by such Holder on such contractual provision or such applicable law; (iv) if such Claim or Interest arises from any failure to perform a nonmonetary obligation other than a default arising from failure to operate under a nonresidential real property lease subject to section 365(b)(1)(A) of the Bankruptcy Code, compensating the Holder of such Claim or Interest (other than the Debtor or an insider of the Debtor) for any actual pecuniary loss incurred by such Holder as the result of such failure; and (v) not otherwise altering the legal, equitable or contractual rights to which such Claim or Interest entitles the Holder thereof.
- Related Persons means, with respect to any Person, such Person's predecessors, successors and assigns (whether by operation of law or otherwise) and their respective present and former Affiliates and each of their respective current and former members, partners, equity holders, controlling persons, officers, directors, employees, managers, shareholders, partners, financial advisors, attorneys, accountants, investment bankers, consultants, agents and professionals each acting in such capacity, and any Person claiming by or through any of them (including their respective officers, directors, managers, shareholders, partners, employees, equity holders, members, and professionals); provided, however, that when used in reference to the Debtor or Reorganized Debtor, the term Related Persons shall not include the Debtor's and Reorganized Debtor's direct or indirect subsidiaries; provided, further, that (x) no Eligible Employee shall be a Related Person unless he executes and delivers a Separation Agreement and (y) Paul Ormond shall not be a Related Person unless he executes and delivers the Former CEO Settlement Agreement, in each case, prior to the Effective Date.
- 1.93 <u>Released Parties</u> means each of the following solely in its capacity as such: (a) the Debtor; (b) the Reorganized Debtor; (c) Carlyle; (d) the Credit Facility Agent; (e) the Credit Facility Lenders; (f) the Plan Sponsor under the Alternative Plan Sponsor Agreement, (g) the Plan Sponsor under the Original Plan Sponsor Agreement, (h) Welltower; and (i) with respect to each of the foregoing parties under (a) through (h), such Entities' Related Persons; <u>provided</u>, if the Closing of the Alternative Transaction does not occur, neither the Plan Sponsor under the Alternative Plan Sponsor Agreement, nor Welltower, or any Related Person to such Entities, shall constitute Released Parties.
- 1.94 <u>Reorganized Debtor</u> means the Debtor and any successors thereto by merger, consolidation, conversion or otherwise, on or after the Effective Date, after giving effect to the transactions implementing this Plan.
- 1.95 <u>Restructuring Support Agreement</u> means the Alternative Restructuring Support Agreement or the Original Restructuring Support Agreement, as applicable.
- 1.96 <u>Section 510(b) Claim</u> means a Claim against the Debtor that is subordinated, or subject to subordination, pursuant to section 510(b) of the Bankruptcy Code, including, without limitation, a claim arising from the rescission or purchase of a sale or security of the Debtor or an affiliate of the Debtor, for damages arising from the purchase or sale of such security or for reimbursement or contribution on account of such Claim pursuant to section 502 of the Bankruptcy Code.

- 1.97 <u>Section 1125(e) Parties</u> means each of the following, solely in their capacity as such: (a) Carlyle; (b) the Credit Facility Agent; (c) the Credit Facility Lenders; (d) the Plan Sponsor under the Alternative Plan Sponsor Agreement; (e) the Plan Sponsor under the Original Plan Sponsor Agreement; (f) Welltower; and (g) with respect to each of the foregoing parties under (a) through (f), such Entities' directors, officers, and professionals; <u>provided</u>, if the Closing of the Alternative Transaction does not occur, the Plan Sponsor under the Alternative Plan Sponsor Agreement, Welltower, and/or any of their directors, officers or professionals, shall constitute Section 1125(e) Parties only to the extent permitted by applicable law.
- 1.98 <u>Secured Claim</u> means any Claim, other than the Credit Facility Claim and the Intercompany Note Claim, secured by a Lien on collateral in which the Estate has an interest, to the extent of the value of such collateral (i) as agreed to by the Holder of such Claim and the Debtor or (ii) as determined pursuant to a Final Order of the Bankruptcy Court in accordance with section 506(a) of the Bankruptcy Code or, in the event that such Claim is subject to setoff under section 553 of the Bankruptcy Code, to the extent of such setoff.
- 1.99 <u>Securities Act</u> means the Securities Act of 1933, as now in effect or hereafter amended, and the rules and regulations of the U.S. Securities and Exchange Commission promulgated thereunder.
- 1.100 <u>Separation Agreement</u> has the meaning given to such term in the Plan Sponsor Agreement.
- 1.101 <u>Severance Claim</u> means a claim against the Debtor for damages resulting from the termination of an employment contract to which the Debtor is a party.
 - 1.102 Total Equity Distribution means Cash in the amount of \$50 million.
- 1.103 <u>Unimpaired</u> means with respect to a Claim, a Claim that is not Impaired, including any Claim that is Reinstated.
- 1.104 <u>United States Trustee</u> means the Office of the United States Trustee for the District of Delaware.
 - 1.105 Welltower means Welltower Inc. and Meerkat I LLC.
- B. Rules of Interpretation. For purposes of this Plan, unless otherwise provided herein: (a) whenever from the context it is appropriate, each term, whether stated in the singular or the plural, will include both the singular and the plural; (b) unless otherwise provided in this Plan, any reference in this Plan to a contract, instrument, release, or other agreement or document being in a particular form or on particular terms and conditions means that such document will be substantially in such form or substantially on such terms and conditions, subject to the Plan Sponsor Agreement; (c) any reference in this Plan to an existing document, schedule or Exhibit filed or to be filed means such document, schedule or Exhibit, as it may have been or may be amended, modified, or supplemented pursuant to this Plan and, as applicable, the Plan Sponsor Agreement; (d) any reference to an entity as a Holder of a Claim or Interest includes that entity's successors and assigns; (e) all references in this Plan to Sections or Articles are references to Sections or Articles of this Plan or the Plan Supplement, as the same may be amended, waived or modified from time to

time; (f) the words "herein," "hereof," "hereto," "hereunder" and other words of similar import refer to this Plan as a whole and not to any particular Section, subsection or clause contained in this Plan; (g) captions and headings to Articles and Sections are inserted for convenience of reference only and are not intended to be a part of or to affect the interpretation of this Plan; (h) the rules of construction set forth in section 102 of the Bankruptcy Code (other than section 102(5) of the Bankruptcy Code) will apply; and (i) any reference to an Entity's "subsidiaries" means its direct and indirect subsidiaries.

- C. <u>Computation of Time.</u> In computing any period of time prescribed or allowed by this Plan, unless otherwise expressly provided, the provisions of Bankruptcy Rule 9006(a) shall apply. In the event that any payment, distribution, act or deadline under this Plan is required to be made or performed or occurs on a day that is not a Business Day, then the making of such payment or distribution, the performance of such act or the occurrence of such deadline shall be deemed to be on the next succeeding Business Day, but if so made, performed or completed by such next succeeding Business Day shall be deemed to have been completed or to have occurred as of the required date.
- D. <u>Exhibits and Plan Supplement</u>. All Exhibits to this Plan, as well as the Plan Supplement, are incorporated into and are a part of this Plan as if set forth in full herein. Holders of Claims and Interests may obtain a copy of the Plan Supplement and the filed Exhibits upon written request to the Debtor. Upon their filing, the Plan Supplement and the Exhibits may be inspected (i) in the office of the Clerk of the Bankruptcy Court during normal business hours, (ii) at the Bankruptcy Court's website at http://www.deb.uscourts.gov, or (iii) free of charge on the Debtor's restructuring website at http://dm.epiq11.com/HCR.
- E. <u>Deemed Acts</u>. Whenever an act or event is expressed under this Plan to have been deemed done or to have occurred, it shall be deemed to have been done or to have occurred by virtue of this Plan and/or Confirmation Order without any further act by any party.

ARTICLE II: TREATMENT OF ADMINISTRATIVE EXPENSE CLAIMS AND PRIORITY TAX CLAIMS

In accordance with section 1123(a)(1) of the Bankruptcy Code, Administrative Expense Claims and Priority Tax Claims have not been classified and thus are excluded from the Classes of Claims and Interests set forth in <u>Article III</u>.

2.1. Administrative Expense Claims.

(a) General.

Subject to the terms of the Confirmation Order and Section 2.1(b) of this Plan, each Holder of an Allowed Administrative Expense Claim shall receive, on account of and in full and complete settlement, release and discharge of, and in exchange for, such Allowed Administrative Expense Claim, payment of Cash equal to the full unpaid amount of such Allowed Administrative Expense Claim (i) on the Effective Date or as soon as reasonably practicable thereafter, (ii) such other date as the Reorganized Debtor and such Holder shall have agreed, or (iii) as otherwise ordered by the Bankruptcy Court, provided, however, that Allowed Administrative Expense Claims not yet due or

that represent obligations incurred by the Debtor in the ordinary course of its business during the Chapter 11 Case, or assumed by the Debtor during the Chapter 11 Case, shall be paid or performed when due in the ordinary course of business and in accordance with the terms and conditions of the particular agreements governing such obligations.

(b) Professionals Fee Claims.

All Professionals or other entities requesting compensation or reimbursement of expenses pursuant to sections 327, 328, 330, 331, 503(b) and/or section 1103 of the Bankruptcy Code for services rendered before the Effective Date (including, without limitation, any compensation requested by any Professional or any other entity for making a substantial contribution in the Chapter 11 Case) shall file and serve final requests for payment of Professional Fee Claims no later than the first Business Day that is forty-five (45) days after the Effective Date. Objections to any Professional Fee Claim must be filed and served on the Reorganized Debtor and the applicable Professional within thirty (30) days after the filing of the final fee application with respect to the Professional Fee Claim. Any such objections that are not consensually resolved may be set for hearing on twenty-one (21) days' notice by the Professional asserting such Professional Fee Claim.

- 2.2. <u>Priority Tax Claims</u>. Each Holder of an Allowed Priority Tax Claim against the Debtor shall receive, on account of and in full and complete settlement, release and discharge of, and in exchange for, such Allowed Priority Tax Claim, payment in full in Cash, on the latest of: (a) the Effective Date; (b) the date such Priority Tax Claim becomes an Allowed Claim; and (c) the date such Allowed Priority Tax Claim becomes due and payable under applicable non-bankruptcy law.
- 2.3. <u>Post-Effective Date Fees and Expenses</u>. Except as otherwise specifically provided in the Plan, from and after the Effective Date, the Reorganized Debtor shall, in the ordinary course of business and without any further notice to or action, order or approval of the Bankruptcy Court, pay in Cash the legal, professional or other fees and expenses related to the implementation and consummation of the Plan incurred by the Reorganized Debtor following the Effective Date. Upon the Effective Date, any requirement that Professionals comply with sections 327 through 331 and 1103 of the Bankruptcy Code in seeking retention or compensation for services rendered after such date shall terminate, and each Reorganized Debtor may employ and pay any professional for services rendered or expenses incurred after the Effective Date in the ordinary course of business without any further notice to or action, order or approval of the Bankruptcy Court.

ARTICLE III: CLASSIFICATION AND TREATMENT OF CLASSIFIED CLAIMS AND INTERESTS

3.1. Summary of Classification and Treatment of Classified Claims and Interests.

3.1.1 General.

(a) Pursuant to sections 1122 and 1123 of the Bankruptcy Code, Claims and Interests are classified for all purposes, including, without limitation, voting, Confirmation and distributions pursuant to this Plan, as set forth herein. A Claim or Interest shall

be deemed classified in a particular Class only to the extent that the Claim or Interest qualifies within the description of that Class, and shall be deemed classified in a different Class to the extent that any remainder of such Claim or Interest qualifies within the description of such different Class. A Claim or Interest is in a particular Class only to the extent that such Claim or Interest is Allowed in that Class and has not been paid or otherwise settled prior to the Effective Date.

- (b) Any Class of Claims or Interests that, as of the commencement of the Confirmation Hearing, does not have at least one Holder of a Claim or Interest that is Allowed in an amount greater than zero for voting purposes shall be considered vacant, deemed eliminated from the Plan for purposes of voting to accept or reject the Plan, and disregarded for purposes of determining whether the Plan satisfies section 1129(a)(8) of the Bankruptcy Code with respect to that Class.
- (c) Pursuant to Bankruptcy Rule 9019 and section 1123(b)(3) of the Bankruptcy Code, and in consideration for the classification, distribution and other benefits provided under the Plan, the provisions of the Plan shall constitute a good faith compromise and settlement of all Claims and controversies resolved pursuant to the Plan, including all Claims, causes of action and controversies arising prior to the Effective Date, whether known or unknown, foreseen or unforeseen, asserted or unasserted, by or against any Released Party, or Holders of Claims, arising out of, relating to or in connection with the business or affairs of or transactions with the Debtor. The entry of the Confirmation Order shall constitute the Bankruptcy Court's approval of each of the foregoing compromises or settlements, and all other compromises and settlements provided for in the Plan, and settlements are in the best interests of the Debtor, the Estate, creditors and other parties in interest, and are fair, equitable and within the range of reasonableness. The provisions of the Plan, including its release, injunction, exculpation and compromise provisions, are mutually dependent and non-severable.

3.1.2 <u>Identification of Classes Against the Debtor</u>. The following chart assigns a number to each Class for purposes of identifying each separate Class:

<u>CLASS</u>	<u>CLAIM OR INTEREST</u>
1	Other Priority Claims
2	Secured Claims
3	Credit Facility Claims
4	QCP Claims
5	General Unsecured Claims
6	Severance Claims
7	Section 510(b) Claims

8A Preferred Interests

8B Common Interests

3.2. <u>Classification and Treatment of Claims Against and Interests in the Debtor</u>

3.2.1 <u>Class 1: Other Priority Claims.</u>

- (a) Classification: Class 1 consists of all Other Priority Claims against the Debtor.
- (b) Treatment: Each Holder of an Allowed Other Priority Claim will receive, in the sole discretion of the Reorganized Debtor, (i) payment in full in Cash as promptly as reasonably practicable on the later of (a) the Effective Date and (b) the date on which such Other Priority Claim becomes an Allowed Claim payable under applicable law or any agreement relating thereto, or (ii) treatment of such Allowed Other Priority Claim in any other manner that renders the claim Unimpaired, including Reinstatement. All Allowed Other Priority Claims not due and payable on or before the Effective Date shall be paid in the ordinary course of business in accordance with the terms thereof.
- (c) Voting: Allowed Claims in Class 1 are Unimpaired. Each Holder of an Allowed Claim in Class 1 shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

3.2.2 Class 2: Secured Claims.

- (a) Classification: Class 2 consists of Secured Claims against the Debtor.
- (b) Treatment: Except to the extent a Holder of a Secured Claim agrees to different treatment of that Secured Claim, each Holder of an Allowed Secured Claim shall, in the sole discretion of the Reorganized Debtor, receive on the Effective Date (or as promptly thereafter as reasonably practicable) or in the ordinary course of the Reorganized Debtor's business (a) payment in full by Reorganized Debtor in Cash, including the payment of any interest Allowed and payable under section 506(b) of the Bankruptcy Code, (b) delivery of the collateral securing such Allowed Secured Claim, or (c) treatment of such Allowed Secured Claim in any other manner that renders the Claim Unimpaired, including Reinstatement.
- (c) Voting: Allowed Claims in Class 2 are Unimpaired. Each Holder of an Allowed Claim in Class 2 shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

3.2.3 <u>Class 3: Credit Facility Claims.</u>

(a) Classification: Class 3 consists of the Credit Facility Claims against the Debtor.

(b) Allowance: If the Closing of the Alternative Transaction occurs, the Credit Facility Claims shall be Allowed in the principal amount of \$550,000,000 plus all unpaid fees, costs, expenses, interest, charges, premiums and other obligations due under the Credit Facility, including, without limitation, the Make-Whole Amount to the extent payable in accordance with the terms of the Credit Facility; provided, however, that if the Closing of the Original Transaction occurs, the Credit Facility Claims shall be Allowed to the extent required to Reinstate the Credit Facility Claims under the applicable provisions of the Bankruptcy Code. For the avoidance of doubt, if the Closing of the Original Transaction occurs and the Credit Facility Claims are Reinstated, the Make-Whole Amount shall not be due and payable at the time of Reinstatement.

(c) Treatment:

- (i) If the Closing of the Alternative Transaction occurs, the Credit Facility Agent shall receive, on the Effective Date, in full and final satisfaction, release, and discharge of (including any Liens related thereto), and in exchange for, the Allowed Credit Facility Claims, payment in full, in Cash, of all such Allowed Credit Facility Claims.
- (ii) If the Closing of the Alternative Transaction does not occur, on the Effective Date, the Credit Facility Claims shall be Reinstated and any liens held by the Credit Facility Agent, on behalf of itself and the Credit Facility Lenders, against the assets of the Debtor securing the Credit Facility Claims shall survive the Effective Date in the same priority as they existed as of the Petition Date.
- (d) Voting: Claims in Class 3 are Unimpaired. Each Holder of an Allowed Claim in Class 3 shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

3.2.4 Class 4: QCP Claims.

- (a) Classification: Class 4 consists of all QCP Claims.
- (b) Allowance: The QCP Claims shall be deemed Allowed on the Effective Date in the aggregate amount of \$445,796,590.

(c) Treatment:

- (i) If the Closing of the Alternative Transaction occurs, on the Effective Date, (x) the Agreed Deferred Rent Obligation shall be paid, in full, in Cash, and (y) the balance of the QCP Claims, including any such QCP Claims that are Administrative Expense Claims, shall be waived and released.
- (ii) If the Closing of the Alternative Transaction does not occur, the Holders of the QCP Claims (or the designee(s) of QCP) shall receive, on the Effective Date, in full and final satisfaction, release, and discharge of, and in exchange for, their QCP Claims, 100% of the New Common Stock of Reorganized Debtor.

(d) Voting: Allowed Claims in Class 4 are Impaired. The Holder of an Allowed QCP Claim in Class 4 shall be entitled to vote to accept or reject this Plan.

3.2.5 Class 5: General Unsecured Claims.

- (a) Classification: Class 5 consists of all General Unsecured Claims.
- (b) Treatment: Except to the extent a Holder of an Allowed General Unsecured Claim agrees to different treatment of that General Unsecured Claim, each Holder of an Allowed General Unsecured Claim shall receive, in the sole discretion of the Reorganized Debtor, (i) payment in full in Cash of such Claim (a) on the Effective Date or as promptly thereafter as reasonably practicable, (b) in the ordinary course of the Reorganized Debtor's business, or (c) on the date such Allowed General Unsecured Claim becomes due and payable according to its terms; or (ii) satisfaction of such Allowed General Unsecured Claim in any other manner that renders the Allowed General Unsecured Claim Unimpaired, including Reinstatement.
- (c) Voting: Allowed Claims in Class 5 are Unimpaired. Each Holder of an Allowed Claim in Class 5 shall be conclusively deemed to have accepted this Plan pursuant to section 1126(f) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan.

3.2.6 Class 6: Severance Claims

- (a) Classification: Class 6 consists of all Severance Claims.
- (b) Allowance: On the Effective Date, the Severance Claims shall be Allowed in the maximum amount permitted under the applicable provisions of the Bankruptcy Code.
- (c) Treatment: Each Holder of an Allowed Severance Claim shall receive, in the sole discretion of the Plan Sponsor, payment in full in Cash on account of such Allowed Severance Claim (a) on the Effective Date or as soon as thereafter as reasonably practicable or (b) as otherwise agreed by the Reorganized Debtor and such Holder of a Severance Claim.
- (d) Voting: Allowed Claims in Class 6 are Unimpaired, and the Holders of such Claims are conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code.

3.2.7 Class 7: Section 510(b) Claims.

(a) Classification: Class 7 consists of all Section 510(b) Claims against the Debtor.

(b) Treatment:

(i) If the Closing of the Alternative Transaction occurs, except to the extent a Holder of an Allowed Section 510(b) Claim, if any, agrees to different

treatment of that Allowed Section 510(b) Claim, each such Holder of an Allowed Section 510(b) Claim shall receive satisfaction of such Allowed Section 510(b) Claim in any manner determined by the Debtor, with the consent of the Plan Sponsor, that renders such Allowed Section 510(b) Claim Unimpaired, including Reinstatement.

(ii) If the Closing of the Alternative Transaction does not occur, on the Effective Date, all Section 510(b) Claims shall be discharged and extinguished and the Holders thereof shall not receive or retain any property under this Plan on account of such Section 510(b) Claims.

(c) Voting: Claims in Class 7 are Impaired because each Holder of an Allowed Section 510(b) Claim, if any, will not be entitled to receive or retain any property under this Plan if the Closing of the Alternative Transaction does not occur. Accordingly, each Holder of an Allowed Claim in Class 7 shall be conclusively deemed to have rejected this Plan pursuant to section 1126(g) of the Bankruptcy Code, and, therefore, shall not be entitled to vote to accept or reject this Plan

3.2.8 Class 8A: Preferred Interests.

- (a) Classification: Class 8A consists of all Preferred Interests.
- (b) Treatment:

(i) If the Closing of the Alternative Transaction occurs, on the Effective Date or as soon thereafter as reasonably practicable, each Holder of an Allowed Preferred Interest shall receive its pro rata share of the Preferred Equity Distribution.

(ii) If the Closing of the Alternative Transaction does not occur, on the Effective Date, all Preferred Interests in the Debtor shall be cancelled, annulled, and extinguished and the Holders of such Preferred Interests shall not receive or retain any property under this Plan on account of such Preferred Interests nor receive any distributions or property under this Plan on account of such Preferred Interests.

(c) Voting: Preferred Interests in Class 8A are Impaired because each Holder of Preferred Interests will not be entitled to receive or retain any property under the Plan if the Closing of the Alternative Transaction does not occur. Each Holder of a Preferred Interest in Class 8A shall be entitled to vote to accept or reject this Plan.

3.2.9 Class 8B: Common Interests.

- (a) Classification: Class 8B consists of all Common Interests.
- (b) Treatment:

(i) If the Closing of the Alternative Transaction occurs, on the Effective Date or as soon thereafter as reasonably practicable, each Holder of an Allowed Common Interest shall receive its pro rata share of the Common Equity Distribution.

- (ii) If the Closing of the Alternative Transaction does not occur, on the Effective Date, all Common Interests in the Debtor shall be cancelled, annulled, and extinguished and the Holders of such Common Interests shall not receive or retain any property under this Plan on account of such Common Interests nor receive any distributions or property under this Plan on account of such Common Interests.
- (c) Voting: Common Interests in Class 8B are Impaired. Each Holder of a Common Interest in Class 8B shall be entitled to vote to accept or reject this Plan.

3.3. Intercompany Claims.

- 3.3.1 On the Effective Date, in the sole discretion of the Reorganized Debtor, all Intercompany Claims shall either be (i) Reinstated, in whole or in part, (ii) deemed satisfied, or (iii) discharged and extinguished, in full or in part, and shall be eliminated as of the Effective Date, in whole or in part, in which case the Holders thereof shall not be entitled to and shall not receive or retain any property or interest on account of such discharged and extinguished portion under this Plan; provided, however, that prior to such discharge and extinguishment such Intercompany Claims may be contributed to capital, transferred, setoff or subject to any other arrangement in the sole discretion of the Reorganized Debtor; provided further, that any Intercompany Claim held by an obligor under the Credit Facility shall not be discharged or extinguished without the express written consent of the Credit Facility Agent.
- 3.3.2 For the avoidance of doubt, the Intercompany Note Claim evidenced by the Intercompany Note shall: (i) if the Closing of the Alternative Transaction occurs, be Allowed in the amount of \$25,000,000 plus accrued and unpaid interest thereon and, in the sole discretion of the Plan Sponsor under the Alternative Plan Sponsor Agreement, either (A) HCR Home Health Care and Hospice, LLC shall receive, on the Effective Date, in full and final satisfaction, release, and discharge of (including any liens related thereto), and in exchange for, the Intercompany Note Claim, payment in full, in Cash, of the Intercompany Note Claim or (B) such Intercompany Note Claim shall be Reinstated under the Plan, and any liens held by HCR Home Health Care and Hospice, LLC against the assets of the Debtor securing the Intercompany Note shall survive the Effective Date with the same priority that such liens held as of the Petition Date; and (ii) if the Closing of the Alternative Transaction does not occur, such Intercompany Note Claims shall be Allowed in the amount of \$25,000,000 plus accrued and unpaid interest thereon and be Reinstated under the Plan, and any liens held by HCR Home Health Care and Hospice, LLC against the assets of the Debtor securing the Intercompany Note shall survive the Effective Date with the same priority that such liens held as of the Petition Date.

3.4. <u>Unimpaired Claims</u>.

3.4.1 Notwithstanding anything to the contrary in this Plan or the Plan Documents, and subject to Section 3.4.2, each Holder of a Claim arising prior to the Effective Date (i) in Classes 1, 2, 5 or 7 (if the Closing of the Alternative Transaction occurs) of this Plan or (ii) in Classes 1, 2, 3, or 5 (if the Closing of the Alternative Transaction does not occur) of the Plan (each of (i) and (ii), as applicable, a "Covered Unimpaired Claim") shall be entitled to enforce its rights in respect of its Covered Unimpaired Claim against the Reorganized Debtor, in each case, until such Claim has been (a) paid in full in accordance with applicable law, or on terms agreed to between the

Holder of such Claim and the Debtor or Reorganized Debtor, or in accordance with the terms and conditions of the particular transaction giving rise to such Claim or (b) otherwise satisfied or disposed of as determined by a court of competent jurisdiction. The Debtor, Reorganized Debtor and any other Entity shall retain all rights and defenses, both legal and equitable, under the Bankruptcy Code or other applicable bankruptcy and non-bankruptcy law with respect to any Covered Unimpaired Claim (other than the power to impair such Claim within the meaning of section 1124 of the Bankruptcy Code), including, but not limited to, legal and equitable rights of setoff and/or recoupment as to Covered Unimpaired Claims; provided, however, that (x) the foregoing shall not apply with respect to the Intercompany Note Claim, which is Allowed in full under this Plan without defenses, counterclaims, setoff or recoupment, and (y) for the avoidance of doubt, the Debtor, the Reorganized Debtor and any other Entity shall not retain such rights and defenses with respect to (i) any Claims or Causes of Action held by the Debtor or its Estate or the Reorganized Debtor that are released pursuant to Section 9.3 of this Plan (the "Debtor Released Claims") or (ii) any Claims of Causes of Action released by certain Holders of Claims pursuant to Section 9.4 of this Plan (the "Third Party Released Claims"). Such Debtor Released Claims and Third Party Released Claims shall be deemed settled, satisfied, resolved, released, discharged, barred and/or enjoined by the applicable provisions under, and pursuant to the express terms of, this Plan and Plan Documents on the Effective Date. Holders of Covered Unimpaired Claims shall not be required to file a Proof of Claim with the Court and shall retain all their rights under applicable non-bankruptcy law to pursue their Covered Unimpaired Claims in any forum with jurisdiction over the parties. If the Debtor or the Reorganized Debtor disputes any Covered Unimpaired Claim, such dispute shall be determined, resolved or adjudicated pursuant to applicable non-bankruptcy law.

3.4.2 Notwithstanding the foregoing or anything else to the contrary in this Plan or the Plan Documents, with respect to all Claims (a) in Class 5 arising from the rejection of an Executory Contract and (b) in Class 6: (i) the Allowed amount of such Claims may be determined by the Bankruptcy Court after notice and a hearing; (ii) such Claims may be paid in full in Cash by the Debtor on the Effective Date or as soon thereafter as reasonably practicable; and (iii) such Claims shall be discharged pursuant to Section 9.2.1 of this Plan and Holders of such Claims shall be subject to the injunction set forth in Section 9.2.2 of this Plan.

ARTICLE IV: ACCEPTANCE OR REJECTION OF THE PLAN

- 4.1. <u>Impaired Classes of Claims or Interests Entitled to Vote on this Plan</u>. Claims in Class 4 (QCP Claims) and Interests in Class 8A (Preferred Interests) and Class 8B (Common Interests) are Impaired, and the Holders of such Claims and Interests are entitled to vote to accept or reject this Plan.
- 4.2. Acceptance by an Impaired Class of Claims or Interests. Pursuant to section 1126(c) of the Bankruptcy Code, an Impaired Class of Claims shall have accepted the Plan if, after excluding any Claims held by any Holder whose Claims have been designated pursuant to section 1126(e) of the Bankruptcy Code, (a) the Holders of at least two-thirds in dollar amount of the Allowed Claims actually voting in such Class have voted to accept such Plan, and (b) more than one-half in number of such Allowed Claims actually voting in such Class have voted to accept the Plan. Pursuant to section 1126(d) of the Bankruptcy Code, an Impaired Class of Interests shall have accepted the Plan if, after excluding any Interests held by any Holder who has been designated

pursuant to section 1126(e) of the Bankruptcy Code, the Holders of more than two-thirds in amount of such Allowed Interests actually voting in such Class have voted to accept the Plan.

- 4.3. <u>Presumed Acceptance by Unimpaired Classes</u>. Classes 1 (Other Priority Claims), 2 (Secured Claims), 3 (Credit Facility Claims), 5 (General Unsecured Claims), and 6 (Severance Claims) are Unimpaired by this Plan. Pursuant to section 1126(f) of the Bankruptcy Code, the Holders of Claims in such Classes are conclusively presumed to have accepted this Plan and therefore shall not be entitled to vote to accept or reject this Plan.
- 4.4. Presumed Rejection by Certain Impaired Classes. Class 7 (Section 510(b) Claims) (if the Closing of the Alternative Transaction does not occur) is Impaired by this Plan. Holders of Claims in Class 7 (Section 510(b) Claims) (if the Closing of the Alternative Transaction does not occur) will not receive or retain any property under this Plan on account of such Claims. Pursuant to section 1126(g) of the Bankruptcy Code, the Holders of Claims in Class 7 (Section 510(b) Claims) are conclusively presumed to have rejected this Plan and therefore shall not be entitled to vote to accept or reject this Plan.
 - 4.5. Confirmability of this Plan and Reservation of Rights.
- 4.5.1 <u>Confirmation</u>. The confirmation requirements of section 1129 of the Bankruptcy Code must be satisfied with respect to the Debtor.
- 4.5.2 <u>Reservation of Rights</u>. The Debtor reserves the right to amend, modify, or supplement this Plan for any reason.

ARTICLE V: MEANS FOR IMPLEMENTATION OF THE PLAN

- 5.1. <u>Alternative Transaction</u>. The Alternative Transaction means the transaction described in the Alternative Plan Sponsor Agreement, pursuant to which ProMedica (or its designee(s)), on the Effective Date, shall provide the ProMedica Plan Contribution in exchange for 100% of the New Common Stock of Reorganized Debtor. In addition, on the Effective Date, the Alternative Master Lease shall be executed and delivered.
- 5.2. Original Transaction. The Original Transaction means the transaction described in the Original Plan Sponsor Agreement, pursuant to which the Holders of the QCP Claims (or the designee(s) of QCP) shall receive on the Effective Date, in full and final satisfaction, release, and discharge of, and in exchange for, their QCP Claims, 100% of the New Common Stock of Reorganized Debtor. The Debtor shall consummate the Original Transaction in accordance with its terms if the Alternative Plan Sponsor Agreement is terminated in accordance with its terms.
- 5.3. Operations Between the Confirmation Date and Effective Date. During the period from the Confirmation Date through and until the Effective Date, the Debtor may continue to operate its business as debtor in possession in the ordinary course in a manner consistent with its obligations under the applicable Plan Sponsor Agreement and the transactions contemplated by the Plan and the Plan Sponsor Agreement, subject to all applicable orders of the Bankruptcy Court and the provisions of the Bankruptcy Code.

5.4. <u>Sources of Cash Consideration for Plan Distributions</u>. The Reorganized Debtor shall fund distributions and satisfy applicable Allowed Claims under the Plan with Cash on hand and/or, if the Closing of the Alternative Transaction occurs, from such Cash on hand together with the ProMedica Plan Contribution.

5.5. New Common Stock.

- (a) If the Closing of the Alternative Transaction occurs, on the Effective Date, ProMedica (or its designee(s)) shall receive, and the Debtor shall, in accordance with the Implementation Memorandum, issue, transfer, and deliver to ProMedica (or its designee(s)), 1,000,000 newly issued shares of New Common Stock of the Debtor, par value \$0.01 per share, which shall constitute all of the issued and outstanding capital stock and rights to purchase or otherwise acquire capital stock of the Debtor, free and clear of any lien, charge, pledge, security interest, claim, or other encumbrance in consideration for the ProMedica Plan Contribution. If the Closing of the Alternative Transaction occurs, ProMedica (or its designee(s)) will receive the New Common Stock in accordance with the Implementation Memorandum.
- (b) If the Closing of the Alternative Transaction does not occur, on the Effective Date, QCP or its designee(s) shall receive, and the Debtor shall issue, transfer, and deliver to QCP or its designee(s), 1,000,000 newly issued shares of New Common Stock of the Reorganized Debtor, par value \$0.01 per share, which shall constitute all of the issued and outstanding capital stock and rights to purchase or otherwise acquire capital stock of the Debtor, free and clear of any lien, charge, pledge, security interest, claim, or other encumbrance, in full satisfaction of all of the QCP Claims.
- 5.6. Rollover Restricted Shares. At the Debtor's option, if the Closing of the Alternative Transaction occurs, subject to the approval of ProMedica Health System, Inc. (which approval shall not be unreasonably withheld, conditioned or delayed), (a) the Debtor shall accelerate the vesting of the 437,540 unvested, restricted common shares of the Debtor that do not vest automatically upon the consummation of the Alternative Transaction pursuant to the terms of the applicable Rollover Stock Agreements (the "Rollover Restricted Shares"), which Rollover Restricted Shares shall be entitled to receive their pro rata share of the Common Equity Distribution or (b) the pro rata share of the Common Equity Distribution applicable to the Rollover Restricted Shares shall be reserved and paid if and when such Rollover Restricted Shares vest.
- 5.7. Section 1145 Exemption. To the maximum extent provided by section 1145 of the Bankruptcy Code and applicable non-bankruptcy law, the offering, issuance and distribution of the New Common Stock of the Debtor shall be exempt from, among other things, the registration and prospectus delivery requirements of section 5 of the Securities Act, and any other applicable state and federal law requiring registration and/or delivery of a prospectus prior to the offering, issuance, distribution or sale of securities, subject to the provisions of section 1145(b)(1) of the Bankruptcy Code relating to the definition of an underwriter in section 2(a)(11) of the Securities Act. In addition, any securities issued under the Plan shall be freely transferable under the Securities Act by the recipients thereof, subject to compliance with any rules and regulations of the U.S. Securities and Exchange Commission, if any, applicable at the time of any future transfer of such Interests and applicable regulatory approval, if any.

5.8. Master Lease Amendment.

- (a) If the Closing of the Alternative Transaction occurs, on the Effective Date, the Reorganized Debtor shall cause HCR III to, and Meerkat I LLC (or its designees) shall, enter into the Alternative Master Lease.
- (b) If the Closing of the Alternative Transaction does not occur, on the Effective Date, Reorganized Debtor shall cause HCR III to, and QCP and the Lessors shall, enter into the Original Master Lease Amendment.
- 5.9. <u>Release of Guaranty</u>. On or prior to the third day after the Effective Date if the Alternative Transaction closes or on or prior to the seventh day after the Effective Date if the Original Transaction closes, HCR III and the Lessors under the MLSA shall terminate, release and discharge the Guaranty pursuant either to the Original Plan Sponsor Agreement or the Alternative Plan Sponsor Agreement, as applicable.

5.10. <u>Corporate Governance, Directors, Officers and Corporate Action.</u>

- 5.10.1 <u>Certificate of Incorporation; By-Laws</u>. On the Effective Date, the Certificate of Incorporation and the By-Laws shall go into effect. Consistent with, but only to the extent required by, section 1123(a)(6) of the Bankruptcy Code, on the Effective Date, the Certificate of Incorporation shall be amended to prohibit the issuance of non-voting equity securities. After the Effective Date, the Reorganized Debtor may amend and restate its certificates or articles of incorporation, by-laws, or similar governing documents, as applicable, as permitted by applicable law.
- 5.10.2 <u>Directors and Officers of the Reorganized Debtor</u>. Subject to any requirement of Bankruptcy Court approval pursuant to section 1129(a)(5) of the Bankruptcy Code, the initial directors and officers of Reorganized Debtor on the Effective Date, shall be as set forth on (i) **Exhibit F** if the Closing of the Alternative Transaction occurs or (ii) **Exhibit G** if the Closing of the Original Transaction occurs. After the Effective Date, the Certificate of Incorporation and the By-Laws, as each may be amended thereafter from time to time, shall govern the designation and election of directors.
- 5.10.3 <u>Corporate Action</u>. On the Effective Date, (i) the selection of directors and officers for Reorganized Debtor, (ii) the issuance and distribution of the New Common Stock, and (iii) all other actions and transactions contemplated by this Plan and the Plan Sponsor Agreement shall be deemed authorized and approved in all respects (subject to the provisions of this Plan and the Plan Sponsor Agreement). All matters provided for in this Plan and the Plan Sponsor Agreement involving the corporate structure of the Debtor or the Reorganized Debtor, and any corporate action required by the Debtor or the Reorganized Debtor in connection with this Plan and the Plan Sponsor Agreement, shall be deemed to have timely occurred in accordance with applicable law and shall be in effect, without any requirement of further action by the security holders or directors of the Debtor or the Reorganized Debtor in accordance with section 303 of the Delaware General Corporation Law and the provisions of the Bankruptcy Code. On and after the Effective Date, the appropriate officers of the Reorganized Debtor and members of the boards of directors or managers of the Reorganized Debtor shall be authorized and directed to issue, execute

and deliver the agreements, documents, securities and instruments contemplated by this Plan and the Plan Sponsor Agreement in the name of and on behalf of the Reorganized Debtor.

- 5.11. Continued Corporate Existence and Vesting of Assets in the Reorganized Debtor. On and after the Effective Date, after giving effect to each of the actions contemplated under this Plan, the Reorganized Debtor shall continue to exist in accordance with the applicable law in the jurisdiction in which it is formed. Pursuant to section 1141(b) of the Bankruptcy Code, except as otherwise provided under this Plan, all property of the Debtor's Estate, including all claims, rights and causes of action and any property acquired by the Debtor or the Reorganized Debtor under or in connection with this Plan, together with any property of the Debtor that is not property of its Estate and that is not specifically disposed of pursuant to this Plan, shall revest in the Reorganized Debtor on the Effective Date free and clear of all Claims, Liens, charges, other encumbrances and Interests, except as specifically provided in this Plan (including Sections 3.2.3(b), 3.2.4(c) and 3.3.2 herein) or the Confirmation Order. Thereafter, the Reorganized Debtor may operate its business and may use, acquire and dispose of property free of any restrictions of the Bankruptcy Code and the Bankruptcy Rules. As of the Effective Date, all property of the Reorganized Debtor shall be free and clear of all Liens and non-Reinstated Claims, except as specifically provided in this Plan (including Sections 3.2.3(b), 3.2.4(c) and 3.3.2 herein) or the Confirmation Order.
- 5.12. Cancellation of Liens. Except as otherwise provided in this Plan, on the Effective Date, in consideration for the distributions to be made on the Effective Date pursuant to this Plan, all Liens, charges, and encumbrances related to any Claim or Interest, other than any Lien securing a Secured Claim, an Intercompany Note Claim or a Credit Facility Claim that is, in each case, Reinstated pursuant to this Plan, shall be terminated, null and void and of no effect. The Holders of Secured Claims (other than a Secured Claim, Intercompany Note Claim or a Credit Facility Claim that is, in each case, Reinstated pursuant to this Plan) shall be authorized and directed to release any collateral or other property of the Debtor (including any Cash collateral) held by such Holder and to take such actions as may be requested by the Debtor (or the Reorganized Debtor, as the case may be) to evidence the release of any Liens, including the execution, delivery, and filing or recording of such release documents as may be requested by the Debtor (or the Reorganized Debtor, as the case may be).
- 5.13. Preservation of Rights of Action and Settlement of Ordinary Litigation Claims. Except as otherwise provided in this Plan, the Confirmation Order, or in any document, instrument, release or other agreement entered into in connection with this Plan, in accordance with section 1123(b)(3)(B) of the Bankruptcy Code, the Debtor and its Estate shall retain the Ordinary Litigation Claims. The Reorganized Debtor, as the successor in interest to the Debtor and the Estate, may enforce, sue on, settle or compromise (or decline to do any of the foregoing) any or all of the Ordinary Litigation Claims or any other claims, rights of action, suits or proceedings that the Debtor or its Estate may hold against any Person, and the failure to specifically identify any Ordinary Litigation Claim as being retained herein shall not effectuate a waiver or release of such Ordinary Litigation Claim.
- 5.14. <u>Registration of New Common Stock</u>. On the Effective Date, the New Common Stock shall not be listed for public trading on any securities exchange, the Reorganized Debtor will not be a reporting company under the Securities Exchange Act of 1934, and the Reorganized

Debtor shall not be required to file reports with the U.S. Securities and Exchange Commission or any other governmental entity.

- 5.15. <u>Additional Transactions Authorized Under this Plan</u>. On or after the Effective Date, the Reorganized Debtor shall be authorized to take any such actions as may be necessary or appropriate to Reinstate Claims or render Claims not Impaired, as provided for under this Plan.
- 5.16. Comprehensive Settlement of Claims and Controversies. Pursuant to Bankruptcy Rule 9019 and in consideration for the distributions and other benefits provided under the Plan, the provisions of the Plan will constitute a good faith compromise and settlement of all Claims or controversies relating to the rights that a Holder of a Claim or Interest may have with respect to any Allowed Claim or Allowed Interest or any distribution to be made pursuant to the Plan on account of any Allowed Claim or Allowed Interest. The entry of the Confirmation Order will constitute the Bankruptcy Court's approval, as of the Effective Date, of the compromise or settlement of all such claims or controversies and the Bankruptcy Court's finding that all such compromises or settlements are in the best interests (i) of the Debtor, the Reorganized Debtor, the Estate and their respective property, and (ii) Claim and Interest holders, and are fair, equitable and reasonable.
- 5.17. <u>Tax Election</u>. On or after the Effective Date, and subject to the Closing of the Alternative Transaction, the Reorganized Debtor shall be authorized to elect to opt out of the application of Section 382(l)(5) of the Internal Revenue Code of 1986, as amended, by filing such election as is required under Section 382(l)(5)(G) of the Internal Revenue Code of 1986, as amended.

ARTICLE VI: TREATMENT OF EXECUTORY CONTRACTS AND INSURANCE POLICIES

Assumption or Rejection of Executory Contracts. On the Effective Date, all Executory Contracts of the Debtor will be deemed assumed in accordance with, and subject to, the provisions and requirements of sections 365 and 1123 of the Bankruptcy Code, unless such Executory Contract (i) was previously assumed or rejected by the Debtor, (ii) previously expired or terminated pursuant to its own terms, or (iii) upon the Plan Sponsor's prior written consent or request, is subject to a motion to reject such Executory Contract filed prior to the Effective Date. Entry of the Confirmation Order by the Bankruptcy Court shall constitute approval of such assumptions and the rejection of any Executory Contract for which a motion to reject has been filed, pursuant to sections 365(a) and 1123 of the Bankruptcy Code. Each Executory Contract assumed pursuant to this Article VI shall revest in and be fully enforceable by the Reorganized Debtor in accordance with its terms, except as modified by the provisions of this Plan, or any order of the Bankruptcy Court authorizing and providing for its assumption. Any motions to assume or reject Executory Contracts pending on the Effective Date shall be subject to approval by the Bankruptcy Court on or after the Effective Date. To the maximum extent permitted by law, to the extent any provision in any Executory Contract assumed pursuant to the Plan restricts or prevents, or purports to restrict or prevent, or is breached or deemed breached by, the assumption of such Executory Contract, including any "change of control" provision, then such provision shall be deemed modified such that the transactions contemplated by the Plan shall not entitle the non-Debtor party thereto to terminate such Executory Contract or to exercise any other default-related rights with respect thereto.

- 6.2. <u>No Cure Obligations</u>. There are no anticipated cure obligations with respect to any Executory Contract to which the Debtor is a party. Unless otherwise determined by the Bankruptcy Court pursuant to a Final Order or agreed to by the parties thereto prior to the Effective Date, no payments shall be required to cure any defaults of the Debtor existing as of the Confirmation Date with respect to any Executory Contract assumed pursuant to this Plan. Subject to the occurrence of the Effective Date, the entry of the Confirmation Order shall constitute a finding by the Bankruptcy Court that (i) each such assumption is in the best interest of the Debtor and its Estate, and (ii) the requirements of section 365(b)(l) of the Bankruptcy Code are deemed satisfied.
- 6.3. <u>Insurance Policies and Agreements</u>. Insurance policies issued to, or insurance agreements entered into by, the Debtor prior to the Petition Date (including, without limitation, any policies covering directors' or officers' conduct) shall continue in effect after the Effective Date. To the extent that such insurance policies or agreements are considered to be Executory Contracts, this Plan shall constitute a motion to assume or ratify such insurance policies and agreements, and, subject to the occurrence of the Effective Date, the entry of the Confirmation Order shall constitute approval of such assumption pursuant to section 365(a) of the Bankruptcy Code and a finding by the Bankruptcy Court that each such assumption is in the best interest of the Debtor and its Estate. Unless otherwise determined by the Bankruptcy Court pursuant to a Final Order or agreed to by the parties thereto prior to the Effective Date, no payments shall be required to cure any defaults of the Debtor existing as of the Confirmation Date with respect to each such insurance policy.
- 6.4. <u>Postpetition Contracts and Leases</u>. All contracts, agreements and leases that were entered into by the Debtor or assumed by the Debtor after the Petition Date shall be deemed assigned by the Debtor to the Reorganized Debtor on the Effective Date.
- 6.5. Modifications, Amendments, Supplements, Restatements or Other Agreements. Unless otherwise provided in the Plan, each Executory Contract that is assumed shall include all modifications, amendments, supplements, restatements or other agreements that in any manner affect such Executory Contract including easements, licenses, permits, rights, privileges, immunities, options, rights of first refusal and any other interests, unless any of the foregoing agreements has been previously rejected or repudiated or is rejected or repudiated under the Plan. Modifications, amendments, supplements and restatements to prepetition Executory Contracts that have been executed by the Debtor during the Chapter 11 Case shall not be deemed to alter the prepetition nature of the Executory Contract, or the validity, priority or amount of any Claims that may arise in connection therewith.

ARTICLE VII: PROVISIONS GOVERNING DISTRIBUTIONS

7.1. <u>Distributions on Account of Claims Allowed as of the Effective Date</u>. Unless the Holder of an Allowed Claim and the Debtor or the Reorganized Debtor agree to a different date, and except as otherwise provided herein or as ordered by the Bankruptcy Court, distributions to be made on account of each Claim that is Allowed as of the Effective Date shall be made as set forth in the provisions of the Plan governing the treatment of such Claim or as soon thereafter as practicable.

- 7.2. <u>Distributions on Account of Claims that Become Allowed after the Effective Date</u>. Unless the Holder of a Claim that becomes an Allowed Claim after the Effective Date and the Reorganized Debtor agrees to a different date, and except as otherwise provided herein or as ordered by the Bankruptcy Court, distributions on account of Claims that become Allowed Claims after the Effective Date shall be made on the later of (a) the date such Claim becomes Allowed and (b) as set forth in the provisions of the Plan governing the treatment of such Claim or as soon thereafter as is practicable.
- 7.3. <u>Distributions by Reorganized Debtor</u>. Other than as specifically set forth in this Plan, the Reorganized Debtor shall make, or cause to be made, all distributions required to be made under this Plan.
- 7.4. Means of Cash Payment. Payments of Cash made pursuant to this Plan shall be in U.S. dollars and shall be made, at the option and in the sole discretion of the Reorganized Debtor, by (a) checks drawn on or (b) wire transfers from a bank selected by the Reorganized Debtor. Cash payments to foreign creditors may be made, at the option of the Reorganized Debtor, in such funds and by such means as are necessary or customary in a particular foreign jurisdiction.
- Withholding and Reporting Requirements. In connection with this Plan and all 7.5. distributions thereunder, the Reorganized Debtor shall comply with all withholding and reporting requirements imposed by any federal, state, local or foreign taxing authority, and all distributions hereunder shall be subject to any such withholding and reporting requirements. The Reorganized Debtor shall be authorized to take any and all actions that may be necessary or appropriate to comply with such withholding and reporting requirements, including liquidating a portion of the distribution to be made under the Plan to generate sufficient funds to pay applicable withholding taxes or establishing such other mechanisms that the Reorganized Debtor believes are reasonable and appropriate. All persons holding Claims or Interests shall be required to provide any information necessary to effect information reporting and the withholding of such taxes. Each Holder of an Allowed Claim that is to receive a distribution pursuant to this Plan shall have sole and exclusive responsibility for the satisfaction and payment of any tax obligations imposed by any governmental unit, including income, withholding and other tax obligations, on account of such distribution. No distribution shall be made to or on behalf of such Holder pursuant to this Plan unless and until such Holder has made arrangements satisfactory to the Reorganized Debtor for the payment and satisfaction of such tax obligations.
- 7.6. <u>Compliance Matters</u>. The Debtor or the Reorganized Debtor, as applicable, reserves the right to allocate and distribute all distributions made under the Plan in compliance with all applicable wage garnishments, alimony, child support and other spousal awards, Liens and similar encumbrances.
- 7.7. Setoff and Recoupment. The Reorganized Debtor may, pursuant to sections 553 and/or 558 of the Bankruptcy Code or applicable non-bankruptcy laws, but shall not be required to, set off and/or recoup against any Claim the payments or other distributions to be made pursuant to this Plan in respect of such Claim, or claims of any nature whatsoever that the Debtor or the Reorganized Debtor may have against the Holder of such Claim; provided, however, that neither the failure to assert such rights of setoff and/or recoupment nor the allowance of any Claim

hereunder shall constitute a waiver or release by the Reorganized Debtor of any claim that the Debtor or the Reorganized Debtor may assert against any Holder of an Allowed Claim.

- 7.8. Reinstated Claims. Notwithstanding anything contained herein to the contrary, nothing shall affect, diminish or impair the Debtor's or the Reorganized Debtor's rights and defenses, both legal and equitable, with respect to any Reinstated Claim, including, but not limited to, legal and equitable rights of setoff and/or recoupment against the Holders of any Reinstated Claims; provided, however, that the foregoing shall not apply with respect to any Credit Facility Claims or Intercompany Note Claims, each of which is Allowed in full under this Plan without defenses, counterclaims, setoff or recoupment. Additionally, and notwithstanding anything contained herein to the contrary, any Unimpaired Claim in Classes 1, 2, 5 (other than a Claim in Class 5 arising from the rejection of an Executory Contract) and, solely in the event the Closing of the Alternative Transaction occurs, Class 7, not paid in full in Cash by the Debtor on the Effective Date or as soon thereafter as reasonably practicable shall constitute a Reinstated Claim.
- 7.9. <u>Undeliverable or Non-Negotiated Distributions</u>. If any distribution is returned as undeliverable, no further distributions to the applicable Holder shall be made unless and until the Reorganized Debtor is notified in writing of such Holder's then-current address, at which time the undelivered distribution shall be made to such Holder without interest or dividends. undeliverable distributions under the Plan that remain unclaimed for one year after attempted distribution shall indefeasibly revert to the Reorganized Debtor. Upon such reversion, the relevant Allowed Claim (and any Claim on account of missed distributions) shall be automatically discharged and forever barred, notwithstanding any federal or state escheat laws to the contrary. Checks issued on account of Allowed Claims shall be null and void if not negotiated within 180 calendar days from and after the date of issuance thereof. Requests for reissuance of any check must be made directly and in writing to the Reorganized Debtor by the Holder of the relevant Allowed Claim within the 180-calendar-day period. After such date, the relevant Allowed Claim (and any Claim for reissuance of the original check) shall be deemed undeliverable as of the date of the original issuance of the check and shall indefeasibly revert to the Reorganized Debtor in accordance with the terms hereof, notwithstanding any federal or state escheat laws to the contrary.
- 7.10. <u>Claims Paid by Third Parties</u>. To the extent a Holder receives a distribution on account of a Claim and also receives payment from a party that is not the Debtor or the Reorganized Debtor on account of such Claim, such Holder shall, within thirty calendar days of receipt thereof, repay or return the distribution to the Reorganized Debtor, to the extent the Holder's total recovery on account of such Claim from the third party and under the Plan exceeds the amount of the Claim as of the date of any such distribution under the Plan.

ARTICLE VIII: CONFIRMATION AND CONSUMMATION OF THE PLAN

- 8.1. <u>Conditions to Effective Date of Plan Implementing the Alternative Transaction</u>. This Plan shall not become effective, the Effective Date shall not occur, and the Alternative Transaction shall not be consummated unless and until the following conditions shall have been satisfied or waived in accordance with Section 8.2 of this Plan:
- 8.1.1 The Confirmation Order confirming this Plan shall have been entered by the Bankruptcy Court and shall not have been stayed or vacated, and there have been no amendments to the Plan or the Confirmation Order except as permitted by the Plan and the Alternative Plan Sponsor Agreement.
- 8.1.2 All conditions precedent to the consummation of the transactions set forth in Article VI of the Alternative Plan Sponsor Agreement (other than the occurrence of the Effective Date) shall have been satisfied.
- 8.1.3 The Alternative Plan Sponsor Agreement shall not have terminated in accordance with its terms.
- 8.1.4 All authorizations, consents, certifications, approvals, rulings, no-action letters, opinions or other documents or actions required by any law, regulation or order to be received or to occur in order to implement this Plan on the Effective Date shall have been obtained or shall have occurred.
- 8.2. <u>Conditions to Effective Date of Plan Implementing the Original Transaction</u>. If the Alternative Plan Sponsor Agreement shall have terminated in accordance with its terms, this Plan shall not become effective, the Effective Date shall not occur, and the Original Transaction shall not be consummated unless and until the following conditions shall have been satisfied or waived in accordance with <u>Section 8.2</u> of this Plan:
- 8.2.1 The Confirmation Order confirming this Plan shall have been entered by the Bankruptcy Court and shall have become a Final Order, and there have been no amendments to the Plan or the Confirmation Order except as permitted by the Plan and the Original Plan Sponsor Agreement.
- 8.2.2 All conditions precedent to the consummation of the transactions set forth in Article VI of the Original Plan Sponsor Agreement (other than the occurrence of the Effective Date) shall have been satisfied.
- 8.2.3 The Original Plan Sponsor Agreement shall not have terminated in accordance with its terms.
- 8.2.4 All authorizations, consents, certifications, approvals, rulings, no-action letters, opinions or other documents or actions required by any law, regulation or order to be received or to occur in order to implement this Plan on the Effective Date shall have been obtained or shall have occurred.

- 8.3. Waiver of Conditions. Each of the conditions set forth in Section 8.1 of this Plan may be waived in whole or in part with the consent of both the Debtor and the Plan Sponsor under the Alternative Plan Sponsor Agreement, in their respective sole discretion; provided, however, that any such waiver which adversely affects the Plan Sponsor under the Original Plan Sponsor Agreement shall require the consent of the Plan Sponsor under the Original Plan Sponsor Agreement. Each of the conditions set forth in Section 8.2 of this Plan may be waived in whole or in part with the consent of both the Debtor and the Plan Sponsor under the Original Plan Sponsor Agreement, in their respective sole discretion.
- 8.4. <u>Vacatur of Confirmation Order</u>. If the Confirmation Order is vacated, (a) this Plan shall be null and void in all respects; (b) any settlement of Claims or Interests provided for hereby shall be null and void without further order of the Bankruptcy Court; and (c) the Debtor shall be entitled, at any time before or after entry of the order vacating the Confirmation Order, to request appropriate relief from the Bankruptcy Court with respect to the treatment of Executory Contracts.
- 8.5. <u>Notice of Effective Date</u>. The Debtor shall file with the Bankruptcy Court a notice of the occurrence of the Effective Date on the Effective Date or as soon as practicable thereafter.

ARTICLE IX: EFFECT OF PLAN CONFIRMATION

Binding Effect. Until the occurrence of the Confirmation Date of this Plan, the 9.1. Bankruptcy Court's order [Docket No. 127] confirming the Modified Prepackaged Plan of Reorganization for HCR ManorCare, Inc. [Docket No. 114] (the "Original Plan") shall remain in full force and effect and binding upon the Debtor, the Reorganized Debtor, all Holders of Claims against and Interests in the Debtor and each such Holder's respective successors and assigns in accordance with the terms thereof. On the Confirmation Date of this Plan, the Original Plan shall be deemed to be modified, amended and superseded in its entirety by this Plan. On the Effective Date of this Plan, except as otherwise provided in section 1141(d)(3) of the Bankruptcy Code, all provisions of this Plan or the Plan Supplement, including all agreements, instruments and other documents filed in connection with this Plan and executed by the Debtor or the Reorganized Debtor in connection with this Plan or the Plan Supplement, shall be binding upon the Debtor, the Reorganized Debtor, all Holders of Claims against and Interests in the Debtor and each such Holder's respective successors and assigns, whether or not the Claim or Interest of such Holder is Impaired under this Plan and whether or not such Holder has accepted this Plan, and all other parties that are affected in any manner by this Plan. Except as expressly provided otherwise in the Plan, all agreements, instruments and other documents filed in connection with this Plan shall be given full force and effect, and shall bind all parties referred to therein as of the Effective Date, whether or not such agreements are actually issued, delivered or recorded on the Effective Date or thereafter and whether or not a party has actually executed such agreement. For the avoidance of doubt, no provision of the Plan Supplement or any agreement, instrument or other document filed in connection with this Plan and executed by the Debtor or the Reorganized Debtor in connection with this Plan or the Plan Supplement shall render Impaired any Claim that is otherwise Unimpaired under this Plan; provided, however, that the Reorganized Debtor may take any action that it determines may be necessary, advisable or appropriate in connection with the operation of the Reorganized Debtor after the consummation of this Plan (without prejudice to the rights of creditors or other parties under contract or applicable nonbankruptcy law).

9.2. <u>Discharge</u>.

- 9.2.1 <u>Discharge of Claims and Termination of Interests</u>. Except as otherwise provided herein or in the Confirmation Order including with respect to any Claims that are Reinstated under this Plan, all consideration distributed under this Plan shall be in exchange for all Claims and Interests of any nature whatsoever, whether known or unknown, against the Debtor or its Estate, assets, properties or interest in property, and shall constitute a complete satisfaction and settlement of all Claims and Interests other than Reinstated Claims, in each case regardless of whether any property shall have been distributed or retained pursuant to this Plan on account of such Claims and Interests. On the Effective Date, except for Reinstated Claims, the Debtor shall be deemed discharged and released under section 1141(d)(l)(A) of the Bankruptcy Code from any and all Claims and Interests, including, but not limited to, demands and liabilities that arose before the Effective Date, and all debts of the kind specified in sections 502(g), 502(h) or 502(i) of the Bankruptcy Code, Section 510(b) Claims, General Unsecured Claims, and Interests in the Debtor.
- 9.2.2 Discharge Injunction. As of the Effective Date, except as otherwise expressly provided in this Plan or the Confirmation Order, all Entities (other than holders of Reinstated Claims solely in their capacities as such) shall be precluded from asserting against the Debtor or the Reorganized Debtor and their respective assets and property or the Estate, any other or further Claims (other than those Reinstated under this Plan) or Interests, or any other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities of any nature whatsoever, relating to the Debtor or Reorganized Debtor or any of their respective assets and property or the Estate, based upon any act, omission, transaction or other activity of any nature that occurred prior to the Effective Date. In accordance with the foregoing, except as expressly provided in this Plan or the Confirmation Order, the Confirmation Order shall constitute a judicial determination, as of the Effective Date, of the discharge of all non-Reinstated Claims or other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities, pursuant to sections 524 and 1141 of the Bankruptcy Code, and such discharge shall void and extinguish any judgment obtained against the Debtor, the Reorganized Debtor, or its respective assets, property and Estate at any time, to the extent such judgment is related to a discharged Claim, debt, liability or Interest. Except as otherwise specifically provided in this Plan or the Confirmation Order, all Persons or Entities who have held, hold or may hold Claims or Interests that arose prior to the Effective Date and all other parties-in-interest, along with their respective present or former employees, agents, officers, directors, principals, representatives and Affiliates, are permanently enjoined, from and after the Effective Date, from (i) commencing or continuing in any manner any action or other proceeding of any kind with respect to any such Claim (including a Section 510(b) Claim) against or Interest in the Reorganized Debtor or property of the Reorganized Debtor, other than to enforce any right to a distribution pursuant to the Plan, (ii) the enforcement, attachment, collection or recovery by any manner or means of any judgment, award, decree or order against the Reorganized Debtor or property of the Reorganized Debtor, other than to enforce any right to a distribution pursuant to this Plan, (iii) creating, perfecting or enforcing any Lien or encumbrance of any kind against the Reorganized Debtor or against the property or interests in property of the Reorganized Debtor, other than to enforce any right to a distribution pursuant to this Plan or (iv) asserting any right of setoff or subrogation of any kind against any obligation due from the

Reorganized Debtor or against the property or interests in property of the Reorganized Debtor, with respect to any such Claim or Interest. Such injunction shall extend to any successors or assignees of the Reorganized Debtor and their respective properties and interest in properties. For the avoidance of doubt, the provisions of this Section 9.2.2 shall not apply with respect to Claims that are Reinstated under this Plan, including, without limitation, the Credit Facility Claims (if the Closing of the Alternative Transaction does not occur) and the Intercompany Note Claim, as applicable.

- Releases by the Debtor. Except as otherwise expressly provided in this Plan or the Confirmation Order, on the Effective Date, for good and valuable consideration, to the fullest extent permissible under applicable law, each of the Debtor and Reorganized Debtor on its own behalf and as a representative of its respective Estate, shall, and shall be deemed to, completely and forever release, waive, void, extinguish and discharge unconditionally, each and all of the Released Parties of and from any and all Claims and Causes of Action (including, without limitation, Avoidance Actions), any and all other obligations, suits, judgments, damages, debts, rights, remedies, causes of action and liabilities of any nature whatsoever, whether liquidated or unliquidated, fixed or contingent, direct or derivative, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are or may be based in whole or part on any act, omission, transaction, event or other circumstance taking place or existing on or prior to the Effective Date (including prior to the Petition Date) in connection with or related to the Debtor, the Reorganized Debtor, their respective assets and property, and the Estate, the Chapter 11 Case, this Plan, the Plan Supplement or the Disclosure Statement that may be asserted by or on behalf of the Debtor, the Reorganized Debtor or the Estate against any of the Released Parties; provided, however, that nothing in this Section 9.3 shall be construed to release any party from fraud, willful misconduct or gross negligence as determined by a Final Order.
- 9.4. Releases by Certain Holders of Claims. Except as otherwise expressly provided in this Plan or the Confirmation Order, on the Effective Date, for good and valuable consideration, to the fullest extent permissible under applicable law, each Holder of a Claim or an Interest that votes to accept this Plan shall be deemed to have completely and forever released, waived, and discharged unconditionally each of the Released Parties of and from any and all Claims, any and all other obligations, suits, judgments, damages, debts, rights, remedies, causes of action and liabilities of any nature whatsoever (including, without limitation, those arising under the Bankruptcy Code), whether liquidated or unliquidated, fixed or contingent, direct or derivative, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are or may be based in whole or part on any act, omission, transaction, event or other circumstance taking place or existing on or prior to the Effective Date (including prior to the Petition Date) in connection with or related to the Debtor, the Reorganized Debtor or their respective assets and property, and the Estate, the Chapter 11 Case, this Plan, the Plan Supplement, and/or the Disclosure Statement; provided, however, that nothing in this Section 9.4 shall be construed to release (x) any party from fraud, willful misconduct or gross negligence as determined by a Final Order, (v) any Reinstated Claim including, without limitation, the Credit Facility Claims (if the Closing of the Alternative Transaction does not occur) and the Intercompany

Note Claims, as applicable or (z) any obligations under the Plan Sponsor Agreement or the Merger Agreement.

Exculpation. From and after the Effective Date, the Exculpated Fiduciaries and, solely to the extent provided by section 1125(e) of the Bankruptcy Code, the Section 1125(e) Parties, shall neither have nor incur any liability to, or be subject to any right of action by, any Holder of a Claim or an Interest, or any other party in interest, or any of their respective employees, representatives, financial advisors, attorneys, or agents acting in such capacity, or Affiliates, or any of their successors or assigns, for any act or omission in connection with, relating to, or arising out of, the Chapter 11 Case, formulating, negotiating or implementing this Plan and/or previous iterations hereof, the Plan Supplement, the Disclosure Statement and/or previous iterations thereof, the Alternative Plan Sponsor Agreement, the Amendment to Original Plan Sponsor Agreement, the Original Plan Sponsor Agreement, the Restructuring Support Agreement, the Alternative Master Lease or the Original Master Lease (as applicable), the solicitation of acceptances of this Plan and/or previous iterations hereof, the pursuit of Confirmation of this Plan, the Confirmation of this Plan, the consummation of this Plan, the administration of this Plan, the property to be distributed under this Plan, the consummation of the transactions contemplated by the Alternative Plan Sponsor Agreement, the Original Plan Sponsor Agreement, or any other act taken or omitted to be taken in connection with or in contemplation of the Chapter 11 Case or implementation of this Plan; provided, however, that this Section 9.5 shall not apply to release (x) obligations under this Plan, and obligations under the Alternative Plan Sponsor Agreement, the Amendment to Original Plan Sponsor Agreement, the Original Plan Sponsor Agreement, the Restructuring Support Agreement, the Alternative Master Lease or the Original Master Lease (as applicable), the Merger Agreement and the contracts, instruments, releases, agreements, and documents delivered, Reinstated or assumed under this Plan (including, without limitation, the Credit Facility (if the Closing of the Alternative Transaction does not occur) and the Intercompany Note, as applicable), and (y) any Claims or Causes of Action arising out of fraud, willful misconduct or gross negligence as determined by a Final Order.

Any of the Exculpated Parties shall be entitled to rely, in all respects, upon the reasonable and informed advice of counsel with respect to their duties and responsibilities under this Plan.

9.6. <u>Injunctions Related to Exculpation and Releases</u>. (a) Except as expressly provided in this Plan or the Confirmation Order, as of the Effective Date, all Persons and Entities that hold, have held, or may hold a Claim or any other obligation, suit, judgment, damages, debt, right, remedy, Cause of Action or liability of any nature whatsoever, of the types described in <u>Section 9.5</u> of this Plan and relating to the Debtor, the Reorganized Debtor or any of their respective assets and property and/or the Estate, are, and shall be, permanently, forever and completely stayed, restrained, prohibited, barred and enjoined from taking any of the following actions against any Exculpated Party or its property on account of such released liabilities, whether directly or indirectly, derivatively or otherwise, on account of or based on the subject matter of such discharged Claims or other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities: (i) commencing, conducting or continuing in any manner, directly or indirectly, any suit,

action or other proceeding (including, without limitation, any judicial, arbitral, administrative or other proceeding) in any forum; (ii) enforcing, attaching (including, without limitation, any prejudgment attachment), collecting, or in any way seeking to recover any judgment, award, decree, or other order; (iii) creating, perfecting or in any way enforcing in any matter, directly or indirectly, any Lien; (iv) setting off, seeking reimbursement or contributions from, or subrogation against, or otherwise recouping in any manner, directly or indirectly, any amount against any liability or obligation that is discharged under Section 9.2 of this Plan; and/or (v) commencing or continuing in any manner any judicial, arbitration or administrative proceeding in any forum, that does not comply with or is inconsistent with the provisions of this Plan or the Confirmation Order.

- Except as expressly provided in this Plan or the Confirmation Order, as of the Effective Date, all Persons and Entities that hold, have held, or may hold a Claim or any other obligation, suit, judgment, damages, debt, right, remedy, Cause of Action or liability of any nature whatsoever, of the types described in Section 9.4 of this Plan and relating to the Debtor, the Reorganized Debtor or any of their respective assets and property and/or the Estate, the Chapter 11 Case, this Plan, the Plan Supplement and/or the Disclosure Statement are, and shall be, permanently, forever and completely stayed, restrained, prohibited, barred and enjoined from taking any of the following actions against any Released Party or its property on account of such released liabilities, whether directly or indirectly, derivatively or otherwise, on account of or based on the subject matter of such discharged Claims or other obligations, suits, judgments, damages, debts, rights, remedies, causes of action or liabilities: (i) commencing, conducting or continuing in any manner, directly or indirectly, any suit, action or other proceeding (including, without limitation, any judicial, arbitral, administrative or other proceeding) in any forum; (ii) enforcing, attaching (including, without limitation, any prejudgment attachment), collecting, or in any way seeking to recover any judgment, award, decree, or other order; (iii) creating, perfecting or in any way enforcing in any matter, directly or indirectly, any Lien; (iv) setting off, seeking reimbursement or contributions from, or subrogation against, or otherwise recouping in any manner, directly or indirectly, any amount against any liability or obligation that is discharged under Section 9.2 of this Plan; and/or (v) commencing or continuing in any manner any judicial, arbitration or administrative proceeding in any forum, that does not comply with or is inconsistent with the provisions of this Plan or the Confirmation Order.
- 9.7. Survival of Indemnification and Exculpation Obligations. The obligations of the Debtor to indemnify and exculpate any past and present directors, officers, agents, employees and representatives who provided services to the Debtor prior to or after the Petition Date, pursuant to certificates or articles of incorporation, by-laws, contracts and/or applicable statutes, in respect of all actions, suits and proceedings against any of such officers, directors, agents, employees and representatives, based upon any act or omission related to service with, for or on behalf of the Debtor, shall not be discharged or Impaired by Confirmation or consummation of this Plan and shall be assumed by the Reorganized Debtor. For the avoidance of doubt, this Section 9.7 affects only the obligations of the Debtor and Reorganized Debtor with respect to any indemnity or exculpation owed to or for the benefit of past and present directors, officers, agents, employees and representatives of the Debtor, and shall have no effect on nor in any way discharge or reduce, in whole or in part, any obligation of any other Person, including any provider of director and officer insurance, owed to or for the benefit of such past and present directors, officers, agents, employees

and representatives of the Debtor. For further avoidance of doubt, the provisions regarding indemnification, exculpation and directors' and officers' insurance contained in section 4.7 of the Plan Sponsor Agreement shall be deemed incorporated into this Plan as if set forth fully herein.

- 9.8. <u>Term of Bankruptcy Injunction or Stays</u>. All injunctions or stays provided for in the Chapter 11 Case under section 105 or section 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the Effective Date.
- 9.9. <u>Liability to Governmental Units</u>. Nothing in the Confirmation Order or the Plan discharges, releases, resolves, precludes, exculpates, or enjoins: (i) any liability to any Governmental Unit that is not a Claim; (ii) any Claim of a Governmental Unit arising on or after the Confirmation Date; (iii) any police or regulatory liability to a Governmental Unit to the extent of such entity's liability under non-bankruptcy law on account of its status as the owner or operator of property after the Confirmation Date; or (iv) any liability to a Governmental Unit on the part of any Person other than the Debtor or Reorganized Debtor. For the avoidance of doubt, the foregoing shall not limit the scope of discharge of all Claims and Interests arising prior to the Effective Date under sections 524 and 1141 of the Bankruptcy Code, or limit the Debtor's or Reorganized Debtor's rights under section 525 of the Bankruptcy Code. Nothing in the Confirmation Order or this Plan shall affect any setoff or recoupment rights of any Governmental Unit.
- 9.10. <u>Receivership Complaint</u>. The Receivership Complaint shall be deemed dismissed with prejudice as of the Effective Date. QCP, with the reasonable cooperation of Debtor, shall cause the Receivership Complaint to be withdrawn from the relevant court docket as soon as practicable after the Effective Date.

ARTICLE X: RETENTION OF JURISDICTION

- 10.1. Retention of Jurisdiction Pursuant to sections 105(c) and 1142 of the Bankruptcy Code and notwithstanding entry of the Confirmation Order and the occurrence of the Effective Date, the Bankruptcy Court shall retain jurisdiction over all matters arising out of, and related to, the Chapter 11 Case and this Plan to the fullest extent permitted by law, including, among other things, jurisdiction to:allow, disallow, determine, liquidate, classify, estimate or establish the priority, secured or unsecured status, or amount of any Claim or Interest, including the resolution of any request for payment of any Administrative Expense Claim or Priority Tax Claim, and the resolution of any objections to the secured or unsecured status, allowance, priority or amount of Claims or Interests;
- (b) resolve any matters related to the assumption or rejection of any Executory Contract to which the Debtor is a party or with respect to which the Debtor or Reorganized Debtor may be liable and to hear, determine, and, if necessary, liquidate any Claims arising therefrom;
- (c) ensure that distributions to Holders of Allowed Claims are accomplished pursuant to the provisions of this Plan and adjudicate any and all disputes from, or relating to distributions under, the Plan;

- (d) decide or resolve any motions, adversary proceedings, contested or litigated matters and any other matters and grant or deny any applications involving the Debtor that may be pending on the Effective Date;
- (e) enter such orders as may be necessary or appropriate to implement or consummate the provisions of this Plan, the Plan Sponsor Agreement, the Restructuring Support Agreement, and all contracts, instruments, releases and other agreements or documents created in connection with this Plan, the Disclosure Statement, the Plan Supplement or the Confirmation Order, and issue injunctions, enter and implement other orders, or take such other actions as may be necessary or appropriate to restrain interference by any entity with consummation, implementation or enforcement of this Plan or the Confirmation Order;
- (f) resolve any cases, controversies, suits or disputes that may arise in connection with the consummation, interpretation, or enforcement of this Plan, the Plan Sponsor Agreement, the Restructuring Support Agreement, or any contract, instrument, release or other agreement or document that is executed or created pursuant to this Plan, or any Entity's rights arising from or obligations incurred in connection with this Plan or such documents, including hearing and determining disputes, cases, or controversies arising in connection with the interpretation, implementation or enforcement of the Plan, Plan Sponsor Agreement, Restructuring Support Agreement or the Confirmation Order;
- (g) enter and enforce any order for the sale of property pursuant to sections 363, 1123, or 1146(a) of the Bankruptcy Code;
- (h) adjudicate, decide or resolve any and all disputes as to the ownership of any Claim or Interest;
- (i) approve any modification of this Plan before or after the Effective Date pursuant to section 1127 of the Bankruptcy Code or approve any modification of the Disclosure Statement, the Confirmation Order or any contract, instrument, release or other agreement or document created in connection with this Plan, the Disclosure Statement, the Plan Supplement or the Confirmation Order, or remedy any defect or omission or reconcile any inconsistency in any Bankruptcy Court order, this Plan, the Disclosure Statement, the Plan Supplement, the Confirmation Order or any contract, instrument, release or other agreement or document created in connection with this Plan, the Disclosure Statement, the Plan Supplement or the Confirmation Order, in such manner as may be necessary or appropriate to consummate this Plan;
- (j) hear and determine all disputes involving the existence, nature or scope of the Debtor's discharge;
- (k) subject to <u>Section 11.1</u>, hear and determine all applications for compensation and reimbursement of expenses of Professionals under this Plan or under sections 330, 331, 363, 503(b), 1103 and 1129(a)(9) of the Bankruptcy Code, which shall be payable by the Debtor only upon allowance thereof pursuant to an order of the Bankruptcy Court;
- (l) hear and determine Causes of Action by or on behalf of the Debtor or the Reorganized Debtor;

- (m) hear and determine matters concerning state, local and federal taxes in accordance with sections 346, 505 and 1146 of the Bankruptcy Code;
- (n) hear and determine any issues arising under, or violations of, section 525 of the Bankruptcy Code;
- (o) enter and implement such orders as are necessary or appropriate if the Confirmation Order is for any reason or in any respect modified, stayed, reversed, revoked or vacated, or if distributions pursuant to this Plan are enjoined or stayed;
- (p) determine any other matters that may arise in connection with or relate to this Plan, the Disclosure Statement, the Plan Supplement, the Confirmation Order, the Plan Sponsor Agreement, the Restructuring Support Agreement, or any contract, instrument, release, or other agreement, or document created in connection with this Plan, the Disclosure Statement, the Plan Supplement or the Confirmation Order;
- (q) enforce all orders, judgments, injunctions, releases, exculpations, indemnifications and rulings entered in connection with the Chapter 11 Case;
- (r) hear and determine all matters related to the property of the Estate from and after the Confirmation Date;
- (s) hear and determine such other matters as may be provided in the Confirmation Order or as may be authorized under the Bankruptcy Code;
- (t) enter a final order or decree concluding or closing the Chapter 11 Case; and
 - (u) hear any other matter not inconsistent with the Bankruptcy Code;

provided, however, that the Bankruptcy Court shall not retain jurisdiction after the Effective Date over disputes concerning documents contained in the Plan Supplement or in respect of the Credit Facility or Intercompany Note that have a jurisdictional, forum selection or dispute resolution clause that refers disputes to a different court, including, for the avoidance of doubt, the Alternative Master Lease or the Original Master Lease (as applicable), and any disputes concerning documents contained in the Plan Supplement that contain such clauses shall be governed in accordance with the provisions of such documents.

ARTICLE XI: MISCELLANEOUS PROVISIONS

11.1. <u>Post-Effective Date Retention of Professionals</u>. On the Effective Date, any requirement that Professionals employed by the Reorganized Debtor comply with sections 327 through 331 of the Bankruptcy Code in seeking retention or compensation for services rendered after such date will terminate, and the Reorganized Debtor will be authorized to employ and compensate professionals in the ordinary course of business and without the need for application to or approval by the Bankruptcy Court, including professionals previously employed by the Debtor.

- 11.2. <u>Effectuating Documents and Further Transactions</u>. Each of the Debtor and the Reorganized Debtor is authorized to execute, deliver, file or record such contracts, instruments, certificates, notes, releases and other agreements or documents and take such actions as may be necessary or appropriate to effectuate, implement and further evidence the terms and conditions of this Plan, the Plan Sponsor Agreement, and the New Common Stock issued under or in connection with this Plan.
- 11.3. Exemption from Transfer Taxes. Pursuant to section 1146(a) of the Bankruptcy Code, (a) the issuance, transfer or exchange of equity securities under this Plan; (b) the creation of any mortgage, deed of trust, lien, pledge or other security interest; or (c) the making or delivery of any deed or other instrument of transfer under this Plan, including, without limitation, merger agreements, agreements of consolidation, restructuring, disposition, liquidation or dissolution, deeds, bills of sale, and transfers of tangible property, will not be subject to any document recording tax, stamp tax, conveyance fee, intangibles or similar tax, mortgage tax, stamp act, real estate transfer tax, mortgage recording tax, Uniform Commercial Code filing or recording fee or other similar tax or governmental assessment in the United States. The Confirmation Order shall direct the appropriate federal, state or local governmental officials or agents to forgo the collection of any such tax or governmental assessment and to accept for filing and recordation any of the foregoing instruments or other documents without the payment of any such tax or governmental assessment.
- 11.4. Payment of Statutory Fees. All fees due and payable pursuant to section 1930(a)(6) of Title 28 of the United States Code ("Quarterly Fees") prior to the Effective Date shall be paid by the Debtor on the Effective Date. After the Effective Date, the Reorganized Debtor shall be liable for any and all Quarterly Fees when they are due and payable after the Effective Date. The Debtor shall file all Quarterly Reports due prior to the Effective Date when they become due, in a form reasonably acceptable to the United States Trustee. After the Effective Date, the Reorganized Debtor shall file with the Bankruptcy Court Quarterly Reports in a form reasonably acceptable to the United States Trustee, which reports shall include a separate schedule of disbursements made by the Reorganized Debtor during the applicable period, attested to by an authorized representative of the Reorganized Debtor. The Reorganized Debtor shall remain obligated to pay Quarterly Fees to the Office of the U.S. Trustee until the earliest of the Debtor's case being closed, dismissed or converted to a case under Chapter 7 of the Bankruptcy Code.
- 11.5. Amendment or Modification of this Plan. Subject to section 1127 of the Bankruptcy Code, the Debtor may alter, amend or modify this Plan or the Exhibits at any time prior to or after the Confirmation Date but prior to the substantial consummation of this Plan, solely in accordance with the Plan Sponsor Agreement; provided, that sections of this Plan pertaining to the the Credit Facility Claims, the Intercompany Note or Intercompany Claims held by obligors under the Credit Facility shall not be altered, amended or modified other than in accordance with the Credit Facility. Any Holder of a Claim that has accepted this Plan shall be deemed to have accepted this Plan, as altered, amended or modified, if the proposed alteration, amendment or modification is made in accordance with the Plan Sponsor Agreement and does not materially and adversely change the treatment of the Claim of such Holder.
- 11.6. <u>Severability of Plan Provisions</u>. If, prior to the Confirmation Date, any term or provision of this Plan is determined by the Bankruptcy Court to be invalid, void or unenforceable, the Bankruptcy Court will have the power to alter and interpret such term or provision to make it

valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision will then be applicable as altered or interpreted, provided that any such holding, alteration or interpretation complies and is consistent with the Plan Sponsor Agreement and does not adversely impact the holders of the Credit Facility Claims. Notwithstanding any such holding, alteration or interpretation, the remainder of the terms and provisions of this Plan will remain in full force and effect and will in no way be affected, impaired or invalidated by such holding, alteration, or interpretation. The Confirmation Order will constitute a judicial determination and will provide that each term and provision of this Plan, as it may have been altered or interpreted in accordance with the foregoing, is valid and enforceable pursuant to its terms.

- 11.7. Successors and Assigns. This Plan shall be binding upon and inure to the benefit of the Debtor, and its successors and assigns, including, without limitation, the Reorganized Debtor. The rights, benefits and obligations of any entity named or referred to in this Plan shall be binding on, and shall inure to the benefit of, any heir, executor, administrator, successor or assign of such entity.
- 11.8. Non-Consummation. If consummation of this Plan does not occur, then (a) this Plan shall be null and void in all respects, (b) any settlement or compromise embodied in this Plan (including the fixing or limiting to an amount certain any Claim or Interest or Class of Claims or Interests), assumption or rejection of Executory Contracts or leases affected by this Plan, and any document or agreement executed pursuant to this Plan shall be deemed null and void and (c) nothing contained in this Plan shall (i) constitute a waiver or release of any Claims by or against, or any Interests in, the Debtor or any other Person, (ii) prejudice in any manner the rights of the Debtor, the Plan Sponsor under the Alternative Plan Sponsor Agreement, the Plan Sponsor under the Original Plan Sponsor Agreement, or any other Person or (iii) constitute an admission of any sort by the Debtor, the Plan Sponsor under the Alternative Plan Sponsor Agreement, the Plan Sponsor under the Original Plan Sponsor Agreement, or any other Person.
- 11.9. Notice. All notices, requests and demands to or upon the Debtor or the Reorganized Debtor to be effective shall be in writing and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when actually delivered or, in the case of notice by facsimile transmission, when received and telephonically confirmed, addressed as follows:

HCR ManorCare, Inc. 333 N. Summit St. Toledo, Ohio 43604

Attn: John R. Castellano, Chief Restructuring Officer

with a copy to:

SIDLEY AUSTIN LLP One South Dearborn Street Chicago, Illinois 60603 Telephone: (312) 853-7000

Facsimile: (312) 853-7036

Attn: Larry J. Nyhan Dennis M. Twomey

-and-

YOUNG CONAWAY STARGATT & TAYLOR, LLP Rodney Square 1000 North King Street Wilmington, Delaware 19801 Telephone: (302) 571-6600

Facsimile: (302) 571-1253 Attn: Edmon L. Morton

Attorneys for the Debtor and Debtor in Possession

- 11.10. <u>Governing Law</u>. Subject to the provisions of any contract, certificates or articles of incorporation, by-laws, instruments, releases, or other agreements or documents entered into in connection with this Plan, and subject further to <u>Section 10.1</u> of this Plan, the rights and obligations arising under this Plan shall be governed by, and construed and enforced in accordance with (i) the Bankruptcy Code, the Bankruptcy Rules or other federal law to the extent applicable and (ii) if none of such law is applicable, the laws of the State of Delaware, without giving effect to the principles of conflicts of law of such jurisdiction.
- 11.11. <u>Tax Reporting and Compliance</u>. The Reorganized Debtor is hereby authorized, on behalf of the Debtor, to request an expedited determination under section 505 of the Bankruptcy Code of the tax liability of the Debtor for all taxable periods ending after the Petition Date through, and including, the Effective Date.
- 11.12. <u>Exhibits</u>. All Exhibits to this Plan are incorporated into and are a part of this Plan as if set forth in full herein.
- 11.13. Filing of Additional Documents. On or before substantial consummation of this Plan, the Reorganized Debtor and the Debtor shall, as applicable, file such agreements and other documents as may be necessary or appropriate to effectuate and evidence further the terms and conditions of this Plan.
- 11.14. <u>Plan Documents</u>. The Plan and the Plan Supplement, including all Exhibits, supplements, appendices and schedules thereto, and any modifications to any of the foregoing, shall be in form and substance acceptable to the Debtor, subject to the Plan Sponsor Agreement.
- 11.15. <u>Reservation of Rights</u>. Except as expressly set forth herein, this Plan shall have no force and effect unless the Bankruptcy Court has entered the Confirmation Order. The filing of this Plan, any statement or provision contained in this Plan, or the taking of any action by the Debtor with respect to this Plan shall not be and shall not be deemed to be an admission or waiver of any rights of the Debtor, the Plan Sponsor under the Alternative Plan Sponsor Agreement, the Plan Sponsor under the Original Plan Sponsor Agreement, the Credit Facility Agent, the Credit Facility Lenders or any other Person with respect to Claims against and Interests in the Debtor.

Dated: May 18, 2018 Wilmington, Delaware Respectfully submitted,

HCR ManorCare, Inc.

/s/ John R. Castellano

John R. Castellano Chief Restructuring Officer

SIDLEY AUSTIN LLP

Larry J. Nyhan Dennis M. Twomey William A. Evanoff Allison Ross Stromberg Matthew E. Linder One South Dearborn Street Chicago, Illinois 60603 Telephone: (312) 853-7000 Facsimile: (312) 853-7036

-and-

YOUNG CONAWAY STARGATT & TAYLOR, LLP

Robert S. Brady (No. 2847) Edmon L. Morton (No. 3856) Justin H. Rucki (No. 5304) Tara C. Pakrouh (No. 6192) Rodney Square 1000 North King Street Wilmington, Delaware 19801

Telephone: (302) 571-6600 Facsimile: (302) 571-1253

Attorneys for the Debtor and Debtor in Possession

Exhibit A

Alternative Master Lease Term Sheet

Master Lease Terms

Facilities: All ManorCare operated facilities that are currently owned by QCP (165

SNFs, 54 ALFs)

Landlord: Joint Venture

Tenant: HCR III. ProMedica will provide a corporate guaranty of the obligations of

Tenant

Term: 15 years with three 5-year extensions

Standard termination provisions will exist, the terms and conditions of which

will be set forth in the definitive documentation

Initial Rent: \$179,000,000

Annual Y1 Rent: \$179 million

Escalators: Y2 Rent: \$179 million + 1.375% = \$181,461,250.

Y3 Rent: Y2 Rent + 2.75%Thereafter: Y3 Rent + 2.75%

Renewal Term

Fair market value

Rent:

Exhibit B

Alternative Plan Sponsor Agreement

ALTERNATIVE PLAN SPONSOR AGREEMENT

among

HCR MANORCARE, INC.,

QUALITY CARE PROPERTIES, INC.,

PROMEDICA HEALTH SYSTEM, INC.

SUBURBAN HEALTHCO, INC.

MEERKAT I LLC

and

THE OTHER LESSORS IDENTIFIED HEREIN

Dated as of April 25, 2018

TABLE OF CONTENTS

	TABLE OF CONTENTS	Page
	ARTICLE I	
	The Transactions	
1.1	Acquisition of Reorganized Company Stock	4
1.2	New Master Lease and Guaranty	5
1.3	Articles of Incorporation and By-Laws of the Reorganized	
	Debtor	
1.4	Closing	
1.5	Severance and Other Payments	
1.6	Deliveries by Debtor	
1.7	Deliveries by ProMedica	
1.8	Deliveries by HCR III and the Lessors	8
	ARTICLE II	
	Representations and Warranties	
2.1	Representations and Warranties of the Debtor	8
2.2	Representations and Warranties of ProMedica	
2.3	Representations and Warranties of QCP	
	ARTICLE III	
	Bankruptcy Matters	
3.1	Confirmation Order	25
3.2	Chapter 11 Obligations	
3.3	Case Financing	
3.4	Release	28
3.5	Release of Non-Debtor Subsidiary Claims	28
	ARTICLE IV	
	Covenants	
4.1	Interim Operations	20
4.1	Interim Operations	
4.3	Access and Records	
4.4	Publicity	
4.5	Employee Matters	
4.6	Expenses	
4.7	Indemnification; Directors' and Officers' Insurance	

4.8	Confidentiality	40
4.9	Financing	40
4.10	Acquisition Proposals.	43
	ARTICLE V	
	Special Covenants	
5.1	Corporate Governance	
5.2	Rent and Forbearance	
5.3	Business Operations	
5.4	QCP Support	
5.5	Current Rent Limitation	48
	ARTICLE VI	
	Conditions	
6.1	Conditions to Each Party's Obligations	49
6.2	Conditions to Obligations of ProMedica	
6.3	Conditions to Obligation of the Debtor	50
	ARTICLE VII	
	Termination	
7.1	Automatic Termination	50
7.2	Termination by Mutual Consent	50
7.3	Termination by ProMedica, QCP, or the Debtor	51
7.4	Termination by the Debtor or QCP	
7.5	Termination by ProMedica or QCP	
7.6	Termination by QCP	53
	ARTICLE VIII	
	Miscellaneous and General	
8.1	No Survival; Effect of Termination and Abandonment	53
8.2	Modification or Amendment	
8.3	Waiver of Conditions	54
8.4	Counterparts	54
8.5	GOVERNING LAW AND VENUE; WAIVER OF JURY	
	TRIAL; SPECIFIC PERFORMANCE	
8.6	Notices	
8.7	Entire Agreement	
8.8	No Third Party Beneficiaries; No Recourse	58

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 122 of 441

8.9	Obliga	tions of ProMedica and of the Debtor	58
		er Taxes	
8.11	Definit	tions	59
8.12	Severa	bility	59
8.13	Interpr	retation; Construction	59
		ment	
8.15	Certair	1 Waivers	60
8.16	No Sol	licitation	61
8.17	Settlen	nent Discussions	61
Annex	A	Defined Terms	
Exhibit	t 1	Amended Plan	
Exhibit	t 2	New Master Lease – Basic Terms	
Exhibit	t 3	Alternative Restructuring Support Agreement	
Exhibit	t 4	Form of Former CEO Settlement Agreement	
Exhibit	t 5	Form of Separation Agreement	

ALTERNATIVE PLAN SPONSOR AGREEMENT

This PLAN SPONSOR AGREEMENT, dated as of April 25, 2018 (hereinafter called this "<u>Agreement</u>"), is among HCR ManorCare, Inc., a Delaware corporation (the "<u>Debtor</u>"), Quality Care Properties, Inc., a Maryland corporation ("<u>QCP</u>"), ProMedica Health System, Inc., an Ohio non-profit corporation ("<u>ProMedica Parent</u>"), Suburban HealthCo, Inc., an Ohio non-profit corporation and a wholly-owned, indirect subsidiary of ProMedica Parent ("<u>Purchaser</u>" and, together with ProMedica Parent, "<u>ProMedica</u>"), and Meerkat I LLC, a Delaware limited liability company (the "<u>JV</u>" and, together with any wholly-owned Subsidiary of the JV so designated by the JV, the "<u>Lessors</u>"). The Debtor, QCP, ProMedica, and the Lessors are each referred to in this Agreement as a "<u>party</u>" and together, the "<u>parties</u>".

RECITALS

WHEREAS, the Debtor and its Subsidiaries are engaged in the business of providing skilled nursing, hospice and other ancillary services (the "**Business**");

WHEREAS, pursuant to the Master Lease and Security Agreement, dated as of April 7, 2011, by and among certain lessors (the "<u>Original Lessors</u>") and HCR III Healthcare, LLC ("<u>HCR III</u>"), as Lessee (as amended from time to time, the "<u>Master Lease</u>"), HCR III leases certain skilled nursing and assisted living facilities (each, a "<u>Leased Facility</u>" and, collectively, the "<u>Leased Facilities</u>") from the Original Lessors;

WHEREAS, the Debtor guarantees the obligations of HCR III under the Master Lease pursuant to that certain Guaranty of Obligations, effective as of February 11, 2013 (as amended or modified from time to time, the "Guaranty");

WHEREAS, HCR III subleases the Leased Facilities to certain of its direct and indirect Subsidiaries and Affiliates (each, a "<u>Sublease OpCo</u>" and, collectively, the "<u>Sublease OpCos</u>") pursuant to certain subleases (the "<u>Subleases</u>"), and the Sublease OpCos operate the Leased Facilities;

WHEREAS, the Original Lessors, HCR III and the Sublease OpCos have entered into certain Agreements Regarding Subleases made and entered as of April 7, 2011, October 23, 2015, February 16, 2016 and March 14, 2016 (each, as amended or modified from time to time, an "Agreement Regarding Subleases" and, collectively, the "Agreements Regarding Subleases");

WHEREAS, QCP, the Debtor and certain of their respective Subsidiaries entered into a Forbearance Agreement, dated as of April 5, 2017 (the "<u>Prior Forbearance Agreement</u>"), pursuant to which, among other things, QCP agreed to defer certain of the Debtor's rent and other obligations under the Master Lease, which such Prior Forbearance Agreement terminated according to its terms on July 5, 2017;

WHEREAS, on August 17, 2017, QCP filed a complaint against the Debtor, HCR III and "John Does 1–50" in Los Angeles County Superior Court, styled as *Quality Care Properties, Inc. and HCP Properties, LP* v. *HCR III Healthcare, LLC et al.*, Los Angeles County Super. Ct. No. BC672837, seeking, among other relief, the appointment of a receiver for HCR III (the "Receivership Proceeding");

WHEREAS, QCP, the Debtor and certain of their respective Subsidiaries entered into that certain Forbearance Agreement and Amendment to Master Lease and Security Agreement, dated as of December 22, 2017 (the "Second Forbearance Agreement"), pursuant to which, among other things, the Original Lessors and HCR III agreed to amend the Master Lease to defer certain of the Debtor's rent and other obligations under the Master Lease and to defer the Debtor's response on the Receivership Proceeding;

WHEREAS, on March 2, 2018, the Debtor, QCP, HCP Mezzanine Lender, LP and the Original Lessors entered into that certain Plan Sponsor Agreement (the "Original Plan Sponsor Agreement");

WHEREAS, prior to the date of the Original Plan Sponsor Agreement, HCR III paid to the Original Lessors the Reduced Cash Rent of \$23,500,000 per month, as more fully described in the Master Lease (the "Reduced Cash Rent");

WHEREAS, as of the date of the Original Plan Sponsor Agreement (and taking into account payments made by HCR III to the Original Lessors on such date), the Original Lessors were owed approximately \$180,571,590 in unpaid Rent currently due and owing under the Master Lease, including \$14,000,000 in unpaid Rent due and owing since the date of the Second Forbearance Agreement until the date of the Original Plan Sponsor Agreement ("Reduced Cash Rent Past Due Balance");

WHEREAS, in addition to the Reduced Cash Rent Past Due Balance, as of the date of the Original Plan Sponsor Agreement, the Original Lessors were owed approximately \$265,225,000 in unpaid Deferred Rent Obligations (as defined in the Master Lease) currently due and owing under the Master Lease;

WHEREAS, on March 4, 2018 (the "<u>Petition Date</u>") the Debtor filed a voluntary petition for relief under chapter 11 of the Bankruptcy Code in the United States Bankruptcy Court for the District of Delaware (the "<u>Bankruptcy Court</u>" and such proceedings, collectively, the "<u>Bankruptcy Case</u>");

WHEREAS, on the Petition Date the Debtor filed the Prepackaged Plan of Reorganization For HCR ManorCare, Inc. (as modified by the Modified Prepackaged Plan of Reorganization For HCR ManorCare, Inc. filed by the Debtors on April 10, 2018 the "Original Plan") and the Disclosure Statement for the Prepackaged Plan of Reorganization for HCR ManorCare, Inc. (as amended, supplemented, or otherwise modified the "Disclosure Statement"), and the Original Plan provided for the transfer of all of the issued and outstanding capital stock and rights to purchase or otherwise acquire capital stock of the Debtor to a wholly owned subsidiary of QCP;

WHEREAS, on April 13, 2018, the Bankruptcy Court entered a combined order approving the Disclosure Statement with respect to the Original Plan and confirming the Original Plan (the "Original Confirmation Order");

WHEREAS, the Original Plan has not yet become effective and the parties desire to amend the Original Plan as provided in the *First Amended Chapter 11 Plan of Reorganization For HCR ManorCare, Inc.* (the "Amended Plan"), a copy of which is attached as Exhibit 1 hereto;

WHEREAS, pursuant to that certain Agreement and Plan of Merger (the "Merger Agreement") dated as of the date hereof, QCP and certain of its Subsidiaries will be merged with and into Potomac Acquisition LLC (the "Acquirer") with Acquirer surviving pursuant to the transactions set forth in the Merger Agreement (the "Merger") and, thereafter, Acquirer will cause all of the property leased to HCR III pursuant to the Master Lease to be transferred to the Lessors;

WHEREAS, the Effective Date (as defined in the Amended Plan) of the Amended Plan will occur concurrently with the closing of the Merger and, in connection with the occurrence of the Effective Date, the transactions contemplated by this Agreement will be consummated, including without limitation, the Plan Acquisition (as defined below), and execution of that certain new master lease by and among the Lessors and HCR III (the "New Master Lease"), the form of which New Master Lease will be agreed by Debtor and ProMedica and filed in the Plan Supplement (as defined in the Amended Plan) and the basic terms of which are attached hereto as Exhibit 2, which lease shall be guaranteed by ProMedica and shall supersede the Master Lease;

WHEREAS, if the transactions contemplated by Merger Agreement and this Plan Sponsor Agreement are not consummated, the Amended Plan provides that the transactions contemplated by the Original Plan Sponsor Agreement shall be implemented;

WHEREAS, the Debtor, with assistance from its professional advisors, has reviewed alternatives to the Transactions contemplated by this Agreement, including transfer of the Business to QCP, on the terms reflected in the Original Plan Sponsor Agreement and the Original Confirmation Order, and has determined that the Amended Plan, and implementation of the Transactions contemplated therein and herein, are in the best interests of the Debtor, its creditors and the bankruptcy estate;

WHEREAS, in connection with the Original Plan, the Debtor entered into that certain restructuring support agreement, dated March 2, 2018 (the "Original Restructuring Support Agreement") with each of (i) Carlyle MC Partners, L.P., a Delaware limited partnership, Carlyle Partners V-A MC, L.P., a Delaware limited partnership, Carlyle Partners V MC, L.P., a Delaware limited partnership, CP V Coinvestment A, L.P., a Delaware limited partnership, and CP V Coinvestment B, L.P., a Delaware limited partnership (collectively, the "Carlyle Holders") and (ii) MC Operations Investments, LLC;

WHEREAS, in connection with the Amended Plan, the Debtor has entered into that certain restructuring support agreement, dated on or prior to the date hereof (the "<u>Alternative</u> <u>Restructuring Support Agreement</u>"); with each of (i) the Carlyle Holders, (ii) MC Operations Investments, LLC, and (iii) ProMedica Parent;

WHEREAS, the Debtor intends to, with the consent of ProMedica and QCP, promptly offer to Paul Ormond (the "<u>Former CEO</u>") a settlement agreement in the form attached hereto as <u>Exhibit 4</u> (the "<u>Former CEO Settlement Agreement</u>"), by and among the Debtor and Paul Ormond, to be effective upon the Closing Date subject to the execution and delivery of the Former CEO Settlement Agreement by Mr. Ormond within 60 days of the date of this Agreement;

WHEREAS, the Debtor entered into a Credit Agreement, dated as of July 17, 2017 (as may be amended, modified or supplemented from time to time, the "Centerbridge Facility"),

by and among HCR Home Health Care and Hospice, LLC, as borrower, HCR Manorcare Inc., HCR Manorcare Operations II, LLC, HCR Manorcare Heartland, LLC, Manor Care, Inc., HCR II Healthcare, LLC, and HCR Healthcare, LLC, each as Holdco, the lender parties thereto, and RD Credit, LLC, as administrative agent and collateral agent; and

WHEREAS, the Debtor, QCP, the Lessors and ProMedica desire to make certain representations, warranties, covenants and agreements in connection with this Agreement.

NOW, THEREFORE, in consideration of the premises, and of the representations, warranties, covenants and agreements contained herein, the parties hereto agree as follows:

ARTICLE I

The Transactions

- 1.1 <u>Acquisition of Reorganized Company Stock.</u>
- (a) On the Closing Date, Purchaser, in exchange for the ProMedica Plan Contribution (as defined below) and the other obligations to be paid or undertaken by the ProMedica as provided herein, shall acquire and accept from the Debtor, and the Debtor shall issue, transfer and deliver (or cause HCR III to deliver as provided in the Implementation Memo (as defined below)) to Purchaser 1,000,000 newly issued shares of the common stock of the Debtor, par value \$0.01 per share (the "Reorganized Company Stock"), which, pursuant to and in accordance with the Amended Plan and the Confirmation Order, shall constitute all of the issued and outstanding capital stock and rights to purchase or otherwise acquire capital stock of the Debtor (such transaction, the "Plan Acquisition"), free and clear of any Liens, other than those created by ProMedica, or the Lessors solely pursuant to the terms of the New Master Lease or imposed by applicable federal or state securities laws.
- (b) The "ProMedica Plan Contribution" means Cash to be invested or contributed by ProMedica to the Reorganized Debtor, in the form, at the election of ProMedica, of (i) a capital contribution to the Reorganized Debtor or (ii) a combination of (A) the capital contribution referred to in clause (i) and (B) an unsecured, subordinated loan to the Reorganized Debtor in a principal amount not to exceed \$550 million, which unsecured loan shall mature no earlier than the third anniversary of the Closing Date and the other terms and conditions of which shall be reasonably acceptable to the Debtor, in an amount sufficient to pay, in full, the Cash distribution to the Holders of Class 3 Claims under the Amended Plan, the Agreed Deferred Rent Obligation and the Total Equity Distribution (as defined in the Amended Plan). The "Agreed Deferred Rent Obligation" means that portion of the Allowed QCP Claims under the Amended Plan equal to \$440,190,183.
- (c) For purposes of this Agreement, "<u>Lien</u>" means any mortgage, lien, pledge, charge, security interest, easement, covenant, or other restriction or title matter or encumbrance of any kind in respect of such asset; and "<u>Permitted Lien</u>" means the following Liens: (i) specified encumbrances described in Section 1.1(a) of the Debtor Disclosure Letter; (ii) encumbrances for Taxes or other governmental charges not yet due and payable or that are being contested in good faith by appropriate proceedings and are reflected on or specifically reserved against or otherwise

disclosed in any consolidated balance sheet included in the Audited Financial Statements; (iii) mechanics', carriers', workmen's, repairmen's or other like encumbrances arising or incurred in the ordinary course of business consistent with past practice relating to obligations as to which there is no default on the part of the Debtor, or the validity or amount of which is being contested in good faith by appropriate proceedings and are reflected on or specifically reserved against or otherwise disclosed in any consolidated balance sheet included in the Audited Financial Statements; (iv) other encumbrances that do not, individually or in the aggregate, materially impair the current use or operation of the specific parcel of owned real property to which they relate or the conduct of the business of the Debtor and its Subsidiaries as presently conducted; (v) easements, rights of way or other similar matters or restrictions or exclusions which would be shown by a current title report or other similar report; (vi) any condition or other matter, if any, that may be shown or disclosed by a current and accurate survey or physical inspection; and (vii) the renewal or reaffirmation by the Debtor or any of its Subsidiaries of any Lien existing on the date hereof granted in favor of RD Credit, LLC and/or the lender parties to the Centerbridge Facility (or any agent on their behalf) to secure obligations thereunder.

1.2 New Master Lease and Guaranty.

- (a) On the Closing Date, (i) the Lessors and HCR III shall enter into the New Master Lease and ProMedica shall enter into a guaranty of the New Master Lease and (ii) HCR III and the applicable Sublease OpCos shall enter into amendments to each Sublease substantially similar to the New Master Lease, *mutatis mutandis* (the "Sublease Amendments").
- (b) On or prior to the third day after the Closing Date, the Original Lessors and HCR III shall terminate, release and discharge, and QCP shall cause the Original Lessors and HCR III to terminate, release and discharge, the Guaranty.
 - 1.3 Articles of Incorporation and By-Laws of the Reorganized Debtor.
- (a) <u>Articles of Incorporation</u>. Pursuant to the Amended Plan, on the Closing Date, the articles of incorporation of the Debtor (the "<u>Charter</u>") shall be amended in their entirety to be in a form set forth in a supplement to the Amended Plan until, subject to Section 4.7, thereafter amended as provided therein or by applicable Law.
- (b) <u>By-Laws</u>. Pursuant to the Amended Plan, on the Closing Date, the by-laws of the Debtor (the "<u>By-Laws</u>") shall be amended in their entirety to be in a form set forth in a supplement to the Amended Plan until, subject to Section 4.7(a), thereafter amended as provided therein or by applicable Law.
- 1.4 <u>Closing</u>. Unless otherwise mutually agreed in writing between the Debtor, QCP and ProMedica, the closing of the Plan Acquisition, the entry into the New Master Lease and the Sublease Amendments and the other transactions contemplated by this Agreement (the "<u>Closing</u>" and such transactions, collectively, the "<u>Transactions</u>") shall take place at the offices of Gibson Dunn & Crutcher LLP, 200 Park Avenue, 47th Floor, New York, New York 10166 at 9:00 a.m. on the second (2nd) business day following the day on which the last to be satisfied or

waived of the conditions set forth in ARTICLE VI (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the fulfillment or waiver of those conditions) shall be satisfied or waived in accordance with this Agreement (the date of the Closing, the "Closing Date"). For purposes of this Agreement, the term "business day" means any day ending at 11:59 p.m. (Eastern Time) other than a Saturday or Sunday or a day on which banks are required or authorized to close in the City of New York.

1.5 <u>Severance and Other Payments.</u>

Subject to Section 1.5(b), on the Closing Date, following the Closing, (i) (a) ProMedica shall cause the Severance Account to be funded with all amounts necessary to make severance payments to Eligible Employees (each as defined in the Amended Plan) in the amounts set forth in Section 1.5(i) of the Debtor Disclosure Letter (the "Excess Severance Payments"), (ii) ProMedica shall cause the Excess Severance Payments due each Eligible Employee to be paid from the Severance Account to such Eligible Employee on the Payment Date with respect to such Eligible Employee (it being understood that, for the purpose of determining entitlement to payment of Excess Severance Payments, the employment of each Eligible Employee who remains employed by the Debtor immediately prior to the Closing shall be deemed to have been terminated without Cause constituting a Qualifying Termination (each as defined in such Eligible Employee's written Employment Agreement with the Debtor (each, an "Employment Agreement")) as of the Closing), (iii) except as provided in clause (iv) of this Section 1.5(a), ProMedica shall pay, or cause to be paid, the other obligations set forth in Section 1.5 of the Debtor Disclosure Letter, in each case, on the applicable Payment Date; provided, that payments made in respect of any nonqualified deferred compensation plan shall be made in accordance with the relevant Eligible Employee's deferral election; provided, further, that payments to be made under the Key Employee Incentive Plan shall be all those due and payable as a result of the Restructuring Completion (as defined in the Key Employee Incentive Plan), (iv) the Debtor shall irrevocably terminate the HCR ManorCare, Inc. Senior Executive Retirement Plan, as amended, as of the Closing Date and ProMedica shall pay or cause to be paid all amounts due thereunder, which amounts the Parties agree are set forth in Section 1.5 of the Debtor Disclosure Letter (and which amounts, for the avoidance of doubt, reflect the actuarial reduction provided for under Section 7.05(b) of the HCR ManorCare, Inc. Senior Executive Retirement Plan) on the applicable Payment Date, (v) ProMedica will execute and deliver the applicable Separation Agreement to each Eligible Employee (if not previously executed and delivered) and (vi) ProMedica will execute and deliver the Former CEO Settlement Agreement to the Former CEO (if not previously executed and delivered), subject to the Former CEO's execution and delivery of the Former CEO Settlement Agreement to ProMedica. The "Payment Date" means (I) with respect to each Eligible Employee, the later of (x) the Closing Date and (y) the eighth (8th) day after such Eligible Employee shall have executed (and not revoked) such Eligible Employee's Separation Agreement, (II) with respect to amounts payable to the Former CEO (other than the severance obligation to the Former CEO set forth on Section 1.5 of the Debtor Disclosure Letter), the Closing Date and (III) with respect to the severance obligation to the Former CEO set forth on Section 1.5 of the Debtor Disclosure Letter, the later of (x) the Closing Date and (y) the eighth (8th) day after the Former CEO shall have executed and delivered to the Debtor (and not revoked) the Former CEO Settlement Agreement. ProMedica shall also fund any employer Taxes related to the amounts payable pursuant to this Section 1.5(a) and timely pay such amounts, plus any applicable withholding Taxes related to such payments, to the applicable Governmental Entity (provided, for the avoidance of doubt, that all amounts payable subject to this Section 1.5 shall be subject to withholding of any Taxes that are required to be withheld pursuant to applicable Laws).

- (b) Notwithstanding anything to the contrary in this Agreement or the Amended Plan, ProMedica shall have no obligation to pay, or cause to be paid, any Excess Severance Payments to any Eligible Employee:
- (i) if such Eligible Employee separates from service without the written consent of the ProMedica prior to the Closing Date; it being understood and agreed that if such Eligible Employee separates from service without the written consent of the ProMedica prior to the Closing Date due to the Eligible Employee's Disability (as defined in such Eligible Employee's written Employment Agreement with the Debtor (the "Employment Agreement")) or death, such separation from service shall be deemed and shall constitute a Qualifying Termination within the meaning of such Eligible Employee's Employment Agreement for all purposes, and the Eligible Employee shall remain entitled to, and shall receive, the Excess Severance Payments (irrespective of whether such separation from service otherwise would have constituted a Qualifying Termination under the Employment Agreement without consideration of this Agreement) upon the Payment Date, subject to Sections 1.5(b)(ii), 1.5(b)(iii) and 1.5(b)(iv) of this Agreement;
- (ii) if such Eligible Employee does not execute and deliver a separation agreement and release in a form attached as Exhibit 4 hereto (each, a "Separation Agreement");
- (iii) to the extent that the Debtor has not obtained the approvals and waivers contemplated by Section 4.5(e)(i) and Section 4.5(e)(ii) with respect to such Eligible Employee (including, for the avoidance of doubt, waivers in respect of any tax gross-ups to which such Eligible Employee would otherwise be entitled); or
- (iv) if (A) there has been a material breach of any covenant or agreement made by the Debtor in this Agreement and such material breach is not curable or, if curable, is not cured by the earlier of (i) Closing and (ii) thirty (30) days after written notice thereof is given by ProMedica to the Debtor and (B) prior to Closing, the Debtor Board determines that such Eligible Employee is responsible for or had the ability to avoid such breach and knowingly failed to do so (it being understood that, if ProMedica asserts in a written notice to the Debtor Board that it believes any Eligible Employee is responsible for any such material breach or had the ability to avoid such material breach and knowingly failed to do so, the Debtor Board shall consider such assertion by ProMedica and make a good faith determination as to whether such Eligible Employee is responsible for any such material breach or had the ability to avoid such material breach and knowingly failed to do so); provided that all determinations by the Debtor Board relating to this Section 1.5(b)(iv) shall be made by the non-employee directors thereof.
- 1.6 <u>Deliveries by Debtor</u>. On the Closing Date, the Debtor shall deliver to ProMedica:
- (a) a certificate executed by the Debtor that it is not a foreign person within the meaning of Section 1445(f)(3) of the Internal Revenue Code of 1986, as amended, and the Treasury Regulations promulgated thereunder (the "Code");

- (b) (i) a certificate representing the Reorganized Company Stock, (ii) evidence of the resignations or removal of those directors and officers of the Debtor and its Subsidiaries, as applicable, effective as of immediately prior to the Closing, identified in writing to the Debtor by ProMedica not less than five (5) business days prior to the Closing Date, (iii) evidence of the appointment and approval by the board of directors of the Debtor (the "**Debtor Board**") of those directors nominated by ProMedica not less than five (5) business days prior to the Closing Date, to the Debtor Board effective as of the Closing, and (iv) evidence of amendments to the constitutive documents of the Debtor and its Subsidiaries effective as of the Closing that may be necessary to implement acquisition of the Reorganized Company Stock as contemplated by the Amended Plan, including in accordance with Section 1.3 of this Agreement, and which shall have been identified by ProMedica in writing to the Debtor not less than five (5) business days prior to the Closing Date; and
- (c) such other documents, instruments and certificates as ProMedica may reasonably request and as are customary, including a certified copy of the Confirmation Order and notice of the Closing Date.
- (d) As used in this Agreement, the term (i) "<u>Subsidiary</u>" means, with respect to any Person, any other Person of which at least a majority of the securities or ownership interests having by their terms ordinary voting power to elect a majority of the board of directors or other persons performing similar functions is directly or indirectly owned or controlled by such Person and/or by one or more of its Subsidiaries; (ii) "<u>Person</u>" means any individual, corporation, limited liability company, partnership, limited partnership, firm, joint venture, association, joint-stock company, trust, unincorporated organization or any other entity; and (iii) "<u>Governmental Entity</u>" means any domestic or foreign governmental or regulatory authority, agency, commission, body, court or other legislative, executive or judicial governmental entity.
- 1.7 <u>Deliveries by ProMedica</u>. At the Closing, ProMedica shall deliver to the Debtor any documents, instruments or certificates as the Debtor may reasonably request and as are customary.
- 1.8 <u>Deliveries by HCR III and the Lessors</u>. At the Closing, the Debtor shall cause HCR III to execute and deliver, and HCR III shall execute and deliver, the New Master Lease, and the JV shall cause the Lessors to execute and deliver, and the Lessors shall execute and deliver, the New Master Lease.

ARTICLE II

Representations and Warranties

2.1 Representations and Warranties of the Debtor. Except as set forth in the corresponding sections or subsections of the Debtor Disclosure Letter (as defined in the Original Plan Sponsor Agreement) (it being agreed that disclosure of any item in any section or subsection of the Debtor Disclosure Letter shall be deemed disclosed with respect to any other section or subsection to which the relevance of such item is reasonably apparent), the Debtor hereby represents and warrants to ProMedica as of the date of the Original Plan Sponsor Agreement

(provided that the representations and warranties set forth in Sections 2.1(a) - (d) shall be deemed to be made as of the date hereof) that:

entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing (with respect to jurisdictions that recognize the concept of good standing) as a foreign corporation or other legal entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or the conduct of its business requires such qualification, except where the failure to be so qualified or in good standing, or to have such power or authority, would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The Debtor has made available to ProMedica complete and correct copies of the Debtor's and its material Subsidiaries' certificates of incorporation and by-laws or comparable governing documents, each as amended to the date of this Agreement, and each as so delivered is in full force and effect.

As used in this Agreement, the term "<u>Material Adverse Effect</u>" with respect to the Debtor means a change, event or occurrence that has a material adverse effect on the financial condition, properties, assets, liabilities, business or results of operations of the Debtor and its Subsidiaries taken as a whole; <u>provided</u>, <u>however</u>, that none of the following, and no changes, events, occurrences or effects, to the extent arising out of or resulting from any of the following, shall constitute or be taken into account in determining whether a Material Adverse Effect has occurred or may, would or could occur with respect to the Debtor:

- (i) changes in or generally affecting the economy, credit, capital or financial markets or political conditions generally in the United States or elsewhere in the world, including changes in interest and exchange rates, or changes that are the result of acts of war (whether or not declared), armed hostilities, sabotage or terrorism, any escalation or worsening of any such acts of war (whether or not declared), epidemics, pandemics, earthquakes, hurricanes, tornados or other natural disasters;
- (ii) changes that are the result of factors generally affecting the industries in which the Debtor and its Subsidiaries operate or in which the services of the Debtor and its Subsidiaries are used and distributed:
- (iii) any loss of, or adverse change or effect in, the relationship of the Debtor with its customers, employees, financing sources or suppliers, caused by entry into or the pendency or the announcement of the Transactions;
- (iv) changes from the entry into, announcement or performance of this Agreement (including any litigation arising from allegations of any breach of fiduciary duty or violation of Law relating to this Agreement or the Transactions, or compliance by the Debtor with the terms of this Agreement);

- (v) changes or prospective changes in United States generally accepted accounting principles ("<u>GAAP</u>") or in any law, statute, rule or regulation after the date of this Agreement or the interpretation or enforcement thereof;
- (vi) any failure by the Debtor to meet any projections, forecasts or estimates of revenues or earnings for any period, <u>provided</u> that the exception in this clause shall not (on its own) prevent or otherwise affect a determination that any change, event or occurrence underlying such failure has resulted in, or contributed to, a Material Adverse Effect;
- (vii) (A) any action taken by the Debtor or its Subsidiaries at ProMedica's written request or (B) the failure to take any action by the Debtor or its Subsidiaries if that action is prohibited by this Agreement to the extent that ProMedica fails to give its consent after receipt of a written request therefor;
- (viii) any change or announcement of a potential change in the credit rating of the Debtor or any of its Subsidiaries or its obligations, <u>provided</u> that the exception in this clause shall not (on its own) prevent or otherwise affect a determination that any change, event or occurrence underlying such failure has resulted in, or contributed to, a Material Adverse Effect; or
- (ix) loss of customers or patients (and loss of revenue associated therewith), change in or tightening of payment or other terms provided by vendors, suppliers or service providers to the Debtor or its Subsidiaries, in each case, occurring since December 31, 2017 through the date of this Agreement;

<u>provided</u>, <u>further</u>, that, with respect to clauses (i), (ii) and (v), such change, event or occurrence does not disproportionately adversely affect the Debtor and its Subsidiaries compared to other companies operating in the principal industries in which the Debtor and its Subsidiaries operate.

(b) Capital Structure.

- (i) The authorized capital stock of the Debtor consists of 55,000,000 shares of common stock of the Debtor, par value \$0.01 per share ("Shares"), and 50,000,000 preferred shares, par value \$0.01 per share ("Preferred Shares"), of which 44,925,848 Shares and 2,000 Preferred Shares, respectively, were outstanding as of the close of business on March 31, 2018. All of the outstanding Shares have been duly authorized and are validly issued, fully paid and nonassessable. Other than 2,049,823 Shares reserved for issuance under the Amended and Restated Equity Incentive Plan of the Debtor (the "Stock Plan"), as of the close of business on March 31, 2018, the Debtor had no Shares reserved for issuance.
- (ii) Section 2.1(b)(ii) of the Debtor Disclosure Letter sets forth (x) each of the Debtor's Subsidiaries and the ownership interest of the Debtor in each such Subsidiary, as well as the ownership interest of any other Person or Persons in each such Subsidiary and (y) the Debtor's or its Subsidiaries' capital stock, equity interest or other direct or indirect ownership interest in any other Person other than securities in a publicly traded company held for investment by the Debtor or any of its Subsidiaries and consisting of less than 1% of the outstanding capital stock of such company.

Corporate Authority and Approval. The Debtor has all requisite corporate power, authority and legal capacity to execute and deliver this Agreement and each other agreement, document, instrument or certificate contemplated by this Agreement or to be executed by the Debtor in connection with the consummation of the Transactions (the "Debtor **Documents**") and to perform its obligations hereunder and thereunder and consummate the Transactions. The execution and delivery of this Agreement and the Debtor Documents and the consummation of the Transactions contemplated thereby have been duly authorized by all requisite corporate action on the part of the Debtor. This Agreement has been, and the Debtor Documents will be at or prior to the Closing, duly and validly executed and delivered by, or on behalf of, the Debtor and (assuming the due authorization, execution and delivery by the other parties hereto and thereto and the entry of the Confirmation Order) this Agreement constitutes, and each of the Debtor Documents when so executed and delivered will constitute, legal, valid and binding obligations of the Debtor and each of its applicable Subsidiaries, enforceable against it in accordance with their respective terms, subject to applicable bankruptcy, insolvency, reorganization, fraudulent transfer, moratorium and similar laws affecting creditors' rights generally and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity) (the "Bankruptcy and Equity Exception").

(d) <u>Governmental Filings; No Violations; Certain Contracts.</u>

- (i) Other than (A) the filings, notices, consents, registrations, approvals, permits or authorizations (including those with respect to state licensing required to operate the Debtor's businesses) set forth on Section 2.1(d)(i) of the Debtor Disclosure Letter (the "Governmental Approvals"), (B) the entry of the Confirmation Order and (C) and compliance with the applicable requirements under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), no notices, reports or other filings are required to be made by the Debtor with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by the Debtor from any Governmental Entity, in connection with the execution, delivery and performance of this Agreement by the Debtor and the consummation of the Transactions, except those failures to make or obtain as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect or prevent, materially delay or materially impair the consummation of the Transactions.
- (ii) The execution, delivery and performance of this Agreement by the Debtor do not, and the consummation of the Transactions will not, constitute or result in (A) a breach or violation of, or a default under, the certificate of incorporation or by-laws of the Debtor or the comparable governing documents of any of its Subsidiaries or (B) with or without notice, lapse of time or both, a breach or violation of, a termination (or right of termination) or default under, the creation or acceleration of any obligations under or the creation of a Lien on any of the assets of the Debtor or any of its Subsidiaries pursuant to any agreement, lease, license, contract, note, mortgage, indenture, arrangement or other obligation, in each case, not otherwise terminable by the other party thereto on sixty (60) days' or less notice (each, a "Contract") to which the Debtor or any of its Subsidiaries or its or any of their respective properties are bound or (C) assuming compliance with the matters referred to in Section 2.1(d)(i), a violation of any Law to which the Debtor or any of its Subsidiaries is subject, except in the case of clause (B) or (C), for any such breach, violation, termination, default,

creation, acceleration or change that is not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect.

(e) <u>Financial Statements</u>.

- (i) Set forth in the Debtor Disclosure Letter are the audited consolidated balance sheets of the Debtor and its Subsidiaries as of December 31, 2016 and December 31, 2017, and the related audited consolidated statements of operations, statements of comprehensive (loss) income, statements of cash flows and statements of equity for the twelve (12)-month periods then ended (the "Audited Financial Statements"). Each of the balance sheets, statements of operations, statements of comprehensive loss, statements of cash flows and statements of equity included in the Audited Financial Statements (including any related notes thereto) fairly presents, in all material respects, the financial condition, results of operations and cash flows, as the case may be, of the Debtor and its Subsidiaries as of the dates and for the periods set forth therein, as applicable, in each case in accordance with GAAP, consistently applied during the periods involved, except as specifically noted therein.
- The Debtor and its Subsidiaries maintain a system of internal accounting controls designed, in all material respects, to provide reasonable assurances (i) that transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP, (ii) that receipts and expenditures of the Debtor and its Subsidiaries are being made in accordance with appropriate authorizations of management and the Debtor Board and (iii) regarding prevention or timely detection of unauthorized acquisition, use or disposition of assets of the Debtor and its Subsidiaries. To the Knowledge of the Debtor, there is not, and in the past five (5) years there has not been, any fraud that involves or involved the Debtor's and/or its Subsidiaries' management or other employees who have or had a significant role in the Debtor's and/or its Subsidiaries' internal control over financial reporting except as would not, individually or in the aggregate, be material to the Debtor and its Subsidiaries. No significant deficiencies or material weaknesses in the design or operation of the Debtor's system of internal control over financial reporting were identified in the Debtor's most recent assessment that would be reasonably likely to materially affect the Debtor's and/or its Subsidiaries' ability to record, process, summarize and report financial information. As used herein, "Knowledge of the **<u>Debtor</u>**" means the actual (but not constructive or imputed) knowledge of the chief executive officer, chief financial officer, chief operating officer or general counsel of the Debtor (without any obligation of further review and/or inquiry).
- (iii) Each of the balance sheets, statements of operations, statements of comprehensive loss, statements of cash flows and statements of equity (including any related notes thereto) required to be delivered to QCP pursuant to the terms of the Master Lease after the date of this Agreement, including the quarterly unaudited consolidated balance sheets of the Debtor and its Subsidiaries as of March 31, 2018 and any subsequent quarterly unaudited consolidated balance sheets of the Debtor and its Subsidiaries, and the related unaudited consolidated statements of operations, statements of comprehensive (loss) income, statements of cash flows and statements of equity, will fairly present, in all material respects, the financial condition, results of operations and cash flows, as the case may be, of the Debtor and its Subsidiaries as of the dates and for the periods set forth therein, as applicable, in each case in accordance with GAAP, consistently applied during the periods involved, except as specifically noted therein, subject, in the case of any unaudited

consolidated balance sheets, statements of operations, statements of comprehensive loss, statements of cash flows and statements of equity, to the absence of footnotes and to year-end adjustments, except as would not, individually or in the aggregate, be material in amount or effect.

- (f) <u>Absence of Certain Changes</u>. Since the date of the most recent Audited Financial Statements through the date hereof, there has not been:
- (i) any material damage, destruction or other casualty loss with respect to any asset or property owned, leased or otherwise used by the Debtor or any of its Subsidiaries, whether or not covered by insurance;
- (ii) any declaration, setting aside or payment of any dividend or other distribution with respect to any shares of capital stock of the Debtor or any of its Subsidiaries (except for dividends or other distributions by any direct or indirect wholly-owned Subsidiary to the Debtor or to any wholly-owned Subsidiary of the Debtor), or any repurchase, redemption or other acquisition by the Debtor or any of its Subsidiaries of any outstanding shares of capital stock or other securities of the Debtor or any of its Subsidiaries;
- (iii) any material change in any method of accounting or accounting practice by the Debtor or any of its Subsidiaries;
- (iv) (A) any material increase in the compensation payable or to become payable to the officers or employees of the Debtor or any of its Subsidiaries or material increases in the benefits of such officers or employees (except for increases in the compensation of non-officer employees in the ordinary course of business and consistent with past practice), (B) any entrance into, adoption, material amendment or termination of any Company Plan (as defined in Section 2.1(h)(i)) (or any arrangement that would have been a Company Plan were it in effect as of the date hereof) that is or would be material to the Debtor and its Subsidiaries, taken as a whole, or (C) any actions taken by the Debtor or any of its Subsidiaries to accelerate the vesting or lapsing of restrictions or payment, or fund or in any other way secure the payment, of any material compensation or benefits under any Company Plan, taken as a whole, except, in each case, to the extent required by applicable Laws; or
 - (v) any agreement to do any of the foregoing.
- civil, criminal or administrative actions, suits, claims, hearings, arbitrations or indictments or other proceedings by or before any Governmental Entity (collectively, "Proceedings") (or to the Knowledge of the Debtor, any audits or investigations) pending or, to the Knowledge of the Debtor, threatened in writing against the Debtor or any of its Subsidiaries, except for any such Proceeding or investigation that is not reasonably likely to result in, individually or in the aggregate, a Material Adverse Effect. As of the date of this Agreement, none of the Debtor nor any of its Subsidiaries is a party to or subject to the provisions of any temporary, preliminary or permanent judgment, order, writ, injunction, ruling, decree, assessment or award of any Governmental Entity or arbitrator (collectively, an "Order") specifically imposed upon the Debtor or any of its Subsidiaries which is, individually or in the aggregate, reasonably likely to have a Material Adverse Effect.

(h) Employee Benefits.

(i) Section 2.1(h)(i)-A of the Debtor Disclosure Letter sets forth an accurate and complete list of each material Company Plan. No Company Plan is maintained outside the jurisdiction of the United States or covers any employees or other service providers of the Debtor or any of its Subsidiaries who primarily reside or work outside the United States.

For purposes of this Agreement, "Company Plan" means any benefit or compensation plan, program, policy, practice, agreement, Contract, arrangement or other obligation, whether or not in writing and whether or not funded (and, in each case, which is sponsored or maintained by, or required to be contributed to, or with respect to which any Liabilities (as defined in Section 2.1(k)) are borne by the Debtor or any of its Subsidiaries. Company Plans include, but are not limited to, "employee benefit plans" within the meaning of Section 3(3) of ERISA ("ERISA Plans"), the Stock Plan, employment, consulting, retirement, severance, termination or change in control agreements, deferred compensation, equity-based, incentive, bonus, supplemental retirement, profit sharing, insurance, medical, welfare, vacation, fringe or other benefits or remuneration of any kind.

- (ii) With respect to each material Company Plan, the Debtor has made available to ProMedica, to the extent applicable, accurate and complete copies of (A) the Company Plan document, including any material amendments thereto, and all related trust documents, insurance contracts or other funding vehicles, (B) a written description of such Company Plan if such plan is not set forth in a written document, (C) the most recent annual actuarial report or annual report on Form 5500 or Form 990, and all schedules and financial statements attached thereto, (D) all material and non-routine correspondence to or from any Governmental Entity received in the last three (3) years with respect to such Company Plan, and (E) the most recent Internal Revenue Service ("<u>IRS</u>") determination or opinion letter, and related trust intended to be qualified under Section 401(a) of the Code and any pending request for such a determination letter.
- (iii) Except as would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect, (A) each Company Plan (including any related trusts) has been established, operated and administered in compliance with its terms and applicable Laws, including, without limitation, ERISA and the Code, (B) all contributions or other amounts payable by the Debtor or a Debtor Subsidiary with respect to each Debtor Plan in respect of current or prior plan years have been paid or accrued in accordance with generally accepted accounting principles, and (C) there are no pending or, to the Knowledge of the Debtor, threatened claims (other than routine claims for benefits) or Proceedings by a Governmental Entity by, on behalf of or against any Company Plan or any trust related thereto.
- (iv) Except as would not reasonably be expected to result in a Material Adverse Effect, each ERISA Plan that is intended to be qualified under Section 401(a) of the Code has been determined by the IRS to be qualified under Section 401(a) of the Code and, to the Knowledge of the Debtor, nothing has occurred that would reasonably be expected to result in the loss of qualification or tax exemption of any such Company Plan. Except as would not reasonably be expected to result in a Material Adverse Effect, with respect to any ERISA Plan, neither the Debtor nor any of its Subsidiaries has engaged in a transaction in connection with which the Debtor or any of its Subsidiaries reasonably would be subject to either a civil penalty assessed

pursuant to Section 409 or 502(i) of ERISA or a Tax imposed pursuant to Section 4975 or 4976 of the Code.

- Except as would not reasonably be expected to result in a Material (v) Adverse Effect, neither the Debtor nor any of its Subsidiaries has or is expected to incur any material liability under subtitles C or D of Title IV of ERISA with respect to any ongoing, frozen or terminated "single-employer plan", within the meaning of Section 4001(a)(15) of ERISA, currently or formerly maintained by any of them or any ERISA Affiliate. Except as would not, individually or in the aggregate, reasonably be expected to result in material liability to the Debtor or any of its Subsidiaries, with respect to any Company Plan subject to the minimum funding requirements of Section 412 of the Code or Title IV of ERISA, (A) no such plan is, or is reasonably expected to be, in "at-risk" status (within the meaning of Section 303(i)(4)(A) of ERISA or Section 430(i)(4)(A) of the Code), (B) as of the last day of the most recent plan year ended prior to the date hereof, the actuarially determined present value of all "benefit liabilities" within the meaning of Section 4001(a)(16) of ERISA did not exceed the then-current value of assets of such Company Plan, (C) no unsatisfied liability (other than for premiums to the PBGC) under Title IV of ERISA has been, or is reasonably expected to be, incurred by the Debtor or any of its Subsidiaries, (D) the Pension Benefit Guaranty Corporation ("PBGC") has not instituted proceedings to terminate any such Company Plan and (E) no "reportable event" within the meaning of Section 4043 of ERISA (excluding any such event for which the thirty (30)-day notice requirement has been waived under the regulations to Section 4043 of ERISA) has occurred, nor has any event described in Section 4062, Section 4063 or Section 4041 of ERISA occurred. For purposes of this Agreement, "ERISA Affiliate" means all employers (whether or not incorporated) that would be treated together with the Debtor or any of its Subsidiaries as a "single employer" within the meaning of Section 414 of the Code.
- (vi) Neither the Debtor nor any of its Subsidiaries has maintained, established, participated in or contributed to, or is or has been obligated to contribute to, or has otherwise incurred any obligation or liability (including any contingent liability) under, any "multiemployer plan" within the meaning of Section 3(37) of ERISA in the last six (6) years.
- (vii) Except as required by applicable Law, no Company Plan provides retiree or post-employment medical, disability, life insurance or other welfare benefits to any Person, and none of the Debtor or any of its Subsidiaries has any obligation to provide such benefits.
- (viii) Section 2.1(h)(viii) of the Debtor Disclosure Letter sets forth the following with respect to each Company Plan that is a "nonqualified deferred compensation plan" (within the meaning of Section 409A of the Code), the Heartland Employment Services, LLC Key Employee Incentive Plan (the "Key Employee Incentive Plan") and the Heartland Employment Services, LLC Key Employee Retention Plan, in each case as of January 31, 2018 and in a manner that is accurate in all material respects: (A) the number of participants in each such Company Plan; (B) the dollar amount of the aggregate balances for all participants (and, with respect to any such nonqualified deferred compensation plan, individual balances for any executive officer) under each such Company Plan; and (C) for any supplemental retirement plan, the amount payable under such Company Plan, assuming the participant undergoes a "separation from service" (within the meaning of Section 409A of the Code) on the Closing Date. In addition, the Debtor has

provided to ProMedica a list of each participant in the Key Employee Incentive Plan and the Heartland Employment Services, LLC Key Employee Retention Plan with individual balances for such participants that are accurate in all material respects as of January 31, 2018.

- (ix) Neither the execution and delivery of this Agreement, nor the consummation of the Transactions will, either alone or in combination with another event, (A) entitle any current or former employee, director, officer or independent contractor of the Debtor or any of its Subsidiaries to material severance pay or any material increase in severance pay, (B) accelerate the time of payment or vesting, or materially increase the amount of compensation due to any such employee, director, officer or independent contractor, (C) directly or indirectly cause the Debtor to transfer or set aside any assets to fund any material benefits under any Company Plan, (D) otherwise give rise to any material liability under any Company Plan, (E) limit or restrict the right to merge, materially amend, terminate or transfer the assets of any Company Plan on or following the Closing Date or (F) result in the payment of any amount that could, individually or in combination with any other such payment, constitute an "excess parachute payment" as defined in Section 280G(b)(1) of the Code.
- (x) Neither the Debtor nor any of its Subsidiaries has any obligation to provide, and no Company Plan or other agreement provides any individual with the right to, a gross up, indemnification, reimbursement or other payment for any excise or additional taxes, interest or penalties incurred pursuant to Section 409A or Section 4999 of the Code or due to the failure of any payment to be deductible under Section 280G of the Code.

(i) Labor Matters.

- Section 2.1(i)(i) of the Debtor Disclosure Letter sets forth an accurate and complete list of any material collective bargaining agreement or other material agreement with a labor union or like organization that the Debtor or any of its Subsidiaries is a party to or otherwise bound by (collectively, the "Debtor Labor Agreements") as of the date of this Agreement and, to the Knowledge of the Debtor, as of the date of this Agreement, there are no activities or Proceedings by any individual or group of individuals, including representatives of any labor organizations or labor unions, to organize any employees of the Debtor or any of its Subsidiaries. Within seven (7) days following the date of this Agreement, the Debtor will make available to ProMedica accurate and complete copies of each Debtor Labor Agreement listed on Section 2.1(i)(i) of the Debtor Disclosure Letter. Neither the execution and delivery of this Agreement (including any shareholder or other approval of this Agreement) nor the consummation of the Transactions would reasonably be expected to, either alone or in combination with another event, entitle any third party (including any labor organization or Governmental Entity) to any material payments or consent rights under any of the Debtor Labor Agreements. Except as would not reasonably be expected to result in a Material Adverse Effect, the Debtor and its Subsidiaries are in compliance with their obligations pursuant to all notification and bargaining obligations arising under any Debtor Labor Agreements.
- (ii) As of the date hereof, there is no strike, lockout, slowdown or work stoppage pending or, to the Knowledge of the Debtor, threatened in writing, that is reasonably expected to interfere in any material respect with the respective business activities of the Debtor or any of its Subsidiaries.

- Compliance with Laws; Licenses. The businesses of each of the Debtor and (i) its Subsidiaries have not, since December 31, 2014, been, and are not being, conducted in violation of any federal, state, local or foreign law, statute or ordinance, common law, or any rule, regulation, standard, Order, agency requirement, license or permit of any Governmental Entity (collectively, "Laws"), except for violations that, individually or in the aggregate, are not reasonably likely to have a Material Adverse Effect. To the Knowledge of the Debtor, as of the date of this Agreement, no investigation by any Governmental Entity with respect to any actual or alleged violation of Law by the Debtor or any of its Subsidiaries is pending or threatened in writing, nor has any Governmental Entity notified the Debtor in writing of an intention to conduct the same, except for such investigations or reviews the outcome of which are not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect. The Debtor and its Subsidiaries each has obtained and is in compliance with all permits, licenses, certifications, approvals, registrations, consents, authorizations, franchises, variances, exemptions and orders issued or granted by a Governmental Entity ("Licenses") necessary to conduct its business as presently conducted, except those failures or the absence of which would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.
- (k) <u>Undisclosed Liabilities</u>. The Debtor and its Subsidiaries do not have any debts, claims, liabilities or obligations, including all costs and expenses relating thereto, whether absolute or contingent, known or unknown, liquidated or unliquidated, whether due or to become due and regardless of when or by whom asserted ("<u>Liabilities</u>") that would have been required to be reflected or reserved against, or otherwise described on a consolidated balance sheet of the Debtor prepared in accordance with GAAP or in the notes thereto, other than Liabilities (i) reflected or reserved against in the balance sheet included in the Audited Financial Statements or disclosed in the notes thereto, (ii) incurred in the ordinary course of the operation of business consistent with past practice of the Debtor and its Subsidiaries since December 31, 2016, (iii) arising or incurred in connection with the Transactions or (iv) which have not had, or would not, individually or in the aggregate, reasonably be expected to have, a Material Adverse Effect.

(1) Material Contracts and Government Contracts.

- (i) Section 2.1(1)(i) of the Debtor Disclosure Letter sets forth a list of each of the following Contracts to which, as of the date hereof, the Debtor or any of its Subsidiaries is a party or by which the Debtor or any of its Subsidiaries or any of their respective assets is bound (each, a "Material Contract"):
- (A) any Contract (or group of related Contracts with respect to a single transaction or series of related transactions) pursuant to which the Debtor or any of its Subsidiaries currently leases or subleases real or personal property to or from any Person (other than QCP or any of its Affiliates) providing for lease payments in excess of \$5,000,000 per annum (in each case, other than the Subleases);
- (B) any Contract pursuant to which any material Intellectual Property Rights are (1) licensed by any Person to the Debtor or any of its Subsidiaries (other than non-exclusive licenses to the Debtor or any of its Subsidiaries for non-customized software that is generally available on commercial terms) or (2) licensed by the Debtor or any of its Subsidiaries to any Person;

- (C) each joint venture, partnership and other similar Contract involving the sharing of profits of the Debtor or any of its Subsidiaries with any third party;
- (D) each Contract that expressly limits the freedom of the Debtor or any of its Subsidiaries (or, after the Closing, ProMedica or any of its Subsidiaries) to compete in any line of business or within any geographic area or with any Person;
- (E) each Contract under which the Debtor or any of its Subsidiaries has borrowed or loaned money, or any note, bond, indenture, mortgage or any guarantee of such indebtedness, in each case, relating to amounts in excess of \$5,000,000;
- (F) each Contract entered into in the past three (3) years relating to the acquisition or disposition of assets (other than in the ordinary course of business) or any capital stock of any enterprise, in each case, in excess of \$10,000,000 individually (other than such Contracts previously approved in writing by QCP or one of its Subsidiaries pursuant to the Master Lease);
- (G) each Contract entered into since January 1, 2015 related to any settlement or stipulation of any action against the Debtor or any of its Subsidiaries by any other Person, other than settlement agreements for cash that do not exceed \$1,000,000 individually as to any such settlement or stipulation (excluding amounts paid by insurers) or \$5,000,000 (including any amounts paid by insurers); and
- (H) each Contract that provides for indemnification of any officer, director or employee of the Debtor or any of its Subsidiaries other than in the ordinary course of business.
- (ii) Each Material Contract and each Government Contract is valid and binding on the Debtor or its Subsidiaries, as the case may be, and, to the Knowledge of the Debtor, each other party thereto, and, to the Knowledge of the Debtor, is in full force and effect, except for such failures to be valid and binding or to be in full force and effect as would not, or would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect. There is no material breach of or default under any such Contracts by the Debtor or its Subsidiaries and no event has occurred that, with the lapse of time or the giving of notice or both, would constitute a material breach of or default thereunder by the Debtor or any of its Subsidiaries or would permit termination, modification or acceleration thereof, in each case except as would not, or would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect. To the Knowledge of the Debtor, no party to any Material Contract or any Government Contract has delivered any written notice to the Debtor or any of its Subsidiaries that it is terminating such Material Contract or Government Contract, except as would not, or would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect.
- (iii) For purposes of this Agreement, the following terms shall have the following meanings:
- (A) "<u>Affiliate</u>" when used with respect to any party shall mean any Person who is an "affiliate" of that party within the meaning of Rule 405 promulgated under the Securities Act of 1933, as amended (the "<u>Securities Act</u>");

- (B) "<u>Government Contract</u>" means any contract to which the Debtor or any of its Subsidiaries is a party, or by which any of them is bound, the ultimate contracting party of which is a Governmental Entity (including any subcontract with a prime contractor or other subcontractor who is a party to any such contract); and
- rights arising under the Laws of any jurisdiction in the world, including all rights in or to any of the following: (1) trademarks, service marks, certification marks, collective marks, d/b/a's, Internet domain names, symbols, trade dress, trade names, and other indicia of origin, all applications and registrations for the foregoing, in each case, including all goodwill associated therewith and symbolized thereby; (2) patents, invention disclosures and applications therefor, including divisions, continuations, continuations-in-part and renewal applications, and including renewals, extensions and reissues; (3) trade secrets, proprietary or confidential information, inventions, know-how, processes, formulae, models, methodologies, formulae, drawings, prototypes, models, designs, customer lists and supplier lists; and (4) published and unpublished works of authorship, copyrights therein and thereto, and registrations and applications therefor, and all renewals, extensions, restorations and reversions thereof.
 - (m) Real Property. The Debtor does not own any real property.
- (n) <u>Takeover Statutes</u>. No "fair price", "moratorium", "control share acquisition" or other similar anti-takeover statute or regulation enacted under state or federal Laws in the United States (each, a "<u>Takeover Statute</u>") or any anti-takeover provision in the Debtor's certificate of incorporation or by-laws is applicable to the Debtor, the Shares or the Transactions.
- (o) <u>Environmental Matters</u>. Except as would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect, to the Knowledge of the Debtor, each real property operated by the Debtor or any of its Subsidiaries and used in the conduct of their respective businesses: (i) is in substantial compliance with all applicable Environmental Laws; (ii) is not the subject of any pending written notice from any Governmental Entity alleging the violation of any applicable Environmental Law; (iii) is not currently subject to any Orders arising under any Environmental Law; and (iv) has not had any emissions or discharges of Hazardous Substances except as permitted under applicable Environmental Laws.

As used herein, (i) the term "Environmental Law" means any federal, state, local or foreign statute, law, regulation, order, decree, permit, authorization or requirement of any Governmental Entity relating to: (A) the protection, investigation or restoration of the environment, natural resources or health or safety (in relation to exposure to Hazardous Substances), (B) the handling, use, presence, disposal, release or threatened release of any Hazardous Substance or (C) noise, odor, indoor air, employee exposure, wetlands, pollution, contamination or any injury or threat of injury to persons or property relating to any Hazardous Substance; and (ii) the term "Hazardous Substance" means any substance that is: (A) listed, classified or regulated pursuant to any Environmental Law; (B) any petroleum product or byproduct, asbestos-containing material, lead-containing paint or plumbing, polychlorinated biphenyls, mold, radioactive material or radon; and (C) any other substance which may be the subject of regulatory action by any Governmental Entity in connection with any Environmental Law.

(p) <u>Taxes</u>.

- (i) The Debtor and each of its Subsidiaries has prepared (or caused to be prepared) and timely filed (taking into account valid extensions of time within which to file) all material Tax Returns required to be filed by any of them, and all such filed Tax Returns (taking into account all amendments thereto) are accurate and complete in all material respects;
- (ii) all material Taxes owed by the Debtor and each of its Subsidiaries that are due and payable (whether or not shown as due on a Tax Return) have been timely paid or, if not yet due and payable, adequately reserved against in accordance with GAAP;
- (iii) no deficiency for any amount of material Taxes has been proposed or asserted in writing or assessed by any Taxing Authority against the Debtor or any of its Subsidiaries that remains unpaid or unresolved;
- (iv) the Debtor and each of its Subsidiaries has not received written notice of any pending Proceedings, examinations or proposed adjustments in respect of any material amount of Taxes of the Debtor or any of its Subsidiaries;
- (v) all material Taxes required to be withheld or collected by the Debtor or any of its Subsidiaries have been withheld and collected and, to the extent required by applicable Law, timely paid to the appropriate Governmental Entity or other person, and all IRS forms required with respect thereto have been properly completed and timely filed;
- (vi) there are no material Liens for Taxes on any of the assets of the Debtor or any of its Subsidiaries, other than Permitted Liens;
- (vii) neither the Debtor nor any of its Subsidiaries has been a "controlled corporation" or a "distributing corporation" in any distribution occurring during the two (2)-year period ending on the date of this Agreement that was purported or intended to be governed by Section 355 of the Code (or any similar provision of state, local or non-U.S. Law);
- (viii) no claim has been received, or is expected by the Debtor to be received, with respect to the Debtor or any of its Subsidiaries from a Taxing Authority in a jurisdiction where such entity does not file Tax Returns that it is or may be subject to taxation by, or required to file any Tax Return in, that jurisdiction which claim has not since been resolved;
- (ix) during the past ten (10) years, neither the Debtor nor any of its Subsidiaries has been a member of an affiliated group of corporations filing a consolidated federal income Tax Return (other than a group the common parent of which is the Debtor) or has had any liability for the Taxes of any person (other than the Debtor or any of its Subsidiaries) under U.S. Treasury Regulation Section 1.1502-6 (or any similar provision of any state, local or non-U.S. Law), as a transferee or successor;
- (x) neither the Debtor nor any of its Subsidiaries is a party to, or is bound by, or has any obligation under, any Tax sharing Contract, Tax allocation agreement, Tax indemnity obligation or similar agreement or practice other than (A) Contracts solely among the Debtor and/or its Subsidiaries and (B) customary Tax indemnification provisions in Contracts

entered into in the ordinary course of business, the primary purpose of which does not relate to Taxes;

- (xi) neither the Debtor nor any of its Subsidiaries has currently in effect any waiver of any statute of limitations in respect of Taxes or any agreement to any extension of time with respect to the filing of a Tax Return or an assessment or deficiency for Taxes (other than pursuant to extensions of time to file Tax Returns obtained in the ordinary course of business);
- (xii) neither the Debtor nor any of its Subsidiaries has participated in any "listed transaction" within the meaning of U.S. Treasury Regulations Section 1.6011-4(b); and
- (xiii) the Debtor has made available to ProMedica accurate and complete copies of any private letter ruling requests, closing agreements or gain recognition agreements with respect to Taxes of the Debtor or any of its Subsidiaries requested or executed in the last five (5) years.

As used in this Agreement, (i) the term "<u>Tax</u>" (including, with correlative meaning, the term "<u>Taxes</u>") means all federal, state, local or foreign taxes, including, without limitation, income, gross receipts, windfall profits, severance, property, production, sales, use, social security, license, excise, franchise, employment, withholding or similar taxes together with any interest, additions or penalties with respect thereto and any interest in respect of such additions or penalties, (ii) the term "<u>Tax Return</u>" means any return, filing, report, questionnaire, information statement, claim for refund or declaration of estimated Taxes, including any schedule or attachment thereto or any amendment thereof, filed or required to be filed with any Taxing Authority in connection with the determination, assessment or collection of any Tax, or the administration of any laws, regulations or administrative requirements relating to any Tax, including consolidated, combined and unitary tax returns and (iii) the term "<u>Taxing Authority</u>" means any Governmental Entity exercising regulatory authority in respect of taxes or responsible for the imposition of any Tax.

(q) Intellectual Property.

- (i) To the Knowledge of the Debtor, the Debtor and its Subsidiaries own or have a valid and enforceable right to use all material Intellectual Property Rights that are used in or necessary for the conduct of their respective businesses as presently conducted. All Intellectual Property Rights that are owned by the Debtor or any of its Subsidiaries ("Owned Intellectual Property") are, to the Knowledge of the Debtor, (A) valid, subsisting and enforceable and (B) not subject to any outstanding Order adversely affecting the Debtor's or any of its Subsidiaries' use of, or rights in or to any such Intellectual Property Rights, except in the case of clauses (A) and (B) as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (ii) To the Knowledge of the Debtor, neither the Debtor nor any of its Subsidiaries is infringing, misappropriating or otherwise violating any Intellectual Property Rights of any Person, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

- (iii) To the Knowledge of the Debtor, no Person is infringing, misappropriating or otherwise violating any Owned Intellectual Property, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (iv) To the Knowledge of the Debtor, no Person has gained unauthorized access to any Debtor IT Asset, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (v) The Debtor and each of its Subsidiaries are in compliance in all material respects with their respective privacy and security policies and with all applicable Laws regarding privacy and personal information, including with respect to the collection, storage, transmission, transfer, disclosure and use of personal information, in each case, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The Debtor and each of its Subsidiaries have taken commercially reasonable measures consistent with customary industry practices to ensure the confidentiality, privacy and security of all personal information of their customers and employees, and no Person has gained unauthorized access to, or misused, any such information, in each case, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (vi) For purposes of this Agreement, "<u>Debtor IT Assets</u>" means all computers, software, firmware, middleware, servers, workstations, routers, hubs, switches, data communications lines, and all other information technology equipment and associated documentation owned or used by the Debtor or any of its Subsidiaries.
- (r) <u>Insurance</u>. The fire and casualty, general liability, business interruption, product liability, and sprinkler and water damage insurance policies maintained by the Debtor or any of its Subsidiaries ("<u>Insurance Policies</u>") provide full and adequate coverage for all normal risks incident to the business of the Debtor and its Subsidiaries and their respective properties and assets, except for any such failures to maintain such insurance policies that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Each of the Insurance Policies is in full force and effect and all premiums due with respect to all Insurance Policies have been paid, in each case, with such exceptions that, individually or in the aggregate, are not reasonably expected to have a Material Adverse Effect.
- (s) <u>Brokers and Finders</u>. Neither the Debtor nor any of its officers, directors or employees has employed any broker or finder or incurred any Liability for any brokerage fees, commissions or finders' fees in connection with the Transactions or otherwise, except for fees payable to Moelis & Company as its financial advisor. The Debtor has made available to ProMedica a complete and accurate copy of all agreements pursuant to which Moelis & Company is entitled to any fees and expenses in connection with any of the Transactions or otherwise.
- 2.2 <u>Representations and Warranties of ProMedica</u>. ProMedica hereby represents and warrants to the Debtor that:
- (a) <u>Organization, Good Standing and Qualification</u>. Each of ProMedica Parent and the Purchaser is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power

and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing (with respect to jurisdictions that recognize the concept of good standing) as a foreign corporation or other entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so organized, qualified or in such good standing, or to have such power or authority, would not, individually or in the aggregate, reasonably be expected to prevent, materially delay or impair the ability of ProMedica to consummate the Transactions.

(b) <u>Corporate Authority</u>. Each of the Purchaser and ProMedica Parent has all requisite corporate or similar power and authority and has taken all corporate or similar action necessary in order to execute, deliver and perform its obligations under this Agreement and to consummate the Transactions. This Agreement has been duly executed and delivered by each of the Purchaser and ProMedica Parent, and is a valid and binding agreement of the ProMedica Entities, enforceable against each of the Purchaser and ProMedica Parent, in accordance with its terms, subject to the Bankruptcy and Equity Exception.

(c) Governmental Filings; No Violations; Etc.

- (i) Other than the Governmental Approvals, and compliance with the applicable requirements under the HSR Act, no notices, reports or other filings are required to be made by ProMedica, with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by ProMedica from, any Governmental Entity in connection with the execution, delivery and performance of this Agreement by ProMedica and the consummation by ProMedica of the Transactions, except those that the failure to make or obtain would not, individually or in the aggregate, reasonably be expected to prevent or materially delay the ability of ProMedica to consummate the Transactions.
- ProMedica do not, and the consummation by ProMedica of the Transactions contemplated hereby will not, constitute or result in (A) a breach or violation of, or a default under, the certificate of incorporation or by-laws of ProMedica or the comparable governing instruments of any of their respective Subsidiaries, (B) with or without notice, lapse of time or both, a breach or violation of, a termination (or right of termination) or a default under, the creation or acceleration of any obligations under or the creation of a Lien on any of the assets of ProMedica Parent, Purchaser or any of their respective Subsidiaries pursuant to, any Contracts binding upon ProMedica or any Laws or governmental or non-governmental permit or license to which ProMedica Parent, Purchaser or any of their respective Subsidiaries is subject, or (C) any change in the rights or obligations of any party under any of such Contracts, except, in the case of clause (B) or (C) above, for any breach, violation, termination, default, creation, acceleration or change that would not, individually or in the aggregate, reasonably be expected to prevent or materially delay the ability of ProMedica to consummate the Transactions.
- (d) <u>Capitalization of ProMedica; Financial Statements</u>. ProMedica Parent and Purchaser are each non-profit Ohio corporations that are managed by their respective duly elected Board of Trustees. ProMedica has, and at the Closing will have, available sufficient funds to pay the ProMedica Plan Contribution and pay all other cash amounts payable pursuant to this

Agreement. ProMedica expressly acknowledges and agrees that its obligations hereunder, including its obligations to consummate the transactions contemplated hereby, are not subject to, or conditioned on, receipt of financing. ProMedica Parent has provided to the Debtor the audited consolidated balance sheets of ProMedica Parent and its Subsidiaries as of December 31, 2017, and the related audited consolidated statements of operations, statements of comprehensive (loss) income, statements of cash flows and statements of equity for the twelve (12)-month periods then ended (the "ProMedica Audited Financial Statements"). Each of the balance sheets, statements of operations, statements of comprehensive loss, statements of cash flows and statements of equity included in the ProMedica Audited Financial Statements (including any related notes thereto) fairly presents, in all material respects, the financial condition, results of operations and cash flows, as the case may be, of ProMedica Parent and its Subsidiaries as of the dates and for the periods set forth therein, as applicable, in each case in accordance with GAAP, consistently applied during the periods involved, except as specifically noted therein.

- (e) <u>Proceedings</u>. As of the date of this Agreement, other than the Bankruptcy Case and chapter 11 proceedings for HCR III and/or its Subsidiaries, there are no Proceedings pending or, to the knowledge of the officers of ProMedica, threatened against ProMedica that seek to enjoin, or would reasonably be expected to have the effect of preventing, making illegal, materially delaying or otherwise interfering with the ability of ProMedica to consummate the Transactions.
- No Other Company Representations or Warranties. Except for the (f) representations and warranties set forth in Section 2.1, ProMedica hereby acknowledges and agrees that neither the Debtor nor any of its Subsidiaries, nor any of their respective stockholders, Affiliates or Representatives, has made or is making any other express or implied representation or warranty with respect to the Debtor or any of its Subsidiaries or their respective businesses or operations, including with respect to any information provided or made available to ProMedica. Subject to Section 8.1(a), neither the Debtor nor any of its Subsidiaries, nor any of their respective stockholders, Affiliates or Representatives, will have or be subject to any liability or indemnification obligation to ProMedica resulting from the delivery, dissemination or any other distribution to ProMedica, or the use by ProMedica, of any such information provided or made available to them by the Debtor or any of its Subsidiaries, or any of their respective stockholders, Affiliates or Representatives, including any information, documents, estimates, projections, forecasts or other forward-looking information, business plans or other material provided or made available to ProMedica or any of their Affiliates or Representatives, in anticipation or contemplation of any of the Transactions.
- (g) Non-Reliance on Debtor Estimates, Projections, Forecasts, Forward-Looking Statements and Business Plans. In connection with the due diligence investigation of the Debtor by ProMedica, ProMedica has received and may continue to receive from the Debtor certain estimates, projections, forecasts and other forward-looking information, as well as certain business plan information, regarding the Debtor, its Subsidiaries and their respective businesses and operations. ProMedica hereby acknowledges that there are uncertainties inherent in attempting to make such estimates, projections, forecasts and other forward-looking statements, as well as in such business plans, with which ProMedica is familiar, that ProMedica is taking full responsibility for making their own evaluation of the adequacy and accuracy of all estimates, projections, forecasts and other forward-looking information, as well as such business

plans, so furnished to them (including the reasonableness of the assumptions underlying such estimates, projections, forecasts, forward-looking information or business plans), and that, subject to Section 8.1(a), ProMedica will have no claim against the Debtor or any of its Subsidiaries, or any of their respective stockholders, Affiliates or Representatives, with respect thereto. Accordingly, ProMedica hereby acknowledges that neither the Debtor nor any of its Subsidiaries, nor any of their respective stockholders, Affiliates or Representatives, has made or is making any representation or warranty with respect to such estimates, projections, forecasts, forward-looking statements or business plans (including the reasonableness of the assumptions underlying such estimates, projections, forecasts, forward-looking statements or business plans).

- 2.3 <u>Representations and Warranties of QCP</u>. QCP hereby represents and warrants that:
- (a) Organization, Good Standing and Qualification. QCP is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing (with respect to jurisdictions that recognize the concept of good standing) as a foreign corporation or other entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so organized, qualified or in such good standing, or to have such power or authority, would not, individually or in the aggregate, reasonably be expected to prevent, materially delay or impair the ability of QCP to consummate the Transactions.
- (b) <u>Corporate Authority</u>. QCP has all requisite corporate or similar power and authority and has taken all corporate or similar action necessary in order to execute, deliver and perform its obligations under this Agreement and to consummate the Transactions. This Agreement has been duly executed and delivered by QCP, and is a valid and binding agreement of QCP, enforceable against QCP, in accordance with its terms, subject to the Bankruptcy and Equity Exception.

ARTICLE III

Bankruptcy Matters

3.1 Case Management.

In addition to any other obligations hereunder, including under Article IV, from and after the date of this Agreement, the Debtor shall deliver an updated Case Budget to QCP and ProMedica from time to time as reasonably requested by QCP or ProMedica. The "<u>Case Budget</u>" shall mean a budget for the Bankruptcy Case prepared in good faith by the Debtor in consultation with QCP and ProMedica based on the 13-Week Cash Budget, which shall include (i) expected Closing Date payments and (ii) funding for the Bankruptcy Case and Closing Date payments.

3.2 ProMedica Approvals.

ProMedica shall use best efforts to take or cause to be taken all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on their part under this

Agreement and applicable Laws to consummate and make effective the Transactions as soon as practicable, including all necessary notices, reports and other filings and to obtain as promptly as practicable all consents, registrations, approvals, permits and authorizations necessary or advisable to be obtained from any third party and/or any Governmental Entity to permit ProMedica to consummate the Plan Acquisition and operate the Leased Facilities (the "ProMedica Approvals").

3.3 Chapter 11 Obligations.

- (a) The Debtor shall file the Amended Plan within five (5) Business Days of the date of this Agreement.
- (b) The Debtor shall file the Approval Motion within ten (10) days of this Agreement and shall use reasonable best efforts to obtain entry of the Approval Order within thirty (30) days thereafter. The "Approval Motion" is a motion seeking entry of an order by the Bankruptcy Court (the "Approval Order") (i) authorizing the Debtor to enter into this Agreement, (ii) approving the supplemental disclosure, voting materials and solicitation procedures regarding the Amended Plan and (iii) establishing the procedures and a hearing date for consideration of entry of the Confirmation Order; provided, however, that in the event the Bankruptcy Court does not grant the relief requested in clause (i) pursuant to the Approval Order, the Debtor shall seek such relief pursuant to the Confirmation Order.
- The Debtor shall use reasonable best efforts to pursue entry of the Confirmation Order by the Bankruptcy Court as promptly as practicable but not later than sixty-five (65) days after the date of this Agreement. For purposes of this Agreement, (i) the "Confirmation Order" shall mean an Order of the Bankruptcy Court, reasonably acceptable to the Debtor, QCP, ProMedica, and JV that, among other things, (A) authorizes the Debtor to enter into this Agreement (to the extent such authorization has not previously been obtained), (B) finds that the Amended Plan is feasible, that adequate notice has been provided to all parties-in-interest in the Bankruptcy Case and that the requirements under the Bankruptcy Code for classification of claims, solicitation of votes, and confirmation of a plan have been satisfied, (C) confirms the Amended Plan, (D) approves this Agreement and all of its terms, and authorizes and directs the Debtor to perform its obligations under this Agreement, (E) approves the transactions contemplated by the Amended Plan and this Agreement, (F) exculpates and releases the members of the Debtor Board and officers of the Debtor from liability with respect to the negotiation, execution and delivery of this Agreement and the Debtor's performance of its obligations hereunder prior to the Closing (or termination of this Agreement in accordance with its terms, as applicable) and (G) approves the releases, exculpations and discharge contemplated by the Amended Plan and enjoins parties from asserting claims subject to such releases, exculpations and/or discharge.
- (d) In the event the entry of the Confirmation Order shall be appealed, the Debtor and ProMedica shall use their respective reasonable best efforts to defend such appeal, including opposing any stay requested in connection therewith.
- (e) Each document filed with the Bankruptcy Court as a supplement to the Amended Plan, and all exhibits, schedules and annexes thereto, shall be consistent with the terms

hereof and otherwise reasonably satisfactory in form and substance to ProMedica and QCP; provided that the Charter and the By-Laws shall be satisfactory to ProMedica in its sole discretion and in compliance with Section 4.7; provided further that any document filed with the Bankruptcy Court as a supplement to the Amended Plan that shall only be effective from and after the Effective Date (as defined in the Amended Plan) and as to which QCP shall not have any rights or obligations (including, without limitation, the Implementation Memorandum and the New Master Lease) shall be satisfactory to ProMedica in its sole discretion, subject to the Debtor's reasonable approval (not to be unreasonably withheld, conditioned or delayed).

- (f) No later than thirty (30) days after execution of this Agreement, ProMedica shall provide the Debtor with an implementation memorandum (the "<u>Implementation</u> <u>Memorandum</u>") detailing the sequence and steps pursuant to which the Transactions will be implemented on the Effective Date of the Amended Plan; <u>provided</u>, <u>however</u>, that nothing in the Implementation Memorandum shall delay or impede ProMedica's obligation to obtain required regulatory approval or adversely impact the distributions to be made to any Holder of a Claim or Interest under the Amended Plan; <u>provided further</u>, <u>however</u>, that the Implementation Memorandum shall be subject to the Debtor's reasonable approval (not to be unreasonably withheld, conditioned, or delayed).
- (g) The Debtor shall not file any motion or pleading or make any statement on the record before the Bankruptcy Court that is inconsistent with the Restructuring Documents or this Agreement or that reasonably could be expected to prevent, delay or impede the successful implementation of the Transactions and the Restructuring Documents.
- (h) The Amended Plan, the Confirmation Order, and the Disclosure Statement (collectively, the "**Restructuring Documents**") may not be amended or modified unless such amendments or modifications are (i) not inconsistent with the terms and conditions of this Agreement and the Amended Plan and (ii) consented to in writing by ProMedica and QCP, such consent not to be unreasonably withheld, conditioned or delayed.
- (i) ProMedica shall, and shall cause each of their Subsidiaries to, (i) support, and take all reasonable actions necessary or reasonably requested by the Debtor to facilitate, the solicitation, confirmation and consummation of the Amended Plan and the transactions contemplated by the Amended Plan; (ii) not take any other action, directly or indirectly, that could prevent, interfere with, delay or impede the solicitation of votes in connection with the Amended Plan or the confirmation or consummation of the Amended Plan; (iii) not object to or otherwise commence any proceeding or take any action opposing the Amended Plan or Disclosure Statement; and (iv) otherwise use its reasonable best efforts to assist the Debtor in obtaining entry of the Confirmation Order by the Bankruptcy Court within the deadline set forth in Section 3.3(c) above.
- (j) At the Closing, ProMedica shall make available, or cause to be made available, to the Debtor such amounts as may be necessary to enable the Debtor to fund or pay all of the Debtor's obligations required to be funded or paid under the Amended Plan or this Agreement (to the extent the Debtor does not have sufficient funds to fund or pay such obligations when due).

- (k) If an Eligible Employee does not execute and deliver a Separation Agreement, or the Former CEO does not execute and deliver the Former CEO Settlement Agreement, in each case prior to the entry of the Confirmation Order, the Debtor shall, at the request of QCP and ProMedica, promptly file with the Bankruptcy Court a motion to reject the Employment Agreement of such Eligible Employee or the Former CEO, as applicable, as of the Closing Date, and shall use reasonable best efforts to pursue entry of an order by the Bankruptcy Court prior to the Closing Date approving such rejection.
- debtor-in-possession financing except with the prior written consent of ProMedica and QCP, not to be unreasonably withheld, conditioned or delayed; it being understood that, for the avoidance of doubt, the granting of security as contemplated by the Centerbridge Facility, or any customary and reasonable adequate protection to be provided to the lenders under the Centerbridge Facility as part of the Bankruptcy Case (other than as a consequence of secured debtor-in-possession financing), shall not be considered a violation of this Section 3.4.
- 3.5 Release of Non-Debtor Subsidiary Claims. QCP, each Lessor, ProMedica and Debtor hereby covenant and agree that, on the Effective Date, for good and valuable consideration, to the fullest extent permissible under applicable law, (a) each Subsidiary of the Debtor shall be deemed to have completely and forever released, waived, and discharged unconditionally each of the shareholders or other equity holders of the Debtor that is not an employee of the Debtor and each director of the Debtor that is not an employee of the Debtor, and any Person claiming by or through any of the foregoing Persons (including their respective officers, directors, managers, shareholders, partners, employees, equity holders, members, and professionals) (the "Non-Debtor Subsidiary Released Parties"), and QCP, ProMedica, and each Lessor shall be deemed to have completely and forever released, waived, and discharged unconditionally each Non-Debtor Subsidiary Released Party, of and from any and all Claims (as such term is defined in the Amended Plan) and all other obligations, suits, judgments, damages, debts, rights, remedies, causes of action and liabilities of any nature whatsoever (including, without limitation, those arising under the Bankruptcy Code), whether liquidated or unliquidated, fixed or contingent, direct or derivative, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are or may be based in whole or part on any act, omission, transaction, event or other circumstance taking place or existing on or prior to the Effective Date (including prior to the Petition Date) in connection with or related to the Debtor, the reorganized Debtor, their respective Subsidiaries, assets, property and estates, the Bankruptcy Case or the Restructuring Documents and (b) following the Closing, promptly upon the written request of any Non-Debtor Subsidiary Released Party, Debtor shall cause each Subsidiary of Debtor to, ProMedica shall, and JV shall cause each Lessor to, execute and deliver to such Non-Debtor Subsidiary Released Party a written instrument evidencing such release.

ARTICLE IV

Covenants

4.1 <u>Interim Operations</u>.

- (a) The Debtor covenants and agrees as to itself and its Subsidiaries that, after the date of this Agreement and prior to the Closing, except (i) as otherwise expressly required or contemplated by this Agreement or the Original Plan Sponsor Agreement, (ii) as QCP and ProMedica may approve in writing (such approval not to be unreasonably withheld, conditioned or delayed), (iii) as otherwise required by applicable Laws or (iv) as set forth in Section 4.1 of the Debtor Disclosure Letter, the Business shall be conducted in the ordinary and usual course and, to the extent consistent therewith, the Debtor and its Subsidiaries shall use their respective commercially reasonable efforts to preserve their business organizations intact, preserve governmental licenses, permits, consents, approvals, authorizations and qualifications and maintain existing relations and goodwill with Governmental Entities, customers, suppliers, distributors, creditors, lessors, employees and business associates, and keep available the services of its and its Subsidiaries' present employees and agents. Without limiting the generality of, and in furtherance of, the foregoing, from the date of this Agreement until the Closing, except (i) as otherwise expressly required or contemplated by this Agreement or the Original Plan Sponsor Agreement, (ii) as QCP and ProMedica may approve in writing (such approval not to be unreasonably withheld, conditioned or delayed), (iii) as otherwise required by applicable Laws or (iv) as set forth in Section 4.1 of the Debtor Disclosure Letter, the Debtor will not and will not permit any of its Subsidiaries to:
- (i) adopt or propose any change in its certificate of incorporation or by-laws or other applicable governing instruments;
- (ii) merge or consolidate the Debtor or any of its Subsidiaries with any other Person, or restructure, reorganize or completely or partially liquidate the Debtor or any of its Subsidiaries or otherwise enter into any agreements providing for the sale of their respective material assets, operations or business (other than (A) the sale or disposition of obsolete or worn-out assets in the ordinary course of business and (B) sales and dispositions of such facilities and related assets as are contemplated by the SNF/AL Remarketing Process (as defined in the Original Plan Sponsor Agreement));
- (iii) acquire assets outside of the ordinary course of business from any other Person with a value or purchase price in the aggregate in excess of \$1,000,000 in any transaction or series of related transactions, other than acquisitions pursuant to Contracts in effect as of the date of this Agreement which have been provided to ProMedica and QCP prior to the date of this Agreement;
- (iv) acquire any corporation, partnership or other business organization or division thereof or collection of assets constituting all or substantially all of a business or business unit, whether by merger or consolidation, purchase of substantial assets or equity interest or any other manner, from any other Person;

- (v) issue, sell, pledge, dispose of, grant, transfer, encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer or encumbrance of, any shares of capital stock of the Debtor or any of its Subsidiaries (other than the issuance of shares by a wholly-owned Subsidiary of the Debtor to the Debtor or another wholly-owned Subsidiary), or securities convertible or exchangeable into or exercisable for any shares of such capital stock, or any options, warrants or other rights of any kind to acquire any shares of such capital stock or such convertible or exchangeable securities, other than such dispositions as are contemplated by the SNF/AL Remarketing Process;
- (vi) create or incur any Lien securing indebtedness for borrowed money (other than a Lien currently provided for under the Centerbridge Facility, any Permitted Lien (other than a Permitted Lien under clause (iv) of such definition) and/or the grant of any cash collateral in respect of letters of credit issued in respect of, or otherwise securing, ordinary course operating liabilities) on any assets of the Debtor or any of its Subsidiaries having a value in excess of \$1,000,000 in the aggregate;
- (vii) make any loans, advances, guarantees or capital contributions to or investments in any Person (other than the Debtor or any direct or indirect wholly-owned Subsidiary of the Debtor);
- (viii) declare, set aside, make or pay any dividend or other distribution, payable in cash, stock, property or otherwise, with respect to any of its capital stock (except for dividends paid by any direct or indirect wholly-owned Subsidiary to the Debtor or to any other direct or indirect wholly-owned Subsidiary) or enter into any agreement with respect to the voting of its capital stock (other than the Original Restructuring Support Agreement and the Alternative Restructuring Support Agreement);
- (ix) reclassify, split, combine, subdivide or redeem, purchase or otherwise acquire, directly or indirectly, any of its capital stock or securities convertible or exchangeable into or exercisable for any shares of its capital stock;
- (x) incur any indebtedness for borrowed money (which, for the avoidance of doubt, shall not include obligations in respect of cash-collateralized letters of credit issued in respect of, or other grants of cash collateral securing, ordinary course operating liabilities) or guarantee such indebtedness of another Person, or issue or sell any debt securities or warrants or other rights to acquire any debt security of the Debtor or any of its Subsidiaries, except for indebtedness for borrowed money incurred in the ordinary course of business consistent with past practice (A) not to exceed \$2,000,000 in the aggregate, (B) guarantees incurred in compliance with this Section 4.1 by the Debtor of indebtedness of wholly-owned Subsidiaries of the Debtor or (C) indebtedness owed to the Debtor or another wholly-owned Subsidiary of the Debtor;
- (xi) except as set forth in the capital expenditures budget set forth in Section 4.1(a)(xi) of the Debtor Disclosure Letter, make or authorize any capital expenditure in excess of \$2,000,000 in the aggregate, excluding any capital expenditure required by any Contract set forth on Section 4.1(a)(xi) of the Debtor Disclosure Letter or capital expenditure determined in good faith by the Debtor Board to be required for (A) the protection of, or to avoid injury to, any

Person, (B) the care or safety of any patient under the care of any facility operated by the Debtor or any of its Subsidiaries or (C) compliance with Law;

- (xii) enter into any Contract that would have been a Material Contract had it been entered into prior to this Agreement, other than such Contracts as are contemplated by the SNF/AL Remarketing Process;
- (xiii) make any material changes with respect to material accounting policies or procedures, except as required by changes in applicable Law or GAAP;
- (xiv) settle any litigation or other Proceeding brought against the Debtor or its Subsidiaries by a Governmental Entity (A) for an amount in excess of \$100,000 individually or \$1,000,000 in the aggregate for all such Proceedings (other than any resolution of claims processing for government reimbursement in the ordinary course of business, and excluding recoupment actions) or (B) in a manner that would impose any restrictions on its assets, operations or businesses or result in any injunction or equitable relief against the Debtor or any of its Subsidiaries:
- (xv) settle any Proceeding other than against or brought by a Governmental Entity, (A) for an amount in excess of \$500,000 individually or \$8,000,000 in the aggregate for all such Proceedings in any one-calendar-month period (in each case, with respect to Proceedings in the state of Pennsylvania, net of applicable insurance proceeds) or (B) in a manner that would impose any restrictions on its assets, operations or businesses or result in any injunction or equitable relief against the Debtor or any of its Subsidiaries;
- (xvi) amend, modify or terminate any Material Contract, including the Centerbridge Facility, in a manner adverse to the Debtor or its Subsidiaries;
- (xvii) (A) change in any material respect any material method of accounting of the Debtor or its Subsidiaries for Tax purposes; (B) enter into any agreement with any Governmental Entity (including a "closing agreement" under Code Section 7121) with respect to any material Tax or Tax Returns of the Debtor or its Subsidiaries; (C) surrender a right of the Debtor or its Subsidiaries to a material Tax refund; (D) change an accounting period of the Debtor or its Subsidiaries with respect to any material Tax; (E) file an amended Tax Return; (F) change or revoke any material election with respect to Taxes; (G) make any material election with respect to Taxes that is inconsistent with past practice; (H) file any Tax Return that is inconsistent with past practice; or (I) consent to any extension or waiver of the limitations period applicable to any material Tax claim or assessment (other than in the ordinary course of business);
- (xviii) transfer, sell, lease, license, mortgage, pledge, divest or otherwise dispose of any material tangible or intangible assets (including Intellectual Property Rights), licenses, operations, rights, product lines, businesses or interests therein of the Debtor or its Subsidiaries, including the capital stock of any of its Subsidiaries, except (A) in connection with services provided in the ordinary course of business and sales or other dispositions of obsolete or worn-out assets, (B) sales, leases, licenses, divestitures, cancellations, abandonments, lapses, expirations or other dispositions of assets with a fair market value not in excess of \$500,000 in the aggregate, (C) pursuant to Contracts in effect prior to the date of this Agreement and (D) such

sales, leases, licenses, divestitures, cancellations, abandonments, lapses, expirations or other dispositions of assets as are contemplated by the SNF/AL Remarketing Process;

(xix) (A) enter into, adopt, amend in any material respect or terminate any Company Plan (other than entry into any new employment agreement with any individual whose hiring is not restricted by, or who is otherwise hired in accordance with, clause (H) below), (B) increase or accelerate the compensation, bonus, pension, welfare, fringe or other benefits, severance or termination pay of any director, officer or employee of the Debtor or any of its Subsidiaries, (C) grant any new awards, or amend or modify the terms of any outstanding awards, under any Company Plan (other than grants of any new awards to any individual whose hiring is not restricted by, or who is otherwise hired in accordance with, clause (H) below), (D) take any action to accelerate the vesting or lapsing of restrictions or payment, or fund or in any other way secure the payment, of compensation or benefits under any Company Plan, (E) change any actuarial or other assumptions used to calculate funding obligations with respect to any Company Plan that is required by applicable Law to be funded or change the manner in which contributions to such plans are made or the basis on which such contributions are determined, except as may be required by the terms of any existing Company Plan set forth on Section 2.1(h)(i) of the Debtor Disclosure Letter or GAAP, (F) forgive any loans or issue any loans (other than routine travel advances issued in the ordinary course of business) to any director, officer or employee of the Debtor or any of its Subsidiaries, (G) terminate the employment of any officer of the Debtor or its Subsidiaries other than for "cause", (H) hire (x) any officer of the Debtor or its Subsidiaries with a title of Vice President or higher or (y) any employee of the Debtor or its Subsidiaries with aggregate annual base salary and target bonus of more than \$250,000, except, in the case of the foregoing clauses (x) and (y), to the extent jointly determined by the Chief Restructuring Officer of the Debtor ("CRO") and the Debtor Board in their reasonable business judgment in good faith necessary in the interests of patient care (any such individual described in the foregoing clauses (x) and (y) and hired to replace any such employee, a "New Hire"), provided that (i) any such officer New Hire (and his or her terms and conditions of employment, including any base and target incentive compensation) hired pursuant to this clause (H) shall be reasonably acceptable to ProMedica and QCP and (ii) the terms and conditions of employment of any New Hire that is not an officer, including base and target incentive compensation, shall be subject to notice and consultation with ProMedica and QCP, or (I) make any incentive payment or payment in respect of severance or any nonqualified deferred compensation entitlement to any current or former director, officer or employee of the Debtor or its Subsidiaries (including making any payments to any rabbi trust or taking any action that would cause the trustee of any rabbi trust to make payments to any current or former director, officer or employee of the Debtor or its Subsidiaries), except, with respect to clause (I) payment of any nondiscretionary incentive payments under existing Company Plans, nondiscretionary severance payments under existing Company Plans, and nondiscretionary payments of nonqualified deferred compensation (other than as set forth on Section 4.1(a)(xix)(I) of the Debtor Disclosure Letter) or as otherwise required by applicable Law; provided that payment in respect of any severance or nonqualified deferred compensation amount in excess of \$200,000 shall be subject to prior notice and consultation with ProMedica and QCP and, with respect to clauses (A) through (H) above, (1) amendments to welfare plans in the ordinary course of business, consistent with past practices that do not materially increase the costs of such welfare plans, (2) with respect to any hourly employees and salaried facility-level employees of the Debtor or its Subsidiaries, and any other employees of the Debtor or its Subsidiaries whose annual base salary does not exceed \$150,000, increases in compensation in the

ordinary course materially consistent with the Debtor's 2018 operating budget or otherwise as reasonably determined by the CRO, in consultation with ProMedica and QCP, to be necessary to respond to market demand, (3) with respect to each other employee of the Debtor or its Subsidiaries whose annual base salary exceeds \$150,000 (other than any Eligible Employee), increases in compensation in the ordinary course of business consistent with past practice that do not exceed 1.5% of the aggregate annual base salaries of such other employees or 7.5% of the annual base salary for any individual and (4) as required pursuant to existing Company Plans, or as otherwise required by applicable Law;

- (xx) become a party to, establish, adopt, amend, commence participation in or terminate any collective bargaining agreement or other agreement with a labor union, works council or similar organization;
- (xxi) enter into any Contract adversely affecting in any material respect the Debtor's or any of its Subsidiaries' ability to use or otherwise exploit any material Intellectual Property Rights;
- (xxii) fail to use commercially reasonable efforts to keep in full force the material Insurance Policies under substantially the same levels of coverage as the current policies of the Debtor and its Subsidiaries;
- (xxiii) change in any material respect any of the Debtor's or its Subsidiaries' material policies or procedures for or timing of the collection of accounts receivable (or any other trade receivables), payment of accounts payable (or any other trade payables), billing of its customers, pricing and payment terms, cash collections, cash payments or terms with suppliers, in each case, other than changes required by suppliers, vendors and service providers;
- (xxiv) dismiss the QCP Consultants other than in accordance with Section 5.1(b);
- (xxv) modify or amend in any respect any Contract pursuant to which HCR III or any of its Subsidiaries currently subleases real property to any other Subsidiary of the Debtor; or
 - (xxvi) agree, authorize or commit to do any of the foregoing.
- (b) ProMedica shall not knowingly take or permit any of its Subsidiaries to take any action that is reasonably likely to prevent or materially impede the consummation of the Transactions.
 - 4.2 <u>Filings; Other Actions; Notification</u>.
- (a) <u>Cooperation</u>. Subject to the terms and conditions set forth in this Agreement, the Debtor, JV and ProMedica (and, to the extent reasonably required, QCP) shall cooperate with each other and use (and shall cause their respective Subsidiaries to use) their respective reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on their part under this Agreement and applicable Laws to consummate and make effective the Transactions as soon as practicable,

including preparing and filing as promptly as practicable all documentation to effect all necessary notices, reports and other filings and to obtain as promptly as practicable all consents, registrations, approvals, permits and authorizations necessary or advisable to be obtained from any third party and/or any Governmental Entity (including the Governmental Approvals and the ProMedica Approvals) and, to the extent applicable, complying with the requirements of the HSR Act. Subject to applicable Laws relating to the exchange of information, the parties shall cooperate on all matters with respect to obtaining the Governmental Approvals and making of filings with any Governmental Entity in connection with this Section 4.2. Each of the Lessors, ProMedica, and the Debtor shall (i) in coordination with each other, file as promptly as practicable after the execution hereof, all applications and documents required, necessary or advisable to obtain the Governmental Approvals and comply with the applicable requirements of the HSR Act as soon as practicable, (ii) submit to customary background checks in connection therewith, (iii) respond promptly to any additional requests from Governmental Entities and (iv) otherwise use its reasonable best efforts to obtain all Governmental Approvals and comply with the applicable requirements of the HSR Act as promptly as practicable (and in any event prior to the Termination Date). The Lessors, ProMedica and the Debtor shall have the right to review in advance (and shall deliver to QCP in advance) and, to the extent practicable, each will consult with the other (and of QCP) on and consider in good faith the views of the other (and of QCP) in connection with, all of the information relating to the Lessors, ProMedica and the Debtor, as the case may be, and any of their respective Subsidiaries, that appears in any filing made with, or written materials submitted to, any third party and/or any Governmental Entity in connection with the Transactions. In exercising the foregoing rights, each of the Lessors, Debtor, and ProMedica shall act reasonably and as promptly as practicable.

- (b) <u>Information; Meetings</u>. The Debtor and ProMedica each shall (and the JV shall cause the Lessors), upon the reasonable request by the other, furnish the other with all information concerning itself, its Subsidiaries, directors, officers and stockholders and such other matters as may be reasonably necessary or advisable in connection with any statement, filing, notice or application made by or on behalf of the Lessors, ProMedica or the Debtor or any of their respective Subsidiaries to any third party and/or any Governmental Entity in connection with the Transactions.
- Status. Subject to applicable Laws and as required by any Governmental (c) Entity, the Debtor, the Lessors and ProMedica each shall keep the other and QCP reasonably apprised of the status of matters relating to completion of the Transactions to the extent such party is aware of such information, including promptly furnishing the other and QCP with copies of notices or other communications received by the Lessors, ProMedica or the Debtor or any of their respective Subsidiaries, as the case may be, from any third party and/or any Governmental Entity with respect to the Transactions. The JV and ProMedica shall give prompt written notice to the Debtor and QCP of any change, fact or condition of which the Lessors and or ProMedica becomes aware that is reasonably expected to result in any failure of any condition to Debtor's obligations to effect the Transactions, and the Debtor shall give prompt written notice to the Lessors, ProMedica and QCP of any change, fact or condition of which the Debtor becomes aware that is reasonably expected to result in any failure of any condition to ProMedica's obligations to effect the Transactions or to materially affect the Business. Neither the Debtor nor the Lessors or ProMedica shall permit any of their respective officers or any other representatives or agents to participate in any meeting with any Governmental Entity in respect of any filings, investigation or

other inquiry relating to the Transactions unless it consults with the other party and QCP in advance and, to the extent permitted by such Governmental Entity, gives the other parties and QCP the opportunity to attend and participate thereat.

- Access and Records. Subject to applicable Law, upon reasonable notice, 4.3 the Debtor shall (and shall cause its Subsidiaries to) afford ProMedica's and the Lessors' officers, directors, employees, agents, counsel, accountants, investment bankers, financing sources and other authorized representatives (together, "Representatives") reasonable access, during normal business hours throughout the period prior to the Closing Date, to its employees, properties, books, contracts and records, and, upon ProMedica's or any Lessor's written request, will use reasonable efforts to provide ProMedica or the Lessors with access to suppliers, creditors and third parties with whom the Debtor or its Subsidiaries does business or by whom there have been asserted any claim, and, during such period, the Debtor shall (and shall cause its Subsidiaries to) use its reasonable best efforts to furnish promptly to ProMedica and the Lessors all information concerning its business, properties, personnel, suppliers, creditors and third parties with whom the Debtor or its Subsidiaries does business as may reasonably be requested, provided that no investigation pursuant to this Section 4.3 shall affect or be deemed to modify any representation or warranty made by the Debtor herein, and provided, further, that the foregoing shall not require the Debtor (i) to permit any inspection, or to disclose any information, that in the reasonable judgment of the Debtor would result in the disclosure of any trade secrets of third parties or violate any of its obligations with respect to confidentiality if the Debtor shall have used its reasonable best efforts to obtain the consent of such third party to such inspection or disclosure or (ii) to disclose any privileged information of the Debtor or any of its Subsidiaries. The Debtor shall continue to deliver its financial information and other reports required pursuant to the terms of the Master Lease. The Debtor shall cooperate to provide ProMedica with access and information reasonably requested by ProMedica to facilitate the consolidation of the Debtor's financial results with and into those of ProMedica following Closing, including establishment by ProMedica of accounting and financial reporting systems, systems of internal controls over financial reporting and disclosure controls as defined in the Exchange Act.
- 4.4 Publicity. ProMedica and Welltower, Inc. (and/or their affiliates) shall be authorized to issue an initial joint press release regarding the Transactions; provided, however, that such press release shall be subject to the consent of the Debtor, which consent shall not be unreasonably withheld, conditioned or delayed; provided further that ProMedica and Welltower, Inc. shall consult with QCP as to the content of such press release. Thereafter, no party shall issue any press release or make any public announcement concerning the Transactions without the prior written consent of the other parties hereto except as may be required by Law, including rules promulgated by the Securities and Exchange Commission (the "SEC") or by obligations pursuant to any rules of any national securities exchange. To the extent any party is required to make any press release or public announcement by Law or by obligations pursuant to any rules of any national securities exchange, such party shall (i) to the extent practicable, consult with the other parties hereto prior to issuing any press releases or otherwise making public announcements with respect to the Transactions and prior to making any filings with any third party and/or any Governmental Entity with respect thereto, except as may be required by Law or by obligations pursuant to any rules of any national securities exchange, (ii) to the extent practicable, provide the other parties hereto with a reasonable opportunity to review and comment on such press release,

announcement or filing and (iii) consider in good faith any comments of the other parties with respect to such press release, announcement or filing.

4.5 <u>Employee Matters</u>.

- (a) Prior to making any broad-based written or oral communications to officers and employees of the Debtor or any of its Subsidiaries pertaining to compensation or benefit matters that are affected by the Transactions, the Debtor shall provide ProMedica and QCP with a copy of the intended communication, ProMedica and QCP each shall have a reasonable period of time to review and comment on the communication, and the Debtor shall consider any such comments in good faith.
- (b) ProMedica shall use all reasonable endeavors to ensure that any employee benefit plans which the employees of the Debtor and its Subsidiaries are entitled to participate in to take into account for purposes of eligibility, vesting and benefit accrual thereunder (other than any benefit accrual under any defined benefit pension plans), service by employees of the Debtor and its Subsidiaries as if such service were with ProMedica, to the same extent such service was credited under a comparable plan of the Debtor or any of its Subsidiaries (except to the extent it would result in a duplication of benefits).
- (c) Nothing contained in this Agreement is intended to: (i) be treated as an amendment of any particular Company Plan; (ii) prevent QCP, ProMedica, the Debtor or any of their Affiliates from amending or terminating any of their benefit plans (or, after the Closing Date, any Company Plan) in accordance with their terms; (iii) prevent QCP, ProMedica, the Debtor or any of its Affiliates, after the Closing Date, from terminating the employment of any employee of the Debtor; or (iv) create any third-party beneficiary rights in any employee of the Debtor, any beneficiary or dependent thereof, or any collective bargaining representative thereof, with respect to the compensation, terms and conditions of employment and/or benefits that may be provided to any such employee by QCP, ProMedica, the Debtor or any of their Affiliates or under any benefit plan which QCP, ProMedica, the Debtor or any of their Affiliates may maintain.
- (d) <u>Settlement Agreement</u>. If the Former CEO Settlement Agreement is executed and delivered by the parties thereto, from the date of execution and delivery through the Closing Date, the Debtor shall comply with the terms of the Former CEO Settlement Agreement and shall not amend, modify or alter such Former CEO Settlement Agreement without the prior written consent of ProMedica and QCP.

(e) 280G Vote.

- (i) At least two (2) business days prior to the Closing Date, the Debtor shall submit for approval by its stockholders, in conformance with Section 280G of the Code and the regulations thereunder (the "280G Stockholder Vote"), any payments that would reasonably be expected to constitute a "parachute payment" pursuant to Section 280G of the Code (each, a "Parachute Payment").
- (ii) The Debtor shall use its commercially reasonably efforts to obtain, at least three (3) business days prior to the Closing Date, irrevocable waivers of the right to any

Parachute Payment from each of the applicable "disqualified individuals" (as defined under Section 280G of the Code and the regulations promulgated thereunder).

(iii) True and complete copies of all disclosure and documents that comprise the stockholder approval of each Parachute Payment shall be delivered to ProMedica and QCP in sufficient time to allow ProMedica and QCP to comment thereon, but no less than five (5) business days prior to the 280G Stockholder Vote, and shall reflect all reasonable comments of ProMedica and QCP thereon.

4.6 Expenses.

- (a) Whether or not the Transactions are consummated, the parties hereto shall bear their own respective expenses (including all compensation and expenses of counsel, financial advisors, consultants, actuaries and independent accountants) incurred in connection with this Agreement and the Transactions.
- (b) The Debtor and its Subsidiaries' sole financial advisor shall be Moelis & Company, and the Debtor shall not, and shall not permit its Subsidiaries to, hire any additional financial advisor. The Debtor shall not agree to any change in the financial terms of any agreement pursuant to which Moelis & Company is entitled to any fees and expenses in connection with any of the Transactions.

4.7 Indemnification; Directors' and Officers' Insurance.

- that it will indemnify, defend and hold harmless each present and former director and officer of the Debtor and its Subsidiaries (in each case, when acting in such capacity) (the "Indemnified Parties") from and against any and all costs or expenses (including reasonable attorneys' fees), judgments, fines, losses, claims, damages or liabilities (whether or not incurred in connection with any Proceeding or investigation, whether civil, criminal, administrative or investigative) arising out of or related to such Indemnified Party's service or status as an officer or director of the Debtor or one of its Subsidiaries or any action or inaction on the part of any such Person in such capacity, in each case, as of prior to the Closing, whether asserted or claimed prior to, at or after the Closing Date, to the fullest extent such Indemnified Party would be permitted to be indemnified by the Debtor or its applicable Subsidiary under applicable Law and the certificate of incorporation and by-laws (or comparable organizational documents) of the Debtor or its applicable Subsidiary in effect on the date of this Agreement; provided that the Person to whom expenses are advanced provides an undertaking to repay such advances if it is ultimately determined that such Person is not entitled to indemnification.
- (b) Prior to or as of the Closing Date, the Debtor shall (after reasonable consultation with ProMedica) and, if the Debtor is unable to, ProMedica shall cause the Debtor as of the Closing Date to obtain and fully pay the premium for "tail" insurance policies for the extension of (i) the directors' and officers' liability coverage of the existing directors' and officers' insurance policies of the Debtor and its Subsidiaries, and (ii) the existing fiduciary liability insurance policies of the Debtor and its Subsidiaries, in each case for a claims reporting or discovery period of at least six (6) years from and after the Closing Date (the "<u>Tail Period</u>") from

an insurance carrier with the same or better credit rating as the Debtor's insurance carrier as of the date hereof with respect to directors' and officers' liability insurance and fiduciary liability insurance (collectively, "D&O Insurance") with terms, conditions, retentions and limits of liability that are at least as favorable to the insureds as the Debtor's existing policies with respect to any matter claimed against a director or officer of the Debtor or any of its Subsidiaries by reason of him or her serving in such capacity that existed or occurred at or prior to the Closing Date (including in connection with this Agreement or the transactions or actions contemplated hereby); provided that in no event shall the aggregate cost of the D&O Insurance exceed during the Tail Period 300% of the current aggregate annual premiums paid by the Debtor for such insurance. If the Debtor for any reason fails to obtain such insurance policies as of the Closing Date, the Debtor shall, and ProMedica shall cause the Debtor to, continue to maintain in effect for the Tail Period the D&O Insurance in place as of the date hereof with terms, conditions, retentions and limits of liability that are at least as favorable to the insureds as provided in the Debtor's existing policies as of the date hereof, or the Debtor shall, and ProMedica shall cause the Debtor to, use reasonable best efforts to purchase comparable D&O Insurance for the Tail Period with terms, conditions, retentions and limits of liability that are at least as favorable to the insureds as provided in the Debtor's existing policies as of the date hereof; provided, however, that in no event shall the annual cost of the D&O Insurance exceed during the Tail Period 150% of the current aggregate annual premiums paid by the Debtor for such insurance; and provided, further, that if the annual premiums of such insurance coverage exceed such amount, the Debtor shall obtain a policy with the greatest coverage available for a cost not exceeding such amount. From and after the Closing Date, except as required by applicable Law, ProMedica shall cause the certificate of incorporation and by-laws (or comparable organizational documents) of each of the Debtor and its Subsidiaries to contain provisions no less favorable to the Indemnified Parties with respect to limitation of liabilities, advancement of expenses and indemnification than are set forth as of the date of this Agreement in each of such documents of the Debtor and its Subsidiaries. In the event of any Proceeding or investigation for which an Indemnified Party is entitled to indemnification pursuant to this Section 4.7, (A) Debtor shall, and ProMedica shall cause the Debtor and its Subsidiaries to, defend the Indemnified Party with respect to any such Proceeding or investigation and (B) ProMedica shall not, and shall cause the Debtor and its Subsidiaries not to, settle, compromise or consent to the entry of any judgment in any Proceeding or investigation pending or threatened in writing to which an Indemnified Party is a party (and in respect of which indemnification could be sought by such Indemnified Party hereunder) without the prior written consent of such Indemnified Party (such consent not to be unreasonably withheld, conditioned or delayed), unless such settlement, compromise or consent includes an unconditional release of such Indemnified Party from all liability arising out of such Proceeding or investigation. ProMedica agrees that all rights of the Indemnified Parties to indemnification and exculpation from liabilities for acts or omissions occurring prior to the Closing as provided in the certificate of incorporation or by-laws (or comparable organizational documents) of the Debtor or any of its Subsidiaries and any indemnification agreements entered into prior to the date hereof, including provisions relating to advancement of expenses incurred in the defense of any Proceeding or investigation, shall survive the Closing Date and shall continue in full force and effect in accordance with their terms. Such rights shall not be amended or otherwise modified in any manner that would adversely affect the rights of any of the Indemnified Parties for any acts or omissions occurring prior to the Closing without the prior written consent of each affected Indemnified Party (such consent not to be unreasonably withheld, conditioned or delayed).

- (c) Any Indemnified Party wishing to claim indemnification under paragraph (a) of this Section 4.7, upon learning of any such Proceeding or investigation, shall promptly notify ProMedica and the Debtor thereof, but the failure to so notify shall not relieve ProMedica or the Debtor of any Liability it may have to such Indemnified Party except to the extent such failure materially prejudices the indemnifying party.
- (d) If ProMedica or the Debtor or any of their respective successors or assigns (i) shall consolidate with or merge into any other corporation or entity and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) shall transfer all or substantially all of its properties and assets to any individual, corporation or other entity, then, and in each such case, proper provisions shall be made so that the successors and assigns of ProMedica or the Debtor shall assume all of the obligations set forth in this Section 4.7.
- (e) The provisions of this Section 4.7 are intended to be for the benefit of, and shall be enforceable by, each of the Indemnified Parties.
- (f) The rights of the Indemnified Parties under this Section 4.7 shall be in addition to any rights such Indemnified Parties may have under the certificate of incorporation or by-laws (or comparable organizational documents) of the Debtor or any of its Subsidiaries, or under any applicable Contracts or Laws. All rights to indemnification and exculpation from liabilities for acts or omissions occurring at or prior to the Closing Date and rights to advancement of expenses relating thereto now existing in favor of any Indemnified Party as provided in the certificate of incorporation, by-laws or comparable governing documents of the Debtor and its Subsidiaries or any indemnification agreement between such Indemnified Party and the Debtor or any of its Subsidiaries shall survive the Transactions and shall not be amended, repealed or otherwise modified in any manner that would adversely affect any right thereunder of any such Indemnified Party.
- ProMedica and the Debtor hereby acknowledge that the Indemnified Parties have or may, in the future, have certain rights to indemnification, advancement of expenses and/or insurance provided by Persons other than ProMedica, the Debtor or its Subsidiaries (collectively, the "Other Indemnitors" and, individually, an "Other Indemnitor") with respect to Proceedings or investigations that are the subject of Section 4.7(a). The Debtor hereby agrees that if any advancement or indemnification obligation is owed, at any time, to an Indemnified Party for the same Proceeding or investigation by both (i) ProMedica or the Debtor, including under Section 4.7(a), and (ii) any Other Indemnitor (whether pursuant to any certificate of incorporation, by-laws or comparable organizational documents, indemnification agreement or other agreements), the Debtor shall (and if it is unable or fails to honor its obligations under this Section 4.7, ProMedica shall) be primarily liable for indemnification and advancement of expenses to such Indemnified Parties in respect of such Proceeding or investigation, and any obligation of an Other Indemnitor to provide indemnification or advancement of expenses shall be secondary to the obligations of the Debtor (or if it is unable or fails to honor its obligations under this Section 4.7, ProMedica) under this Section 4.7. ProMedica and the Debtor irrevocably waive, relinquish and release the Other Indemnitors from any and all claims (A) against the Other Indemnitors for contribution, subrogation, indemnification or any other recovery of any kind in respect of the indemnification arrangements provided to such Indemnified Parties by such Other Indemnitors and (B) that the Indemnified Party must seek expense advancement or reimbursement, or indemnification, from

any Other Indemnitor before ProMedica and the Debtor must perform their expense advancement and reimbursement, and indemnification obligations, under this Agreement. If any Other Indemnitor pays or causes to be paid, for any reason, any amounts otherwise indemnifiable or subject to advancement under this Section 4.7, then (1) such Other Indemnitor shall be fully subrogated to all rights of the Indemnified Parties with respect to the payments actually made and (2) the Debtor shall reimburse the Other Indemnitor for the payments actually made. Notwithstanding anything to the contrary in this Agreement, for the avoidance of doubt, ProMedica's rights against the Debtor for contribution, subrogation, indemnification or any other recovery of any kind in respect of the indemnification arrangements provided by ProMedica and Debtor to the Indemnified Parties pursuant to this Section 4.7 are hereby preserved.

Confidentiality. Unless and until Closing occurs, no party hereto shall disclose any information provided to such party or its Representatives pursuant to Section 4.3 or any other provision of this Agreement, or any non-public information related to the negotiation, execution and delivery of this Agreement and the Transactions, other than (a) disclosure to such party's Representatives (provided that such Representatives agree to comply with this Section 4.8 as if they were a party hereto and that such party shall be responsible for any Representative of such party to so comply with this Section 4.8), (b) as required by Law or by obligations pursuant to any rules of any national securities exchange, the SEC or by the request of any Governmental Entity, and (c) subject to reasonable customary confidentiality arrangements reasonably acceptable to the Debtor, any Person who expresses a bona fide interest in acquiring all or part of the Business or any business entities, properties or assets of the Debtors and its Subsidiaries following the Closing, including any potential joint venture partner (a "Potential Acquiror") and actual or potential purchasers, replacement tenants, third-party operators or other partners of ProMedica with respect to the remarketing of the Leased Facilities pursuant to the SNF/AL Remarketing Process. Notwithstanding the foregoing, nothing herein shall in any way limit any existing rights of any party to the Master Lease to use or disclose information under the terms of the Master Lease.

4.9 Financing.

The Debtor shall, and shall cause its Subsidiaries to, use reasonable best efforts to provide and direct its and their respective Representatives to use their respective reasonable best efforts to provide all cooperation reasonably requested by ProMedica in connection with the arrangement, syndication and consummation of any debt financing by ProMedica or its Subsidiaries and the Debtor or any of its Subsidiaries, any equity financing of ProMedica or any minority investment in the Debtor or any of its Subsidiaries (not to exceed ten percent (10%) of the fully-diluted equity of the Debtor or such Subsidiary) including the refinancing of the indebtedness under the Centerbridge Facility, whether at the Debtor, ProMedica or any of their Subsidiaries (such financing, the "Financing"). Such cooperation shall include (i) furnishing ProMedica and its Debt Financing Sources with the Audited Financial Statements and with audited financial statements for the Debtor and its Subsidiaries for each subsequent fiscal year within fifty (50) days after the end of such fiscal year, and quarterly and interim unaudited financial statements for the Debtor and its Subsidiaries for the fiscal quarters and interim periods ended March 31, 2018, June 30, 2018 and September 30, 2018 and for each subsequent fiscal quarter within thirty-five (35) days after the end of such fiscal quarter, in each case, with comparative financial information for the equivalent period of the prior year; (ii) using reasonable

best efforts to furnish ProMedica and its Debt Financing Sources with information, audit reports, historical business and other financial data and any supplements thereto regarding the Debtor and its Subsidiaries customarily included in information memoranda and other syndication materials for revolving, term loan credit facilities, bond offering documents and equity offering documents, including assisting in preparing pro forma financial information for the Debtor and its Subsidiaries; (iii) to the extent reasonably requested by ProMedica, using reasonable best efforts to prepare carve-out audited and unaudited financial statements for the time period required or desirable to any Debt Financing Source; (iv) using reasonable best efforts to cause members of management and other senior officers to participate in a reasonable number of meetings (including one-on-one meetings or conference calls), lender presentations, due diligence sessions and sessions with rating agencies, prospective lenders and investors and other syndication and marketing activities; (v) reasonably assisting ProMedica and its Debt Financing Sources in the preparation of any syndication and offering documents and materials, including information memoranda, lender presentations, offering memoranda, registration statements, prospectuses and other marketing documents (collectively, the "Marketing Documentation") and provide and execute a customary authorization letter with respect thereto; (vi) reasonably cooperating in marketing efforts of ProMedica; (vii) reasonably assisting in the negotiation and preparation of any credit agreements, indentures, underwriting agreements, purchase agreements, pledge and security documents, mortgages, guarantees, hedging agreements and other definitive documents, a certificate of the chief financial officer of the Debtor with respect to solvency matters in a form acceptable to ProMedica and any other certificates, letters and documents (including any schedules and exhibits in connection with the foregoing) as may be reasonably requested by ProMedica; (viii) reasonably assisting in the obtaining of (A) audit reports, authorization letters, comfort letters and consents of accountants and auditors with respect to financial statements for the Debtor and its Subsidiaries for inclusion in any Marketing Documentation and (B) payoff letters, instruments of discharge and Lien terminations; (ix) providing information regarding the Debtor and its Subsidiaries required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including, without limitation, the USA Patriot Act of 2001, to the extent requested by its Debt Financing Sources at least eight (8) business days prior to the Closing; (x) reasonably assisting ProMedica in obtaining corporate, facility and debt security ratings from rating agencies; (xi) reasonably cooperating with ProMedica's legal counsel (including providing customary back-up certificates) in connection with any legal opinions that such legal counsel may be required to deliver in connection with the Financing; (xii) taking such reasonable actions as may be reasonably requested by ProMedica necessary to permit its Debt Financing Sources to evaluate the Debtor's assets and cash management policies and procedures relating thereto for the purposes of establishing collateral arrangements as of the Closing and to cooperate with other due diligence conducted by its Debt Financing Sources; (xiii) using reasonable best efforts to obtain customary evidence of authority, customary officer's certificates, good standing certificates (to the extent applicable) in the respective jurisdictions of organization of the Debtor and its Subsidiaries, customary lien searches with respect to the Debtor and its Subsidiaries and insurance certificates; (xiv) using reasonable best efforts to facilitate the granting of a security interest (or perfection thereof) in collateral by the Debtor and its Subsidiaries to secure the Financing at Closing; and (xv) if applicable, reasonably cooperating with ProMedica in connection with the payoff of the Centerbridge Facility at Closing.

- (b) The Debtor hereby consents to the use of its and its Subsidiaries' logos in connection with the Financing; <u>provided</u> that such logos are used solely in a manner that is not intended to, nor reasonably likely to, harm or disparage the Debtor or its Subsidiaries.
- (c) For purposes of this Agreement, "<u>Debt Financing Sources</u>" means the Persons that will provide or otherwise enter into agreements in connection with the Financing, together with, in each case, their Affiliates and Representatives.
- Notwithstanding anything in this Agreement to the contrary, none of the Debtor, its Subsidiaries or their respective Affiliates, or Representatives shall (i) be required to pay any commitment or other similar fee other than any such fee that may be payable by the Debtor or its Subsidiaries at the Closing, (ii) be required to incur any other liability in connection with the Financing other than any such liabilities to be incurred by the Debtor or its Subsidiaries arising after the Closing occurs or liabilities for costs and expenses to the extent required to be indemnified or reimbursed by ProMedica, (iii) otherwise be required to take any action in violation of any of the Debtor's or any of its Subsidiaries' organizational documents or that Debtor determines in good faith could reasonably be expected to violate any applicable Law or Contract, (iv) be required to waive or amend any terms of this Agreement or terms of any other Contract to which the Debtor or its Subsidiaries is a party, (v) be required to provide cooperation to the extent it would (y) cause any condition to the Closing set forth in ARTICLE VI not to be satisfied or (z) cause a breach of this Agreement, (vi) be required to deliver or obtain opinions of internal or external counsel or (vii) provide access to or disclose information where the Debtor reasonably determines that such access or disclosure could jeopardize the attorney-client privilege or contravene any Law or Contract to which it or any of its Subsidiaries is a party, provided that in the event that the Debtor or any of its Subsidiaries does not provide information in reliance on the exclusions in this subclause (vii), the Debtor and its Subsidiaries shall use commercially reasonable efforts to provide notice to ProMedica promptly upon obtaining knowledge that such information is being withheld (but solely if providing such notice would not violate such obligation of confidentiality). No director or officer of the Debtor or its Subsidiaries, acting in such capacity, shall be required to execute, deliver or enter into or perform any agreement, document or instrument, including any definitive documentation with respect to the Financing, or adopt any resolutions approving the agreements, documents and instruments pursuant to which the Financing is obtained, except with respect to the delivery of customary representation and authorization letters in connection with the Marketing Documentation. Nothing hereunder will require any officer or representative of the Debtor or any of its Subsidiaries to deliver any certificate or opinion or take any other action that would result in personal liability to such officer or representative. ProMedica shall indemnify, defend and hold harmless the Debtor and its Subsidiaries, and their respective officers, employees and Representatives, from and against any and all liabilities or losses suffered or incurred by them in connection with the arrangement of the Financing and any information utilized in connection therewith, except in the event such liabilities or losses arose out of or result from (A) the willful misconduct, gross negligence or bad faith of the Debtor and its subsidiaries or (B) historical information furnished in writing by or on behalf of the Debtor and its Subsidiaries, including financial statements. Notwithstanding anything in this paragraph, the Debtor and its Subsidiaries shall not be required to provide any such requested cooperation that shall unreasonably disrupt or interfere with the ongoing operations of the Debtor or its Subsidiaries. ProMedica shall, promptly upon request by the Debtor, reimburse the Debtor and its Subsidiaries for all reasonable and documented out-of-pocket costs and expenses incurred

by the Debtor and its Subsidiaries (including those of its Affiliates and Representatives) in connection with taking actions requested by ProMedica pursuant to this Section 4.9, which for the avoidance of doubt will not include audit fees and other expenses relating to the preparation of financial statements for the Business to the extent required by the Master Lease or otherwise historically prepared by the Debtors and its Subsidiaries. Notwithstanding anything to the contrary, the condition set forth in Section 6.2(b) as it applies to the Debtor's obligations under this Section 4.9 shall be deemed satisfied unless there has occurred a breach by Debtor of its obligations under this Section 4.9 that directly results in the Financing not being available on the Closing Date.

4.10 Acquisition Proposals.

- (a) <u>No Solicitation or Negotiation</u>. At all times during the period commencing with the execution and delivery of this Agreement and continuing until the earlier to occur of the termination of this Agreement pursuant to ARTICLE VII and the Closing, except as expressly permitted by this Section 4.10, the Debtor and its Subsidiaries and its and its Subsidiaries' respective directors, officers and employees shall not, and the Debtor shall direct its other agents and representatives not to, directly or indirectly:
- (i) initiate, solicit, propose or knowingly encourage or otherwise knowingly facilitate any inquiry or the making of any proposal or offer that constitutes or, would reasonably be expected to lead to, an Acquisition Proposal;
- (ii) engage in, continue or otherwise participate in any discussions or negotiations relating to any Acquisition Proposal or any inquiry, proposal or offer that would reasonably be expected to lead to an Acquisition Proposal (other than to state that the terms of this Section 4.10 prohibit such discussions);
- (iii) provide any non-public information or data concerning the Debtor or its Subsidiaries, or access to the Debtor or its Subsidiaries' properties, books and records to any Person, in each case, in connection with any Acquisition Proposal or any inquiry, proposal or offer that constitutes or would reasonably be expected to lead to an Acquisition Proposal;
- (iv) otherwise knowingly facilitate any effort or attempt to make an Acquisition Proposal;
- (v) approve or recommend or publicly declare advisable any Acquisition Proposal or proposal reasonably expected to lead to an Acquisition Proposal or approve or recommend, or publicly declare advisable or publicly propose to enter into, any Alternative Acquisition Agreement;
 - (vi) execute or enter into an Alternative Acquisition Agreement; or
 - (vii) agree, authorize or commit to do any of the foregoing.
- (b) <u>Notice of Acquisition Proposals</u>. The Debtor shall promptly (but, in any event, within twenty-four hours) give notice to ProMedica and QCP if (i) any bona fide inquiries, proposals or offers with respect to an Acquisition Proposal or that would reasonably be expected to

lead to an Acquisition Proposal are received by the Debtor or any of its Subsidiaries, or any of its or their respective Representatives of which the Debtor or its Subsidiaries are made aware (it being understood that the Debtor and its Subsidiaries shall instruct its and their respective Representatives to promptly make them aware of any such bona fide inquiries, proposals or offers), or (ii) any bona fide request is received by the Debtor or any of its Subsidiaries, or any of its or their respective Representatives of which the Debtor or its Subsidiaries are made aware (it being understood that the Debtor and its Subsidiaries shall instruct its and their respective Representatives to promptly make them aware of any such bona fide requests), for the Debtor or any of its Subsidiaries to engage in discussions or negotiations relating to an Acquisition Proposal or that would reasonably be expected to lead to an Acquisition Proposal, setting forth in such notice the name of the Person making any such inquiry, proposal, offer or request and the material terms and conditions of any such Acquisition Proposal or inquiry, proposal or offer and the scope of any such request (including, if applicable, correct and complete copies of any such written Acquisition Proposals, inquiries, proposals or offers, including proposed agreements, or requests (or where no such copies are available, a reasonably detailed written description thereof)), and thereafter shall keep ProMedica and QCP reasonably informed, on a current basis of the status and terms and conditions of any such Acquisition Proposals, inquiries, proposals or offers or requests (including any amendments or supplements thereto) and the status of any such discussions or negotiations, including any change in such Person's intentions as previously notified.

- (c) Existing Discussions. The Debtor (i) agrees, that as of the date hereof, it has ceased and caused to be terminated any activities, solicitations, discussions and negotiations with any Person (other than QCP and ProMedica, and their Subsidiaries) conducted prior to the date hereof with respect to any Acquisition Proposal, or proposal that would reasonably be expected to lead to an Acquisition Proposal (including access to any physical or electronic data rooms) and (ii) shall promptly (but in any event within twenty-four hours of the execution and delivery of this Agreement) (A) deliver a written notice to each such Person providing only that the Debtor is ending all discussions and negotiations with such Person with respect to any Acquisition Proposal, or proposal or transaction that would reasonably be expected to lead to an Acquisition Proposal and is requesting the prompt return or destruction of all confidential information concerning the Debtor and any of its Subsidiaries and (B) if applicable, terminate any physical and electronic data or other diligence access previously granted to such Persons.
- (d) <u>Standstill Provisions</u>. From that date of this Agreement and continuing until the earlier to occur of the termination of this Agreement pursuant to ARTICLE VII and the Closing, the Debtor shall not terminate, amend or otherwise modify or waive any provision of any confidentiality, "standstill" or similar agreement to which the Debtor or any of its Subsidiaries is a party and, in the event of any breach of such agreement of which the Debtor Board becomes aware, the Debtor shall use reasonable best efforts to enforce, to the fullest extent permitted under applicable Law, the provisions of any such agreement, including by obtaining injunctions to prevent any breaches of such agreements and to enforce specifically the terms and provisions thereof.
- (e) For purposes of this Agreement, the term "<u>Acquisition Proposal</u>" means any (i) proposal, offer, inquiry or indication of interest relating to a merger, joint venture, partnership, consolidation, dissolution, liquidation, tender offer, spin-off, share exchange, business combination or similar transaction involving the Debtor or any of its Subsidiaries or (ii)

acquisition by any Person or group (as defined under Section 13 of the Exchange Act), resulting in, or any proposal, offer, inquiry or indication of interest that if consummated would result in, any Person or group (as defined under Section 13 of the Exchange Act) becoming the beneficial owner of, directly or indirectly, in one or a series of related transactions, ten percent (10%) or more of the consolidated net revenues, net income or total assets (it being understood that total assets include equity securities of Subsidiaries) of the Debtor, other than (A) the transactions contemplated by the Original Plan Sponsor Agreement and (B) the Transactions or the disposition of those Leased Facilities set forth in Section 5.3(b) of the Debtor Disclosure Letter; and the term "Alternative Acquisition Agreement" means any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement, option agreement, joint venture agreement, partnership agreement or other similar agreement relating to any Acquisition Proposal.

(f) Notwithstanding anything to the contrary set forth herein or otherwise, nothing in this Section 4.10 shall prohibit the Debtor from cooperating with any action contemplated by Section 7.4 of the Merger Agreement taken by QCP or any of its Subsidiaries or with the exercise by QCP of any rights or remedies under the Merger Agreement

ARTICLE V

Special Covenants

- 5.1 Corporate Governance.
- (a) <u>Debtor Restructuring Committee</u>. After the date of this Agreement and prior to the Closing Date, the power and authority of the special restructuring committee of the Debtor Board (the "<u>Debtor Restructuring Committee</u>") shall not be changed, and no new or replacement member of the Debtor Restructuring Committee, the sole members of which are Sherman Edmiston III and Kevin P. Collins in their capacity as members of the Debtor Board, shall be appointed without the consent of ProMedica and QCP, such consent not to be unreasonably withheld, conditioned or delayed.
- (b) <u>QCP Consultants</u>. After the date of this Agreement and prior to the Closing, the Debtor will comply with Section 5.1(b) of the Original Plan Sponsor Agreement.
- (c) <u>Chief Restructuring Officer</u>. After the date hereof, the CRO shall have those responsibilities set forth in Section 5.1(c) of the Debtor Disclosure Letter; <u>provided</u>, <u>however</u>, that at all times the responsibilities of the CRO shall include responsibility for allocation of costs under the Administrative Support Services Agreement (as referred to in the Centerbridge Facility) unless otherwise provided in the Centerbridge Facility. After the date of this Agreement and prior to the Closing Date, the power and authority of the CRO shall not be changed, and the CRO shall not be removed or replaced, without the consent of ProMedica and QCP, such consent not to be unreasonably withheld, conditioned or delayed. If the CRO voluntary terminates his employment or is otherwise unable to perform his or her responsibilities, a replacement CRO shall be appointed, subject to ProMedica's and QCP's reasonable consent (not to be unreasonably withheld, conditioned or delayed).

- 5.2 Rent and Forbearance.
- (a) Adjustment to Rent; Forbearance.
- (i) Upon the termination of the Master Lease with respect to any Leased Facility set forth on Section 5.2(a)(i) of the Debtor Disclosure Letter in connection with the releasing or sale of any Leased Facility by QCP or its Subsidiaries to any third-party purchasers, replacement tenants or other partners of QCP, there shall be no adjustment to Reduced Cash Rent. Upon the termination of the Master Lease with respect to any Leased Facility other than those set forth on Section 5.2(a)(i) of the Debtor Disclosure Letter in connection with the releasing or sale of any Leased Facility by QCP or its Subsidiaries to any third-party purchasers, replacement tenants or other partners of QCP, the parties to the Master Lease shall amend the Master Lease to reduce Minimum Rent (as defined in the Master Lease) and Reduced Cash Rent in a manner reflecting the net sales proceeds of such disposition.
- Forbearance. From the date of this Agreement until the earlier of the (b) Closing Date or termination of this Agreement in accordance with ARTICLE VII hereof, QCP agrees to forbear from bringing suit or exercising remedies under the Guaranty, the Master Lease or Agreements Regarding Subleases arising out of or related to (i) the failure to pay Rent (as defined in the Master Lease), (ii) notwithstanding Section 4.4 of the Master Lease, the failure to deposit or pay any amounts required to be deposited pursuant to Section 4.4 of the Master Lease prior to the termination of this Agreement, (iii) the failure to make capital expenditures required by the Master Lease, (iv) the occurrence of any "Event of Default" described in Section 16.1(g) or Section 16.1(h) of the Master Lease as a result of the transactions and other actions contemplated by this Agreement, (v) any "Event of Default" under clause (ii) of Section 16.1(l) of the Master Lease resulting from the failure of the Debtor to make any payment due under the Guaranty, (vi) the 2017 default under and acceleration of amounts owing under that certain Credit Agreement, dated as of April 6, 2011, among Holdings (as defined therein), HCR Healthcare, LLC, the several banks and other financial institutions or entities from time to time parties thereto, JPMorgan Chase Bank, N.A., as Administrative Agent and Collateral Agent, and J.P. Morgan Securities Inc., Credit Suisse Securities (USA) LLC and Merrill Lynch, Pierce, Fenner & Smith Incorporated, as co-Lead Arrangers and Joint Bookrunners, as amended by that certain First Amendment to Credit Agreement, dated as of March 21, 2014 and that certain Second Amendment to Credit Agreement, dated as of May 6, 2015 (the "JP Morgan Credit Facility"); or (vii) the failure of the Debtor to make any payment due under the Guaranty, including, in each case, by delivering a notice of default under the Master Lease, terminating the Master Lease, declaring an "Event of Default" under the Master Lease or exercising any other right or remedy under the Master Lease and/or Agreements Regarding Subleases against the Debtor or its direct and indirect Subsidiaries and, with respect to each, their property and interests in property.
- (c) <u>Forbearance Stipulations</u>. The Debtor and its Subsidiaries hereby admit, stipulate and agree as follows:
- (i) as of the date of the Original Plan Sponsor Agreement (and taking into account payments made by HCR III to the Original Lessors on the date hereof), the Original Lessors were owed the Reduced Cash Rent Past Due Balance;

- (ii) as of the date of the Original Plan Sponsor Agreement, the Original Lessors were owed approximately \$265,225,000 in unpaid Deferred Rent Obligations (as defined in the Master Lease) then due and owing under the Master Lease;
- (iii) as of the date of the Original Plan Sponsor Agreement, the Rent and other obligations of HCR III under the Master Lease and the Debtor under the Guaranty constituted legal, valid and binding obligations of HCR III and the Debtor, respectively, enforceable against HCR III and the Debtor in accordance with the terms of the Master Lease and the Guaranty;
- (iv) no portion of the Rent and other obligations of HCR III under the Master Lease, the Debtor under the Guaranty, or the Sublease OpCos under the Agreements Regarding Subleases is subject to avoidance, recharacterization, recovery or subordination pursuant to the Bankruptcy Code or applicable nonbankruptcy law;
- (v) the Debtor and HCR III do not have any claims, counterclaims, causes of action, defenses or setoff rights arising under or in connection with the Master Lease, the Guaranty or the Agreements Regarding Subleases against ProMedica, QCP, the Original Lessors or their respective Representatives (in their capacities as such) with respect to any unpaid Rent (including the Deferred Rent Obligation); and
- (vi) other than (A) as a result of HCR III's failure to pay all due and owing Rent in full, the failure to deposit or pay any amounts required to be deposited pursuant to Section 4.4 of the Master Lease or the failure to make capital expenditures required by the Master Lease, (B) any "Event of Default" under clause (ii) of Section 16.1(l) of the Master Lease resulting from the failure of the Debtor to make any payment due under the Guaranty, (C) the 2017 default under and acceleration of amounts owing under the JP Morgan Credit Facility and (D) the other matters described in Section 5.2(b), no default or Event of Default (as defined in the Master Lease) under the Master Lease is continuing.
- 5.3 <u>Business Operations</u>. Subject to the terms and conditions hereof, from the date of this Agreement until the Closing Date, the Debtor shall, and shall cause its Subsidiaries to:
- (a) comply in all material respects with all non-monetary obligations under the Master Lease, Guaranty and Agreements Regarding Subleases (it being understood that an obligation to make capital expenditures shall constitute a monetary obligation);
- (b) comply with Section 5.3(b), (c) and (d) of the Original Plan Sponsor Agreement;
- (c) keep on the premises of each Leased Facility all material personal property utilized in the ordinary course of operations at each such Leased Facility; and
- (d) deliver to the QCP Consultants, QCP and ProMedica (i) an updated 13-Week Cash Budget on a day during the last week of April 2018 and on a day during the last week of each calendar month thereafter and (ii) a report on a weekly basis, beginning on the third business day of the week after the date of this Agreement, setting forth any line-item variances

from the most recent 13-Week Cash Budget delivered to ProMedica, QCP and the QCP Consultants.

- 5.4 <u>QCP Support</u>. Subject to the terms and conditions hereof and for so long as this Agreement has not been terminated in accordance with its terms, and unless compliance is waived in writing by the JV and ProMedica, QCP agrees to, and to cause its direct and indirect subsidiaries and affiliates to:
- (a) support, and take all reasonable actions reasonably requested by the Debtor or ProMedica and the JV to facilitate the solicitation, confirmation and consummation of the Amended Plan and the transactions contemplated by the Amended Plan, including, without limitation, voting in favor of the Amended Plan and, upon the Effective Date of the Amended Plan, taking such actions as are reasonably requested by the Reorganized Debtor or ProMedica to dismiss with prejudice the Receivership Proceeding; provided that nothing set forth in this Section 5.4(a) shall require QCP to make any payment, transfer any asset or otherwise grant any consideration to any Person;
- (b) not take any other action, directly or indirectly, that would reasonably be expected to prevent, materially interfere with, materially delay or materially impede the solicitation of votes in connection with the Amended Plan, the confirmation of the Amended Plan or the consummation of the Amended Plan and the transactions contemplated thereunder;
- (c) not to take any action, directly or indirectly, with respect to the Receivership Proceeding (except as contemplated by clause (a) above);
- (d) not, without the prior written consent of the Debtor (such consent not to be unreasonably conditioned, withheld or delayed), approve any amendment or modification to the Merger Agreement or grant any waiver under the Merger Agreement that, in each case, would reasonably be expected to prevent or materially delay the consummation of the Closing;
- (e) not to object to or otherwise commence, or encourage any other person to commence, any proceeding or take any action opposing the Amended Plan or Disclosure Statement that would reasonably be expected to have a material adverse effect on the consummation of the Amended Plan and the transactions contemplated thereunder; and
- (f) Upon satisfaction of the condition set forth in Section 6.1(e)(i) and the satisfaction of all conditions in Article VI (other than Section 6.1(e)(ii)), QCP will take such action within its power to satisfy the condition set forth in Section 6.1(e)(ii) (including enforcing its rights under the Merger Agreement).

Notwithstanding anything to the contrary in this Agreement or otherwise, it shall not be deemed a breach or violation of this Agreement for QCP, its Subsidiaries and/or any of their respective representatives to take any actions contemplated by Section 7.4 of the Merger Agreement or for QCP to exercise any rights or remedies under the Merger Agreement.

5.5 <u>Restructuring Costs</u>. Subject to the limitations set forth in this Section 5.5, ProMedica shall reimburse the Debtor, promptly upon demand, for all Restructuring Costs paid by the Debtor during the Fee Reimbursement Period. "<u>Restructuring Costs</u>" means all fees, costs

and expenses paid by the Debtor to professionals (to the extent either retained by the Debtor or retained by any other Person entitled to reimbursement by the Debtor) during the Fee Reimbursement Period in connection with (i) the Original Plan, (ii) the Amended Plan, (iii) the Bankruptcy Case, or (iv) the restructuring or reorganization of the Debtor. Notwithstanding the foregoing, ProMedica shall not be required by this Section 5.5 to reimburse the Debtor for Restructuring Costs in excess of \$2 million per calendar month or, with respect to the calendar month in which the Fee Reimbursement Period ends, to the extent the Fee Reimbursement Period ends on a day other than the last day of such calendar month, a pro rata portion of such amount; provided that, to the extent that the amount of reimbursed Restructuring Costs in any calendar month is less than \$2 million, the unused portion may be applied in future calendar months. The "Fee Reimbursement Period" means the period from (and including) May 1, 2018 through (and including) the earlier of (i) the Effective Date (as defined in the Amended Plan) or (ii) the date of termination of this Agreement.

ARTICLE VI

Conditions

- 6.1 <u>Conditions to Each Party's Obligations</u>. The respective obligations of each party to effect the Transactions are subject to the satisfaction or waiver at or prior to the Closing of each of the following conditions:
- (a) <u>Litigation</u>. No court or other Governmental Entity of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any Law (whether temporary, preliminary or permanent) that is in effect and restrains, enjoins or otherwise prohibits consummation of the Transactions. No Law shall be in effect enjoining or otherwise prohibiting the consummation of the Transactions.
- (b) <u>Approvals</u>. The Governmental Approvals set forth on Section 6.1(b) of the Debtor Disclosure Letter shall have been obtained or made, except to the extent the failure to obtain such Governmental Approvals would not, in the aggregate, reasonably be expected to result in losses, costs, liabilities or expenses to ProMedica, the Debtor and their respective Subsidiaries in excess of \$5,000,000, and the applicable waiting period under the HSR Act shall have expired or been terminated.
- (c) <u>Confirmation Order</u>. The Bankruptcy Court shall have entered the Confirmation Order, there shall have been no amendment to the Amended Plan or the Confirmation Order except as permitted hereby and all conditions to the effectiveness of the Amended Plan shall have been satisfied or waived in accordance with the terms of the Plan.
- (d) <u>Bankruptcy Court Orders</u>. The Bankruptcy Court shall not have entered an Order (i) dismissing the Bankruptcy Case or converting the Bankruptcy Case to a case under Chapter 7 of the Bankruptcy Code or (ii) materially inconsistent with this Agreement, the Amended Plan or the Confirmation Order in a manner adverse to ProMedica or QCP.

(e) Merger Agreement.

- (i) All conditions to the consummation of the Merger in the Merger Agreement (other than the occurrence of the Closing and such conditions to the consummation of the Merger as, by their nature, are to be satisfied at such consummation) shall have been satisfied or waived in accordance with the terms of the Merger Agreement; and
- (ii) The parties to the Merger Agreement have delivered the Requisite Merger Filings (as defined in the Merger Agreement) in escrow to Wachtell, Lipton, Rosen & Katz LLP to be released and filed with the relevant Governmental Entities immediately after the Closing, pursuant to the terms of the Merger Agreement.
- 6.2 <u>Conditions to Obligations of ProMedica</u>. The obligations of ProMedica to effect the Transactions are also subject to the satisfaction or waiver by ProMedica at or prior to the Closing Date of the following conditions:
- (a) <u>Performance of Obligations of the Debtor</u>. A court of competent jurisdiction shall not have determined that the Debtor has breached in any material respect any material covenants required to be performed by it under this Agreement on or prior to the Closing Date and such material breach either is not curable or has not been cured.
- (b) Officer's Certificate. ProMedica shall have received a certificate dated as of the Closing Date, signed by the chief executive officer of the Debtor, stating that the condition specified in Section 6.2(a) has been waived or satisfied.
- 6.3 <u>Conditions to Obligation of the Debtor</u>. The obligation of the Debtor to effect the Transactions is also subject to the satisfaction or waiver by the Debtor at or prior to the Closing of the following conditions:
- (a) <u>Performance of Obligations of ProMedica</u>. ProMedica shall have performed and complied, in all material respects, all material obligations required to be performed by it under this Agreement on or prior to the Closing Date, other than any material breach which is curable and has been cured.
- (b) Officer's Certificate. The Debtor shall have received a certificate dated as of the Closing Date, signed by the chief executive officer of ProMedica Parent, stating that the condition specified in Section 6.3(a) has been waived or satisfied.

ARTICLE VII

Termination

- 7.1 <u>Automatic Termination</u>. Subject to Section 8.1, this Agreement will expire by its own terms and terminate automatically, without notice or further action by any party, if the Merger Agreement is terminated in accordance with its terms.
- 7.2 <u>Termination by Mutual Consent</u>. This Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing, by mutual written consent of the Debtor, ProMedica and QCP, by action of their respective boards of directors.

- 7.3 <u>Termination by ProMedica, QCP or the Debtor</u>. This Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing by any of ProMedica, QCP or the Debtor if:
- (a) the Transactions shall not have been consummated by 11:59 p.m. NYC time on October 15, 2018 (the "<u>Termination Date</u>") (or 11:59 p.m. NYC time on October 12, 2018, if the termination right is exercised by QCP);
- Bankruptcy Court enters an Order (i) denying confirmation of the Amended Plan, (ii) dismissing the Bankruptcy Case, or (iii) converting the Bankruptcy Case of the Debtor to a case under chapter 7 of the Bankruptcy Code, and such Order (x) is or becomes a Final Order or (y) has been in effect for thirty (30) days and is not subject to stay; provided that a "Final Order" shall mean an Order which has not been stayed (or with respect to which any stay has been lifted) and (A) as to which the time to file an appeal, a motion for rehearing or reconsideration (excluding any motion under Section 60(b) of the Federal Rules of Civil Procedure) or a petition for writ of certiorari has expired and no appeal, motion, stay or petition is pending, or (B) in the event that such an appeal or petition thereof has been sought, either (1) such Order shall have been affirmed by the highest court to which such Order was appealed or certiorari shall have been denied, and the time to take any further appeal or petition of certiorari shall have expired or (2) such appeal, motion, stay or petition shall not have been granted and shall no longer be pending and the time for seeking such appeal, motion, stay or petition shall have been expired; or
- (c) any Order permanently restraining, enjoining or otherwise prohibiting consummation of the Transactions shall become a Final Order;

<u>provided</u>, <u>however</u>, that the right to terminate this Agreement pursuant to this Section 7.3 shall not be available to any party that has breached in any material respect its obligations under this Agreement in any manner that shall have proximately contributed to the occurrence of the failure of a condition to the consummation of the Transactions.

- 7.4 <u>Termination by the Debtor or QCP</u>. This Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing by the Debtor or QCP if:
- (a) a court of competent jurisdiction shall have determined that there has been a material breach of any covenant or agreement made by ProMedica in this Agreement such that the condition set forth in Section 6.3(a) would not be satisfied and such material breach is not curable or, if curable, is not cured within the earlier of (x) thirty (30) days after written notice thereof is given by the Debtor or QCP, as applicable, to ProMedica and the other parties hereto and (y) the Termination Date; provided that the Debtor shall not have the right to terminate this Agreement pursuant to this Section 7.4(a) if the Debtor is then in material breach of any of its covenants or agreements contained in this Agreement such that the condition set forth in Section 6.2(a) would not be satisfied; provided further that QCP shall not have the right to terminate this Agreement pursuant to this Section 7.4(a) if QCP is then in material breach of any of its covenants or agreements contained in Section 5.4 of this Agreement; or

- (b) if any condition set forth in Section 6.1 shall have become incapable of fulfillment prior to the Termination Date other than as a result of a material breach by the Debtor or QCP, as applicable, of any covenant or agreement contained in this Agreement, and such condition is not waived by the Debtor or QCP, as applicable.
- 7.5 <u>Termination by ProMedica or QCP</u>. This Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing by ProMedica or QCP if:
- (a) a court of competent jurisdiction shall have determined that there has been a material breach of any covenant or agreement made by the Debtor in this Agreement such that the condition set forth in Section 6.2(a) would not be satisfied and such material breach is not curable or, if curable, is not cured within the earlier of (x) thirty (30) days after written notice thereof is given by ProMedica or QCP, as applicable, to the Debtor and the other parties hereto and (y) the Termination Date; provided that ProMedica shall not have the right to terminate this Agreement pursuant to this Section 7.5(a) if ProMedica is then in material breach of any of its covenants or agreements contained in this Agreement; provided further that QCP shall not have the right to terminate this Agreement pursuant to this Section 7.5(a) if QCP is then in material breach of any of its covenants or agreements contained in Section 5.4 of this Agreement;
- (b) if any condition set forth in Section 6.1 shall have become incapable of fulfillment prior to the Termination Date other than as a result of a material breach by ProMedica or QCP, as applicable, of any covenant or agreement contained in this Agreement, and such condition is not waived by ProMedica or QCP, as applicable;
- (c) any Subsidiary of the Debtor commences proceedings under the Bankruptcy Code without the prior written consent of ProMedica or QCP, which may be withheld in such Person's sole discretion;
- (d) the Bankruptcy Court or any court exercising appellate jurisdiction over the Bankruptcy Court enters an Order (i) appointing a trustee with expanded powers pursuant to section 1104 of the Bankruptcy Code or (ii) that is materially inconsistent with the Amended Plan or the Transactions in a manner adverse to ProMedica or QCP, and such Order (x) is or becomes a Final Order or (y) has been in effect for thirty (30) days and is not subject to stay;
- (e) if the Carlyle Holders have (i) filed any federal or state tax return, or any amendment to such a return, claiming any deduction for worthlessness of its Debtor Shares (as defined in the Alternative Restructuring Support Agreement) or (ii) Transferred (as defined in the Alternative Restructuring Support Agreement) any Debtor Shares other than in accordance with the Alternative Restructuring Support Agreement;
- (f) other than in accordance with the Amended Plan, if there has been (i) a termination or rejection of the Master Lease or any Agreement Regarding Subleases by the lessee or any other Subsidiary of the Debtor, or (ii) the occurrence of a Default or an Event of Default (each as defined under the Centerbridge Facility) under the Centerbridge Facility that would permit the acceleration of Obligations (as defined under the Centerbridge Facility) and a failure by the Debtor to cure such Default or Event of Default within thirty (30) days of the occurrence of

such a Default or Event of Default, or the acceleration of Obligations (as defined under the Centerbridge Facility); or

- 7.6 <u>Termination by QCP</u>. This Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing by QCP if:
- (a) the Confirmation Order has not been entered by the Bankruptcy Court within sixty-five (65) days after the date of this Agreement; or
- (b) HCR III fails to pay such cash and cash equivalents available to pay part or all of the Reduced Cash Rent due from and after the date hereof after (A) making all transfers of funds from the Debtor's home health care, hospice and other ancillary businesses other than the Debtor's skilled nursing and inpatient rehabilitation facilities, memory care facilities and assisted living facilities (the "SNF Business") that are permitted under the Centerbridge Facility and (B) paying all available cash and cash equivalents from the SNF Business subject to retaining such reserves and making such other expenditures as either (x) the CRO or (y) the Debtor Board has determined in good faith, after consulting with QCP, are necessary to allow the Debtor to operate safely, prudently and in the ordinary course of business, and HCR III does not cure such failure within five (5) business days of such failure.

ARTICLE VIII

Miscellaneous and General

- 8.1 No Survival; Effect of Termination and Abandonment.
- Except as provided in Section 8.1(b) below, in the event of termination of this Agreement and the abandonment of the Transactions pursuant to ARTICLE VII, this Agreement shall become void and be of no effect with no liability to any Person on the part of any party hereto (or of any of its Representatives or Affiliates); provided, however, and notwithstanding anything in the foregoing to the contrary, that (i) no such termination shall relieve any party of any liability or damages resulting from any fraud by such party or Willful Breach of this Agreement on the part of such party occurring prior to such termination and (ii) the provisions set forth in this ARTICLE VIII, ProMedica's expense reimbursement and indemnification obligations set forth in Section 4.9(d) and the agreements of the parties contained in Section 4.4 (*Publicity*), Section 4.6 (*Expenses*) and Section 4.8 (*Confidentiality*) shall survive the termination of this Agreement. For purposes of this Agreement, "Willful Breach" shall mean a material breach of, or failure to perform, any of the covenants or other agreements contained in this Agreement (including failure to close the Transactions in accordance with Section 1.4) that is a consequence of an act, or failure to act, by the breaching or non-performing Person with actual knowledge, or knowledge that a Person acting reasonably under the circumstances would reasonably be expected to have, that such Person's act or failure to act would, or would be reasonably expected to, result in or constitute a breach of, or failure of, performance under this Agreement.
- (b) <u>Waiver</u>; <u>Automatic Stay</u>. The failure of a party to exercise its right to terminate this Agreement under, or the extension of any time period in, any provision of ARTICLE

VII at any time will not constitute a waiver of any such right. The Debtor hereby waives any requirement that ProMedica or QCP seek or obtain relief from the automatic stay under section 362 of the Bankruptcy Code for purposes of providing notice under, or terminating, this Agreement, and the Debtor agrees not to object to ProMedica or QCP seeking, if necessary, to lift the automatic stay in connection therewith.

- (c) This ARTICLE VIII and the agreements of the Debtor and ProMedica contained in Section 4.7 (*Indemnification; Directors' and Officers' Insurance*) shall survive the consummation of the Transactions. All other representations, warranties, covenants and agreements in this Agreement and in any certificate or other writing delivered pursuant hereto shall not survive consummation of the Transactions and, from and after the Closing Date, no party shall have any claim against any other party for breach of any representation or warranty set forth herein.
- 8.2 <u>Modification or Amendment</u>. Subject to the provisions of applicable Laws, at any time prior to the Closing Date, the parties hereto may modify or amend this Agreement, by written agreement executed and delivered by duly authorized officers of the respective parties.
- 8.3 <u>Waiver of Conditions</u>. The conditions to each of the parties' obligations to consummate the Transactions and the other transactions contemplated by this Agreement are for the sole benefit of such party and may be waived by such party in whole or in part to the extent permitted by applicable Laws.
- 8.4 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each such counterpart being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement.

8.5 <u>GOVERNING LAW AND VENUE; WAIVER OF JURY TRIAL;</u> SPECIFIC PERFORMANCE.

THIS AGREEMENT SHALL BE DEEMED TO BE MADE IN AND IN ALL RESPECTS SHALL BE INTERPRETED, CONSTRUED AND GOVERNED BY AND IN ACCORDANCE WITH THE LAW OF THE STATE OF DELAWARE WITHOUT REGARD TO THE CONFLICTS OF LAW PRINCIPLES THEREOF TO THE EXTENT THAT SUCH PRINCIPLES WOULD DIRECT A MATTER TO ANOTHER JURISDICTION. The parties hereby irrevocably submit to the personal jurisdiction of the courts of the State of Delaware and the federal courts of the United States of America located in the State of Delaware solely in respect of the interpretation and enforcement of the provisions of this Agreement and of the documents referred to in this Agreement, and in respect of the Transactions, and hereby waive, and agree not to assert, as a defense in any Proceeding for the interpretation or enforcement hereof or of any such document, that it is not subject thereto or that such Proceeding may not be brought or is not maintainable in said courts or that the venue thereof may not be appropriate or that this Agreement or any such document may not be enforced in or by such courts, and the parties hereto irrevocably agree that all claims relating to such Proceeding or transactions shall be heard and determined in such a Delaware State or federal court. The parties hereby consent to and grant any such court jurisdiction over the person of such parties and, to the extent permitted by law, over the subject matter of such dispute and agree that mailing of process or other papers in connection with any

such Proceeding in the manner provided in Section 8.6 or in such other manner as may be permitted by Law shall be valid and sufficient service thereof. Notwithstanding the foregoing consent to jurisdiction in either a state or federal court of competent jurisdiction in the State of Delaware, each of the parties hereby agrees that during the pendency of the Bankruptcy Case, the Bankruptcy Court shall have exclusive jurisdiction over all matters arising out of or in connection with this Agreement.

- (b) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (i) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (ii) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (iii) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (iv) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 8.5.
- (c) The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in the Court of Chancery of the State of Delaware or, during the Bankruptcy Case, in the Bankruptcy Court, this being in addition to any other remedy to which such party is entitled at law or in equity. Each of the parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that (x) the other party has an adequate remedy at law or (y) an award of specific performance is not an appropriate remedy for any reason at law or in equity, and that it waives any requirement for the securing or posting of a bond in connection with any such remedy.
- 8.6 <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by any party to the other parties shall be in writing and delivered personally or sent by registered or certified mail, postage prepaid, by facsimile or overnight courier:

If to QCP:

Quality Care Properties, Inc.
7315 Wisconsin Avenue, Suite 550 East
Bethesda, Maryland 20814
Attention: David Haddock, General Counsel

Email: dhaddock@qcpcorp.com

with a copy to:

Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019

Attention: Adam O. Emmerich and Scott K. Charles Email: aoemmerich@wlrk.com and skcharles@wlrk.com

and

Sullivan & Cromwell LLP 125 Broad Street New York, New York 10004

Attention: Audra D. Cohen and Andrew G. Dietderich

Email: cohena@sullcrom.com and dietdericha@sullcrom.com

If to the Debtor:

HCR ManorCare, Inc. 333 North Summit Street Toledo, Ohio 43604

Attention: Richard A. Parr, General Counsel

John Castellano, Chief Restructuring Officer

Email: rparr@hcr-manorcare.com jcastellano@alixpartners.com

with a copy to:

Sidley Austin LLP One South Dearborn Chicago, IL 60603 Attention: Larry Nyhan Email: lnyhan@sidley.com

and

Latham & Watkins LLP 555 Eleventh Street, NW, Suite 1000 Washington, D.C. 20004-1304 Attention: Daniel T. Lennon Roger G. Schwartz

J. Cory Tull

Email: daniel.lennon@lw.com roger.schwartz@lw.com cory.tull@lw.com

If to ProMedica:

ProMedica Health System, Inc. 100 Madison Avenue Toledo, OH 43604 Attention: Jeff Kuhn, Chief Legal Officer and General Counsel

Email: jeff.kuhn@promedica.orgwith a copy to:

Shumaker, Loop & Kendrick, LLP 1000 Jackson Street Toledo, OH 43604

Attention: James I Rothschild

David Coyle

Email: jrothschild@slk-law.com dcoyle@slk-law.com

If to the Meerkat I LLC or the Lessors:

Meerkat I LLC 4500 Dorr Street Toledo, OH 43615 Attn: General Counsel with a copy to:

Gibson Dunn & Crutcher LLP 200 Park Avenue New York, New York 10166 Attn: Dennis Friedman

Michael A. Rosenthal

Email: dfriedman@gibsondunn.com mrosenthal@gibsondunn.com

or to such other persons or addresses as may be designated in writing by the party to receive such notice as provided above. Any notice, request, instruction or other document given as provided above shall be deemed given to the receiving party: (i) upon actual receipt, if delivered personally; (ii) three (3) business days after deposit in the mail, if sent by registered or certified mail; (iii) upon confirmation of successful transmission if sent by facsimile (provided that if given by facsimile, such notice, request, instruction or other document shall be followed up within one (1) business day by dispatch pursuant to one of the other methods described herein); or (iv) on the next business day after deposit with an overnight courier, if sent by an overnight courier.

8.7 <u>Entire Agreement</u>. This Agreement (including any exhibits hereto), the Debtor Disclosure Letter and the Alternative Restructuring Support Agreement constitute the entire agreement and supersede all other prior agreements, understandings, representations and warranties both written and oral, among the parties, with respect to the subject matter hereof. EACH PARTY HERETO AGREES THAT, EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS AGREEMENT, NEITHER PROMEDICA, THE

LESSORS NOR THE DEBTOR MAKES OR RELIES ON ANY OTHER REPRESENTATIONS, WARRANTIES OR INDUCEMENTS, AND EACH HEREBY DISCLAIMS ANY OTHER REPRESENTATIONS, WARRANTIES OR INDUCEMENTS, EXPRESS OR IMPLIED, RELATED TO THE DEBTOR, THE TRANSACTIONS OR ANY OTHER MATTER, INCLUDING REPRESENTATIONS OR WARRANTIES AS TO THE ACCURACY OR COMPLETENESS OF ANY OTHER INFORMATION, MADE BY, OR MADE AVAILABLE BY, ITSELF OR ANY OF ITS REPRESENTATIVES, WITH RESPECT TO, OR IN CONNECTION WITH, THE NEGOTIATION, EXECUTION OR DELIVERY OF THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY, NOTWITHSTANDING THE DELIVERY OR DISCLOSURE TO THE OTHER OR THE OTHER'S REPRESENTATIVES OF ANY DOCUMENTATION OR OTHER INFORMATION WITH RESPECT TO ANY ONE OR MORE OF THE FOREGOING. No party shall be bound by, or be liable for, any alleged representation, promise, inducement or statement of intention not contained herein.

- 8.8 No Third Party Beneficiaries; No Recourse. Except as provided in Section 4.7 (Indemnification; Directors' and Officers' Insurance) and Section 8.15 (Certain Waivers) only, ProMedica, the Lessors and the Debtor hereby agree that their respective representations, warranties and covenants set forth herein are solely for the benefit of the other parties hereto, in accordance with and subject to the terms of this Agreement, and this Agreement is not intended to, and does not, confer upon any Person other than the parties hereto any rights or remedies hereunder, including, without limitation, the right to rely upon the representations and warranties set forth herein. The representations and warranties in this Agreement are the product of negotiations among the parties hereto and are for the sole benefit of the parties hereto. Any inaccuracies in such representations and warranties are subject to waiver by the parties hereto in accordance with Section 8.3 without notice or liability to any other Person. In some instances, the representations and warranties in this Agreement may represent an allocation among the parties hereto of risks associated with particular matters regardless of the knowledge of any of the parties hereto. Consequently, Persons other than the parties hereto may not rely upon the representations and warranties in this Agreement as characterizations of actual facts or circumstances as of the date of this Agreement or as of any other date. This Agreement may only be enforced against, and any claim or cause of action based upon, arising out of, or related to this Agreement or the Transactions may only be brought against, the entities that are expressly named as parties hereto and then only with respect to the specific obligations set forth herein with respect to such party. Except to the extent a named party to this Agreement (and then only to the extent of the specific obligations undertaken by such named party in this Agreement and not otherwise), no past, present or future director, officer, employee, incorporator, member, partner, stockholder, Affiliate, agent, attorney, advisor or representative or Affiliate of any of the foregoing shall have any liability (whether in contract, tort, equity or otherwise) for any one or more of the representations, warranties, covenants, agreements or other obligations or liabilities of any one or more of the Debtor, ProMedica or the Lessors under this Agreement (whether for indemnification or otherwise) or of or for any claim based on, arising out of, or related to this Agreement or the Transactions.
- 8.9 <u>Obligations of ProMedica and of the Debtor</u>. Whenever this Agreement requires a Subsidiary of ProMedica to take any action, such requirement shall be deemed to include an undertaking on the part of ProMedica to cause such Subsidiary to take such action.

Whenever this Agreement requires a Subsidiary of the Debtor to take any action, such requirement shall be deemed to include an undertaking on the part of the Debtor to cause such Subsidiary to take such action.

- 8.10 <u>Transfer Taxes</u>. All transfer, documentary, sales, use, stamp, registration and other such Taxes and fees (including penalties and interest) incurred in connection with the Transactions shall be paid by ProMedica when due, and ProMedica will indemnify the Debtor against liability for any such Taxes. Notwithstanding the foregoing sentence, responsibility for the payment of any real property transfer taxes that are payable as a result of the consummation of the Merger shall be as set forth in the Merger Agreement.
- 8.11 <u>Definitions</u>. Each of the terms set forth in Annex A is defined in the Section of this Agreement set forth opposite such term.
- 8.12 Severability. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application of such provision to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application of such provision, in any other jurisdiction.
- 8.13 <u>Interpretation; Construction</u>. The table of contents and headings herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof. Where a reference in this Agreement is made to a Section, Annex or Exhibit, such reference shall be to a Section of or Annex or Exhibit to this Agreement unless otherwise indicated. Whenever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation".
- (a) The parties have participated jointly in negotiating and drafting this Agreement. In the event that an ambiguity or a question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.
- (b) The Debtor and ProMedica have or may have set forth information in the Debtor Disclosure Letter in a section of such Debtor Disclosure Letter that corresponds to the section of this Agreement to which it relates. The fact that any item of information is disclosed in the Debtor Disclosure Letter to this Agreement shall not be construed to mean that such information is required to be disclosed by this Agreement.
- 8.14 <u>Assignment</u>. This Agreement shall not be, directly or indirectly, assignable by operation of law or otherwise; <u>provided</u>, <u>however</u>, that (i) ProMedica Parent may designate, by

written notice to the Debtor and QCP, another wholly-owned direct or indirect wholly-owned Subsidiary to be a party in lieu of Purchaser (provided that such Subsidiary remains wholly-owned by ProMedica Parent through the Closing Date), in which event all references herein to Purchaser shall be deemed references to such other Subsidiary, except that all representations and warranties made herein with respect to Purchaser as of the date of this Agreement shall be deemed representations and warranties made with respect to such other Subsidiary as of the date of such designation; provided that any such designation shall not violate the Merger Agreement, materially impede or delay the consummation of the Transactions or otherwise materially impede the rights of the stockholders of the Debtor under this Agreement and (ii) the Merger Agreement and consummation of the transactions contemplated thereby shall not constitute an assignment of this Agreement. Any purported assignment in violation of this Agreement is void and such designation shall not relieve ProMedica of any obligation or liability hereunder. The sale or transfer by ProMedica of any direct or indirect equity interest in Purchaser shall constitute an assignment of this Agreement.

Certain Waivers. ProMedica and the Debtor (on behalf of itself and its Affiliates) acknowledge that Latham & Watkins LLP and other legal counsel ("Prior Debtor Counsel") have, on or prior to the Closing Date, represented (i) the Carlyle Holders in connection with the Transactions and (ii) the Debtor and its Subsidiaries and other Affiliates in one or more matters, including relating to this Agreement and the Transactions (each, an "Existing **Representation**"), and that, in the event of any post-Closing matters (x) relating to this Agreement or any other agreements or Transactions (including any matter that may be related to a litigation, claim or dispute arising under or related to this Agreement or such other agreements or in connection with such transactions) and (y) in which ProMedica or any of its Affiliates (including the Debtor and its Subsidiaries), on the one hand, and any Carlyle Holder or any of its Affiliates (each, a "Designated Person"), on the other hand, are or may be adverse to each other (each, a "Post-Closing Matter"), the Designated Persons reasonably anticipate that Prior Debtor Counsel will represent them in connection with such matters. Accordingly, ProMedica and the Debtor (on behalf of itself and its Affiliates) hereby (i) waive and shall not assert, and agree after the Closing to cause their Affiliates to waive and to not assert, any conflict of interest arising out of or relating to the representation by one or more Prior Debtor Counsel of one or more Designated Persons in connection with one or more Post-Closing Matters (the "Post-Closing Representations"), and (ii) agree that, in the event that a Post-Closing Matter arises, Prior Debtor Counsel may represent one or more Designated Persons in the Post-Closing Matter even though the interests of such Person(s) may be directly adverse to ProMedica or any of its Affiliates (including the Debtor and its Subsidiaries), and even though Prior Debtor Counsel may (i) have represented the Debtor or its Subsidiaries in a matter substantially related to such dispute or (ii) be currently representing the Debtor or any of its respective Affiliates. Without limiting the foregoing, ProMedica and the Debtor (on behalf of itself and its Affiliates) consent to the disclosure by Prior Debtor Counsel, in connection with one or more Post-Closing Representations, to the Designated Persons of any information learned by Prior Debtor Counsel in the course of one or more Existing Representations, whether or not such information is subject to the attorney-client privilege of the Debtor or any of its Subsidiaries or Prior Debtor Counsel's duty of confidentiality as to the Debtor or any of its Subsidiaries and whether or not such disclosure is made before or after the Closing. The Carlyle Holders and Latham & Watkins LLP are intended third-party beneficiaries, and are relying on the provisions of, of this Section 8.15.

- 8.16 <u>No Solicitation</u>. This Agreement, the Transactions and the Amended Plan are the product of negotiations among the parties, together with their respective representatives. Notwithstanding anything herein to the contrary, this Agreement is not, and shall not be deemed to be, a solicitation of votes for the acceptance of the Amended Plan or any plan of reorganization for the purposes of sections 1125 and 1126 of the Bankruptcy Code or otherwise.
- 8.17 <u>Settlement Discussions</u>. This Agreement is part of a proposed settlement of matters that could otherwise be the subject of litigation among the parties. Nothing herein shall be deemed an admission of any kind. Pursuant to the agreement of the parties, Federal Rule of Evidence 408, any applicable state rules of evidence and any other applicable law, foreign or domestic, this Agreement, all exhibits, schedules and appendices hereto, and all negotiations relating hereto shall not be admissible into evidence in any proceeding other than to prove the existence of this Agreement or in a proceeding to enforce the terms of this Agreement.

HCR MANORCARE, INC.

Chief Restructuring Officer Title: QUALITY CARE PROPERTIES, INC. By: Name: Title: PROMEDICA HEALTH SYSTEMS, INC. By: Name: Title: MEERKAT I, LLC Name: Title: SUBURBAN HEALTHCO, INC. Name: Title:

[SIGNATURE PAGE CONTINUES]

[Signature Page to Plan Sponsor Agreement]

HCR MANORCARE, INC.
By: Name: Title:
QUALITY CARE PROPERTIES, INC.
By: Name: C. Marc Richards Title: Chief Financial Officer
PROMEDICA HEALTH SYSTEMS, INC.
By: Name: Title:
MEERKAT I, LLC
By: Name: Title:
SUBURBAN HEALTHCO, INC.
By:
Name: Title:

[SIGNATURE PAGE CONTINUES]

Зу:	*	
Name:		
Title:		5.
84		
QUALITY CARE PROPERTIES, INC.		
		0.59
	8	
By: Name:		
Title:		
1100.		
	~	
ROMEDICA HEALTH SYSTEM, INC	С,	*
By:		97
By: Name:	CONTRICTION OF THE STATE OF THE	98 70
		95 55
Name:	-	#* ##
Title:		# # # # # # # # # # # # # # # # # # #
Name: Title:		<u>e</u> t
Name: Title: MEERKAT I LLC		<u>.</u>
Name: Title: MEERKAT I LLC By:		
Name: Title: MEERKAT I LLC By: Name:		
Name: Title: MEERKAT I LLC By:		
Name: Title: MEERKAT I LLC By: Name:		
Name: Title: MEERKAT I LLC By: Name: Title:		
Name: Title: MEERKAT I LLC By: Name: Title:		
Name: Title: MEERKAT I LLC By: Name:		

SIGNATURE PAGE CONTINUES

HCR MANORCARE, INC.

~
By: Name: Title:
QUALITY CARE PROPERTIES, INC.
By: Name: Title:
PROMEDICA HEALTH SYSTEM, INC. By: Name: Randy Oostra President & CEO
MEERKAT, LLC
By: Name: Title:
SUBURBAN HEALTHCO, INC. By: Name: Randy Oostra Title: Prosident
President \

[SIGNATURE PAGE CONTINUES]

ANNEX A

DEFINED TERMS

<u>Terms</u>	Section
13-Week Cash Budget	5.1(b)(ii)(C)
280G Stockholder Vote	
Acquisition Proposal	
Affiliate	
Alternative Acquisition Agreement	
Alternative Restructuring Support Agreement	Recitals
Agreement	
Agreement Regarding Subleases, Agreements Regarding Subleases	
Amended Plan	
Applicable Restriction	
Approved Fees	` ' ' '
Audited Financial Statements	
Authorization Order	` / ` /
Bankruptcy and Equity Exception	` /
Bankruptcy Case	` /
Bankruptcy Code	Recitals
Bankruptcy Court	
Business	Recitals
business day	1.4
By-Laws	1.3(b)
Carlyle Holders	Recitals
Case Budget	3.1(d)
Centerbridge Facility	Recitals
Charter	1.3(a)
Closing	1.4
Closing Date	1.4
Code	1.6(a)
Company Plan	2.1(h)(i)
Confirmation Order	
Contract	2.1(d)(ii)
CEO	5.1(b)(i)
CFO	5.1(b)(i)
CRO	4.1(a)(xix)
D&O Insurance	4.7(b)
Debt Exchange	1.1(a)
Debt Financing Sources	4.9(c)
Debtor	Preamble
Debtor Board	1.6(b)
Debtor Disclosure Letter	2.1
Debtor Documents	2.1(c)

Debtor IT Assets	2.1(q)(vi)
Debtor Labor Agreements	
Debtor Restructuring Committee	5.1(a)
Deferred Rent Obligation	1.1(a)
Designated Person	8.15
Disclosure Statement	3.1(b)
Employment Agreement	1.5(b)(i)
Environmental Law	2.1(o)
ERISA Affiliate	2.1(h)(v)
ERISA Plans	` / ` /
Exchange Act	* * * * *
Excess Severance Payments	
Existing Representation	. ,
Fee Reimbursement Period	
Final Order	
Financing	
Former CEO	. ,
Former CEO Settlement Agreement	Recitals
Fundamental Representations of the Debtor	
GAAP	
Governmental Approvals	
Governmental Entity	
Government Contract	· /
Guaranty	
Hazardous Substance	
HCR III	. ,
Indemnified Parties	
Insurance Policies	
Intellectual Property Rights	
IRS	
JP Morgan Credit Facility	
Key Employee Incentive Plan	· /
	2.1(n)(viii)
Laws	
Leased Facility, Leased Facilities	
Lessors	
Liabilities	
Licenses	. ,
Lien	= :
Manorcare Financing	` '
Marketing Documentation	· /
Master Lease	
New Master Lease	
Material Adverse Effect	
	` /
Material Contract	* / * /
New Hire	4.1(a)(X1X)

Non-Debtor Subsidiary Released Parties	3.4
Order	2.1(g)
Original Confirmation Order	Recitals
Original Lessors	Recitals
Original Plan	Recitals
Original Plan Sponsor Agreement	Recitals
Other Indemnitors	4.7(g)
Owned Intellectual Property	2.1(q)(i)
Parachute Payment	4.5(e)(i)
ProMedica	Preamble
party, parties	Preamble
Payment Date	1.5(a)
PBGC	2.1(h)(v)
Person	1.6(d)
Permitted Lien	1.1(b)
Petition Date	3.2(b)
Plan Acquisition	` '
Post-Closing Matter	
Post-Closing Representations	
Potential Acquiror	
Preferred Shares	
Prior Debtor Counsel	` / ` /
Prior Forbearance Agreement	
Proceedings	
ProMedica	(C)
ProMedica Approvals	3.2
ProMedica Parent	
Purchaser	
QCP	
QCP Consultants	
QCP Entities	` ' ' '
Receivership Proceeding	
Reduced Cash Rent	
Reduced Cash Rent Past Due Balance	Recitals
Reorganized Company Stock	
Representatives	
Restructuring Documents	
SEC	· /
Second Forbearance Agreement	
Securities Act	
Separation Agreement	
Shares	
SNF/AL Remarketing Process	
SNF Business	
Stock Plan	
Sublease Amendments	

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 191 of 441

Subleases	Recitals
Sublease OpCo, Sublease OpCos	Recitals
Subsidiary	1.6(d)
Tail Period	4.7(b)
Takeover Statute	2.1(n)
Tax, Taxes	2.1(p)
Tax Return	2.1(p)
Taxing Authority	2.1(p)
Termination Date	
Transactions	1.4
Willful Breach	8.1(a)

Amended Plan

New Master Lease – Basic Terms

Alternative Restructuring Support Agreement

RESTRUCTURING SUPPORT AGREEMENT

This RESTRUCTURING SUPPORT AGREEMENT (this "<u>Agreement</u>") is made and entered into as of April 25, 2018, by and among (i) HCR ManorCare, Inc., a Delaware corporation (the "<u>Debtor</u>"), (ii) Carlyle MC Partners, L.P., a Delaware limited partnership, Carlyle Partners V-A MC, L.P., a Delaware limited partnership, Carlyle Partners V MC, L.P., a Delaware limited partnership, Carlyle Partners V MC, L.P., a Delaware limited partnership, collectively, the "<u>Majority Holders</u>"), ProMedica Health System, Inc., an Ohio non-profit corporation ("<u>ProMedica</u>" or the "<u>Alternative Plan Sponsor</u>"), and MC Operations Investments, LLC (the "<u>OCP Holder</u>," together with ProMedica and the Majority Holders, the "<u>Supporting Parties</u>," and together with the Debtor, the "<u>Parties</u>").

RECITALS

WHEREAS, the Parties have engaged in arm's length, good faith discussions regarding a restructuring of the Debtor and certain of its subsidiaries (the "Restructuring");

WHEREAS, on the date hereof, the Debtor, ProMedica, Quality Care Properties, Inc. ("QCP"), and certain other parties, have entered into an Alternative Plan Sponsor Agreement (the "Alternative PSA"), which, among other things, contemplates the acquisition by a subsidiary of the Alternative Plan Sponsor of 100% of the equity of the reorganized Debtor through the attached First Amended Chapter 11 Plan of Reorganization for HCR ManorCare, Inc. (the "Plan"; capitalized terms not otherwise defined in this Agreement shall have the meaning given to such terms in the Alternative PSA); and

WHEREAS, the Supporting Parties intend to support the Restructuring and the Plan upon the terms and conditions set forth herein:

AGREEMENT

NOW, THEREFORE, in consideration of the conditional promises and mutual covenants and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

1. Representations and Warranties

- a. <u>Representations and Warranties of the Debtor</u>. The Debtor represents and warrants to the Supporting Parties that, as of the date hereof:
 - i. It has all requisite power and authority to enter into this Agreement and to carry out the transactions contemplated by, and perform its obligations under, this Agreement.
 - ii. The execution and delivery of this Agreement and the performance of its obligations hereunder have been duly authorized by all necessary action on its part.
 - iii. Subject to the provisions of sections 1125 and 1126 of the Bankruptcy Code, and except as set forth herein, this Agreement is the legally valid and binding obligation of it, enforceable against it in accordance with its terms, except as enforcement may be limited by applicable laws relating

to or limiting creditors' rights generally or by equitable principles relating to enforceability.

- b. <u>Representations and Warranties of the Supporting Parties.</u> Each Supporting Party represents and warrants to the Debtor and to each other Supporting Party that, as of the date hereof:
 - i. (A) Such Supporting Party is the sole record and beneficial owner of the shares of common stock of the Debtor set forth on its signature page hereto ("**Debtor Shares**"), (B) the Debtor Shares of such Supporting Party constitute the percentage of all shares of common stock of the Debtor set forth on such signature pages hereto, and (C) such Supporting Party has full power and authority to bind and act on behalf of, vote and consent to matters concerning such shares and to dispose of, exchange, assign and transfer such shares.
 - ii. Such Supporting Party has made no prior assignment, sale, participation, grant, conveyance or other transfer of, and has not entered into any other agreement to assign, sell, participate, grant, convey or otherwise transfer, in whole or in part, any portion of its right, title or interest in any of its Debtor Shares that conflict with the representations and warranties of such Supporting Party herein or would render such Supporting Party otherwise unable to comply with this Agreement and perform its obligations hereunder.
 - iii. This Agreement is the legally valid and binding obligation of each such Supporting Party, enforceable against it in accordance with its terms, except as enforcement may be limited by applicable laws relating to bankruptcy or limiting creditors' rights generally or by equitable principles relating to enforceability.

2. Covenants of the Majority Holders

Subject to the terms and conditions hereof and for so long as this Agreement has not been terminated in accordance with its terms, each Majority Holder, severally (and not jointly), agrees to, and to cause its direct and indirect subsidiaries, and its and their directors, officers, employees, agents and other representatives, to:

- a. <u>Support of Plan</u>. Support, and take all reasonable action necessary or reasonably requested by the Debtor to support and facilitate, the solicitation, confirmation and consummation of the Plan and the transactions contemplated by the Plan;
- b. <u>Vote for Plan</u>. (i) Vote all Equity Interests, Claims and other instruments carrying voting rights held or controlled, directly or indirectly, by such Majority Holder to accept the Plan (and any amendments, waivers, or consents required in connection with the Plan that are recommended or requested by the Debtor); (ii) timely deliver its duly executed and completed ballot promptly following the commencement of the solicitation of acceptances of the Plan, and (iii) not withdraw, change, or revoke (or cause to be withdrawn, changed, or revoked) its vote with respect to the Plan

- c. <u>No Competing Plans</u>. Other than for the Plan, not pursue, propose or support, or encourage the pursuit, proposal or support of, any chapter 11 plan or other restructuring or reorganization for, or the liquidation of, the Debtor (directly or indirectly);
- d. No Actions Against Plan. (i) Not take any action that would reasonably be expected to prevent, interfere with, delay or impede the solicitation of votes in connection with the Plan or the confirmation or consummation of the Plan, and (ii) not object to or otherwise commence, or encourage, join in or support any other person to commence, any proceeding or other analogous action opposing the Plan; and
- e. <u>Tax Matters</u>. Not file any federal or state tax return, or any amendment to such a return, claiming any deduction for worthlessness of its Debtor Shares.

3. Covenants of the QCP Holder

Subject to the terms and conditions hereof and for so long as this Agreement has not been terminated in accordance with its terms, the QCP Holder agrees to, and to cause its direct and indirect subsidiaries and affiliates to:

- a. support, and take all reasonable actions necessary or reasonably requested by the Debtor to facilitate the solicitation, confirmation and consummation of the Plan and the transactions contemplated by the Plan, including, without limitation, voting in favor of the Plan;
- b. not take any other action, directly or indirectly, that could prevent, interfere with, delay or impede the solicitation of votes in connection with the Plan or the confirmation or consummation of the Plan;
- c. not object to or otherwise commence, or encourage any other person to commence, any proceeding or take any action opposing the Plan or Disclosure Statement: and
- d. not file any federal or state tax return, or any amendment to such a return, claiming any deduction for worthlessness of its Debtor Shares.

4. Information about the Majority Holders

Subject to the terms and conditions hereof and for so long as this Agreement has not been terminated in accordance with its terms, each Majority Holder, severally (and not jointly), agrees to:

a. use its reasonable best efforts to provide such information requested by the Alternative Plan Sponsor, QCP or the Debtor as is necessary regarding such Majority Holder for the Alternative Plan Sponsor, QCP, the Debtor and their affiliates to make or obtain any filings, notices, consents, registrations, approvals, permits or authorizations (including those with respect to state licensing required to operate the Debtor's businesses) in connection with the Plan or the transactions contemplated thereby; and

b. upon the reasonable request of the Debtor, QCP or the Alternative Plan Sponsor, use its reasonable best efforts to provide such information concerning such Majority Holder in connection with any filings, notices, consents, registrations, approvals, permits or authorizations (including those with respect to state licensing required to operate the Debtor's businesses) in connection with the Plan or the transactions contemplated thereby.

5. Transfer of Shares

Each Supporting Party agrees that, so long as this Agreement has not been terminated in accordance with its terms, it shall not directly or indirectly sell, pledge, hypothecate or otherwise transfer or dispose of or grant, issue or sell any option, right to acquire, voting, participation or other interest in any Debtor Shares (each a "Transfer"), unless the transferee thereof, prior to such Transfer, agrees in writing for the benefit of the Parties to become subject to the terms and conditions of this Agreement as a "Supporting Party" and to be bound by this Agreement by executing the joinder attached hereto as Exhibit B (the "Joinder Agreement"), and delivering an executed copy thereof, within two (2) business days of such execution, to the Debtor, in which event (i) the transferee shall be deemed to be a Supporting Party hereunder and (ii) the transferor shall be deemed to relinquish its rights and be released from its obligations under this Agreement to the extent of such transferred rights and obligations. Each Supporting Party agrees that any Transfer that does not comply with the foregoing shall be deemed void ab initio, and the Debtor and each other Supporting Party shall have the right to avoid such Transfer. This Agreement shall in no way be construed to preclude any shareholder from acquiring additional shares; provided that any such additional shares shall, upon acquisition, automatically be deemed to be subject to all the terms of this Agreement.

6. ProMedica Support

ProMedica shall, and shall cause each of its Subsidiaries to, (i) support, and take all reasonable actions necessary or reasonably requested by the Debtor to facilitate, the solicitation, confirmation and consummation of the Plan and the transactions contemplated by the Plan; (ii) not take any other action, directly or indirectly, that could prevent, interfere with, delay or impede the solicitation of votes in connection with the Plan or the confirmation or consummation of the Plan; (iii) not object to or otherwise commence any proceeding or take any action opposing the Plan or Disclosure Statement; and (iv) otherwise use its reasonable best efforts to assist the Debtor in obtaining entry of the Confirmation Order by the Bankruptcy Court within the deadline set forth in Section 3.3(c) of the Alternative Plan Sponsor Agreement.

7. Additional Supporting Parties

A shareholder that is not a Supporting Party as of the date of this Agreement will become a Party to this Agreement as a Supporting Party on the date that it agrees in writing, for the benefit of the Parties, to become subject to the terms and conditions of this Agreement as a "Supporting Party" and to be bound by this Agreement by executing the Joinder Agreement and delivering an executed copy thereof, within two (2) business days of such execution, to the

Debtor. Upon such delivery, such Supporting Party shall immediately thereafter send a copy of such Joinder Agreement to all other Supporting Parties.

8. Termination

This Agreement and all obligations of the Parties hereunder shall immediately terminate and be of no further force and effect as follows:

- a. upon the Effective Date of the Plan; and
- b. upon the termination of the Alternative PSA and the Original PSA.

Upon termination of this Agreement in accordance with its terms, this Agreement shall forthwith become void and of no further force or effect, each Party shall be released from its commitments, undertakings and agreements under or related to this Agreement, and there shall be no liability or obligation on the part of any Party.

9. No Monetary Liability.

Notwithstanding anything to the contrary contained in this Agreement or provided for under any applicable law, no Party shall be liable to any other person, either in contract or in tort, for any money damages relating to any breach of this Agreement.

10. Specific Performance

It is understood and agreed by the Parties that money damages would not be a sufficient or appropriate remedy for any breach of this Agreement by any Party and each non-breaching Party shall be entitled to specific performance and injunctive or other equitable relief as a remedy of any such breach. Each Party agrees to waive any requirement for the securing or posting of a bond in connection with such remedy.

11. Entire Agreement; Prior Negotiations

This Agreement, including all exhibits attached hereto, constitutes the entire agreement of the Parties and supersedes all prior negotiations and documents reflecting such prior negotiations between and among the Parties (and their respective advisors) with respect to the subject matter of this Agreement.

12. Amendments

Except as otherwise provided herein, this Agreement may not be modified, amended or supplemented without prior written agreement signed by the Debtor, each Supporting Party and the Alternative Plan Sponsor.

13. Governing Law

This Agreement shall be governed by, and construed in accordance with, the internal laws of the State of Delaware. By its execution and delivery of this Agreement, each of the parties hereto hereby irrevocably and unconditionally agrees for itself that any legal action, suit or proceeding against it with respect to any matter under or arising out of or in connection with this Agreement or for recognition or enforcement of any judgment rendered in any such action, suit or proceeding, may be brought in either a state or federal court of competent jurisdiction

in the State of Delaware. By execution and delivery of this Agreement, each of the parties hereto hereby irrevocably accepts and submits itself to the nonexclusive jurisdiction of each such court, generally and unconditionally, with respect to any such action, suit or proceeding. Notwithstanding the foregoing, each of the parties hereto hereby agrees that, while the Bankruptcy Case is pending, the Bankruptcy Court shall have exclusive jurisdiction of all matters arising out of or in connection with this Agreement.

14. Effectiveness

This Agreement shall become effective and binding upon each Party upon the execution and delivery of this Agreement by such Party.

15. No Solicitation

Notwithstanding anything to the contrary, this Agreement is not and shall not be deemed to be (a) a solicitation of consents to the Plan or (b) an offer for the issuance, purchase, sale exchange, hypothecation or other transfer of securities or a solicitation of an offer to purchase or otherwise acquire securities for purposes of the Securities Act of 1933 and the Securities Exchange Act of 1934, each as amended.

16. Third-Party Beneficiary

This Agreement is intended for the benefit of the parties hereto and no other person shall have any rights hereunder; <u>provided</u>, <u>however</u>, that QCP shall be an express third party beneficiary of this Agreement, entitled to enforce this Agreement against the Parties as if it were itself a party hereto.

17. Counterparts

This Agreement may be executed in several counterparts, each of which shall be deemed to be an original, and all of which together shall be deemed to be one and the same agreement. Execution copies of this agreement may be delivered by electronic mail (in ".pdf" or ".tif" format), facsimile or otherwise, which shall be deemed to be an original for the purposes of this Agreement.

18. [Intentionally Omitted]

19. No Waiver of Participation and Preservation of Rights

Except as provided in this Agreement, nothing herein is intended to, does or shall be deemed in any manner to waive, limit, impair or restrict the ability of any Party to protect and preserve its rights, remedies and interests, including, but not limited to, claims against the Debtor, liens or security interests it may have in any assets of the Debtor, or its rights to participate fully in the Bankruptcy Case. Without limiting the foregoing sentence in any way, if this Agreement is terminated in accordance with its terms for any reason, the Parties each fully reserve any and all of their respective rights, remedies and interests.

20. Notices

All notices hereunder shall be deemed given if in writing and delivered, if sent by facsimile, courier or by registered or certified mail (return receipt requested) to the following

addresses and facsimile numbers (or at such other addresses or facsimile numbers as shall be specified by like notice):

If to the Debtor, to counsel at the following address:

Sidley Austin LLP One South Dearborn Chicago, IL 60603 Attention: Larry Nyhan Email: lnyhan@sidley.com

If to the Alternative Plan Sponsor to counsel at the following address:

Shumaker, Loop & Kendrick, LLP 1000 Jackson Street Toledo, OH 43604 Attention: James I Rothschild David Coyle

Email: jrothschild@slk-law.com dcoyle@slk-law.com

If to the Majority Holders, to counsel at the following address:

Latham & Watkins LLP
555 Eleventh Street, NW, Suite 1000
Washington, D.C. 20004-1304
Attention: Daniel T. Lennon
Roger G. Schwartz
J. Cory Tull
Email: daniel.lennon@lw.com

Email: daniel.lennon@lw.com roger.schwartz@lw.com cory.tull@lw.com

If to the QCP Holder, to counsel at the following address:

Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019

Attention: Adam O. Emmerich and Scott K. Charles Email: aoemmerich@wlrk.com skcharles@wlrk.com

[Signature Pages Follow]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed and delivered by their respective duly authorized officers or other agents, solely in their respective capacity as officers or other agents of the undersigned and not in any other capacity, as of the date first set forth above.

HCR MANORCARE, INC.

Name: John R. Castellano

Title: Chief Restructuring Officer

CARLYLE MC PARTNERS, L.P

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.C., its general partner

By:_____

Name: Jeremy Anderson Title: Authorized Signatory

Debtor Shares: 6,908,455

Debtor Shares Percentage: 15.377%

CARLYLE PARTNERS V-A MC, L.P.

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.C., its general partner

By:_____

Name: Jeremy Anderson Title: Authorized Signatory

Debtor Shares: 527,141

Debtor Shares Percentage: 1.173%

CARLYLE PARTNERS V MC, L.P.

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.C., its general partner

By:

Name: Jeremy Anderson Title: Authorized Signatory

Debtor Shares: 26,089,114

Debtor Shares Percentage: 58.072%

CP V COINVESTMENT A, L.	CP	\mathbf{V}	COIN	IVES	TMEN	TA.	. L.F
-------------------------	-----------	--------------	------	------	------	-----	-------

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.C., its general partner

By:____

Name: Jeremy Anderson Title: Authorized Signatory

Debtor Shares: 1,015,490

Debtor Shares Percentage: 2.260%

CP V COINVESTMENT B, L.P.

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.C., its general partner

By:_____

Name: Jeremy Anderson Title: Authorized Signatory

Debtor Shares: 129,357

Debtor Shares Percentage: 0.288%

PROMEDICA HEALTH SYSTEM, INC.

Name: Title:

Randy Obstra

President & CEO

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 206 of 441

	MC OPERATIONS	INVESTMENTS.
--	---------------	--------------

LLC

Name: C. Marc Richards Title: Chief Financial Officer

Debtor Shares: 4,232,244

Debtor Shares Percentage:

EXHIBIT A

<u>Plan</u>

EXHIBIT B

Joinder Agreement

Form of Former CEO Settlement Agreement

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 210 of 441

CONFIDENTIAL FOR DISCUSSION AND SETTLEMENT PURPOSES ONLY SUBJECT TO FRE 408 AND STATE LAW ANALOGUES

[Address]

[•], 2018

Dear Mr. Ormond,

Reference is made to (i) the Plan Sponsor Agreement, dated as of [•], 2018, among Quality Care Properties, Inc. ("QCP"), HCR ManorCare, Inc. (the "Company"), and certain other parties (together with its related exhibits, the "PSA") (ii) the Amended and Restated Employment Agreement between you and the Company, dated as of April 7, 2011 (as amended, the "Employment Agreement"), (iii) the HCR ManorCare Senior Management Savings Plan for Corporate Officers (as amended and restated, the "SMSPCO") and (iv) that certain Letter Agreement between you and the Company regarding your employment and compensation arrangements with the Company and its subsidiaries, dated as of April 7, 2011 (the "Letter Agreement"). The Employment Agreement, the SMSPCO and the Letter Agreement are collectively referred to herein as the "Documents."

This agreement (the "Settlement Letter") is intended to resolve any outstanding disputes between yourself and the Company regarding amounts that you have claimed are owed to you by the Company or its subsidiaries in connection with your employment by the Company. As to the matters described herein and except as otherwise provided herein, this Settlement Letter supersedes any provision contained in any of the Documents or any other written or unwritten agreement, plan or arrangement of the Company or any of its subsidiaries. By countersigning this Settlement Letter in the space provided below, you indicate your acknowledgement of, agreement with and consent to the following:

- 1. Effectiveness of Settlement Letter Conditioned Upon Closing. The parties agree that the effectiveness of this Settlement Letter shall occur on the occurrence of the Closing (as such term is defined in the PSA). If the PSA terminates prior to the Closing, this Settlement Letter shall be void *ab initio* and be of no further force or effect.
- **2. Settlement Amounts.** On the later of (a) the Effective Date, and (b) the Closing Date (as such term is defined in the PSA), you shall be entitled to the following amounts and payments (the "Settlement Amounts"):
- A. In respect of your claimed interest in the Frozen SERP (as such term is defined in the Letter Agreement) you will be entitled to receive your full benefit under the Frozen SERP, which was approximately \$60,763,043 on January 31, 2018 and is subject to change based on investment changes as provided in the Letter Agreement;

- B. In respect of your claimed interest in the SMSPCO, you will be entitled to receive your full benefit under the SMSPCO, which was approximately \$42,040,673 on January 31, 2018 and subject to change based on investment changes as provided in the SMSPCO.
- C. In respect of your claimed interest in the SERP Replacement (as such term is defined in the Letter Agreement), you will be entitled to receive a cash payment equal to \$9,875,000.
- D. In respect of your claimed entitlement to severance under the Employment Agreement, you will be entitled to a cash payment in the amount of \$3,978,000.
- E. Continued payment of your medical, dental and vision premiums during the Continuation Period, as provided in your Employment Agreement.
- F. Any 280G Gross-Up Payment as provided in Section 8 of your Employment Agreement.
- G. Continued use of office space, furnishings and secretarial support services during the Continuation Period, as provided in your Employment Agreement, at a reasonable cost to the Company; provided that such accommodations shall be provided at a location other than the premises of the Company that has a reasonable commute to your primary residence (which location shall be subject to prior notice and consultation with you) and shall be provided at a reasonably comparable quality to such accommodations as were made available to you prior to your termination of employment with the Company.

3. Release.

You knowingly and voluntarily waive, terminate, cancel, release and discharge forever the Company and each of its subsidiaries, as well as their respective current and former stockholders (including, for the avoidance of doubt, QCP), beneficial owners of their stock, their current or former officers, directors, employees, members, attorneys and agents, and their predecessors, successors, and assigns, individually and in their official capacities (together, the "Company Released Parties") from any and all suits, actions, causes of action, claims, allegations, rights, obligations, liabilities, demands, entitlements or charges (collectively, "Claims") that you (or your heirs, executors, administrators, successors and assigns) has or may have, whether known, unknown or unforeseen, vested or contingent, by reason of any matter, cause or thing occurring at any time before and including the date of this Settlement Letter, including all claims arising under or in connection with your employment or termination of employment with the Company, including, without limitation: Claims under United States federal, state or local law and the national or local law of any foreign country (statutory or decisional), for wrongful, abusive, constructive or unlawful discharge or dismissal, for breach of any contract, or for discrimination based upon race, color, ethnicity, sex, age, national origin, religion, disability, sexual orientation, or any other unlawful criterion or circumstance, including rights or Claims under the Age Discrimination in Employment Act of 1967 ("ADEA"), the Older Workers Benefit Protection Act of 1990 ("OWBPA"), violations of the Equal Pay Act, Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1991, the Americans with Disabilities Act of 1991, the Employee Retirement Income Security Act of 1974 ("ERISA"), the Fair Labor

Form of Separation Agreement

CONFIDENTIAL

SEPARATION AGREEMENT, GENERAL RELEASE AND WAIVER OF

CLAIMS (this "*Release*"), in entered into by and between [●] ("*Employee*"), Quality Care Properties, Inc., a Maryland corporation ("*QCP*"), and HCR ManorCare, Inc., a Delaware corporation ("*HCRMC*", and together with QCP and their respective subsidiaries, the "*Company*"). The Company and the Company's affiliates, stockholders, beneficial owners of its stock, its current or former officers, directors, employees, members, attorneys and agents, and their predecessors, successors and assigns, individually and in their official capacities shall be referred to herein as the "*Released Parties*".

WHEREAS, Employee has been employed as [title] of HCRMC;

WHEREAS, QCP, HCRMC, and certain other entities have entered into that certain Plan Sponsor Agreement (the "*PSA*") dated as of [•], 2018;

WHEREAS, Employee and HCRMC have entered into that certain Employment Agreement, dated [•] (the "Employment Agreement");

WHEREAS, Employee's employment with the Company shall terminate by the Company without Cause (as defined in the Employment Agreement) as of the Closing (as defined in the PSA) (the "*Termination Date*"), which termination constitutes a Qualifying Termination under the Employment Agreement entitling Employee to "severance after a Change in Control" pursuant to Section 6 of the Employment Agreement; ¹ and

WHEREAS, Section 1.5(a) of the PSA provides that (i) QCP shall fund and pay, or cause to be funded and paid, to the Employee certain Excess Severance Payments (as such term is defined in the PSA) in the amounts set forth in Schedule 1, subject to certain terms and conditions as described therein, including the execution of this Release and (ii) QCP shall pay, or cause to be paid, certain other obligations set forth in Schedule 1.²

NOW, THEREFORE, in consideration of the covenants and agreements hereinafter set forth, the parties agree as follows:

1. General Release. Except as to the Surviving Claims (defined in Section 2 below), Employee knowingly and voluntarily waives, terminates, cancels, releases and discharges forever the Released Parties from any and all suits, actions, causes of action, claims, allegations, rights, obligations, liabilities, demands, entitlements or charges (collectively, "Claims") that Employee (or Employee's heirs, executors, administrators, successors and assigns) has or may have, whether known, unknown or unforeseen, vested or contingent,

US-DOCS\99184296.10

Note: To provide for payments upon the Closing Date, releases will be executed shortly before Closing Date so that non-revocability period has expired as of Closing Date.

Note: Schedule 1 to reflect the Excess Severance Amounts and all other amounts or benefits required to be paid, other than pursuant to ERISA plans.

by reason of any matter, cause or thing occurring at any time before and including the date of this Release, including all claims arising under or in connection with Employee's employment or termination of employment with the Company, including, without limitation: Claims under United States federal, state or local law and the national or local law of any foreign country (statutory or decisional), for wrongful, abusive, constructive or unlawful discharge or dismissal, for breach of any contract, or for discrimination based upon race, color, ethnicity, sex, age, national origin, religion, disability, sexual orientation, or any other unlawful criterion or circumstance, including rights or Claims under the Age Discrimination in Employment Act of 1967 ("ADEA"), the Older Workers Benefit Protection Act of 1990 ("OWBPA"), violations of the Equal Pay Act, Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1991, the Americans with Disabilities Act of 1991, the Employee Retirement Income Security Act of 1974 ("ERISA"), the Fair Labor Standards Act, the Worker Adjustment Retraining and Notification Act, the Family Medical Leave Act, including all amendments to any of the aforementioned acts; and violations of any other federal, state, or municipal fair employment statutes or laws, including, without limitation, violations of any other law, rule, regulation, or ordinance pertaining to employment, wages, compensation, hours worked, or any other Claims for compensation or bonuses, whether or not paid under any compensation plan or arrangement; breach of contract; tort and other common law Claims; defamation; libel; slander; impairment of economic opportunity defamation; sexual harassment; retaliation; attorneys' fees; emotional distress; intentional infliction of emotional distress; assault; battery, pain and suffering; and punitive or exemplary damages (the "Released Matters"). In addition and except as to Surviving Claims, in consideration of the provisions of this Release, Employee further agrees to waive any and all rights under the laws of any jurisdiction in the United States, or any other country, that limit a general release to those Claims that are known or suspected to exist in Employee's favor as of the Effective Date (as defined below).

- 2. <u>Surviving Claims</u>. On the Payment Date (as defined in the PSA), QCP shall pay or cause to be paid to Employee the amounts required to be paid to Employee pursuant to Section 1.5 of the PSA. Notwithstanding anything herein to the contrary, Employee's Release set forth in Section 1 above shall not:
 - (i) waive or release any Claims for payment of amounts or benefits payable as set forth on <u>Schedule 1</u>³ hereto;
 - (ii) waive or release any Claim to enforce this Release;
 - (iii) waive or release any Claim for vested employee benefits under plans covered by ERISA;
 - (iv) waive or release any Claim that may not lawfully be waived;

Note: Schedule 1 to reflect the Excess Severance Amounts and all other amounts or benefits required to be paid (including pursuant to Section 1.5 of the PSA and the Plan), other than pursuant to plans subject to ERISA.

- (v) waive or release Claims described in Section 4(a) of this Release;
- (vi) waive or release any Claim for indemnification, advancement, and D&O insurance, including Claims provided for pursuant to Section 4.7 of the PSA; or
- (vii) limit Employee's rights under applicable law to provide truthful information to any governmental or regulatory entity or to file a charge with or participate in an investigation conducted by any governmental or regulatory entity.

Notwithstanding the foregoing, Employee agrees to waive Employee's right to recover monetary damages in connection with any charge, complaint or lawsuit unrelated to any Claim described in clauses (i), (ii) and (iv) of the immediately preceding sentence filed by Employee or anyone else on Employee's behalf (whether involving a governmental entity or not) with respect to Released Matters; provided that Employee is not agreeing to waive, and this Release shall not be read as requiring Employee to waive, any right Employee may have to receive an award for information provided to any governmental or regulatory entity or respond to a subpoena or other legal requirement.

- 3. <u>Additional Representations</u>. Employee further represents and warrants that Employee has not filed any civil action, suit, arbitration, administrative charge, or legal proceeding against any Released Party nor has Employee assigned, pledged, or hypothecated as of the Effective Date any Claim to any person and no other person has an interest in the Claims that he is releasing.
- **4.** Acknowledgements by Employee. Employee acknowledges and agrees that Employee has read this Release in its entirety and that this Release is a general release of all known and unknown Claims with respect to Released Matters. Employee further acknowledges and agrees that:
 - a. this Release does not release, waive or discharge any rights or Claims that may arise for actions or omissions after the Effective Date of this Release and Employee acknowledges that he is not releasing, waiving or discharging any ADEA Claims that may arise after the Effective Date of this Release;
 - b. Employee is entering into this Release and releasing, waiving and discharging rights or Claims only in exchange for consideration which he is not already entitled to receive;
 - c. Employee's receipt of the Excess Severance Payments is subject to the terms and conditions set forth in Section 1.5 of the PSA (and, for the avoidance of doubt, the non-occurrence of any of the events set forth in Section 1.5(b) of the PSA that by their terms would terminate QCP's obligation to pay such Excess Severance Payments pursuant to Section 1.5(b) of the PSA);

- d. Employee has been advised, and is being advised by the Release, to consult with an attorney before executing this Release; Employee acknowledges that he has consulted with counsel of his choice concerning the terms and conditions of this Release; and
- e. Employee has been advised, and is being advised by this Release, that he has been given at least twenty-one (21) days within which to consider the Release, but Employee can execute this Release at any time prior to the expiration of such review period.
- 5. Effective Date of Release: Employee is aware that Sections 1,2,3,4,6 and 7 of this Release shall become null and void if he revokes his agreement to this Release within seven (7) days following the date of execution of this Release (or, if later, prior to the Closing Date). Employee may revoke this Release (solely with respect to Sections 1,2,3,4,6 and 7 hereof) at any time during such period by delivering (or causing to be delivered) to the Company written notice of his revocation of this Release no later than the later to occur of (i) 5:00 p.m. Eastern time on the seventh (7th) full day following the date of execution of this Release and (ii) the Closing (the "Effective Date"). Employee agrees and acknowledges that a notice of revocation that is not received by such date and time will be invalid and will not revoke this Release with respect to Sections 1 through 4 hereof. The Excess Severance Payments shall be paid to Employee upon the later to occur of the day following the Effective Date and the Termination Date. 4 If Employee fails to execute this Release on or prior to the 21st day following receipt of this Release or exercises his right to revoke Sections 1,2,3,4,6 and 7 of this Release prior to the Effective Date, he shall forfeit his right to receive any portion of the Excess Severance Payments. For the avoidance of doubt, Sections 8-17 of this Release shall survive Employee's revocation of his agreement to this Release in accordance with this Section 5 and the payment of the remaining benefits under Schedule 1 shall not be affected by such revocation.
- 6. Company/QCP Release. The Company and QCP on behalf of themselves and each of their affiliates and any of its or their successors and assigns, hereby irrevocably and unconditionally waives and releases any Claims, both known and unknown, in law or in equity, which the Company, QCP or any of their subsidiaries or affiliates ever had, now has or may have against Employee, Employee's family members and any of their respective past or present successors, heirs, assigns, or agents (collectively, the "Employee Released Party"), including, but not limited to, claims that in any way relate to: (1) the Employee's employment or service as an officer or director with the Company or any of its affiliates; (2) any claims to attorneys' fees or other indemnities; and (3) any federal, state, local or foreign laws and regulations. In addition, in consideration of the provisions of this Release, the Company and QCP further agree on behalf of themselves and each of their affiliates to waive any and all rights under the laws of any jurisdiction in the United States, or any other country, that limit a general release to those Claims that

Note: Releases to be provided in advance of Closing and can be executed 8 days prior to Closing so that Effective Date occurs on or before Closing.

are known or suspected to exist in its favor as of the date of this Release, except for the Company/QCP Unreleased Claims. The Company and QCP further represent and warrant on behalf of themselves and their affiliates that none of them has filed any civil action, suit, arbitration, administrative charge, or legal proceeding against any Employee Released Party nor have any of them assigned, pledged, or hypothecated as of the Termination Date any Claim to any person and no other person has an interest in the Claims that the Company or QCP is releasing. Notwithstanding anything herein in to the contrary, neither the Company nor QCP waives or releases any claims with respect to (i) any Claim that may not lawfully be waived, (ii) any right or claim that arises after the date this Release is executed, or (iii) any Claim to enforce this Release (the "Company/QCP Unreleased Claims").

- 7. <u>Company Acknowledgement</u>. The Company acknowledges and agrees that an authorized representative of the Company has read this Release in its entirety and that this Release is a general release of all known and unknown Claims with respect to the released matters. The Company is entering into this Release and releasing, waiving and discharging rights or Claims only in exchange for consideration which it is not already entitled to receive.
- 8. Officer and Director Resignations. Employee hereby confirms his resignation, effective upon the Termination Date, as a [member of the Board of Directors of the Company] and as an officer of the Company and any other officer or director position Employee held with the Company or any of its subsidiaries, affiliates, joint ventures or other related entities, and agrees to execute such documents as the Company reasonably deems to be appropriate to facilitate such resignations.

9. Cooperation.

- a. For two years following the Termination Date, Employee agrees to make himself reasonably available to the Company upon reasonable advance notice to respond to requests by the Company for information concerning any litigation, regulatory inquiry or investigation, involving facts or events relating to the Company that may reasonably be within his knowledge. Employee will reasonably cooperate with the Company in connection with any and all future litigation or regulatory proceedings brought by or against the Company to the extent the Company reasonably deems Employee's cooperation necessary and such cooperation does not interfere with performance of Employee's duties to Employee's then current employer. Employee will be entitled to reimbursement of reasonable out-of-pocket expenses incurred in connection with fulfilling his obligations under this Section 9(a) and a per hour rate of \$350. This Section 9(a) is subject to Section 2(vii) above.
- b. For two years following the Termination Date, Company agrees to make itself, its appropriate representatives, and Company records and information reasonably available to Employee upon reasonable advance notice to respond to requests by Employee for information concerning any matters involving Employee, including, without limitation, any litigation, governmental or regulatory inquiry or

investigation, involving facts or events relating to Employee that may be eligible for indemnification under Section 4.7 of the PSA.

10. Affirmation of Restrictive Covenants; Return of Property. Employee agrees and acknowledges that his obligations under Section [10] of the Employment Agreement (collectively, the "Restrictive Covenants"), will continue to apply after the Termination Date as specified in such provisions. Employee affirms that such Restrictive Covenants are not unduly burdensome to him and are reasonably necessary to protect the legitimate interests of the Company.

Employee covenants to return to the Company all property or information, including, without limitation, all reports, files, memos, plans, lists, or other records (whether electronically stored or not) belonging to the Company, including copies, extracts or other documents derived from such property or information. The Company agrees to (a) to the extent identified by Employee to QCP prior to Closing, maintain copies of financial and other information disclosed to QCP prior to the Effective Date that Employee reasonably believes may be relevant to the shareholder and derivative actions instituted against QCP and the Company prior to, on or following the Effective Date, and (b) make available to Employee such information to the extent reasonably necessary for Employee to defend himself against actions instituted against QCP or any of its affiliates, the Company or any of its affiliates or Employee.

- 11. Non-Disparagement. Each party (which, in the case of the Company, shall mean the officers and the members of the board of directors of QCP and HCRMC) agrees to refrain from Disparaging (as defined below) the other party and its affiliates, including, in the case of the Company, any of its services, technologies or practices, or any of its directors, officers, agents, representatives or stockholders (including QCP), either orally or in writing. Nothing in this paragraph shall preclude any party from making truthful statements that are reasonably necessary to comply with applicable law, regulation or legal process, to defend or enforce a party's rights under this Release, or that are subject to Section 2(vii) above. For purposes of this Release, "Disparaging" means making remarks, comments or statements, whether written or oral, that materially impugn the character, integrity, reputation or abilities of the person being disparaged with the knowledge on the part of the person making such remarks, comments or statements that such remarks, comments or statements would so materially impugn the character, integrity, reputation or abilities of the person being disparaged.
- 12. Non-Admission. Nothing in this Release is intended to or shall be construed as an admission by the Company or any of the other Released Parties that any of the them violated any law, interfered with any right, breached any obligation or otherwise engaged in any improper or illegal conduct with respect to Employee or otherwise. The Company and the other Released Parties expressly deny any such illegal or wrongful conduct. Nothing in this Release is intended to or shall be construed as an admission by Employee that he violated any law, interfered with any right, breached any obligation or otherwise engaged in any improper or illegal conduct with respect to the Company or otherwise. Employee expressly denies any such illegal or wrongful conduct.

- **13.** <u>Amendment</u>. This Release may be modified only in a written agreement signed by the parties, and any party's failure to enforce this Release in the event of one or more events which violate this Release shall not constitute a waiver of any right to enforce this Release against subsequent violations.
- **14.** Governing Law. To the extent not subject to federal law, this Release shall be governed in accordance with the laws of Delaware without reference to principles of conflict of laws.
- **15.** <u>Severability</u>. If any provision of this Release should be declared to be unenforceable by any administrative agency or court of law, then remainder of the Release shall remain in full force and effect.
- **16.** <u>Captions</u>; <u>Section Headings</u>. Captions and section headings used herein are for convenience only and are not a part of this Release and shall not be used in construing it.
- 17. <u>Counterparts</u>; <u>Facsimile Signatures</u>. This Release may be executed in any number of counterparts, each of which when so executed and delivered shall be deemed an original instrument without the production of any other counterpart. Any signature on this Release, delivered by either party by photographic, facsimile or PDF shall be deemed to be an original signature thereto.

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 220 of 441

IN WITNESS WHEREOF, Employee has s	signed this Release on this of
·	
Accepted and Agreed:	
WGD M. G. A	
HCR ManorCare, Inc.	
D.,,	
By:	
Accepted and Agreed:	
Quality Care Properties, Inc.	
Quanty Care Properties, Inc.	
Ву:	

Schedule 1

Excess Severance Payments

• Severance pursuant to Section [6(a)(i)] of the Employment Agreement: \$[•]⁵

Other Obligations

- Employee's accrued unpaid wages, employee benefits, and unreimbursed business expenses through and including the Termination Date.
- Employee's participant balance in the Senior Management Savings Plan for Corporate Officers (the "SMSPCO"): \$[●], which is subject to change based on investment changes as provided in the SMSPCO.
- Pay the full premium payments for group medical, dental and vision benefits continuation pursuant to Section [6(a)(ii)] of the Employment Agreement under the Company's group medical, dental and vision plans, which coverage will continue for three years following termination, cover Employee and his eligible dependents and will be primary coverage.
- [Lump-sum amount payable to Employee in connection with SERP: \$[•].]⁶
- KEIP: [●]
- [Retention Payment: \$250,000]⁷

Payment Timing: All payments will be paid to Employee as provided in the PSA.

All payments are subject to withholding of applicable Taxes as required by Law (as such terms are defined in the PSA).

Note: To reflect the applicable amount for each relevant Employee.

Note: Cavanaugh only.

Note: Cavanaugh only.

Standards Act, the Worker Adjustment Retraining and Notification Act, the Family Medical Leave Act, including all amendments to any of the aforementioned acts; and violations of any other federal, state, or municipal fair employment statutes or laws, including, without limitation, violations of any other law, rule, regulation, or ordinance pertaining to employment, wages, compensation, hours worked, or any other Claims for compensation or bonuses, whether or not paid under any compensation plan or arrangement; breach of contract; tort and other common law Claims; defamation; libel; slander; impairment of economic opportunity defamation; sexual harassment; retaliation; attorneys' fees; emotional distress; intentional infliction of emotional distress; assault; battery, pain and suffering; and punitive or exemplary damages (the "Ormond Released Matters"). In addition, in consideration of the provisions of this Settlement Letter, you further agree to waive any and all rights under the laws of any jurisdiction in the United States, or any other country, that limit a general release to those Claims that are known or suspected to exist in your favor as of the date of this Settlement Letter, except for the Ormond Unreleased Claims. You further represent and warrant that you have not filed any civil action, suit, arbitration, administrative charge, or legal proceeding against any Company Released Party nor, have you assigned, pledged, or hypothecated as of the Effective Date any Claim to any person and no other person has an interest in the Claims that you are releasing. You acknowledge and agree that you have read this Settlement Letter in its entirety and that this Section 3.A is a general release of all known and unknown Claims. You further acknowledge and agree that:

- (i) this Release does not release, waive or discharge any rights or Claims that may arise for actions or omissions after the Effective Date and you acknowledge that you are not releasing, waiving or discharging any ADEA Claims that may arise after the Effective Date;
- (ii) you are entering into this Settlement Letter and releasing, waiving and discharging rights or Claims in exchange for consideration that is presently disputed by the Company and other good and valuable consideration:
- (iii) you have been advised, and are being advised by the Settlement Letter, to consult with an attorney before executing the Settlement Letter; you acknowledge that you have consulted with counsel of his choice concerning the terms and conditions of this Settlement Letter;
- (iv) you have been advised, and are being advised by this Settlement Letter, that you have been given at least twenty-one (21) days within which to consider the Settlement Letter, but you can execute this Settlement Letter at any time prior to the expiration of such review period; and
- (v) you are aware that this Settlement Letter shall become null and void if you revoke your agreement to this Settlement Letter within seven (7) days following the date of execution of this Settlement Letter. You may revoke this Settlement Letter at any time during such seven-day period by delivering (or causing to be delivered) to the Company written notice of your revocation of this Settlement Letter no later than 5:00 p.m. Eastern time on the seventh (7th) full day following the date of execution of this

Release (the "<u>Effective Date</u>"). You acknowledge that a letter of revocation that is not received by such date and time will be invalid and will not revoke this Release.

- B. Ormond Surviving Claims. Notwithstanding anything herein to the contrary, by executing this Settlement Letter you do not: (i) release any Claim that may not lawfully be waived; (ii) become subject to any prohibition from reporting possible violations of federal law or regulation or making other disclosures that are protected under (or claiming any award under) the whistleblower provisions of federal law or regulation; or (iii) release any Claim to enforce this Settlement Letter (collectively, the "Ormond Unreleased Claims").
- Company and QCP Release. The Company and QCP on behalf of themselves and each of their subsidiaries and any of its or their successors and assigns, hereby irrevocably and unconditionally waives and releases any Claims, both known and unknown, in law or in equity, which the Company, QCP or any of their subsidiaries or affiliates ever had, now has or may have against you, your family members and their respective past or present successors, heirs, assigns, or agents (collectively, the "Ormond Released Party"), including, but not limited to, claims that in any way relate to: (1) the your employment with the Company; (2) any claims to attorneys' fees or other indemnities; and (3) any federal, state, local or foreign laws and regulations. In addition, in consideration of the provisions of this Settlement Letter, the Company and QCP further agree to waive any and all rights under the laws of any jurisdiction in the United States, or any other country, that limit a general release to those Claims that are known or suspected to exist in its favor as of the date of this Settlement Letter, except for the Company/QCP Unreleased Claims. The Company and QCP further represent and warrant that they have not filed any civil action, suit, arbitration, administrative charge, or legal proceeding against any Ormond Released Party nor has the Company or QCP assigned, pledged, or hypothecated as of the Effective Date any Claim to any person and no other person has an interest in the Claims that the Company or QCP is releasing.
- D. <u>Company and QCP Surviving Claims</u>. Notwithstanding anything herein in to the contrary, neither the Company nor QCP waives or releases claims with respect to (i) any Claim that may not lawfully be waived, (ii) any right or claim that arises after the date this Settlement Letter is executed, or (iii) any Claim to enforce this Settlement Letter (the "<u>Company/QCP Unreleased Claims</u>").
- **4. Restrictive Covenants.** You agree that you will continue to be subject to Section 10 of the Employment Agreement (Non-Competition/Non-Solicitation and Confidentiality) for the applicable periods set forth therein.
- **5. Non-Disparagement**. You agree not to make any defamatory or derogatory statements concerning the Company, QCP or any of their respective affiliates or predecessors or their respective directors, officers and employees. The Company and QCP agree to instruct their executive officers and directors not to make any defamatory restrict or limit you, the Company, QCP or any of the Company's or QCP's executive officers or directors from providing truthful information in response to a subpoena, other legal process or to a governmental or regulatory body or in the event of litigation between you and the Company, QCP or their respective affiliates.

- **6. Acknowledgement**. You acknowledge that you have read and understand this Settlement Letter, are fully aware of its terms and legal effect and have not acted in reliance upon any representations or promises made by the Company or any other party other than those contained in writing herein. You are advised to consult your personal tax and legal advisors in connection with the matters described in this Settlement Letter, and by countersigning this Settlement Letter in the space provided below, you represent and warrant that you have given the opportunity to consult with, and have consulted with, such advisors to the extent deemed advisable by you.
- **7. Withholding**. The Company or one of its subsidiaries has the authority to deduct or withhold from any compensation payable to you, whether or not such compensation is described herein, or to require you to remit to the Company, an amount sufficient to satisfy any applicable federal, state, local and foreign taxes (including any FICA obligation) required by law to be withheld with respect to any compensation payable to you as provided herein.
- **8. Entire Agreement**. This Settlement Letter sets forth the entire agreement of the parties hereto in respect of the subject matter contained herein and supersedes all prior agreements, term sheets, promises, covenants, arrangements, communications, representations or warranties, whether oral or written, by any officer, employee or representative of any party hereto in respect of such subject matter. Any other prior agreement of the parties hereto in respect of the subject matter contained herein, including, without limitation, the Documents, is hereby terminated and cancelled.
- **9. Miscellaneous**. This Settlement Letter may be executed in counterparts. This Settlement Letter and any related dispute shall be governed by, and construed and interpreted in accordance with, the laws of the State of Delaware applicable to contracts executed in and to be performed in that State. Each of the parties hereto (A) consents to submit itself to the personal jurisdiction of any state or federal court located in the State of Delaware in the event any dispute arises out of this Settlement Letter or any of the transactions contemplated by this Settlement Letter; (B) agrees that it will not attempt to deny or defeat such personal jurisdiction or venue by motion or other request for leave from any such court; and (C) agrees that it will not bring any action relating to this Settlement Letter or any of the transactions contemplated by this Settlement Letter in any court other than such courts sitting in the State of Delaware. EACH OF THE PARTIES HERETO HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS SETTLEMENT LETTER OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY.

[Signature Page Follows]

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 225 of 441

Please indicate your acknowledgement of, agreement with and consent to the foregoing by countersigning the enclosed copy of this Settlement Letter in the space provided below and

returning the same to the Company.

Sincerely,

HCR MANORCARE, INC.

QUALITY CARE PROPERTIES, INC.

I hereby acknowledge and agree and consent to the foregoing as of the date first written above.

[EXECUTIVE]

Exhibit C

Original Master Lease Amendment

AMENDMENT TO MASTER LEASE AND SECURITY AGREEMENT

This AMENDMENT TO MASTER LEASE AND SECURITY AGREEMENT, dated as of [•], 2018 (this "<u>Amendment</u>"), is entered into by and among HCR ManorCare, Inc. (the "<u>Debtor</u>"), HCR III Healthcare, LLC ("<u>HCR III</u>," and together with the Debtor, the "<u>Company</u>"), Quality Care Properties, Inc. ("<u>QCP Inc.</u>"), and the affiliates of QCP Inc. party hereto (the "<u>Lessor Entities</u>," and together with QCP Inc., "<u>QCP</u>").

RECITALS

WHEREAS, certain real property comprising skilled nursing and assisted living facilities (the "<u>Leased Facilities</u>"), is leased to HCR III by the Lessor Entities pursuant to the Master Lease and Security Agreement, dated as of April 7, 2011 (as modified and amended from time to time, including by Section 1 of this Amendment, the "<u>MLSA</u>"), by and between the Lessor Entities, as Lessor (as defined in the MLSA), and HCR III, as Lessee (as defined in the MLSA);

WHEREAS, on March 4, 2018, the Debtor commenced proceedings under chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>") in the United States Bankruptcy Court for the District of Delaware (the "<u>Bankruptcy Court</u>");

WHEREAS, on [•], 2018, the Bankruptcy Court confirmed a prepackaged chapter 11 plan of reorganization for the Debtor (the "Prepackaged Plan");

WHEREAS the Prepackaged Plan contemplates that the MLSA shall be amended as set forth herein as part of a series of transactions on the Effective Date (as defined in the Prepackaged Plan); and

WHEREAS, QCP and the Company (collectively, the "Parties") desire to amend the MLSA as set forth herein:

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing and of the covenants and agreements hereinafter set forth, and intending to be legally bound hereby, the Parties agree as follows:

1. Amendment of MLSA.

(a) Section 2.1 of the MLSA is hereby amended and modified to insert the following new definitions in appropriate alphabetical order:

"Deferral Period": [•], 2018 until [•], 2025.1

"Monthly Free Cash Flow": For any calendar month, such cash and cash equivalents of Lessee available on the first Business Day of such calendar month to pay part or all of the Minimum Rent due for such

Note to draft: Insert effective date and seventh anniversary of Effective Date, respectively.

calendar month after Lessee retains amounts reasonably required, taking into account projected receipts, to satisfy Lessee's monthly cash needs (including all Additional Charges, other amounts required to be paid or expended by Lessee under this Lease) and liabilities reasonably anticipated to be paid in cash within 90 days of the first Business Day of such calendar month.

"New Deferred Rent Obligation": All Minimum Rent deferred in accordance with Section 3.1.3(b), together with interest accruing at the rate of four percent (4.0%) per annum, compounded monthly.

- (i) Section 3.1 of the MLSA is hereby amended and modified to insert the following at the end of such section:
 - 3.1.3 <u>Partial Cash Payment of Minimum Rent</u>. Notwithstanding Section 3.1.1, during the Deferral Period, Lessee shall pay to Lessor Minimum Rent monthly, in the amounts set forth in or determined pursuant to the applicable formula(s) set forth in Exhibit D attached hereto, as follows:
 - (a) Lessee shall pay in cash on the first Business Day of such calendar month, as partial payment of Minimum Rent for such month, all Monthly Free Cash Flow for such month (not to exceed the total Minimum Rent payable in respect of such calendar month).
 - (b) The remaining Minimum Rent (if any) for each calendar month during the Deferral Period shall be added to the amount of the New Deferred Rent Obligation (it being understood that failure of Lessee to pay such remaining Minimum Rent in cash in respect of any calendar month during the Deferral Period shall not constitute a breach or default under this Lease, subject to the acceleration of the New Deferred Rent Obligation pursuant to Section 3.1.4).
 - 3.1.4 Repayment of New Deferred Rent Obligation. The entire New Deferred Rent Obligation shall be due and payable by Lessee to Lessor on the earliest to occur of: (a) [•], 2025; (b) (i) Lessee shall have been adjudicated as bankrupt or a court of competent jurisdiction having entered an order or decree appointing a receiver of Lessee or of the whole or substantially all of its property, rents or profits, or approving a petition filed against it seeking reorganization or arrangement of Lessee under the Federal bankruptcy laws, in each case, with the Lessee and its Affiliates having used reasonable best efforts to contest such order, decree or petition, and (ii) such judgment, order or decree shall have not been vacated or set aside or stayed within sixty (60) days from the date thereof; or (c) (i) either (I) a single judgment of a court of competent jurisdiction shall have been entered against Lessee or any of its Subsidiaries providing for the payment of money damages to an unaffiliated third party in excess

of the Repayment Trigger Amount (net of amounts payable by insurers of Lessee and its Subsidiaries) or (II) more than one judgments of courts of competent jurisdiction shall have been entered against Lessee and/or its Subsidiaries in any calendar month providing for the aggregate payment of money damages to an unaffiliated third party in excess of the Repayment Trigger Amount (net of amounts payable by insurers of Lessee and its Subsidiaries) and (ii) Lessee shall have (I) filed a petition in bankruptcy or a petition to take advantage of any insolvency act, (II) made an assignment for the benefit of its creditors, (III) consented to the appointment of a receiver of itself or of the whole or any substantial part of its property, rents or profits, or (IV) filed a petition or answer seeking reorganization or arrangement under the Federal bankruptcy laws or any other applicable law or statute of the United States of America or any state thereof. "Repayment Trigger Amount" means \$25,000,000.

- (ii) Section 3.1 of the MLSA is hereby amended and modified to delete Section 3.1.2 in its entirety and to replace it with "[Reserved]".
- (iii) Article 3 of the MLSA is hereby amended and modified to delete Section 3.5 in its entirety.
- (iv) Section 16.1 of the MLSA is hereby amended and modified to insert the following at the end of such section:
 - (q) Lessee shall fail to pay in cash any installment of Monthly Free Cash Flow when the same becomes due and payable and such failure is not cured by Lessee within a period of five (5) Business Days after notice thereof from Lessor; provided, however, that such notice shall be in lieu of and not in addition to any notice required under applicable law;
- (v) Section 45.1.8 is hereby amended and modified to delete the first paragraph of the section in its entirety and replace it with the following:

Termination Date. In addition to other rights of Lessor to terminate this Lease under any other provision hereof, at any time and from time to time, Lessor shall have the right to terminate this Lease with respect to any one or more (including all, if so elected by Lessor) of the seventy-five (75) Facilities identified on Schedule 45.1.8-1 after prior written notice to Lessee no less than fourteen (14) days prior to the applicable termination date. If this Lease is terminated by Lessor under this Section 45.1.8 or by Lessor or Lessee under any other provision hereof, in each case, with respect to any one or more (including all, if applicable) of the Facilities, or upon the expiration of the Term applicable to a Facility (collectively, the "termination date"), the following shall pertain:

- (vi) All references in the MLSA to the "Master Lease" shall be deemed to be references to the MLSA as amended hereby.
- (b) Except as amended as set forth herein, the MLSA shall remain in full force and effect.
- QCP agrees and acknowledges that (i) upon the Effective Date, all (c) amounts due and payable under the MLSA immediately prior to the Effective Date have been satisfied and discharged by the Debtor, as guarantor of the Lessee's obligations under the MLSA pursuant to that certain Guaranty of Obligations, effective as of February 11, 2013 (as amended or modified from time to time, the "Guaranty"), for good and valuable consideration in connection with the Prepackaged Plan and (ii) Lessee shall have no liability for such amounts. QCP hereby agrees that (i) any breach of or default under the MLSA or the Guaranty occurring on or prior to the Effective Date is hereby irrevocably waived and deemed cured, (ii) Lessor shall have no right to exercise any remedies under the MLSA with respect to such breaches or defaults (including any rights described under Article XVI of the MLSA and any right to indemnification for such breaches or defaults under Section 23.1 of the MLSA) and (iii) no action, inaction, failure, occurrence or event occurring on or prior to, and no state of facts existing on, the Effective Date (other than the terms of the MLSA and the execution and delivery of the MLSA and all amendments thereto) that has resulted in a breach of or default under the MLSA occurring on or prior to the Effective Date shall, in whole or in part, be considered in determining whether any breach or default under the MLSA shall have occurred after the Effective Date.
- (d) The Debtor hereby consents to the amendment of the MLSA as set forth herein.
- 2. <u>Binding Effect; Assignment</u>. This Amendment and all of the provisions hereof shall be binding upon and inure to the benefit of the Parties hereto and their respective heirs, executors, successors and permitted assigns, but except as contemplated herein, neither this Amendment nor any of the rights, interests or obligations hereunder shall be assigned, directly or indirectly, by any Party without the prior written consent of the other Parties hereto.
- 3. <u>No Third-Party Beneficiaries</u>. This Amendment is not intended and shall not be deemed to confer upon or give any person, except the Parties hereto, their respective successors and permitted assigns, any remedy, claim, liability, reimbursement, cause of action or other right under or by reason of this Amendment.
- 4. <u>GOVERNING LAW</u>. THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE INTERNAL LAWS OF THE STATE OF CALIFORNIA (WITHOUT REGARD OF PRINCIPLES OR CONFLICTS OF LAW) AND ANY APPLICABLE LAWS OF THE UNITED STATES OF AMERICA, EXCEPT THAT ALL PROVISIONS HEREOF RELATING TO THE CREATION OF THE LEASEHOLD ESTATE AND ALL REMEDIES SET FORTH IN ARTICLE XVI RELATING TO RECOVERY OF POSSESSION OF THE LEASED PROPERTY OF ANY FACILITY (SUCH AS AN ACTION FOR UNLAWFUL DETAINER OR OTHER SIMILAR ACTION) SHALL BE CONSTRUED AND ENFORCED ACCORDING TO, AND

GOVERNED BY, THE LAWS OF THE STATE IN WHICH THE LEASED PROPERTY OF SUCH FACILITY IS LOCATED.

- 5. Forum; Service of Process. Any proceeding brought by any Party arising out of or based upon this Amendment shall only be instituted in any federal or state court in Los Angeles County, California, and each Party waives any objection which it may now or hereafter have to the laying of venue of any such proceeding, and irrevocably submits to the jurisdiction of such courts in any such Proceeding. Each Party further agrees that service of any process, summons, notice or document by U.S. registered mail to such Party's respective last known address shall be effective service of process for any action, suit or proceeding in California with respect to any matters to which it has submitted to jurisdiction in this Section 5.
- WAIVER OF JURY TRIAL. TO THE FULLEST EXTENT PERMITTED BY 6. LAW, THE PARTIES HERETO HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF THIS AMENDMENT OR ANY DEALINGS BETWEEN THEM RELATING TO THE SUBJECT MATTER OF THIS TRANSACTION. THE SCOPE OF THIS WAIVER IS INTENDED TO BE ALL-ENCOMPASSING OF ANY AND ALL DISPUTES THAT MAY BE FILED IN ANY COURT AND THAT RELATE TO THE SUBJECT MATTER OF THIS AMENDMENT, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS, AND ALL OTHER COMMON LAW AND STATUTORY CLAIMS. PARTIES HERETO ACKNOWLEDGE THAT THIS WAIVER IS A MATERIAL INDUCEMENT TO ENTER INTO A BUSINESS RELATIONSHIP, THAT EACH HAS ALREADY RELIED ON THE WAIVER IN ENTERING INTO THIS AMENDMENT AND THAT EACH WILL CONTINUE TO RELY ON THE WAIVER IN THEIR RELATED FUTURE DEALINGS. THE PARTIES HERETO FURTHER WARRANT AND REPRESENT THAT EACH HAS REVIEWED THIS WAIVER WITH ITS OR HIS, AS THE CASE MAY BE, LEGAL COUNSEL, AND THAT EACH KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH LEGAL COUNSEL. THIS WAIVER IS IRREVOCABLE, MEANING THAT IT MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING, AND THE WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS TO THIS AMENDMENT OR TO ANY OTHER DOCUMENTS OR AGREEMENTS RELATING TO THE TRANSACTIONS CONTEMPLATED HEREBY. IN THE EVENT OF LITIGATION, THIS AMENDMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.
- 7. <u>Entire Agreement</u>. The MLSA, as amended by this Amendment, contains the entire agreement between the Parties and all understandings and agreements previously made between the Parties (and not referred to above) are merged herein with respect to the subject matter hereof, which, together with any contemporaneously executed agreements, alone fully and completely expresses their agreement.
- 8. <u>No Presumption</u>. This Amendment has been freely negotiated by the Parties and in any controversy, dispute, or contest over the meaning, interpretation, validity, or enforceability of this Amendment or any of its terms or conditions, there shall be no inference,

presumption, or conclusion drawn whatsoever against either Party by virtue of that Party having drafted this Amendment or any portion thereof.

- 9. <u>Captions; Counterparts</u>. The captions in this Amendment are for convenience only and shall not be considered a part of or affect the construction or interpretation of any provision of this Amendment. This Amendment may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Any facsimile or .pdf copies hereof or signature hereon shall, for all purposes, be deemed originals.
- 10. <u>Changes in Writing</u>. This Amendment may not be orally changed or terminated, nor any of its provisions waived, except by an agreement in writing signed by the Party against whom enforcement of any changes, termination or waiver is sought.
- 11. <u>Authority</u>. Each Party hereby represents and warrants that it has full right, power and authority to enter into this Amendment and that the person executing this Amendment on its behalf is duly authorized to do so.

[Remainder of page intentionally left blank]

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 233 of 441

IN WITNESS WHEREOF, the Parties have executed this Amendment as of the day and year first above written.

QUALITY CARE PROPERTIES, INC.
Ву:
Name:
Title:

HCP PROPERTIES, LP, a Delaware limited partnership

By: HCP I-B Properties, LLC, a Delaware limited liability company, its General Partner

HCP WEST VIRGINIA PROPERTIES, LLC, a Delaware limited liability company

HCP PROPERTIES OF ALEXANDRIA VA, LLC, a Delaware limited liability company

HCP PROPERTIES OF ARLINGTON VA, LLC, a Delaware limited liability company

HCP PROPERTIES OF MIDWEST CITY OK, LLC, a Delaware limited liability company

HCP PROPERTIES OF OKLAHOMA CITY (NORTHWEST), LLC, a Delaware limited liability company

HCP PROPERTIES OF OKLAHOMA CITY (SOUTHWEST), LLC, a Delaware limited liability company

HCP PROPERTIES OF TULSA OK, LLC, a Delaware limited liability company

HCP PROPERTIES-ARDEN COURTS OF ANNANDALE VA, LLC, a Delaware limited liability company

HCP PROPERTIES-CHARLESTON OF HANAHAN SC, LLC, a Delaware limited liability company

HCP PROPERTIES-COLUMBIA SC, LLC, a Delaware limited liability company

HCP PROPERTIES-FAIR OAKS OF FAIRFAX VA, LLC, a Delaware limited liability company

HCP PROPERTIES-IMPERIAL OF RICHMOND VA, LLC, a Delaware limited liability company

HCP PROPERTIES-LEXINGTON SC, LLC, a Delaware limited liability company

HCP PROPERTIES-MEDICAL CARE CENTER-LYNCHBURG VA, LLC, a Delaware limited liability company

HCP PROPERTIES-OAKMONT EAST-GREENVILLE SC, LLC, a Delaware limited liability company

HCP PROPERTIES-OAKMONT OF UNION SC, LLC, a Delaware limited liability company

HCP PROPERTIES-OAKMONT WEST-GREENVILLE SC, LLC, a Delaware limited liability company

HCP PROPERTIES-STRATFORD HALL OF RICHMOND VA, LLC, a Delaware limited liability company

HCP PROPERTIES-WEST ASHLEY-CHARLESTON SC, LLC, a Delaware limited liability company

HCP MARYLAND PROPERTIES, LLC, a Delaware limited liability company

HCP PROPERTIES-SALMON CREEK WA, LLC, a Delaware limited liability company

HCP PROPERTIES-WINGFIELD HILLS NV, LLC, a Delaware limited liability company

HCP PROPERTIES-UTICA RIDGE IA, LLC, a Delaware limited liability company

HCP TWINSBURG OH PROPERTY, LLC, a Delaware limited liability company

Ву:	
Name:	
Title:	

HCP STERLING HEIGHTS MI PROPERTY, LLC, a Delaware limited liability company

By: HCR Schoenherr-Road Property, LLC, a Delaware limited liability company, its sole member

By: HCP Properties, LP, a Delaware limited partnership, its sole member

By: HCP I-B Properties, LLC, a Delaware limited liability company, its General Partner

By:	
Name:	
Title:	

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 237 of 441

HCR MANORCARE, INC.
By:
Name:
Title:
HCR III HEALTHCARE, LLC
By:
Name:
Title:

Exhibit D

Original Plan Sponsor Agreement

PLAN SPONSOR AGREEMENT

among

HCR MANORCARE, INC.,

QUALITY CARE PROPERTIES, INC.,

HCP MEZZANINE LENDER, LP

and

THE LESSORS IDENTIFIED HEREIN

Dated as of March 2, 2018

TABLE OF CONTENTS

	TABLE OF CONTENTS	Page
	ARTICLE I	
	The Transactions	
1 1	Dela Errelana	2
1.1 1.2	Debt Exchange Master Lease Amendment and Guaranty	
1.2	Articles of Incorporation and By-Laws of the Reorganized	4
1.5	Debtor	4
1.4	Closing	
1.5	Severance and Other Payments	
1.6	Deliveries by Debtor	
1.7	Deliveries by Parent and Purchaser	
	ARTICLE II	
	Representations and Warranties	
2.1	Representations and Warranties of the Debtor	7
2.2	Representations and Warranties of the Purchaser Entities	
	ARTICLE III	
	ARTICLE III	
	Bankruptcy Matters	
3.1	Pre-Filing Period	24
3.2	Chapter 11 Obligations	
3.3	Case Financing.	26
3.4	Release	26
	ARTICLE IV	
	Covenants	
1 1	Interior Operations	27
4.1 4.2	Interim Operations Eilings: Other Actions: Notification	
4.2	Filings; Other Actions; Notification Access and Records	
4.4	Publicity	
4.5	Employee Matters	
4.6	Expenses	
4.7	Indemnification; Directors' and Officers' Insurance	
4.8	Confidentiality	
4.9	Financing.	38
4.10	Acquisition Proposals.	41

ARTICLE V

Special Covenants

5.1 5.2 5.3	Corporate Governance Rent and Forbearance Business Operations	46
	ARTICLE VI	
	Conditions	
6.1 6.2 6.3	Conditions to Each Party's Obligations Conditions to Obligations of Purchaser Entities Conditions to Obligation of the Debtor	49
	ARTICLE VII	
	Termination	
7.1 7.2 7.3 7.4 7.5	Automatic Termination Termination by Mutual Consent Termination by Either Parent or the Debtor Termination by the Debtor Termination by Parent	52 52 52
	ARTICLE VIII	
	Miscellaneous and General	
8.1	No Survival; Effect of Termination and Abandonment	53
8.2	Modification or Amendment	
8.3	Waiver of Conditions	
8.4	<u>Counterparts</u>	54
8.5	GOVERNING LAW AND VENUE; WAIVER OF JURY	
0.6	TRIAL; SPECIFIC PERFORMANCE	
8.6	Notices	56
8.7	Entire Agreement	
8.8 8.9	Obligations of Parent and of the Debtor	
8.10	Transfer Taxes	
8.11	Definitions	
8.12	Severability	
8.13	Interpretation; Construction	
8.14	Assignment	
8.15	Certain Waivers	
_		

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 242 of 441

8.16	No S	Solicitation	60
8.17	Settle	ement Discussions	60
Annex	A	Defined Terms	
Exhibi	t 1	Plan Term Sheet	
Exhibi	t 2	Restructuring Support Agreement	
Exhibi	t 3	Form of Former CEO Settlement Agreement	
Exhibi	t 4	Form of Master Lease Amendment	
Exhibi	t 5	Form of Separation Agreement	

AGREEMENT

This PLAN SPONSOR AGREEMENT, dated as of March 2, 2018 (hereinafter called this "<u>Agreement</u>"), is among HCR ManorCare, Inc., a Delaware corporation (the "<u>Debtor</u>"), Quality Care Properties, Inc., a Maryland corporation ("<u>Parent</u>"), HCP Mezzanine Lender, LP, a Delaware limited partnership and a wholly-owned subsidiary of Parent ("<u>Purchaser</u>"), and the parties signatory hereto as lessors (collectively, the "<u>Lessors</u>" and, together with Parent and Purchaser, the "<u>Purchaser Entities</u>"). The Debtor, Parent, Purchaser and the Lessors are each referred to in this Agreement as a "party" and together, the "parties".

RECITALS

WHEREAS, the Debtor and its Subsidiaries are engaged in the business of providing skilled nursing, hospice and other ancillary services (the "**Business**");

WHEREAS, pursuant to the Master Lease and Security Agreement, dated as of April 7, 2011, by and among the Lessors and HCR III Healthcare, LLC ("<u>HCR III</u>"), as Lessee (as amended from time to time, the "<u>Master Lease</u>"), HCR III leases certain skilled nursing and assisted living facilities (each, a "<u>Leased Facility</u>" and, collectively, the "<u>Leased Facilities</u>") from Lessors;

WHEREAS, the Debtor guarantees the obligations of HCR III under the Master Lease pursuant to that certain Guaranty of Obligations, effective as of February 11, 2013 (as amended or modified from time to time, the "Guaranty");

WHEREAS, HCR III subleases the Leased Facilities to certain of its direct and indirect Subsidiaries and Affiliates (each, a "<u>Sublease OpCo</u>" and, collectively, the "<u>Sublease OpCos</u>") pursuant to certain subleases (the "<u>Subleases</u>"), and the Sublease OpCos operate the Leased Facilities;

WHEREAS, Lessors, HCR III and the Sublease OpCos have entered into certain Agreements Regarding Subleases made and entered as of April 7, 2011, October 23, 2015, February 16, 2016 and March 14, 2016 (each, as amended or modified from time to time, an "Agreement Regarding Subleases" and, collectively, the "Agreements Regarding Subleases");

WHEREAS, Parent, the Debtor and certain of their respective Subsidiaries entered into a Forbearance Agreement, dated as of April 5, 2017 (the "Prior Forbearance Agreement"), pursuant to which, among other things, Parent agreed to defer certain of the Debtor's rent and other obligations under the Master Lease, which such Prior Forbearance Agreement terminated according to its terms on July 5, 2017;

WHEREAS, on August 17, 2017, Parent filed a complaint against the Debtor, HCR III and "John Does 1–50" in Los Angeles County Superior Court, styled as *Quality Care Properties, Inc. and HCP Properties, LP* v. *HCR III Healthcare, LLC et al.*, Los Angeles County Super. Ct. No. BC672837, seeking, among other relief, the appointment of a receiver for HCR III (the "Receivership Proceeding");

WHEREAS, Parent, the Debtor and certain of their respective Subsidiaries entered into that certain Forbearance Agreement and Amendment to Master Lease and Security Agreement, dated as of December 22, 2017 (the "Second Forbearance Agreement"), pursuant to which, among other things, the Lessors and HCR III agreed to amend the Master Lease to defer certain of the Debtor's rent and other obligations under the Master Lease and to defer the Debtor's response on the Receivership Proceeding;

WHEREAS, on or prior to the date of this Agreement, HCR III has paid to the Lessors the Reduced Cash Rent of \$23,500,000 per month, as more fully described in the Master Lease (the "Reduced Cash Rent"), due and payable pursuant to the terms of the Master Lease since the date of the Second Forbearance Agreement until the date of this Agreement, other than the Reduced Cash Rent Past Due Balance;

WHEREAS, to implement the Transactions, the Debtor has agreed, subject to the terms and conditions of this Agreement, to file a voluntary petition for relief under chapter 11 of the Bankruptcy Code in the United States Bankruptcy Court for the District of Delaware (the "<u>Bankruptcy Court</u>" and such proceedings, collectively, the "<u>Bankruptcy Case</u>") on or prior to March 4, 2018;

WHEREAS, the Debtor, with assistance from professional advisors, has reviewed alternatives to the transfer of the Business to Purchaser and has determined that this Agreement and the Transactions are in its best interests;

WHEREAS, in the Bankruptcy Case, the Debtor proposes to treat the claims of creditors of the Debtor in accordance with a plan of reorganization (as it may be amended from time to time consistent with this Agreement, the "Plan") under chapter 11 of title 11 of the United States Code (the "Bankruptcy Code") in form and substance consistent with the term sheet attached as Exhibit 1 hereto (the "Plan Term Sheet");

WHEREAS, in connection with the Plan, the Debtor has entered into that certain restructuring support agreement, dated on or prior to the date hereof (the "**Restructuring Support Agreement**"), with each of (i) Carlyle MC Partners, L.P., a Delaware limited partnership, Carlyle Partners V-A MC, L.P., a Delaware limited partnership, Carlyle Partners V MC, L.P., a Delaware limited partnership, and CP V Coinvestment B, L.P., a Delaware limited partnership (collectively, the "**Carlyle Holders**"), (ii) Parent, and (iii) MC Operations Investments, LLC, the form of which is attached hereto as Exhibit 2;

WHEREAS, the Debtor intends to, with the consent of Parent, promptly offer to Paul Ormond (the "<u>Former CEO</u>") a settlement agreement in the form attached hereto as <u>Exhibit 3</u> (the "<u>Former CEO Settlement Agreement</u>"), by and among the Debtor and Paul Ormond, to be effective upon the Closing Date subject to the execution and delivery of the Former CEO Settlement Agreement by Mr. Ormond within 60 days of the date of this Agreement;

WHEREAS, the Debtor entered into a Credit Agreement, dated as of July 17, 2017 (as may be amended, modified or supplemented from time to time, the "Centerbridge"

<u>Facility</u>"), by and among HCR Home Health Care and Hospice, LLC, as borrower, HCR Manorcare Inc., HCR Manorcare Operations II, LLC, HCR Manorcare Heartland, LLC, Manor Care, Inc., HCR II Healthcare, LLC, and HCR Healthcare, LLC, each as Holdco, the lender parties thereto, and RD Credit, LLC, as administrative agent and collateral agent; and

WHEREAS, the Debtor and the Purchaser Entities desire to make certain representations, warranties, covenants and agreements in connection with this Agreement.

NOW, THEREFORE, in consideration of the premises, and of the representations, warranties, covenants and agreements contained herein, the parties hereto agree as follows:

ARTICLE I

The Transactions

1.1 <u>Debt Exchange</u>.

- (a) On the Closing Date, pursuant to the Plan, in exchange for the discharge under the Plan of the QCP Claims (as defined in the Plan Term Sheet) (the "<u>Debt Exchange</u>"), Purchaser shall acquire and accept from the Debtor, and the Debtor shall issue, transfer and deliver to Purchaser 1,000,000 newly issued shares of the common stock of the Debtor, par value \$0.01 per share (the "<u>Reorganized Company Stock</u>"), which, pursuant to and in accordance with the Plan and Confirmation Order, shall constitute all of the issued and outstanding capital stock and rights to purchase or otherwise acquire capital stock of the Debtor (such transaction, the "<u>Plan Acquisition</u>"), free and clear of any Liens, other than those created by Parent, Purchaser or the Lessors or imposed by applicable federal or state securities laws.
- For purposes of this Agreement, "Lien" means any mortgage, lien, pledge, charge, security interest, easement, covenant, or other restriction or title matter or encumbrance of any kind in respect of such asset; and "Permitted Lien" means the following Liens: (i) specified encumbrances described in Section 1.1(a) of the Debtor Disclosure Letter; (ii) encumbrances for Taxes or other governmental charges not yet due and payable or that are being contested in good faith by appropriate proceedings and are reflected on or specifically reserved against or otherwise disclosed in any consolidated balance sheet included in the Audited Financial Statements; (iii) mechanics', carriers', workmen's, repairmen's or other like encumbrances arising or incurred in the ordinary course of business consistent with past practice relating to obligations as to which there is no default on the part of the Debtor, or the validity or amount of which is being contested in good faith by appropriate proceedings and are reflected on or specifically reserved against or otherwise disclosed in any consolidated balance sheet included in the Audited Financial Statements; (iv) other encumbrances that do not, individually or in the aggregate, materially impair the current use or operation of the specific parcel of owned real property to which they relate or the conduct of the business of the Debtor and its Subsidiaries as presently conducted; (v) easements, rights of way or other similar matters or restrictions or exclusions which would be shown by a current title report or other similar report; (vi) any condition or other matter, if any, that may be shown or disclosed by a current and accurate survey or physical inspection; and (vii) the renewal or reaffirmation of any Lien existing on the

date hereof granted in favor of RD Credit, LLC and/or the lender parties to the Centerbridge Facility (or any agent on their behalf) to secure obligations thereunder.

1.2 Master Lease Amendment and Guaranty.

- (a) On the Closing Date, (i) the Lessors and HCR III shall enter into an amendment to the Master Lease substantially in the form attached hereto as Exhibit 4 (the "Master Lease Amendment"); (ii) HCR III and the applicable Sublease OpCos shall enter into amendments to each Sublease substantially similar to the Master Lease Amendment, *mutatis mutandis* (the "Sublease Amendments"); provided that the Repayment Trigger Amount in each Sublease Amendment shall equal the Repayment Trigger Amount in the Master Lease Amendment *multiplied by* the ratio of (x) the Allocated Minimum Rent (as defined in the Master Lease) attributable to the applicable Facility to (y) Minimum Rent (as defined in the Master Lease).
- (b) On or prior to the seventh day after the Closing Date, the Lessors and HCR III shall terminate, release and discharge, and Parent shall cause the Lessors and HCR III to terminate, release and discharge, the Guaranty.
 - 1.3 Articles of Incorporation and By-Laws of the Reorganized Debtor.
- (a) <u>Articles of Incorporation</u>. Pursuant to the Plan, on the Closing Date, the articles of incorporation of the Debtor (the "<u>Charter</u>") shall be amended in their entirety to be in a form set forth in a supplement to the Plan until, subject to Section 4.7, thereafter amended as provided therein or by applicable Law.
- (b) <u>By-Laws</u>. Pursuant to the Plan, on the Closing Date, the by-laws of the Debtor (the "<u>By-Laws</u>") shall be amended in their entirety to be in a form set forth in a supplement to the Plan until, subject to Section 4.7(a), thereafter amended as provided therein or by applicable Law.
- 1.4 Closing. Unless otherwise mutually agreed in writing between the Debtor and Parent, the closing of the Debt Exchange, the Plan Acquisition, the entry into the Master Lease Amendment and the Sublease Amendments and the other transactions contemplated by this Agreement (the "Closing" and such transactions, collectively, the "Transactions") shall take place at the offices of Sullivan & Cromwell LLP, 125 Broad Street, New York, New York, at 9:00 a.m. on the second (2nd) business day following the day on which the last to be satisfied or waived of the conditions set forth in ARTICLE VI (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the fulfillment or waiver of those conditions) shall be satisfied or waived in accordance with this Agreement; provided, however, that Parent may elect, in its reasonable discretion, by written notice to the Debtor, to defer the Closing to the last business day of the calendar month during which all of the conditions set forth in ARTICLE VI (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the fulfillment or waiver of those conditions) shall be satisfied or waived in accordance with this Agreement (the date of the Closing, the "Closing Date"); provided that, if Parent elects to defer the Closing, then Parent will be deemed to have waived all conditions set forth in ARTICLE VI following such election (other than those conditions that by their nature

are to be satisfied at the Closing). For purposes of this Agreement, the term "<u>business day</u>" means any day ending at 11:59 p.m. (Eastern Time) other than a Saturday or Sunday or a day on which banks are required or authorized to close in the City of New York.

1.5 Severance and Other Payments.

Subject to Section 1.5(b), on the Closing Date, following the Closing, (a) (i) Parent shall cause the Severance Account to be funded with all amounts necessary to make severance payments to Eligible Employees (each as defined in the Plan Term Sheet) in the amounts set forth in Section 1.5(i) of the Debtor Disclosure Letter (the "Excess Severance Payments"), (ii) Parent shall cause the Excess Severance Payments due each Eligible Employee to be paid from the Severance Account to such Eligible Employee on the Payment Date with respect to such Eligible Employee (it being understood that, for the purpose of determining entitlement to payment of Excess Severance Payments, the employment of each Eligible Employee who remains employed by the Debtor immediately prior to the Closing shall be deemed to have been terminated without Cause constituting a Qualifying Termination (each as defined in such Eligible Employee's written Employment Agreement with the Debtor (each, an "Employment Agreement")) as of the Closing), (iii) except as provided in clause (iv) of this Section 1.5(a), Parent shall pay, or cause to be paid, the other obligations set forth in Section 1.5 of the Debtor Disclosure Letter, in each case, on the applicable Payment Date; provided, that payments made in respect of any nonqualified deferred compensation plan shall be made in accordance with the relevant Eligible Employee's deferral election; provided, further, that payments to be made under the Key Employee Incentive Plan shall be all those due and payable as a result of the Restructuring Completion (as defined in the Key Employee Incentive Plan), (iv) the Debtor shall irrevocably terminate the HCR ManorCare, Inc. Senior Executive Retirement Plan, as amended, as of the Closing Date and Parent shall pay or cause to be paid all amounts due thereunder, which amounts the Parties agree are set forth in Section 1.5 of the Debtor Disclosure Letter (and which amounts, for the avoidance of doubt, reflect the actuarial reduction provided for under Section 7.05(b) of the HCR ManorCare, Inc. Senior Executive Retirement Plan) on the applicable Payment Date, (v) QCP will execute and deliver the applicable Separation Agreement to each Eligible Employee (if not previously executed and delivered) and (vi) QCP will execute and deliver the Former CEO Settlement Agreement to the Former CEO (if not previously executed and delivered), subject to the Former CEO's execution and delivery of the Former CEO Settlement Agreement to QCP. The "Payment Date" means (I) with respect to each Eligible Employee, the later of (x) the Closing Date and (y) the eighth (8th) day after such Eligible Employee shall have executed (and not revoked) such Eligible Employee's Separation Agreement, (II) with respect to amounts payable to the Former CEO (other than the severance obligation to the Former CEO set forth on Section 1.5 of the Debtor Disclosure Letter), the Closing Date and (III) with respect to the severance obligation to the Former CEO set forth on Section 1.5 of the Debtor Disclosure Letter, the later of (x) the Closing Date and (y) the eighth (8th) day after the Former CEO shall have executed and delivered to the Debtor (and not revoked) the Former CEO Settlement Agreement. Parent shall also fund any employer Taxes related to the amounts payable pursuant to this Section 1.5(a) and timely pay such amounts, plus any applicable withholding Taxes related to such payments, to the applicable Governmental Entity (provided, for the avoidance of doubt, that all amounts payable subject to this Section 1.5 shall be subject to withholding of any Taxes that are required to be withheld pursuant to applicable Laws).

- (b) Notwithstanding anything to the contrary in this Agreement, the Plan Term Sheet or the Plan, Parent shall have no obligation to pay, or cause to be paid, any Excess Severance Payments to any Eligible Employee:
- (i) if such Eligible Employee separates from service without the written consent of the Parent prior to the Closing Date; it being understood and agreed that if such Eligible Employee separates from service without the written consent of the Parent prior to the Closing Date due to the Eligible Employee's Disability (as defined in such Eligible Employee's written Employment Agreement with the Debtor (the "Employment Agreement")) or death, such separation from service shall be deemed and shall constitute a Qualifying Termination within the meaning of such Eligible Employee's Employment Agreement for all purposes, and the Eligible Employee shall remain entitled to, and shall receive, the Excess Severance Payments (irrespective of whether such separation from service otherwise would have constituted a Qualifying Termination under the Employment Agreement without consideration of this Agreement) upon the Payment Date, subject to Sections 1.5(b)(ii), 1.5(b)(iii) and 1.5(b)(iv) of this Agreement;
- (ii) if such Eligible Employee does not execute and deliver a separation agreement and release in a form attached as <u>Exhibit 5</u> hereto (each, a "<u>Separation</u> **Agreement**");
- (iii) to the extent that the Debtor has not obtained the approvals and waivers contemplated by Section 4.5(e)(i) and Section 4.5(e)(ii) with respect to such Eligible Employee (including, for the avoidance of doubt, waivers in respect of any tax gross-ups to which such Eligible Employee would otherwise be entitled); or
- (iv) if (A) there has been a material breach of any covenant or agreement made by the Debtor in this Agreement and such material breach is not curable or, if curable, is not cured by the earlier of (i) Closing and (ii) thirty (30) days after written notice thereof is given by Parent to the Debtor and (B) prior to Closing, the Debtor Board determines that such Eligible Employee is responsible for or had the ability to avoid such breach and knowingly failed to do so (it being understood that, if Parent asserts in a written notice to the Debtor Board that it believes any Eligible Employee is responsible for any such material breach or had the ability to avoid such material breach and knowingly failed to do so, the Debtor Board shall consider such assertion by Parent and make a good faith determination as to whether such Eligible Employee is responsible for any such material breach or had the ability to avoid such material breach and knowingly failed to do so); provided that all determinations by the Debtor Board relating to this Section 1.5(b)(iv) shall be made by the non-employee directors thereof.
- 1.6 <u>Deliveries by Debtor</u>. On the Closing Date, the Debtor shall deliver to the Purchaser Entities:
- (a) a certificate executed by the Debtor that it is not a foreign person within the meaning of Section 1445(f)(3) of the Internal Revenue Code of 1986, as amended, and the Treasury Regulations promulgated thereunder (the "Code");

- (ii) evidence of the resignations or removal of those directors and officers of the Debtor and its Subsidiaries, as applicable, effective as of immediately prior to the Closing, identified in writing to the Debtor by Parent not less than five (5) business days prior to the Closing Date, (iii) evidence of the appointment and approval by the board of directors of the Debtor (the "**Debtor Board**") of those directors nominated by Parent not less than five (5) business days prior to the Closing Date, to the Debtor Board effective as of the Closing, and (iv) evidence of amendments to the constitutive documents of the Debtor and its Subsidiaries effective as of the Closing that may be necessary to implement acquisition of the Reorganized Company Stock as contemplated by the Plan, including in accordance with Section 1.3 of this Agreement, and which shall have been identified by Parent in writing to the Debtor not less than five (5) business days prior to the Closing Date; and
- (c) such other documents, instruments and certificates as the Purchaser Entities may reasonably request and as are customary, including a certified copy of the Confirmation Order and notice of the Closing Date.
- (d) As used in this Agreement, the term (i) "<u>Subsidiary</u>" means, with respect to any Person, any other Person of which at least a majority of the securities or ownership interests having by their terms ordinary voting power to elect a majority of the board of directors or other persons performing similar functions is directly or indirectly owned or controlled by such Person and/or by one or more of its Subsidiaries; (ii) "<u>Person</u>" means any individual, corporation, limited liability company, partnership, limited partnership, firm, joint venture, association, joint-stock company, trust, unincorporated organization or any other entity; and (iii) "<u>Governmental Entity</u>" means any domestic or foreign governmental or regulatory authority, agency, commission, body, court or other legislative, executive or judicial governmental entity.
- 1.7 <u>Deliveries by Parent and Purchaser</u>. At the Closing, Parent and Purchaser shall deliver to the Debtor any documents, instruments or certificates as the Debtor may reasonably request and as are customary.

ARTICLE II

Representations and Warranties

- 2.1 <u>Representations and Warranties of the Debtor</u>. Except as set forth in the corresponding sections or subsections of the disclosure letter delivered to Parent by the Debtor simultaneously with the execution of this Agreement (the "<u>Debtor Disclosure Letter</u>") (it being agreed that disclosure of any item in any section or subsection of the Debtor Disclosure Letter shall be deemed disclosed with respect to any other section or subsection to which the relevance of such item is reasonably apparent), the Debtor hereby represents and warrants to the Purchaser Entities that:
- (a) <u>Organization, Good Standing and Qualification</u>. The Debtor is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power and authority to own,

lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing (with respect to jurisdictions that recognize the concept of good standing) as a foreign corporation or other legal entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or the conduct of its business requires such qualification, except where the failure to be so qualified or in good standing, or to have such power or authority, would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The Debtor has made available to Parent complete and correct copies of the Debtor's and its material Subsidiaries' certificates of incorporation and by-laws or comparable governing documents, each as amended to the date of this Agreement, and each as so delivered is in full force and effect.

As used in this Agreement, the term "<u>Material Adverse Effect</u>" with respect to the Debtor means a change, event or occurrence that has a material adverse effect on the financial condition, properties, assets, liabilities, business or results of operations of the Debtor and its Subsidiaries taken as a whole; <u>provided</u>, <u>however</u>, that none of the following, and no changes, events, occurrences or effects, to the extent arising out of or resulting from any of the following, shall constitute or be taken into account in determining whether a Material Adverse Effect has occurred or may, would or could occur with respect to the Debtor:

- (i) changes in or generally affecting the economy, credit, capital or financial markets or political conditions generally in the United States or elsewhere in the world, including changes in interest and exchange rates, or changes that are the result of acts of war (whether or not declared), armed hostilities, sabotage or terrorism, any escalation or worsening of any such acts of war (whether or not declared), epidemics, pandemics, earthquakes, hurricanes, tornados or other natural disasters;
- (ii) changes that are the result of factors generally affecting the industries in which the Debtor and its Subsidiaries operate or in which the services of the Debtor and its Subsidiaries are used and distributed;
- (iii) any loss of, or adverse change or effect in, the relationship of the Debtor with its customers, employees, financing sources or suppliers, caused by entry into or the pendency or the announcement of the Transactions;
- (iv) changes from the entry into, announcement or performance of this Agreement (including any litigation arising from allegations of any breach of fiduciary duty or violation of Law relating to this Agreement or the Transactions, or compliance by the Debtor with the terms of this Agreement);
- (v) changes or prospective changes in United States generally accepted accounting principles ("<u>GAAP</u>") or in any law, statute, rule or regulation after the date of this Agreement or the interpretation or enforcement thereof;
- (vi) any failure by the Debtor to meet any projections, forecasts or estimates of revenues or earnings for any period, <u>provided</u> that the exception in this clause shall not (on its own) prevent or otherwise affect a determination that any change, event or occurrence underlying such failure has resulted in, or contributed to, a Material Adverse Effect;

- (vii) (A) any action taken by the Debtor or its Subsidiaries at Parent's written request or (B) the failure to take any action by the Debtor or its Subsidiaries if that action is prohibited by this Agreement to the extent that Parent fails to give its consent after receipt of a written request therefor;
- (viii) any change or announcement of a potential change in the credit rating of the Debtor or any of its Subsidiaries or its obligations, <u>provided</u> that the exception in this clause shall not (on its own) prevent or otherwise affect a determination that any change, event or occurrence underlying such failure has resulted in, or contributed to, a Material Adverse Effect; or
- (ix) loss of customers or patients (and loss of revenue associated therewith), change in or tightening of payment or other terms provided by vendors, suppliers or service providers to the Debtor or its Subsidiaries, in each case, occurring since December 31, 2017 through the date of this Agreement;

<u>provided</u>, <u>further</u>, that, with respect to clauses (i), (ii) and (v), such change, event or occurrence does not disproportionately adversely affect the Debtor and its Subsidiaries compared to other companies operating in the principal industries in which the Debtor and its Subsidiaries operate.

(b) <u>Capital Structure</u>.

- (i) The authorized capital stock of the Debtor consists of 55,000,000 shares of common stock of the Debtor, par value \$0.01 per share ("Shares"), and 50,000,000 preferred shares, par value \$0.01 per share ("Preferred Shares"), of which 44,944,057 Shares and 2,000 Preferred Shares, respectively, were outstanding as of the close of business on February 28, 2018. All of the outstanding Shares have been duly authorized and are validly issued, fully paid and nonassessable. Other than 2,031,614 Shares reserved for issuance under the Amended and Restated Equity Incentive Plan of the Debtor (the "Stock Plan"), the Debtor has no Shares reserved for issuance.
- (ii) Section 2.1(b)(ii) of the Debtor Disclosure Letter sets forth (x) each of the Debtor's Subsidiaries and the ownership interest of the Debtor in each such Subsidiary, as well as the ownership interest of any other Person or Persons in each such Subsidiary and (y) the Debtor's or its Subsidiaries' capital stock, equity interest or other direct or indirect ownership interest in any other Person other than securities in a publicly traded company held for investment by the Debtor or any of its Subsidiaries and consisting of less than 1% of the outstanding capital stock of such company.
- (c) <u>Corporate Authority and Approval</u>. The Debtor has all requisite corporate power, authority and legal capacity to execute and deliver this Agreement and each other agreement, document, instrument or certificate contemplated by this Agreement or to be executed by the Debtor in connection with the consummation of the Transactions (the "<u>Debtor Documents</u>") and to perform its obligations hereunder and thereunder and consummate the Transactions. The execution and delivery of this Agreement and the Debtor Documents and the consummation of the Transactions contemplated thereby have been duly authorized by all requisite corporate action on the part of the Debtor. This Agreement has been, and the Debtor

Documents will be at or prior to the Closing, duly and validly executed and delivered by, or on behalf of, the Debtor and (assuming the due authorization, execution and delivery by the other parties hereto and thereto and the entry of the Confirmation Order) this Agreement constitutes, and each of the Debtor Documents when so executed and delivered will constitute, legal, valid and binding obligations of the Debtor and each of its applicable Subsidiaries, enforceable against it in accordance with their respective terms, subject to applicable bankruptcy, insolvency, reorganization, fraudulent transfer, moratorium and similar laws affecting creditors' rights generally and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity) (the "Bankruptcy and Equity Exception").

(d) Governmental Filings; No Violations; Certain Contracts.

- (i) Other than (A) the filings, notices, consents, registrations, approvals, permits or authorizations (including those with respect to state licensing required to operate the Debtor's businesses) set forth on Section 2.1(d)(i) of the Debtor Disclosure Letter (the "Governmental Approvals") and (B) the entry of the Confirmation Order, no notices, reports or other filings are required to be made by the Debtor with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by the Debtor from any Governmental Entity, in connection with the execution, delivery and performance of this Agreement by the Debtor and the consummation of the Transactions, except those failures to make or obtain as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect or prevent, materially delay or materially impair the consummation of the Transactions.
- The execution, delivery and performance of this Agreement by the (ii) Debtor do not, and the consummation of the Transactions will not, constitute or result in (A) a breach or violation of, or a default under, the certificate of incorporation or by-laws of the Debtor or the comparable governing documents of any of its Subsidiaries or (B) with or without notice, lapse of time or both, a breach or violation of, a termination (or right of termination) or default under, the creation or acceleration of any obligations under or the creation of a Lien on any of the assets of the Debtor or any of its Subsidiaries pursuant to any agreement, lease, license, contract, note, mortgage, indenture, arrangement or other obligation, in each case, not otherwise terminable by the other party thereto on sixty (60) days' or less notice (each, a "Contract") to which the Debtor or any of its Subsidiaries is a party or by which the Debtor or any of its Subsidiaries or its or any of their respective properties are bound or (C) assuming compliance with the matters referred to in Section 2.1(d)(i), a violation of any Law to which the Debtor or any of its Subsidiaries is subject, except in the case of clause (B) or (C), for any such breach, violation, termination, default, creation, acceleration or change that is not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect.

(e) Financial Statements.

(i) Set forth in the Debtor Disclosure Letter are the audited consolidated balance sheets of the Debtor and its Subsidiaries as of December 31, 2016 and December 31, 2017, and the related audited consolidated statements of operations, statements of comprehensive (loss) income, statements of cash flows and statements of equity for the twelve

- (12)-month periods then ended (the "<u>Audited Financial Statements</u>"). Each of the balance sheets, statements of operations, statements of comprehensive loss, statements of cash flows and statements of equity included in the Audited Financial Statements (including any related notes thereto) fairly presents, in all material respects, the financial condition, results of operations and cash flows, as the case may be, of the Debtor and its Subsidiaries as of the dates and for the periods set forth therein, as applicable, in each case in accordance with GAAP, consistently applied during the periods involved, except as specifically noted therein.
- The Debtor and its Subsidiaries maintain a system of internal accounting controls designed, in all material respects, to provide reasonable assurances (i) that transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP, (ii) that receipts and expenditures of the Debtor and its Subsidiaries are being made in accordance with appropriate authorizations of management and the Debtor Board and (iii) regarding prevention or timely detection of unauthorized acquisition, use or disposition of assets of the Debtor and its Subsidiaries. To the Knowledge of the Debtor, there is not, and in the past five (5) years there has not been, any fraud that involves or involved the Debtor's and/or its Subsidiaries' management or other employees who have or had a significant role in the Debtor's and/or its Subsidiaries' internal control over financial reporting except as would not, individually or in the aggregate, be material to the Debtor and its Subsidiaries. No significant deficiencies or material weaknesses in the design or operation of the Debtor's system of internal control over financial reporting were identified in the Debtor's most recent assessment that would be reasonably likely to materially affect the Debtor's and/or its Subsidiaries' ability to record, process, summarize and report financial information. As used herein, "Knowledge of **the Debtor**" means the actual (but not constructive or imputed) knowledge of the chief executive officer, chief financial officer, chief operating officer or general counsel of the Debtor (without any obligation of further review and/or inquiry).
- Each of the balance sheets, statements of operations, statements of (iii) comprehensive loss, statements of cash flows and statements of equity (including any related notes thereto) required to be delivered to Parent pursuant to the terms of the Master Lease after the date of this Agreement, including the quarterly unaudited consolidated balance sheets of the Debtor and its Subsidiaries as of March 31, 2018 and any subsequent quarterly unaudited consolidated balance sheets of the Debtor and its Subsidiaries, and the related unaudited consolidated statements of operations, statements of comprehensive (loss) income, statements of cash flows and statements of equity, will fairly present, in all material respects, the financial condition, results of operations and cash flows, as the case may be, of the Debtor and its Subsidiaries as of the dates and for the periods set forth therein, as applicable, in each case in accordance with GAAP, consistently applied during the periods involved, except as specifically noted therein, subject, in the case of any unaudited consolidated balance sheets, statements of operations, statements of comprehensive loss, statements of cash flows and statements of equity, to the absence of footnotes and to year-end adjustments, except as would not, individually or in the aggregate, be material in amount or effect.
- (f) <u>Absence of Certain Changes</u>. Since the date of the most recent Audited Financial Statements through the date hereof, there has not been:

- (i) any material damage, destruction or other casualty loss with respect to any asset or property owned, leased or otherwise used by the Debtor or any of its Subsidiaries, whether or not covered by insurance;
- (ii) any declaration, setting aside or payment of any dividend or other distribution with respect to any shares of capital stock of the Debtor or any of its Subsidiaries (except for dividends or other distributions by any direct or indirect wholly-owned Subsidiary to the Debtor or to any wholly-owned Subsidiary of the Debtor), or any repurchase, redemption or other acquisition by the Debtor or any of its Subsidiaries of any outstanding shares of capital stock or other securities of the Debtor or any of its Subsidiaries;
- (iii) any material change in any method of accounting or accounting practice by the Debtor or any of its Subsidiaries;
- (iv) (A) any material increase in the compensation payable or to become payable to the officers or employees of the Debtor or any of its Subsidiaries or material increases in the benefits of such officers or employees (except for increases in the compensation of non-officer employees in the ordinary course of business and consistent with past practice), (B) any entrance into, adoption, material amendment or termination of any Company Plan (as defined in Section 2.1(h)(i)) (or any arrangement that would have been a Company Plan were it in effect as of the date hereof) that is or would be material to the Debtor and its Subsidiaries, taken as a whole, or (C) any actions taken by the Debtor or any of its Subsidiaries to accelerate the vesting or lapsing of restrictions or payment, or fund or in any other way secure the payment, of any material compensation or benefits under any Company Plan, taken as a whole, except, in each case, to the extent required by applicable Laws; or
 - (v) any agreement to do any of the foregoing.
- civil, criminal or administrative actions, suits, claims, hearings, arbitrations or indictments or other proceedings by or before any Governmental Entity (collectively, "Proceedings") (or to the Knowledge of the Debtor, any audits or investigations) pending or, to the Knowledge of the Debtor, threatened in writing against the Debtor or any of its Subsidiaries, except for any such Proceeding or investigation that is not reasonably likely to result in, individually or in the aggregate, a Material Adverse Effect. As of the date of this Agreement, none of the Debtor nor any of its Subsidiaries is a party to or subject to the provisions of any temporary, preliminary or permanent judgment, order, writ, injunction, ruling, decree, assessment or award of any Governmental Entity or arbitrator (collectively, an "Order") specifically imposed upon the Debtor or any of its Subsidiaries which is, individually or in the aggregate, reasonably likely to have a Material Adverse Effect.

(h) Employee Benefits.

(i) Section 2.1(h)(i)-A of the Debtor Disclosure Letter sets forth an accurate and complete list of each material Company Plan. No Company Plan is maintained outside the jurisdiction of the United States or covers any employees or other service providers of the Debtor or any of its Subsidiaries who primarily reside or work outside the United States.

For purposes of this Agreement, "<u>Company Plan</u>" means any benefit or compensation plan, program, policy, practice, agreement, Contract, arrangement or other obligation, whether or not in writing and whether or not funded (and, in each case, which is sponsored or maintained by, or required to be contributed to, or with respect to which any Liabilities (as defined in Section 2.1(k)) are borne by the Debtor or any of its Subsidiaries. Company Plans include, but are not limited to, "employee benefit plans" within the meaning of Section 3(3) of ERISA ("<u>ERISA Plans</u>"), the Stock Plan, employment, consulting, retirement, severance, termination or change in control agreements, deferred compensation, equity-based, incentive, bonus, supplemental retirement, profit sharing, insurance, medical, welfare, vacation, fringe or other benefits or remuneration of any kind.

- (ii) With respect to each material Company Plan, the Debtor has made available to Parent, to the extent applicable, accurate and complete copies of (A) the Company Plan document, including any material amendments thereto, and all related trust documents, insurance contracts or other funding vehicles, (B) a written description of such Company Plan if such plan is not set forth in a written document, (C) the most recent annual actuarial report or annual report on Form 5500 or Form 990, and all schedules and financial statements attached thereto, (D) all material and non-routine correspondence to or from any Governmental Entity received in the last three (3) years with respect to such Company Plan, and (E) the most recent Internal Revenue Service ("IRS") determination or opinion letter, and related trust intended to be qualified under Section 401(a) of the Code and any pending request for such a determination letter.
- (iii) Except as would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect, (A) each Company Plan (including any related trusts) has been established, operated and administered in compliance with its terms and applicable Laws, including, without limitation, ERISA and the Code, (B) all contributions or other amounts payable by the Debtor or a Debtor Subsidiary with respect to each Debtor Plan in respect of current or prior plan years have been paid or accrued in accordance with generally accepted accounting principles, and (C) there are no pending or, to the Knowledge of the Debtor, threatened claims (other than routine claims for benefits) or Proceedings by a Governmental Entity by, on behalf of or against any Company Plan or any trust related thereto.
- (iv) Except as would not reasonably be expected to result in a Material Adverse Effect, each ERISA Plan that is intended to be qualified under Section 401(a) of the Code has been determined by the IRS to be qualified under Section 401(a) of the Code and, to the Knowledge of the Debtor, nothing has occurred that would reasonably be expected to result in the loss of qualification or tax exemption of any such Company Plan. Except as would not reasonably be expected to result in a Material Adverse Effect, with respect to any ERISA Plan, neither the Debtor nor any of its Subsidiaries has engaged in a transaction in connection with which the Debtor or any of its Subsidiaries reasonably would be subject to either a civil penalty assessed pursuant to Section 409 or 502(i) of ERISA or a Tax imposed pursuant to Section 4975 or 4976 of the Code.
- (v) Except as would not reasonably be expected to result in a Material Adverse Effect, neither the Debtor nor any of its Subsidiaries has or is expected to incur any material liability under subtitles C or D of Title IV of ERISA with respect to any ongoing, frozen

or terminated "single-employer plan", within the meaning of Section 4001(a)(15) of ERISA, currently or formerly maintained by any of them or any ERISA Affiliate. Except as would not, individually or in the aggregate, reasonably be expected to result in material liability to the Debtor or any of its Subsidiaries, with respect to any Company Plan subject to the minimum funding requirements of Section 412 of the Code or Title IV of ERISA, (A) no such plan is, or is reasonably expected to be, in "at-risk" status (within the meaning of Section 303(i)(4)(A) of ERISA or Section 430(i)(4)(A) of the Code), (B) as of the last day of the most recent plan year ended prior to the date hereof, the actuarially determined present value of all "benefit liabilities" within the meaning of Section 4001(a)(16) of ERISA did not exceed the then-current value of assets of such Company Plan, (C) no unsatisfied liability (other than for premiums to the PBGC) under Title IV of ERISA has been, or is reasonably expected to be, incurred by the Debtor or any of its Subsidiaries, (D) the Pension Benefit Guaranty Corporation ("PBGC") has not instituted proceedings to terminate any such Company Plan and (E) no "reportable event" within the meaning of Section 4043 of ERISA (excluding any such event for which the thirty (30)-day notice requirement has been waived under the regulations to Section 4043 of ERISA) has occurred, nor has any event described in Section 4062, Section 4063 or Section 4041 of ERISA occurred. For purposes of this Agreement, "ERISA Affiliate" means all employers (whether or not incorporated) that would be treated together with the Debtor or any of its Subsidiaries as a "single employer" within the meaning of Section 414 of the Code.

- (vi) Neither the Debtor nor any of its Subsidiaries has maintained, established, participated in or contributed to, or is or has been obligated to contribute to, or has otherwise incurred any obligation or liability (including any contingent liability) under, any "multiemployer plan" within the meaning of Section 3(37) of ERISA in the last six (6) years.
- (vii) Except as required by applicable Law, no Company Plan provides retiree or post-employment medical, disability, life insurance or other welfare benefits to any Person, and none of the Debtor or any of its Subsidiaries has any obligation to provide such benefits.
- (viii) Section 2.1(h)(viii) of the Debtor Disclosure Letter sets forth the following with respect to each Company Plan that is a "nonqualified deferred compensation plan" (within the meaning of Section 409A of the Code), the Heartland Employment Services, LLC Key Employee Incentive Plan (the "Key Employee Incentive Plan") and the Heartland Employment Services, LLC Key Employee Retention Plan, in each case as of January 31, 2018 and in a manner that is accurate in all material respects: (A) the number of participants in each such Company Plan; (B) the dollar amount of the aggregate balances for all participants (and, with respect to any such nonqualified deferred compensation plan, individual balances for any executive officer) under each such Company Plan; and (C) for any supplemental retirement plan, the amount payable under such Company Plan, assuming the participant undergoes a "separation from service" (within the meaning of Section 409A of the Code) on the Closing Date. In addition, the Debtor has provided to Parent a list of each participant in the Key Employee Incentive Plan and the Heartland Employment Services, LLC Key Employee Retention Plan with individual balances for such participants that are accurate in all material respects as of January 31, 2018.

- (ix) Neither the execution and delivery of this Agreement, nor the consummation of the Transactions will, either alone or in combination with another event, (A) entitle any current or former employee, director, officer or independent contractor of the Debtor or any of its Subsidiaries to material severance pay or any material increase in severance pay, (B) accelerate the time of payment or vesting, or materially increase the amount of compensation due to any such employee, director, officer or independent contractor, (C) directly or indirectly cause the Debtor to transfer or set aside any assets to fund any material benefits under any Company Plan, (D) otherwise give rise to any material liability under any Company Plan, (E) limit or restrict the right to merge, materially amend, terminate or transfer the assets of any Company Plan on or following the Closing Date or (F) result in the payment of any amount that could, individually or in combination with any other such payment, constitute an "excess parachute payment" as defined in Section 280G(b)(1) of the Code.
- (x) Neither the Debtor nor any of its Subsidiaries has any obligation to provide, and no Company Plan or other agreement provides any individual with the right to, a gross up, indemnification, reimbursement or other payment for any excise or additional taxes, interest or penalties incurred pursuant to Section 409A or Section 4999 of the Code or due to the failure of any payment to be deductible under Section 280G of the Code.

(i) Labor Matters.

- Section 2.1(i)(i) of the Debtor Disclosure Letter sets forth an (i) accurate and complete list of any material collective bargaining agreement or other material agreement with a labor union or like organization that the Debtor or any of its Subsidiaries is a party to or otherwise bound by (collectively, the "Debtor Labor Agreements") as of the date of this Agreement and, to the Knowledge of the Debtor, as of the date of this Agreement, there are no activities or Proceedings by any individual or group of individuals, including representatives of any labor organizations or labor unions, to organize any employees of the Debtor or any of its Subsidiaries. Within fourteen (14) days following the date of this Agreement, the Debtor will make available to Parent accurate and complete copies of each Debtor Labor Agreement listed on Section 2.1(i)(i) of the Debtor Disclosure Letter. Neither the execution and delivery of this Agreement (including any shareholder or other approval of this Agreement) nor the consummation of the Transactions would reasonably be expected to, either alone or in combination with another event, entitle any third party (including any labor organization or Governmental Entity) to any material payments or consent rights under any of the Debtor Labor Agreements. Except as would not reasonably be expected to result in a Material Adverse Effect, the Debtor and its Subsidiaries are in compliance with their obligations pursuant to all notification and bargaining obligations arising under any Debtor Labor Agreements.
- (ii) As of the date hereof, there is no strike, lockout, slowdown or work stoppage pending or, to the Knowledge of the Debtor, threatened in writing, that is reasonably expected to interfere in any material respect with the respective business activities of the Debtor or any of its Subsidiaries.
- (j) <u>Compliance with Laws; Licenses</u>. The businesses of each of the Debtor and its Subsidiaries have not, since December 31, 2014, been, and are not being, conducted in violation of any federal, state, local or foreign law, statute or ordinance, common law, or any

rule, regulation, standard, Order, agency requirement, license or permit of any Governmental Entity (collectively, "Laws"), except for violations that, individually or in the aggregate, are not reasonably likely to have a Material Adverse Effect. To the Knowledge of the Debtor, as of the date of this Agreement, no investigation by any Governmental Entity with respect to any actual or alleged violation of Law by the Debtor or any of its Subsidiaries is pending or threatened in writing, nor has any Governmental Entity notified the Debtor in writing of an intention to conduct the same, except for such investigations or reviews the outcome of which are not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect. The Debtor and its Subsidiaries each has obtained and is in compliance with all permits, licenses, certifications, approvals, registrations, consents, authorizations, franchises, variances, exemptions and orders issued or granted by a Governmental Entity ("Licenses") necessary to conduct its business as presently conducted, except those failures or the absence of which would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.

(k) <u>Undisclosed Liabilities</u>. The Debtor and its Subsidiaries do not have any debts, claims, liabilities or obligations, including all costs and expenses relating thereto, whether absolute or contingent, known or unknown, liquidated or unliquidated, whether due or to become due and regardless of when or by whom asserted ("<u>Liabilities</u>") that would have been required to be reflected or reserved against, or otherwise described on a consolidated balance sheet of the Debtor prepared in accordance with GAAP or in the notes thereto, other than Liabilities (i) reflected or reserved against in the balance sheet included in the Audited Financial Statements or disclosed in the notes thereto, (ii) incurred in the ordinary course of the operation of business consistent with past practice of the Debtor and its Subsidiaries since December 31, 2016, (iii) arising or incurred in connection with the Transactions or (iv) which have not had, or would not, individually or in the aggregate, reasonably be expected to have, a Material Adverse Effect.

(1) Material Contracts and Government Contracts.

- (i) Section 2.1(l)(i) of the Debtor Disclosure Letter sets forth a list of each of the following Contracts to which, as of the date hereof, the Debtor or any of its Subsidiaries is a party or by which the Debtor or any of its Subsidiaries or any of their respective assets is bound (each, a "Material Contract"):
- (A) any Contract (or group of related Contracts with respect to a single transaction or series of related transactions) pursuant to which the Debtor or any of its Subsidiaries currently leases or subleases real or personal property to or from any Person (other than Parent or any of its Affiliates) providing for lease payments in excess of \$5,000,000 per annum (in each case, other than the Subleases);
- (B) any Contract pursuant to which any material Intellectual Property Rights are (1) licensed by any Person to the Debtor or any of its Subsidiaries (other than non-exclusive licenses to the Debtor or any of its Subsidiaries for non-customized software that is generally available on commercial terms) or (2) licensed by the Debtor or any of its Subsidiaries to any Person;

- (C) each joint venture, partnership and other similar Contract involving the sharing of profits of the Debtor or any of its Subsidiaries with any third party;
- (D) each Contract that expressly limits the freedom of the Debtor or any of its Subsidiaries (or, after the Closing, Parent or any of its Subsidiaries) to compete in any line of business or within any geographic area or with any Person;
- (E) each Contract under which the Debtor or any of its Subsidiaries has borrowed or loaned money, or any note, bond, indenture, mortgage or any guarantee of such indebtedness, in each case, relating to amounts in excess of \$5,000,000;
- (F) each Contract entered into in the past three (3) years relating to the acquisition or disposition of assets (other than in the ordinary course of business) or any capital stock of any enterprise, in each case, in excess of \$10,000,000 individually (other than such Contracts previously approved in writing by Parent or one of its Subsidiaries pursuant to the Master Lease);
- (G) each Contract entered into since January 1, 2015 related to any settlement or stipulation of any action against the Debtor or any of its Subsidiaries by any other Person, other than settlement agreements for cash that do not exceed \$1,000,000 individually as to any such settlement or stipulation (excluding amounts paid by insurers) or \$5,000,000 (including any amounts paid by insurers); and
- (H) each Contract that provides for indemnification of any officer, director or employee of the Debtor or any of its Subsidiaries other than in the ordinary course of business.
- (ii) Each Material Contract and each Government Contract is valid and binding on the Debtor or its Subsidiaries, as the case may be, and, to the Knowledge of the Debtor, each other party thereto, and, to the Knowledge of the Debtor, is in full force and effect, except for such failures to be valid and binding or to be in full force and effect as would not, or would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect. There is no material breach of or default under any such Contracts by the Debtor or its Subsidiaries and no event has occurred that, with the lapse of time or the giving of notice or both, would constitute a material breach of or default thereunder by the Debtor or any of its Subsidiaries or would permit termination, modification or acceleration thereof, in each case except as would not, or would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect. To the Knowledge of the Debtor, no party to any Material Contract or any Government Contract has delivered any written notice to the Debtor or any of its Subsidiaries that it is terminating such Material Contract or Government Contract, except as would not, or would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect.
- (iii) For purposes of this Agreement, the following terms shall have the following meanings:

- (A) "<u>Affiliate</u>" when used with respect to any party shall mean any Person who is an "affiliate" of that party within the meaning of Rule 405 promulgated under the Securities Act of 1933, as amended (the "<u>Securities Act</u>");
- (B) "<u>Government Contract</u>" means any contract to which the Debtor or any of its Subsidiaries is a party, or by which any of them is bound, the ultimate contracting party of which is a Governmental Entity (including any subcontract with a prime contractor or other subcontractor who is a party to any such contract); and
- (C) "<u>Intellectual Property Rights</u>" means all intellectual property rights arising under the Laws of any jurisdiction in the world, including all rights in or to any of the following: (1) trademarks, service marks, certification marks, collective marks, d/b/a's, Internet domain names, symbols, trade dress, trade names, and other indicia of origin, all applications and registrations for the foregoing, in each case, including all goodwill associated therewith and symbolized thereby; (2) patents, invention disclosures and applications therefor, including divisions, continuations, continuations-in-part and renewal applications, and including renewals, extensions and reissues; (3) trade secrets, proprietary or confidential information, inventions, know-how, processes, formulae, models, methodologies, formulae, drawings, prototypes, models, designs, customer lists and supplier lists; and (4) published and unpublished works of authorship, copyrights therein and thereto, and registrations and applications therefor, and all renewals, extensions, restorations and reversions thereof.
 - (m) Real Property. The Debtor does not own any real property.
- (n) <u>Takeover Statutes</u>. No "fair price", "moratorium", "control share acquisition" or other similar anti-takeover statute or regulation enacted under state or federal Laws in the United States (each, a "<u>Takeover Statute</u>") or any anti-takeover provision in the Debtor's certificate of incorporation or by-laws is applicable to the Debtor, the Shares or the Transactions.
- (o) Environmental Matters. Except as would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect, to the Knowledge of the Debtor, each real property operated by the Debtor or any of its Subsidiaries and used in the conduct of their respective businesses: (i) is in substantial compliance with all applicable Environmental Laws; (ii) is not the subject of any pending written notice from any Governmental Entity alleging the violation of any applicable Environmental Law; (iii) is not currently subject to any Orders arising under any Environmental Law; and (iv) has not had any emissions or discharges of Hazardous Substances except as permitted under applicable Environmental Laws.

As used herein, (i) the term "<u>Environmental Law</u>" means any federal, state, local or foreign statute, law, regulation, order, decree, permit, authorization or requirement of any Governmental Entity relating to: (A) the protection, investigation or restoration of the environment, natural resources or health or safety (in relation to exposure to Hazardous Substances), (B) the handling, use, presence, disposal, release or threatened release of any Hazardous Substance or (C) noise, odor, indoor air, employee exposure, wetlands, pollution, contamination or any injury or threat of injury to persons or property relating to any Hazardous Substance; and (ii) the term "<u>Hazardous Substance</u>" means any substance that is: (A) listed,

classified or regulated pursuant to any Environmental Law; (B) any petroleum product or byproduct, asbestos-containing material, lead-containing paint or plumbing, polychlorinated biphenyls, mold, radioactive material or radon; and (C) any other substance which may be the subject of regulatory action by any Governmental Entity in connection with any Environmental Law.

(p) <u>Taxes</u>.

- (i) The Debtor and each of its Subsidiaries has prepared (or caused to be prepared) and timely filed (taking into account valid extensions of time within which to file) all material Tax Returns required to be filed by any of them, and all such filed Tax Returns (taking into account all amendments thereto) are accurate and complete in all material respects;
- (ii) all material Taxes owed by the Debtor and each of its Subsidiaries that are due and payable (whether or not shown as due on a Tax Return) have been timely paid or, if not yet due and payable, adequately reserved against in accordance with GAAP;
- (iii) no deficiency for any amount of material Taxes has been proposed or asserted in writing or assessed by any Taxing Authority against the Debtor or any of its Subsidiaries that remains unpaid or unresolved;
- (iv) the Debtor and each of its Subsidiaries has not received written notice of any pending Proceedings, examinations or proposed adjustments in respect of any material amount of Taxes of the Debtor or any of its Subsidiaries;
- (v) all material Taxes required to be withheld or collected by the Debtor or any of its Subsidiaries have been withheld and collected and, to the extent required by applicable Law, timely paid to the appropriate Governmental Entity or other person, and all IRS forms required with respect thereto have been properly completed and timely filed;
- (vi) there are no material Liens for Taxes on any of the assets of the Debtor or any of its Subsidiaries, other than Permitted Liens;
- (vii) neither the Debtor nor any of its Subsidiaries has been a "controlled corporation" or a "distributing corporation" in any distribution occurring during the two (2)-year period ending on the date of this Agreement that was purported or intended to be governed by Section 355 of the Code (or any similar provision of state, local or non-U.S. Law);
- (viii) no claim has been received, or is expected by the Debtor to be received, with respect to the Debtor or any of its Subsidiaries from a Taxing Authority in a jurisdiction where such entity does not file Tax Returns that it is or may be subject to taxation by, or required to file any Tax Return in, that jurisdiction which claim has not since been resolved;
- (ix) during the past ten (10) years, neither the Debtor nor any of its Subsidiaries has been a member of an affiliated group of corporations filing a consolidated federal income Tax Return (other than a group the common parent of which is the Debtor) or has had any liability for the Taxes of any person (other than the Debtor or any of its Subsidiaries)

under U.S. Treasury Regulation Section 1.1502-6 (or any similar provision of any state, local or non-U.S. Law), as a transferee or successor;

- (x) neither the Debtor nor any of its Subsidiaries is a party to, or is bound by, or has any obligation under, any Tax sharing Contract, Tax allocation agreement, Tax indemnity obligation or similar agreement or practice other than (A) Contracts solely among the Debtor and/or its Subsidiaries and (B) customary Tax indemnification provisions in Contracts entered into in the ordinary course of business, the primary purpose of which does not relate to Taxes;
- (xi) neither the Debtor nor any of its Subsidiaries has currently in effect any waiver of any statute of limitations in respect of Taxes or any agreement to any extension of time with respect to the filing of a Tax Return or an assessment or deficiency for Taxes (other than pursuant to extensions of time to file Tax Returns obtained in the ordinary course of business);
- (xii) neither the Debtor nor any of its Subsidiaries has participated in any "listed transaction" within the meaning of U.S. Treasury Regulations Section 1.6011-4(b); and
- (xiii) the Debtor has made available to the Purchaser Entities accurate and complete copies of any private letter ruling requests, closing agreements or gain recognition agreements with respect to Taxes of the Debtor or any of its Subsidiaries requested or executed in the last five (5) years.

As used in this Agreement, (i) the term "<u>Tax</u>" (including, with correlative meaning, the term "<u>Taxes</u>") means all federal, state, local or foreign taxes, including, without limitation, income, gross receipts, windfall profits, severance, property, production, sales, use, social security, license, excise, franchise, employment, withholding or similar taxes together with any interest, additions or penalties with respect thereto and any interest in respect of such additions or penalties, (ii) the term "<u>Tax Return</u>" means any return, filing, report, questionnaire, information statement, claim for refund or declaration of estimated Taxes, including any schedule or attachment thereto or any amendment thereof, filed or required to be filed with any Taxing Authority in connection with the determination, assessment or collection of any Tax, or the administration of any laws, regulations or administrative requirements relating to any Tax, including consolidated, combined and unitary tax returns and (iii) the term "<u>Taxing Authority</u>" means any Governmental Entity exercising regulatory authority in respect of taxes or responsible for the imposition of any Tax.

(q) Intellectual Property.

(i) To the Knowledge of the Debtor, the Debtor and its Subsidiaries own or have a valid and enforceable right to use all material Intellectual Property Rights that are used in or necessary for the conduct of their respective businesses as presently conducted. All Intellectual Property Rights that are owned by the Debtor or any of its Subsidiaries ("Owned Intellectual Property") are, to the Knowledge of the Debtor, (A) valid, subsisting and enforceable and (B) not subject to any outstanding Order adversely affecting the Debtor's or any

of its Subsidiaries' use of, or rights in or to any such Intellectual Property Rights, except in the case of clauses (A) and (B) as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

- (ii) To the Knowledge of the Debtor, neither the Debtor nor any of its Subsidiaries is infringing, misappropriating or otherwise violating any Intellectual Property Rights of any Person, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (iii) To the Knowledge of the Debtor, no Person is infringing, misappropriating or otherwise violating any Owned Intellectual Property, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (iv) To the Knowledge of the Debtor, no Person has gained unauthorized access to any Debtor IT Asset, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (v) The Debtor and each of its Subsidiaries are in compliance in all material respects with their respective privacy and security policies and with all applicable Laws regarding privacy and personal information, including with respect to the collection, storage, transmission, transfer, disclosure and use of personal information, in each case, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The Debtor and each of its Subsidiaries have taken commercially reasonable measures consistent with customary industry practices to ensure the confidentiality, privacy and security of all personal information of their customers and employees, and no Person has gained unauthorized access to, or misused, any such information, in each case, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (vi) For purposes of this Agreement, "<u>Debtor IT Assets</u>" means all computers, software, firmware, middleware, servers, workstations, routers, hubs, switches, data communications lines, and all other information technology equipment and associated documentation owned or used by the Debtor or any of its Subsidiaries.
- (r) <u>Insurance</u>. The fire and casualty, general liability, business interruption, product liability, and sprinkler and water damage insurance policies maintained by the Debtor or any of its Subsidiaries ("<u>Insurance Policies</u>") provide full and adequate coverage for all normal risks incident to the business of the Debtor and its Subsidiaries and their respective properties and assets, except for any such failures to maintain such insurance policies that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Each of the Insurance Policies is in full force and effect and all premiums due with respect to all Insurance Policies have been paid, in each case, with such exceptions that, individually or in the aggregate, are not reasonably expected to have a Material Adverse Effect.
- (s) <u>Brokers and Finders</u>. Neither the Debtor nor any of its officers, directors or employees has employed any broker or finder or incurred any Liability for any brokerage fees, commissions or finders' fees in connection with the Transactions or otherwise, except for fees payable to Moelis & Company as its financial advisor. The Debtor has made available to Parent

a complete and accurate copy of all agreements pursuant to which Moelis & Company is entitled to any fees and expenses in connection with any of the Transactions or otherwise.

- 2.2 <u>Representations and Warranties of the Purchaser Entities</u>. Except as set forth in the forms, statements, certifications, reports and documents filed with or furnished to the SEC prior to the date of this Agreement by Parent pursuant to the Exchange Act of 1934, as amended (the "<u>Exchange Act</u>"), or the Securities Act (excluding, in each case, any disclosures set forth in any risk factor section or in any other section to the extent they are forward-looking statements or cautionary, predictive or forward-looking in nature), the Purchaser Entities each hereby represent and warrant to the Debtor that:
- (a) Organization, Good Standing and Qualification. Each of the Purchaser Entities is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing (with respect to jurisdictions that recognize the concept of good standing) as a foreign corporation or other entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so organized, qualified or in such good standing, or to have such power or authority, would not, individually or in the aggregate, reasonably be expected to prevent, materially delay or impair the ability of the Purchaser Entities to consummate the Transactions.
- (b) <u>Corporate Authority</u>. No vote of holders of capital stock of Parent is necessary to approve this Agreement or the other Transactions. Each of the Purchaser Entities has all requisite corporate or similar power and authority and has taken all corporate or similar action necessary in order to execute, deliver and perform its obligations under this Agreement and to consummate the Transactions. This Agreement has been duly executed and delivered by each of the Purchaser Entities and is a valid and binding agreement of the Purchaser Entities, enforceable against each of Purchaser Entities in accordance with its terms, subject to the Bankruptcy and Equity Exception.

(c) Governmental Filings; No Violations; Etc.

- (i) Other than the Governmental Approvals, no notices, reports or other filings are required to be made by the Purchaser Entities with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by the Purchaser Entities from, any Governmental Entity in connection with the execution, delivery and performance of this Agreement by the Purchaser Entities and the consummation by the Purchaser Entities of the Transactions, except those that the failure to make or obtain would not, individually or in the aggregate, reasonably be expected to prevent or materially delay the ability of the Purchaser Entities to consummate the Transactions.
- (ii) The execution, delivery and performance of this Agreement by the Purchaser Entities do not, and the consummation by the Purchaser Entities of the Transactions contemplated hereby will not, constitute or result in (A) a breach or violation of, or a default under, the certificate of incorporation or by-laws of the Purchaser Entities or the comparable

governing instruments of any of their respective Subsidiaries, (B) with or without notice, lapse of time or both, a breach or violation of, a termination (or right of termination) or a default under, the creation or acceleration of any obligations under or the creation of a Lien on any of the assets of the Purchaser Entities or any of their respective Subsidiaries pursuant to, any Contracts binding upon the Purchaser Entities or any Laws or governmental or non-governmental permit or license to which the Purchaser Entities or any of their respective Subsidiaries is subject, or (C) any change in the rights or obligations of any party under any of such Contracts, except, in the case of clause (B) or (C) above, for any breach, violation, termination, default, creation, acceleration or change that would not, individually or in the aggregate, reasonably be expected to prevent or materially delay the ability of the Purchaser Entities to consummate the Transactions.

- (d) <u>Capitalization of Purchaser</u>. All of the issued and outstanding partnership interests of Purchaser are, and at the Closing Date will be, owned by Parent or a direct or indirect wholly-owned Subsidiary of Parent. All of the partnership interests of Purchaser have been duly authorized and validly issued in accordance with the Amended and Restated Limited Partnership Agreement of Purchaser, dated as of October 31, 2016 (the "<u>Purchaser LP Agreement</u>"), and, except as set forth in the Purchaser LP Agreement, are fully paid (to the extent required under the Purchaser LP Agreement) and nonassessable (except as such nonassessability may be affected by Section 17-303, 17-607 and 17-804 of the Delaware Revised Uniform Limited Partnership Act, as amended, and the Purchaser LP Agreement).
- (e) <u>Proceedings</u>. As of the date of this Agreement, other than the Bankruptcy Case and chapter 11 proceedings for HCR III and/or its Subsidiaries, there are no Proceedings pending or, to the knowledge of the officers of the Purchaser Entities, threatened against the Purchaser Entities that seek to enjoin, or would reasonably be expected to have the effect of preventing, making illegal, materially delaying or otherwise interfering with the ability of the Purchaser Entities to consummate the Transactions.
- No Other Company Representations or Warranties. Except for the representations and warranties set forth in Section 2.1, the Purchaser Entities hereby acknowledge and agree that neither the Debtor nor any of its Subsidiaries, nor any of their respective stockholders, Affiliates or Representatives, has made or is making any other express or implied representation or warranty with respect to the Debtor or any of its Subsidiaries or their respective businesses or operations, including with respect to any information provided or made available to the Purchaser Entities. Subject to Section 8.1(a), neither the Debtor nor any of its Subsidiaries, nor any of their respective stockholders, Affiliates or Representatives, will have or be subject to any liability or indemnification obligation to the Purchaser Entities resulting from the delivery, dissemination or any other distribution to the Purchaser Entities, or the use by the Purchaser Entities, of any such information provided or made available to them by the Debtor or any of its Subsidiaries, or any of their respective stockholders, Affiliates or Representatives, including any information, documents, estimates, projections, forecasts or other forward-looking information, business plans or other material provided or made available to Parent, the Purchaser Entities or any of their Affiliates or Representatives, in anticipation or contemplation of any of the Transactions.
- (g) <u>Non-Reliance on Debtor Estimates, Projections, Forecasts, Forward-</u> <u>Looking Statements and Business Plans</u>. In connection with the due diligence investigation of

the Debtor by the Purchaser Entities, the Purchaser Entities have received and may continue to receive from the Debtor certain estimates, projections, forecasts and other forward-looking information, as well as certain business plan information, regarding the Debtor, its Subsidiaries and their respective businesses and operations. The Purchaser Entities hereby acknowledge that there are uncertainties inherent in attempting to make such estimates, projections, forecasts and other forward-looking statements, as well as in such business plans, with which Purchaser Entities are familiar, that the Purchaser Entities are taking full responsibility for making their own evaluation of the adequacy and accuracy of all estimates, projections, forecasts and other forward-looking information, as well as such business plans, so furnished to them (including the reasonableness of the assumptions underlying such estimates, projections, forecasts, forwardlooking information or business plans), and that, subject to Section 8.1(a), the Purchaser Entities will have no claim against the Debtor or any of its Subsidiaries, or any of their respective stockholders, Affiliates or Representatives, with respect thereto. Accordingly, the Purchaser Entities hereby acknowledge that neither the Debtor nor any of its Subsidiaries, nor any of their respective stockholders, Affiliates or Representatives, has made or is making any representation or warranty with respect to such estimates, projections, forecasts, forward-looking statements or business plans (including the reasonableness of the assumptions underlying such estimates, projections, forecasts, forward-looking statements or business plans).

ARTICLE III

Bankruptcy Matters

3.1 <u>Pre-Filing Period</u>.

- (a) Prior to the Petition Date, Parent and the Debtor shall work together in good faith to prepare the Plan, proposed Confirmation Order and Disclosure Statement (collectively, the "**Restructuring Documents**") and "first day" and "second day" motions, and any other motions, pleadings, notices and other documents to be filed during the Bankruptcy Case.
- (b) For purposes of this Agreement, (i) the "Confirmation Order" shall mean an Order of the Bankruptcy Court, reasonably acceptable to the Debtor and Parent that, among other things, (A) approves the Disclosure Statement, voting materials and procedures, and solicitation, (B) finds that the Plan is feasible, that adequate notice has been provided to all parties-in-interest in the Bankruptcy Case and that the requirements under the Bankruptcy Code for classification of claims, solicitation of votes, and confirmation of a Plan have been satisfied, (C) confirms the Plan, (D) approves this Agreement and all of its terms, and authorizes and directs the Debtor to perform its obligations under this Agreement, (E) approves the transactions contemplated by the Plan and this Agreement, (F) exculpates and releases the members of the Debtor Board and officers of the Debtor from liability with respect to the negotiation, execution and delivery of this Agreement and the Debtor's performance of its obligations hereunder prior to the Closing (or termination of this Agreement in accordance with its terms, as applicable) and (G) approves the releases, exculpations and discharge contemplated by the Plan and enjoins parties from asserting claims subject to such releases, exculpations and/or discharge; and (ii) the "Disclosure Statement" shall mean the disclosure statement with respect to the Plan, which

shall contain "adequate information" as defined by section 1125(a) of the Bankruptcy Code and shall be reasonably acceptable to Parent and the Debtor.

- (c) The Debtor shall give Parent reasonable advance notice of, and opportunity to review and comment on, the "first day" and "second day" motions, and any other motions, pleadings, notices and other documents to be filed during the Bankruptcy Case and shall incorporate such comments to the extent reasonable and consistent with the terms of this Agreement and the Plan.
- (d) The Debtor has provided Parent with a Case Budget on the date hereof and shall deliver an updated Case Budget to Parent from time to time as reasonably requested by Parent. The "Case Budget" shall mean a budget for the Bankruptcy Case prepared in good faith by the Debtor in consultation with Parent based on the 13-Week Cash Budget, which shall include (i) expected Closing Date payments and (ii) funding for the Bankruptcy Case and Closing Date payments.

3.2 Chapter 11 Obligations.

- (a) The Debtor shall commence solicitation of the Plan on or prior to one (1) day after the date hereof.
- (b) On or prior to March 4, 2018, the Debtor shall file or cause to be filed a petition for relief under Chapter 11 of the Bankruptcy Code with the Bankruptcy Court. On the date of such filing (the "<u>Petition Date</u>"), the Debtor shall file with the Bankruptcy Court the Restructuring Documents and a motion seeking entry of (i) an order approving the Disclosure Statement and (ii) the Confirmation Order.
- (c) The Debtor shall use reasonable best efforts to pursue entry of the Confirmation Order by the Bankruptcy Court within 40 days of the Petition Date.
- (d) In the event the entry of the Confirmation Order shall be appealed, the Debtor and the Purchaser Entities shall use their respective reasonable best efforts to defend such appeal, including opposing any stay requested in connection therewith.
- (e) Each document filed with the Bankruptcy Court as a supplement to the Plan, and all exhibits, schedules and annexes thereto, shall be consistent with the terms hereof and the Plan Term Sheet and otherwise reasonably satisfactory in form and substance to Parent; provided that the Charter and the By-Laws shall be satisfactory to Parent in its sole discretion and in compliance with Section 4.7.
- (f) The Debtor shall not file any motion or pleading or make any statement on the record before the Bankruptcy Court that is inconsistent with the Restructuring Documents or this Agreement or that reasonably could be expected to prevent, delay or impede the successful implementation of the Transactions and the Restructuring Documents.
- (g) The Restructuring Documents may not be amended or modified unless such amendments or modifications are (i) not inconsistent with the terms and conditions of this

Agreement and the Plan Term Sheet and (ii) consented to in writing by Parent, such consent not to be unreasonably withheld, conditioned or delayed.

- (h) Parent shall, and shall cause each Subsidiary of Parent to, (i) support, and take all reasonable actions necessary or reasonably requested by the Debtor to facilitate, the solicitation, confirmation and consummation of the Plan and the transactions contemplated by the Plan, including voting in favor of the Plan; (ii) not take any other action, directly or indirectly, that could prevent, interfere with, delay or impede the solicitation of votes in connection with the Plan or the confirmation or consummation of the Plan; (iii) not object to or otherwise commence any proceeding or take any action opposing the Plan or Disclosure Statement; and (iv) otherwise use its reasonable best efforts to assist the Debtor in obtaining entry of the Confirmation Order by the Bankruptcy Court within 40 days of the Petition Date.
- (i) At the Closing, Parent shall make available, or cause to be made available, to the Debtor such amounts as may be necessary to enable the Debtor to fund or pay all of the Debtor's obligations required to be funded or paid under the Plan or this Agreement (to the extent the Debtor does not have sufficient funds to fund or pay such obligations when due).
- (j) If an Eligible Employee does not execute and deliver a Separation Agreement, or the Former CEO does not execute and deliver the Former CEO Settlement Agreement, in each case prior to the entry of the Confirmation Order, the Debtor shall, at the request of Parent, promptly file with the Bankruptcy Court a motion to reject the Employment Agreement of such Eligible Employee or the Former CEO, as applicable, as of the Closing Date, and shall use reasonable best efforts to pursue entry of an order by the Bankruptcy Court prior to the Closing Date approving such rejection.
- 3.3 <u>Case Financing</u>. The Debtor agrees that it shall not incur any secured debtor-in-possession financing except with the prior written consent of Parent, not to be unreasonably withheld, conditioned or delayed; it being understood that, for the avoidance of doubt, the granting of security as contemplated by the Centerbridge Facility, or any customary and reasonable adequate protection to be provided to the lenders under the Centerbridge Facility as part of the Bankruptcy Case (other than as a consequence of secured debtor-in-possession financing), shall not be considered a violation of this Section 3.3.
- 3.4 Release of Non-Debtor Subsidiary Claims. Parent and Debtor hereby covenant and agree that, on the Effective Date, for good and valuable consideration, to the fullest extent permissible under applicable law, (a) each Subsidiary of the Debtor shall be deemed to have completely and forever released, waived, and discharged unconditionally each of the shareholders or other equity holders of the Debtor that is not an employee of the Debtor and each director of the Debtor that is not an employee of the Debtor, and any Person claiming by or through any of the foregoing Persons (including their respective officers, directors, managers, shareholders, partners, employees, equity holders, members, and professionals) (the "Non-Debtor Subsidiary Released Parties"), and Parent, Purchaser and each Lessor shall be deemed to have completely and forever released, waived, and discharged unconditionally each Non-Debtor Subsidiary Released Party, of and from any and all Claims (as such term is defined in the Plan) and all other obligations, suits, judgments, damages, debts, rights, remedies, causes of action and liabilities of any nature whatsoever (including, without limitation, those arising under

the Bankruptcy Code), whether liquidated or unliquidated, fixed or contingent, direct or derivative, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity or otherwise that are or may be based in whole or part on any act, omission, transaction, event or other circumstance taking place or existing on or prior to the Effective Date (including prior to the Petition Date) in connection with or related to the Debtor, the reorganized Debtor, their respective Subsidiaries, assets, property and estates, the Bankruptcy Case or the Restructuring Documents and (b) following the Closing, promptly upon the written request of any Non-Debtor Subsidiary Released Party, Parent and Debtor shall cause each Subsidiary of Debtor to (and Parent shall and shall cause Purchaser and each Lessor to) execute and deliver to such Non-Debtor Subsidiary Released Party a written instrument evidencing such release.

ARTICLE IV

Covenants

4.1 <u>Interim Operations</u>.

- The Debtor covenants and agrees as to itself and its Subsidiaries that, after (a) the date of this Agreement and prior to the Closing, except (i) as otherwise expressly required or contemplated by this Agreement, (ii) as Parent may approve in writing (such approval not to be unreasonably withheld, conditioned or delayed), (iii) as otherwise required by applicable Laws or (iv) as set forth in Section 4.1 of the Debtor Disclosure Letter, the Business shall be conducted in the ordinary and usual course and, to the extent consistent therewith, the Debtor and its Subsidiaries shall use their respective commercially reasonable efforts to preserve their business organizations intact, preserve governmental licenses, permits, consents, approvals, authorizations and qualifications and maintain existing relations and goodwill with Governmental Entities, customers, suppliers, distributors, creditors, lessors, employees and business associates, and keep available the services of its and its Subsidiaries' present employees and agents. Without limiting the generality of, and in furtherance of, the foregoing, from the date of this Agreement until the Closing, except (i) as otherwise expressly required or contemplated by this Agreement, (ii) as Parent may approve in writing (such approval not to be unreasonably withheld, conditioned or delayed), (iii) as otherwise required by applicable Laws or (iv) as set forth in Section 4.1 of the Debtor Disclosure Letter, the Debtor will not and will not permit any of its Subsidiaries to:
- (i) adopt or propose any change in its certificate of incorporation or by-laws or other applicable governing instruments;
- (ii) merge or consolidate the Debtor or any of its Subsidiaries with any other Person, or restructure, reorganize or completely or partially liquidate the Debtor or any of its Subsidiaries or otherwise enter into any agreements providing for the sale of their respective material assets, operations or business (other than the sale or disposition of obsolete or worn-out assets in the ordinary course of business);
- (iii) acquire assets outside of the ordinary course of business from any other Person with a value or purchase price in the aggregate in excess of \$1,000,000 in any transaction or series of related transactions, other than acquisitions pursuant to Contracts in effect

as of the date of this Agreement which have been provided to Parent prior to the date of this Agreement;

- (iv) acquire any corporation, partnership or other business organization or division thereof or collection of assets constituting all or substantially all of a business or business unit, whether by merger or consolidation, purchase of substantial assets or equity interest or any other manner, from any other Person;
- (v) issue, sell, pledge, dispose of, grant, transfer, encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer or encumbrance of, any shares of capital stock of the Debtor or any of its Subsidiaries (other than the issuance of shares by a wholly-owned Subsidiary of the Debtor to the Debtor or another wholly-owned Subsidiary), or securities convertible or exchangeable into or exercisable for any shares of such capital stock, or any options, warrants or other rights of any kind to acquire any shares of such capital stock or such convertible or exchangeable securities;
- (vi) create or incur any Lien securing indebtedness for borrowed money (other than a Lien currently provided for under the Centerbridge Facility, any Permitted Lien (other than clause (d) of such definition) and/or the grant of any cash collateral in respect of letters of credit issued in respect of, or otherwise securing, ordinary course operating liabilities) on any assets of the Debtor or any of its Subsidiaries having a value in excess of \$1,000,000 in the aggregate;
- (vii) make any loans, advances, guarantees or capital contributions to or investments in any Person (other than the Debtor or any direct or indirect wholly-owned Subsidiary of the Debtor);
- (viii) declare, set aside, make or pay any dividend or other distribution, payable in cash, stock, property or otherwise, with respect to any of its capital stock (except for dividends paid by any direct or indirect wholly-owned Subsidiary to the Debtor or to any other direct or indirect wholly-owned Subsidiary) or enter into any agreement with respect to the voting of its capital stock (other than the Restructuring Support Agreement);
- (ix) reclassify, split, combine, subdivide or redeem, purchase or otherwise acquire, directly or indirectly, any of its capital stock or securities convertible or exchangeable into or exercisable for any shares of its capital stock;
- (x) incur any indebtedness for borrowed money (which, for the avoidance of doubt, shall not include obligations in respect of cash-collateralized letters of credit issued in respect of, or other grants of cash collateral securing, ordinary course operating liabilities) or guarantee such indebtedness of another Person, or issue or sell any debt securities or warrants or other rights to acquire any debt security of the Debtor or any of its Subsidiaries, except for indebtedness for borrowed money incurred in the ordinary course of business consistent with past practice (A) not to exceed \$2,000,000 in the aggregate, (B) guarantees incurred in compliance with this Section 4.1 by the Debtor of indebtedness of wholly-owned Subsidiaries of the Debtor or (C) indebtedness owed to the Debtor or another wholly-owned Subsidiary of the Debtor;

- (xi) except as set forth in the capital expenditures budget set forth in Section 4.1(a)(xi) of the Debtor Disclosure Letter, make or authorize any capital expenditure in excess of \$2,000,000 in the aggregate, excluding any capital expenditure required by any Contract set forth on Section 4.1(a)(xi) of the Debtor Disclosure Letter or capital expenditure determined in good faith by the Debtor Board to be required for (A) the protection of, or to avoid injury to, any Person, (B) the care or safety of any patient under the care of any facility operated by the Debtor or any of its Subsidiaries or (C) compliance with Law;
- (xii) enter into any Contract that would have been a Material Contract had it been entered into prior to this Agreement;
- (xiii) make any material changes with respect to material accounting policies or procedures, except as required by changes in applicable Law or GAAP;
- (xiv) settle any litigation or other Proceeding brought against the Debtor or its Subsidiaries by a Governmental Entity (A) for an amount in excess of \$100,000 individually or \$1,000,000 in the aggregate for all such Proceedings (other than any resolution of claims processing for government reimbursement in the ordinary course of business, and excluding recoupment actions) or (B) in a manner that would impose any restrictions on its assets, operations or businesses or result in any injunction or equitable relief against the Debtor or any of its Subsidiaries;
- (xv) settle any Proceeding other than against or brought by a Governmental Entity, (A) for an amount in excess of \$500,000 individually or \$8,000,000 in the aggregate for all such Proceedings in any one-calendar-month period (in each case, with respect to Proceedings in the state of Pennsylvania, net of applicable insurance proceeds) or (B) in a manner that would impose any restrictions on its assets, operations or businesses or result in any injunction or equitable relief against the Debtor or any of its Subsidiaries;
- (xvi) amend, modify or terminate any Material Contract, including the Centerbridge Facility, in a manner adverse to the Debtor or its Subsidiaries;
- (xvii) (A) change in any material respect any material method of accounting of the Debtor or its Subsidiaries for Tax purposes; (B) enter into any agreement with any Governmental Entity (including a "closing agreement" under Code Section 7121) with respect to any material Tax or Tax Returns of the Debtor or its Subsidiaries; (C) surrender a right of the Debtor or its Subsidiaries to a material Tax refund; (D) change an accounting period of the Debtor or its Subsidiaries with respect to any material Tax; (E) file an amended Tax Return; (F) change or revoke any material election with respect to Taxes; (G) make any material election with respect to Taxes that is inconsistent with past practice; (H) file any Tax Return that is inconsistent with past practice; or (I) consent to any extension or waiver of the limitations period applicable to any material Tax claim or assessment (other than in the ordinary course of business);
- (xviii) transfer, sell, lease, license, mortgage, pledge, divest or otherwise dispose of any material tangible or intangible assets (including Intellectual Property Rights), licenses, operations, rights, product lines, businesses or interests therein of the Debtor or its

Subsidiaries, including the capital stock of any of its Subsidiaries, except in connection with services provided in the ordinary course of business and sales or other dispositions of obsolete or worn-out assets and except for sales, leases, licenses, divestitures, cancellations, abandonments, lapses, expirations or other dispositions of assets with a fair market value not in excess of \$500,000 in the aggregate, other than pursuant to Contracts in effect prior to the date of this Agreement;

(A) enter into, adopt, amend in any material respect or terminate any Company Plan (other than entry into any new employment agreement with any individual whose hiring is not restricted by, or who is otherwise hired in accordance with, clause (H) below), (B) increase or accelerate the compensation, bonus, pension, welfare, fringe or other benefits, severance or termination pay of any director, officer or employee of the Debtor or any of its Subsidiaries, (C) grant any new awards, or amend or modify the terms of any outstanding awards, under any Company Plan (other than grants of any new awards to any individual whose hiring is not restricted by, or who is otherwise hired in accordance with, clause (H) below), (D) take any action to accelerate the vesting or lapsing of restrictions or payment, or fund or in any other way secure the payment, of compensation or benefits under any Company Plan, (E) change any actuarial or other assumptions used to calculate funding obligations with respect to any Company Plan that is required by applicable Law to be funded or change the manner in which contributions to such plans are made or the basis on which such contributions are determined, except as may be required by the terms of any existing Company Plan set forth on Section 2.1(h)(i) of the Debtor Disclosure Letter or GAAP, (F) forgive any loans or issue any loans (other than routine travel advances issued in the ordinary course of business) to any director, officer or employee of the Debtor or any of its Subsidiaries, (G) terminate the employment of any officer of the Debtor or its Subsidiaries other than for "cause", (H) hire (x) any officer of the Debtor or its Subsidiaries with a title of Vice President or higher or (y) any employee of the Debtor or its Subsidiaries with aggregate annual base salary and target bonus of more than \$250,000, except, in the case of the foregoing clauses (x) and (y), to the extent jointly determined by the Chief Restructuring Officer of the Debtor ("CRO") and the Debtor Board in their reasonable business judgment in good faith necessary in the interests of patient care (any such individual described in the foregoing clauses (x) and (y) and hired to replace any such employee, a "New Hire"), provided that (i) any such officer New Hire (and his or her terms and conditions of employment, including any base and target incentive compensation) hired pursuant to this clause (H) shall be reasonably acceptable to Parent and (ii) the terms and conditions of employment of any New Hire that is not an officer, including base and target incentive compensation, shall be subject to notice and consultation with Parent, or (I) make any incentive payment or payment in respect of severance or any nonqualified deferred compensation entitlement to any current or former director, officer or employee of the Debtor or its Subsidiaries (including making any payments to any rabbi trust or taking any action that would cause the trustee of any rabbi trust to make payments to any current or former director, officer or employee of the Debtor or its Subsidiaries), except, with respect to clause (I) payment of any nondiscretionary incentive payments under existing Company Plans, nondiscretionary severance payments under existing Company Plans, and nondiscretionary payments of nonqualified deferred compensation (other than as set forth on Section 4.1(a)(xix)(I) of the Debtor Disclosure Letter) or as otherwise required by applicable Law; provided that payment in respect of any severance or nonqualified deferred compensation amount in excess of \$200,000 shall be subject to prior notice and consultation with Parent and, with respect to clauses (A) through (H) above,

- (1) amendments to welfare plans in the ordinary course of business, consistent with past practices that do not materially increase the costs of such welfare plans, (2) with respect to any hourly employees and salaried facility-level employees of the Debtor or its Subsidiaries, and any other employees of the Debtor or its Subsidiaries whose annual base salary does not exceed \$150,000, increases in compensation in the ordinary course materially consistent with the Debtor's 2018 operating budget or otherwise as reasonably determined by the CRO, in consultation with the QCP Consultants, to be necessary to respond to market demand, (3) with respect to each other employee of the Debtor or its Subsidiaries whose annual base salary exceeds \$150,000 (other than any Eligible Employee), increases in compensation in the ordinary course of business consistent with past practice that do not exceed 1.5% of the aggregate annual base salaries of such other employees or 7.5% of the annual base salary for any individual and (4) as required pursuant to existing Company Plans, or as otherwise required by applicable Law;
- (xx) become a party to, establish, adopt, amend, commence participation in or terminate any collective bargaining agreement or other agreement with a labor union, works council or similar organization;
- (xxi) enter into any Contract adversely affecting in any material respect the Debtor's or any of its Subsidiaries' ability to use or otherwise exploit any material Intellectual Property Rights;
- (xxii) fail to use commercially reasonable efforts to keep in full force the material Insurance Policies under substantially the same levels of coverage as the current policies of the Debtor and its Subsidiaries;
- (xxiii) change in any material respect any of the Debtor's or its Subsidiaries' material policies or procedures for or timing of the collection of accounts receivable (or any other trade receivables), payment of accounts payable (or any other trade payables), billing of its customers, pricing and payment terms, cash collections, cash payments or terms with suppliers, in each case, other than changes required by suppliers, vendors and service providers;
- (xxiv) dismiss the QCP Consultants other than in accordance with Section 5.1(b);
- (xxv) modify or amend in any respect any Contract pursuant to which HCR III or any of its Subsidiaries currently subleases real property to any other Subsidiary of the Debtor; or
 - (xxvi) agree, authorize or commit to do any of the foregoing.
- (b) Parent shall not knowingly take or permit any of its Subsidiaries to take any action that is reasonably likely to prevent or materially impede the consummation of the Transactions.
 - 4.2 Filings; Other Actions; Notification.

- Cooperation. Subject to the terms and conditions set forth in this (a) Agreement, the Debtor and Parent shall cooperate with each other and use (and shall cause their respective Subsidiaries to use) their respective reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on their part under this Agreement and applicable Laws to consummate and make effective the Transactions as soon as practicable, including preparing and filing as promptly as practicable all documentation to effect all necessary notices, reports and other filings and to obtain as promptly as practicable all consents, registrations, approvals, permits and authorizations necessary or advisable to be obtained from any third party and/or any Governmental Entity (including the Governmental Approvals). Subject to applicable Laws relating to the exchange of information, the parties shall cooperate on all matters with respect to obtaining the Governmental Approvals and making of filings with any Governmental Entity in connection with this Section 4.2. Each of Parent and the Debtor shall (i) in coordination with each other, file as promptly as practicable after the execution hereof, all applications and documents required, necessary or advisable to obtain the Governmental Approvals as soon as practicable, (ii) submit to customary background checks in connection therewith, (iii) respond promptly to any additional requests from Governmental Entities and (iv) otherwise use its reasonable best efforts to obtain all Governmental Approvals as promptly as practicable (and in any event prior to the Termination Date). Parent and the Debtor shall have the right to review in advance and, to the extent practicable, each will consult with the other on and consider in good faith the views of the other in connection with, all of the information relating to Parent and the Debtor, as the case may be, and any of their respective Subsidiaries, that appears in any filing made with, or written materials submitted to, any third party and/or any Governmental Entity in connection with the Transactions. In exercising the foregoing rights, each of the Debtor and Parent shall act reasonably and as promptly as practicable.
- (b) <u>Information; Meetings</u>. The Debtor and Parent each shall, upon the reasonable request by the other, furnish the other with all information concerning itself, its Subsidiaries, directors, officers and stockholders and such other matters as may be reasonably necessary or advisable in connection with any statement, filing, notice or application made by or on behalf of Parent or the Debtor or any of their respective Subsidiaries to any third party and/or any Governmental Entity in connection with the Transactions.
- Entity, the Debtor and Parent each shall keep the other reasonably apprised of the status of matters relating to completion of the Transactions to the extent such party is aware of such information, including promptly furnishing the other with copies of notices or other communications received by Parent or the Debtor or any of their respective Subsidiaries, as the case may be, from any third party and/or any Governmental Entity with respect to the Transactions. Parent shall give prompt written notice to the Debtor of any change, fact or condition of which Parent becomes aware that is reasonably expected to result in any failure of any condition to Debtor's obligations to effect the Transactions, and the Debtor shall give prompt written notice to Parent of any change, fact or condition of which the Debtor becomes aware that is reasonably expected to result in any failure of any condition to Parent's or the Purchaser's obligations to effect the Transactions or to materially affect the Business. Neither the Debtor nor Parent shall permit any of their respective officers or any other representatives or agents to participate in any meeting with any Governmental Entity in respect of any filings,

investigation or other inquiry relating to the Transactions unless it consults with the other party in advance and, to the extent permitted by such Governmental Entity, gives the other parties the opportunity to attend and participate thereat.

- 4.3 Access and Records. Subject to applicable Law, upon reasonable notice, the Debtor shall (and shall cause its Subsidiaries to) afford Parent's officers, directors, employees, agents, counsel, accountants, investment bankers, financing sources and other authorized representatives (together, "Representatives") reasonable access, during normal business hours throughout the period prior to the Closing Date, to its employees, properties, books, contracts and records, and, upon Parent's written request, will use reasonable efforts to provide Parent with access to suppliers, creditors and third parties with whom the Debtor or its Subsidiaries does business or by whom there have been asserted any claim, and, during such period, the Debtor shall (and shall cause its Subsidiaries to) use its reasonable best efforts to furnish promptly to Parent all information concerning its business, properties, personnel, suppliers, creditors and third parties with whom the Debtor or its Subsidiaries does business as may reasonably be requested, provided that no investigation pursuant to this Section 4.3 shall affect or be deemed to modify any representation or warranty made by the Debtor herein, and provided, further, that the foregoing shall not require the Debtor (i) to permit any inspection, or to disclose any information, that in the reasonable judgment of the Debtor would result in the disclosure of any trade secrets of third parties or violate any of its obligations with respect to confidentiality if the Debtor shall have used its reasonable best efforts to obtain the consent of such third party to such inspection or disclosure or (ii) to disclose any privileged information of the Debtor or any of its Subsidiaries. The Debtor shall continue to deliver its financial information and other reports required pursuant to the terms of the Master Lease. The Debtor shall cooperate to provide Parent with access and information reasonably requested by Parent to facilitate the consolidation of the Debtor's financial results with and into those of Parent following Closing, including establishment by Parent of accounting and financial reporting systems, systems of internal controls over financial reporting and disclosure controls as defined in the Exchange Act. To the extent not otherwise provided in this Agreement, the Debtor shall, upon Parent's request in writing, use its reasonable best efforts to provide to Parent those financial statements and information as may be reasonably required to comply with Parent's obligations under the Exchange Act.
- 4.4 <u>Publicity</u>. The initial press release regarding the Transactions shall be a joint press release issued by the Debtor and Parent. Thereafter, no party shall issue any press release or make any public announcement concerning the Transactions without the prior written consent of the other parties hereto except as may be required by Law, including rules promulgated by the Securities and Exchange Commission (the "<u>SEC</u>") or by obligations pursuant to any rules of any national securities exchange. To the extent any party is required to make any press release or public announcement by Law or by obligations pursuant to any rules of any national securities exchange, such party shall (i) to the extent practicable, consult with the other parties hereto prior to issuing any press releases or otherwise making public announcements with respect to the Transactions and prior to making any filings with any third party and/or any Governmental Entity with respect thereto, except as may be required by Law or by obligations pursuant to any rules of any national securities exchange, (ii) to the extent practicable, provide the other parties hereto with a reasonable opportunity to review and

comment on such press release, announcement or filing and (iii) consider in good faith any comments of the other parties with respect to such press release, announcement or filing.

4.5 <u>Employee Matters</u>.

- (a) Prior to making any broad-based written or oral communications to officers and employees of the Debtor or any of its Subsidiaries pertaining to compensation or benefit matters that are affected by the Transactions, the Debtor shall provide Parent with a copy of the intended communication, Parent shall have a reasonable period of time to review and comment on the communication, and the Debtor shall consider any such comments in good faith.
- (b) Parent shall use all reasonable endeavors to procure that any employee benefit plans which the employees of the Debtor and its Subsidiaries are entitled to participate in to take into account for purposes of eligibility, vesting and benefit accrual thereunder (other than any benefit accrual under any defined benefit pension plans), service by employees of the Debtor and its Subsidiaries as if such service were with Parent, to the same extent such service was credited under a comparable plan of the Debtor or any of its Subsidiaries (except to the extent it would result in a duplication of benefits).
- (c) Nothing contained in this Agreement is intended to: (i) be treated as an amendment of any particular Company Plan; (ii) prevent Parent, the Debtor or any of their Affiliates from amending or terminating any of their benefit plans (or, after the Closing Date, any Company Plan) in accordance with their terms; (iii) prevent Parent, the Debtor or any of its Affiliates, after the Closing Date, from terminating the employment of any employee of the Debtor; or (iv) create any third-party beneficiary rights in any employee of the Debtor, any beneficiary or dependent thereof, or any collective bargaining representative thereof, with respect to the compensation, terms and conditions of employment and/or benefits that may be provided to any such employee by Parent, the Debtor or any of their Affiliates or under any benefit plan which Parent, the Debtor or any of their Affiliates may maintain.
- (d) <u>Settlement Agreement</u>. If the Former CEO Settlement Agreement is executed and delivered by the parties thereto, from the date of execution and delivery through the Closing Date, the Debtor shall comply with the terms of the Former CEO Settlement Agreement and shall not amend, modify or alter such Former CEO Settlement Agreement without the prior written consent of Parent.

(e) <u>280G Vote</u>.

- (i) At least two (2) business days prior to the Closing Date, the Debtor shall submit for approval by its stockholders, in conformance with Section 280G of the Code and the regulations thereunder (the "280G Stockholder Vote"), any payments that would reasonably be expected to constitute a "parachute payment" pursuant to Section 280G of the Code (each, a "Parachute Payment").
- (ii) The Debtor shall use its commercially reasonably efforts to obtain, at least three (3) business days prior to the Closing Date, irrevocable waivers of the right to any Parachute Payment from each of the applicable "disqualified individuals" (as defined under Section 280G of the Code and the regulations promulgated thereunder).

(iii) True and complete copies of all disclosure and documents that comprise the stockholder approval of each Parachute Payment shall be delivered to Parent in sufficient time to allow Parent to comment thereon, but no less than five (5) business days prior to the 280G Stockholder Vote, and shall reflect all reasonable comments of Parent thereon.

4.6 Expenses.

- (a) Whether or not the Transactions are consummated, the parties hereto shall bear their own respective expenses (including all compensation and expenses of counsel, financial advisors, consultants, actuaries and independent accountants) incurred in connection with this Agreement and the Transactions.
- (b) The Debtor and its Subsidiaries' sole financial advisor shall be Moelis & Company, and the Debtor shall not, and shall not permit its Subsidiaries to, hire any additional financial advisor. The Debtor shall not agree to any change in the financial terms of any agreement pursuant to which Moelis & Company is entitled to any fees and expenses in connection with any of the Transactions.

4.7 Indemnification; Directors' and Officers' Insurance.

- From and after the Closing Date, each of Parent and the Debtor agrees that (a) it will indemnify, defend and hold harmless each present and former director and officer of the Debtor and its Subsidiaries (in each case, when acting in such capacity) (the "Indemnified Parties") from and against any and all costs or expenses (including reasonable attorneys' fees), judgments, fines, losses, claims, damages or liabilities (whether or not incurred in connection with any Proceeding or investigation, whether civil, criminal, administrative or investigative) arising out of or related to such Indemnified Party's service or status as an officer or director of the Debtor or one of its Subsidiaries or any action or inaction on the part of any such Person in such capacity, in each case, as of prior to the Closing, whether asserted or claimed prior to, at or after the Closing Date, to the fullest extent such Indemnified Party would be permitted to be indemnified by the Debtor or its applicable Subsidiary under applicable Law and the certificate of incorporation and by-laws (or comparable organizational documents) of the Debtor or its applicable Subsidiary in effect on the date of this Agreement; provided that the Person to whom expenses are advanced provides an undertaking to repay such advances if it is ultimately determined that such Person is not entitled to indemnification.
- (b) Prior to or as of the Closing Date, the Debtor shall (after reasonable consultation with Parent) and, if the Debtor is unable to, Parent shall cause the Debtor as of the Closing Date to obtain and fully pay the premium for "tail" insurance policies for the extension of (i) the directors' and officers' liability coverage of the existing directors' and officers' insurance policies of the Debtor and its Subsidiaries, and (ii) the existing fiduciary liability insurance policies of the Debtor and its Subsidiaries, in each case for a claims reporting or discovery period of at least six (6) years from and after the Closing Date (the "Tail Period") from an insurance carrier with the same or better credit rating as the Debtor's insurance carrier as of the date hereof with respect to directors' and officers' liability insurance and fiduciary liability insurance (collectively, "D&O Insurance") with terms, conditions, retentions and limits of liability that are at least as favorable to the insureds as the Debtor's existing policies with respect

to any matter claimed against a director or officer of the Debtor or any of its Subsidiaries by reason of him or her serving in such capacity that existed or occurred at or prior to the Closing Date (including in connection with this Agreement or the transactions or actions contemplated hereby); provided that in no event shall the aggregate cost of the D&O Insurance exceed during the Tail Period 300% of the current aggregate annual premiums paid by the Debtor for such insurance. If the Debtor for any reason fails to obtain such insurance policies as of the Closing Date, the Debtor shall, and Parent shall cause the Debtor to, continue to maintain in effect for the Tail Period the D&O Insurance in place as of the date hereof with terms, conditions, retentions and limits of liability that are at least as favorable to the insureds as provided in the Debtor's existing policies as of the date hereof, or the Debtor shall, and Parent shall cause the Debtor to, use reasonable best efforts to purchase comparable D&O Insurance for the Tail Period with terms, conditions, retentions and limits of liability that are at least as favorable to the insureds as provided in the Debtor's existing policies as of the date hereof; provided, however, that in no event shall the annual cost of the D&O Insurance exceed during the Tail Period 150% of the current aggregate annual premiums paid by the Debtor for such insurance; and provided, further, that if the annual premiums of such insurance coverage exceed such amount, the Debtor shall obtain a policy with the greatest coverage available for a cost not exceeding such amount. From and after the Closing Date, except as required by applicable Law, Parent shall cause the certificate of incorporation and by-laws (or comparable organizational documents) of each of the Debtor and its Subsidiaries to contain provisions no less favorable to the Indemnified Parties with respect to limitation of liabilities, advancement of expenses and indemnification than are set forth as of the date of this Agreement in each of such documents of the Debtor and its Subsidiaries. In the event of any Proceeding or investigation for which an Indemnified Party is entitled to indemnification pursuant to this Section 4.7, (A) Debtor shall, and Parent shall cause the Debtor and its Subsidiaries to, defend the Indemnified Party with respect to any such Proceeding or investigation and (B) Parent shall not, and shall cause the Debtor and its Subsidiaries not to, settle, compromise or consent to the entry of any judgment in any Proceeding or investigation pending or threatened in writing to which an Indemnified Party is a party (and in respect of which indemnification could be sought by such Indemnified Party hereunder) without the prior written consent of such Indemnified Party (such consent not to be unreasonably withheld, conditioned or delayed), unless such settlement, compromise or consent includes an unconditional release of such Indemnified Party from all liability arising out of such Proceeding or investigation. Parent agrees that all rights of the Indemnified Parties to indemnification and exculpation from liabilities for acts or omissions occurring prior to the Closing as provided in the certificate of incorporation or by-laws (or comparable organizational documents) of the Debtor or any of its Subsidiaries and any indemnification agreements entered into prior to the date hereof, including provisions relating to advancement of expenses incurred in the defense of any Proceeding or investigation, shall survive the Closing Date and shall continue in full force and effect in accordance with their terms. Such rights shall not be amended or otherwise modified in any manner that would adversely affect the rights of any of the Indemnified Parties for any acts or omissions occurring prior to the Closing without the prior written consent of each affected Indemnified Party (such consent not to be unreasonably withheld, conditioned or delayed).

(c) Any Indemnified Party wishing to claim indemnification under paragraph (a) of this Section 4.7, upon learning of any such Proceeding or investigation, shall promptly notify Parent and the Debtor thereof, but the failure to so notify shall not relieve Parent

or the Debtor of any Liability it may have to such Indemnified Party except to the extent such failure materially prejudices the indemnifying party.

- (d) If Parent or the Debtor or any of their respective successors or assigns (i) shall consolidate with or merge into any other corporation or entity and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) shall transfer all or substantially all of its properties and assets to any individual, corporation or other entity, then, and in each such case, proper provisions shall be made so that the successors and assigns of Parent or the Debtor shall assume all of the obligations set forth in this Section 4.7.
- (e) The provisions of this Section 4.7 are intended to be for the benefit of, and shall be enforceable by, each of the Indemnified Parties.
- (f) The rights of the Indemnified Parties under this Section 4.7 shall be in addition to any rights such Indemnified Parties may have under the certificate of incorporation or by-laws (or comparable organizational documents) of the Debtor or any of its Subsidiaries, or under any applicable Contracts or Laws. All rights to indemnification and exculpation from liabilities for acts or omissions occurring at or prior to the Closing Date and rights to advancement of expenses relating thereto now existing in favor of any Indemnified Party as provided in the certificate of incorporation, by-laws or comparable governing documents of the Debtor and its Subsidiaries or any indemnification agreement between such Indemnified Party and the Debtor or any of its Subsidiaries shall survive the Transactions and shall not be amended, repealed or otherwise modified in any manner that would adversely affect any right thereunder of any such Indemnified Party.
- Each of Parent and the Debtor hereby acknowledges that the Indemnified Parties have or may, in the future, have certain rights to indemnification, advancement of expenses and/or insurance provided by Persons other than Parent, the Debtor or its Subsidiaries (collectively, the "Other Indemnitors" and, individually, an "Other Indemnitor") with respect to Proceedings or investigations that are the subject of Section 4.7(a). The Debtor hereby agrees that if any advancement or indemnification obligation is owed, at any time, to an Indemnified Party for the same Proceeding or investigation by both (i) Parent or the Debtor, including under Section 4.7(a), and (ii) any Other Indemnitor (whether pursuant to any certificate of incorporation, by-laws or comparable organizational documents, indemnification agreement or other agreements), the Debtor shall (and if it is unable or fails to honor its obligations under this Section 4.7, Parent shall) be primarily liable for indemnification and advancement of expenses to such Indemnified Parties in respect of such Proceeding or investigation, and any obligation of an Other Indemnitor to provide indemnification or advancement of expenses shall be secondary to the obligations of the Debtor (or if it is unable or fails to honor its obligations under this Section 4.7, Parent) under this Section 4.7. Parent and the Debtor irrevocably waive, relinquish and release the Other Indemnitors from any and all claims (A) against the Other Indemnitors for contribution, subrogation, indemnification or any other recovery of any kind in respect of the indemnification arrangements provided to such Indemnified Parties by such Other Indemnitors and (B) that the Indemnified Party must seek expense advancement or reimbursement, or indemnification, from any Other Indemnitor before Parent and the Debtor must perform their expense advancement and reimbursement, and indemnification obligations, under this Agreement. If any Other Indemnitor pays or causes to be paid, for any reason, any amounts

otherwise indemnifiable or subject to advancement under this Section 4.7, then (1) such Other Indemnitor shall be fully subrogated to all rights of the Indemnified Parties with respect to the payments actually made and (2) the Debtor shall reimburse the Other Indemnitor for the payments actually made. Notwithstanding anything to the contrary in this Agreement, for the avoidance of doubt, Parent's rights against the Debtor for contribution, subrogation, indemnification or any other recovery of any kind in respect of the indemnification arrangements provided by Parent and Debtor to the Indemnified Parties pursuant to this Section 4.7 are hereby preserved.

4.8 Confidentiality. Unless and until Closing occurs, no party hereto shall disclose any information provided to such party or its Representatives pursuant to Section 4.3 or any other provision of this Agreement, or any non-public information related to the negotiation, execution and delivery of this Agreement and the Transactions, other than (a) disclosure to such party's Representatives (provided that such Representatives agree to comply with this Section 4.8 as if they were a party hereto and that such party shall be responsible for any Representative of such party to so comply with this Section 4.8), (b) as required by Law or by obligations pursuant to any rules of any national securities exchange, the SEC or by the request of any Governmental Entity, and (c) subject to reasonable customary confidentiality arrangements reasonably acceptable to the Debtor, any Person who expresses a bona fide interest in acquiring all or part of the Business or any business entities, properties or assets of the Debtors and its Subsidiaries following the Closing, including any potential joint venture partner (a "Potential Acquiror") and actual or potential purchasers, replacement tenants, third-party operators or other partners of Parent with respect to the remarketing of the Leased Facilities pursuant to the SNF/AL Remarketing Process. Notwithstanding the foregoing, nothing herein shall in any way limit any existing rights of any party to the Master Lease to use or disclose information under the terms of the Master Lease.

4.9 Financing.

(a) The Debtor shall, and shall cause its Subsidiaries to, use reasonable best efforts to provide and direct its and their respective Representatives to use their respective reasonable best efforts to provide all cooperation reasonably requested by Parent in connection with the arrangement, syndication and consummation of any debt financing by Parent or its Subsidiaries and the Debtor or any of its Subsidiaries, any equity financing of Parent or any minority investment in the Debtor or any of its Subsidiaries (not to exceed ten percent (10%) of the fully-diluted equity of the Debtor or such Subsidiary) including (i) the refinancing of the indebtedness under the Centerbridge Facility, whether at the Debtor, Parent or any of their Subsidiaries (such financing, the "Manorcare Financing"), (ii) the refinancing of the indebtedness under the First Lien Credit and Guaranty Agreement, dated as of October 31, 2016, by and among QCP SNF West REIT, LLC, QCP SNF Central REIT, LLC, QCP SNF East REIT, LLC, QCP AL REIT, LLC, QCP Holdco REIT, LLC, Parent, certain Subsidiaries of Parent from time to time party to the credit agreement as guarantors, the lenders party thereto and Barclays Bank PLC, as Administrative Agent and Swing Line Lender, and the L/C Issuers at Parent or any of its Subsidiaries and (iii) the refinancing of the \$750 million senior secured notes due 2023 issued pursuant to that certain Indenture, dated as of October 17, 2016, among QCP SNF West REIT, LLC, QCP SNF Central REIT, LLC, QCP SNF East REIT, LLC and QCP AL REIT, LLC, as issuers, and Wilmington Trust, National Association, as trustee and notes collateral

agent (such financing, together with the Manorcare Financing, the "Financing"). Such cooperation shall include (i) furnishing Parent and its Debt Financing Sources with the Audited Financial Statements and with audited financial statements for the Debtor and its Subsidiaries for each subsequent fiscal year within fifty (50) days after the end of such fiscal year, and quarterly and interim unaudited financial statements for the Debtor and its Subsidiaries for the fiscal quarters and interim periods ended March 31, 2018, June 30, 2018 and September 30, 2018 and for each subsequent fiscal quarter within thirty-five (35) days after the end of such fiscal quarter, in each case, with comparative financial information for the equivalent period of the prior year; (ii) using reasonable best efforts to furnish Parent and its Debt Financing Sources with information, audit reports, historical business and other financial data and any supplements thereto regarding the Debtor and its Subsidiaries customarily included in information memoranda and other syndication materials for revolving, term loan credit facilities, bond offering documents and equity offering documents, including assisting in preparing pro forma financial information for the Debtor and its Subsidiaries; (iii) to the extent reasonably requested by Parent, using reasonable best efforts to prepare carve-out audited and unaudited financial statements for the time period required or desirable to any Debt Financing Source; (iv) using reasonable best efforts to cause members of management and other senior officers to participate in a reasonable number of meetings (including one-on-one meetings or conference calls), lender presentations, due diligence sessions and sessions with rating agencies, prospective lenders and investors and other syndication and marketing activities; (v) reasonably assisting Parent and its Debt Financing Sources in the preparation of any syndication and offering documents and materials, including information memoranda, lender presentations, offering memoranda, registration statements, prospectuses and other marketing documents (collectively, the "Marketing Documentation") and provide and execute a customary authorization letter with respect thereto; (vi) reasonably cooperating in marketing efforts of Parent; (vii) reasonably assisting in the negotiation and preparation of any credit agreements, indentures, underwriting agreements, purchase agreements, pledge and security documents, mortgages, guarantees, hedging agreements and other definitive documents, a certificate of the chief financial officer of the Debtor with respect to solvency matters in a form acceptable to Parent and any other certificates, letters and documents (including any schedules and exhibits in connection with the foregoing) as may be reasonably requested by Parent; (viii) reasonably assisting in the obtaining of (A) audit reports, authorization letters, comfort letters and consents of accountants and auditors with respect to financial statements for the Debtor and its Subsidiaries for inclusion in any Marketing Documentation and (B) payoff letters, instruments of discharge and Lien terminations; (ix) providing information regarding the Debtor and its Subsidiaries required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including, without limitation, the USA Patriot Act of 2001, to the extent requested by its Debt Financing Sources at least eight (8) business days prior to the Closing; (x) reasonably assisting Parent in obtaining corporate, facility and debt security ratings from rating agencies; (xi) reasonably cooperating with Parent's legal counsel (including providing customary back-up certificates) in connection with any legal opinions that such legal counsel may be required to deliver in connection with the Financing; (xii) taking such reasonable actions as may be reasonably requested by Parent necessary to permit its Debt Financing Sources to evaluate the Debtor's assets and cash management policies and procedures relating thereto for the purposes of establishing collateral arrangements as of the Closing and to cooperate with other due diligence conducted by its Debt Financing Sources; (xiii) using reasonable best efforts to

obtain customary evidence of authority, customary officer's certificates, good standing certificates (to the extent applicable) in the respective jurisdictions of organization of the Debtor and its Subsidiaries, customary lien searches with respect to the Debtor and its Subsidiaries and insurance certificates; (xiv) using reasonable best efforts to facilitate the granting of a security interest (or perfection thereof) in collateral by the Debtor and its Subsidiaries to secure the Financing at Closing; and (xv) if applicable, reasonably cooperating with Parent in connection with the payoff of the Centerbridge Facility at Closing.

- (b) The Debtor hereby consents to the use of its and its Subsidiaries' logos in connection with the Financing; <u>provided</u> that such logos are used solely in a manner that is not intended to, nor reasonably likely to, harm or disparage the Debtor or its Subsidiaries.
- (c) For purposes of this Agreement, "<u>Debt Financing Sources</u>" means the Persons that will provide or otherwise enter into agreements in connection with the Financing, together with, in each case, their Affiliates and Representatives.
- Notwithstanding anything in this Agreement to the contrary, none of the (d) Debtor, its Subsidiaries or their respective Affiliates, or Representatives shall (i) be required to pay any commitment or other similar fee other than any such fee that may be payable by the Debtor or its Subsidiaries at the Closing, (ii) be required to incur any other liability in connection with the Financing other than any such liabilities to be incurred by the Debtor or its Subsidiaries arising after the Closing occurs or liabilities for costs and expenses to the extent required to be indemnified or reimbursed by Parent, (iii) otherwise be required to take any action in violation of any of the Debtor's or any of its Subsidiaries' organizational documents or that Debtor determines in good faith could reasonably be expected to violate any applicable Law or Contract, (iv) be required to waive or amend any terms of this Agreement or terms of any other Contract to which the Debtor or its Subsidiaries is a party, (v) be required to provide cooperation to the extent it would (y) cause any condition to the Closing set forth in ARTICLE VI not to be satisfied or (z) cause a breach of this Agreement, (vi) be required to deliver or obtain opinions of internal or external counsel or (vii) provide access to or disclose information where the Debtor reasonably determines that such access or disclosure could jeopardize the attorney-client privilege or contravene any Law or Contract to which it or any of its Subsidiaries is a party, provided that in the event that the Debtor or any of its Subsidiaries does not provide information in reliance on the exclusions in this subclause (vii), the Debtor and its Subsidiaries shall use commercially reasonable efforts to provide notice to Parent promptly upon obtaining knowledge that such information is being withheld (but solely if providing such notice would not violate such obligation of confidentiality). No director or officer of the Debtor or its Subsidiaries, acting in such capacity, shall be required to execute, deliver or enter into or perform any agreement, document or instrument, including any definitive documentation with respect to the Financing, or adopt any resolutions approving the agreements, documents and instruments pursuant to which the Financing is obtained, except with respect to the delivery of customary representation and authorization letters in connection with the Marketing Documentation. Nothing hereunder will require any officer or representative of the Debtor or any of its Subsidiaries to deliver any certificate or opinion or take any other action that would result in personal liability to such officer or representative. Parent shall indemnify, defend and hold harmless the Debtor and its Subsidiaries, and their respective officers, employees and Representatives, from and against any and all liabilities or losses suffered or incurred by them in connection with the arrangement of

the Financing and any information utilized in connection therewith, except in the event such liabilities or losses arose out of or result from (A) the willful misconduct, gross negligence or bad faith of the Debtor and its subsidiaries or (B) historical information furnished in writing by or on behalf of the Debtor and its Subsidiaries, including financial statements. Notwithstanding anything in this paragraph, the Debtor and its Subsidiaries shall not be required to provide any such requested cooperation that shall unreasonably disrupt or interfere with the ongoing operations of the Debtor or its Subsidiaries. Parent shall, promptly upon request by the Debtor, reimburse the Debtor and its Subsidiaries for all reasonable and documented out-of-pocket costs and expenses incurred by the Debtor and its Subsidiaries (including those of its Affiliates and Representatives) in connection with taking actions requested by Parent pursuant to this Section 4.9, which for the avoidance of doubt will not include audit fees and other expenses relating to the preparation of financial statements for the Business to the extent required by the Master Lease or otherwise historically prepared by the Debtors and its Subsidiaries. Notwithstanding anything to the contrary, the condition set forth in Section 6.2(b) as it applies to the Debtor's obligations under this Section 4.9 shall be deemed satisfied unless there has occurred a breach by Debtor of its obligations under this Section 4.9 that directly results in the Financing not being available on the Closing Date.

4.10 <u>Acquisition Proposals</u>.

- (a) <u>No Solicitation or Negotiation</u>. At all times during the period commencing with the execution and delivery of this Agreement and continuing until the earlier to occur of the termination of this Agreement pursuant to ARTICLE VII and the Closing, except as expressly permitted by this Section 4.10, the Debtor and its Subsidiaries and its and its Subsidiaries' respective directors, officers and employees shall not, and the Debtor shall direct its other agents and representatives not to, directly or indirectly:
- (i) initiate, solicit, propose or knowingly encourage or otherwise knowingly facilitate any inquiry or the making of any proposal or offer that constitutes or, would reasonably be expected to lead to, an Acquisition Proposal;
- (ii) engage in, continue or otherwise participate in any discussions or negotiations relating to any Acquisition Proposal or any inquiry, proposal or offer that would reasonably be expected to lead to an Acquisition Proposal (other than to state that the terms of this Section 4.10 prohibit such discussions);
- (iii) provide any non-public information or data concerning the Debtor or its Subsidiaries, or access to the Debtor or its Subsidiaries' properties, books and records to any Person, in each case, in connection with any Acquisition Proposal or any inquiry, proposal or offer that constitutes or would reasonably be expected to lead to an Acquisition Proposal;
- (iv) otherwise knowingly facilitate any effort or attempt to make an Acquisition Proposal;
- (v) approve or recommend or publicly declare advisable any Acquisition Proposal or proposal reasonably expected to lead to an Acquisition Proposal or

approve or recommend, or publicly declare advisable or publicly propose to enter into, any Alternative Acquisition Agreement;

- (vi) execute or enter into an Alternative Acquisition Agreement; or
- (vii) agree, authorize or commit to do any of the foregoing.
- Notice of Acquisition Proposals. The Debtor shall promptly (but, in any (b) event, within twenty-four hours) give notice to Parent if (i) any bona fide inquiries, proposals or offers with respect to an Acquisition Proposal or that would reasonably be expected to lead to an Acquisition Proposal are received by the Debtor or any of its Subsidiaries, or any of its or their respective Representatives of which the Debtor or its Subsidiaries are made aware (it being understood that the Debtor and its Subsidiaries shall instruct its and their respective Representatives to promptly make them aware of any such bona fide inquiries, proposals or offers), or (ii) any bona fide request is received by the Debtor or any of its Subsidiaries, or any of its or their respective Representatives of which the Debtor or its Subsidiaries are made aware (it being understood that the Debtor and its Subsidiaries shall instruct its and their respective Representatives to promptly make them aware of any such bona fide requests), for the Debtor or any of its Subsidiaries to engage in discussions or negotiations relating to an Acquisition Proposal or that would reasonably be expected to lead to an Acquisition Proposal, setting forth in such notice the name of the Person making any such inquiry, proposal, offer or request and the material terms and conditions of any such Acquisition Proposal or inquiry, proposal or offer and the scope of any such request (including, if applicable, correct and complete copies of any such written Acquisition Proposals, inquiries, proposals or offers, including proposed agreements, or requests (or where no such copies are available, a reasonably detailed written description thereof)), and thereafter shall keep Parent reasonably informed, on a current basis of the status and terms and conditions of any such Acquisition Proposals, inquiries, proposals or offers or requests (including any amendments or supplements thereto) and the status of any such discussions or negotiations, including any change in such Person's intentions as previously notified.
- (c) Existing Discussions. The Debtor (i) agrees, that as of the date hereof, it has ceased and caused to be terminated any activities, solicitations, discussions and negotiations with any Person (other than Parent and its Subsidiaries) conducted prior to the date hereof with respect to any Acquisition Proposal, or proposal that would reasonably be expected to lead to an Acquisition Proposal (including access to any physical or electronic data rooms) and (ii) shall promptly (but in any event within twenty-four hours of the execution and delivery of this Agreement) (A) deliver a written notice to each such Person providing only that the Debtor is ending all discussions and negotiations with such Person with respect to any Acquisition Proposal, or proposal or transaction that would reasonably be expected to lead to an Acquisition Proposal and is requesting the prompt return or destruction of all confidential information concerning the Debtor and any of its Subsidiaries and (B) if applicable, terminate any physical and electronic data or other diligence access previously granted to such Persons.
- (d) <u>Standstill Provisions</u>. From that date of this Agreement and continuing until the earlier to occur of the termination of this Agreement pursuant to ARTICLE VII and the Closing, the Debtor shall not terminate, amend or otherwise modify or waive any provision of

any confidentiality, "standstill" or similar agreement to which the Debtor or any of its Subsidiaries is a party and, in the event of any breach of such agreement of which the Debtor Board becomes aware, the Debtor shall use reasonable best efforts to enforce, to the fullest extent permitted under applicable Law, the provisions of any such agreement, including by obtaining injunctions to prevent any breaches of such agreements and to enforce specifically the terms and provisions thereof.

(e) For purposes of this Agreement, the term "Acquisition Proposal" means any (i) proposal, offer, inquiry or indication of interest relating to a merger, joint venture, partnership, consolidation, dissolution, liquidation, tender offer, spin-off, share exchange, business combination or similar transaction involving the Debtor or any of its Subsidiaries or (ii) acquisition by any Person or group (as defined under Section 13 of the Exchange Act), resulting in, or any proposal, offer, inquiry or indication of interest that if consummated would result in, any Person or group (as defined under Section 13 of the Exchange Act) becoming the beneficial owner of, directly or indirectly, in one or a series of related transactions, ten percent (10%) or more of the consolidated net revenues, net income or total assets (it being understood that total assets include equity securities of Subsidiaries) of the Debtor, other than the Transactions or the disposition of those Leased Facilities set forth in Section 5.3(b) of the Debtor Disclosure Letter; and the term "Alternative Acquisition Agreement" means any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement, option agreement, joint venture agreement, partnership agreement or other similar agreement relating to any Acquisition Proposal.

ARTICLE V

Special Covenants

5.1 Corporate Governance.

(a) <u>Debtor Restructuring Committee</u>. After the date of this Agreement and prior to the Closing Date, the power and authority of the special restructuring committee of the Debtor Board (the "<u>Debtor Restructuring Committee</u>") shall not be changed, and no new or replacement member of the Debtor Restructuring Committee, the sole members of which are Sherman Edmiston III and Kevin P. Collins in their capacity as members of the Debtor Board, shall be appointed without the consent of Parent, such consent not to be unreasonably withheld, conditioned or delayed.

(b) <u>QCP Consultants</u>.

(i) From and after the date of this Agreement, Guy Sansone and Laura Linynsky shall serve as consultants (the "QCP Consultants"). In such capacity, the QCP Consultants shall report to Parent and shall have the following rights and access: (i) email addresses with the Debtor's domain name and full access to the Debtor's corporate email network, (ii) full view-only access to all financial information of the Debtor and its Subsidiaries available to the current Chief Executive Officer ("CEO") and the current Chief Financial Officer ("CFO") of the Debtor, (iii) announcement of the QCP Consultants internally and externally as the future management team, including in the Plan and Disclosure Statement, (iv) access to any

written reports provided to the current CEO and CFO, (v) opportunity to review and comment on all external and broadly-distributed internal communications by the Debtor's management, (vi) attendance at any and all telephonic and in-person meetings of the Debtor Board (other than executive sessions of the Debtor Board and other than the portion of any meeting from which employee members of the Debtor Board are excluded or involving, in the good faith judgment of the Debtor Board, any actual or potential conflict of interest with the Purchaser Entities or as may be reasonably necessary to preserve attorney client privilege, in the good faith judgment of the Debtor Board), (vii) bi-weekly calls with independent directors of the Debtor to address access or other concerns; provided that the CRO shall also participate in such calls, and (viii) direct access to all employees of the Debtor and its Subsidiaries, subject to coordination with, and reasonable approval of, the CRO. If either of the QCP Consultants is no longer able to perform the functions of such position due to death or disability or such individual resigns or is convicted of a felony, he or she shall be discharged of his or her duties by Parent, and a replacement QCP Consultant shall be appointed by Parent, subject to reasonable approval of such person by the Debtor Board and the CRO. If (i) the Debtor Board (A) determines that either of the QCP Consultants willfully fails to materially comply with any restrictions on the authority and activities of the QCP Consultants set forth in Section 5.1(b)(iv) (the restriction not so complied with being referred to as the "Applicable Restriction")) and (B) notifies Parent in writing of such determination, and (ii) such QCP Consultant thereafter again willfully fails to materially comply with such Applicable Restriction, such QCP Consultant may be terminated as a QCP Consultant by the Debtor and denied the rights and access described in this Section 5.1(b) with the approval of the Debtor Board and the CRO, in which case a replacement QCP Consultant shall be appointed by Parent, subject to reasonable approval of such person by the Debtor Board and the CRO. For the avoidance of doubt, each of the QCP Consultants shall be deemed to be a Representative of Parent for purposes of Section 4.8 and Parent shall be responsible for any failure of either QCP Consultant to comply with Section 4.8 as if such QCP Consultant were a party thereto.

- (ii) Subject to the other terms of this Section 5.1(b), the Debtor shall promptly provide information to, and access to its properties and personnel to, the QCP Consultants as reasonably requested by Parent or the QCP Consultants so as to enable the QCP Consultants to accomplish the following:
- (A) prior to the Closing Date, formulating a comprehensive, detailed business plan that could begin to be implemented immediately following the Closing, including (1) the identification of potential cost reductions in the operations of the Debtor and its Subsidiaries and analysis of the operations of the remaining Leased Facilities following the completion of the SNF/AL Remarketing Process, (2) the development of a business plan for the Debtor's and its Subsidiaries' home health care and hospice businesses, (3) the development of a business plan for the Debtor's and its Subsidiaries' rehabilitation and other ancillary businesses, (4) a full analysis of the operational and sales relationships between the SNF Business and the Debtor's and its Subsidiaries' home health care and hospice businesses, and (5) an analysis of the overhead scale and regional structure of the Debtor and its Subsidiaries;
- (B) collaborating with Debtor in the remarketing of the Leased Facilities in accordance with the SNF/AL Remarketing Process (as defined in Section 5.3(b)), which remarketings may occur prior to or after the Closing Date as Parent reasonably requests;

- (C) reviewing and commenting on the rolling thirteen (13)-week budget for consolidated cash disbursements of the Debtor, the initial version of which is set forth in Section 5.1(b)(ii)(C) of the Debtor Disclosure Letter (the "13-Week Cash Budget");
- (D) monitoring the Debtor's segregation of receipts and expenditures allocable to its various Subsidiaries;
- (E) monitoring the operating and financial performance of the Debtor's businesses;
- (F) providing input into the Debtor's development of its overall business plan reflecting the transactions contemplated by Section 5.3 of this Agreement;
- (G) monitoring the status of material litigation and consulting with the Debtor generally on the resolution of such litigation;
 - (H) monitoring intercompany transfers and liabilities; and
- (I) subject to the approval of the CRO, carrying out such other customary transition-related activities as the QCP Consultants reasonably determine to be necessary or advisable in support of an orderly transition of operations at Closing.
- (iii) The QCP Consultants may (A) share information they acquire in their capacities as QCP Consultants with Parent and its Representatives, <u>provided</u> that Parent shall keep such information confidential pursuant to Section 4.8, (B) facilitate attendance by the management of Parent at meetings (not including Debtor Board meetings) with officers and employees of the Debtor, subject to approval of the CRO, and (C) share information they acquire in their capacities as QCP Consultants with any other Persons subject to appropriate confidentiality arrangements with Debtor pursuant to Section 4.8 (which, for the avoidance of doubt, shall be reasonably acceptable to the Debtor).
- The Debtor and Parent shall consult with one another and (iv) cooperate in good faith to facilitate the exercise of the rights set forth in this Section 5.1(b), which respect to which time is of the essence. For the avoidance of doubt, the operation of the Debtor and its Subsidiaries shall continue to be directed by the CRO and current CEO, at the direction of, and with oversight by, the Debtor Board, unless and until the Closing occurs. Prior to the Closing, neither QCP Consultant shall have the authority to, and neither QCP Consultant will, (A) instruct or direct, or authorize any action of, any employee or other Representative of the Debtor or any of its Subsidiaries, except as authorized by the CRO; (B) bind the Debtor and/or its Subsidiaries, commit the Debtor and/or its Subsidiaries to execute any agreements or instruments in the name of the Debtor and/or any of its Subsidiaries; (C) represent to any third party that such QCP Consultant has authority to bind the Debtor or any of its Subsidiaries prior to the Closing; (D) disburse or expend funds of the Debtor or its Subsidiaries; (E) knowingly contradict or countermand to any officer or employee of the Debtor or its Subsidiaries any directive or instruction of the CRO or current CEO or CFO or their direct reports or any written policy of the Debtor or its Subsidiaries; (F) inform any employee of the Debtor or its Subsidiaries (other than the Eligible Employees) of any determination or decision concerning such employee's continued employment by the Debtor or its Subsidiaries without the prior

approval of the CRO (which approval will not be unreasonably withheld, conditioned or delayed); (G) initiate discussions or communications with any suppliers, vendors, service providers, patients or regulators of the Debtor or its Subsidiaries without the prior approval of the CRO (which approval will not be unreasonably withheld, conditioned or delayed), except for such discussions or communications with regulators as is reasonably required to allow Parent to satisfy its obligations pursuant to Section 4.2; provided that, to the extent any QCP Consultant engages in any substantive discussion or communication with any such supplier, vendor, service provider, patient or regulator, which discussion or communication was initiated by such supplier, vendor, service provider, patient or regulator, such QCP Consultant shall provide to the CRO, within one (1) business day thereof, a reasonable description of such discussion or communication; or (H) broadly distribute written communications to the personnel of the Debtor and its Subsidiaries without the approval of the CRO (which approval will not be unreasonably withheld, conditioned or delayed).

- (c) <u>Chief Restructuring Officer</u>. After the date hereof, the CRO shall have those responsibilities set forth in Section 5.1(c) of the Debtor Disclosure Letter; <u>provided</u>, <u>however</u>, that at all times the responsibilities of the CRO shall include responsibility for allocation of costs under the Administrative Support Services Agreement unless otherwise provided in the Centerbridge Facility. After the date of this Agreement and prior to the Closing Date, the power and authority of the CRO shall not be changed, and the CRO shall not be removed or replaced, without the consent of Parent, such consent not to be unreasonably withheld, conditioned or delayed. If the CRO voluntary terminates his employment or is otherwise unable to perform his or her responsibilities, a replacement CRO shall be appointed, subject to Parent's reasonable consent (not to be unreasonably withheld, conditioned or delayed).
 - 5.2 Rent and Forbearance.
 - (a) Adjustment to Rent; Forbearance.
- (i) Upon the termination of the Master Lease with respect to any Leased Facility set forth on Section 5.2(a)(i) of the Debtor Disclosure Letter in connection with the releasing or sale of any Leased Facility by Parent or its Subsidiaries to any third-party purchasers, replacement tenants or other partners of Parent, there shall be no adjustment to Reduced Cash Rent. Upon the termination of the Master Lease with respect to any Leased Facility other than those set forth on Section 5.2(a)(i) of the Debtor Disclosure Letter in connection with the releasing or sale of any Leased Facility by Parent or its Subsidiaries to any third-party purchasers, replacement tenants or other partners of Parent, the parties to the Master Lease shall amend the Master Lease to reduce Minimum Rent (as defined in the Master Lease) and Reduced Cash Rent in a manner reflecting the net sales proceeds of such disposition.
- (b) <u>Forbearance</u>. From the date of this Agreement until the earlier of the Closing Date or termination of this Agreement in accordance with ARTICLE VII hereof, Parent agrees to forbear from bringing suit or exercising remedies under the Guaranty, the Master Lease or Agreements Regarding Subleases arising out of or related to (i) the failure to pay Rent (as defined in the Master Lease), (ii) notwithstanding Section 4.4 of the Master Lease, the failure to deposit or pay any amounts required to be deposited pursuant to Section 4.4 of the Master Lease prior to the termination of this Agreement, (iii) the failure to make capital expenditures required

by the Master Lease, (iv) the occurrence of any "Event of Default" described in Section 16.1(g) or Section 16.1(h) of the Master Lease as a result of the transactions and other actions contemplated by this Agreement, (v) any "Event of Default" under clause (ii) of Section 16.1(1) of the Master Lease resulting from the failure of the Debtor to make any payment due under the Guaranty, (vi) the 2017 default under and acceleration of amounts owing under that certain Credit Agreement, dated as of April 6, 2011, among Holdings (as defined therein), HCR Healthcare, LLC, the several banks and other financial institutions or entities from time to time parties thereto, JPMorgan Chase Bank, N.A., as Administrative Agent and Collateral Agent, and J.P. Morgan Securities Inc., Credit Suisse Securities (USA) LLC and Merrill Lynch, Pierce, Fenner & Smith Incorporated, as co-Lead Arrangers and Joint Bookrunners, as amended by that certain First Amendment to Credit Agreement, dated as of March 21, 2014 and that certain Second Amendment to Credit Agreement, dated as of May 6, 2015 (the "JP Morgan Credit **Facility**"); or (vii) the failure of the Debtor to make any payment due under the Guaranty, including, in each case, by delivering a notice of default under the Master Lease, terminating the Master Lease, declaring an "Event of Default" under the Master Lease or exercising any other right or remedy under the Master Lease and/or Agreements Regarding Subleases against the Debtor or its direct and indirect Subsidiaries and, with respect to each, their property and interests in property.

- (c) <u>Forbearance Stipulations</u>. The Debtor and its Subsidiaries hereby admit, stipulate and agree as follows:
- (i) as of the date hereof (and taking into account payments made by HCR III to the Lessors on the date hereof), the Lessors were owed approximately \$180,571,590 in unpaid Rent currently due and owing under the Master Lease, including \$14,000,000 in unpaid Rent due and owing since the date of the Second Forbearance Agreement until the date of this Agreement ("Reduced Cash Rent Past Due Balance");
- (ii) as of the date hereof, the Lessors were owed approximately \$265,225,000 in unpaid Deferred Rent Obligations (as defined in the Master Lease) currently due and owing under the Master Lease;
- (iii) the Rent and other obligations of HCR III under the Master Lease and the Debtor under the Guaranty constitute legal, valid and binding obligations of HCR III and the Debtor, respectively, enforceable against HCR III and the Debtor in accordance with the terms of the Master Lease and the Guaranty;
- (iv) no portion of the Rent and other obligations of HCR III under the Master Lease, the Debtor under the Guaranty, or the Sublease OpCos under the Agreements Regarding Subleases is subject to avoidance, recharacterization, recovery or subordination pursuant to the Bankruptcy Code or applicable nonbankruptcy law;
- (v) the Debtor and HCR III do not have any claims, counterclaims, causes of action, defenses or setoff rights arising under or in connection with the Master Lease, the Guaranty or the Agreements Regarding Subleases against Parent, the Lessors or their respective Representatives (in their capacities as such) with respect to any unpaid Rent (including the Deferred Rent Obligation); and

- (vi) other than (A) as a result of HCR III's failure to pay all due and owing Rent in full, the failure to deposit or pay any amounts required to be deposited pursuant to Section 4.4 of the Master Lease or the failure to make capital expenditures required by the Master Lease, (B) any "Event of Default" under clause (ii) of Section 16.1(l) of the Master Lease resulting from the failure of the Debtor to make any payment due under the Guaranty, (C) the 2017 default under and acceleration of amounts owing under the JP Morgan Credit Facility and (D) the other matters described in Section 5.2(b), no default or Event of Default (as defined in the Master Lease) under the Master Lease is continuing.
- 5.3 <u>Business Operations</u>. Subject to the terms and conditions hereof, from the date of this Agreement until the Closing Date, the Debtor shall, and shall cause its Subsidiaries to:
- (a) comply in all material respects with all non-monetary obligations under the Master Lease, Guaranty and Agreements Regarding Subleases (it being understood that an obligation to make capital expenditures shall constitute a monetary obligation);
- (b) cooperate with Parent in the sale, remarketing or re-leasing of any Leased Facilities pursuant to the terms of this Agreement and the protocol set forth in Section 5.3(b) of the Debtor Disclosure Letter (the "SNF/AL Remarketing Process"), including taking all steps required under the Centerbridge Facility to facilitate such process; provided that any such sale, remarketing or re-leasing (or any other disposition of any SNF Business) prior the Closing shall be subject to appropriate disclosure to the Bankruptcy Court;
- (c) provide Parent and any actual or potential purchasers or replacement tenants of the Leased Facilities with customary access and information in support of the SNF/AL Remarketing Process, subject to entry by such third-party acquiror or lessee into appropriate confidentiality arrangements pursuant to Section 4.8;
- (d) provide such access and information to any Potential Acquiror and each of its Representatives as such Potential Acquiror may reasonably request from time to time in order to facilitate its due diligence investigations, subject to such limitations on access and information, *mutatis mutandis*, as set forth in Section 26.1 of the Master Lease, and subject to entry by such Potential Acquiror into reasonable customary confidentiality arrangements pursuant to Section 4.8 in a form that is reasonably acceptable to the Debtor; it being understood that such information may include information that has not previously been provided to the Purchaser Entities in connection with their respective entry into this Agreement or pursuant to the Master Lease;
- (e) keep on the premises of each Leased Facility all material personal property utilized in the ordinary course of operations at each such Leased Facility;
- (f) deliver to Parent the auditor's consent relating to the Audited Financial Statement for the year ended December 31, 2017 in connection with Parent's timely filing with the SEC of its Annual Report on Form 10-K for the year ended December 31, 2017; and
- (g) deliver to the QCP Consultants and Parent (i) an updated 13-Week Cash Budget on the date that is four weeks after the date of this Agreement and every four weeks

thereafter and (ii) a report on a weekly basis, beginning on the third business day of the week after the date of this Agreement, setting forth any line-item variances from the most recent 13-Week Cash Budget delivered to Parent and the QCP Consultants.

ARTICLE VI

Conditions

- 6.1 <u>Conditions to Each Party's Obligations</u>. The respective obligations of each party to effect the Transactions are subject to the satisfaction or waiver at or prior to the Closing of each of the following conditions:
- (a) <u>Litigation</u>. No court or other Governmental Entity of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any Law (whether temporary, preliminary or permanent) that is in effect and restrains, enjoins or otherwise prohibits consummation of the Transactions. No Law shall be in effect enjoining or otherwise prohibiting the consummation of the Transactions.
- (b) <u>Approvals</u>. The Governmental Approvals set forth on Section 6.1(b) of the Debtor Disclosure Letter shall have been obtained or made, except to the extent the failure to obtain such Governmental Approvals would not, in the aggregate, reasonably be expected to result in losses, costs, liabilities or expenses to the Purchaser Entities, the Debtor and their respective Subsidiaries in excess of \$5,000,000.
- Confirmation Order and such Order shall have become a Final Order, there shall have been no amendment to the Plan or the Confirmation Order except as permitted hereby and all conditions to the effectiveness of the Plan shall have occurred. "Final Order" shall mean an Order which has not been stayed (or with respect to which any stay has been lifted) and (i) as to which the time to file an appeal, a motion for rehearing or reconsideration (excluding any motion under Section 60(b) of the Federal Rules of Civil Procedure) or a petition for writ of certiorari has expired and no appeal, motion, stay or petition is pending, or (ii) in the event that such an appeal or petition thereof has been sought, either (A) such Order shall have been affirmed by the highest court to which such Order was appealed or certiorari shall have been denied, and the time to take any further appeal or petition of certiorari shall have expired or (B) such appeal, motion, stay or petition shall not have been granted and shall no longer be pending and the time for seeking such appeal, motion, stay or petition shall have been expired.
- (d) <u>Bankruptcy Court Orders</u>. The Bankruptcy Court shall not have entered an Order (i) dismissing the Bankruptcy Case or converting the Bankruptcy Case to a case under Chapter 7 of the Bankruptcy Code or (ii) materially inconsistent with this Agreement, the Plan or the Confirmation Order in a manner adverse to the Purchaser Entities.
- 6.2 <u>Conditions to Obligations of Purchaser Entities</u>. The obligations of Purchaser Entities to effect the Transactions are also subject to the satisfaction or waiver by Parent at or prior to the Closing Date of the following conditions:

- Representations and Warranties. (i) The representations and warranties of (a) the Debtor (other than the representations and warranties set forth in Sections 2.1(a) (Organization, Good Standing and Qualification), 2.1(c) (Corporate Authority and Approval), 2.1(s) (Brokers and Finders) and the first sentence of Section 2.1(b) (Capital Structure) (the "Fundamental Representations of the Debtor")) in this Agreement (without giving effect to any limitations as to "materiality" or "Material Adverse Effect" set forth therein) shall be true and correct as of the date hereof and true and correct as of the Closing as though made on and as of the Closing (except for representations and warranties that are made expressly as of a specific date, which representations and warranties shall be true and correct as of such date), in each case, except for such failures to be so true and correct as shall not have had, and would not, individually or in the aggregate, reasonably be likely to have a Material Adverse Effect, and (ii) the Fundamental Representations of the Debtor shall be true and correct in all respects as of the date hereof and true and correct in all respects as of the Closing as though made on and as of the Closing (except for representations and warranties that are made expressly as of a specific date, which representations and warranties shall be true and correct as of such date).
- (b) <u>Performance of Obligations of the Debtor</u>. The Debtor shall have performed and complied in all material respects with all covenants required to be performed by it under this Agreement on or prior to the Closing Date.
- (c) <u>Officer's Certificate</u>. The Purchaser Entities shall have received a certificate dated as of the Closing Date, signed by the chief executive officer of the Debtor, stating that the conditions specified in Sections 6.2(a) and 6.2(b) have been waived or satisfied.
- 6.3 <u>Conditions to Obligation of the Debtor</u>. The obligation of the Debtor to effect the Transactions is also subject to the satisfaction or waiver by the Debtor at or prior to the Closing of the following conditions:
- (a) Representations and Warranties. The representations and warranties of the Purchaser Entities set forth in this Agreement shall be true and correct as of the date hereof and true and correct as of the Closing as though made on and as of the Closing (except for representations and warranties that are made expressly as of a specific date, which representations and warranties shall be true and correct as of such date), except in all cases where the failure of such representations and warranties to be true and correct does not have a material adverse effect on the Purchaser Entities' ability to timely consummate the Transactions.
- (b) <u>Performance of Obligations of Purchaser Entities</u>. Each of the Purchaser Entities shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date.
- (c) <u>Officer's Certificate</u>. The Debtor shall have received a certificate dated as of the Closing Date, signed by the chief executive officer of the Parent, stating that the conditions specified in Sections 6.3(a) and 6.3(b) have been waived or satisfied.

ARTICLE VII

Termination

- 7.1 <u>Automatic Termination</u>. Subject to Section 8.1, this Agreement will expire by its own terms and terminate automatically, without notice or further action by any party, if:
- (a) the solicitation of the Plan or Petition Date does not occur on or prior to the applicable dates required by Section 3.2; <u>provided</u> that Parent shall have the option to extend such dates without limit (but subject to the other termination events described herein) by written notice to the Debtor from time to time;
- (b) any of the Restructuring Documents is not filed on the Petition Date; <u>provided</u> that Parent shall have the option to extend such date without limit (but subject to the other termination events described herein) by written notice to the Debtor from time to time;
- (c) a hearing to consider entry of the Confirmation Order is not scheduled within forty (40) days of the Petition Date; <u>provided</u> that Parent shall have the option to extend such date without limit (but subject to the other termination events described herein) by written notice to the Debtor, from time to time;
- (d) the Confirmation Order has not been entered by the Bankruptcy Court within forty-five (45) days after the Petition Date; <u>provided</u> that Parent shall have the option to extend such date without limit (but subject to the other termination events described herein) by written notice to the Debtor, from time to time;
- (e) the Confirmation Order has not become a Final Order by September 30, 2018; <u>provided</u> that Parent shall have the option to extend such date without limit (but subject to the other termination events described herein) by written notice to the Debtor, from time to time;
- (f) the Debtor or any of its Subsidiaries files any motion or other request for relief seeking to (i) dismiss the Bankruptcy Case, (ii) convert the Bankruptcy Case to a case under chapter 7 of the Bankruptcy Code or (iii) appoint a trustee or an examiner with expanded power pursuant to section 1104 of the Bankruptcy Code;
- (g) the Bankruptcy Court or any court exercising appellate jurisdiction over the Bankruptcy Court enters an Order (i) denying confirmation of the Plan, (ii) converting the Bankruptcy Case of the Debtor to a case under chapter 7 of the Bankruptcy Code, (iii) appointing a trustee or examiner with expanded powers pursuant to section 1104 of the Bankruptcy Code or (iv) that is materially inconsistent with the Plan or the Transactions in a manner adverse to the Purchaser Entities, and such Order (x) is or becomes a Final Order or (y) has been in effect for thirty (30) days and is not subject to stay; or
- (h) any Subsidiary of the Debtor commences proceedings under the Bankruptcy Code without the prior written consent of Parent, which may be withheld in its sole discretion.

- 7.2 <u>Termination by Mutual Consent</u>. This Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing, by mutual written consent of the Debtor and Parent, by action of their respective boards of directors.
- 7.3 Termination by Either Parent or the Debtor. This Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing by action of the board of directors of either Parent or the Debtor if (a) the Transactions shall not have been consummated by September 30, 2018 (the "Termination Date") or (b) any Order permanently restraining, enjoining or otherwise prohibiting consummation of the Transactions shall become final and non-appealable; provided that the right to terminate this Agreement pursuant to this Section 7.3 shall not be available to any party that has breached in any material respect its obligations under this Agreement in any manner that shall have proximately contributed to the occurrence of the failure of a condition to the consummation of the Transactions.
- 7.4 <u>Termination by the Debtor</u>. This Agreement may be terminated by the Debtor and the Transactions may be abandoned:
- (a) if there has been a material breach of any representation, warranty, covenant or agreement made by the Purchaser Entities in this Agreement, or any such representation and warranty shall have become untrue after the date of this Agreement, such that Section 6.3(a) or Section 6.3(b) would not be satisfied and such material breach or condition is not curable or, if curable, is not cured within the earlier of (x) thirty (30) days after written notice thereof is given by the Debtor to Parent and (y) the Termination Date; provided that the Debtor shall not have the right to terminate this Agreement pursuant to this Section 7.4 if the Debtor is then in material breach of any of its representations, warranties, covenants or agreements contained in this Agreement such that the conditions set forth in Section 6.2(a) or Section 6.2(b) would not be satisfied; or
- (b) if any condition set forth in Section 6.1 shall have become incapable of fulfillment prior to the Termination Date other than as a result of a material breach by the Debtor of any covenant or agreement contained in this Agreement, and such condition is not waived by the Debtor.
- 7.5 <u>Termination by Parent</u>. This Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing by Parent:
- (a) if there has been a material breach of any representation, warranty, covenant or agreement made by the Debtor in this Agreement, or any such representation and warranty shall have become untrue after the date of this Agreement, such that Section 6.2(a) or Section 6.2(b) would not be satisfied and such material breach or condition is not curable or, if curable, is not cured within the earlier of (x) thirty (30) days after written notice thereof is given by Parent to the Debtor and (y) the Termination Date; provided that Parent shall not have the right to terminate this Agreement pursuant to this Section 7.5 if any of the Purchaser Entities is then in material breach of any of its representations, warranties, covenants or agreements contained in this Agreement such that the conditions set forth in Section 6.3(a) or Section 6.3(b) would not be satisfied:

- (b) if the Carlyle Holders have (i) filed any federal or state tax return, or any amendment to such a return, claiming any deduction for worthlessness of its Debtor Shares (as defined in the Restructuring Support Agreement) or (ii) Transferred (as defined in the Restructuring Support Agreement) any Debtor Shares other than in accordance with the Restructuring Support Agreement;
- (c) if any condition set forth in Section 6.1 shall have become incapable of fulfillment prior to the Termination Date other than as a result of a material breach by the Purchaser Entities of any covenant or agreement contained in this Agreement, and such condition is not waived by Parent;
- (d) if there has been a material breach of the obligations set forth in Section 4.10;
- (e) other than in accordance with the Plan, if there has been (i) a termination or rejection by the Lessor of the Master Lease or any Agreement Regarding Subleases by the Lessor or any other Subsidiary of the Debtor, or (ii) the occurrence of a Default or an Event of Default (each as defined under the Centerbridge Facility) under the Centerbridge Facility that would permit the acceleration of Obligations (as defined under the Centerbridge Facility) and a failure by the Debtor to cure such Default or Event of Default within thirty (30) days of the occurrence of such a Default or Event of Default, or the acceleration of Obligations (as defined under the Centerbridge Facility); or
- (f) if (i) Debtor fails to comply with Section 5.3(f) or (ii) HCR III fails to pay such cash and cash equivalents available to pay part or all of the Reduced Cash Rent due from and after the date hereof after (A) making all transfers of funds from the Debtor's home health care, hospice and other ancillary businesses other than the Debtor's skilled nursing and inpatient rehabilitation facilities, memory care facilities and assisted living facilities (the "SNF Business") that are permitted under the Centerbridge Facility and (B) paying all available cash and cash equivalents from the SNF Business subject to retaining such reserves and making such other expenditures as either (x) the CRO or (y) the Debtor Board has determined in good faith, after consulting with Parent, are necessary to allow the Debtor to operate safely, prudently and in the ordinary course of business, and, in case of either of clauses (i) or (ii), HCR III does not cure such failure within five (5) business days of such failure.

ARTICLE VIII

Miscellaneous and General

- 8.1 No Survival; Effect of Termination and Abandonment.
- (a) Except as provided in Section 8.1(b) below, in the event of termination of this Agreement and the abandonment of the Transactions pursuant to ARTICLE VII, this Agreement shall become void and be of no effect with no liability to any Person on the part of any party hereto (or of any of its Representatives or Affiliates); provided, however, and notwithstanding anything in the foregoing to the contrary, that (i) no such termination shall relieve any party of any liability or damages resulting from any fraud by such party or Willful

Breach of this Agreement on the part of such party occurring prior to such termination and (ii) the provisions set forth in this ARTICLE VIII, Parent's expense reimbursement and indemnification obligations set forth in Section 4.9(d) and the agreements of the Debtor and the Purchaser Entities contained in Section 4.4 (*Publicity*), Section 4.6 (*Expenses*) and Section 4.8 (*Confidentiality*) shall survive the termination of this Agreement. For purposes of this Agreement, "Willful Breach" shall mean a material breach of, or failure to perform, any of the covenants or other agreements contained in this Agreement (including failure to close the Transactions in accordance with Section 1.4) that is a consequence of an act, or failure to act, by the breaching or non-performing Person with actual knowledge, or knowledge that a Person acting reasonably under the circumstances would reasonably be expected to have, that such Person's act or failure to act would, or would be reasonably expected to, result in or constitute a breach of, or failure of, performance under this Agreement.

- (b) <u>Waiver</u>; <u>Automatic Stay</u>. The failure of a party to exercise its right to terminate this Agreement under, or the extension of any time period in, any provision of ARTICLE VII at any time will not constitute a waiver of any such right. The Debtor hereby waives any requirement that any Purchaser Entity seek or obtain relief from the automatic stay under section 362 of the Bankruptcy Code for purposes of providing notice under, or terminating, this Agreement, and the Debtor agrees not to object to the Purchaser Entities seeking, if necessary, to lift the automatic stay in connection therewith.
- (c) This ARTICLE VIII and the agreements of the Debtor and the Purchaser Entities contained in Section 4.7 (*Indemnification; Directors' and Officers' Insurance*) shall survive the consummation of the Transactions. All other representations, warranties, covenants and agreements in this Agreement and in any certificate or other writing delivered pursuant hereto shall not survive consummation of the Transactions and, from and after the Closing Date, no party shall have any claim against any other party for breach of any representation or warranty set forth herein.
- 8.2 <u>Modification or Amendment</u>. Subject to the provisions of applicable Laws, at any time prior to the Closing Date, the parties hereto may modify or amend this Agreement, by written agreement executed and delivered by duly authorized officers of the respective parties.
- 8.3 <u>Waiver of Conditions</u>. The conditions to each of the parties' obligations to consummate the Transactions and the other transactions contemplated by this Agreement are for the sole benefit of such party and may be waived by such party in whole or in part to the extent permitted by applicable Laws.
- 8.4 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each such counterpart being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement.

8.5 <u>GOVERNING LAW AND VENUE; WAIVER OF JURY TRIAL;</u> SPECIFIC PERFORMANCE.

- THIS AGREEMENT SHALL BE DEEMED TO BE MADE IN AND IN (a) ALL RESPECTS SHALL BE INTERPRETED, CONSTRUED AND GOVERNED BY AND IN ACCORDANCE WITH THE LAW OF THE STATE OF DELAWARE WITHOUT REGARD TO THE CONFLICTS OF LAW PRINCIPLES THEREOF TO THE EXTENT THAT SUCH PRINCIPLES WOULD DIRECT A MATTER TO ANOTHER JURISDICTION. The parties hereby irrevocably submit to the personal jurisdiction of the courts of the State of Delaware and the federal courts of the United States of America located in the State of Delaware solely in respect of the interpretation and enforcement of the provisions of this Agreement and of the documents referred to in this Agreement, and in respect of the Transactions, and hereby waive, and agree not to assert, as a defense in any Proceeding for the interpretation or enforcement hereof or of any such document, that it is not subject thereto or that such Proceeding may not be brought or is not maintainable in said courts or that the venue thereof may not be appropriate or that this Agreement or any such document may not be enforced in or by such courts, and the parties hereto irrevocably agree that all claims relating to such Proceeding or transactions shall be heard and determined in such a Delaware State or federal court. The parties hereby consent to and grant any such court jurisdiction over the person of such parties and, to the extent permitted by law, over the subject matter of such dispute and agree that mailing of process or other papers in connection with any such Proceeding in the manner provided in Section 8.6 or in such other manner as may be permitted by Law shall be valid and sufficient service thereof. Notwithstanding the foregoing consent to jurisdiction in either a state or federal court of competent jurisdiction in the State of Delaware, each of the parties hereby agrees that during the pendency of the Bankruptcy Case, the Bankruptcy Court shall have exclusive jurisdiction over all matters arising out of or in connection with this Agreement.
- EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY (b) CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (i) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (ii) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (iii) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (iv) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 8.5.
- (c) The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the

terms and provisions of this Agreement in the Court of Chancery of the State of Delaware or, during the Bankruptcy Case, in the Bankruptcy Court, this being in addition to any other remedy to which such party is entitled at law or in equity. Each of the parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that (x) the other party has an adequate remedy at law or (y) an award of specific performance is not an appropriate remedy for any reason at law or in equity, and that it waives any requirement for the securing or posting of a bond in connection with any such remedy.

8.6 <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by any party to the other parties shall be in writing and delivered personally or sent by registered or certified mail, postage prepaid, by facsimile or overnight courier:

<u>If to Parent, Purchaser or the Lessors</u>:

Quality Care Properties, Inc. 7315 Wisconsin Avenue, Suite 550 East Bethesda, Maryland 20814 Attention: David Haddock, General Counsel

Email: dhaddock@qcpcorp.com

with a copy to:

Sullivan & Cromwell LLP 125 Broad Street New York, New York 10004 Attention: Audra D. Cohen and Andrew G. Dietderich

Email: cohena@sullcrom.com and dietdericha@sullcrom.com

If to the Debtor:

HCR ManorCare, Inc. 333 North Summit Street Toledo, Ohio 43604

Attention: Richard A. Parr, General Counsel

John Castellano, Chief Restructuring Officer

Email: rparr@hcr-manorcare.com jcastellano@alixpartners.com

with a copy to:

Sidley Austin LLP One South Dearborn Chicago, IL 60603 Attention: Larry Nyhan Email: lnyhan@sidley.com

and

Latham & Watkins LLP
555 Eleventh Street, NW, Suite 1000
Washington, D.C. 20004-1304
Attention: Daniel T. Lennon
Roger G. Schwartz
J. Cory Tull

Email: daniel.lennon@lw.com roger.schwartz@lw.com cory.tull@lw.com

or to such other persons or addresses as may be designated in writing by the party to receive such notice as provided above. Any notice, request, instruction or other document given as provided above shall be deemed given to the receiving party: (i) upon actual receipt, if delivered personally; (ii) three (3) business days after deposit in the mail, if sent by registered or certified mail; (iii) upon confirmation of successful transmission if sent by facsimile (provided that if given by facsimile, such notice, request, instruction or other document shall be followed up within one (1) business day by dispatch pursuant to one of the other methods described herein); or (iv) on the next business day after deposit with an overnight courier, if sent by an overnight courier.

- Entire Agreement. This Agreement (including any exhibits hereto), the 8.7 Debtor Disclosure Letter and the Restructuring Support Agreement constitute the entire agreement and supersede all other prior agreements, understandings, representations and warranties both written and oral, among the parties, with respect to the subject matter hereof. EACH PARTY HERETO AGREES THAT, EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS AGREEMENT, NEITHER PARENT, PURCHASER, THE LESSORS NOR THE DEBTOR MAKES OR RELIES ON ANY OTHER REPRESENTATIONS, WARRANTIES OR INDUCEMENTS, AND EACH HEREBY DISCLAIMS ANY OTHER REPRESENTATIONS, WARRANTIES OR INDUCEMENTS, EXPRESS OR IMPLIED, RELATED TO THE DEBTOR, THE TRANSACTIONS OR ANY OTHER MATTER, INCLUDING REPRESENTATIONS OR WARRANTIES AS TO THE ACCURACY OR COMPLETENESS OF ANY OTHER INFORMATION, MADE BY, OR MADE AVAILABLE BY, ITSELF OR ANY OF ITS REPRESENTATIVES, WITH RESPECT TO, OR IN CONNECTION WITH, THE NEGOTIATION, EXECUTION OR DELIVERY OF THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY, NOTWITHSTANDING THE DELIVERY OR DISCLOSURE TO THE OTHER OR THE OTHER'S REPRESENTATIVES OF ANY DOCUMENTATION OR OTHER INFORMATION WITH RESPECT TO ANY ONE OR MORE OF THE FOREGOING. No party shall be bound by, or be liable for, any alleged representation, promise, inducement or statement of intention not contained herein.
- 8.8 No Third Party Beneficiaries; No Recourse. Except as provided in Section 4.7 (*Indemnification; Directors' and Officers' Insurance*) and Section 8.15 (*Certain Waivers*) only, Parent, Purchaser, the Lessors and the Debtor hereby agree that their respective representations, warranties and covenants set forth herein are solely for the benefit of the other parties hereto, in accordance with and subject to the terms of this Agreement, and this Agreement is not intended to, and does not, confer upon any Person other than the parties hereto

any rights or remedies hereunder, including, without limitation, the right to rely upon the representations and warranties set forth herein. The representations and warranties in this Agreement are the product of negotiations among the parties hereto and are for the sole benefit of the parties hereto. Any inaccuracies in such representations and warranties are subject to waiver by the parties hereto in accordance with Section 8.3 without notice or liability to any other Person. In some instances, the representations and warranties in this Agreement may represent an allocation among the parties hereto of risks associated with particular matters regardless of the knowledge of any of the parties hereto. Consequently, Persons other than the parties hereto may not rely upon the representations and warranties in this Agreement as characterizations of actual facts or circumstances as of the date of this Agreement or as of any other date. This Agreement may only be enforced against, and any claim or cause of action based upon, arising out of, or related to this Agreement or the Transactions may only be brought against, the entities that are expressly named as parties hereto and then only with respect to the specific obligations set forth herein with respect to such party. Except to the extent a named party to this Agreement (and then only to the extent of the specific obligations undertaken by such named party in this Agreement and not otherwise), no past, present or future director, officer, employee, incorporator, member, partner, stockholder, Affiliate, agent, attorney, advisor or representative or Affiliate of any of the foregoing shall have any liability (whether in contract, tort, equity or otherwise) for any one or more of the representations, warranties, covenants, agreements or other obligations or liabilities of any one or more of the Debtor, Parent, Purchaser or the Lessors under this Agreement (whether for indemnification or otherwise) or of or for any claim based on, arising out of, or related to this Agreement or the Transactions.

- 8.9 Obligations of Parent and of the Debtor. Whenever this Agreement requires a Subsidiary of Parent to take any action, such requirement shall be deemed to include an undertaking on the part of Parent to cause such Subsidiary to take such action. Whenever this Agreement requires a Subsidiary of the Debtor to take any action, such requirement shall be deemed to include an undertaking on the part of the Debtor to cause such Subsidiary to take such action.
- 8.10 <u>Transfer Taxes</u>. All transfer, documentary, sales, use, stamp, registration and other such Taxes and fees (including penalties and interest) incurred in connection with the Transactions shall be paid by the Purchaser Entities when due, and the Purchaser Entities will indemnify the Debtor against liability for any such Taxes.
- 8.11 <u>Definitions</u>. Each of the terms set forth in Annex A is defined in the Section of this Agreement set forth opposite such term.
- 8.12 <u>Severability</u>. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application of such provision to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability, nor shall such

invalidity or unenforceability affect the validity or enforceability of such provision, or the application of such provision, in any other jurisdiction.

- 8.13 <u>Interpretation; Construction</u>. The table of contents and headings herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof. Where a reference in this Agreement is made to a Section, Annex or Exhibit, such reference shall be to a Section of or Annex or Exhibit to this Agreement unless otherwise indicated. Whenever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation".
- (a) The parties have participated jointly in negotiating and drafting this Agreement. In the event that an ambiguity or a question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.
- (b) The Debtor and Parent have or may have set forth information in the Debtor Disclosure Letter in a section of such Debtor Disclosure Letter that corresponds to the section of this Agreement to which it relates. The fact that any item of information is disclosed in the Debtor Disclosure Letter to this Agreement shall not be construed to mean that such information is required to be disclosed by this Agreement.
- 8.14 <u>Assignment</u>. This Agreement shall not be, directly or indirectly, assignable by operation of law or otherwise; <u>provided</u>, <u>however</u>, that Parent may designate, by written notice to the Debtor, another wholly-owned direct or indirect wholly-owned Subsidiary to be a party in lieu of Purchaser (provided that such Subsidiary remains wholly-owned by Parent through the Closing Date), in which event all references herein to Purchaser shall be deemed references to such other Subsidiary, except that all representations and warranties made herein with respect to Purchaser as of the date of this Agreement shall be deemed representations and warranties made with respect to such other Subsidiary as of the date of such designation; <u>provided</u> that any such designation shall not materially impede or delay the consummation of the Transactions or otherwise materially impede the rights of the stockholders of the Debtor under this Agreement. Any purported assignment in violation of this Agreement is void and such designation shall not relieve Parent of any obligation or liability hereunder. The sale or transfer by Parent of any direct or indirect equity interest in Purchaser shall constitute an assignment of this Agreement.
- 8.15 <u>Certain Waivers</u>. Each Purchaser Entity and the Debtor (on behalf of itself and its Affiliates) acknowledge that Latham & Watkins LLP and other legal counsel ("<u>Prior Debtor Counsel</u>") have, on or prior to the Closing Date, represented (i) the Carlyle Holders in connection with the Transactions and (ii) the Debtor and its Subsidiaries and other Affiliates in one or more matters, including relating to this Agreement and the Transactions (each, an "<u>Existing Representation</u>"), and that, in the event of any post-Closing matters (x) relating to this Agreement or any other agreements or Transactions (including any matter that may be related to a litigation, claim or dispute arising under or related to this Agreement or such other agreements or in connection with such transactions) and (y) in which Parent or any of its

Affiliates (including the Debtor and its Subsidiaries), on the one hand, and any Carlyle Holder or any of its Affiliates (each, a "Designated Person"), on the other hand, are or may be adverse to each other (each, a "Post-Closing Matter"), the Designated Persons reasonably anticipate that Prior Debtor Counsel will represent them in connection with such matters. Accordingly, each Purchaser Entity and the Debtor (on behalf of itself and its Affiliates) hereby (i) waive and shall not assert, and agree after the Closing to cause their Affiliates to waive and to not assert, any conflict of interest arising out of or relating to the representation by one or more Prior Debtor Counsel of one or more Designated Persons in connection with one or more Post-Closing Matters (the "Post-Closing Representations"), and (ii) agree that, in the event that a Post-Closing Matter arises, Prior Debtor Counsel may represent one or more Designated Persons in the Post-Closing Matter even though the interests of such Person(s) may be directly adverse to Parent or any of its Affiliates (including the Debtor and its Subsidiaries), and even though Prior Debtor Counsel may (i) have represented the Debtor or its Subsidiaries in a matter substantially related to such dispute or (ii) be currently representing the Debtor or any of its respective Affiliates. Without limiting the foregoing, each Purchaser Entity and the Debtor (on behalf of itself and its Affiliates) consent to the disclosure by Prior Debtor Counsel, in connection with one or more Post-Closing Representations, to the Designated Persons of any information learned by Prior Debtor Counsel in the course of one or more Existing Representations, whether or not such information is subject to the attorney-client privilege of the Debtor or any of its Subsidiaries or Prior Debtor Counsel's duty of confidentiality as to the Debtor or any of its Subsidiaries and whether or not such disclosure is made before or after the Closing. The Carlyle Holders and Latham & Watkins LLP are intended third-party beneficiaries, and are relying on the provisions of, of this Section 8.15.

- 8.16 No Solicitation. This Agreement, the Transactions and the Plan are the product of negotiations among the parties, together with their respective representatives. Notwithstanding anything herein to the contrary, this Agreement is not, and shall not be deemed to be, a solicitation of votes for the acceptance of the Plan or any plan of reorganization for the purposes of sections 1125 and 1126 of the Bankruptcy Code or otherwise. The Debtor will not solicit acceptances of the Plan from Parent until Parent has been provided with copies of a Disclosure Statement containing adequate information as required by section 1125 of the Bankruptcy Code.
- 8.17 Settlement Discussions. This Agreement is part of a proposed settlement of matters that could otherwise be the subject of litigation among the parties. Nothing herein shall be deemed an admission of any kind. Pursuant to the agreement of the parties, Federal Rule of Evidence 408, any applicable state rules of evidence and any other applicable law, foreign or domestic, this Agreement, all exhibits, schedules and appendices hereto, and all negotiations relating hereto shall not be admissible into evidence in any proceeding other than to prove the existence of this Agreement or in a proceeding to enforce the terms of this Agreement.

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 303 of 441

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the duly authorized officers of the parties hereto as of the date first written above.

HCR	MANORCARE, INC.
Ву:	Name: John R. Castellano Title: Chief Rostructuring Office
QUA	ALITY CARE PROPERTIES, INC.
By:	
	Name: Title:
НСР	MEZZANINE LENDER, LP
By:	
Ву:	Name: Title:

[SIGNATURE PAGE CONTINUES]

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the duly authorized officers of the parties hereto as of the date first written above.

HCR MANORCARE, INC.

By:	
Name:	
Title:	
QUALITY CARE PROPERTIES, INC.	/
1////	

By: Name:

Title:

Marc Richards

Chief Financial Officer

HCP MEZZANINE LENDER, LP

By:

Name:

Marc Richards

Title:

Chief Financial Officer

[SIGNATURE PAGE CONTINUES]

"LESSORS"

HCP PROPERTIES, LP, a Delaware limited partnership

By: HCP I-B Properties, LLC, a Delaware limited liability company, its General Partner

HCP WEST VIRGINIA PROPERTIES, LLC, a Delaware limited liability company

HCP PROPERTIES OF ALEXANDRIA VA, LLC, a Delaware limited liability company

HCP PROPERTIES OF ARLINGTON VA, LLC, a Delaware limited liability company

HCP PROPERTIES OF MIDWEST CITY OK, LLC, a Delaware limited liability company

HCP PROPERTIES OF OKLAHOMA CITY (NORTHWEST), LLC, a Delaware limited liability company

HCP PROPERTIES OF OKLAHOMA CITY (SOUTHWEST), LLC, a Delaware limited liability company

HCP PROPERTIES OF TULSA OK, LLC, a Delaware limited liability company

HCP PROPERTIES-ARDEN COURTS OF ANNANDALE VA, LLC, a Delaware limited liability company

HCP PROPERTIES-CHARLESTON OF HANAHAN SC, LLC, a Delaware limited liability company

HCP PROPERTIES-COLUMBIA SC, LLC, a Delaware limited liability company

HCP PROPERTIES-FAIR OAKS OF FAIRFAX VA, LLC, a Delaware limited liability company

HCP PROPERTIES-IMPERIAL OF RICHMOND VA, LLC, a Delaware limited liability company

HCP PROPERTIES-LEXINGTON SC, LLC, a Delaware limited liability company

HCP PROPERTIES-MEDICAL CARE CENTER-LYNCHBURG VA, LLC, a Delaware limited liability company

HCP PROPERTIES-OAKMONT EAST-GREENVILLE SC, LLC, a Delaware limited liability company

HCP PROPERTIES-OAKMONT OF UNION SC, LLC, a Delaware limited liability company

HCP PROPERTIES-OAKMONT WEST-GREENVILLE SC, LLC, a Delaware limited liability company

HCP PROPERTIES-STRATFORD HALL OF RICHMOND VA, LLC, a Delaware limited liability company

HCP PROPERTIES-WEST ASHLEY-CHARLESTON SC, LLC, a Delaware limited liability company

HCP MARYLAND PROPERTIES, LLC, a Delaware limited liability company

HCP PROPERTIES-SALMON CREEK WA, LLC, a Delaware limited liability company

HCP PROPERTIES-WINGFIELD HILLS NV, LLC, a Delaware limited liability company

HCP PROPERTIES-UTICA RIDGE IA, LLC, a Delaware limited liability company

HCP TWINSBURG OH PROPERTY, LLC, a Delaware limited liability company

By: HCP Schoenherr Road Property, LLC, a Delaware limited liability company, its sole member

By: HCP Properties, LP, a Delaware limited partnership, its sole member

By: HCP I-B Properties, LLC, a Delaware limited liability company, its General

Partner

By: ____

Name: Title:

Marc Richards

Chief Financial Officer

HCP STERLING HEIGHTS MI PROPERTY, LLC, a Delaware limited liability company

By: HCP Schoenherr Road Property, LLC, a Delaware limited liability company, its sole member

By: HCP Properties, LP, a Delaware limited partnership, its sole member

By: HCP I-B Properties, LLC, a Delaware limited liability company, its General

Partner

By:

Name:

Marc Richards

Title:

Chief Financial Officer

ANNEX A

DEFINED TERMS

<u>Terms</u>	Section
13-Week Cash Budget	5.1(b)(ii)(C)
280G Stockholder Vote	4.5(e)(i)
Acquisition Proposal	4.10(e)
Affiliate	2.1(1)(iii)(A)
Alternative Acquisition Agreement	4.10(e)
Agreement	Preamble
Agreement Regarding Subleases, Agreements Regarding Subleases	Recitals
Applicable Restriction	5.1(b)(i)
Audited Financial Statements	2.1(e)(i)
Bankruptcy and Equity Exception	2.1(c)
Bankruptcy Case	Recitals
Bankruptcy Code	Recitals
Bankruptcy Court	Recitals
Business	Recitals
business day	1.4
By-Laws	1.3(b)
Carlyle Holders	Recitals
Case Budget	3.1(d)
Centerbridge Facility	Recitals
Charter	1.3(a)
Closing	1.4
Closing Date	1.4
Code	1.6(a)
Company Plan	2.1(h)(i)
Confirmation Order	3.1(b)
Contract	2.1(d)(ii)
CEO	5.1(b)(i)
CFO	5.1(b)(i)
CRO	4.1(a)(xix)
D&O Insurance	4.7(b)
Debt Exchange	1.1(a)
Debt Financing Sources	4.9(c)
Debtor	Preamble
Debtor Board	1.6(b)
Debtor Disclosure Letter	2.1
Debtor Documents	2.1(c)
Debtor IT Assets	2.1(q)(vi)
Debtor Labor Agreements	2.1(i)(i)
Debtor Restructuring Committee	5.1(a)
Designated Person	8.15

Disclosure Statement	3.1(b)
Employment Agreement	1.5(b)(i)
Environmental Law	` '
ERISA Affiliate	2.1(h)(v)
ERISA Plans	2.1(h)(i)
Exchange Act	2.2
Excess Severance Payments	1.5(a)
Existing Representation	8.15
Final Order	
Financing	4.9(a)
Former CEO	Recitals
Former CEO Settlement Agreement	Recitals
Fundamental Representations of the Debtor	
GAAP	
Governmental Approvals	2.1(d)(i)
Governmental Entity	1.6(d)
Government Contract	
Guaranty	Recitals
Hazardous Substance	2.1(o)
HCR III	Recitals
Indemnified Parties	4.7(a)
Insurance Policies	2.1(r)
Intellectual Property Rights	
IRS	2.1(h)(ii)
JP Morgan Credit Facility	5.2(b)
Key Employee Incentive Plan	2.1(h)(viii)
Knowledge of the Debtor	2.1(e)(ii)
Laws	2.1(j)
Leased Facility, Leased Facilities	Recitals
Lessors	Preamble
Liabilities	2.1(k)
Licenses	2.1(j)
Lien	1.1(b)
Manorcare Financing	4.9(a)
Marketing Documentation	4.9(a)
Master Lease	Recitals
Master Lease Amendment	1.2
Material Adverse Effect	2.1(a)
Material Contract	2.1(l)(i)
New Hire	4.1(a)(xix)
Non-Debtor Subsidiary Released Parties	3.4
Order	2.1(g)
Other Indemnitors	
Owned Intellectual Property	
Parachute Payment	
Parent	Preamble

party, parties	Preamble
Payment Date	1.5(a)
PBGC	2.1(h)(v)
Person	1.6(d)
Permitted Lien	1.1(b)
Petition Date	3.2(b)
Plan	Recitals
Plan Acquisition	1.1(a)
Plan Term Sheet	Recitals
Post-Closing Matter	8.15
Post-Closing Representations	8.15
Potential Acquiror	4.8
Preferred Shares	2.1(b)(i)
Prior Debtor Counsel	8.15
Prior Forbearance Agreement	Recitals
Proceedings	2.1(g)
Purchaser	Preamble
Purchaser Entities	Preamble
Purchaser LP Agreement	2.2(d)
QCP Consultants	5.1(b)(i)
Receivership Proceeding	Recitals
Reduced Cash Rent	
Reduced Cash Rent Past Due Balance	
Reorganized Company Stock	1.1(a)
Representatives	
Restructuring Documents	
Restructuring Support Agreement	Recitals
SEC	
Second Forbearance Agreement	Recitals
Securities Act	
Separation Agreement	
Shares	
SNF/AL Remarketing Process	1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1
SNF Business	
Stock Plan	
Sublease Amendments	1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1
Subleases	
Sublease OpCo, Sublease OpCos	
Subsidiary	
Tail Period	
Takeover Statute	* /
Tax, Taxes	
Tax Return	A .
Taxing Authority	2 ·
Termination Date	2 ·
Transactions	

Case 18-10467-KG	Doc 197	Filed 05/18/18	Page 311 of 443

Exhibit Version

EXHIBIT 1

HCR MANORCARE, INC. RESTRUCTURING TERM SHEET SUMMARY OF PRINCIPAL TERMS AND CONDITIONS March 2, 2018

The following (this "<u>Plan Term Sheet</u>") describes the material terms of a proposed restructuring of HCR ManorCare, Inc. (the "<u>Debtor</u>"), which will be implemented through a prepackaged plan of reorganization (the "<u>Prepackaged Plan</u>") under title 11 of the United States Code (the "<u>Bankruptcy Code</u>"). The Debtor and Quality Care Properties, Inc. and certain of its subsidiaries (collectively, "<u>QCP</u>") are parties to the Plan Sponsor Agreement, dated as of March 2, 2018 (the "<u>PSA</u>"), which, among other things, set forth the manner in which the Prepackaged Plan will be pursued. Capitalized terms not otherwise defined herein shall have the meanings set forth in the PSA.

In addition, the Debtor, the Carlyle Holders and QCP are party to a Restructuring Support Agreement (the "RSA") pursuant to which, among other things, the Carlyle Holders and QCP have agreed to support the Prepackaged Plan described by the Plan Term Sheet.

This Plan Term Sheet outlines many of the principal terms of the Prepackaged Plan, but other material terms will be set forth in definitive documentation, including the PSA and RSA. Nothing contained herein shall constitute an offer susceptible to acceptance by the Debtor or any other party, or a legally binding obligation of any party, other than as set forth in the PSA and RSA. This Plan Term Sheet is a settlement proposal in furtherance of settlement discussions and is entitled to protection from any use or disclosure to any party or person pursuant to Federal Rule of Evidence 408 or any other applicable rule of evidence. THIS PLAN TERM SHEET DOES NOT CONSTITUTE A SOLICITATION OF VOTES FOR A PLAN OF REORGANIZATION FOR PURPOSES OF SECTIONS 1125 AND 1126 OF THE BANKRUPTCY CODE OR AN OFFER WITH RESPECT TO ANY SECURITIES. SUCH OFFER OR SOLICITATION WILL BE MADE IN COMPLIANCE WITH THE PSA AND THE RSA AND ALL APPLICABLE SECURITIES LAWS AND/OR PROVISIONS OF THE BANKRUPTCY CODE.

Transaction Overview	
Solicitation	The Debtor will pursue a prepetition solicitation of votes on the Prepackaged Plan (the "Solicitation") in accordance with the PSA and RSA.

		-
Prepackaged Case	The Debtor will file a voluntary petition for reorganization under chapter 11 of the Bankruptcy Code, in the United States Bankruptcy Court for the District of Delaware (the "Bankruptcy Court"), in accordance with its obligations under the PSA and RSA. The date on which the Debtor's chapter 11 case is commenced shall be the "Petition Date." In addition, on the Petition Date the Debtor will file "first-day" and "second-day" pleadings with the Bankruptcy Court.	
Funding of Chapter 11 Case	The cost of administering the Debtor's chapter 11 c by cash of the Debtor, ordinary course intercompant the cash management system of the Debtor and its senecessary, additional intercompany borrowing from Services, LLC in accordance with the Centerbridge Debtor shall not incur any debtor-in-possession final avoidance of doubt, the Debtor may seek entry of an chapter 11 case authorizing the Debtor to use the case Centerbridge and providing Centerbridge reasonable protections in connection therewith.	y transfers through subsidiaries, and, if a HCR ManorCare Facility. The ancing. For the n order in the ash collateral of
Treatment of Claims and l	interests	
Administrative Claims	Each holder of an Allowed ¹ Administrative Claim ² will receive, in the sole discretion of the reorganized Debtor, payment in full in cash of the unpaid portion of such Allowed Administrative Claim (a) on the date on which the Prepackaged Plan becomes effective (the "Effective Date") or as promptly thereafter as reasonably practicable, (b) in the ordinary course of the reorganized Debtor's business, or (c) as otherwise agreed by the reorganized Debtor and such holder. ³	In accordance with section 1123(a)(1) of the Bankruptcy Code, the Prepackaged Plan does not classify Administrative Claims.

[&]quot;Allowed" means any claim that is determined to be Allowed (as defined in the Prepackaged Plan).

² "<u>Administrative Claim</u>" means a claim arising under sections 503(b), 507(b) or to the extent applicable, 1114(e)(2) of the Bankruptcy Code.

Administrative Claims that become due prior to the Effective Date may be paid by the Debtor in the ordinary course during the chapter 11 case, subject to the PSA.

Priority Tax Claims	Each holder of an Allowed Priority Tax Claim ⁴ will receive, in the sole discretion of the reorganized Debtor, payment in cash by the Reorganized Debtor in an amount of such holder's Allowed Priority Tax Claim on the later of the Effective Date or when such Allowed Priority Tax Claim becomes due.	In accordance with section 1123(a)(1) of the Bankruptcy Code, the Prepackaged Plan does not classify Priority Tax Claims.
Class 1 – Other Priority Claims	Each holder of an Allowed Other Priority Claim ⁵ will receive payment in full in cash as promptly as reasonably practicable on the later of (a) the Effective Date and (b) the date on which such Other Priority Claim becomes an Allowed claim payable under applicable law or any agreement relating thereto.	Unimpaired – Deemed to Accept
Class 2 – Secured Claims	Each holder of an Allowed secured claim will receive, in the sole discretion of the reorganized Debtor, on the Effective Date (or as promptly thereafter as reasonably practicable) or in the ordinary course of the reorganized Debtor's business, (a) payment in full by the reorganized Debtor in cash, including the payment of any interest Allowed and payable under section 506(b) of the Bankruptcy Code, (b) delivery of the collateral securing such Allowed secured claim or (c) treatment of such Allowed secured claim in any other matter that renders the claim unimpaired.	Unimpaired – Deemed to Accept
Class 3 – Centerbridge Claim	On the Effective Date, Centerbridge's ⁶ secured contingent guaranty claim against the Debtor (the " <u>Centerbridge Claim</u> ") will be reinstated and any liens held by Centerbridge against the assets of the Debtor securing the Centerbridge Claim shall survive the Effective Date, rendering the Centerbridge Claim unimpaired.	Unimpaired – Deemed to Accept

⁴ "Priority Tax Claim" means a claim under section 507(a)(8) of the Bankruptcy Code.

⁵ "<u>Other Priority Claim</u>" means an Allowed claim under section 507(a) of the Bankruptcy other than an Administrative Expense Claim or Priority Tax Claim.

⁶ "Centerbridge" means RD Credit, LLC.

Class 4 – QCP Claims	The QCP Claims ⁷ are deemed Allowed. Each holder of a QCP Claim (or its designee) shall receive on the Effective Date, in full and final satisfaction, release, and discharge of, and in exchange for, its QCP Claim, 100% of the common stock in the reorganized Debtor.	Impaired – Entitled to Vote
Class 5 – General Unsecured Claims	Each holder of an unsecured claim against the Debtor other than a QCP Guarantee Claim or a Severance Claim (a "General Unsecured Claim") will receive, in the sole discretion of the reorganized Debtor, (a) payment in full in cash of such claim (i) on the Effective Date or as promptly thereafter as reasonably practicable or (ii) in the ordinary course of the reorganized Debtor's business or (b) satisfaction of such claim in any other manner that renders the General Unsecured Claim unimpaired or as otherwise agreed by the reorganized Debtor and such holder.	Unimpaired – Deemed to Accept
Class 6 – Severance Claims	Each holder of a claim against the Debtor for damages resulting from the termination of an employment contract (a "Severance Claim) will receive, in the sole discretion of QCP, payment in full in cash of the Allowed amount of such claim (for the avoidance of doubt, subject to applicable limitations on the Allowed amount of such claim under the Bankruptcy Code) (a) on the Effective Date or as soon thereafter as reasonably practicable or (b) as otherwise agreed by the reorganized Debtor and such holder.	Unimpaired – Deemed to Accept
Class 7 – Section 510(b) Claims	On the Effective Date, section 510(b) claims (if any) will be discharged and eliminated, and the holders of any such claims will not receive any distributions or property under the Prepackaged Plan on account of such claims (if any such claims exist).	Impaired – Deemed to Reject Not entitled to any distributions under the Prepackaged Plan

⁷ "<u>OCP Claims</u>" means all claims of QCP and its subsidiaries against the Debtor arising under the Guaranty and due and unpaid as of the Effective Date.

Class 8 – Equity Interests	On the Effective Date, all existing common and preferred shares, options, warrants, rights, similar instruments, and other prepetition equity interests in the Debtor will be extinguished without payment. Holders of such interests will not retain any such interests from or after the Effective Date or receive any distributions or property under the Prepackaged Plan on account of such interests.	Impaired – Deemed to Reject Not entitled to any distributions under the Prepackaged Plan
----------------------------	---	--

Other Provisions		
Executory Contracts / Unexpired Leases	All executory contracts and unexpired leases to which the Debtor is a party and are otherwise susceptible to assumption/assignment will be assumed on the Effective Date pursuant to sections 365 and 1123 of the Bankruptcy Code other than any executory contract or unexpired lease to which the Debtor is a party and which is rejected by the Debtor in accordance with the PSA or with the prior written consent of QCP (in its sole discretion).	
Bar Date	None.	
Vesting of Assets Free and Clear of Liens and Claims	As of the Effective Date, all property of the Debtor shall vest in the reorganized Debtor free and clear of all liens, claims, charges or other encumbrances or interests, <u>provided</u> , <u>however</u> , that any liens (a) held by Centerbridge against the assets of the Debtor securing the Centerbridge Claim and (b) held by HCR Home Health Care and Hospice, LLC securing the \$25 million Subordinated Secured Demand Note, dated March [•], 2018, made by the Debtor to HCR Home Health Care and Hospice, LLC, in each case, shall survive the Effective Date.	
New Master Lease	On the Effective Date, HCR III and the Lessor will amend the Master Lease in accordance with the PSA. On or prior to the seventh day after the Effective Date, the Guaranty will be terminated and released in accordance with the PSA.	
Claim Allowance Process	The Debtor shall have the right to object to or otherwise agree to the Allowance of all claims (other than QCP Claims) in accordance with its duties as debtor-in-possession, subject to QCP's right to object to such Allowance.	
Excess Severance	Subject to Section 1.5(b) of the PSA, (a) on the Effective Date, QCP shall establish an account for the purpose of funding Excess Severance Payments (the "Severance Account") and, following the Closing, shall cause the Severance Account to be funded with all amounts necessary to make Excess Severance Payments and (b) each of the current CEO, CFO, GC and COO of the Debtor (an "Eligible Employee"), shall, unless otherwise agreed in writing between the reorganized Debtor and such person, receive from the Severance Account such Eligible Employee's Excess Severance Payment when due and payable in accordance with the terms of the applicable Separation Agreement.	

Tax Treatment of the Restructuring	The restructuring of the Debtor is intended to qualify under section 382(l)(5) of the Internal Revenue Code of 1986, as amended, in order to preserve the net operating losses and other tax attributes available to the Debtor.
Exemption from Registration	The issuance of common stock of the reorganized Debtor will be exempt from SEC registration under applicable law.
Governance of the Reorganized Debtor	Board of Directors and officers to be determined by QCP.
Releases and Exculpation	Customary releases and exculpation by (x) the Debtor and (y) holders of claims and interests in the Debtor of: (a) the Debtor and the reorganized Debtor, (b) the current and former directors, officers, equity holders, controlling persons, employees, agents, attorneys, financial advisors, restructuring advisors, investment bankers, accountants, and other professional representatives of the Debtor and the reorganized Debtor, in their capacities as such, (c) QCP, and (d) with respect to each entity named in the preceding (a) through (c), each such entity's directors, officers, members, managers, equity holders, general or limited partners, controlling persons, employees, agents, affiliates, parents, subsidiaries, predecessors, successors, heirs, executors and assigns, attorneys, financial advisors, restructuring advisors, investment bankers, accountants and other professionals or representatives when acting in any such capacities, in each case, other than direct or indirect subsidiaries of the Debtor.
Conditions to the Effective Date	 The Confirmation Order shall have become a Final Order; All conditions precedent to the consummation of the transactions contemplated by the PSA (other than the occurrence of the Effective Date) shall have been satisfied; The PSA shall not have terminated in accordance with its terms; and All other actions necessary to implement the terms of the Prepackaged Plan on the Effective Date shall have been taken.

RESTRUCTURING SUPPORT AGREEMENT

This RESTRUCTURING SUPPORT AGREEMENT (this "<u>Agreement</u>") is made and entered into as of March 2, 2018, by and among (i) HCR ManorCare, Inc., a Delaware corporation (the "<u>Debtor</u>"), (ii) Carlyle MC Partners, L.P., a Delaware limited partnership, Carlyle Partners V-A MC, L.P., a Delaware limited partnership, Carlyle Partners V MC, L.P., a Delaware limited partnership, CP V Coinvestment A, L.P., a Delaware limited partnership, and CP V Coinvestment B, L.P., a Delaware limited partnership (collectively, the "<u>Majority Holders</u>"), and (iii) MC Operations Investments, LLC (the "<u>OCP Holder</u>," together with the Majority Holders, the "<u>Parties</u>").

RECITALS

WHEREAS, the Parties have engaged in arm's length, good faith discussions regarding a restructuring of the Debtor and certain of its subsidiaries (the "Restructuring");

WHEREAS, on the date hereof, the Debtor, Quality Care Properties, Inc. ("QCP"), and certain of QCP's direct and indirect subsidiaries have entered into a Plan Sponsor Agreement (the "PSA"), which, among other things, contemplates the acquisition by a subsidiary of QCP of 100% of the equity of the reorganized Debtor through a prepackaged Plan in form and substance consistent with the term sheet attached as Exhibit 1 hereto (the "Plan Term Sheet"); 1 and

WHEREAS, the Supporting Parties intend to support the Restructuring and the prepackaged Plan upon the terms and conditions set forth herein:

AGREEMENT

1. Representations and Warranties

- a. <u>Representations and Warranties of the Debtor</u>. The Debtor represents and warrants to the Supporting Parties that, as of the date hereof:
 - i. It has all requisite power and authority to enter into this Agreement and to carry out the transactions contemplated by, and perform its obligations under, this Agreement.
 - ii. The execution and delivery of this Agreement and the performance of its obligations hereunder have been duly authorized by all necessary action on its part.
 - iii. Subject to the provisions of sections 1125 and 1126 of the Bankruptcy Code, and except as set forth herein, this Agreement is the legally valid and binding obligation of it, enforceable against it in accordance with its terms, except as enforcement may be limited by applicable laws relating to or limiting creditors' rights generally or by equitable principles relating to enforceability.

SC1:4573100.7

-

Capitalized terms not otherwise defined in this Agreement shall have the meaning given to such terms in the PSA.

- b. <u>Representations and Warranties of the Supporting Parties.</u> Each Supporting Party represents and warrants to the Debtor and to each other Supporting Party that, as of the date hereof:
 - i. Such Supporting Party (A) is the sole beneficial owner of the shares set forth on its signature page hereto ("<u>Debtor Shares</u>") and (B) has full power and authority to bind or act on behalf of, vote and consent to matters concerning such shares and to dispose of, exchange, assign and transfer such shares.
 - ii. Such Supporting Party has made no prior assignment, sale, participation, grant, conveyance or other transfer of, and has not entered into any other agreement to assign, sell, participate, grant, convey or otherwise transfer, in whole or in part, any portion of its right, title or interests in any of its shares that are subject to this Agreement that conflict with the representations and warranties of such Supporting Party herein or would render such Supporting Party otherwise unable to comply with this Agreement and perform its obligations hereunder.
 - iii. This Agreement is the legally valid and binding obligation of each such Supporting Party hereto, enforceable against it in accordance with its terms.

2. Covenants of the Supporting Parties

Subject to the terms and conditions hereof and for so long as this Agreement has not been terminated in accordance with its terms, and unless compliance is waived in writing by the Debtor and each Supporting Party, each Supporting Party, severally (and not jointly), agrees to, and to cause its direct and indirect subsidiaries and affiliates to:

- a. support, and take all reasonable actions necessary or reasonably requested by the Debtor to facilitate the solicitation, confirmation and consummation of the Plan and the transactions contemplated by the Plan, including, without limitation, voting in favor of the Plan;
- b. not take any other action, directly or indirectly, that could prevent, interfere with, delay or impede the solicitation of votes in connection with the Plan or the confirmation or consummation of the Plan;
- not object to or otherwise commence, or encourage any other person to commence, any proceeding or take any action opposing the Plan or Disclosure Statement; and
- d. not file any federal or state tax return, or any amendment to such a return, claiming any deduction for worthlessness of its Debtor Shares.

3. Information about the Majority Holders

Subject to the terms and conditions hereof and for so long as this Agreement has not been terminated in accordance with its terms, and unless compliance is waived in writing by the Debtor and QCP, each Majority Holder, severally (and not jointly), agrees to:

- a. use their reasonable best efforts to provide such information requested in writing by Parent or the Debtor as is necessary regarding the Majority Holders for the Purchaser Entities to make or obtain the Governmental Approvals; and
- b. upon the reasonable written request by the Debtor or QCP, use their reasonable best efforts to furnish all information concerning such Majority Holders in connection with any statement, filing, notice or application made by or on behalf of a Purchaser Entity or the Debtor or any of their respective subsidiaries to any Governmental Entity in connection with making or obtaining the Governmental Approvals.

4. Transfer of Shares

Each Supporting Party agrees that so long as this Agreement has not been terminated in accordance with its terms it shall not directly or indirectly sell, pledge, hypothecate or otherwise transfer or dispose of or grant, issue or sell any option, right to acquire, voting, participation or other interest in any Debtor Shares (each a "Transfer"), unless the transferee thereof, prior to such Transfer, agrees in writing for the benefit of the Parties to become subject to the terms and conditions of this Agreement as a "Supporting Party" and to be bound by this Agreement by executing the joinder attached hereto as Exhibit B (the "Joinder **Agreement**"), and delivering an executed copy thereof, within two (2) business days of such execution, to the Debtor, in which event (i) the transferee shall be deemed to be a Supporting Party hereunder and (ii) the transferor shall be deemed to relinquish its rights and be released from its obligations under this Agreement to the extent of such transferred rights and obligations. Each Supporting Party agrees that any Transfer that does not comply with the foregoing shall be deemed void ab initio, and the Debtor and each other Supporting Party shall have the right to avoid such Transfer. This Agreement shall in no way be construed to preclude any shareholder from acquiring additional shares; provided that any such additional shares shall, upon acquisition, automatically be deemed to be subject to all the terms of this Agreement.

5. Additional Supporting Parties

A shareholder that is not a Supporting Party as of the date of this Agreement will become a Party to this Agreement as a Supporting Party on the date that it agrees in writing, for the benefit of the Parties, to become subject to the terms and conditions of this Agreement as a "Supporting Party" and to be bound by this Agreement by executing the Joinder Agreement and delivering an executed copy thereof, within two (2) business days of such execution, to the Debtor. Upon such delivery, such Supporting Party shall immediately thereafter send a copy of such Joinder Agreement to all other Supporting Parties.

6. Termination

This Agreement and all obligations of the Parties hereunder shall immediately terminate and be of no further force and effect as follows:

- a. upon the Effective Date of the Plan; and
- b. upon the termination of PSA.

Upon termination of this Agreement in accordance with its terms, this Agreement shall forthwith become void and of no further force or effect, each Party shall be released from its commitments, undertakings and agreements under or related to this Agreement, and there shall be no liability or obligation on the part of any Party.

7. No Monetary Liability

Notwithstanding anything to the contrary contained in this Agreement or provided for under any applicable law, no Party shall be liable to any other person, either in contract or in tort, for any money damages relating to any breach of this Agreement.

8. Specific Performance

It is understood and agreed by the Parties that money damages would not be a sufficient or appropriate remedy for any breach of this Agreement by any Party and each non-breaching Party shall be entitled to seek specific performance and injunctive or other equitable relief as a remedy of any such breach. Each Party agrees to waive any requirement for the securing or posting of a bond in connection with such remedy.

9. Entire Agreement; Prior Negotiations

This Agreement, including all exhibits attached hereto, constitutes the entire agreement of the Parties and supersedes all prior negotiations and documents reflecting such prior negotiations between and among the Parties (and their respective advisors) with respect to the subject matter of this Agreement.

10. Amendments

Except as otherwise provided herein, this Agreement may not be modified, amended or supplemented without prior written agreement signed by the Debtor and each Supporting Party.

11. Governing Law

This Agreement shall be governed by, and construed in accordance with, the internal laws of the State of Delaware. By its execution and delivery of this Agreement, each of the parties hereto hereby irrevocably and unconditionally agrees for itself that any legal action, suit or proceeding against it with respect to any matter under or arising out of or in connection with this Agreement or for recognition or enforcement of any judgment rendered in any such action, suit or proceeding, may be brought in either a state or federal court of competent jurisdiction in the State of Delaware. By execution and delivery of this Agreement, each of the parties hereto hereby irrevocably accepts and submits itself to the nonexclusive jurisdiction of each such court, generally and unconditionally, with respect to any such action, suit or proceeding. Notwithstanding the foregoing, upon the commencement of the Bankruptcy Case, each of the parties hereto hereby agrees that, if the petition has been filed and the Bankruptcy Case is pending, the Bankruptcy Court shall have exclusive jurisdiction of all matters arising out of or in connection with this Agreement.

12. Effective Date; Conditions to Effectiveness

This Agreement shall become effective and binding upon each of the Parties upon the execution and delivery of this Agreement by each Party hereto.

13. No Solicitation

Notwithstanding anything to the contrary, this Agreement is not and shall not be deemed to be (a) a solicitation of consents to the Plan or (b) an offer for the issuance, purchase, sale exchange, hypothecation or other transfer of securities or a solicitation of an offer to purchase or otherwise acquire securities for purposes of the Securities Act of 1933 and the Securities Exchange Act of 1934, each as amended.

14. Third-Party Beneficiary

This Agreement is intended for the benefit of the parties hereto and no other person shall have any rights hereunder.

15. Counterparts

This Agreement may be executed in several counterparts, each of which shall be deemed to be an original, and all of which together shall be deemed to be one and the same agreement. Execution copies of this agreement may be delivered by electronic mail (in ".pdf" or ".tif" format), facsimile or otherwise, which shall be deemed to be an original for the purposes of this Agreement.

16. Settlement Discussions

This Agreement and the Plan Term Sheet are part of a proposed settlement of matters that could otherwise be the subject of litigation among the parties hereto. Nothing herein shall be deemed an admission of any kind. Pursuant to Federal Rule of Evidence 408 and any applicable state rules of evidence, this Agreement and all negotiations relating thereto shall not be admissible into evidence in any proceeding other than a proceeding to enforce the terms of this Agreement.

17. No Waiver of Participation and Preservation of Rights

Except as provided in this Agreement, nothing herein is intended to, does or shall be deemed in any manner to waive, limit, impair or restrict the ability of any Party to protect and preserve its rights, remedies and interests, including, but not limited to, claims against the Debtor, liens or security interests it may have in any assets of the Debtor, or its rights to participate fully in the Bankruptcy Case. Without limiting the foregoing sentence in any way, if this Agreement is terminated in accordance with its terms for any reason, the Parties each fully reserve any and all of their respective rights, remedies and interests.

18. **Notices**

All notices hereunder shall be deemed given if in writing and delivered, if sent by facsimile, courier or by registered or certified mail (return receipt requested) to the following addresses and facsimile numbers (or at such other addresses or facsimile numbers as shall be specified by like notice):

If to the Debtor, to counsel at the following address:

Sidley Austin LLP One South Dearborn Chicago, IL 60603 Attention: Larry Nyhan Email: lnyhan@sidley.com

If to the QCP Holder, to counsel at the following address:

Sullivan & Cromwell LLP 125 Broad Street New York, NY 10004-2498 Attn: Andrew G. Dietderich

Fax: (212) 558-3588

Email: dietdericha@sullcrom.com

If to the Majority Holders, to counsel at the following address:

Latham & Watkins LLP 555 Eleventh Street, NW, Suite 1000 Washington, D.C. 20004-1304 Attention: Daniel T. Lennon Roger G. Schwartz

J. Cory Tull

Email: daniel.lennon@lw.com roger.schwartz@lw.com cory.tull@lw.com

[Signature Pages Follow]

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 325 of 441

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed and delivered by their respective duly authorized officers or other agents, solely in their respective capacity as officers or other agents of the undersigned and not in any other capacity, as of the date first set forth above.

HCR MANORCARE, INC.

By:	 	
Name:		
Title:		

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 326 of 441

MC OPERATIONS INVESTME LLC	ENTS,
Ву:	
Name: Title:	
Debtor Shares:	

Ву:		
Name: Title:		
Debtor S	hares:	
CARLY	LE PARTNERS V-A MC, I	L .F
Ву:		
Name: Title:		
Debtor S		
	hares: LE PARTNERS V MC, L.F	
CARLY		
CARLY By: Name: Title:	LE PARTNERS V MC, L.F	
CARLY By: Name: Title: Debtor S	LE PARTNERS V MC, L.F	
CARLY By: Name: Title: Debtor S CP V CO	LE PARTNERS V MC, L.F	
CARLY By: Name: Title: Debtor S CP V CO	LE PARTNERS V MC, L.F.	

CARLYLE MC PARTNERS, L.P.

[Signature Page to Restructuring Support Agreement]

CDV	COINI	ESTM	FNT	D	T D
	CUIN	COLIVI		D.	L.F.

Бу:	
Name: Title:	
Debtor Shares:	

EXHIBIT A

Plan Term Sheet

EXHIBIT B

Joinder Agreement

This Joinder Agreement to the Restructuring Support Agreement, dated as of [•], 2018, by and among (a) HCR Manorcare, Inc. (the "Debtor"), (b) MC Operations Investments, LLC (the "QCP Holder") and (c) Carlyle MC Partners, L.P., a Delaware limited partnership, Carlyle Partners V-A MC, L.P., a Delaware limited partnership, Carlyle Partners V MC, L.P., a Delaware limited partnership, and CP V Coinvestment B, L.P., a Delaware limited partnership (collectively, the "Majority Holders") (as set forth in Annex I hereto, the "Restructuring Support Agreement"), is executed and delivered by [______] (the "Joining Supporting Party") as of [_____], 2018. Each capitalized term used herein but not otherwise defined shall have the meanings set forth in the Restructuring Support Agreement.

- Agreement to be Bound. The Joining Supporting Party hereby agrees to be bound by all of the terms of the Restructuring Support Agreement (as the same may be hereafter amended, restated or otherwise modified from time to time). The Joining Supporting Party shall hereafter be deemed to be a "Supporting Party" and a Party for all purposes under the Restructuring Support Agreement.
- 2. <u>Representations and Warranties</u>. With respect to the aggregate principal amount of shares held by the Joining Supporting Party upon consummation of the Transfer of such shares, the Joining Supporting Party hereby makes the representations and warranties of the Supporting Parties set forth in Section 1 of the Restructuring Support Agreement to each of the other Parties.
- 3. <u>Governing Law</u>. This Joinder Agreement shall be governed by and construed in accordance with the internal laws of the State of Delaware, without regard to any conflicts of law provisions which would require the application of the law of any other jurisdiction.

* * * * *

[SIGNATURE PAGE FOLLOWS]

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 331 of 441

IN WITNESS WHEREOF, the Joining Supporting Party has caused this Joinder Agreement to be executed as of the date first written above.

	Name:		
	Title:		
tice	Address and E-	-mail:	

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 332 of 441

Annex 1 to Joinder Agreement

Restructuring Support Agreement

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 333 of 441

CONFIDENTIAL FOR DISCUSSION AND SETTLEMENT PURPOSES ONLY SUBJECT TO FRE 408 AND STATE LAW ANALOGUES

Exhibit 3

[Address]

[•], 2018

Dear Mr. Ormond,

Reference is made to (i) the Plan Sponsor Agreement, dated as of [•], 2018, among Quality Care Properties, Inc. ("QCP"), HCR ManorCare, Inc. (the "Company"), and certain other parties (together with its related exhibits, the "PSA") (ii) the Amended and Restated Employment Agreement between you and the Company, dated as of April 7, 2011 (as amended, the "Employment Agreement"), (iii) the HCR ManorCare Senior Management Savings Plan for Corporate Officers (as amended and restated, the "SMSPCO") and (iv) that certain Letter Agreement between you and the Company regarding your employment and compensation arrangements with the Company and its subsidiaries, dated as of April 7, 2011 (the "Letter Agreement"). The Employment Agreement, the SMSPCO and the Letter Agreement are collectively referred to herein as the "Documents."

This agreement (the "Settlement Letter") is intended to resolve any outstanding disputes between yourself and the Company regarding amounts that you have claimed are owed to you by the Company or its subsidiaries in connection with your employment by the Company. As to the matters described herein and except as otherwise provided herein, this Settlement Letter supersedes any provision contained in any of the Documents or any other written or unwritten agreement, plan or arrangement of the Company or any of its subsidiaries. By countersigning this Settlement Letter in the space provided below, you indicate your acknowledgement of, agreement with and consent to the following:

- 1. Effectiveness of Settlement Letter Conditioned Upon Closing. The parties agree that the effectiveness of this Settlement Letter shall occur on the occurrence of the Closing (as such term is defined in the PSA). If the PSA terminates prior to the Closing, this Settlement Letter shall be void *ab initio* and be of no further force or effect.
- **2. Settlement Amounts.** On the later of (a) the Effective Date, and (b) the Closing Date (as such term is defined in the PSA), you shall be entitled to the following amounts and payments (the "Settlement Amounts"):
- A. In respect of your claimed interest in the Frozen SERP (as such term is defined in the Letter Agreement) you will be entitled to receive your full benefit under the Frozen SERP, which was approximately \$60,763,043 on January 31, 2018 and is subject to change based on investment changes as provided in the Letter Agreement;

- B. In respect of your claimed interest in the SMSPCO, you will be entitled to receive your full benefit under the SMSPCO, which was approximately \$42,040,673 on January 31, 2018 and subject to change based on investment changes as provided in the SMSPCO.
- C. In respect of your claimed interest in the SERP Replacement (as such term is defined in the Letter Agreement), you will be entitled to receive a cash payment equal to \$9,875,000.
- D. In respect of your claimed entitlement to severance under the Employment Agreement, you will be entitled to a cash payment in the amount of \$3,978,000.
- E. Continued payment of your medical, dental and vision premiums during the Continuation Period, as provided in your Employment Agreement.
- F. Any 280G Gross-Up Payment as provided in Section 8 of your Employment Agreement.
- G. Continued use of office space, furnishings and secretarial support services during the Continuation Period, as provided in your Employment Agreement, at a reasonable cost to the Company; provided that such accommodations shall be provided at a location other than the premises of the Company that has a reasonable commute to your primary residence (which location shall be subject to prior notice and consultation with you) and shall be provided at a reasonably comparable quality to such accommodations as were made available to you prior to your termination of employment with the Company.

3. Release.

You knowingly and voluntarily waive, terminate, cancel, release and discharge forever the Company and each of its subsidiaries, as well as their respective current and former stockholders (including, for the avoidance of doubt, QCP), beneficial owners of their stock, their current or former officers, directors, employees, members, attorneys and agents, and their predecessors, successors, and assigns, individually and in their official capacities (together, the "Company Released Parties") from any and all suits, actions, causes of action, claims, allegations, rights, obligations, liabilities, demands, entitlements or charges (collectively, "Claims") that you (or your heirs, executors, administrators, successors and assigns) has or may have, whether known, unknown or unforeseen, vested or contingent, by reason of any matter, cause or thing occurring at any time before and including the date of this Settlement Letter, including all claims arising under or in connection with your employment or termination of employment with the Company, including, without limitation: Claims under United States federal, state or local law and the national or local law of any foreign country (statutory or decisional), for wrongful, abusive, constructive or unlawful discharge or dismissal, for breach of any contract, or for discrimination based upon race, color, ethnicity, sex, age, national origin, religion, disability, sexual orientation, or any other unlawful criterion or circumstance, including rights or Claims under the Age Discrimination in Employment Act of 1967 ("ADEA"), the Older Workers Benefit Protection Act of 1990 ("OWBPA"), violations of the Equal Pay Act, Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1991, the Americans with Disabilities Act of 1991, the Employee Retirement Income Security Act of 1974 ("ERISA"), the Fair Labor

Standards Act, the Worker Adjustment Retraining and Notification Act, the Family Medical Leave Act, including all amendments to any of the aforementioned acts; and violations of any other federal, state, or municipal fair employment statutes or laws, including, without limitation, violations of any other law, rule, regulation, or ordinance pertaining to employment, wages, compensation, hours worked, or any other Claims for compensation or bonuses, whether or not paid under any compensation plan or arrangement; breach of contract; tort and other common law Claims; defamation; libel; slander; impairment of economic opportunity defamation; sexual harassment; retaliation; attorneys' fees; emotional distress; intentional infliction of emotional distress; assault; battery, pain and suffering; and punitive or exemplary damages (the "Ormond Released Matters"). In addition, in consideration of the provisions of this Settlement Letter, you further agree to waive any and all rights under the laws of any jurisdiction in the United States, or any other country, that limit a general release to those Claims that are known or suspected to exist in your favor as of the date of this Settlement Letter, except for the Ormond Unreleased Claims. You further represent and warrant that you have not filed any civil action, suit, arbitration, administrative charge, or legal proceeding against any Company Released Party nor, have you assigned, pledged, or hypothecated as of the Effective Date any Claim to any person and no other person has an interest in the Claims that you are releasing. You acknowledge and agree that you have read this Settlement Letter in its entirety and that this Section 3.A is a general release of all known and unknown Claims. You further acknowledge and agree that:

- (i) this Release does not release, waive or discharge any rights or Claims that may arise for actions or omissions after the Effective Date and you acknowledge that you are not releasing, waiving or discharging any ADEA Claims that may arise after the Effective Date;
- you are entering into this Settlement Letter and releasing, waiving and discharging rights or Claims in exchange for consideration that is presently disputed by the Company and other good and valuable consideration;
- (iii) you have been advised, and are being advised by the Settlement Letter, to consult with an attorney before executing the Settlement Letter; you acknowledge that you have consulted with counsel of his choice concerning the terms and conditions of this Settlement Letter;
- (iv) you have been advised, and are being advised by this Settlement Letter, that you have been given at least twenty-one (21) days within which to consider the Settlement Letter, but you can execute this Settlement Letter at any time prior to the expiration of such review period; and
- (v) you are aware that this Settlement Letter shall become null and void if you revoke your agreement to this Settlement Letter within seven (7) days following the date of execution of this Settlement Letter. You may revoke this Settlement Letter at any time during such seven-day period by delivering (or causing to be delivered) to the Company written notice of your revocation of this Settlement Letter no later than 5:00 p.m. Eastern time on the seventh (7th) full day following the date of execution of this

Release (the "<u>Effective Date</u>"). You acknowledge that a letter of revocation that is not received by such date and time will be invalid and will not revoke this Release.

- B. Ormond Surviving Claims. Notwithstanding anything herein to the contrary, by executing this Settlement Letter you do not: (i) release any Claim that may not lawfully be waived; (ii) become subject to any prohibition from reporting possible violations of federal law or regulation or making other disclosures that are protected under (or claiming any award under) the whistleblower provisions of federal law or regulation; or (iii) release any Claim to enforce this Settlement Letter (collectively, the "Ormond Unreleased Claims").
- Company and QCP Release. The Company and QCP on behalf of themselves and each of their subsidiaries and any of its or their successors and assigns, hereby irrevocably and unconditionally waives and releases any Claims, both known and unknown, in law or in equity, which the Company, QCP or any of their subsidiaries or affiliates ever had, now has or may have against you, your family members and their respective past or present successors, heirs, assigns, or agents (collectively, the "Ormond Released Party"), including, but not limited to, claims that in any way relate to: (1) the your employment with the Company; (2) any claims to attorneys' fees or other indemnities; and (3) any federal, state, local or foreign laws and regulations. In addition, in consideration of the provisions of this Settlement Letter, the Company and QCP further agree to waive any and all rights under the laws of any jurisdiction in the United States, or any other country, that limit a general release to those Claims that are known or suspected to exist in its favor as of the date of this Settlement Letter, except for the Company/QCP Unreleased Claims. The Company and QCP further represent and warrant that they have not filed any civil action, suit, arbitration, administrative charge, or legal proceeding against any Ormond Released Party nor has the Company or QCP assigned, pledged, or hypothecated as of the Effective Date any Claim to any person and no other person has an interest in the Claims that the Company or QCP is releasing.
- D. <u>Company and QCP Surviving Claims</u>. Notwithstanding anything herein in to the contrary, neither the Company nor QCP waives or releases claims with respect to (i) any Claim that may not lawfully be waived, (ii) any right or claim that arises after the date this Settlement Letter is executed, or (iii) any Claim to enforce this Settlement Letter (the "<u>Company/QCP Unreleased Claims</u>").
- **4. Restrictive Covenants.** You agree that you will continue to be subject to Section 10 of the Employment Agreement (Non-Competition/Non-Solicitation and Confidentiality) for the applicable periods set forth therein.
- **5. Non-Disparagement**. You agree not to make any defamatory or derogatory statements concerning the Company, QCP or any of their respective affiliates or predecessors or their respective directors, officers and employees. The Company and QCP agree to instruct their executive officers and directors not to make any defamatory restrict or limit you, the Company, QCP or any of the Company's or QCP's executive officers or directors from providing truthful information in response to a subpoena, other legal process or to a governmental or regulatory body or in the event of litigation between you and the Company, QCP or their respective affiliates.

- **6. Acknowledgement**. You acknowledge that you have read and understand this Settlement Letter, are fully aware of its terms and legal effect and have not acted in reliance upon any representations or promises made by the Company or any other party other than those contained in writing herein. You are advised to consult your personal tax and legal advisors in connection with the matters described in this Settlement Letter, and by countersigning this Settlement Letter in the space provided below, you represent and warrant that you have given the opportunity to consult with, and have consulted with, such advisors to the extent deemed advisable by you.
- **7. Withholding**. The Company or one of its subsidiaries has the authority to deduct or withhold from any compensation payable to you, whether or not such compensation is described herein, or to require you to remit to the Company, an amount sufficient to satisfy any applicable federal, state, local and foreign taxes (including any FICA obligation) required by law to be withheld with respect to any compensation payable to you as provided herein.
- **8. Entire Agreement**. This Settlement Letter sets forth the entire agreement of the parties hereto in respect of the subject matter contained herein and supersedes all prior agreements, term sheets, promises, covenants, arrangements, communications, representations or warranties, whether oral or written, by any officer, employee or representative of any party hereto in respect of such subject matter. Any other prior agreement of the parties hereto in respect of the subject matter contained herein, including, without limitation, the Documents, is hereby terminated and cancelled.
- Miscellaneous. This Settlement Letter may be executed in counterparts. This Settlement Letter and any related dispute shall be governed by, and construed and interpreted in accordance with, the laws of the State of Delaware applicable to contracts executed in and to be performed in that State. Each of the parties hereto (A) consents to submit itself to the personal jurisdiction of any state or federal court located in the State of Delaware in the event any dispute arises out of this Settlement Letter or any of the transactions contemplated by this Settlement Letter; (B) agrees that it will not attempt to deny or defeat such personal jurisdiction or venue by motion or other request for leave from any such court; and (C) agrees that it will not bring any action relating to this Settlement Letter or any of the transactions contemplated by this Settlement Letter in any court other than such courts sitting in the State of Delaware. EACH OF THE PARTIES HERETO HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS SETTLEMENT LETTER OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY.

[Signature Page Follows]

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 338 of 441

Please indicate your acknowledgement of, agreement with and consent to the foregoing by countersigning the enclosed copy of this Settlement Letter in the space provided below and

returning the same to the Company.

Sincerely,
HCR MANORCARE, INC.
QUALITY CARE PROPERTIES, INC.
I hereby acknowledge and agree and consent to the foregoing as of the date first written above.

[EXECUTIVE]

CONFIDENTIAL

EXHIBIT 5

SEPARATION AGREEMENT, GENERAL RELEASE AND WAIVER OF

CLAIMS (this "*Release*"), in entered into by and between [●] ("*Employee*"), Quality Care Properties, Inc., a Maryland corporation ("*QCP*"), and HCR ManorCare, Inc., a Delaware corporation ("*HCRMC*", and together with QCP and their respective subsidiaries, the "*Company*"). The Company and the Company's affiliates, stockholders, beneficial owners of its stock, its current or former officers, directors, employees, members, attorneys and agents, and their predecessors, successors and assigns, individually and in their official capacities shall be referred to herein as the "*Released Parties*".

WHEREAS, Employee has been employed as [title] of HCRMC;

WHEREAS, QCP, HCRMC, and certain other entities have entered into that certain Plan Sponsor Agreement (the "*PSA*") dated as of [•], 2018;

WHEREAS, Employee and HCRMC have entered into that certain Employment Agreement, dated [•] (the "Employment Agreement");

WHEREAS, Employee's employment with the Company shall terminate by the Company without Cause (as defined in the Employment Agreement) as of the Closing (as defined in the PSA) (the "*Termination Date*"), which termination constitutes a Qualifying Termination under the Employment Agreement entitling Employee to "severance after a Change in Control" pursuant to Section 6 of the Employment Agreement; ¹ and

WHEREAS, Section 1.5(a) of the PSA provides that (i) QCP shall fund and pay, or cause to be funded and paid, to the Employee certain Excess Severance Payments (as such term is defined in the PSA) in the amounts set forth in Schedule 1, subject to certain terms and conditions as described therein, including the execution of this Release and (ii) QCP shall pay, or cause to be paid, certain other obligations set forth in Schedule 1.²

NOW, THEREFORE, in consideration of the covenants and agreements hereinafter set forth, the parties agree as follows:

1. General Release. Except as to the Surviving Claims (defined in Section 2 below), Employee knowingly and voluntarily waives, terminates, cancels, releases and discharges forever the Released Parties from any and all suits, actions, causes of action, claims, allegations, rights, obligations, liabilities, demands, entitlements or charges (collectively, "Claims") that Employee (or Employee's heirs, executors, administrators, successors and assigns) has or may have, whether known, unknown or unforeseen, vested or contingent,

US-DOCS\99184296.10

Note: To provide for payments upon the Closing Date, releases will be executed shortly before Closing Date so that non-revocability period has expired as of Closing Date.

Note: Schedule 1 to reflect the Excess Severance Amounts and all other amounts or benefits required to be paid, other than pursuant to ERISA plans.

by reason of any matter, cause or thing occurring at any time before and including the date of this Release, including all claims arising under or in connection with Employee's employment or termination of employment with the Company, including, without limitation: Claims under United States federal, state or local law and the national or local law of any foreign country (statutory or decisional), for wrongful, abusive, constructive or unlawful discharge or dismissal, for breach of any contract, or for discrimination based upon race, color, ethnicity, sex, age, national origin, religion, disability, sexual orientation, or any other unlawful criterion or circumstance, including rights or Claims under the Age Discrimination in Employment Act of 1967 ("ADEA"), the Older Workers Benefit Protection Act of 1990 ("OWBPA"), violations of the Equal Pay Act, Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1991, the Americans with Disabilities Act of 1991, the Employee Retirement Income Security Act of 1974 ("ERISA"), the Fair Labor Standards Act, the Worker Adjustment Retraining and Notification Act, the Family Medical Leave Act, including all amendments to any of the aforementioned acts; and violations of any other federal, state, or municipal fair employment statutes or laws, including, without limitation, violations of any other law, rule, regulation, or ordinance pertaining to employment, wages, compensation, hours worked, or any other Claims for compensation or bonuses, whether or not paid under any compensation plan or arrangement; breach of contract; tort and other common law Claims; defamation; libel; slander; impairment of economic opportunity defamation; sexual harassment; retaliation; attorneys' fees; emotional distress; intentional infliction of emotional distress; assault; battery, pain and suffering; and punitive or exemplary damages (the "Released Matters"). In addition and except as to Surviving Claims, in consideration of the provisions of this Release, Employee further agrees to waive any and all rights under the laws of any jurisdiction in the United States, or any other country, that limit a general release to those Claims that are known or suspected to exist in Employee's favor as of the Effective Date (as defined below).

- 2. <u>Surviving Claims</u>. On the Payment Date (as defined in the PSA), QCP shall pay or cause to be paid to Employee the amounts required to be paid to Employee pursuant to Section 1.5 of the PSA. Notwithstanding anything herein to the contrary, Employee's Release set forth in Section 1 above shall not:
 - (i) waive or release any Claims for payment of amounts or benefits payable as set forth on <u>Schedule 1</u>³ hereto;
 - (ii) waive or release any Claim to enforce this Release;
 - (iii) waive or release any Claim for vested employee benefits under plans covered by ERISA;
 - (iv) waive or release any Claim that may not lawfully be waived;

Note: Schedule 1 to reflect the Excess Severance Amounts and all other amounts or benefits required to be paid (including pursuant to Section 1.5 of the PSA and the Plan), other than pursuant to plans subject to ERISA.

- (v) waive or release Claims described in Section 4(a) of this Release;
- (vi) waive or release any Claim for indemnification, advancement, and D&O insurance, including Claims provided for pursuant to Section 4.7 of the PSA; or
- (vii) limit Employee's rights under applicable law to provide truthful information to any governmental or regulatory entity or to file a charge with or participate in an investigation conducted by any governmental or regulatory entity.

Notwithstanding the foregoing, Employee agrees to waive Employee's right to recover monetary damages in connection with any charge, complaint or lawsuit unrelated to any Claim described in clauses (i), (ii) and (iv) of the immediately preceding sentence filed by Employee or anyone else on Employee's behalf (whether involving a governmental entity or not) with respect to Released Matters; provided that Employee is not agreeing to waive, and this Release shall not be read as requiring Employee to waive, any right Employee may have to receive an award for information provided to any governmental or regulatory entity or respond to a subpoena or other legal requirement.

- 3. <u>Additional Representations</u>. Employee further represents and warrants that Employee has not filed any civil action, suit, arbitration, administrative charge, or legal proceeding against any Released Party nor has Employee assigned, pledged, or hypothecated as of the Effective Date any Claim to any person and no other person has an interest in the Claims that he is releasing.
- **4.** <u>Acknowledgements by Employee</u>. Employee acknowledges and agrees that Employee has read this Release in its entirety and that this Release is a general release of all known and unknown Claims with respect to Released Matters. Employee further acknowledges and agrees that:
 - a. this Release does not release, waive or discharge any rights or Claims that may arise for actions or omissions after the Effective Date of this Release and Employee acknowledges that he is not releasing, waiving or discharging any ADEA Claims that may arise after the Effective Date of this Release;
 - b. Employee is entering into this Release and releasing, waiving and discharging rights or Claims only in exchange for consideration which he is not already entitled to receive;
 - c. Employee's receipt of the Excess Severance Payments is subject to the terms and conditions set forth in Section 1.5 of the PSA (and, for the avoidance of doubt, the non-occurrence of any of the events set forth in Section 1.5(b) of the PSA that by their terms would terminate QCP's obligation to pay such Excess Severance Payments pursuant to Section 1.5(b) of the PSA);

- d. Employee has been advised, and is being advised by the Release, to consult with an attorney before executing this Release; Employee acknowledges that he has consulted with counsel of his choice concerning the terms and conditions of this Release; and
- e. Employee has been advised, and is being advised by this Release, that he has been given at least twenty-one (21) days within which to consider the Release, but Employee can execute this Release at any time prior to the expiration of such review period.
- 5. Effective Date of Release: Employee is aware that Sections 1,2,3,4,6 and 7 of this Release shall become null and void if he revokes his agreement to this Release within seven (7) days following the date of execution of this Release (or, if later, prior to the Closing Date). Employee may revoke this Release (solely with respect to Sections 1,2,3,4,6 and 7 hereof) at any time during such period by delivering (or causing to be delivered) to the Company written notice of his revocation of this Release no later than the later to occur of (i) 5:00 p.m. Eastern time on the seventh (7th) full day following the date of execution of this Release and (ii) the Closing (the "Effective Date"). Employee agrees and acknowledges that a notice of revocation that is not received by such date and time will be invalid and will not revoke this Release with respect to Sections 1 through 4 hereof. The Excess Severance Payments shall be paid to Employee upon the later to occur of the day following the Effective Date and the Termination Date. 4 If Employee fails to execute this Release on or prior to the 21st day following receipt of this Release or exercises his right to revoke Sections 1,2,3,4,6 and 7 of this Release prior to the Effective Date, he shall forfeit his right to receive any portion of the Excess Severance Payments. For the avoidance of doubt, Sections 8-17 of this Release shall survive Employee's revocation of his agreement to this Release in accordance with this Section 5 and the payment of the remaining benefits under Schedule 1 shall not be affected by such revocation.
- 6. Company/QCP Release. The Company and QCP on behalf of themselves and each of their affiliates and any of its or their successors and assigns, hereby irrevocably and unconditionally waives and releases any Claims, both known and unknown, in law or in equity, which the Company, QCP or any of their subsidiaries or affiliates ever had, now has or may have against Employee, Employee's family members and any of their respective past or present successors, heirs, assigns, or agents (collectively, the "Employee Released Party"), including, but not limited to, claims that in any way relate to: (1) the Employee's employment or service as an officer or director with the Company or any of its affiliates; (2) any claims to attorneys' fees or other indemnities; and (3) any federal, state, local or foreign laws and regulations. In addition, in consideration of the provisions of this Release, the Company and QCP further agree on behalf of themselves and each of their affiliates to waive any and all rights under the laws of any jurisdiction in the United States, or any other country, that limit a general release to those Claims that

Note: Releases to be provided in advance of Closing and can be executed 8 days prior to Closing so that Effective Date occurs on or before Closing.

are known or suspected to exist in its favor as of the date of this Release, except for the Company/QCP Unreleased Claims. The Company and QCP further represent and warrant on behalf of themselves and their affiliates that none of them has filed any civil action, suit, arbitration, administrative charge, or legal proceeding against any Employee Released Party nor have any of them assigned, pledged, or hypothecated as of the Termination Date any Claim to any person and no other person has an interest in the Claims that the Company or QCP is releasing. Notwithstanding anything herein in to the contrary, neither the Company nor QCP waives or releases any claims with respect to (i) any Claim that may not lawfully be waived, (ii) any right or claim that arises after the date this Release is executed, or (iii) any Claim to enforce this Release (the "Company/QCP Unreleased Claims").

- 7. Company Acknowledgement. The Company acknowledges and agrees that an authorized representative of the Company has read this Release in its entirety and that this Release is a general release of all known and unknown Claims with respect to the released matters. The Company is entering into this Release and releasing, waiving and discharging rights or Claims only in exchange for consideration which it is not already entitled to receive.
- 8. Officer and Director Resignations. Employee hereby confirms his resignation, effective upon the Termination Date, as a [member of the Board of Directors of the Company] and as an officer of the Company and any other officer or director position Employee held with the Company or any of its subsidiaries, affiliates, joint ventures or other related entities, and agrees to execute such documents as the Company reasonably deems to be appropriate to facilitate such resignations.

9. Cooperation.

- a. For two years following the Termination Date, Employee agrees to make himself reasonably available to the Company upon reasonable advance notice to respond to requests by the Company for information concerning any litigation, regulatory inquiry or investigation, involving facts or events relating to the Company that may reasonably be within his knowledge. Employee will reasonably cooperate with the Company in connection with any and all future litigation or regulatory proceedings brought by or against the Company to the extent the Company reasonably deems Employee's cooperation necessary and such cooperation does not interfere with performance of Employee's duties to Employee's then current employer. Employee will be entitled to reimbursement of reasonable out-of-pocket expenses incurred in connection with fulfilling his obligations under this Section 9(a) and a per hour rate of \$350. This Section 9(a) is subject to Section 2(vii) above.
- b. For two years following the Termination Date, Company agrees to make itself, its appropriate representatives, and Company records and information reasonably available to Employee upon reasonable advance notice to respond to requests by Employee for information concerning any matters involving Employee, including, without limitation, any litigation, governmental or regulatory inquiry or

investigation, involving facts or events relating to Employee that may be eligible for indemnification under Section 4.7 of the PSA.

10. Affirmation of Restrictive Covenants; Return of Property. Employee agrees and acknowledges that his obligations under Section [10] of the Employment Agreement (collectively, the "Restrictive Covenants"), will continue to apply after the Termination Date as specified in such provisions. Employee affirms that such Restrictive Covenants are not unduly burdensome to him and are reasonably necessary to protect the legitimate interests of the Company.

Employee covenants to return to the Company all property or information, including, without limitation, all reports, files, memos, plans, lists, or other records (whether electronically stored or not) belonging to the Company, including copies, extracts or other documents derived from such property or information. The Company agrees to (a) to the extent identified by Employee to QCP prior to Closing, maintain copies of financial and other information disclosed to QCP prior to the Effective Date that Employee reasonably believes may be relevant to the shareholder and derivative actions instituted against QCP and the Company prior to, on or following the Effective Date, and (b) make available to Employee such information to the extent reasonably necessary for Employee to defend himself against actions instituted against QCP or any of its affiliates, the Company or any of its affiliates or Employee.

- 11. Non-Disparagement. Each party (which, in the case of the Company, shall mean the officers and the members of the board of directors of QCP and HCRMC) agrees to refrain from Disparaging (as defined below) the other party and its affiliates, including, in the case of the Company, any of its services, technologies or practices, or any of its directors, officers, agents, representatives or stockholders (including QCP), either orally or in writing. Nothing in this paragraph shall preclude any party from making truthful statements that are reasonably necessary to comply with applicable law, regulation or legal process, to defend or enforce a party's rights under this Release, or that are subject to Section 2(vii) above. For purposes of this Release, "Disparaging" means making remarks, comments or statements, whether written or oral, that materially impugn the character, integrity, reputation or abilities of the person being disparaged with the knowledge on the part of the person making such remarks, comments or statements that such remarks, comments or statements would so materially impugn the character, integrity, reputation or abilities of the person being disparaged.
- 12. <u>Non-Admission</u>. Nothing in this Release is intended to or shall be construed as an admission by the Company or any of the other Released Parties that any of the them violated any law, interfered with any right, breached any obligation or otherwise engaged in any improper or illegal conduct with respect to Employee or otherwise. The Company and the other Released Parties expressly deny any such illegal or wrongful conduct. Nothing in this Release is intended to or shall be construed as an admission by Employee that he violated any law, interfered with any right, breached any obligation or otherwise engaged in any improper or illegal conduct with respect to the Company or otherwise. Employee expressly denies any such illegal or wrongful conduct.

- **13.** <u>Amendment</u>. This Release may be modified only in a written agreement signed by the parties, and any party's failure to enforce this Release in the event of one or more events which violate this Release shall not constitute a waiver of any right to enforce this Release against subsequent violations.
- **14.** Governing Law. To the extent not subject to federal law, this Release shall be governed in accordance with the laws of Delaware without reference to principles of conflict of laws.
- **15.** <u>Severability</u>. If any provision of this Release should be declared to be unenforceable by any administrative agency or court of law, then remainder of the Release shall remain in full force and effect.
- **16.** <u>Captions</u>; <u>Section Headings</u>. Captions and section headings used herein are for convenience only and are not a part of this Release and shall not be used in construing it.
- 17. <u>Counterparts</u>; <u>Facsimile Signatures</u>. This Release may be executed in any number of counterparts, each of which when so executed and delivered shall be deemed an original instrument without the production of any other counterpart. Any signature on this Release, delivered by either party by photographic, facsimile or PDF shall be deemed to be an original signature thereto.

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 346 of 441

IN WITNESS WHEREOF, Employee has signed this Release on this of	
·	
Accepted and Agreed:	
HCR ManorCare, Inc.	
By:	
Accepted and Agreed:	
Quality Care Properties, Inc.	
By:	

Schedule 1

Excess Severance Payments

• Severance pursuant to Section [6(a)(i)] of the Employment Agreement: \$[•]⁵

Other Obligations

- Employee's accrued unpaid wages, employee benefits, and unreimbursed business expenses through and including the Termination Date.
- Employee's participant balance in the Senior Management Savings Plan for Corporate Officers (the "SMSPCO"): \$[●], which is subject to change based on investment changes as provided in the SMSPCO.
- Pay the full premium payments for group medical, dental and vision benefits continuation pursuant to Section [6(a)(ii)] of the Employment Agreement under the Company's group medical, dental and vision plans, which coverage will continue for three years following termination, cover Employee and his eligible dependents and will be primary coverage.
- [Lump-sum amount payable to Employee in connection with SERP: \$[•].]⁶
- KEIP: [●]
- [Retention Payment: \$250,000]⁷

Payment Timing: All payments will be paid to Employee as provided in the PSA.

All payments are subject to withholding of applicable Taxes as required by Law (as such terms are defined in the PSA).

Note: To reflect the applicable amount for each relevant Employee.

Note: Cavanaugh only.

Note: Cavanaugh only.

Exhibit E

Amendment to Original Plan Sponsor Agreement

AMENDMENT TO PLAN SPONSOR AGREEMENT

This AMENDMENT (this "Amendment") to the Plan Sponsor Agreement (defined below) is entered into as of April 25, 2018, by and among HCR ManorCare, Inc., a Delaware corporation (the "Debtor"), Quality Care Properties, Inc., a Maryland corporation ("Parent"), HCP Mezzanine Lender, LP, a Delaware limited partnership and a wholly-owned subsidiary of Parent ("Purchaser"), and the parties signatory hereto as lessors (collectively, the "Lessors" and, together with Parent and Purchaser, the "Purchaser Entities").

RECITALS

- A. The Debtor and the Purchaser Entities entered into that certain Plan Sponsor Agreement, dated as of March 2, 2018 (the "Original PSA").
- B. Contemporaneously with this Amendment, the Debtor, ProMedica Health System, Inc. (the "<u>Alternative Plan Sponsor</u>"), Parent and certain other parties have entered into an Alternative Plan Sponsor Agreement (the "<u>Alternative PSA</u>"), which, among other things, contemplates the acquisition by a subsidiary of the Alternative Plan Sponsor of 100% of the equity of the reorganized Debtor through the *First Amended Chapter 11 Plan of Reorganization for HCR ManorCare, Inc.* in the form attached to the Alternative PSA (the "<u>Amended Plan</u>");
- C. In accordance with Section 8.2 of the Original PSA, the Debtor and the Purchaser Entities desire to amend certain terms of the Original PSA as expressly provided in this Amendment.

AGREEMENT

In consideration of the mutual promises herein contained and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby acknowledge and agree as follows:

- 1. <u>Article I.</u> Article I of the Original PSA is hereby amended by inserting the following new Section 1.8 after Section 1.7 and before Article II:
 - "Section 1.8 <u>Alternative Plan Sponsor Agreement.</u> The parties acknowledge that the Debtor, Parent, ProMedica Health System, Inc. (the "<u>Alternative Plan Sponsor</u>") and certain other parties have entered into an Alternative Plan Sponsor Agreement (the "<u>Alternative PSA</u>"), dated as of April 25, 2018, which, among other things, contemplates the acquisition by a subsidiary of the Alternative Plan Sponsor of 100% of the equity of the reorganized Debtor through the *First Amended Chapter 11 Plan of Reorganization for HCR ManorCare, Inc.* in the form attached to the Alternative PSA (the "<u>Amended Plan</u>") as an alternative to the transactions contemplated by this Agreement and that the Alternative PSA was entered into by the Debtor at the request of the Purchaser Entities (which Purchaser Entities would derive significant benefits from the consummation of the transactions contemplated by the Alternative PSA).

The parties further acknowledge that, at the request of the Purchaser Entities, the Debtor has entered into that certain restructuring support agreement, dated as of April 25, 2018 (the "Alternative RSA"), with Carlyle MC Partners, L.P., a Delaware limited partnership, Carlyle Partners V-A MC, L.P., a Delaware limited partnership, Carlyle Partners V MC, L.P., a Delaware limited partnership, CP V Coinvestment A, L.P., a Delaware limited partnership, and CP V Coinvestment B, L.P., a Delaware limited partnership (collectively, the "Carlyle Holders"), MC Operations Investments, LLC and the Alternative Plan Sponsor. Notwithstanding anything in this Agreement or any other Debtor Document to the contrary, each of the Purchaser Entities hereby consents to (a) the execution, delivery and performance of the Alternative PSA and Alternative RSA by the Debtor (and, with respect to the Alternative RSA, the Carlyle Holders), the filing of the Amended Plan and all motions or pleadings in connection therewith and the consummation of the transactions contemplated thereby and (b) agrees that neither the execution, delivery or performance of the Alternative PSA and Alternative RSA by the Debtor (and, with respect to the Alternative RSA, the Carlyle Holders) nor and the consummation of the transactions contemplated thereby shall constitute, or be deemed in any manner whatsoever to constitute, any breach, default or violation of any provision of this Agreement or any other Debtor Document (including the Restructuring Support Agreement)."

2. <u>Section 4.2(a)</u>. Section 4.2(a) of the Original PSA is hereby amended by inserting the following at the end of such section:

"The parties agree that (x) notwithstanding anything to the contrary in this Section 4.2(a), from and after the execution of the Alternative PSA, neither party shall be required to, and neither party shall (without the consent of the Debtor and Parent), pursue the Governmental Approvals or make any filings with any Governmental Entity in connection with the transactions contemplated by this Agreement until such time (if any) as the condition in Section 6.1(e) shall have been satisfied, and (y) in the event the condition in Section 6.1(e) is satisfied, the foregoing provisions of this Section 4.2(a) shall continue in full force and effect and the parties shall cooperate to obtain the Governmental Approvals and make such other filings with Governmental Entities in accordance with such foregoing provisions of this Section 4.2(a) as promptly as practicable."

- 3. <u>Section 4.10(e)</u>. Section 4.10(e) of the Original PSA is hereby amended by inserting the following after the words "other than" and before the words "the Transactions or the disposition of those Leased Facilities set forth in Section 5.3(b) of the Debtor Disclosure Letter":
 - "(A) the transactions contemplated by the Alternative Plan Sponsor Agreement and (B)"
- 4. <u>Section 6.1</u>. Section 6.1 of the Original PSA is hereby amended by inserting the following after subsection (d) thereof:

- "(e) <u>Termination of Alternative PSA</u>. The Alternative PSA shall have terminated in accordance with its terms (it being understood and agreed that if the Debtor has elected to terminate the Alternative PSA pursuant to Section 7.3 or 7.4 of the Alternative PSA (or has provided written notice to the Alternative Plan Sponsor that the Alternative PSA has terminated automatically in accordance with Section 7.1 of the Alternative PSA, this condition shall be deemed satisfied irrespective of any dispute among any of the parties to the Alternative PSA as to the effectiveness of such termination)."
- 5. <u>Section 6.2(a)</u>. Section 6.2(a) of the Original PSA is hereby amended and restated by deleting in its entirety all of the text of such section and replacing such text with the following:
 - "Representations and Warranties. (i) The representations and warranties of the Debtor (other than the representations and warranties set forth in Sections 2.1(a) (Organization, Good Standing and Qualification), 2.1(c) (Corporate Authority and Approval), 2.1(s) (Brokers and Finders) and the first sentence of Section 2.1(b) (Capital Structure) (the "Fundamental Representations of the **<u>Debtor</u>**")) in this Agreement (without giving effect to any limitations as to "materiality" or "Material Adverse Effect" set forth therein) shall have been true and correct as of the date hereof and true and correct as of June 30, 2018 as though made on and as of June 30, 2018 (except for representations and warranties that are made expressly as of a specific date, which representations and warranties shall have been true and correct as of such date), in each case, except for such failures to be so true and correct as shall not have had, and would not, individually or in the aggregate, reasonably be likely to have a Material Adverse Effect, and (ii) the Fundamental Representations of the Debtor shall be true and correct in all respects as of the date hereof and true and correct in all respects as of the Closing as though made on and as of the Closing (except for representations and warranties that are made expressly as of a specific date, which representations and warranties shall be true and correct as of such date)."
- 6. <u>Section 7.1</u>. Section 7.1 of the Original PSA is hereby amended and restated by deleting the word "or" after subsection (g) thereof and inserting the following before the period in subsection (h) thereof:

"; or

- (i) the transactions contemplated by the Alternative PSA are consummated."
- 7. <u>Section 7.3</u>. Section 7.3 of the Original PSA is hereby amended and restated by replacing "September 30, 2018" with "January 15, 2019".
- 8. <u>Effect of Amendment</u>. This Amendment shall not constitute an amendment or waiver of any provision of the Original PSA not expressly amended or waived herein and shall not be construed as an amendment, waiver or consent to any action that would require an amendment, waiver or consent except as expressly stated

herein. The Original PSA, as amended by this Amendment, is and shall continue to be in full force and effect and is in all respects ratified and confirmed hereby.

- 9. References. On and after the date hereof, each reference in the Original PSA to "this Agreement," "hereunder," "hereof," "herein" or words of like import referring to the Original PSA, and each reference in any other document relating to the "Plan Sponsor Agreement," "Agreement," "thereunder," "thereof" or words of like import referring to the Original PSA, means and references the Original PSA as amended hereby.
- 10. <u>Miscellaneous</u>. The provisions contained in Sections 8.2, 8.4, 8.5, 8.12 and 8.13 of the Original PSA are incorporated by reference in this Amendment *mutatis mutandis*.

[The remainder of this page is intentionally left blank]

IN WITNESS WHEREOF the parties have hereunto caused this Amendment to be duly executed as of the date hereof.

[SIGNATURE PAGE CONTINUES]

IN WITNESS WHEREOF the parties have hereunto caused this Amendment to be duly executed as of the date hereof.

HCR MANORCARE, INC.

By:		
150	Name:	
	Title:	

QUALITY CARE PROPERTIES, INC.

By:

Name: C. Marc Richards Title: Chief Financial Officer

HCP MEZZANINE LENDER, LP

By:

Name: C. Marc Richards Title: Chief Financial Officer

[SIGNATURE PAGE CONTINUES]

"LESSORS"

HCP PROPERTIES, LP, a Delaware limited partnership

By: HCP I-B Properties, LLC, a Delaware limited liability company, its General Partner

HCP WEST VIRGINIA PROPERTIES, LLC, a Delaware limited liability company

HCP PROPERTIES OF ALEXANDRIA VA, LLC, a Delaware limited liability company

HCP PROPERTIES OF ARLINGTON VA, LLC, a Delaware limited liability company

HCP PROPERTIES OF MIDWEST CITY OK, LLC, a Delaware limited liability company

HCP PROPERTIES OF OKLAHOMA CITY (NORTHWEST), LLC, a Delaware limited liability company

HCP PROPERTIES OF OKLAHOMA CITY (SOUTHWEST), LLC, a Delaware limited liability company

HCP PROPERTIES OF TULSA OK, LLC, a Delaware limited liability company

HCP PROPERTIES-ARDEN COURTS OF ANNANDALE VA, LLC, a Delaware limited liability company

HCP PROPERTIES-CHARLESTON OF HANAHAN SC, LLC, a Delaware limited liability company

HCP PROPERTIES-COLUMBIA SC, LLC, a Delaware limited liability company

HCP PROPERTIES-FAIR OAKS OF FAIRFAX VA, LLC, a Delaware limited liability company

HCP PROPERTIES-IMPERIAL OF RICHMOND VA, LLC, a Delaware limited liability company

HCP PROPERTIES-LEXINGTON SC, LLC, a Delaware limited liability company

HCP PROPERTIES-MEDICAL CARE CENTER-LYNCHBURG VA, LLC, a Delaware limited liability company

HCP PROPERTIES-OAKMONT EAST-GREENVILLE SC, LLC, a Delaware limited liability company

HCP PROPERTIES-OAKMONT OF UNION SC, LLC, a Delaware limited liability company

HCP PROPERTIES-OAKMONT WEST-GREENVILLE SC, LLC, a Delaware limited liability company

HCP PROPERTIES-STRATFORD HALL OF RICHMOND VA, LLC, a Delaware limited liability company

HCP PROPERTIES-WEST ASHLEY-CHARLESTON SC, LLC, a Delaware limited liability company

HCP MARYLAND PROPERTIES, LLC, a Delaware limited liability company

HCP PROPERTIES-SALMON CREEK WA, LLC, a Delaware limited liability company

HCP PROPERTIES-WINGFIELD HILLS NV, LLC, a Delaware limited liability company

HCP PROPERTIES-UTICA RIDGE IA, LLC, a Delaware limited liability company

HCP TWINSBURG OH PROPERTY, LLC, a Delaware limited liability company

By: HCR Schoenherr Road Property, LLC, a Delaware limited liability company, its sole member

By: HCP Properties, LP, a Delaware limited partnership, its sole member

By: HCP I-B Properties, LLC, a Delaware limited liability company, its General Partner

By:

Name: C. Marc Richards
Title: Chief Financial Officer

HCP STERLING HEIGHTS MI PROPERTY, LLC, a Delaware limited liability company

By: HCR Schoenherr Road Property, LLC, a Delaware limited liability company, its sole member

By: HCP Properties, LP, a Delaware limited partnership, its sole member

By: HCP I-B Properties, LLC, a Delaware limited liability company, its General Partner

By:

Name: C. Marc Richards
Title: Chief Financial Officer

Exhibit F

Directors and Officers of Reorganized Debtor – Alternative Transaction

Officers of the Reorganized Debtor

President and Chief Executive Officer Steven Cavanaugh

David Parker Executive Vice President and Chief Operating Officer

Vice President and Chief Financial Officer Matthew Kang

Daniel Kite Vice President and Treasurer

Richard Parr Vice President, General Counsel, and Secretary

Vice President and Controller Kathy Hoops

Directors of the Reorganized Debtor

Robert W. LaClair Chairperson, ProMedica Health System, Inc.

> Board Member, ProMedica Health System, Inc. Foundation Board Member, ProMedica Continuum Services Board

Member, PPG Lenawee Long Term Care

Board Member, ProMedica Insurance Corporation

Board Member, Paramount Care, Inc.

Board Member, Paramount Preferred Options, Inc. Board Member, Paramount Benefits Agency

Board Member, Paramount Advantage

Board Member, Paramount Insurance Company

Board Member, Health Resources, Inc. Board Member, Indemnity Corporation

Stephen H. Staelin Board Member, ProMedica Health System, Inc.

Board Member, ProMedica Insurance Corporation

Board Member, Paramount Care, Inc.

Board Member, Paramount Preferred Options, Inc. Board Member, Paramount Benefits Agency Board Member, Paramount Advantage

Board Member, Paramount Insurance Company

Board Member, Health Resources, Inc. Board Member, Indemnity Corporation

James F. White, Jr. Vice Chair, ProMedica Health System, Inc.

Board Member, ProMedica Insurance Corporation

Board Member, Paramount Care, Inc.

Board Member, Paramount Preferred Options, Inc.

Board Member, Paramount Benefits Agency

Board Member, Paramount Advantage

Board Member, Paramount Insurance Company Board Member, Health Resources, Inc.

<u>Identity and Nature of Compensation for Insiders</u>

Steven Cavanaugh, David Parker, Matthew Kang, Daniel Kite, Richard Parr, and Kathy Hoops are insiders of the Debtor who will serve as officers of the Reorganized Debtor. The officers of the Debtor receive salary and benefits and participate in certain key-employee and long-term incentive plans.

Exhibit G

Directors and Officers of the Reorganized Debtor – Original Transaction

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 362 of 441

Directors and Officers of the Reorganized Debtor

Guy Sansone Director

Chief Executive Officer & President

Managing Director, Alvarez & Marsal

Laura Linynsky Director

Chief Financial Officer, Treasurer & Executive Vice President

Senior Vice President,

Quality Care Properties, Inc.

Richard Parr Director

Secretary & Vice President

Identity and Nature of Compensation for Insider

Richard Parr is an insider of the Debtor who will serve as a director of the Reorganized Debtor. Mr. Parr will receive board fees for his service as a director of the Reorganized Debtor.

Exhibit H

Certificate of Incorporation – Original Transaction

FORM OF AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF HCR MANORCARE, INC.

FIRST. The name of the corporation is HCR ManorCare, Inc. (the "Corporation")

SECOND. The address of the Corporation's registered office in the State of Delaware is Corporation Trust Center, 1209 Orange Street in the City of Wilmington, County of New Castle. The name of its registered agent at such address is The Corporation Trust Company.

THIRD. The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of Delaware.

FOURTH. The Corporation is authorized to issue one class of stock, to be designated "Common Stock," with a par value of \$0.01 per share. The total number of shares of Common Stock that the Corporation shall have authority to issue is 1,000,000.

FIFTH. The board of directors of the Corporation is expressly authorized to adopt, amend or repeal by-laws of the Corporation (the "By-laws").

SIXTH. Elections of directors need not be by written ballot except and to the extent provided in the By-laws.

SEVENTH. Any action required or permitted to be taken by the holders of Common Stock, including but not limited to the election of directors, may be taken by written consent or consents but only if such consent or consents are signed by all holders of Common Stock.

EIGHTH. A director of the Corporation shall not be liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except to the extent that such exemption from liability or limitation thereof is not permitted under the Delaware General Corporation Law as currently in effect or as the same may hereafter be amended. No amendment, modification or repeal of this Article EIGHTH shall adversely

affect any right or protection of, or any limitation of the liability of, a director that exists at the time of such amendment, modification or repeal, including any right or protection with respect to acts or omissions occurring prior to such amendment, modification or repeal.

NINTH. To the fullest extent permitted by applicable law, the Corporation is authorized to provide indemnification of (and advancement of expenses to) agents of the Corporation (and any other persons to which the Delaware General Corporation Law permits the corporation to provide indemnification) through by-law provisions or agreements with such agents or other persons, by vote of stockholders or disinterested directors or otherwise, in excess of the indemnification and advancement otherwise permitted by Section 145 of the Delaware General Corporation Law, subject only to limits created by the Delaware General Corporation Law and applicable decisional law, with respect to actions for breach of duty to the corporation, its stockholders, and others.

TENTH. Unless the Corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware (or, if such court does not have jurisdiction, the Superior Court of the State of Delaware, or, if such other court does not have jurisdiction, the United States District Court for the District of Delaware) shall, to the fullest extent permitted by law, be the sole and exclusive forum for (a) any derivative action or proceeding brought on behalf of the Corporation, (b) any action asserting a claim of breach of a fiduciary duty owed by any director, officer or stockholder of the Corporation to the Corporation or the Corporation's stockholders, (c) any action asserting a claim arising pursuant to any provision of the DGCL, this Amended and Restated Certificate of Incorporation (as may be amended, altered, changed or repealed) or the By-laws or (d) any action asserting a claim governed by the internal affairs doctrine.

ELEVENTH. The Corporation shall not issue any class of non-voting equity securities unless and solely to the extent permitted by Section 1123(a)(6) of title 11 of the United States Code (the "Bankruptcy Code") as in effect on the date of filing of this Certificate of Incorporation with the Secretary of State of the State of Delaware; provided, however, that this ELEVENTH (a) will have no further force and effect beyond that required under Section 1123(a)(6) of the Bankruptcy Code, (b) will have such force and effect, if any, only for so long as Section 1123(a)(6) of the Bankruptcy Code is in effect and applicable to

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 366 of 441

the Corporation and (c) in all events may be amended or eliminated from time to time in accordance with applicable law.

IN WITNESS WHEREOF, I have signed this Amended and Restated
Certificate of Incorporation this day of, 2018.
[Authorized Officer]

Exhibit I

By-Laws – Original Transaction

FORM OF AMENDED AND RESTATED BY-LAWS

OF

HCR MANORCARE, INC. (a Delaware corporation)

Adopted as of [____], 2018

TABLE OF CONTENTS

	Pa	age
ARTICLE I. IDEN	NTIFICATION; OFFICES	1
SECTION 1.	NAME	1
SECTION 2.	PRINCIPAL AND BUSINESS OFFICES	
SECTION 3.	REGISTERED AGENT AND OFFICE	
SECTION 4.	PLACE OF KEEPING CORPORATE RECORDS	
	OCKHOLDERS	
	ANNUAL MEETING	
SECTION 1. SECTION 2.	SPECIAL MEETING	
SECTION 2. SECTION 3.	PLACE OF STOCKHOLDER MEETINGS	
SECTION 3. SECTION 4.	NOTICE OF MEETINGS	
SECTION 5.	QUORUMADJOURNED MEETINGS	2
SECTION 6.	FIXING OF RECORD DATE	
SECTION 7. SECTION 8.	VOTING UF RECORD DATE	
SECTION 8. SECTION 9.		
SECTION 9. SECTION 10.	VOTINGPROXIES	
	RATIFICATION OF ACTS OF DIRECTORS AND OFFICERS	
SECTION 11.	CONDUCT OF MEETINGS	
SECTION 12.		
SECTION 13.	ACTION WITHOUT MEETING	
ARTICLE III. DII	RECTORS	6
SECTION 1.	GENERAL POWERS	
SECTION 2.	NUMBER AND TENURE OF DIRECTORS	6
SECTION 3.	ELECTION OF DIRECTORS	
SECTION 4.	CHAIRMAN OF THE BOARD; VICE CHAIRMAN OF THE BOARD	6
SECTION 5.	QUORUM	6
SECTION 6.	VOTING	7
SECTION 7.	VACANCIES	7
SECTION 8.	REMOVAL OF DIRECTORS	7
SECTION 9.	RESIGNATION	
SECTION 10.	REGULAR MEETINGS	7
SECTION 11.	SPECIAL MEETINGS	
SECTION 12.	NOTICE OF SPECIAL MEETINGS OF THE BOARD OF DIRECTORS	7
SECTION 13.	WRITTEN ACTION BY DIRECTORS	
SECTION 14.	PARTICIPATION BY CONFERENCE TELEPHONE	8
SECTION 15.	COMMITTEES	8
SECTION 16.	COMPENSATION OF DIRECTORS	9
ARTICLE IV. OF	FICERS	9
SECTION 1.	GENERAL PROVISIONS	9
SECTION 2.		
SECTION 3.	RESIGNATION AND REMOVAL OF OFFICERS	

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 370 of 441

SECTION 4.	VACANCIES	9
SECTION 5.	THE CHIEF EXECUTIVE OFFICER	10
SECTION 6.	THE CHIEF FINANCIAL OFFICER	10
SECTION 7.	THE PRESIDENT	
SECTION 8.	THE VICE PRESIDENT	10
SECTION 9.	THE SECRETARY	
SECTION 10.	THE ASSISTANT SECRETARY	11
SECTION 11.	THE TREASURER	
SECTION 12.	THE ASSISTANT TREASURER	
SECTION 13.	OTHER OFFICERS, ASSISTANT OFFICERS AND AGENTS	12
SECTION 14.	ABSENCE OF OFFICERS	
SECTION 15.	COMPENSATION	12
ARTICLE V. CAI	PITAL STOCK	12
SECTION 1.	ISSUANCE OF STOCK	
SECTION 2.	CERTIFICATES OF SHARES; UNCERTIFICATED SHARES	
SECTION 3.	SIGNATURES OF FORMER OFFICER, TRANSFER AGENT OR	
CECTION 4	REGISTRARTRANSFER OF SHARES	
SECTION 4.	LOST, DESTROYED OR STOLEN CERTIFICATES	
SECTION 5. SECTION 6.	REGULATIONS	
ARTICLE VI. IN	DEMNIFICATION	14
SECTION 1.	RIGHT TO INDEMNIFICATION OF DIRECTORS, OFFICERS AND EMPLOYEES	
ARTICLE VII. D	IVIDENDS	14
SECTION 1.	DECLARATIONS OF DIVIDENDS	14
SECTION 2.	SPECIAL PURPOSES RESERVES	
	IOTICE BY ELECTRONIC TRANSMISSION	
SECTION 1.	NOTICE BY ELECTRONIC TRANSMISSION	
SECTION 2.	DEFINITION OF ELECTRONIC TRANSMISSION	
SECTION 3.	INAPPLICABILITY	15
ARTICLE IX. GE	NERAL PROVISIONS	15
SECTION 1.	FISCAL YEAR	15
SECTION 2.	SEAL	
SECTION 3.	WRITTEN WAIVER OF NOTICE	
SECTION 4.	ATTENDANCE AS WAIVER OF NOTICE	
SECTION 5.	CONTRACTS	
SECTION 6.	LOANS	
SECTION 7.	CHECKS, DRAFTS, ETC.	
SECTION 8.	DEPOSITS	
SECTION 9.	ANNUAL STATEMENT	
SECTION 10.	VOTING OF SECURITIES.	
SECTION 11.	EVIDENCE OF AUTHORITY	

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 371 of 441

SECTION 12.	CERTIFICATE OF INCORPORATION	16
SECTION 13.	SEVERABILITY	16
SECTION 14.	PRONOUNS	16
ADTICLE V AM	IENDMENTS	17
SECTION 1.	BY THE BOARD OF DIRECTORS	17
SECTION 2.	BY THE STOCKHOLDERS	17

ARTICLE I. IDENTIFICATION; OFFICES

- SECTION 1. NAME. The name of the corporation is HCR ManorCare, Inc. (the "Corporation").
- SECTION 2. PRINCIPAL AND BUSINESS OFFICES. The Corporation may have such principal and other business offices, either within or outside of the state of Delaware, as the Board of Directors may designate or as the Corporation's business may require from time to time.
- SECTION 3. REGISTERED AGENT AND OFFICE. The address of the Corporation's registered office in the State of Delaware is Corporation Trust Center, 1209 Orange Street in the City of Wilmington, County of New Castle. The name of its registered agent at such address is The Corporation Trust Company.
- SECTION 4. PLACE OF KEEPING CORPORATE RECORDS. The records and documents required by law to be kept by the Corporation permanently shall be kept at the Corporation's principal office or as the Board of Directors may designate.

ARTICLE II. STOCKHOLDERS

- SECTION 1. ANNUAL MEETING. An annual meeting of the stockholders shall be held on such date and time as may be designated by the Board of Directors, the Chairman of the Board, the Chief Executive Officer or the President. At each annual meeting, the stockholders shall elect directors to hold office for the term provided in Section 2 of Article III of these Bylaws and transact such other business as may properly be brought before the meeting. The Board of Directors may postpone, cancel or reschedule any previously scheduled annual meeting of stockholders.
- SECTION 2. SPECIAL MEETING. A special meeting of the stockholders for any purpose or purposes may be called at any time only by the President, the Board of Directors, the Chairman of the Board, the Chief Executive Officer or any other person designated by the Board of Directors. The Board of Directors may postpone, cancel or reschedule any previously scheduled special meeting of stockholders. Business transacted at any special meeting of stockholders shall be limited to matters relating to the purpose or purposes stated in the notice of meeting.
- SECTION 3. PLACE OF STOCKHOLDER MEETINGS. The Board of Directors, the Chairman of the Board, the Chief Executive Officer or the President may designate any place, either within or without the State of Delaware, as the place of meeting for any annual meeting or for any special meeting. If no such place is designated by the Board of Directors, the place of meeting will be the principal business office of the Corporation or the Board of Directors may, in its sole discretion, determine that the meeting shall not be held at any place, but will instead be held solely by means of remote communication as provided under Section 211 of the Delaware General Corporation Law.

SECTION 4. NOTICE OF MEETINGS. Except as otherwise provided by law or waived as herein provided, whenever stockholders are required or permitted to take any action at a meeting, whether annual or special, written notice of the meeting shall be given stating the place, if any, date and hour of the meeting, the means of remote communications, if any, by which stockholders may be deemed to be present in person and vote at such meeting and, in the case of a special meeting, the purpose or purposes for which the meeting is called. Unless otherwise required by law, such written notice shall be given not less than 10 days nor more than 60 days before the date of the meeting to each stockholder entitled to vote at the meeting. If mailed, notice is given when deposited in the United States mail, postage prepaid, directed to the stockholder at the stockholder's address as it appears on the records of the Corporation. If electronically transmitted (in a manner consistent with Section 232 of the Delaware General Corporation Law), then notice is deemed given when transmitted and directed to a facsimile number or electronic mail address at which the stockholder has consented to receive notice. An affidavit of the secretary or of the transfer agent or other agent of the Corporation that the notice has been given by a form of electronic transmission shall, in the absence of fraud, be *prima facie* evidence of the facts stated therein.

When a meeting is adjourned to reconvene at the same or another place, if any, or by means of remote communications, if any, in accordance with <u>Section 6</u> of Article II of these Bylaws, notice need not be given of the adjourned meeting if the time and place thereof are announced at the meeting at which the adjournment is taken.

SECTION 5. QUORUM. Unless otherwise provided by law, the Corporation's Amended and Restated Certificate of Incorporation (the "Certificate of Incorporation") or these By-laws, the holders of a majority in voting power of the shares of the capital stock of the Corporation issued and outstanding and entitled to vote at the meeting, present in person, present by means of remote communication in a manner, if any, authorized by the Board of Directors in its sole discretion, or represented by proxy, shall constitute a quorum for the transaction of business. If a quorum is present in person or represented by proxy at such meeting, such stockholders may continue to transact business until adjournment, notwithstanding the withdrawal of such number of stockholders as may leave less than a quorum.

SECTION 6. ADJOURNED MEETINGS. Any meeting of stockholders may be adjourned from time to time to any other time and date and to any other place (or by means of remote communications, if any) at which a meeting of stockholders may be held under these Bylaws by the chairman of the meeting or by a majority of the stockholders present or represented at the meeting and entitled to vote, although less than a quorum. It shall not be necessary to notify any stockholder of any adjournment of less than 30 days if the time and place, if any, of the adjourned meeting, and the means of remote communication, if any, by which stockholders and proxyholders may be deemed to be present in person and vote at such adjourned meeting, are announced at the meeting at which adjournment is taken, unless after the adjournment a new record date is fixed for the adjourned meeting. At the adjourned meeting, the Corporation may transact any business which might have been transacted at the original meeting.

SECTION 7. FIXING OF RECORD DATE.

- (a) The Board of Directors may fix in advance a date as a record date for the determination of the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof. Such record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall not be more than 60 days nor less than 10 days before the date of such meeting. If no record date is fixed by the Board of Directors, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting.
- For the purpose of determining stockholders entitled to consent to (b) corporate action in writing without a meeting, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is established by the Board of Directors, and which date shall not be more than 10 days after the date on which the resolution fixing the record date is adopted by the Board of Directors. If no record date has been fixed by the Board of Directors, the record date for determining stockholders entitled to consent to corporate action in writing without a meeting, when no prior action by the Board of Directors is required by law, shall be the first date on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Corporation by delivery to its registered office in the State of Delaware, its principal office, or an officer or agent of the Corporation having custody of the book in which the proceedings of meetings of stockholders are recorded. Delivery to the Corporation's registered office shall be by hand or by certified or registered mail, return receipt requested. If no record date has been fixed by the Board of Directors and prior action by the Board of Directors is required by law, the record date for determining stockholders' consent to corporate action in writing without a meeting shall be the close of business on the day on which the Board of Directors adopts the resolution taking such prior action.
- (c) For the purpose of determining the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights or the stockholders entitled to exercise any rights in respect to any change, conversion or exchange of stock, or for the purpose of any other lawful action, the Board of Directors may fix the record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than 60 days prior to such action. If no record date is fixed, the record date for determining the stockholders for any such purpose shall be the close of business on the day on which the Board of Directors adopts the resolution relating thereto.
- SECTION 8. VOTING LIST. Corporation shall prepare, at least 10 days before every meeting of stockholders, a complete list of stockholders entitled to vote at the meeting, arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the name of each stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, for a period of at least 10 days prior to the meeting, (a) by a reasonably accessible electronic network, provided that the information required to gain access to such list is provided with the notice of the meeting, or (b) during

ordinary business hours, at the principal place of business of the Corporation. In the event that the Corporation determines to make the list available on an electronic network, the Corporation may take reasonable steps to ensure that such information is available only to the stockholders of the Corporation. If the meeting is to be held at a place, then the list shall be produced and kept at the time and place of the meeting during the whole time thereof and may be inspected by any stockholder who is present. If the meeting is to be held solely by means of remote communication, then the list shall also be open to the examination of any stockholder during the whole time of the meeting on a reasonably accessible electronic network, and the information required to access such list shall be provided with the notice of the meeting. Except as otherwise provided by law, such list shall be the only evidence as to the identity of stockholders entitled to examine the list of stockholders required by this Section 8 or to vote in person or by proxy at any meeting of the stockholders. The Corporation shall not be required to include electronic mail addresses or other electronic contact information on such list.

SECTION 9. VOTING. Unless otherwise provided by the Certificate of Incorporation, each holder of Common Stock shall be entitled to one vote for each share of Common Stock held by each stockholder. When a quorum is present at any meeting, in all matters other than the election of directors, the affirmative vote of the majority in voting power of the outstanding shares of stock present in person or represented by proxy at the meeting and entitled to vote on the subject matter shall be the act of the stockholders, except when a different vote is required by law, the Certificate of Incorporation or these By-laws. When a quorum is present at any meeting, directors shall be elected by plurality of the votes of the shares present in person or represented by a proxy at the meeting entitled to vote on the election of directors.

SECTION 10. PROXIES. Each stockholder entitled to vote at a meeting of stockholders or to express consent or dissent to corporate action in writing without a meeting (including by means of remote communications, if any, by which stockholders may be deemed to be present in person and vote at such meeting) may authorize another person or persons to act for him by proxy (executed or transmitted in a manner permitted by the Delaware General Corporation Law), but no such proxy shall be voted or acted upon after three years from its date, unless the proxy provides for a longer period. A duly executed proxy shall be irrevocable if it states that it is irrevocable and if, and only as long as, it is coupled with an interest sufficient in law to support an irrevocable power. A proxy may remain irrevocable regardless of whether the interest with which it is coupled is an interest in the stock itself or an interest in the Corporation generally.

SECTION 11. RATIFICATION OF ACTS OF DIRECTORS AND OFFICERS. Except as otherwise provided by law or by the Certificate of Incorporation of the Corporation, any transaction or contract or act of the Corporation or of the directors or the officers of the Corporation may be ratified by the affirmative vote of the holders of the number of shares which would have been necessary to approve such transaction, contract or act at a meeting of stockholders, or by the written consent of stockholders in lieu of a meeting.

SECTION 12. CONDUCT OF MEETINGS.

(a) <u>Chairman of Meeting</u>. Meetings of stockholders shall be presided over by the Chairman of the Board, if any, or in the Chairman's absence by the Vice Chairman of the Board, if any, or in the Vice Chairman's absence by the Chief Executive Officer, or in the Chief

Executive Officer's absence, by the President, or in the President's absence by a Vice President, or in the absence of all of the foregoing persons by a chairman designated by the Board of Directors, or in the absence of such designation by a chairman chosen by vote of the stockholders at the meeting. The Secretary shall act as secretary of the meeting, but in the Secretary's absence the chairman of the meeting may appoint any person to act as secretary of the meeting.

Rules, Regulations and Procedures. The Board of Directors may adopt by resolution such rules, regulations and procedures for the conduct of any meeting of stockholders of the Corporation as it shall deem appropriate including, without limitation, such guidelines and procedures as it may deem appropriate regarding the participation by means of remote communication of stockholders and proxyholders not physically present at a meeting. Except to the extent inconsistent with such rules, regulations and procedures as adopted by the Board of Directors, the chairman of any meeting of stockholders shall have the right and authority to convene and (for any or no reason) to recess and/or adjorn the meeting and to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such chairman, are appropriate for the proper conduct of the meeting. Such rules, regulations or procedures, whether adopted by the Board of Directors or prescribed by the chairman of the meeting, may include, without limitation, the following: (i) the establishment of an agenda or order of business for the meeting; (ii) rules and procedures for maintaining order at the meeting and the safety of those present; (iii) limitations on attendance at or participation in the meeting to stockholders of record of the Corporation, their duly authorized and constituted proxies or such other persons as shall be determined; (iv) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (v) limitations on the time allotted to questions or comments by participants. Unless and to the extent determined by the Board of Directors or the chairman of the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure.

SECTION 13. ACTION WITHOUT MEETING.

- (a) Any action required or permitted to be taken at any annual or special meeting of stockholders of the Corporation, may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action so taken, shall be delivered to the Corporation signed by the holders of shares of outstanding Common Stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted.
- (b) Prompt notice of the taking of the corporate action without a meeting by less than unanimous consent shall be given to those stockholders who have not consented in writing and who, if the action had been taken at a meeting, would have been entitled to notice of the meeting if the record date for such meeting had been the date that written consents signed by a sufficient number of holders to take the action were delivered to the Corporation.
- (c) A telegram, cablegram or other electronic transmission consenting to an action to be taken and transmitted by a stockholder or proxy holder, or by a person or persons authorized to act for a stockholder or proxy holder, shall be deemed to be written and signed for the purposes of this section, provided that any such telegram, cablegram or other electronic

transmission sets forth or is delivered with information from which the Corporation can determine (i) that the telegram, cablegram or other electronic transmission was transmitted by the stockholder or proxy holder or by a person or persons authorized to act for the stockholder or proxy holder and (ii) the date on which such stockholder or proxy holder or authorized person or persons transmitted such telegram, cablegram or electronic transmission. No consent given by telegram, cablegram or other electronic transmission shall be deemed to have been delivered until such consent is reproduced in paper form and until such paper form shall be delivered to the Corporation by delivery to its registered office in the State of Delaware, its principal place of business or to an officer or agent of the Corporation having custody of the book in which the proceedings of meetings of stockholders are recorded. Any copy, facsimile or other reliable reproduction of a consent in writing may be substituted or used in lieu of the original writing for any and all purposes for which the original writing could be used, provided that such copy, facsimile or other reproduction shall be a complete reproduction of the entire original writing.

ARTICLE III. DIRECTORS

SECTION 1. GENERAL POWERS. The business and affairs of the Corporation shall be managed by or under the direction of a Board of Directors, who may exercise all of the powers of the Corporation except as otherwise provided by law or the Certificate of Incorporation.

SECTION 2. NUMBER AND TENURE OF DIRECTORS. The number of directors of the Corporation shall be determined from time to time by the stockholders or the Board of Directors in a resolution adopted by the Board of Directors. Each director shall hold office until the next annual meeting of stockholders and until such director's successor is elected and qualified or until such director's earlier death, resignation or removal.

SECTION 3. ELECTION OF DIRECTORS. Except as otherwise provided in these Bylaws, directors shall be elected at the annual meeting of stockholders by such stockholders as have the right to vote on such election. Directors need not be residents of the State of Delaware. Directors need not be stockholders of the Corporation. Elections of directors need not be by written ballot.

SECTION 4. CHAIRMAN OF THE BOARD; VICE CHAIRMAN OF THE BOARD. The Board of Directors may appoint from its members a Chairman of the Board and a Vice Chairman of the Board, neither of whom need be an employee or officer of the Corporation. If the Board of Directors appoints a Chairman of the Board, such Chairman shall perform such duties and possess such powers as are assigned by the Board of Directors. If the Board of Directors appoints a Vice Chairman of the Board, such Vice Chairman shall perform such duties and possess such powers as are assigned by the Board of Directors. Unless otherwise provided by the Board of Directors, the Chairman of the Board or, in the Chairman's absence, the Vice Chairman of the Board, if any, shall preside at all meetings of the Board of Directors.

SECTION 5. QUORUM. The greater of (a) a majority of the directors at any time in office and (b) one-third of the number of directors fixed pursuant to <u>Section 2</u> of Article III of these By-laws shall constitute a quorum of the Board of Directors. If less than a quorum are

present at a meeting of the Board of Directors, a majority of the directors present may adjourn the meeting from time to time without further notice other than announcement at the meeting, until such quorum shall be present.

SECTION 6. VOTING. The vote of the majority of the directors present at a meeting at which a quorum is present shall be the act of the Board of Directors, unless the Delaware General Corporation Law or the Certificate of Incorporation requires a vote of a greater number.

SECTION 7. VACANCIES. Any vacancy or newly-created directorship on the Board of Directors, however occurring, may be filled by vote of a majority of the directors then in office, although less than a quorum, or by a sole remaining director. A director elected to fill a vacancy shall be elected for the unexpired term of such director's predecessor in office, and a director chosen to fill a position resulting from a newly-created directorship shall hold office until the next annual meeting of stockholders and until a successor is elected and qualified, or until such director's earlier death, resignation or removal.

SECTION 8. REMOVAL OF DIRECTORS. Except as otherwise provided by the General Corporation Law of the State of Delaware, a director, or the entire Board of Directors, may be removed, with or without cause, by the holders of a majority in voting power of the outstanding shares of stock then entitled to vote at an election of directors.

SECTION 9. RESIGNATION. Any director may resign by delivering a resignation in writing or by electronic transmission to the Corporation at its principal office or to the Chairman of the Board, the Chief Executive Officer, the President or the Secretary. Such resignation shall be effective upon delivery unless it is specified to be effective at some later time or upon the happening of some later event.

SECTION 10. REGULAR MEETINGS. Regular meetings of the Board of Directors may be held without notice at such date, time, place and manner as shall be determined from time to time by the Board of Directors; provided that any director who is absent when such a determination is made shall be given notice of the determination. A regular meeting of the Board of Directors may be held without notice immediately after and at the same place as the annual meeting of stockholders.

SECTION 11. SPECIAL MEETINGS. Special meetings of the Board of Directors may be called by or at the request of the Chairman of the Board, the Chief Executive Officer, the President, two or more directors or by one director in the event that there is only a single director in office. The person or persons authorized to call special meetings of the Board of Directors may fix any time, date or place, either within or without the State of Delaware, for holding any special meeting of the Board of Directors called by them.

SECTION 12. NOTICE OF SPECIAL MEETINGS OF THE BOARD OF DIRECTORS. Notice of the date, place, if any, and time of any special meeting of the Board of Directors shall be given to each director by the Secretary or by the officer or one of the directors calling the meeting. Notice shall be duly given to each director (a) in person, by telephone, fax or by electronic transmission at least 24 hours in advance of the meeting, (b) by sending written notice by reputable overnight courier or delivering written notice by hand, to such director's last known

business, home or facsimile address at least 48 hours in advance of the meeting, or (c) by sending written notice by first-class mail to such director's last known business or home address at least 72 hours in advance of the meeting. A notice or waiver of notice of a meeting of the Board of Directors need not specify the purposes of the meeting.

SECTION 13. WRITTEN ACTION BY DIRECTORS. Unless otherwise restricted by the Certificate of Incorporation or these By-laws, any action required or permitted to be taken at any meeting of the Board of Directors, or of any committee thereof, may be taken without a meeting if all members of the Board of Directors or committee, as the case may be, consent thereto in writing or by electronic transmission, and the writing or writings or electronic transmissions are filed with the minutes of proceedings of the Board of Directors or committee. Such filing shall be in paper form if the minutes are maintained in paper form and shall be in electronic form if the minutes are maintained in electronic form.

SECTION 14. PARTICIPATION BY CONFERENCE TELEPHONE. Members of the Board of Directors, or any committee designated by such board, may participate in a meeting of the Board of Directors, or committee thereof, by means of conference telephone other communications equipment as long as all persons participating in the meeting can speak with and hear each other, and participation by a director pursuant to this section shall constitute presence in person at such meeting.

SECTION 15. COMMITTEES. The Board of Directors may designate one or more committees, each committee to consist of one or more of the directors of the Corporation with such lawfully delegable powers and duties as the Board of Directors thereby confers, to serve at the pleasure of the Board of Directors. The Board may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee. In the absence or disqualification of a member at any meeting of a committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not such member or members constitute a quorum, may unanimously appoint another member of the Board of Directors to act at the meeting in the place of any such absent or disqualified member. Any such committee, to the extent provided in the resolution of the Board of Directors and to the extent permitted by law, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers which may require it, but no such committee shall have the power or authority in reference to the following matters: (a) approving or adopting, or recommending to the stockholders, any action or matter (other than the election or removal of directors) expressly required by law to be submitted to stockholders for approval or (b) adopting, amending or repealing any bylaw of the Corporation. Each such committee shall keep minutes and make such reports as the Board of Directors may from time to time request. Except as the Board of Directors may otherwise determine, any committee may make rules for the conduct of its business, but unless otherwise provided by the directors or in such rules, its business shall be conducted as nearly as possible in the same manner as is provided in these By-laws for the Board of Directors. Except as otherwise provided in the Certificate of Incorporation, these By-laws, or the resolution of the Board of Directors designating the committee, a committee may create one or more subcommittees, each subcommittee to consist of one or more members of the committee, and delegate to a subcommittee any or all of the powers and authority of the committee.

SECTION 16. COMPENSATION OF DIRECTORS. Unless otherwise restricted by the Certificate of Incorporation or these By-laws, the Board of Directors shall have the authority to fix the compensation of directors. The directors may be paid their expenses, if any, of attendance at each meeting of the Board of Directors and may be paid a fixed sum for attendance at each meeting of the Board of Directors or a stated salary as director. No such payment shall preclude any director from serving the Corporation in any other capacity and receiving compensation therefore. Members of special or standing committees may be allowed like compensation for attending committee meetings.

ARTICLE IV. OFFICERS

SECTION 1. GENERAL PROVISIONS. The officers of the Corporation shall consist of a Chief Executive Officer, Chief Financial Officer, a President, a Secretary, a Treasurer and such other officers with such other titles as the Board of Directors shall determine, including one or more Vice Presidents, Assistant Treasurers and Assistant Secretaries. The Board of Directors may appoint such other officers as it may deem appropriate. No officer need be a stockholder. Any two or more offices may be held by the same person. The officers elected by the Board of Directors shall have such duties as are hereafter described and such additional duties as the Board of Directors may from time to time prescribe.

SECTION 2. ELECTION AND TERM OF OFFICE. The Chief Executive Officer, Chief Financial Officer, President, Treasurer and Secretary shall be elected annually by the Board of Directors at the regular meeting of the Board of Directors held after each annual meeting of the stockholders. If the election of officers is not held at such meeting, such election shall be held as soon thereafter as may be convenient. Other officers may be appointed at any time by the Board of Directors. Except as otherwise provided by law, by the Certificate of Incorporation or by these By-laws, each officer shall hold office until his successor has been duly elected and qualified, unless a different term is specified in the resolution electing or appointing such officer, or until his earlier death, resignation or removal. Election or appointment of an officer or agent shall not of itself create contract rights.

SECTION 3. RESIGNATION AND REMOVAL OF OFFICERS. Any officer may resign by delivering a written resignation to the Corporation at its principal office or to the Chief Executive Officer, Chief Financial Officer, the President or the Secretary. Such resignation shall be effective upon receipt unless it is specified to be effective at some later time or upon the happening of some later event. Any officer may be removed at any time, with or without cause, by vote of a majority of the directors then in office. Except as the Board of Directors may otherwise determine, no officer who resigns or is removed shall have any right to any compensation as an officer for any period following such officer's resignation or removal, or any right to damages on account of such removal, whether such officer's compensation be by the month or by the year or otherwise, unless such compensation is expressly provided for in a duly authorized written agreement with the Corporation.

SECTION 4. VACANCIES. The Board of Directors may fill any vacancy occurring in any office for any reason and may, in its discretion, leave unfilled for such period as it may determine any offices other than those of Chief Executive Officer, Chief Financial Officer,

President, Treasurer and Secretary. Each such successor shall hold office for the unexpired term of such officer's predecessor and until a successor is elected and qualified, or until such officer's earlier death, resignation or removal.

SECTION 5. THE CHIEF EXECUTIVE OFFICER. Unless the Board of Directors has designated another person as the Corporation's Chief Executive Officer, the President shall be the Chief Executive Officer of the Corporation. The Chief Executive Officer shall have general charge and supervision of the business and affairs of the Corporation subject to the direction of the Board of Directors, and shall perform all duties and have all powers that are commonly incident to the office of chief executive or that are delegated to such officer by the Board of Directors. The Chief Executive Officer shall see that orders and resolutions of the Board of Directors are carried into effect. The Chief Executive Officer may sign bonds, mortgages, certificates for shares and all other contracts and documents whether or not under the seal of the Corporation except in cases where the signing and execution thereof shall be expressly delegated by law, by the Board of Directors or by these By-laws to some other officer or agent of the Corporation. The Chief Executive Officer shall have general powers of supervision and shall be the final arbiter of all differences between officers of the Corporation and his decision as to any matter affecting the Corporation shall be final and binding as between the officers of the Corporation subject only to the Board of Directors.

SECTION 6. THE CHIEF FINANCIAL OFFICER. The Chief Financial Officer shall be the principal financial and accounting officer of the Corporation and shall have all powers that are commonly incident to the office of chief financial officer, or that are delegated to such officer by the Board of Directors, the Chief Executive Officer or the President.

SECTION 7. THE PRESIDENT. In the absence of the Chief Executive Officer or in the event of his inability or refusal to act, the President shall perform the duties of the Chief Executive Officer, and when so acting, shall have all the powers of and be subject to all the restrictions upon the Chief Executive Officer. At all other times the President shall have the active management of the business of the Corporation under the general supervision of the Chief Executive Officer or the Board of Directors. The President shall have concurrent power with the Chief Executive Officer to sign bonds, mortgages, certificates for shares and other contracts and documents, whether or not under the seal of the Corporation except in cases where the signing and execution thereof shall be expressly delegated by law, by the Board of Directors, or by these By-laws to some other officer or agent of the Corporation. In general, the President shall perform all duties incident to the office of president and such other duties as the Chief Executive Officer (if the President is not the Chief Executive Officer) or the Board of Directors may from time to time prescribe.

SECTION 8. THE VICE PRESIDENT. In the absence of the President or in the event of his inability or refusal to act, the Vice President (or in the event there be more than one Vice President, the Executive Vice President and then the other Vice President or Vice Presidents in the order designated, or in the absence of any designation, then in the order of their election) shall perform the duties of the President, and when so acting, shall have all the powers of and be subject to all the restrictions upon the President. The Vice Presidents shall perform such other duties and have such other powers as the Chief Executive Officer or the Board of Directors may from time to time prescribe.

SECTION 9. THE SECRETARY. The Secretary shall perform such duties and shall have such powers as the Board of Directors or the Chief Executive Officer may from time to time prescribe. The Secretary shall perform such duties and have such powers as are incident to the office of the secretary, including without limitation the duty and power to attend all meetings of the Board of Directors and all meetings of the stockholders and record all the proceedings in a book to be kept for that purpose and shall perform like duties for the standing committees when required and to maintain a stock ledger and prepare lists of stockholders and their addresses as The Secretary shall give, or cause to be given, notice of all meetings of the stockholders and special meetings of the Board of Directors, and shall perform such other duties as may be prescribed by the Board of Directors or the Chief Executive Officer, under whose supervision he shall be. The Secretary shall have custody of the corporate records and the corporate seal of the Corporation and the Secretary, or an Assistant Secretary, shall have authority to affix the same to any instrument requiring it and when so affixed, it may be attested by his signature or by the signature of such Assistant Secretary. The Board of Directors may give general authority to any other officer to affix the seal of the Corporation and to attest the affixing by his signature.

SECTION 10. THE ASSISTANT SECRETARY. The Assistant Secretary, or if there be more than one, the Assistant Secretaries in the order determined by the Board of Directors (or if there be no such determination, then in the order of their election), shall, in the absence of the Secretary or in the event of his inability or refusal to act, perform the duties and exercise the powers of the Secretary and shall perform such other duties and have such other powers as the Chief Executive Officer, the Board of Directors or the Secretary may from time to time prescribe. In the absence of the Secretary or any Assistant Secretary at any meeting of stockholders or directors, the chairman of the meeting shall designate a temporary secretary to keep a record of the meeting.

SECTION 11. THE TREASURER. The Treasurer shall perform such duties and shall have such powers as may from time to time be assigned by the Board of Directors or the Chief Executive Officer. In addition, the Treasurer shall perform such duties and have such powers as are incident to the office of treasurer, including without limitation, the duty and power to have the custody of the corporate funds and securities and to keep full and accurate accounts of receipts and disbursements in books belonging to the Corporation and deposit all moneys and other valuable effects in the name and to the credit of the Corporation in such depositories as may be designated by the Board of Directors. The Treasurer shall disburse the funds of the Corporation as may be ordered by the Board of Directors, taking proper vouchers for such disbursements, and shall render to the President and the Board of Directors, as required by the Board of Directors, an account of all his transactions as Treasurer and of the financial condition of the Corporation. If required by the Board of Directors, the Treasurer shall give the Corporation a bond (which shall be renewed every six years) in such sum and with such surety or sureties as shall be satisfactory to the Board of Directors for the faithful performance of the duties of his office and for the restoration to the Corporation, in case of his death, resignation, retirement or removal from office, of all books, papers, vouchers, money and other property of whatever kind in his possession or under his control belonging to the Corporation.

SECTION 12. THE ASSISTANT TREASURER. The Assistant Treasurer, or if there shall be more than one, the Assistant Treasurers in the order determined by the Board of

Directors (or if there be no such determination, then in the order of their election), shall, in the absence of the Treasurer or in the event of his inability or refusal to act, perform the duties and exercise the powers of the Treasurer and shall perform such other duties and have such other powers as the Chief Executive Officer, the Board of Directors or the Treasurer may from time to time prescribe.

SECTION 13. OTHER OFFICERS, ASSISTANT OFFICERS AND AGENTS. Officers, Assistant Officers and Agents, if any, other than those whose duties are provided for in these Bylaws, shall have such authority and perform such duties as may from time to time be prescribed by resolution of the Board of Directors.

SECTION 14. ABSENCE OF OFFICERS, DELEGATION OF AUTHORITY. In the absence of any officer of the Corporation, or for any other reason the Board of Directors may deem sufficient, the Board of Directors may from time to time delegate the powers or duties, or any of such powers or duties, of any officers or officer to any other officer or to any director.

SECTION 15. COMPENSATION. The Board of Directors shall have the authority to establish reasonable salaries, compensation or reimbursement of all officers for services to the Corporation.

ARTICLE V. CAPITAL STOCK

SECTION 1. ISSUANCE OF STOCK. Subject to the provisions of the Certificate of Incorporation and applicable law, the whole or any part of any unissued balance of the authorized capital stock of the Corporation or the whole or any part of any shares of the authorized capital stock of the Corporation held in the Corporation's treasury, if any, may be issued, sold, transferred or otherwise disposed of by vote of the Board of Directors in such manner, for such lawful consideration and on such terms as the Board of Directors may determine.

SECTION 2. CERTIFICATES OF SHARES; UNCERTIFICATED SHARES.

- (a) The shares of the Corporation shall be represented by certificates, provided that the Board of Directors of the Corporation may provide by resolution or resolutions that some or all of any or all classes or series of its stock shall be uncertificated shares. Any such resolution shall not apply to shares represented by a certificate until such certificate is surrendered to the Corporation. Every holder of stock represented by certificates shall be entitled to have a certificate, in such form as may be prescribed by law and by the Board of Directors, signed in a manner that complies with Section 158 of the Delaware General Corporation Law, representing the number of shares held by such holder registered in certificate form. Any or all the signatures on the certificate may be a facsimile or pdf.
- (b) Each certificate for shares of stock which are subject to any restriction on transfer pursuant to the Certificate of Incorporation, these By-laws, applicable securities laws or any agreement among any number of stockholders or among such holders and the Corporation shall have conspicuously noted on the face or back of the certificate either the full text of the restriction or a statement of the existence of such restriction.

(c) Within a reasonable time after the issuance or transfer of uncertificated shares, the Corporation shall send to the registered owner thereof a written notice containing the information required to be set forth or stated on certificates pursuant to Sections 151, 156, 202(a) or 218(a) of the General Corporation Law of the State of Delaware or, with respect to Section 151 of the General Corporation Law of the State of Delaware, a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights.

SECTION 3. SIGNATURES OF FORMER OFFICER, TRANSFER AGENT OR REGISTRAR. In case any officer, transfer agent, or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if such person or entity were such officer, transfer agent or registrar at the date of issue.

SECTION 4. TRANSFER OF SHARES. Transfers of shares of the Corporation shall be made only on the books of the Corporation, or by transfer agents designated to transfer shares of the Corporation. Subject to applicable law, shares of stock represented by certificates shall be transferred only on the books of the Corporation by the surrender to the Corporation or its transfer agent of the certificate representing such shares properly endorsed or accompanied by a written assignment or power of attorney properly executed, and with such proof of authority or the authenticity of signature as the Corporation or its transfer agent may reasonably require. Except as may be otherwise required by law, by the Certificate of Incorporation or by these Bylaws, the Corporation shall be entitled to treat the record holder of stock as shown on its books as the owner of such stock for all purposes, including the payment of dividends and the right to vote with respect to such stock, regardless of any transfer, pledge or other disposition of such stock until the shares have been transferred on the books of the Corporation in accordance with the requirements of these By-laws.

SECTION 5. LOST, DESTROYED OR STOLEN CERTIFICATES. Whenever a certificate representing shares of the Corporation has been lost, destroyed or stolen, the holder thereof may file in the office of the Corporation an affidavit setting forth, to the best of his knowledge and belief, the time, place, and circumstance of such loss, destruction or theft together with a statement of indemnity and posting of such bond sufficient in the opinion of the Board of Directors to indemnify the Corporation against any claim that may be made against it on account of the alleged loss of any such certificate. Thereupon the Board of Directors may cause to be issued to such person or such person's legal representative a new certificate or a duplicate of the certificate alleged to have been lost, destroyed or stolen. In the exercise of its discretion, the Board of Directors may waive the indemnification and bond requirements provided herein.

SECTION 6. REGULATIONS. The issue, transfer, conversion and registration of shares of stock of the Corporation shall be governed by such other regulations as the Board of Directors may establish.

ARTICLE VI. INDEMNIFICATION

SECTION 1. RIGHT TO INDEMNIFICATION OF DIRECTORS, OFFICERS AND EMPLOYEES. The Corporation shall indemnify every person who is or was a party or is or was threatened to be made a party to any action, suit, or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that he/she is or was a director or officer of the Corporation or, while a director or officer or employee of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee, agent or trustee of another corporation, partnership, joint venture, trust, employee benefit plan or other enterprise, against expenses (including counsel fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by him/her in connection with such action, suit or proceeding, to the full extent permitted by applicable law.

ARTICLE VII. DIVIDENDS

SECTION 1. DECLARATIONS OF DIVIDENDS. Dividends upon the Common Stock of the Corporation, subject to the provisions of the Certificate of Incorporation, if any, may be declared by the Board of Directors, pursuant to law. Dividends may be paid in cash, in property, or in shares of the Common Stock, subject to the provisions of the Certificate of Incorporation.

SECTION 2. SPECIAL PURPOSES RESERVES. The Board of Directors may set apart out of any of the funds of the Corporation available for dividends a reserve or reserves for any proper purpose and may abolish any such reserve.

ARTICLE VIII. NOTICE BY ELECTRONIC TRANSMISSION

SECTION 1. NOTICE BY ELECTRONIC TRANSMISSION. Without limiting the manner by which notice otherwise may be given effectively to stockholders pursuant to the Delaware General Corporation Law, the Certificate of Incorporation or these By-laws, any notice to stockholders given by the Corporation under any provision of the Delaware General Corporation Law, the Certificate of Incorporation or these By-laws shall be effective if given by a form of electronic transmission consented to by the stockholder to whom the notice is given. Any such consent shall be revocable by the stockholder by written notice to the Corporation. Any such consent shall be deemed revoked if:

- (a) the Corporation is unable to deliver by electronic transmission two consecutive notices given by the Corporation in accordance with such consent; and
- (b) such inability becomes known to the secretary or an assistant secretary of the Corporation or to the transfer agent, or other person responsible for the giving of notice.

However, the inadvertent failure to treat such inability as a revocation shall not invalidate any meeting or other action.

Any notice given pursuant to the preceding paragraph shall be deemed given:

- (c) if by facsimile telecommunication, when directed to a number at which the stockholder has consented to receive notice;
- (d) if by electronic mail, when directed to an electronic mail address at which the stockholder has consented to receive notice;
- (e) if by a posting on an electronic network together with separate notice to the stockholder of such specific posting, upon the later of (i) such posting and (ii) the giving of such separate notice; and
- (f) if by any other form of electronic transmission, when directed to the stockholder.

An affidavit of the secretary or an assistant secretary or of the transfer agent or other agent of the Corporation that the notice has been given by a form of electronic transmission shall, in the absence of fraud, be *prima facie* evidence of the facts stated therein.

- SECTION 2. DEFINITION OF ELECTRONIC TRANSMISSION. An "electronic transmission" means any form of communication, not directly involving the physical transmission of paper, that creates a record that may be retained, retrieved, and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process.
- SECTION 3. INAPPLICABILITY. Notice by a form of electronic transmission shall not apply to Sections 164, 296, 311, 312 or 324 of the Delaware General Corporation Law.

ARTICLE IX. GENERAL PROVISIONS

- SECTION 1. FISCAL YEAR. The fiscal year of the Corporation shall be fixed by resolution of the Board of Directors.
- SECTION 2. SEAL. The corporate seal shall have inscribed thereon the name of the Corporation, the year of its organization and the words "Corporate Seal, Delaware" or such other form as shall be approved by the Board of Directors. Said seal may be used by causing it or a facsimile thereof to be impressed or affixed or reproduced or otherwise.
- SECTION 3. WRITTEN WAIVER OF NOTICE. A written waiver of any notice required to be given by law, the Certificate of Incorporation or by these By-laws, signed by or electronically transmitted by the person entitled to notice, whether before, at or after the time of the event for which notice is to be given, shall be deemed equivalent to notice required to be given to such person. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of stockholders, directors or members of a committee of directors need be specified in any written waiver of notice.
- SECTION 4. ATTENDANCE AS WAIVER OF NOTICE. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting, except when the person attends a

meeting for the express purpose of objecting at the beginning of the meeting, and objects, to the transaction of any business because the meeting is not lawfully called or convened.

- SECTION 5. CONTRACTS. The Board of Directors may authorize any officer or officers, agent or agents, to enter into any contract or execute and deliver any instrument in the name of and on behalf of the Corporation, and such authority may be general or confined to specific instances.
- SECTION 6. LOANS. No loans shall be contracted on behalf of the Corporation and no evidences of indebtedness shall be issued in its name unless authorized by a resolution of the Board of Directors. Such authority may be general or confined to specific instances.
- SECTION 7. CHECKS, DRAFTS, ETC. All checks, drafts or other orders for the payment of money, notes or other evidences of indebtedness issued in the name of the Corporation shall be signed by one or more officers or agents of the Corporation and in such manner as shall from time to time be determined by resolution of the Board of Directors.
- SECTION 8. DEPOSITS. The funds of the Corporation may be deposited or invested in such bank account, in such investments or with such other depositaries as determined by the Board of Directors.
- SECTION 9. ANNUAL STATEMENT. The Board of Directors shall present at each annual meeting, and at any special meeting of the stockholders when called for by vote of the stockholders, a full and clear statement of the business and condition of the Corporation.
- SECTION 10. VOTING OF SECURITIES. Except as the Board of Directors may otherwise designate, the Chief Executive Officer, Chief Financial Officer, the President or the Treasurer may waive notice of, vote, or appoint any person or persons to vote, on behalf of the Corporation at, and act as, or appoint any person or persons to act as, proxy or attorney-in-fact for this Corporation (with or without power of substitution) at, any meeting of stockholders or securityholders of any other entity, the securities of which may be held by this Corporation.
- SECTION 11. EVIDENCE OF AUTHORITY. A certificate by the Secretary, or an Assistant Secretary, or a temporary Secretary, as to any action taken by the stockholders, directors, a committee or any officer or representative of the Corporation shall as to all persons who rely on the certificate in good faith be conclusive evidence of such action.
- SECTION 12. CERTIFICATE OF INCORPORATION. All references in these By-laws to the Certificate of Incorporation shall be deemed to refer to the Certificate of Incorporation of the Corporation, as amended and in effect from time to time.
- SECTION 13. SEVERABILITY. Any determination that any provision of these By-laws is for any reason inapplicable, illegal or ineffective shall not affect or invalidate any other provision of these By-laws.
- SECTION 14. PRONOUNS. All pronouns used in these By-laws shall be deemed to refer to the masculine, feminine or neuter, singular or plural, as the identity of the person or persons may require.

ARTICLE X. AMENDMENTS

SECTION 1. BY THE BOARD OF DIRECTORS. These By-laws may be altered, amended or repealed, in whole or in part, or new By-laws may be adopted by the Board of Directors, when such power is conferred upon the Board of Directors by the Certificate of Incorporation.

SECTION 2. BY THE STOCKHOLDERS. These By-laws may be altered, amended or repealed, in whole or in part, or new By-laws may be adopted, by the affirmative vote of the holders of a majority in voting power of the outstanding shares of the capital stock of the Corporation issued and outstanding and entitled to vote. If the power to adopt, amend or repeal By-laws is conferred upon the Board of Directors by the Certificate of Incorporation it shall not divest or limit the power of the stockholders to adopt, amend or repeal By-laws.

EXHIBIT B

ALTERNATIVE RESTRUCTURING SUPPORT AGREEMENT

RESTRUCTURING SUPPORT AGREEMENT

This RESTRUCTURING SUPPORT AGREEMENT (this "<u>Agreement</u>") is made and entered into as of April 25, 2018, by and among (i) HCR ManorCare, Inc., a Delaware corporation (the "<u>Debtor</u>"), (ii) Carlyle MC Partners, L.P., a Delaware limited partnership, Carlyle Partners V-A MC, L.P., a Delaware limited partnership, Carlyle Partners V MC, L.P., a Delaware limited partnership, CP V Coinvestment A, L.P., a Delaware limited partnership, and CP V Coinvestment B, L.P., a Delaware limited partnership (collectively, the "<u>Majority Holders</u>"), ProMedica Health System, Inc., an Ohio non-profit corporation ("<u>ProMedica</u>" or the "<u>Alternative Plan Sponsor</u>"), and MC Operations Investments, LLC (the "<u>OCP Holder</u>," together with ProMedica and the Majority Holders, the "<u>Supporting Parties</u>," and together with the Debtor, the "<u>Parties</u>").

RECITALS

WHEREAS, the Parties have engaged in arm's length, good faith discussions regarding a restructuring of the Debtor and certain of its subsidiaries (the "Restructuring");

WHEREAS, on the date hereof, the Debtor, ProMedica, Quality Care Properties, Inc. ("QCP"), and certain other parties, have entered into an Alternative Plan Sponsor Agreement (the "Alternative PSA"), which, among other things, contemplates the acquisition by a subsidiary of the Alternative Plan Sponsor of 100% of the equity of the reorganized Debtor through the attached First Amended Chapter 11 Plan of Reorganization for HCR Manor Care, Inc. (the "Plan"; capitalized terms not otherwise defined in this Agreement shall have the meaning given to such terms in the Alternative PSA); and

WHEREAS, the Supporting Parties intend to support the Restructuring and the Plan upon the terms and conditions set forth herein:

AGREEMENT

NOW, THEREFORE, in consideration of the conditional promises and mutual covenants and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

1. Representations and Warranties

- a. <u>Representations and Warranties of the Debtor</u>. The Debtor represents and warrants to the Supporting Parties that, as of the date hereof:
 - i. It has all requisite power and authority to enter into this Agreement and to carry out the transactions contemplated by, and perform its obligations under, this Agreement.
 - ii. The execution and delivery of this Agreement and the performance of its obligations hereunder have been duly authorized by all necessary action on its part.
 - iii. Subject to the provisions of sections 1125 and 1126 of the Bankruptcy Code, and except as set forth herein, this Agreement is the legally valid and binding obligation of it, enforceable against it in accordance with its terms, except as enforcement may be limited by applicable laws relating

to or limiting creditors' rights generally or by equitable principles relating to enforceability.

- b. <u>Representations and Warranties of the Supporting Parties</u>. Each Supporting Party represents and warrants to the Debtor and to each other Supporting Party that, as of the date hereof:
 - i. (A) Such Supporting Party is the sole record and beneficial owner of the shares of common stock of the Debtor set forth on its signature page hereto ("Debtor Shares"), (B) the Debtor Shares of such Supporting Party constitute the percentage of all shares of common stock of the Debtor set forth on such signature pages hereto, and (C) such Supporting Party has full power and authority to bind and act on behalf of, vote and consent to matters concerning such shares and to dispose of, exchange, assign and transfer such shares.
 - ii. Such Supporting Party has made no prior assignment, sale, participation, grant, conveyance or other transfer of, and has not entered into any other agreement to assign, sell, participate, grant, convey or otherwise transfer, in whole or in part, any portion of its right, title or interest in any of its Debtor Shares that conflict with the representations and warranties of such Supporting Party herein or would render such Supporting Party otherwise unable to comply with this Agreement and perform its obligations hereunder.
 - iii. This Agreement is the legally valid and binding obligation of each such Supporting Party, enforceable against it in accordance with its terms, except as enforcement may be limited by applicable laws relating to bankruptcy or limiting creditors' rights generally or by equitable principles relating to enforceability.

2. Covenants of the Majority Holders

Subject to the terms and conditions hereof and for so long as this Agreement has not been terminated in accordance with its terms, each Majority Holder, severally (and not jointly), agrees to, and to cause its direct and indirect subsidiaries, and its and their directors, officers, employees, agents and other representatives, to:

- a. <u>Support of Plan</u>. Support, and take all reasonable action necessary or reasonably requested by the Debtor to support and facilitate, the solicitation, confirmation and consummation of the Plan and the transactions contemplated by the Plan;
- b. <u>Vote for Plan</u>. (i) Vote all Equity Interests, Claims and other instruments carrying voting rights held or controlled, directly or indirectly, by such Majority Holder to accept the Plan (and any amendments, waivers, or consents required in connection with the Plan that are recommended or requested by the Debtor); (ii) timely deliver its duly executed and completed ballot promptly following the commencement of the solicitation of acceptances of the Plan, and (iii) not withdraw, change, or revoke (or cause to be withdrawn, changed, or revoked) its vote with respect to the Plan

- c. <u>No Competing Plans</u>. Other than for the Plan, not pursue, propose or support, or encourage the pursuit, proposal or support of, any chapter 11 plan or other restructuring or reorganization for, or the liquidation of, the Debtor (directly or indirectly);
- d. <u>No Actions Against Plan</u>. (i) Not take any action that would reasonably be expected to prevent, interfere with, delay or impede the solicitation of votes in connection with the Plan or the confirmation or consummation of the Plan, and (ii) not object to or otherwise commence, or encourage, join in or support any other person to commence, any proceeding or other analogous action opposing the Plan; and
- e. <u>Tax Matters</u>. Not file any federal or state tax return, or any amendment to such a return, claiming any deduction for worthlessness of its Debtor Shares.

3. Covenants of the QCP Holder

Subject to the terms and conditions hereof and for so long as this Agreement has not been terminated in accordance with its terms, the QCP Holder agrees to, and to cause its direct and indirect subsidiaries and affiliates to:

- a. support, and take all reasonable actions necessary or reasonably requested by the Debtor to facilitate the solicitation, confirmation and consummation of the Plan and the transactions contemplated by the Plan, including, without limitation, voting in favor of the Plan;
- b. not take any other action, directly or indirectly, that could prevent, interfere with, delay or impede the solicitation of votes in connection with the Plan or the confirmation or consummation of the Plan;
- c. not object to or otherwise commence, or encourage any other person to commence, any proceeding or take any action opposing the Plan or Disclosure Statement; and
- d. not file any federal or state tax return, or any amendment to such a return, claiming any deduction for worthlessness of its Debtor Shares.

4. Information about the Majority Holders

Subject to the terms and conditions hereof and for so long as this Agreement has not been terminated in accordance with its terms, each Majority Holder, severally (and not jointly), agrees to:

a. use its reasonable best efforts to provide such information requested by the Alternative Plan Sponsor, QCP or the Debtor as is necessary regarding such Majority Holder for the Alternative Plan Sponsor, QCP, the Debtor and their affiliates to make or obtain any filings, notices, consents, registrations, approvals, permits or authorizations (including those with respect to state licensing required to operate the Debtor's businesses) in connection with the Plan or the transactions contemplated thereby; and

b. upon the reasonable request of the Debtor, QCP or the Alternative Plan Sponsor, use its reasonable best efforts to provide such information concerning such Majority Holder in connection with any filings, notices, consents, registrations, approvals, permits or authorizations (including those with respect to state licensing required to operate the Debtor's businesses) in connection with the Plan or the transactions contemplated thereby.

5. Transfer of Shares

Each Supporting Party agrees that, so long as this Agreement has not been terminated in accordance with its terms, it shall not directly or indirectly sell, pledge, hypothecate or otherwise transfer or dispose of or grant, issue or sell any option, right to acquire, voting, participation or other interest in any Debtor Shares (each a "Transfer"), unless the transferee thereof, prior to such Transfer, agrees in writing for the benefit of the Parties to become subject to the terms and conditions of this Agreement as a "Supporting Party" and to be bound by this Agreement by executing the joinder attached hereto as Exhibit B (the "Joinder Agreement"), and delivering an executed copy thereof, within two (2) business days of such execution, to the Debtor, in which event (i) the transferee shall be deemed to be a Supporting Party hereunder and (ii) the transferor shall be deemed to relinquish its rights and be released from its obligations under this Agreement to the extent of such transferred rights and obligations. Each Supporting Party agrees that any Transfer that does not comply with the foregoing shall be deemed void ab initio, and the Debtor and each other Supporting Party shall have the right to avoid such Transfer. This Agreement shall in no way be construed to preclude any shareholder from acquiring additional shares; provided that any such additional shares shall, upon acquisition, automatically be deemed to be subject to all the terms of this Agreement.

6. ProMedica Support

ProMedica shall, and shall cause each of its Subsidiaries to, (i) support, and take all reasonable actions necessary or reasonably requested by the Debtor to facilitate, the solicitation, confirmation and consummation of the Plan and the transactions contemplated by the Plan; (ii) not take any other action, directly or indirectly, that could prevent, interfere with, delay or impede the solicitation of votes in connection with the Plan or the confirmation or consummation of the Plan; (iii) not object to or otherwise commence any proceeding or take any action opposing the Plan or Disclosure Statement; and (iv) otherwise use its reasonable best efforts to assist the Debtor in obtaining entry of the Confirmation Order by the Bankruptcy Court within the deadline set forth in Section 3.3(c) of the Alternative Plan Sponsor Agreement.

7. Additional Supporting Parties

A shareholder that is not a Supporting Party as of the date of this Agreement will become a Party to this Agreement as a Supporting Party on the date that it agrees in writing, for the benefit of the Parties, to become subject to the terms and conditions of this Agreement as a "Supporting Party" and to be bound by this Agreement by executing the Joinder Agreement and delivering an executed copy thereof, within two (2) business days of such execution, to the

Debtor. Upon such delivery, such Supporting Party shall immediately thereafter send a copy of such Joinder Agreement to all other Supporting Parties.

8. Termination

This Agreement and all obligations of the Parties hereunder shall immediately terminate and be of no further force and effect as follows:

- a. upon the Effective Date of the Plan; and
- b. upon the termination of the Alternative PSA and the Original PSA.

Upon termination of this Agreement in accordance with its terms, this Agreement shall forthwith become void and of no further force or effect, each Party shall be released from its commitments, undertakings and agreements under or related to this Agreement, and there shall be no liability or obligation on the part of any Party.

9. No Monetary Liability.

Notwithstanding anything to the contrary contained in this Agreement or provided for under any applicable law, no Party shall be liable to any other person, either in contract or in tort, for any money damages relating to any breach of this Agreement.

10. Specific Performance

It is understood and agreed by the Parties that money damages would not be a sufficient or appropriate remedy for any breach of this Agreement by any Party and each non-breaching Party shall be entitled to specific performance and injunctive or other equitable relief as a remedy of any such breach. Each Party agrees to waive any requirement for the securing or posting of a bond in connection with such remedy.

11. Entire Agreement; Prior Negotiations

This Agreement, including all exhibits attached hereto, constitutes the entire agreement of the Parties and supersedes all prior negotiations and documents reflecting such prior negotiations between and among the Parties (and their respective advisors) with respect to the subject matter of this Agreement.

12. Amendments

Except as otherwise provided herein, this Agreement may not be modified, amended or supplemented without prior written agreement signed by the Debtor, each Supporting Party and the Alternative Plan Sponsor.

13. Governing Law

This Agreement shall be governed by, and construed in accordance with, the internal laws of the State of Delaware. By its execution and delivery of this Agreement, each of the parties hereto hereby irrevocably and unconditionally agrees for itself that any legal action, suit or proceeding against it with respect to any matter under or arising out of or in connection with this Agreement or for recognition or enforcement of any judgment rendered in any such action, suit or proceeding, may be brought in either a state or federal court of competent jurisdiction

in the State of Delaware. By execution and delivery of this Agreement, each of the parties hereto hereby irrevocably accepts and submits itself to the nonexclusive jurisdiction of each such court, generally and unconditionally, with respect to any such action, suit or proceeding. Notwithstanding the foregoing, each of the parties hereto hereby agrees that, while the Bankruptcy Case is pending, the Bankruptcy Court shall have exclusive jurisdiction of all matters arising out of or in connection with this Agreement.

14. Effectiveness

This Agreement shall become effective and binding upon each Party upon the execution and delivery of this Agreement by such Party.

15. No Solicitation

Notwithstanding anything to the contrary, this Agreement is not and shall not be deemed to be (a) a solicitation of consents to the Plan or (b) an offer for the issuance, purchase, sale exchange, hypothecation or other transfer of securities or a solicitation of an offer to purchase or otherwise acquire securities for purposes of the Securities Act of 1933 and the Securities Exchange Act of 1934, each as amended.

16. Third-Party Beneficiary

This Agreement is intended for the benefit of the parties hereto and no other person shall have any rights hereunder; <u>provided</u>, <u>however</u>, that QCP shall be an express third party beneficiary of this Agreement, entitled to enforce this Agreement against the Parties as if it were itself a party hereto.

17. Counterparts

This Agreement may be executed in several counterparts, each of which shall be deemed to be an original, and all of which together shall be deemed to be one and the same agreement. Execution copies of this agreement may be delivered by electronic mail (in ".pdf" or ".tif" format), facsimile or otherwise, which shall be deemed to be an original for the purposes of this Agreement.

18. [Intentionally Omitted]

19. No Waiver of Participation and Preservation of Rights

Except as provided in this Agreement, nothing herein is intended to, does or shall be deemed in any manner to waive, limit, impair or restrict the ability of any Party to protect and preserve its rights, remedies and interests, including, but not limited to, claims against the Debtor, liens or security interests it may have in any assets of the Debtor, or its rights to participate fully in the Bankruptcy Case. Without limiting the foregoing sentence in any way, if this Agreement is terminated in accordance with its terms for any reason, the Parties each fully reserve any and all of their respective rights, remedies and interests.

20. Notices

All notices hereunder shall be deemed given if in writing and delivered, if sent by facsimile, courier or by registered or certified mail (return receipt requested) to the following

addresses and facsimile numbers (or at such other addresses or facsimile numbers as shall be specified by like notice):

If to the Debtor, to counsel at the following address:

Sidley Austin LLP One South Dearborn Chicago, IL 60603 Attention: Larry Nyhan Email: lnyhan@sidley.com

If to the Alternative Plan Sponsor to counsel at the following address:

Shumaker, Loop & Kendrick, LLP 1000 Jackson Street Toledo, OH 43604 Attention: James I Rothschild

David Coyle

Email: jrothschild@slk-law.com dcoyle@slk-law.com

If to the Majority Holders, to counsel at the following address:

Latham & Watkins LLP 555 Eleventh Street, NW, Suite 1000 Washington, D.C. 20004-1304 Attention: Daniel T. Lennon Roger G. Schwartz J. Cory Tull

Email: daniel.lennon@lw.com roger.schwartz@lw.com cory.tull@lw.com

If to the QCP Holder, to counsel at the following address:

Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019

Attention: Adam O. Emmerich and Scott K. Charles Email: aoemmerich@wlrk.com skcharles@wlrk.com

[Signature Pages Follow]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed and delivered by their respective duly authorized officers or other agents, solely in their respective capacity as officers or other agents of the undersigned and not in any other capacity, as of the date first set forth above.

HCR MANORCARE, INC.

Name: John R. Castellano

Title: Chief Restructuring Officer

CARLYLE MC PARTNERS, L	.Р.
------------------------	-----

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.C., its general partner

By:_____

Name: Jeremy Anderson Title: Authorized Signatory

Debtor Shares: 6,908,455

Debtor Shares Percentage: 15.377%

CARLYLE PARTNERS V-A MC, L.P.

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.C., its general partner

By:_____

Name: Jeremy Anderson Title: Authorized Signatory

Debtor Shares: 527,141

Debtor Shares Percentage: 1.173%

CARLYLE PARTNERS V MC, L.P.

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.C., its general partner

By:

Name: Jeremy Anderson Title: Authorized Signatory

Debtor Shares: 26,089,114

Debtor Shares Percentage: 58.072%

CP V COINVESTMENT A, L	"P	. 1	Α.	IТ		ÎΕ	M	Г	S	E	7]	П	\bigcap	\mathbf{O}	C	\mathbf{V}	CP	
------------------------	----	-----	----	----	--	----	---	---	---	---	----	---	-----------	--------------	---	--------------	----	--

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.C., its general partner

By:_____

Name: Jeremy Anderson Title: Authorized Signatory

Debtor Shares: 1,015,490

Debtor Shares Percentage: 2.260%

CP V COINVESTMENT B, L.P.

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.C., its general partner

By:_____

Name: Jeremy Anderson Title: Authorized Signatory

Debtor Shares: 129,357

Debtor Shares Percentage: 0.288%

PROMEDICA HEALTH SYSTEM, INC.

Name:

Randy Obstra

Title: Randy

President & CEO

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 401 of 441

MC OPERATIONS	INVESTM	ENTS,
LLC		1

Name: C. Marc Richards Title: Chief Financial Officer

Debtor Shares: 4,232,244

Debtor Shares Percentage:

EXHIBIT B

Joinder Agreement

This Joinder Agreement to the Restructuring Support Agreement, dated as of [•],
2018, by and among (a) HCR Manorcare, Inc. (the "Debtor"), (b) MC Operations
Investments, LLC (the "QCP Holder") and (c) Carlyle MC Partners, L.P., a Delaware limited
partnership, Carlyle Partners V-A MC, L.P., a Delaware limited partnership, Carlyle Partners
V MC, L.P., a Delaware limited partnership, CP V Coinvestment A, L.P., a Delaware limited
partnership, and CP V Coinvestment B, L.P., a Delaware limited partnership (collectively, the
"Majority Holders") (as set forth in Annex I hereto, the "Restructuring Support Agreement")
is executed and delivered by [] (the "Joining Supporting Party") as of
[], 2018. Each capitalized term used herein but not otherwise defined shall
have the meanings set forth in the Restructuring Support Agreement.

- 1. <u>Agreement to be Bound</u>. The Joining Supporting Party hereby agrees to be bound by all of the terms of the Restructuring Support Agreement (as the same may be hereafter amended, restated or otherwise modified from time to time), including, without limitation, the covenants set forth in Section 2 of the Restructuring Support Agreement. The Joining Supporting Party shall hereafter be deemed to be a "Supporting Party" and a Party for all purposes under the Restructuring Support Agreement.
- 2. <u>Representations and Warranties</u>. With respect to the aggregate principal amount of shares held by the Joining Supporting Party upon consummation of the Transfer of such shares, the Joining Supporting Party hereby makes the representations and warranties of the Supporting Parties set forth in Section 1 of the Restructuring Support Agreement to each of the other Parties.
- 3. <u>Governing Law</u>. This Joinder Agreement shall be governed by and construed in accordance with the internal laws of the State of Delaware, without regard to any conflicts of law provisions which would require the application of the law of any other jurisdiction.

* * * * *

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the Joining Supporting Party has caused this Joinder Agreement to be executed as of the date first written above.

Ву:	Name:		
	Title:		
lotic	e Address and E-n	nail:	

Annex 1 to Joinder Agreement

Restructuring Support Agreement

EXHIBIT C

ORIGINAL RESTRUCTURING SUPPORT AGREEMENT

RESTRUCTURING SUPPORT AGREEMENT

This RESTRUCTURING SUPPORT AGREEMENT (this "<u>Agreement</u>") is made and entered into as of March 2, 2018, by and among (i) HCR ManorCare, Inc., a Delaware corporation (the "<u>Debtor</u>"), (ii) Carlyle MC Partners, L.P., a Delaware limited partnership, Carlyle Partners V-A MC, L.P., a Delaware limited partnership, Carlyle Partners V MC, L.P., a Delaware limited partnership, CP V Coinvestment A, L.P., a Delaware limited partnership, and CP V Coinvestment B, L.P., a Delaware limited partnership (collectively, the "<u>Majority Holders</u>"), and (iii) MC Operations Investments, LLC (the "<u>QCP Holder</u>," together with the Majority Holders, the "<u>Supporting Parties</u>," and together with the Debtor, the "<u>Parties</u>").

RECITALS

WHEREAS, the Parties have engaged in arm's length, good faith discussions regarding a restructuring of the Debtor and certain of its subsidiaries (the "**Restructuring**");

WHEREAS, on the date hereof, the Debtor, Quality Care Properties, Inc. ("QCP"), and certain of QCP's direct and indirect subsidiaries have entered into a Plan Sponsor Agreement (the "PSA"), which, among other things, contemplates the acquisition by a subsidiary of QCP of 100% of the equity of the reorganized Debtor through a prepackaged Plan in form and substance consistent with the term sheet attached as Exhibit A hereto (the "Plan Term Sheet"); and

WHEREAS, the Supporting Parties intend to support the Restructuring and the prepackaged Plan upon the terms and conditions set forth herein:

AGREEMENT

1. Representations and Warranties

- a. <u>Representations and Warranties of the Debtor</u>. The Debtor represents and warrants to the Supporting Parties that, as of the date hereof:
 - i. It has all requisite power and authority to enter into this Agreement and to carry out the transactions contemplated by, and perform its obligations under, this Agreement.
 - ii. The execution and delivery of this Agreement and the performance of its obligations hereunder have been duly authorized by all necessary action on its part.
 - iii. Subject to the provisions of sections 1125 and 1126 of the Bankruptcy Code, and except as set forth herein, this Agreement is the legally valid and binding obligation of it, enforceable against it in accordance with its terms, except as enforcement may be limited by applicable laws relating to or limiting creditors' rights generally or by equitable principles relating to enforceability.

Capitalized terms not otherwise defined in this Agreement shall have the meaning given to such terms in the PSA.

- b. <u>Representations and Warranties of the Supporting Parties</u>. Each Supporting Party represents and warrants to the Debtor and to each other Supporting Party that, as of the date hereof:
 - i. Such Supporting Party (A) is the sole beneficial owner of the shares set forth on its signature page hereto ("**Debtor Shares**") and (B) has full power and authority to bind or act on behalf of, vote and consent to matters concerning such shares and to dispose of, exchange, assign and transfer such shares.
 - ii. Such Supporting Party has made no prior assignment, sale, participation, grant, conveyance or other transfer of, and has not entered into any other agreement to assign, sell, participate, grant, convey or otherwise transfer, in whole or in part, any portion of its right, title or interests in any of its shares that are subject to this Agreement that conflict with the representations and warranties of such Supporting Party herein or would render such Supporting Party otherwise unable to comply with this Agreement and perform its obligations hereunder.
 - iii. This Agreement is the legally valid and binding obligation of each such Supporting Party hereto, enforceable against it in accordance with its terms.

2. Covenants of the Supporting Parties

Subject to the terms and conditions hereof and for so long as this Agreement has not been terminated in accordance with its terms, and unless compliance is waived in writing by the Debtor and each Supporting Party, each Supporting Party, severally (and not jointly), agrees to, and to cause its direct and indirect subsidiaries and affiliates to:

- a. support, and take all reasonable actions necessary or reasonably requested by the Debtor to facilitate the solicitation, confirmation and consummation of the Plan and the transactions contemplated by the Plan, including, without limitation, voting in favor of the Plan;
- b. not take any other action, directly or indirectly, that could prevent, interfere with, delay or impede the solicitation of votes in connection with the Plan or the confirmation or consummation of the Plan;
- c. not object to or otherwise commence, or encourage any other person to commence, any proceeding or take any action opposing the Plan or Disclosure Statement; and
- d. not file any federal or state tax return, or any amendment to such a return, claiming any deduction for worthlessness of its Debtor Shares.

3. Information about the Majority Holders

Subject to the terms and conditions hereof and for so long as this Agreement has not been terminated in accordance with its terms, and unless compliance is waived in writing by the Debtor and QCP, each Majority Holder, severally (and not jointly), agrees to:

- a. use their reasonable best efforts to provide such information requested in writing by Parent or the Debtor as is necessary regarding the Majority Holders for the Purchaser Entities to make or obtain the Governmental Approvals; and
- b. upon the reasonable written request by the Debtor or QCP, use their reasonable best efforts to furnish all information concerning such Majority Holders in connection with any statement, filing, notice or application made by or on behalf of a Purchaser Entity or the Debtor or any of their respective subsidiaries to any Governmental Entity in connection with making or obtaining the Governmental Approvals.

4. Transfer of Shares

Each Supporting Party agrees that so long as this Agreement has not been terminated in accordance with its terms it shall not directly or indirectly sell, pledge, hypothecate or otherwise transfer or dispose of or grant, issue or sell any option, right to acquire, voting, participation or other interest in any Debtor Shares (each a "Transfer"), unless the transferee thereof, prior to such Transfer, agrees in writing for the benefit of the Parties to become subject to the terms and conditions of this Agreement as a "Supporting Party" and to be bound by this Agreement by executing the joinder attached hereto as Exhibit B (the "Joinder **Agreement**"), and delivering an executed copy thereof, within two (2) business days of such execution, to the Debtor, in which event (i) the transferee shall be deemed to be a Supporting Party hereunder and (ii) the transferor shall be deemed to relinquish its rights and be released from its obligations under this Agreement to the extent of such transferred rights and obligations. Each Supporting Party agrees that any Transfer that does not comply with the foregoing shall be deemed void ab initio, and the Debtor and each other Supporting Party shall have the right to avoid such Transfer. This Agreement shall in no way be construed to preclude any shareholder from acquiring additional shares; provided that any such additional shares shall, upon acquisition, automatically be deemed to be subject to all the terms of this Agreement.

5. Additional Supporting Parties

A shareholder that is not a Supporting Party as of the date of this Agreement will become a Party to this Agreement as a Supporting Party on the date that it agrees in writing, for the benefit of the Parties, to become subject to the terms and conditions of this Agreement as a "Supporting Party" and to be bound by this Agreement by executing the Joinder Agreement and delivering an executed copy thereof, within two (2) business days of such execution, to the Debtor. Upon such delivery, such Supporting Party shall immediately thereafter send a copy of such Joinder Agreement to all other Supporting Parties.

6. Termination

This Agreement and all obligations of the Parties hereunder shall immediately terminate and be of no further force and effect as follows:

- a. upon the Effective Date of the Plan; and
- b. upon the termination of PSA.

Upon termination of this Agreement in accordance with its terms, this Agreement shall forthwith become void and of no further force or effect, each Party shall be released from its commitments, undertakings and agreements under or related to this Agreement, and there shall be no liability or obligation on the part of any Party.

7. No Monetary Liability

Notwithstanding anything to the contrary contained in this Agreement or provided for under any applicable law, no Party shall be liable to any other person, either in contract or in tort, for any money damages relating to any breach of this Agreement.

8. Specific Performance

It is understood and agreed by the Parties that money damages would not be a sufficient or appropriate remedy for any breach of this Agreement by any Party and each non-breaching Party shall be entitled to seek specific performance and injunctive or other equitable relief as a remedy of any such breach. Each Party agrees to waive any requirement for the securing or posting of a bond in connection with such remedy.

9. Entire Agreement; Prior Negotiations

This Agreement, including all exhibits attached hereto, constitutes the entire agreement of the Parties and supersedes all prior negotiations and documents reflecting such prior negotiations between and among the Parties (and their respective advisors) with respect to the subject matter of this Agreement.

10. Amendments

Except as otherwise provided herein, this Agreement may not be modified, amended or supplemented without prior written agreement signed by the Debtor and each Supporting Party.

11. Governing Law

This Agreement shall be governed by, and construed in accordance with, the internal laws of the State of Delaware. By its execution and delivery of this Agreement, each of the parties hereto hereby irrevocably and unconditionally agrees for itself that any legal action, suit or proceeding against it with respect to any matter under or arising out of or in connection with this Agreement or for recognition or enforcement of any judgment rendered in any such action, suit or proceeding, may be brought in either a state or federal court of competent jurisdiction in the State of Delaware. By execution and delivery of this Agreement, each of the parties hereto hereby irrevocably accepts and submits itself to the nonexclusive jurisdiction of each such court, generally and unconditionally, with respect to any such action, suit or proceeding. Notwithstanding the foregoing, upon the commencement of the Bankruptcy Case, each of the parties hereto hereby agrees that, if the petition has been filed and the Bankruptcy Case is pending, the Bankruptcy Court shall have exclusive jurisdiction of all matters arising out of or in connection with this Agreement.

12. Effective Date; Conditions to Effectiveness

This Agreement shall become effective and binding upon each of the Parties upon the execution and delivery of this Agreement by each Party hereto.

13. No Solicitation

Notwithstanding anything to the contrary, this Agreement is not and shall not be deemed to be (a) a solicitation of consents to the Plan or (b) an offer for the issuance, purchase, sale exchange, hypothecation or other transfer of securities or a solicitation of an offer to purchase or otherwise acquire securities for purposes of the Securities Act of 1933 and the Securities Exchange Act of 1934, each as amended.

14. Third-Party Beneficiary

This Agreement is intended for the benefit of the parties hereto and no other person shall have any rights hereunder.

15. Counterparts

This Agreement may be executed in several counterparts, each of which shall be deemed to be an original, and all of which together shall be deemed to be one and the same agreement. Execution copies of this agreement may be delivered by electronic mail (in ".pdf" or ".tif" format), facsimile or otherwise, which shall be deemed to be an original for the purposes of this Agreement.

16. Settlement Discussions

This Agreement and the Plan Term Sheet are part of a proposed settlement of matters that could otherwise be the subject of litigation among the parties hereto. Nothing herein shall be deemed an admission of any kind. Pursuant to Federal Rule of Evidence 408 and any applicable state rules of evidence, this Agreement and all negotiations relating thereto shall not be admissible into evidence in any proceeding other than a proceeding to enforce the terms of this Agreement.

17. No Waiver of Participation and Preservation of Rights

Except as provided in this Agreement, nothing herein is intended to, does or shall be deemed in any manner to waive, limit, impair or restrict the ability of any Party to protect and preserve its rights, remedies and interests, including, but not limited to, claims against the Debtor, liens or security interests it may have in any assets of the Debtor, or its rights to participate fully in the Bankruptcy Case. Without limiting the foregoing sentence in any way, if this Agreement is terminated in accordance with its terms for any reason, the Parties each fully reserve any and all of their respective rights, remedies and interests.

18. Notices

All notices hereunder shall be deemed given if in writing and delivered, if sent by facsimile, courier or by registered or certified mail (return receipt requested) to the following addresses and facsimile numbers (or at such other addresses or facsimile numbers as shall be specified by like notice):

If to the Debtor, to counsel at the following address:

Sidley Austin LLP One South Dearborn Chicago, IL 60603 Attention: Larry Nyhan Email: lnyhan@sidley.com

If to the QCP Holder, to counsel at the following address:

Sullivan & Cromwell LLP 125 Broad Street New York, NY 10004-2498 Attn: Andrew G. Dietderich

Fax: (212) 558-3588

Email: dietdericha@sullcrom.com

If to the Majority Holders, to counsel at the following address:

Latham & Watkins LLP
555 Eleventh Street, NW, Suite 1000
Washington, D.C. 20004-1304
Attention: Daniel T. Lennon
Roger G. Schwartz
J. Cory Tull

Email: daniel.lennon@lw.com roger.schwartz@lw.com cory.tull@lw.com

[Signature Pages Follow]

Case 18-10467-KG Doc 197 Filed 05/18/18 Page 412 of 441

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed and delivered by their respective duly authorized officers or other agents, solely in their respective capacity as officers or other agents of the undersigned and not in any other capacity, as of the date first set forth above.

HCR MANORCARE, INC.

Name: John R. Castellano
Title: Chief Restructoring Officer

MC OPERATIONS INVESTMENTS,

LLC

Dy.____

Marc Richards

Name:

Chief Financial Officer

Title:

Debtor Shares: 4, 232, 244

CARLYLE MC PARTNER	S,	L.P.
--------------------	----	------

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.Ca its general partner

Name: David B. Pearson

Title: Authorized Person

Debtor Shares: 6,908, 455

CARLYLE PARTNERS V-A MC, L.P.

By: TC Group V, L.P., its general partner

By: TC Group V L.L.C, its general partner

Name: David B. Pearson

Title: Authorized Person

Debtor Shares: 527, 141

CARLYLE PARTNERS V MC, L.P.

By: TC Group V, L.P., its general partner

By: To Group Y. L.I. , its general partner

Name: David B. Pearson

Title: Authorized Person

Debtor Shares: 26,089,114

CP V COINVESTMENT A, L.P.

By: TC Group V, L.P., its general partner

By: TC Group V, L.L.C., its general partner

Name: David B. Pearson

Title: Authorized Person

Debtor Shares: \\ 015, 490

CP V COINVESTMENT B, L.P.

By: TC Group V, L.P., its general partner

By: TC Group V. L.L.C., its general partner

Name: David B. Pearson

Title: Authorized Person

Debtor Shares: 129, 357

EXHIBIT A

Plan Term Sheet

Exhibit Version

EXHIBIT A

HCR MANORCARE, INC. RESTRUCTURING TERM SHEET SUMMARY OF PRINCIPAL TERMS AND CONDITIONS March 2, 2018

The following (this "<u>Plan Term Sheet</u>") describes the material terms of a proposed restructuring of HCR ManorCare, Inc. (the "<u>Debtor</u>"), which will be implemented through a prepackaged plan of reorganization (the "<u>Prepackaged Plan</u>") under title 11 of the United States Code (the "<u>Bankruptcy Code</u>"). The Debtor and Quality Care Properties, Inc. and certain of its subsidiaries (collectively, "<u>QCP</u>") are parties to the Plan Sponsor Agreement, dated as of March 2, 2018 (the "<u>PSA</u>"), which, among other things, set forth the manner in which the Prepackaged Plan will be pursued. Capitalized terms not otherwise defined herein shall have the meanings set forth in the PSA.

In addition, the Debtor, the Carlyle Holders and QCP are party to a Restructuring Support Agreement (the "RSA") pursuant to which, among other things, the Carlyle Holders and QCP have agreed to support the Prepackaged Plan described by the Plan Term Sheet.

This Plan Term Sheet outlines many of the principal terms of the Prepackaged Plan, but other material terms will be set forth in definitive documentation, including the PSA and RSA. Nothing contained herein shall constitute an offer susceptible to acceptance by the Debtor or any other party, or a legally binding obligation of any party, other than as set forth in the PSA and RSA. This Plan Term Sheet is a settlement proposal in furtherance of settlement discussions and is entitled to protection from any use or disclosure to any party or person pursuant to Federal Rule of Evidence 408 or any other applicable rule of evidence. THIS PLAN TERM SHEET DOES NOT CONSTITUTE A SOLICITATION OF VOTES FOR A PLAN OF REORGANIZATION FOR PURPOSES OF SECTIONS 1125 AND 1126 OF THE BANKRUPTCY CODE OR AN OFFER WITH RESPECT TO ANY SECURITIES. SUCH OFFER OR SOLICITATION WILL BE MADE IN COMPLIANCE WITH THE PSA AND THE RSA AND ALL APPLICABLE SECURITIES LAWS AND/OR PROVISIONS OF THE BANKRUPTCY CODE.

Transaction Overview	
Solicitation	The Debtor will pursue a prepetition solicitation of votes on the Prepackaged Plan (the "Solicitation") in accordance with the PSA and RSA.

Prepackaged Case	The Debtor will file a voluntary petition for reorganization under chapter 11 of the Bankruptcy Code, in the United States Bankruptcy Court for the District of Delaware (the "Bankruptcy Court"), in accordance with its obligations under the PSA and RSA. The date on which the Debtor's chapter 11 case is commenced shall be the "Petition Date." In addition, on the Petition Date the Debtor will file "first-day" and "second-day" pleadings with the Bankruptcy Court.		
Funding of Chapter 11 Case	The cost of administering the Debtor's chapter 11 c by cash of the Debtor, ordinary course intercompant the cash management system of the Debtor and its senecessary, additional intercompany borrowing from Services, LLC in accordance with the Centerbridge Debtor shall not incur any debtor-in-possession finate avoidance of doubt, the Debtor may seek entry of an accordance of doubt, the Debtor to use the case Centerbridge and providing Centerbridge reasonable protections in connection therewith.	y transfers through subsidiaries, and, if a HCR ManorCare Facility. The ancing. For the n order in the sh collateral of	
Treatment of Claims and	Interests		
Administrative Claims	Each holder of an Allowed ¹ Administrative Claim ² will receive, in the sole discretion of the reorganized Debtor, payment in full in cash of the unpaid portion of such Allowed Administrative Claim (a) on the date on which the Prepackaged Plan becomes effective (the "Effective Date") or as promptly thereafter as reasonably practicable, (b) in the ordinary course of the reorganized Debtor's business, or (c) as otherwise agreed by the reorganized Debtor and such holder. ³	In accordance with section 1123(a)(1) of the Bankruptcy Code, the Prepackaged Plan does not classify Administrative Claims.	

[&]quot;Allowed" means any claim that is determined to be Allowed (as defined in the Prepackaged Plan).

² "<u>Administrative Claim</u>" means a claim arising under sections 503(b), 507(b) or to the extent applicable, 1114(e)(2) of the Bankruptcy Code.

Administrative Claims that become due prior to the Effective Date may be paid by the Debtor in the ordinary course during the chapter 11 case, subject to the PSA.

Priority Tax Claims	Each holder of an Allowed Priority Tax Claim ⁴ will receive, in the sole discretion of the reorganized Debtor, payment in cash by the Reorganized Debtor in an amount of such holder's Allowed Priority Tax Claim on the later of the Effective Date or when such Allowed Priority Tax Claim becomes due.	In accordance with section 1123(a)(1) of the Bankruptcy Code, the Prepackaged Plan does not classify Priority Tax Claims.
Class 1 – Other Priority Claims	Each holder of an Allowed Other Priority Claim ⁵ will receive payment in full in cash as promptly as reasonably practicable on the later of (a) the Effective Date and (b) the date on which such Other Priority Claim becomes an Allowed claim payable under applicable law or any agreement relating thereto.	Unimpaired – Deemed to Accept
Class 2 – Secured Claims	Each holder of an Allowed secured claim will receive, in the sole discretion of the reorganized Debtor, on the Effective Date (or as promptly thereafter as reasonably practicable) or in the ordinary course of the reorganized Debtor's business, (a) payment in full by the reorganized Debtor in cash, including the payment of any interest Allowed and payable under section 506(b) of the Bankruptcy Code, (b) delivery of the collateral securing such Allowed secured claim or (c) treatment of such Allowed secured claim in any other matter that renders the claim unimpaired.	Unimpaired – Deemed to Accept
Class 3 – Centerbridge Claim	On the Effective Date, Centerbridge's ⁶ secured contingent guaranty claim against the Debtor (the " <u>Centerbridge Claim</u> ") will be reinstated and any liens held by Centerbridge against the assets of the Debtor securing the Centerbridge Claim shall survive the Effective Date, rendering the Centerbridge Claim unimpaired.	Unimpaired – Deemed to Accept

⁴ "Priority Tax Claim" means a claim under section 507(a)(8) of the Bankruptcy Code.

⁵ "Other Priority Claim" means an Allowed claim under section 507(a) of the Bankruptcy other than an Administrative Expense Claim or Priority Tax Claim.

⁶ "Centerbridge" means RD Credit, LLC.

Class 4 – QCP Claims	The QCP Claims ⁷ are deemed Allowed. Each holder of a QCP Claim (or its designee) shall receive on the Effective Date, in full and final satisfaction, release, and discharge of, and in exchange for, its QCP Claim, 100% of the common stock in the reorganized Debtor.	Impaired – Entitled to Vote
Class 5 – General Unsecured Claims	Each holder of an unsecured claim against the Debtor other than a QCP Guarantee Claim or a Severance Claim (a "General Unsecured Claim") will receive, in the sole discretion of the reorganized Debtor, (a) payment in full in cash of such claim (i) on the Effective Date or as promptly thereafter as reasonably practicable or (ii) in the ordinary course of the reorganized Debtor's business or (b) satisfaction of such claim in any other manner that renders the General Unsecured Claim unimpaired or as otherwise agreed by the reorganized Debtor and such holder.	Unimpaired – Deemed to Accept
Class 6 – Severance Claims	Each holder of a claim against the Debtor for damages resulting from the termination of an employment contract (a "Severance Claim) will receive, in the sole discretion of QCP, payment in full in cash of the Allowed amount of such claim (for the avoidance of doubt, subject to applicable limitations on the Allowed amount of such claim under the Bankruptcy Code) (a) on the Effective Date or as soon thereafter as reasonably practicable or (b) as otherwise agreed by the reorganized Debtor and such holder.	Unimpaired – Deemed to Accept
Class 7 – Section 510(b) Claims	On the Effective Date, section 510(b) claims (if any) will be discharged and eliminated, and the holders of any such claims will not receive any distributions or property under the Prepackaged Plan on account of such claims (if any such claims exist).	Impaired – Deemed to Reject Not entitled to any distributions under the Prepackaged Plan

⁷ "<u>OCP Claims</u>" means all claims of QCP and its subsidiaries against the Debtor arising under the Guaranty and due and unpaid as of the Effective Date.

Class 8 – Equity Interests	On the Effective Date, all existing common and preferred shares, options, warrants, rights, similar instruments, and other prepetition equity interests in the Debtor will be extinguished without payment. Holders of such interests will not retain any such interests from or after the Effective Date or receive any distributions or property under the Prepackaged Plan on account of such interests.	Impaired – Deemed to Reject Not entitled to any distributions under the Prepackaged Plan
----------------------------	---	--

Other Provisions	
Executory Contracts / Unexpired Leases	All executory contracts and unexpired leases to which the Debtor is a party and are otherwise susceptible to assumption/assignment will be assumed on the Effective Date pursuant to sections 365 and 1123 of the Bankruptcy Code other than any executory contract or unexpired lease to which the Debtor is a party and which is rejected by the Debtor in accordance with the PSA or with the prior written consent of QCP (in its sole discretion).
Bar Date	None.
Vesting of Assets Free and Clear of Liens and Claims	As of the Effective Date, all property of the Debtor shall vest in the reorganized Debtor free and clear of all liens, claims, charges or other encumbrances or interests, provided, however, that any liens (a) held by Centerbridge against the assets of the Debtor securing the Centerbridge Claim and (b) held by HCR Home Health Care and Hospice, LLC securing the \$25 million Subordinated Secured Demand Note, dated March [•], 2018, made by the Debtor to HCR Home Health Care and Hospice, LLC, in each case, shall survive the Effective Date.
New Master Lease	On the Effective Date, HCR III and the Lessor will amend the Master Lease in accordance with the PSA. On or prior to the seventh day after the Effective Date, the Guaranty will be terminated and released in accordance with the PSA.
Claim Allowance Process	The Debtor shall have the right to object to or otherwise agree to the Allowance of all claims (other than QCP Claims) in accordance with its duties as debtor-in-possession, subject to QCP's right to object to such Allowance.
Excess Severance	Subject to Section 1.5(b) of the PSA, (a) on the Effective Date, QCP shall establish an account for the purpose of funding Excess Severance Payments (the "Severance Account") and, following the Closing, shall cause the Severance Account to be funded with all amounts necessary to make Excess Severance Payments and (b) each of the current CEO, CFO, GC and COO of the Debtor (an "Eligible Employee"), shall, unless otherwise agreed in writing between the reorganized Debtor and such person, receive from the Severance Account such Eligible Employee's Excess Severance Payment when due and payable in accordance with the terms of the applicable Separation Agreement.

Tax Treatment of the Restructuring	The restructuring of the Debtor is intended to qualify under section 382(1)(5) of the Internal Revenue Code of 1986, as amended, in order to preserve the net operating losses and other tax attributes available to the Debtor.						
Exemption from Registration	The issuance of common stock of the reorganized Debtor will be exempt from SEC registration under applicable law.						
Governance of the Reorganized Debtor	Board of Directors and officers to be determined by QCP.						
Releases and Exculpation	Customary releases and exculpation by (x) the Debtor and (y) holders of claims and interests in the Debtor of: (a) the Debtor and the reorganized Debtor, (b) the current and former directors, officers, equity holders, controlling persons, employees, agents, attorneys, financial advisors, restructuring advisors, investment bankers, accountants, and other professional representatives of the Debtor and the reorganized Debtor, in their capacities as such, (c) QCP, and (d) with respect to each entity named in the preceding (a) through (c), each such entity's directors, officers, members, managers, equity holders, general or limited partners, controlling persons, employees, agents, affiliates, parents, subsidiaries, predecessors, successors, heirs, executors and assigns, attorneys, financial advisors, restructuring advisors, investment bankers, accountants and other professionals or representatives when acting in any such capacities, in each case, other than direct or indirect subsidiaries of the Debtor.						
Conditions to the Effective Date	 The Confirmation Order shall have become a Final Order; All conditions precedent to the consummation of the transactions contemplated by the PSA (other than the occurrence of the Effective Date) shall have been satisfied; The PSA shall not have terminated in accordance with its terms; and All other actions necessary to implement the terms of the Prepackaged Plan on the Effective Date shall have been taken. 						

EXHIBIT B

Joinder Agreement

This Joinder Agreement to the Restructuring Support Agreement, dated as of [•],
2018, by and among (a) HCR Manorcare, Inc. (the "Debtor"), (b) MC Operations
Investments, LLC (the "QCP Holder") and (c) Carlyle MC Partners, L.P., a Delaware limited
partnership, Carlyle Partners V-A MC, L.P., a Delaware limited partnership, Carlyle Partners
V MC, L.P., a Delaware limited partnership, CP V Coinvestment A, L.P., a Delaware limited
partnership, and CP V Coinvestment B, L.P., a Delaware limited partnership (collectively, the
"Majority Holders") (as set forth in Annex I hereto, the "Restructuring Support Agreement"),
is executed and delivered by [] (the "Joining Supporting Party") as of
[], 2018. Each capitalized term used herein but not otherwise defined shall
have the meanings set forth in the Restructuring Support Agreement.

- Agreement to be Bound. The Joining Supporting Party hereby agrees to be bound by all of the terms of the Restructuring Support Agreement (as the same may be hereafter amended, restated or otherwise modified from time to time). The Joining Supporting Party shall hereafter be deemed to be a "Supporting Party" and a Party for all purposes under the Restructuring Support Agreement.
- 2. <u>Representations and Warranties</u>. With respect to the aggregate principal amount of shares held by the Joining Supporting Party upon consummation of the Transfer of such shares, the Joining Supporting Party hereby makes the representations and warranties of the Supporting Parties set forth in Section 1 of the Restructuring Support Agreement to each of the other Parties.
- 3. <u>Governing Law</u>. This Joinder Agreement shall be governed by and construed in accordance with the internal laws of the State of Delaware, without regard to any conflicts of law provisions which would require the application of the law of any other jurisdiction.

* * * * *

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the Joining Supporting Party has caused this Joinder Agreement to be executed as of the date first written above.

SUPI	PORTING	PARTY	Z .	
By:	Name: Title:			_
Notic	ee Address	and E-r	nail:	
			@	

Annex 1 to Joinder Agreement

Restructuring Support Agreement

EXHIBIT D

HCR FINANCIAL PROJECTIONS

Financial Information and Projections – Assumptions Based Upon Alternative Transaction

I. Financial Projections

For purposes of demonstrating feasibility of the Alternative Transaction embodied in the Plan, the Debtor, with the assistance of its advisors, has developed a post-Effective Date income statement (the "Financial Projections" or the "Projections") for the quarter period ending December 31, 2018 and annual periods ending December 31, 2019 (fiscal year 2019) through December 31, 2021 (fiscal year 2021) (the "Projection Period"). These Financial Projections assume an Effective Date of October 1, 2018 and represent the consolidated forecasted projections of the Long-Term Care Business excluding the facilities currently being marketed for sale (the "RemainCo Long-Term Care Business"), the 4H Business and the Outpatient Rehab and Other Businesses. These Projections have been prepared based on a number of assumptions made by the Debtor's management, in consultation with the Debtor's financial advisors, as to the future performance of the Reorganized Debtor following the consummation of the Alternative Transaction, and reflect Debtor management's judgement and expectations regarding the Debtor's future operations and financial position.

These Projections have been reviewed with ProMedica Parent and its advisors and the Debtor believes these Projections are reasonable especially in light of the transactions contemplated by the Alternative Plan Sponsor Agreement and the Alternative Master Lease. While there are risk and uncertainties with any projections (as further noted below), under the Alternative Plan Sponsor Agreement, the Debtor's senior management team will remain in their current positions and will be in a position to implement a long-term strategy developed jointly between the Debtor and ProMedica Parent. While ProMedica Parent has not finalized a long-term strategy, these Projections reflect a fair presentation of the potential business results of the Company, including the non-debtor subsidiaries of the Debtor.

The Financial Projections are subject to inherent risks and uncertainties, several of which are difficult to predict and many of which are beyond management's control. Factors that may cause actual results to differ from expected results include, but are not limited to:

- (i) changes in the reimbursement rates or methods or timing of payment from third party payors, including the Medicare and Medicaid programs, or the implementation of other measures to reduce reimbursement for the RemainCo Long-Term Care Business, the 4H Business and the Outpatient Rehab and Other Businesses' services and products;
- (ii) changes in the acuity of patients, as well as payor mix and payment methodologies;
- (iii) changes in existing or future laws and regulations or the enforcement of these laws and regulations, including but not limited to future adjustments in healthcare regulation;
- (iv) further consolidation of Managed Care organizations and other third-party payors;
- (v) the uncertainty inherent in estimating changes in population demographics and healthcare service requirements;

- (vi) changes in future operating costs, most notably employee wages, could differ from historical trends due to a competitive labor market; and
- (vii) changes in the competitive landscape.

Should one or more of the risks or uncertainties referenced above occur, or should underlying assumptions prove incorrect, actual results could differ materially from those expressed in the Financial Projections. Further, new factors could cause actual results to differ materially from those described in the Financial Projections, and it is not possible to predict all such factors, or the extent to which any such factor or combination of factors may cause actual results to differ from those contained in the Financial Projections.

For the avoidance of doubt, QCP has not prepared the Financial Projections or expressed a view on their accuracy or the basis on which they were prepared. The Financial Projections reflect neither the future business plans of QCP (regardless of whether the Alternative Transaction or the Original Transaction is consummated) nor changes to the business that might be implemented by QCP after the Effective Date if the Original Transaction (as opposed to the Alternative Transaction) were consummated, including asset dispositions, new investments and changes in operating performance.

II. Accounting Policies and Disclaimer

THESE FINANCIAL PROJECTIONS WERE NOT PREPARED WITH A VIEW TOWARD COMPLIANCE WITH PUBLISHED GUIDELINES OF THE UNITED STATES SECURITIES AND EXCHANGE COMMISSION OR THE AMERICAN INSTITUTE OF CERTIFIED PUBLIC ACCOUNTANTS FOR PREPARATION AND PRESENTATION OF PROSPECTIVE FINANCIAL INFORMATION. THE FINANCIAL PROJECTIONS DO NOT REFLECT THE FORMAL IMPLEMENTATION OF REORGANIZATION ACCOUNTING PURSUANT TO FINANCIAL ACCOUNTING STANDARDS BOARD ACCOUNTING STANDARDS CODIFICATION TOPIC 852, REORGANIZATIONS ("ASC 852"). OVERALL, THE IMPLEMENTATION OF ASC 852 IS NOT ANTICIPATED TO HAVE A MATERIAL IMPACT ON THE UNDERLYING ECONOMICS OF THE PLAN. THE FINANCIAL PROJECTIONS HAVE BEEN PREPARED USING GENERALLY ACCEPTED ACCOUNTING POLICIES ("GAAP") THAT ARE MATERIALLY CONSISTENT WITH THOSE APPLIED IN THE DEBTOR'S HISTORICAL FINANCIAL STATEMENTS. THE FINANCIAL PROJECTIONS HAVE NOT BEEN AUDITED OR REVIEWED BY A REGISTERED INDEPENDENT ACCOUNTING FIRM.

ALTHOUGH MANAGEMENT HAS PREPARED THE PROJECTIONS IN GOOD FAITH AND BELIEVES THE ASSUMPTIONS TO BE REASONABLE, THE DEBTOR AND THE REORGANIZED DEBTOR CAN PROVIDE NO ASSURANCE THAT SUCH ASSUMPTIONS WILL BE REALIZED. THE FINANCIAL PROJECTIONS HEREIN ARE NOT, AND MUST NOT BE VIEWED AS, A REPRESENTATION OF FACT, PREDICTION, OR GUARANTY OF THE COMPANY'S FUTURE PERFORMANCE. AS DESCRIBED IN DETAIL IN THE DISCLOSURE STATEMENT, A VARIETY OF RISK FACTORS COULD AFFECT THE REORGANIZED DEBTOR'S FINANCIAL RESULTS AND MUST BE CONSIDERED. ACCORDINGLY, ANY REVIEW OF THE PROJECTIONS SHOULD TAKE INTO ACCOUNT THE RISK FACTORS SET FORTH IN THE DISCLOSURE STATEMENT AND THE ASSUMPTIONS DESCRIBED HEREIN, INCLUDING ALL RELEVANT QUALIFICATIONS AND FOOTNOTES.

MOREOVER, THE PROJECTIONS CONTAIN CERTAIN STATEMENTS THAT ARE "FORWARD-LOOKING STATEMENTS" WITHIN THE MEANING OF THE PRIVATE SECURITIES LITIGATION REFORM ACT OF 1995. THESE STATEMENTS ARE SUBJECT TO A NUMBER OF ASSUMPTIONS, RISKS, AND UNCERTAINTIES, MANY OF WHICH ARE BEYOND THE CONTROL OF THE DEBTOR, INCLUDING THE IMPLEMENTATION OF THE PLAN, THE CONTINUING AVAILABILITY OF SUFFICIENT BORROWING CAPACITY OR OTHER FINANCING TO FUND OPERATIONS, ACHIEVING OPERATING EFFICIENCIES, EXISTING AND FUTURE GOVERNMENTAL REGULATIONS AND ACTIONS OF GOVERNMENTAL BODIES, INDUSTRY-SPECIFIC RISK FACTORS, AND OTHER MARKET AND COMPETITIVE CONDITIONS, INCLUDING WITHOUT LIMITATION THOSE SET FORTH HEREIN. HOLDERS OF CLAIMS AND INTERESTS ARE CAUTIONED THAT THE FORWARD-LOOKING STATEMENTS AS OF THE DATE MADE AND ARE NOT GUARANTEES OF FUTURE PERFORMANCE. ACTUAL RESULTS OR DEVELOPMENTS MAY DIFFER MATERIALLY FROM THE EXPECTATIONS EXPRESSED OR IMPLIED IN THE FORWARD-LOOKING STATEMENTS, AND THE DEBTOR UNDERTAKES NO OBLIGATION TO UPDATE ANY SUCH STATEMENTS. THE PROJECTIONS, WHILE PRESENTED WITH NUMERICAL SPECIFICITY, ARE NECESSARILY BASED ON A VARIETY OF ESTIMATES AND ASSUMPTIONS WHICH, THOUGH CONSIDERED REASONABLE BY THE DEBTOR, MAY NOT BE REALIZED AND ARE INHERENTLY SUBJECT TO SIGNIFICANT BUSINESS, ECONOMIC, INDUSTRY, REGULATORY, LEGAL, MARKET, AND FINANCIAL UNCERTAINTIES AND CONTINGENCIES, MANY OF WHICH ARE BEYOND THE REORGANIZED DEBTOR'S CONTROL. THE DEBTOR CAUTIONS THAT NO REPRESENTATIONS CAN BE MADE OR ARE MADE AS TO THE ACCURACY OF THE PROJECTIONS OR TO THE REORGANIZED DEBTOR'S ABILITY TO ACHIEVE THE PROJECTED RESULTS. SOME ASSUMPTIONS INEVITABLY WILL BE INCORRECT. MOREOVER, EVENTS AND CIRCUMSTANCES OCCURRING SUBSEQUENT TO THE DATE ON WHICH THE DEBTOR PREPARED THESE PROJECTIONS MAY BE DIFFERENT FROM THOSE ASSUMED, OR, ALTERNATIVELY, MAY HAVE BEEN UNANTICIPATED, AND THUS THE OCCURRENCE OF THESE EVENTS MAY AFFECT FINANCIAL RESULTS IN A MATERIALLY ADVERSE OR MATERIALLY BENEFICIAL MANNER. EXCEPT AS OTHERWISE PROVIDED IN THE PLAN OR THIS DISCLOSURE STATEMENT, THE DEBTOR AND REORGANIZED DEBTOR, AS APPLICABLE, DO NOT INTEND AND UNDERTAKE NO OBLIGATION TO UPDATE OR OTHERWISE REVISE THE PROJECTIONS TO REFLECT EVENTS OR CIRCUMSTANCES EXISTING OR ARISING AFTER THE DATE HEREOF OR TO REFLECT THE OCCURRENCE OF UNANTICIPATED EVENTS. THEREFORE, THE PROJECTIONS MAY NOT BE RELIED UPON AS A GUARANTEE OR OTHER ASSURANCE OF THE ACTUAL RESULTS THAT WILL OCCUR. IN DECIDING WHETHER TO VOTE TO ACCEPT OR REJECT THE PLAN, HOLDERS OF CLAIMS MUST MAKE THEIR OWN DETERMINATIONS AS TO THE REASONABLENESS OF SUCH ASSUMPTIONS AND THE RELIABILITY OF THE PROJECTIONS AND SHOULD CONSULT WITH THEIR OWN ADVISERS.

III. General Assumptions

a. Overview

The Debtor is a holding company that conducts business operations through more than 300 wholly owned or controlled subsidiaries, which, together with the Debtor, comprise the Company. The Company operates three business segments – the Long-Term Care Business, the 4H Business and the Outpatient Rehab and Other Businesses. The Long-Term Care Business is primarily conducted through non-debtor subsidiaries of non-debtor HCR III, an intermediate holding company and indirect subsidiary of the Debtor.

HCR III, which currently is subject to the MLSA, will be subject to terms of the Alternative Master Lease after the Effective Date. During the Projection Period, HCR III will be able to timely satisfy all of its financial obligations as they come due in the ordinary course. The accompanying projections provide a perspective of the earnings potential of the combined results of the RemainCo Long-Term Care Business, 4H Business and the Outpatient Rehab and Other Businesses. The RemainCo Long-Term Care Business provides skilled nursing and inpatient rehabilitation, memory care and assisted living services. The 4H Business provides Home Health and Hospice services. The Outpatient Rehab and Other Businesses provide comprehensive rehabilitation and various therapeutic services.

These Projections are intended to demonstrate the free cash flow generation of the Company, including the Debtor's non-debtor subsidiaries, after reflecting the transactions described in the Alternative Plan Sponsor Agreement. ProMedica Parent plans to significantly reduce leverage of the Debtor's subsidiaries through a series of transactions including the ProMedica Plan Contribution, which will be in excess of \$1 Billion. ProMedica Parent has also agreed to limit the amount of debt financing necessary to make this capital contribution to the Debtor to \$550 million. Furthermore, the Alternative Master Lease represents a more than 60% reduction in the rental expense as compared to the Original Master Lease Amendment. While there are risks inherent in these Projections as noted above, the Debtor along with ProMedica Parent and their respective advisors have reviewed these Projections and the corresponding results.

b. Methodology

In preparing the Projections, management considered the key revenue drivers including the Average Daily Rates ("ADRs") charged for the healthcare services provided by, and recent trends in the admissions, Average Length of Stay ("ALOS") and Average Daily Census ("ADC") of, the RemainCo Long-Term Care Business, the 4H Business and the Outpatient Rehab and Other Businesses. The key revenue drivers were combined with historical trends in operating costs, initiatives to reduce operating and non-operating costs, and non-operating aspects of the RemainCo Long-Term Care Business, the 4H Business and the Outpatient Rehab and Other Businesses assessed at a granular level.

Assumptions With Respect to the Projected Income Statement

a. Revenues

Revenues are derived by multiplying the ADRs and the projected patient days.

The ADRs are driven by reimbursement rates, which may vary depending on: the patients' conditions, the necessary healthcare services provided, and the patients' funding sources (including, but not limited to, Private Pay, Managed Care, Medicare and Medicaid).

Government agencies historically adjust rates on an annual basis. The rates charged to Private Pay and other funding sources may vary dependent on market influences. Due to the variance in reimbursement rates, the ADR is assessed based on the patients' expected funding source.

Projected patient days for the RemainCo Long-Term Care Business' Medicare and Managed Care funding sources are derived from patient admissions. ALOS is forecasted based on historical trends, adjusted to

reflect input from management on changes to population demographics and future patient choices regarding healthcare services. Projected patient days for the RemainCo Long-Term Care Business' other funding sources are derived from the ADC and are forecasted based on historical trends, adjusted to reflect input from management on changes to population demographics and future patient choices regarding healthcare services.

Projected patient days for the 4H Business and the Outpatient Rehab and Other Businesses are derived from the ADC and are forecast based on historical trends, adjusted to reflect input from management on changes to population demographics and future patient choices regarding healthcare services.

Participation rates in various public and private healthcare plans can influence the nature and duration of treatment and are forecast based on historical trends adjusted for management's input regarding changes in future patient choices.

The Outpatient Rehab and Other Businesses provide outpatient rehabilitation and ancillary healthcare services which are primarily charged at an hourly rate. The rate varies depending on healthcare service provided and is forecast based on historical trends.

b. Operating Expenses

The operating expenses for the RemainCo Long-Term Care Business, the 4H Business and the Outpatient Rehab and Other Businesses are primarily comprised of labor costs and other expenses associated with such businesses. These costs are based on historical trends adjusted for input from management regarding labor market influences during the Projection Period.

Nursing Expenses primarily represent the labor costs associated with the Nursing professionals who provide services for the RemainCo Long-Term Care Business, the 4H Business and the Outpatient Rehab and Other Businesses.

Ancillary Expenses represent supporting healthcare professionals, pharmaceutical and medical supplies.

Administrative Expenses primarily represent the labor costs associated with the administrative and management professionals supporting the RemainCo Long-Term Care Business, the 4H Business and the Outpatient Rehab and Other Businesses.

Other Expenses are comprised of costs associated with support staff and miscellaneous supplies necessary for the RemainCo Long-Term Care Business, the 4H Business and the Outpatient Rehab and Other Businesses

c. General and Administrative

General and Administrative Costs ("**G&A**") are primarily forecasted through labor costs and other expenses associated with the Company's corporate overhead. These costs are based on historical G&A trends, which have been adjusted for cost reduction efforts and the impact of operating the Reorganized Debtor post emergence.

d. Equity Earnings

The Equity Earnings represent the Debtor's share of the equity earnings in a joint venture with Omnicare, Inc. (a subsidiary of CVS Health Corporation), which provides pharmaceutical services to certain of the Debtor subsidiaries.

e. Capital Expenditures

The Capital Expenditures represent the Debtor's expected amount of funds needed for capital projects to repair and maintain their skilled nursing, memory care and assisted living facilities and maintain or upgrade their information technology operating infrastructure. These costs are based on historical trends, end-of-life estimates, and are adjusted for input from management during the Projection Period.

f. Interest Expense

The Interest expense is based on a \$550 million unsecured subordinated loan, the maximum allowable amount under the Alternative Plan Sponsor Agreement, at an annual interest rate of 5.0%.

g. Rent

The rent is based on the year 1 to 4 of the Alternative Master Lease between HCR III and the Lessors with year 1 of the lease starting on October 1, 2018.

<u>Financial Projections for the Combined RemainCo Long-Term Care Business, 4H Business and Outpatient Rehab and Other Business (Based Upon the Alternative Transaction)</u>

	Fiscal Periods							
USD Millions	20	018E (1)		2019E	2020E			2021E
Income Statement								
Revenue								
Medicare A	\$	308.4	\$	1,193.5	\$	1,244.7	\$	1,301.4
Managed Care		115.2		457.9		491.0		526.8
Private		32.6		131.1		133.7		137.0
Assisted Living		58.2		234.2		242.3		250.6
Hospice & VA		16.0		65.4		66.7		68.4
Medicaid		228.7		943.6		981.2		1,025.2
Medicare B		4.9		19.7		20.1		20.6
Other Revenue		22.7		90.4		91.4		92.3
Total Revenues	\$	786.5	\$	3,135.9	\$	3,271.1	\$	3,422.2
Operating Expenses								
Nursing	\$	244.2	\$	993.4	\$	1,039.0	\$	1,086.9
Ancilliary		107.0		441.3		464.2		488.5
Dietary		38.6		158.6		165.8		173.3
Utilities		11.9		49.5		51.7		54.0
Maintenance		9.5		39.0		40.7		42.4
Laundry		4.2		17.3		18.1		18.9
Housekeeping		12.3		50.6		52.9		55.3
Activities		18.4		75.8		79.1		82.6
Administrative		133.9		524.1		546.2		569.4
Other Expenses		50.3		203.1		210.1		217.3
Total Operating Expenses	\$	630.4	\$	2,552.7	\$	2,667.8	\$	2,788.6
Total Contribution Margin	\$	156.1	\$	583.1	\$	603.3	\$	633.7
Other Operating Expenses								
Insurance	\$	17.2	\$	71.5	\$	73.2	\$	75.0
Non-Income Taxes		8.5		36.1		37.0		37.9
Lease Expenses		3.1		12.8		13.0		13.3
Total Other Operating Expenses	\$	28.9	\$	120.4	\$	123.3	\$	126.2
Total Operating Income	\$	127.2	\$	462.8	\$	480.1	\$	507.4
01%		16.2%		14.8%		14.7%		14.8%
General & Administrative Expenses	\$	34.9	\$	101.3	\$	103.8	\$	106.4
Consolidated Earnings Before Interest, Tax,		\$92.3		\$361.5		\$376.2		\$401.0
Depreciation, Amortization & Rent								
Other Adjustments	÷	2.4	ć	0.4	ć	0.0	ċ	0.3
Equity Earnings from JV Capital Expenditures	\$	2.1 (13.8)	Ģ	8.4 (55.0)	Ş	8.8 (57.4)	Ş	9.3 (60.0)
Interest Expense		(6.9)		(27.5)		(27.5)		(27.5)
Alternative Master Lease Rent		(44.8)		(179.9)		(183.0)		(188.0)
Total Non Operating Expenses (Income)	\$	(63.4)	\$	(254.0)	\$	(259.0)	\$	(266.3)
Cash Generation Prior to Working Capital Adjustments	\$	28.9	\$	107.5	\$	117.2	\$	134.7
rajastricita								

EXHIBIT E

4H GLOBAL FINANCIAL PROJECTIONS

<u>Financial Information and Projections – Assumptions Based Upon the Original Transaction</u>

I. Financial Projections

For purposes of demonstrating feasibility of the Original Transaction embodied in the Plan, the Debtor, with the assistance of its advisors, has prepared forecasted, post-Effective Date income statements (the "4H Financial Projections" or the "4H Projections") for annual periods ending December 31, 2018 (fiscal year 2018) through December 31, 2021 (fiscal year 2021) (the "Projection Period"). The 4H Financial Projections represent the consolidated forecasted projections of the 4H Business and the Outpatient Rehab and Other Businesses. These 4H Projections have been prepared based on a number of assumptions made by the Debtor's management, in consultation with the Debtor's financial advisors, as to the future performance of the Reorganized Debtor following the consummation of the Original Transaction, and reflect management's judgement and expectations regarding its future operations and financial position.

The 4H Financial Projections are subject to inherent risks and uncertainties, several of which are difficult to predict and many of which are beyond management's control. Factors that may cause actual results to differ from expected results include, but are not limited to:

- (i) changes in the reimbursement rates or methods or timing of payment from third party payors, including the Medicare and Medicaid programs, or the implementation of other measures to reduce reimbursement for the 4H Business and the Outpatient Rehab and Other Businesses' services and products;
- (ii) changes in the acuity of patients, as well as payor mix and payment methodologies;
- (iii) changes in existing or future laws and regulations or the enforcement of these laws and regulations, including but not limited to future adjustments in healthcare regulation;
- (iv) further consolidation of Managed Care organizations and other third party payors; and
- (v) the uncertainty inherent in estimating changes in population demographics and healthcare service requirements.

Should one or more of the risks or uncertainties referenced above occur, or should underlying assumptions prove incorrect, actual results could differ materially from those expressed in the 4H Financial Projections. Further, new factors could cause actual results to differ materially from those described in the 4H Financial Projections, and it is not possible to predict all such factors, or the extent to which any such factor or combination of factors may cause actual results to differ from those contained in the 4H Financial Projections.

In addition, QCP, who will control the Reorganized Debtor if the Original Transaction closes, has not prepared the 4H Financial Projections or expressed a view on their accuracy or the basis on which they were prepared. The 4H Financial Projections reflect neither the future business plans of QCP nor changes to the business that may be implemented by QCP after the Effective Date if the Original Transaction closes, including asset dispositions, new investments and changes in operating performance.

II. Accounting Policies and Disclaimer

THESE 4H FINANCIAL PROJECTIONS WERE NOT PREPARED WITH A VIEW TOWARD COMPLIANCE WITH PUBLISHED GUIDELINES OF THE UNITED STATES SECURITIES AND EXCHANGE COMMISSION OR THE AMERICAN INSTITUTE OF CERTIFIED PUBLIC ACCOUNTANTS FOR PREPARATION AND PRESENTATION OF PROSPECTIVE FINANCIAL INFORMATION. THE 4H FINANCIAL PROJECTIONS DO NOT REFLECT THE FORMAL IMPLEMENTATION OF REORGANIZATION ACCOUNTING PURSUANT TO FINANCIAL ACCOUNTING STANDARDS BOARD ACCOUNTING STANDARDS CODIFICATION TOPIC 852, REORGANIZATIONS ("ASC 852"). OVERALL, THE IMPLEMENTATION OF ASC 852 IS NOT ANTICIPATED TO HAVE A MATERIAL IMPACT ON THE UNDERLYING ECONOMICS OF THE PLAN. THE 4H FINANCIAL PROJECTIONS HAVE BEEN PREPARED USING GENERALLY ACCEPTED ACCOUNTING POLICIES ("GAAP") THAT ARE MATERIALLY CONSISTENT WITH THOSE APPLIED IN THE DEBTOR'S HISTORICAL FINANCIAL STATEMENTS. THE 4H FINANCIAL PROJECTIONS HAVE NOT BEEN AUDITED OR REVIEWED BY A REGISTERED INDEPENDENT ACCOUNTING FIRM.

ALTHOUGH MANAGEMENT HAS PREPARED THE 4H PROJECTIONS IN GOOD FAITH AND BELIEVES THE ASSUMPTIONS TO BE REASONABLE, THE DEBTOR AND THE REORGANIZED DEBTOR CAN PROVIDE NO ASSURANCE THAT SUCH ASSUMPTIONS WILL BE REALIZED. THE 4H FINANCIAL PROJECTIONS HEREIN ARE NOT, AND MUST NOT BE VIEWED AS, A REPRESENTATION OF FACT, PREDICTION, OR GUARANTY OF THE COMPANY'S FUTURE PERFORMANCE. AS DESCRIBED IN DETAIL IN THE DISCLOSURE STATEMENT, A VARIETY OF RISK FACTORS COULD AFFECT THE REORGANIZED DEBTOR'S FINANCIAL RESULTS AND MUST BE CONSIDERED. ACCORDINGLY, ANY REVIEW OF THE 4H PROJECTIONS SHOULD TAKE INTO ACCOUNT THE RISK FACTORS SET FORTH IN THE DISCLOSURE STATEMENT AND THE ASSUMPTIONS DESCRIBED HEREIN, INCLUDING ALL RELEVANT QUALIFICATIONS AND FOOTNOTES.

MOREOVER, THE 4H PROJECTIONS CONTAIN CERTAIN STATEMENTS THAT ARE "FORWARD-LOOKING STATEMENTS" WITHIN THE MEANING OF THE PRIVATE SECURITIES LITIGATION REFORM ACT OF 1995. THESE STATEMENTS ARE SUBJECT TO A NUMBER OF ASSUMPTIONS, RISKS, AND UNCERTAINTIES, MANY OF WHICH ARE BEYOND THE CONTROL OF THE DEBTOR, INCLUDING THE IMPLEMENTATION OF THE PLAN AND THE CONSUMMATION OF THE ORIGINAL TRANSACTION, THE CONTINUING AVAILABILITY OF SUFFICIENT BORROWING CAPACITY OR OTHER FINANCING TO FUND OPERATIONS, ACHIEVING OPERATING EFFICIENCIES, EXISTING AND FUTURE GOVERNMENTAL REGULATIONS AND ACTIONS OF GOVERNMENTAL BODIES, INDUSTRY-SPECIFIC RISK FACTORS, AND OTHER MARKET AND COMPETITIVE CONDITIONS, INCLUDING WITHOUT LIMITATION THOSE SET FORTH HEREIN. HOLDERS OF CLAIMS AND INTERESTS ARE CAUTIONED THAT THE FORWARD-LOOKING STATEMENTS AS OF THE DATE MADE AND ARE NOT GUARANTEES OF FUTURE PERFORMANCE. ACTUAL RESULTS OR DEVELOPMENTS MAY DIFFER MATERIALLY FROM THE EXPECTATIONS EXPRESSED OR IMPLIED IN THE FORWARD-LOOKING STATEMENTS, AND THE DEBTOR UNDERTAKES NO OBLIGATION TO UPDATE ANY SUCH STATEMENTS. THE 4H PROJECTIONS, WHILE PRESENTED WITH NUMERICAL SPECIFICITY, ARE NECESSARILY BASED ON A VARIETY OF ESTIMATES AND ASSUMPTIONS WHICH, THOUGH CONSIDERED REASONABLE BY THE DEBTOR, MAY NOT BE REALIZED AND ARE INHERENTLY SUBJECT TO SIGNIFICANT BUSINESS, ECONOMIC, INDUSTRY, REGULATORY, LEGAL, MARKET, AND FINANCIAL UNCERTAINTIES AND CONTINGENCIES, MANY OF WHICH ARE BEYOND THE REORGANIZED DEBTOR'S CONTROL. THE DEBTOR CAUTIONS THAT NO

REPRESENTATIONS CAN BE MADE OR ARE MADE AS TO THE ACCURACY OF THE 4H PROJECTIONS OR TO THE REORGANIZED DEBTOR'S ABILITY TO ACHIEVE THE PROJECTED RESULTS. SOME ASSUMPTIONS INEVITABLY WILL BE INCORRECT. MOREOVER, EVENTS AND CIRCUMSTANCES OCCURRING SUBSEQUENT TO THE DATE ON WHICH THE DEBTOR PREPARED THESE 4H PROJECTIONS MAY BE DIFFERENT FROM THOSE ASSUMED, OR, ALTERNATIVELY, MAY HAVE BEEN UNANTICIPATED, AND THUS THE OCCURRENCE OF THESE EVENTS MAY AFFECT FINANCIAL RESULTS IN A MATERIALLY ADVERSE OR MATERIALLY BENEFICIAL MANNER. EXCEPT AS OTHERWISE PROVIDED IN THE PLAN OR THIS DISCLOSURE STATEMENT, THE DEBTOR AND REORGANIZED DEBTOR, AS APPLICABLE, DO NOT INTEND AND UNDERTAKE NO OBLIGATION TO UPDATE OR OTHERWISE REVISE THE 4H PROJECTIONS TO REFLECT EVENTS OR CIRCUMSTANCES EXISTING OR ARISING AFTER THE DATE HEREOF OR TO REFLECT THE OCCURRENCE OF UNANTICIPATED EVENTS. THEREFORE, THE 4H PROJECTIONS MAY NOT BE RELIED UPON AS A GUARANTEE OR OTHER ASSURANCE OF THE ACTUAL RESULTS THAT WILL OCCUR. IN DECIDING WHETHER TO VOTE TO ACCEPT OR REJECT THE PLAN, HOLDERS OF CLAIMS AND INTERESTS MUST MAKE THEIR OWN DETERMINATIONS AS TO THE REASONABLENESS OF SUCH ASSUMPTIONS AND THE RELIABILITY OF THE 4H PROJECTIONS AND SHOULD CONSULT WITH THEIR OWN ADVISERS.

III. General Assumptions

a. Overview

The Debtor is a holding company that conducts business operations through more than 300 wholly owned or controlled subsidiaries, which, together with the Debtor, comprise the Company. The Company operates three business segments – the Long-Term Care Business, the 4H Business and the Outpatient Rehab and Other Businesses. The Long-Term Care Business is primarily conducted through non-debtor subsidiaries of non-debtor HCR III, an intermediate holding company and indirect subsidiary of the Debtor. If the Original Transaction closes, HCR III, which currently is subject to the MLSA, will be subject to terms of the Original Master Lease after the Effective Date. During the Projection Period, HCR III will be able to timely satisfy all of its financial obligations as they come due in the ordinary course. The accompanying projections provide a perspective of the earnings potential of the combined results of the 4H Business and the Outpatient Rehab and Other Businesses. The 4H Business provides Home Health and Hospice Services. The Outpatient Rehab and Other Businesses provide comprehensive rehabilitation and various therapeutic services.

b. Methodology

In preparing the 4H Projections, management considered the key revenue drivers including the Average Daily Rates ("ADRs") charged for the healthcare services provided by, and recent trends in the Average Daily Census ("ADC") of, the 4H Business and the Outpatient Rehab and Other Businesses. The key revenue drivers were combined with historical trends in operating costs, initiatives to reduce operating and non-operating costs, and non-operating aspects of the 4H Business and the Outpatient Rehab and Other Businesses assessed at a granular level.

Assumptions With Respect to the Projected Income Statement

a. Revenues

Revenues are derived by multiplying the ADRs and the projected patient days.

The ADRs are driven by reimbursement rates, these vary depending on the patients' conditions, the necessary healthcare services provided and the patients' funding sources (including, but not limited to, Private Pay, Managed Care, Medicare and Medicaid).

Government agencies historically adjust rates on an annual basis. The rates charged to Private Pay and other funding sources may vary dependent on market influences. Due to the variance in reimbursement rates, the ADR is assessed based on the patients' expected funding source.

Projected patient days for the 4H Business and the Outpatient Rehab and Other Businesses are derived from the ADC and are forecast based on historical trends, adjusted to reflect input from management on changes to population demographics and future patient choices regarding healthcare services.

Participation rates in various public and private healthcare plans can influence the nature and duration of treatment and are forecast based on historical trends adjusted for management's input regarding changes in future patient choices.

The Outpatient Rehab and Other Businesses provide outpatient rehabilitation and ancillary healthcare services which are primarily charged at an hourly rate. The rate varies depending on healthcare service provided and is forecast based on historical trends.

b. Operating Expenses

The operating expenses for the 4H Business and the Outpatient Rehab and Other Businesses are primarily comprised of labor costs and other expenses associated with such businesses. These costs are based on historical trends adjusted for input from management regarding labor market influences during the Projection Period.

Nursing Expenses primarily represent the labor costs associated with the Nursing professionals who provide services for the 4H Business and the Outpatient Rehab and Other Businesses.

Ancillary Expenses represent supporting healthcare professionals, pharmaceutical and medical supplies.

Administrative Expenses primarily represent the labor costs associated with the administrative and management professionals supporting the 4H Business and the Outpatient Rehab and Other Businesses.

Other Expenses are comprised of costs associated with support staff and miscellaneous supplies necessary for the 4H Business and the Outpatient Rehab and Other Businesses

c. General and Administrative

General and Administrative Costs ("**G&A**") are primarily forecasted through an allocation of labor costs and other expenses associated with the Company's corporate overhead. The allocation rate is based on historical G&A costs, adjusted for cost reduction efforts and the impact of operating the Reorganized Debtor post emergence.

d. Equity Earnings

The Equity Earnings represent the Debtor's share of the equity earnings in a joint venture with Omnicare, Inc. (a subsidiary of CVS Health Corporation), which provides pharmaceutical services to certain of the Debtor subsidiaries.

<u>Financial Projections for the Combined 4H Business and Outpatient Rehab and Other Business</u> (Based Upon the Original Transaction)

	Fiscal Periods								
USD Millions	2018E (1)			2019E		2020E		2021E	
Income Statement									
Revenue									
Medicare A	\$	598.0	\$	630.2	\$	662.1	\$	691.7	
Managed Care		37.7		39.1		40.9		43.3	
Private		6.3		6.5		6.8		7.0	
Assisted Living		0.0		0.0		0.0		0.0	
Hospice & VA		2.3		2.3		2.4		2.5	
Medicaid		44.2		46.3		48.2		50.4	
Medicare B		0.8		0.8		0.8		0.8	
Other Revenue		78.8		80.2		81.4		82.6	
Total Revenues	\$	768.1	\$	805.4	\$	842.6	\$	878.3	
Operating Expenses									
Nursing	\$	169.7	\$	175.2	\$	182.6	\$	190.2	
Ancilliary		65.4		66.2		68.9		71.6	
Dietary		5.4		5.5		5.6		5.8	
Utilities		2.0		2.0		2.0		2.1	
Maintenance		1.0		1.1		1.1		1.1	
Laundry		0.6		0.6		0.6		0.6	
Housekeeping		1.7		1.7		1.7		1.8	
Activities		43.7		50.0		52.2		54.5	
Administrative		172.3		174.9		182.5		190.4	
Other Expenses		144.5		153.5		158.1		162.9	
Total Operating Expenses	\$	606.3	\$	630.6	\$	655.4	\$	681.1	
Total Contribution Margin	\$	161.8	\$	174.8	\$	187.2	\$	197.2	
Other Operating Expenses									
Insurance	\$	3.9	\$	3.9	\$	4.0	\$	4.1	
Non-Income Taxes		2.7		2.9		2.9		3.0	
Lease Expenses		12.4		12.7		12.9		13.2	
Total Other Operating Expenses	\$	18.9	\$	19.4	\$	19.8	\$	20.3	
Total Operating Income	\$	143.0	\$	155.4	\$	167.4	\$	177.0	
01%		18.6%		19.3%		19.9%		20.1%	
Allocated General & Administrative Costs	\$	32.5	\$	31.8	\$	32.6	\$	34.2	
Consolidated Earnings Before Interest, Tax, Depreciation and Amortization		\$110.5		\$123.6		\$134.7		\$142.8	
Other Income and (Expenses)									
Equity Earnings from JV	\$	8.6		8.4		8.4	\$	8.4	
Capital Expenditure		(6.9)		(7.9)		(9.1)		(10.5	
Cash Generation Prior To Working Capital Adjustments and Interest	\$	112.1	\$	124.1	\$	134.0	\$	140.7	

⁽¹⁾ Fiscal Period 2018 includes provisional unaudited financial information for January 2018 together with Financial Projections for February 2018 to December 2018